

# NOTICE OF THE UNSECURED CREDITORS (INCLUDING UNSECURED NON-CONVERTIBLE DEBENTURE HOLDERS) OF TATA REALTY AND INFRASTRUCTURE LIMITED (CONVENED PURSUANT TO THE DIRECTIONS OF THE HON'BLE NATIONAL COMPANY LAW TRIBUNAL, MUMBAI BENCH)

Day	Monday
Date	July 26, 2021
Time	12.00 Noon (IST)
Mode of	In view of the prevailing Covid-19 pandemic and related social distancing
Meeting	norms, the Hon'ble National Company Law Tribunal, Mumbai Bench has
	directed that the meeting of the Unsecured Creditors of Tata Realty and
	Infrastructure Limited be held through video conferencing or other audio-
	visual means.

Sr. No.	Contents	Page No.
1)	Notice of the Meeting of the Unsecured Creditors (Including Unsecured Non-Convertible Debenture Holders) of Tata Realty and Infrastructure Limited in terms of provisions of Sections 230 to 232 of the Companies Act, 2013 read with Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 (as amended) convened pursuant to the directions issued by the Hon'ble National Company Law Tribunal, Mumbai Bench.	01-07
2)	Explanatory Statement pursuant to Section 230(3) and Section 102 of the Companies Act, 2013 read with Rule 6(3) of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 (as amended) as Annexure A.	08-27
3)	Amended Scheme of Amalgamation of MIA Infrastructure Private Limited and Wellkept Facility Management Services Private Limited and Acme Living Solutions Private Limited and TRIF Gurgaon Housing Projects Private Limited ("Transferor Companies") with Tata Realty and Infrastructure Limited ("Transferee Company") and their respective shareholders and creditors under Sections 230 to 232 of the Companies Act, 2013 (as amended) as Annexure B.	28-55
4)	Audited Financial Statements of Transferor Companies for the year ended 31 <sup>st</sup> March, 2019 and 31 <sup>st</sup> March, 2020 as Annexure C.	56-200
5)	Audited Financial Statement of Transferee Company for the year ended 31 <sup>st</sup> March, 2019 and 31 <sup>st</sup> March, 2020 as Annexure D.	201-435

#### TATA REALTY AND INFRASTRUCTURE LIMITED

CIN: U70102MH2007PLC168300 E Block, Voltas Premises, T. B. Kadam Marg, Chinchpokli, Mumbai – 400 033 India. Tel. 91 22 6661 4444 Fax: 91 22 6661 4452 Website: www.tril.co.in

# BEFORE THE NATIONAL COMPANY LAW TRIBUNAL, MUMBAI BENCH CA (CAA) No. 702/MB/2019

FORM NO. CAA. 2

[Pursuant to Section 230(5) of the Companies Act, 2013 and Rule 8 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016]

In the matter of Companies Act, 2013;

And

In the matter of Application under Sections 230 to 232 of the Companies Act, 2013 read with the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 (as amended);

#### AND

In the matter of Scheme of Amalgamation of

- 1. MIA Infrastructure Private Limited And
- 2. Wellkept Facility Management Services Private Limited And
- 3. Acme Living Solutions Private Limited And
- 4. TRIF Gurgaon Housing Projects Private Limited, .....the Transferor Companies

With

Tata Realty and Infrastructure Limited .....the Transferee Company

# Tata Realty and Infrastructure Limited ]A Company incorporated under the ]Companies Act, 1956 and a public limited companywithin the meaning of the Companies Act, 2013 and having its ]Registered Office at E Block, Voltas ]Premises, T B Kadam Marg, Chinchpokli, ]Mumbai – 400033. ]...... Third Applicant Company

#### NOTICE OF THE UNSECURED CREDITORS (INCLUDING UNSECURED NON-CONVERTIBLE DEBENTURE HOLDERS) OF TATA REALTY AND INFRASTRUCTURE LIMITED (CONVENED PURSUANT TO THE DIRECTIONS OF THE HON'BLE NATIONAL COMPANY LAW TRIBUNAL, MUMBAI BENCH)

#### To,

The Unsecured Creditors (including Unsecured Non-Convertible Debenture holders) of Tata Realty and Infrastructure Limited ("Third Applicant Company" or "the Company" or "Transferee Company")

Notice is hereby given that by order dated the 12<sup>th</sup> March, 2020 and 2<sup>nd</sup> November, 2020 ("Said Orders"), the Hon'ble National Company Law Tribunal, Mumbai Bench has directed a meeting, *inter alia*, of the Unsecured Creditors (including Unsecured Non-Convertible Debenture holders) of Tata Realty and Infrastructure Limited ("Tribunal Convened Meeting"), the Applicant Company, to be held for the purpose of considering, and if thought fit, approving, with or without modification, the proposed amended Scheme of Amalgamation proposed to be made between the Transferor Company and Transferee Company and their respective shareholders and creditors, Under Sections 230 to 232 read with other applicable provisions of the Companies Act, 2013 (as amended) ('the Act') and the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 (as amended) (' the Rules').

In pursuance of the said orders and as directed therein, further notice is hereby given that a Tribunal Convened Meeting of the Unsecured Creditors (including Unsecured Non-Convertible Debenture holders) of the Transferee Company will be held through Video Conferencing or Other Audio-Visual Means ('VC or OAVM') on Monday, July 26, 2021 at 12.00 Noon, to consider, and, if thought fit, to pass the following Resolution for approval of the Scheme by requisite majority as prescribed under Section 230(1) read with Section 232(1) of the Act.

Copies of the said amended Scheme of Amalgamation, Explanatory Statement pursuant to Sections 102, 230(3) and 232(2) of the Act and Rule 6 of the Rules (as amended) in respect of business as set out below are annexed to this notice. A copy of this Notice together with the relevant documents will be made available on the website of the Company at www.tril.co.in under Investor Info and also on the website of the stock exchange, i.e., BSE Limited at www.bseindia.com, where Non-Convertible Debt Instruments of the Company are listed and also can be obtained free of charge at the registered office of the company or at the office of its advocates, Mr. Shankarlal Raheja, at D-87-88, Mulund Darshan CHS. Ltd., Near Jai Bharat High School, Mulund Colony, Mulund West, Mumbai – 400082.

The Hon'ble National Company Law Tribunal, Mumbai Bench has appointed Mr. Sanjay Dutt (DIN: 05251670), Managing Director & CEO of the Company, and failing him, Mr. Farokh Subedar (DIN: 00028428), Director, as Chairperson and Mr. D. A. Kamat, Proprietor of M/s. D. A. Kamat & Co. Practicing Company Secretary (ICSI Membership No: 3843, COP: 4965) as the Scrutinizer for the Tribunal Convened Meeting. The above-mentioned scheme of Amalgamation, if approved by the meeting, will be subject to the subsequent approval of the tribunal. The Authorised Representatives of the Company shall also attend the Tribunal Convened Meeting.

The Board of Directors of the Company, at its meeting held on 19<sup>th</sup> day of November, 2018, approved the Scheme of Amalgamation in the nature of Merger and subsequently at its meeting held on 13<sup>th</sup> January, 2021, approved the amended Scheme of Amalgamation, subject to the approval of the Equity Shareholders and Creditors of the company, as may be required, and subject to approval of the Hon'ble National Company Law Tribunal, Mumbai Bench.

TAKE NOTICE that the following Resolutions are proposed under Section 230(3) and other applicable provisions of the Act (including any statutory modification(s) or re-enactment thereof for the time being in force), and the provisions of the Memorandum of Association and Articles of Association of the Company, for the purpose of considering, and, if thought ft, approving, the amended Scheme:

"RESOLVED THAT pursuant to the provisions of Sections 230 to 232 of the Companies Act, 2013 ('the Act') read with Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 and other applicable provisions, if any, of the Act, read with National Company Law Tribunal Rules, 2016, Circulars, Notifications, Clarifications, issued thereunder (including any statutory modification(s) or re-enactment thereof for the time being in force) as may be applicable and subject to the enabling provisions in the Memorandum and Articles of Association of the Company and subject to the approval of the Hon'ble National Company Law Tribunal ("NCLT"), Mumbai Bench and New Delhi Bench and such other approvals, permissions and sanctions of regulatory or Governmental and other authorities or Tribunal, as may be necessary, and subject to such conditions and modification(s) as may be prescribed or imposed by the Mumbai Bench and New Delhi Bench of the NCLT, or by any regulatory or other authorities or tribunal, while granting such consents, approvals and permissions, which may be agreed to by the Board of Directors of the Company, the proposed amended Scheme of Amalgamation of MIA Infrastructure Private Limited and Wellkept Facility Management Services Private Limited and Acme Living Solutions Private Limited and TRIF Gurgaon Housing Projects Private Limited with Tata Realty and Infrastructure Limited and their respective shareholders and creditors, placed before this meeting and initialed by the Chairperson of the meeting for the purpose of identification, be and is hereby approved with or without modification and for conditions, if any, which may be required and/or imposed and/or permitted by the Mumbai Bench and New Delhi Bench of the National Company Law Tribunal while sanctioning the Scheme and/or by any Governmental authority.

**RESOLVED FURTHER THAT** for the purpose of giving effect to the above resolutions and for removal of any difficulties or doubts, the Board of Directors of the Company (hereinafter referred to as "the Board", which term shall deem to include any committee or any person(s) which the Board may nominate or constitute or delegate) to exercise its powers, including the powers conferred under above resolutions, be and is hereby authorised to do all such acts, deeds, matters and things as it may, in its absolute discretion, deem necessary, expedient, usual or proper and to settle any questions or difficulties that may arise with regard to the implementation of the above resolution, including passing of such accounting entries and/or making such adjustments in the books of accounts as considered necessary to give effect to the above resolutions or to carry out such modifications / directions as may be ordered by the Mumbai Bench and New Delhi Bench of the National Company Law Tribunal to implement the aforesaid resolution."

Sanjay Dutt DIN: 05251670 Chairman appointed for the aforesaid Tribunal Convened Meeting

Date: June 21, 2021 Place: Mumbai

#### Notes:

- In view of the global outbreak of the Covid-19 pandemic and related social distancing norms, 1. to be followed and pursuant to the orders dated March 12, 2020 and November 2, 2020, in Company Application No. CA (CAA) No. 702/MB/2019 passed by the Hon'ble National Company Law Tribunal, Mumbai Bench ("NCLT"), the meeting of the Unsecured Creditors (including unsecured debenture holders) of Tata Realty and Infrastructure Limited ("Tribunal Convened Meeting") is being convened on Monday, July 26, 2021 at 12:00 Noon (IST) through Video Conferencing ("VC") or Other Audio Visual Means ("OAVM") without the physical presence of the Unsecured Creditors (including unsecured debenture holders) at a common venue, as per applicable procedure mentioned in the General Circular No. 14/ 2020 dated April 8, 2020, General Circular No. 17/ 2020 dated April 13, 2020, General Circular No. 20/2020 dated May 5, 2020, (collectively referred to as "MCA Circulars"), for the purpose of considering, and if thought fit, approving, with or without modification(s), the Scheme of Amalgamation between MIA Infrastructure Private Limited, Wellkept Facility Management Services Private Limited, Acme Living Solutions Private Limited and TRIF Gurgaon Housing Projects Private Limited ("Transferors Company") and Tata Realty and Infrastructure Limited ("Transferee Company") and their respective shareholders and creditors on a going concern basis, under the provisions of Sections 230 to 232 of the Companies Act, 2013 ("the Act") and other relevant provisions of the Act and rules framed thereunder ("Scheme").
- 2. The Explanatory Statement, in terms of requirement of provisions of under Sections 230, 232 and 102 of the Companies Act, 2013 read with Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 (as amended) to the Tribunal Convened Meeting, is annexed with the Notice.
- 3. The voting rights of Unsecured Creditors (including unsecured debenture holders) shall be in proportion to their outstanding amount as per the books of accounts of the Transferee Company as on February 28, 2021, being the cut-off date ("Cut-off Date").
- 4. Since this Tribunal Convened Meeting is being held in compliance with the MCA Circulars through VC or OAVM, therefore the requirement of physical attendance of the Unsecured Creditors (including unsecured debenture holders) has been dispensed with. Accordingly, in terms of MCA Circulars, the facility for appointment of proxies by the Unsecured Creditors (including unsecured debenture holder) under Sections 113 and 230 and any other applicable sections of the Act, will not be available for this Tribunal Convened Meeting and hence the Proxy Form, Attendance Slip and Route Map are not annexed to this Notice.

However, the authorised representatives of the Unsecured Creditors (including unsecured debenture holders) may be appointed to attend the Tribunal Convened Meeting and may cast their votes during the Tribunal Convened Meeting, are required to send a scanned copy (PDF / JPG format) of the certified copy of Board or governing body Resolution / Authorization Letter / Power of Attorney issued by the Board of Directors along with a specimen signature of the duly authorized signatory(ies) who are authorized to vote, to the Company at trilsec@tatarealty.in, authorising them to attend and vote through VC or OAVM on their behalf at the Tribunal Convened Meeting.

5. The Tribunal has appointed Mr. Sanjay Dutt (DIN: 05251670), Managing Director & CEO of the Transferee Company and failing him, Mr. Farokh Subedar (DIN: 00028428), Director to act as the Chairman of the Tribunal Convened Meeting including any adjournment(s) thereof.

- 6. The Unsecured Creditors (including unsecured debenture holders) can join this Tribunal Convened Meeting through VC or OAVM mode 15 minutes before the scheduled time of the commencement of the Meeting by following the procedure mentioned hereinbelow.
- 7. Necessary link to attend the Tribunal Convened Meeting of the Company shall be provided through e-mail at the registered e-mail address of the Unsecured Creditors (including unsecured debenture holders) / at the e-mail address of the Authorized Representative updated with the Company, as the case may be. The said link would facilitate them to attend the Tribunal Convened Meeting through VC or OAVM. The Unsecured Creditors (including unsecured debenture holders) will be required to use Internet with a good speed to avoid any disturbance during the Tribunal Convened Meeting.
- 8. In terms of the directions contained in the Order dated March 12, 2020, the quorum for the Tribunal Convened Meeting shall be in terms of Section 103 of the Act. Accordingly, the attendance of the Unsecured Creditors (including unsecured debenture holders) attending the Tribunal Convened Meeting through VC or OAVM will be counted for the purpose of reckoning the quorum under Section 103 of the Act.

The Chairman may decide to conduct vote on proposed Resolution by show of hands, unless demand for a poll is made by any Unsecured Creditors (including unsecured debenture holders) in accordance with Section 109 of the Act.

- 9. In terms of Sections 230 to 232 of the Act, the Scheme shall be considered approved by the unsecured creditors of the Company, if the resolution mentioned above in the Notice has been approved at the Meeting by a majority of unsecured creditors with three fourth of the value in rupees.
- 10. In line with the MCA Circulars, the Notice of the Tribunal Convened Meeting together with relevant documents are being sent only through electronic mode to those Unsecured Creditors (including unsecured debenture holders), whose name is appearing in the list of Creditors of the Company as on February 28, 2021 and whose e-mail addresses are also registered with the Company or Depository Participants (DPs).

Any person / entity who is not an Unsecured Creditors (including unsecured debenture holders) as on the date referred above, should treat this notice for information purposes only and will not be entitled to attend and/or vote at the Tribunal Convened Meeting.

The Unsecured Creditors (including unsecured debenture holders) may note that this Notice together with the relevant documents will be made available on the website of the Company at <u>www.tril.co.in</u> under Investor Info and also on the website of the stock exchange, i.e., BSE Limited at <u>www.bseindia.com</u>, where Non-Convertible Debt Instruments of the Company are listed.

- 11. The Unsecured Creditors (including unsecured debenture holder), who need assistance in connection with using the technology before or during the Tribunal Convened Meeting, may reach out to the Company officials at e-mail id: <u>trilsec@tatarealty.in</u> or Mobile: +91 8898778859.
- 12. The Unsecured Creditors (including unsecured debenture holder), who would like to express their views or ask questions with regard to the resolution to be placed at the Tribunal Convened Meeting, may raise the same at the meeting or send an e-mail to trilsec@tatarealty.in in advance (mentioning their name, type of creditor, demat account number, PAN, vendor code and other credentials, if any), at least 3 days prior to the date of the Tribunal Convened Meeting. Such questions by the Unsecured Creditors (including

unsecured debenture holder) shall be taken up during the Tribunal Convened Meeting and replied by the Company suitably.

- 13. All documents referred to in the Notice and the Explanatory Statement shall be made available for inspection by the Unsecured Creditors (including unsecured debenture holders) of the Company, without payment of fees upto and including the date of Tribunal Convened Meeting. The Unsecured Creditors (including unsecured debenture holder), who wish to inspect the relevant documents referred to in the Notice can send an e-mail to trilsec@tatarealty.in from their registered e-mail addresses (mentioning their name, type of creditor, demat account number, PAN, vendor code and other credentials, if any) up to the conclusion of this Tribunal Convened Meeting.
- 14. The Tribunal has appointed M/s. D. A. Kamat & Co, Practicing Company Secretary, as the Scrutinizer to scrutinize the voting process undertaken by the Chairman at the Tribunal Convened Meeting. The Scrutinizer will submit his report to the Chairman of the Tribunal Convened Meeting after completion of the scrutiny of the votes cast by the Unsecured Creditors (including unsecured debenture holders) of the Company, in a fair and transparent manner. The Scrutinizers decision on the validity of the vote(s) shall be final and binding.
- 15. The Results shall be declared by the Chairman of the Tribunal Convened Meeting or a person authorized by him not later than 48 (forty-eight) hours of the conclusion of the Meeting, upon receipt of Scrutinizer's Report. The declared Results along with the report of the Scrutinizer shall be placed on the website of the Company at <u>www.tril.co.in</u> under Investor Info and on the notice board of the Company and also on the website of the stock exchange, i.e., BSE Limited at <u>www.bseindia.com</u>, where Non-Convertible Debt Instruments of the Company are listed.
- 16. The Scheme of Amalgamation, if approved at the Tribunal Convened Meeting, will be subject to the approval of the NCLT and any such other approvals as may be required, if any.

#### Explanatory Statement to the Notice of the Meeting of Unsecured Creditors of Tata Realty and Infrastructure Limited under 102 read with Sections 230 to 232 of Rule 6 of the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 as per the directions of Hon'ble National Company Law Tribunal, Mumbai Bench

1. Pursuant to the Order dated 12<sup>th</sup> March, 2020 and 2<sup>nd</sup> November, 2020 passed by the Mumbai Bench of the National Company Law Tribunal (NCLT), in the Company Application Number 702 of 2019 (CA(CAA) No. 702/MB/2019) (Order), a meeting of the Unsecured Creditors of the Company ('Tribunal Convened Meeting'), is being convened on Monday, July 26, 2021 at 12.00 Noon for the purpose of considering and, if thought fit, approving with or without modification(s), the proposed amended Scheme of Amalgamation of MIA Infrastructure Private Limited and Wellkept Facility Management Services Private Limited and Acme Living Solutions Private Limited and TRIF Gurgaon Housing Projects Private Limited with Tata Realty and Infrastructure Limited and their respective shareholders and creditors under Sections 230 to 232 of the Companies Act, 2013 ("Act") (including any statutory modification or re-enactment or amendment thereof) read with the rules issued thereunder.

Subsequent to the filing of Scheme with NCLT, the Board of Directors vide its meeting dated 14<sup>th</sup> August, 2018 has changed the registered office of the Company Elphinstone Building, 2<sup>nd</sup> Floor, 10, Veer Nariman Road, Fort, Mumbai – 400001 to E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai 400033, with effect from 16<sup>th</sup> January, 2019. Further, the Board of Directors of the Company vide its meeting dated 23<sup>rd</sup> September, 2019 and 9<sup>th</sup> April, 2020 has increased the Authorised and Paid-up Share Capital respectively. Therefore, the amended Scheme is attached herewith.

The other definition mentioned in the scheme shall also applicable to the explanatory statement.

2. The Authorized, Issued, Subscribed and Paid-up share capital of MIA Infrastructure Private Limited **<u>First Transferor Company</u>** as on date is as under:

Particulars	Amount in Rs.
Authorised Capital	
28,00,000 Equity Shares of Rs. 10 each	2,80,00,000
Total	2,80,00,000
Issued, Subscribed and Paid-up Capital	
27,70,000 Equity Shares of Rs. 10 each	2,77,00,000
Total	2,77,00,000

 The Authorized, Issued, Subscribed and Paid-up share capital of Wellkept Facility Management Services Private Limited, <u>Second Transferor Company</u> as on date is as under:

Particulars	Amount in Rs.
Authorised Capital	
10,00,000 Equity Shares of Rs. 10 each	1,00,00,000
Total	1,00,00,000
Issued, Subscribed and Paid-up Capital	
4,00,000 Equity Shares of Rs. 10 each	40,00,000
Total	40,00,000

4. The Authorized, Issued, Subscribed and Paid-up share capital of Acme Living Solutions Private Limited, **Third Transferor Company** as on date is as under:

Particulars	Amount in Rs.
Authorised Capital	
50,000 Equity Shares of Rs. 10 each	5,00,000
Total	5,00,000
Issued, Subscribed and Paid-up Capital	
50,000 Equity Shares of Rs. 10 each	5,00,000
Total	5,00,000

5. The Authorized, Issued, Subscribed and Paid-up share capital of TRIF Gurgaon Housing Projects Private Limited, **Fourth Transferor Company** as on date is as under:

Particulars	Amount in Rs.
Authorised Capital	
50,000 Equity Shares of Rs. 10 each	5,00,000
Total	5,00,000
Issued, Subscribed and Paid-up Capital	
50,000 Equity Shares of Rs. 10 each 5	
Total	5,00,000

6. The Authorized, Issued, Subscribed and Paid-up share capital of TATA Realty and Infrastructure Limited the **Transferee Company** as on date is as under:

Particulars	Amount in Rs.
Authorised Capital	
8,00,00,00,000 Equity Shares of Rs. 10 each	80,00,00,00,000
Total	80,00,00,00,000
Issued, Subscribed and Paid-up Capital	
1,61,73,07,692 Equity Shares of Rs. 10 each	16,17,30,76,920
Total	16,17,30,76,920

7. Names of the directors of the **<u>Transferor Companies</u>** along with their addresses:

	MIA INFRASTRUCTURE PRIVATE LIMITED CIN:U74900MH2014PTC252385 (First Transferor Company)				
Sr. No.	DIN	Name	Address		
1	00047995	Ritesh Narendrabhai Kamdar	704, Gaurav Jhankar CHSL, Gaurav Zankar Bundurpakhadi Road, Gaurav Garden, Kandi vali (West Mumbai - 400 067		
2	08049549	Santosh Sitaram Mhadgut	Near Western Express Highway, Room No. 40, Building No 1, Teachers Colony, Bandra (East) Mumbai - 400 051		

	WELLKEPT FACILITY MANAGEMENT SERVICES PRIVATE LIMITED				
	CIN:	U93000MH20	008PTC177346 (	Second Transferor Company)	
Sr.	DIN	N	lame	Address	
No.					
1	00047995	Ritesh	Narendrabhai	704, Gaurav Jhankar CHSL, Gaurav	
		Kamdar		Zankar Bundurpakhadi Road,	
				Gaurav Garden, Kandi vali (West	
				Mumbai - 400 067	
2	03081163	Sunil Bharat	Dhagat	B-1002, Iraisaa Chs Ltd, 10th Floor,	
			-	Sector 19 Plot No. 1, Sanpada, Navi	
				Mumbai – 400 705	
3	07997663	Amit Bl	hupendrakumar	B/7, Adinath Apt, Sai Baba Nagar,	
		Sheth	-	Near J.B. Khot School, Borivali West,	
				Mumbai 400 092.	

	ACME LIVING SOLUTIONS PRIVATE LIMITED				
	CIN	N: U45209DL2008PTC178023 (	(Third Transferor Company)		
Sr.	DIN	Name	Address		
No.					
1	03081163	Sunil Bharat Dhagat	B-1002, Iraisaa Chs Ltd, 10th Floor,		
			Sector 19 Plot No. 1, Sanpada, Navi		
			Mumbai – 400 705		
2	06967259	Vaidehi Saiprasad Modi	Flat Np. 501, Western Palace Chs		
			Ltd, Near Yogi Nagar, Link Road,		
			Borilvi West, Mumbai 400 091.		
3	00047995	Ritesh Narendrabhai	704, Gaurav Jhankar CHSL, Gaurav		
		Kamdar	Zankar Bundurpakhadi Road,		
			Gaurav Garden, Kandi vali (West)		
			Mumbai 400067		

	TRIF GURGAON HOUSING PROJECTS PRIVATE LIMITED CIN: U74900DL2009PTC188404 (Fourth Transferor Company)				
Sr. No.	DIN	Name	Address		
1	07997663	Amit Bhupendrakumar Sheth	B/7, Adinath Apt, Sai Baba Nagar, Near J.B. Khot School, Borivali West, Mumbai – 400 092.		
2	08049549	Santosh Sitaram Mhadgut	Near Western Express Highway, Room No. 40, Building No 1, Teachers Colony, Bandra (East) Mumbai – 400 051.		
3	00047995	Ritesh Narendrabhai Kamdar	704, Gaurav Jhankar CHSL, Gaurav Zankar Bundurpakhadi Road, Gaurav Garden, Kandi vali (West) Mumbai 400067		

- TATA Realty and Infrastructure Limited CIN U70102MH2007PLC168300 (the Transferee Company) DIN/PAN Sr. No. Name Designation Address Vivarea Residencies, B-05251670 Sanjay Managing 1 Bhupender Dutt Director and 1102, 11th Floor, Sane Guruji Marg, Mahalaxmi, CEO(KMP) Mumbai -400 011 00057333 Rajiv Sabharwal Director C / 183, Kalpataru Sparkle, N. 2 Dharmadhikari Road, Gandhinagar, Ban dra East, Mumbai 400051 00120029 Banmali Director Ashford Apartment, Flat 3 No.03, 3rd Floor, 1/26A Agrawala Ridge Road, Malabar Hill Mumbai 400006 4 00028428 Farokh Nariman Director 1, Wadia Building, 6 Subedar Babulnath Road, Mumbai 400007 B 1501, Ashok Towers, AAIPS2689D Sanjay Sharma CFO(KMP) 5 Dr. B. A. Road, Parel, Mumbai 400012 AALPS9504Q 1203, Rustomjee Ozone 6 Sudhakar Company Hiriyanna Secretary Tower 1, off S V Road, Shetty Goregaon West, Mumbai-400062
- 8. Names of the directors and Key Managerial Personnel ("KMP") of <u>the Transferee</u> <u>Company</u> along with their addresses:

#### 9. The details of MIA Infrastructure Private Limited (First Transferor Company):

Particulars	First Transferor Company
Corporate Identification Number (CIN)	U74900MH2014PTC252385
Permanent Account Number (PAN)	AAJCM4215B
Name of the company	MIA Infrastructure Private Limited
Date of Incorporation	22/01/2014
Type of the company	Private Company
Registered office address	E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai 400 033.
e-mail address	trilsec@tatarealty.in
Summary of main object	Engaged in the business of development of infrastructure
as per the memorandum	facilities including allied/incidental services related to
of association and main	infrastructure business.
business carried on by	
the company	

Details of change of name, registered office and objects of the company during the last five years	There is no change of name and objects of the Company during last five years. However, the registered office of the Company has been shifted from Elphinstone Building, 2 <sup>nd</sup> Floor, 10, Veer Nariman Road, Fort, Mumbai – 400001 to E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai 400033, with effect from January 16, 2019.
Details of the capital structure of the company including authorised, issued, subscribed and paid up Share Capital	As per para 2 of the Explanatory Statement
Names of the promoters and directors along with their addresses	Promoter- Tata Realty and Infrastructure Limited Registered office: E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai- 400033 Director- As per para 7 of the Explanatory Statement
The date of the board meeting at which the scheme was approved by the board of directors including the name of the directors who voted in favour of the resolution, who voted against the resolution and who did not vote or participate on such resolution	2 <sup>nd</sup> January, 2019 and 20 <sup>th</sup> January, 2021 The meeting was attended by all the Directors (list of directors is as mentioned Above in para 7 of Explanatory Statement and the resolution was passed unanimously.
Explanatory Statement dis including:-	sclosing details of the scheme of compromise or arrangement

Parties involved in such compromise or	"Transferor Companies"
arrangement	1. MIA Infrastructure Private Limited,
	2. Wellkept Facility Management Services Private Limited,
	3. Acme Living Solutions Private Limited; and
	4. TRIF Gurgaon Housing Projects Private Limited
	"Transferee Company"
	TATA Realty and Infrastructure Limited
	The Transferor Companies are Wholly Owned Subsidiaries of Transferee Company.
Appointed Date	1 <sup>st</sup> April, 2019

Effective Date	means the last of the dates on which the certified copies or the authenticated copies of the order of National Company Law Tribunal sanctioning of the Scheme, are filed with the Respective Registrar of Companies by the Transferor Companies and the Transferee Company. Any reference to this scheme to the date of "coming into effect of this scheme" or "effective date of this scheme" or scheme taking effect, shall means the effective date;
Share Exchange Ratio and other considerations, if any	the entire issued, subscribed and paid-up share capital of the Transferor Companies is held by the Transferee Company along with its nominee. Hence, there is no share exchange ratio involved in the scheme. Upon the Scheme becoming effective, no shares of the Transferee Company shall be allotted in lieu or exchange of its holding in the Transferor Companies and the share capital of the Transferor Companies shall stand cancelled.
Summary of Valuation report (if applicable) including basis of valuation and fairness opinion of the registered valuer, if any, and the declaration that the valuation report is available for inspection at registered office of the Company.	As mentioned above, Transferor companies are the wholly owned subsidiaries of Transferee company and no shares being issued by transferee company to the members of the transferor company, Therefore, no valuation is required to be done.
Details of capital or debt	The Scheme dees not involve any capital or debt
restructuring	The Scheme does not involve any capital or debt restructuring.
Rationale for the	Refer Point 3 of the Scheme.
compromise or	
arrangement	
Benefits of the	Refer Point 3 of the Scheme.
compromise or	
arrangement as	
perceived by the Board of directors to the	
company, members,	
creditors and others.	
Amount due to	Rs. 40,16,290/-
unsecured creditors as	
on 30 <sup>th</sup> November, 2018	
Amount due to	
Amount due to unsecured creditors as	Rs. 0/-
on 28 <sup>th</sup> February, 2021	
	the compromise or arrangement on:
Key Managerial	No Effect
personnel (KMP)	
KMP (other than	No Effect as there is no KMP.
Directors)	

Directors	As the Transferor Companies are ceased to exist, the
	question of any change in the director does not arise.
Promoters	TATA Realty and Infrastructure Limited
Non-promoter members	Nil
Depositors	Not applicable
Creditors	Creditors, if any, shall become the creditors of the Transferee
	Company and paid off in the ordinary course of business. No Effect
Debenture holders	- Not applicable
Deposit trustee &	- Not applicable
Debenture Trustee	
Employees of the	No Effect
company	
	compromise or arrangement on material interest of Directors,
	(KMP)and debenture trustee:
Directors	Nil
Key Managerial	-Not applicable
personnel (KMP)	
Debenture Trustee	Not applicable
Investigation or	- Nil
proceedings, if	
any, pending against the	
company under the Act	
In addition to the documents annexed hereto, the copies of the following document	
will be open for inspection at the Registered Office of the Company between 11:00 a.m	
to 01:00 p.m. on all working days, except Saturdays, Sundays and Public Holidays, up	
to 1 (one) day prior to the date of the Tribunal Convened Meeting'.	

I. Copies of the latest audited financial statements of Applicant Company.

II. Copy of the order of Tribunal dated 12th March, 2020 and 2nd November, 2020.

III. Copy of Scheme of Compromise or Arrangement.

IV. Memorandum and Articles of Association of the Applicant Company.

- V. The certificate issued by Auditor of the company to the effect that the accounting treatment, if any, proposed in the scheme of compromise or arrangement is in conformity with the Accounting Standards prescribed under Section 133 of the Companies Act, 2013.
- VI. Form No. GNL-1 filed by the Applicant Company with the Ministry of Corporate Affairs, evidencing filing of the scheme.
- VII. Register of Directors and key Managerial Personnel and their Shareholding.

A statement to the effect	The person to whom this notice is sent may vote in the
	meeting either in person or through Authorised
whom the notice issued	Representative.
may vote in the meeting	
either in person or	
through Authorised	
Representative.	

10. The details of Wellkept Facility Management Services Private Limited **(Second Transferor Company)**:

Particulars	Second Transferor Company
Corporate Identification Number (CIN)	U93000MH2008PTC177346
Permanent Account Number (PAN)	AACCT8259R
Name of the company	Wellkept Facility Management Services Private Limited
Date of Incorporation	04/01/2008
Type of the company	Private Company
Registered office address	E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai 400 033.
e-mail address	trilsec@tatarealty.in
Summary of main object as per the memorandum of association and main business carried on by the company	Engaged in the business of real estate including allied/incidental services related to infrastructure business.
Details of change of name, registered office and objects of the company during the last five years	There is no change of name and objects of the Company during last five years. However, the registered office of the Company has been shifted from Elphinstone Building, 2nd Floor, 10, Veer Nariman Road, Fort, Mumbai – 400001 to E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai 400033, with effect from January 16, 2019.
Details of the capital structure of the company including authorised, issued, subscribed and paid up Share Capital	As per para 3 of the Explanatory Statement
Names of the promoters and directors along with their addresses	Promoter- Tata Realty and Infrastructure Limited Registered office: E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai- 400033 Director- As per para 7 of the Explanatory Statement
The date of the board meeting at which the scheme was approved by the board	2 <sup>nd</sup> January, 2019 and 20 <sup>th</sup> January, 2021 The meeting was attended by all the Directors (list of
of directors including the name of the directors who voted in favour of the resolution, who voted against the resolution and who did not vote or	directors is as mentioned Above in para 7 of Explanatory Statement and the resolution was passed unanimously.
participate	
on such resolution Explanatory Statement disc arrangement including:-	losing details of the scheme of compromise or
Parties involved in such	"Transferor Companies"
compromise or arrangement	1. MIA Infrastructure Private Limited,

	2. Wellkept Facility Management Services Private Limited,
	3. Acme Living Solutions Private Limited And
	4. TRIF Gurgaon Housing Projects Private Limited
	"Transferee Company"
	TATA Realty and Infrastructure Limited
	The Transferor Companies are Wholly Owned Subsidiaries of Transferee Company
Appointed Date	1 <sup>st</sup> April, 2019
Effective Date	means the last of the dates on which the certified copies or the authenticated copies of the order of National Company Law Tribunal sectioning of the Scheme, are filed with the Respective Registrar of Companies by the Transferor Companies and the Transferee Company. Any reference to this scheme to the date of "coming into effect of this scheme" or "effective date of this scheme" or scheme taking effect, shall means the effective date;
Share Exchange Ratio and other considerations, if any	The entire issued, subscribed and paid-up share capital of the Transferor Companies is held by the Transferee Company along with its nominee. So there is no share exchange ratio involved in the scheme. Upon the Scheme becoming effective, no shares of the Transferee Company shall be allotted in lieu or exchange of its holding in the Transferor Companies and the share capital of the Transferor Companies shall stand cancelled.
Summary of Valuation report (if applicable) including basis of valuation and fairness opinion of the registered valuer, if any, and the declaration that the valuation report is available for inspection at registered office of the Company.	As mentioned above, Transferor companies are the wholly owned subsidiaries of Transferee company and no shares being issued by transferee company to the members of the transferor company, Therefore, no valuation is required to be done.
Details of capital or debtrestructuringRationaleforthecompromise or arrangement	The Scheme does not involve any capital or debt restructuring. Refer Point 3 of the Scheme.
Benefits of the compromise or arrangement as perceived by the Board of directors to the company, members, creditors and others.	Refer Point 3 of the Scheme.

Amount due to unsecured creditors as on 30 <sup>th</sup>	Rs. 1,212/-	
November, 2018		
Amount due to unsecured creditors as on 28 <sup>th</sup> February, 2021	Rs. 0/-	
Disclosure about effect of the	compromise or arrangement on:	
Key Managerial personnel	No Effect	
(KMP) (other than Directors)	No Effect as there is no KMP.	
Directors	As the Transferor Companies are ceased to exist, the question of any change in the director does not arise.	
Promoters	TATA Realty and Infrastructure Limited	
Non-promoter members	Nil	
Depositors	Not applicable	
Creditors	Creditors, if any, shall become the creditors of the Transferee Company and paid off in the ordinary course of business. No Effect	
Debenture holders	- Not applicable	
Deposit trustee & Debenture	- Not applicable	
Trustee		
Employees of the company	No Effect	
Disclosure about effect of a	compromise or arrangement on material interest of	
	sonnel (KMP)and debenture trustee:	
Directors	Nil	
Key Managerial personnel (KMP)	-Not applicable	
Debenture Trustee	Not applicable	
Investigation or proceedings,	- Nil	
any, pending against the		
company under the Act		
	annexed hereto, the copies of the following documents	
will be open for inspection at the Registered Office of the Company between 11:00		
a.m. to 01:00 p.m. on all working days, except Saturdays, Sundays and Public		
	ior to the date of the Tribunal Convened Meeting'.	
I. Copies of the latest aud	ited financial statements of Applicant Company.	
	punal dated 12th March, 2020 and 2nd November, 2020.	
	promise or Arrangement.	
IV. Memorandum and Artic	es of Association of the Applicant Company.	
	by Auditor of the company to the effect that the	
-	if any, proposed in the scheme of compromise or	
	prmity with the Accounting Standards prescribed under	
Section 133 of the Com		
Affairs, evidencing filing		
	d key Managerial Personnel and their Shareholding.	
A statement to the effect that	The person to whom this notice is sent may vote in	
the persons to whom the	the meeting either in person or through Authorised	
notice issued may vote in the	Representative.	
meeting either in person or through Authorised		
through Authorised		

Re	presentative.	

11. The details of Acme Living Solutions Private Limited (Third Transferor Company):

Particulars	Third Transferor Company	
Corporate Identification Number (CIN)	U45209DL2008PTC178023	
Permanent Account Number (PAN)	AACCT8259R	
Name of the company	Acme Living Solutions Private Limited	
Date of Incorporation	13/05/2008	
Type of the company	Private Company	
Registered office address	C/O Tata Services Ltd., Jeevan Bharti, Tower - 1, 10th Floor, 124, Connaught Circus, New Delhi - 110001.	
e-mail address	trilsec@tatarealty.in	
Summary of main object as per the memorandum of association and main business carried on by the company	Engaged in the business of real estate and infrastructure facilities including allied/incidental services related to real estate and infrastructure business.	
Details of change of name, registered office and objects of the company during the last five years	There is no change of name, registered office and objects of the Company during last five years.	
Details of the capital structure of the company including authorised, issued, subscribed and paid up Share Capital	As per para 4 of the Explanatory Statement	
Names of the promoters and directors along with their addresses	Promoter- Tata Realty and Infrastructure Limited Registered office: E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai- 400033 Director- As per para 7 of the Explanatory Statement	
The date of the board meeting at which the scheme	2 <sup>nd</sup> January, 2019 and 20 <sup>th</sup> January, 2021	
was approved by the board of directors including the name of the directors who voted in favour of the resolution, who voted against the resolution and who did not vote or participate	The meeting was attended by all the Directors (list of directors is as mentioned Above in para 7 of Explanatory Statement and the resolution was passed unanimously.	
on such resolution Explanatory Statement disc arrangement including:-	losing details of the scheme of compromise or	
Parties involved in such compromise or arrangement	"Transferor Companies"	
	1. MIA Infrastructure Private Limited,	
	2. Wellkept Facility Management Services Private Limited,	

	3. Acme Living Solutions Private Limited And
	-
	4. TRIF Gurgaon Housing Projects Private Limited
	"Transferee Company"
	TATA Realty and Infrastructure Limited
	The Transferor Companies are Wholly Owned Subsidiaries of Transferee Company
Appointed Date	1 <sup>st</sup> April, 2019
Effective Date	means the last of the dates on which the certified copies or the authenticated copies of the order of National Company Law Tribunal sectioning of the Scheme, are filed with the Respective Registrar of Companies by the Transferor Companies and the Transferee Company. Any reference to this scheme to the date of "coming into effect of this scheme" or "effective date of this scheme" or scheme taking effect, shall means the effective date;
Share Exchange Ratio and other considerations, if any	The entire issued, subscribed and paid-up share capital of the Transferor Companies is held by the Transferee Company along with its nominee. So there is no share exchange ratio involved in the scheme. Upon the Scheme becoming effective, no shares of the Transferee Company shall be allotted in lieu or exchange of its holding in the Transferor Companies and the share capital of the Transferor Companies shall stand cancelled.
Summary of Valuation report (if applicable) including basis of valuation and fairness opinion of the registered valuer, if any, and the declaration that the valuation report is available for inspection at registered office of the Company.	As mentioned above, Transferor companies are the wholly owned subsidiaries of Transferee company and no shares being issued by transferee company to the members of the transferor company, Therefore, no valuation is required to be done.
Details of capital or debt restructuring Rationale for the	The Scheme does not involve any capital or debt restructuring. Refer Point 3 of the Scheme.
compromise or arrangement	
Benefits of the compromise or arrangement as perceived by the Board of directors to the company, members, creditors and others.	Refer Point 3 of the Scheme.
Amount due to unsecured creditors as on 30 <sup>th</sup> November, 2018	Rs. 54,000/-

Amount due to unsecured	Rs. 0/-
creditors as on 28 <sup>th</sup> February, 2021	
Disclosure about effect of the	compromise or arrangement on:
Key Managerial personnel	No Effect
(KMP) (other than Directors)	No Effect as there is no KMP.
Directors	As the Transferor Companies are ceased to exist, the
	question of any change in the director does not arise.
Promoters	TATA Realty and Infrastructure Limited
Non-promoter members	Nil
Depositors	Nil
Creditors	Creditors, if any, shall become the creditors of the Transferee Company and paid off in the ordinary course of business. No Effect
Debenture holders	-Nil
Deposit trustee & Debenture Trustee	-Nil
Employees of the company	No Effect
	compromise or arrangement on material interest of
	sonnel (KMP)and debenture trustee:
Directors	Nil
Key Managerial personnel (KMP)	-Not applicable
Debenture Trustee	Not applicable
Investigation or proceedings,	- Nil
if	
any, pending against the	
company under the Act	
will be open for inspection at a.m. to 01:00 p.m. on all w	annexed hereto, the copies of the following documents the Registered Office of the Company between 11:00 working days, except Saturdays, Sundays and Public for to the date of the Tribunal Convened Meeting'.
<ul><li>II. Copy of the order of Trib</li><li>III. Copy of Scheme of Com</li><li>IV. Memorandum and Artic</li><li>V. The certificate issued accounting treatment,</li></ul>	lited financial statements of Applicant Company. bunal dated 12th March, 2020 and 2nd November, 2020. hpromise or Arrangement. les of Association of the Applicant Company. by Auditor of the company to the effect that the if any, proposed in the scheme of compromise or
Section 133 of the Com	y the Applicant Company with the Ministry of Corporate
	d key Managerial Personnel and their Shareholding.
Details of approvals,	The Company had filed a joint application before the
sanctions or no-objection(s),	National Company Law Tribunal, New Delhi on March

	a key hanagenar i ersenner and then endrenerangi
Details of approvals,	The Company had filed a joint application before the
sanctions or no-objection(s),	National Company Law Tribunal, New Delhi on March
if any, from regulatory or any	29, 2019. The National Company Law Tribunal, New
other governmental	Delhi had granted dispensation from holding meetings
authorities	of the shareholders and creditors vide order dated
required, received or	December 17, 2019.
pending for the proposed	
scheme of compromise or	The Company is not registered with any sectoral
arrangement	regulators and as such no specific approval is
	required to be obtained from any sectoral regulator.

	In compliance with the Directions of the National Company Law Tribunal, New Delhi. The Company has served notice of the application with the Registrar of Companies, Regional Director, Income Tax Authority and Official Liquidator. The observation, received or to be received is being/ will be addressed by the Company in due time. Form GNL 1 filed by Company with Registrar of Companies evidencing filing of scheme
A statement to the effect that the persons to whom the notice issued may vote in the meeting either in person or through Authorised Representative.	the meeting either in person or through Authorised

12. The details of TRIF Gurgaon Housing Projects Private Limited (Fourth Transferor Company):

Particulars	Fourth Transferor Company
Corporate Identification Number (CIN)	U74900DL2009PTC188404
Permanent Account Number (PAN)	AADCT1808K
Name of the company	TRIF Gurgaon Housing Projects Private Limited
Date of Incorporation	13/03/2009
Type of the company	Private Company
Registered office address	C/O Tata Services Ltd., Jeevan Bharti, Tower - 1, 10th Floor, 124, Connaught Circus, New Delhi - 110001.
e-mail address	trilsec@tatarealty.in
Summary of main object as per the memorandum of association and main business carried on by the company	Engaged in the Business of real estate and infrastructure facilities including allied/incidental services related to real estate and infrastructure business
Details of change of name, registered office and objects of the company during the last five years	There is no change of name, registered office and objects of the Company during last five years.
Details of the capital structure of the company including authorised, issued, subscribed and paid up Share Capital	As per para 5 of the Explanatory Statement
Names of the promoters and directors along with their addresses	Promoter- Tata Realty and Infrastructure Limited Registered office: E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai- 400033 Director- As per para 7 of the Explanatory Statement

The date of the board meeting at which the scheme	2 <sup>nd</sup> January, 2019 and 20 <sup>th</sup> January, 2021
was approved by the board of directors including the name of the directors who voted in favour of the resolution, who voted against the resolution and who did not vote or participate on such resolution <i>Explanatory Statement disc</i>	The meeting was attended by all the Directors (list of directors is as mentioned Above in para 7 of Explanatory Statement and the resolution was passed unanimously.
arrangement including:-	
Parties involved in such compromise or arrangement	"Transferor Companies"
	1. MIA Infrastructure Private Limited,
	2. Wellkept Facility Management Services Private Limited,
	3. Acme Living Solutions Private Limited And
	4. TRIF Gurgaon Housing Projects Private Limited
	"Transferee Company"
	TATA Realty and Infrastructure Limited
	The Transferor Companies are Wholly Owned Subsidiaries of Transferee Company
Appointed Date	1 <sup>st</sup> April, 2019
Effective Date	means the last of the dates on which the certified copies or the authenticated copies of the order of National Company Law Tribunal sectioning of the Scheme, are filed with the Respective Registrar of Companies by the Transferor Companies and the Transferee Company. Any reference to this scheme to the date of "coming into effect of this scheme" or "effective date of this scheme" or scheme taking effect, shall means the effective date;
Share Exchange Ratio and other considerations, if any Summary of Valuation report	The entire issued, subscribed and paid-up share capital of the Transferor Companies is held by the Transferee Company along with its nominee. So there is no share exchange ratio involved in the scheme. Upon the Scheme becoming effective, no shares of the Transferee Company shall be allotted in lieu or exchange of its holding in the Transferor Companies and the share capital of the Transferor Companies shall stand cancelled. As mentioned above, Transferor companies are the
(if applicable) including basis of valuation and fairness opinion of the registered	wholly owned subsidiaries of Transferee company and no shares being issued by transferee company to the

valuer, if any, and the	
declaration that the valuation	valuation is required to be done.
report is available for	
inspection at registered	
office of the Company.	The Coheme date not involve any emited on date
Details of capital or debt	The Scheme does not involve any capital or debt
restructuring	restructuring.
Rationale for the	Refer Point 3 of the Scheme.
compromise or arrangement	
Benefits of the compromise	Refer Point 3 of the Scheme.
or arrangement as perceived	
by the Board of directors to	
the company, members,	
creditors and others.	
Amount due to unsecured	Rs. 54,000/-
creditors as on 30 <sup>th</sup>	
November, 2018	
Amount due to unsecured	Rs. 0/-
creditors as on 28 <sup>th</sup> February,	
2021	
	compromise or arrangement on:
Key Managerial personnel	No Effect
(KMP) (other than Directors)	No Effect as there is no KMP.
Directors	As the Transferor Companies are ceased to exist, the
	question of any change in the director does not arise.
Promoters	TATA Realty and Infrastructure Limited
Non-promoter members	Nil
Depositors	Nil
Creditors	Creditors, if any, shall become the creditors of the
	Transferee Company and paid off in the ordinary
	course of business. No Effect
Debenture holders	-Nil
Deposit trustee & Debenture	-Nil
Trustee	
Employees of the company	No Effect
	compromise or arrangement on material interest of
	sonnel (KMP)and debenture trustee:
Directors	Nil
Key Managerial personnel	-Not applicable
(KMP)	
Debenture Trustee	Not applicable
Investigation or proceedings,	- Nil
if	
any, pending against the	
company under the Act	
	annexed hereto, the copies of the following documents
	the Registered Office of the Company between 11:00
	working days, except Saturdays, Sundays and Public
	rior to the date of the Tribunal Convened Meeting'.
	To the date of the Hibdhar Convened Meeting.
I. Copies of the latest aud	lited financial statements of Applicant Company
	lited financial statements of Applicant Company.
<ul> <li>II. Copy of the order of Tribunal dated 12<sup>th</sup> March, 2020 and 2<sup>nd</sup> November, 2020.</li> <li>III. Copy of Scheme of Compromise or Arrangement.</li> </ul>	

IV. Memorandum a	ad Auticles of Association of the Applicant Company
	nd Articles of Association of the Applicant Company.
	issued by Auditor of the company to the effect that the
	tment, if any, proposed in the scheme of compromise or
	in conformity with the Accounting Standards prescribed under
	he Companies Act, 2013.
	filed by the Applicant Company with the Ministry of Corporate ng filing of the scheme.
	ctors and key Managerial Personnel and their Shareholding.
	rovals, The Company had filed a joint application before the
sanctions or no-object	
if any, from regulatory	
other govern	
authorities required, re	
or pending for the pro	
scheme of comprom	
arrangement	The Company is not registered with any sectoral regulators and as such no specific approval is required to be obtained from any sectoral regulator. In compliance with the Directions of the National Company Law Tribunal, New Delhi The Company has served notice of the application with the Registrar of Companies, Regional Director, Income Tax Authority and Official Liquidator, observation, if any,received or to be received is being / will be addressed by the Company in compliance. Form GNL 1 filed by Company with Registrar of Companies evidencing filing of scheme
A statement to the effe	
the persons to who	
notice issued may vote	e in the Representative.
meeting either in per	son or
	norised
Representative.	

# 13. The details of Tata Realty and Infrastructure Limited (Transferee Company):

Particulars	Transferee Company
Corporate Identification Number (CIN)	U70102MH2007PLC168300
Permanent Account Number (PAN)	AACCT6242L
Name of the company	Tata Realty and Infrastructure Limited
Date of Incorporation	02/03/2007
Type of the company	Public Company
Registered office address	E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai 400 033.
e-mail address	trilsec@tatarealty.in
Website	www.tril.co.in
Summary of main object as per the memorandum of association and main	Engaged in the Business of real estate and infrastructure facilities including allied/incidental

business carried on by the company	services related to real estate and infrastructure business
Details of change of name, registered office and objects of the company during the last five years	There is no change of name and objects of the Company during last five years. However, the registered office of the Transferee Company has been shifted from Elphinstone Building, 2nd Floor, 10 Veer Nariman Road, Fort, Mumbai – 400001 to E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai 400 033, with effect from January 16, 2019.
Details of the capital structure of the company including authorised, issued, subscribed and paid up Share Capital	As per para 6 of the Explanatory Statement
Names of the promoters and directors along with their addresses	Promoters- Tata Sons Private Limited Registered office: Bombay House, 24 Homi Mody Street, Mumbai 400001 Director- As per para 7 of the Explanatory Statement
The date of the board meeting at which the scheme was approved by the board of directors including the name of the directors who voted in favour of the resolution, who voted against the resolution and who did not vote or participate on such resolution	18 <sup>th</sup> November, 2018 and 13 <sup>th</sup> January, 2021 The meeting was attended by all the Directors (list of directors is as mentioned Above in para 7 of Explanatory Statement and the resolution was passed unanimously.
	losing details of the scheme of compromise or
arrangement including:-	
Parties involved in such compromise or arrangement	"Transferor Companies"
	1. MIA Infrastructure Private Limited,
	2. Wellkept Facility Management Services Private Limited,
	3. Acme Living Solutions Private Limited And
	4. TRIF Gurgaon Housing Projects Private Limited
	"Transferee Company"
	TATA Realty and Infrastructure Limited
	The Transferor Companies are Wholly Owned Subsidiaries of Transferee Company
Appointed Date	1 <sup>st</sup> April, 2019
Effective Date	means the last of the dates on which the certified copies or the authenticated copies of the order of National Company Law Tribunal sectioning of the Scheme, are filed with the Respective Registrar of Companies by the Transferor Companies and the

Γ	
	Transferee Company. Any reference to this scheme to
	the date of "coming into effect of this scheme" or
	"effective date of this scheme" or scheme taking
	effect, shall means the effective date;
Share Exchange Ratio and	Not Applicable being a Transferee Company.
other considerations, if any	
Summary of Valuation report	Transferor companies are the wholly owned
(if applicable) including basis	subsidiaries of Transferee company and no shares
of valuation and fairness	being issued by transferee company to the members
opinion of the registered	of the transferor company, Therefore, no valuation is
valuer, if any, and the	required to be done.
declaration that the valuation	
report is available for	
inspection at registered	
office of the Company.	
Details of capital or debt	The Scheme does not involve any capital or debt
restructuring	restructuring.
Rationale for the	Refer Point 3 of the Scheme.
compromise or arrangement	
Benefits of the compromise	Also provided in the rational for amalgamation in Point
or arrangement as perceived	3 of the scheme.
by the Board of directors to	
the company, members,	
creditors and others.	
Amount due to unsecured	The aggregate amount of the dues to unsecured
creditors as on 30 <sup>th</sup>	creditors (including unsecured Non-convertible
November, 2018	Debenture Holders) is Rs. 1,834,87,28,391/-
Amount due to unsecured	The aggregate amount of the dues to unsecured
creditors as on 28 <sup>th</sup> February,	creditors (including unsecured Non-convertible
2021	Debenture Holders) is Rs. 2622,92,29,266/-
Disclosure about effect of the	compromise or arrangement on:
Key Managerial personnel	No Effect
(KMP) (other than Directors)	No Effect
Directors	No Effect
Promoters	No Effect
Non-promoter members	No Effect
Depositors	Not Applicable
Creditors	No Effect
Debenture holders	No Effect
Deposit trustee & Debenture	No Effect
Trustee	
Employees of the company	No Effect
	compromise or arrangement on material interest of
	sonnel (KMP)and debenture trustee:
Directors	No Effect
Key Managerial personnel	No Effect
(KMP)	
Debenture Trustee	No Effect
Investigation or proceedings,	Nil
if any, pending against the	
company under the Act	

In addition to the documents annexed hereto, the copies of the following documents will be open for inspection at the Registered Office of the Company between 11:00 a.m. to 01:00 p.m. on all working days, except Saturdays, Sundays and Public Holidays, up to 1 (one) day prior to the date of the Tribunal Convened meeting.

These documents are also available on website of the Company, i.e., <u>www.tril.co.in</u>.

- I. Copies of the latest audited financial statements of Applicant Company.
- II. Copy of the order of Tribunal dated 12th March, 2020 and 2nd November, 2020.
- III. Copy of Scheme of Compromise or Arrangement.
- IV. Memorandum and Articles of Association of the Applicant Company.
- V. The certificate issued by Auditor of the company to the effect that the accounting treatment, if any, proposed in the scheme of compromise or arrangement is in conformity with the Accounting Standards prescribed under Section 133 of the Companies Act, 2013.
- VI. Form No. GNL-1 filed by the Applicant Company with the Ministry of Corporate Affairs, evidencing filing of the scheme.

*VII.* Register of Directors and key Managerial Personnel and their Shareholding.

Copy of Scheme of	Enclosed to this Notice.
Compromise or Arrangement	
Contracts or Agreements	Nil
material to the compromise	
or arrangement	
A statement to the effect that	The person to whom this notice is sent may vote in
the persons to whom the	5 1 5
notice issued may vote in the	Representative.
meeting either in person or	
through Authorised	
Representative.	

Sanjay Dutt

DIN: 05251670 Chairman appointed for the aforesaid Tribunal Convened Meeting

Date: June 21, 2021 Place: Mumbai

# THE SCHEME OF AMALGAMATION

### UNDER SECTION 230 TO 232 OF THE COMPANIES ACT, 2013

OF

# MIA INFRASTRUCTURE PRIVATE LIMITED

#### AND

# WELLKEPT FACILITY MANAGEMENT SERVICES PRIVATE LIMITED AND

# ACME LIVING SOLUTIONS PRIVATE LIMITED

# AND

# TRIF GURGAON HOUSING PROJECTS PRIVATE LIMITED

#### WITH

# TATA REALTY AND INFRASTRUCTURE LIMITED AND THEIR RESPECTIVE SHAREHOLDERS, CREDITORS

**1.** PREAMBLE

This Scheme of Amalgamation is presented pursuant to the provisions of Section 230 to 232 and other relevant provisions of the Companies Act, 2013 for Amalgamation of **MIA Infrastructure Private Limited** and **Wellkept Facilty Management Services Private Limited** and **Acme Living Solutions**  **Private Limited** and **TRIF Gurgaon Housing Projects Private Limited** with **Tata Realty and Infrastructure Limited**.

- 2. Background and Description of Companies
  - I. TRANSFEROR COMPANIES: The following Transferor Company hereinafter collectively called as "Transferor Companies".
  - a. MIA INFRASTRUCTURE PRIVATE LIMITED, The First Transferor Company (CIN: U74900MH2014PTC252385) was incorporated on 22<sup>nd</sup> January, 2014, under the Companies Act, 1956. The Registered Office of the Company is situated at E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai - 400033.

The First Transferor Company is engaged in the business of development of infrastructure facilities including allied/incidental services related to infrastructure business.

The First Transferor Company is wholly owned subsidiary of the Transferee Company.

E-Mail ID: trilsec@tatarealty.in Website: Not Applicable

 WELLKEPT FACILITY MANAGEMENT SERVICES PRIVATE LIMITED, The Second Transferor Company (CIN: U93000MH2008PTC177346) was incorporated on 04<sup>th</sup> January, 2008 under the Companies Act, 1956. The Registered Office of the Company is situated at E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai – 400033.

The Second Transferor Company is engaged in the business of real estate including allied/incidental services related to infrastructure business.

E-Mail ID: trilsec@tatarealty.in Website: Not Applicable

The Second Transferor Company is wholly owned subsidiary of the Transferee Company.

c. ACME LIVING SOLUTIONS PRIVATE LIMITED, The Third Transferor Company (CIN: U45209DL2008PTC178023) was incorporated on 13<sup>th</sup> May, 2008, under the Companies Act, 1956. The Registered Office of the Company is situated at C/O Tata Services Ltd., Jeevan Bharti, Tower - 1, 10th Floor,124, Connaught Circus, New Delhi - 110001.

E-Mail ID: trilsec@tatarealty.in Website: Not Applicable

The Third Transferor Company is engaged in the business of real estate and infrastructure facilities including allied/incidental services related to real estate and infrastructure business.

The Third Transferor Company is wholly owned subsidiary of the Transferee Company.

d. TRIF GURGAON HOUSING PROJECTS PRIVATE LIMITED, The Fourth Transferor Company (CIN: U74900DL2009PTC188404) was incorporated on 13<sup>th</sup> March, 2009 under the Companies Act, 1956. The Registered Office of the Company is situated at C/O Tata Services Ltd., Jeevan Bharti, Tower - 1, 10th Floor, 124, Connaught Circus, New Delhi – 110001.

The Fourth Transferor Company is engaged in the Business of real estate and infrastructure facilities including allied/incidental services related to real estate and infrastructure business.

E-Mail ID: trilsec@tatarealty.in

Website: Not Applicable

The Fourth Transferor Company is wholly owned subsidiary of the Transferee Company.

# II. TRANSFEREE COMPANY:

**TATA REALTY AND INFRASTRUCTURE LIMITED**, the Transferee Company (CIN: U70102MH2007PLC168300) was incorporated on 2<sup>nd</sup> March, 2007, under the Companies Act, 1956. The Registered Office of the

Company is situated at E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai - 400033.

The Transferee Company is engaged in the Business of real estate and infrastructure facilities including allied/incidental services related to real estate and infrastructure business.

E-Mail ID: trilsec@tatarealty.in Website: www.tril.co.in

**III.** This Scheme of Amalgamation provides for the Amalgamation of the Transferor Companies with the Transferee Company pursuant to section 230 to 232 and other relevant provisions of the Companies Act, 2013.

# 3. Rational of the Company

The Transferee Company is holding the entire share capital of the Transferor Companies. Both the Transferor and Transferee Companies are engaged in the business of real estate and infrastructure facilities including allied/incidental services related to real estate and infrastructure business. The Amalgamation of Transferor Companies with the Transferee Company would *inter alia* have following benefits:

- a. To reduce overlaps which are necessarily involved in running multiple entities, to reduce administrative cost and multiple layer inefficiencies so as to achieve operational and management efficiency.
- b. Cost savings are expected to flow from more focused operational efforts, rationalization, standardization and simplification of business processes, and the elimination of duplication, and rationalization of administrative expenses.
- c. Simplifying corporate governance procedures, resulting in structural efficiency, better utilization of capital and a simplified business structure.

In view of the aforesaid, The Board of Directors of the Transferor Companies and the Transferee Company have considered and proposed the amalgamation of the entire undertaking, operations and businesses of the Transferor Companies with the Transferee Company in order to benefit the stakeholders of all the Companies. Accordingly, the Board of Directors of the Transferor Companies and the Transferee Company have formulated this scheme of Amalgamation for the transfer and vesting of entire undertaking and business of the Transferor Companies with and into the Transferee Company pursuant to the provisions of section 230 to section 232 and other relevant provisions of the Act.

# 4. Parts of the Scheme:

This scheme of Amalgamation is divided into the following parts

- Part I deals with definition of the terms used in the scheme of Amalgamation and sets out the share capital of the Transferor Companies and the Transferee Company;
- Part II deals with the Transfer and vesting of Undertaking (as hereinafter defined) of the Transferor Companies to and in the Transferee Company;
- iii. Part III deals with the cancellation of entire issued, subscribed and paid-up share capital of the respective Transferor Company, which is held by the Transferee Company along with itsjoint shareholders
- iv. Part IV deals with the accounting treatment of the amalgamation in the books of the Transferee Company.
- Part V deals with the dissolution of the Transferor Companies and the general terms and conditions applicable to this scheme of amalgamation and other matters consequential and integrally connected thereto.
- **A.** The amalgamation of the Transferor Companies with the Transferee Company, pursuant to and in accordance with this Scheme, shall take place with effect from the Appointed Date and shall be in accordance with the relevant provisions of the Income Tax Act, 1961 including but not limited to Section 2(1B) and Section 47 thereof.

# PART I

#### **DEFINITIONS AND SHARE CAPITAL**

#### 5. DEFINITIONS

In this scheme, unless repugnant to the meaning or context thereof, the following expressions shall have the following meaning:

- 5.1 **"ACT"** means the Companies Act, 2013 the rules and regulations made thereunder and will include any statutory re-enactment or amendment(s) thereto, from time to time;
- 5.2 "Appointed Date" means 01<sup>st</sup> April, 2019 or such other date as the NCLT may direct.
- 5.3 **"Board of Directors"** or **"Board"** in relation to each of the Transferor Company and Transferee Company as the case may be, means the Board of Directors of such Company(s) and shall include a Committee duly constituted and authorized for the purposes of matters pertaining to the amalgamation, the Scheme and/or any other matter relating thereto.
- 5.4 **"Companies"** means the Transferor Companies and the Transferee Company referred collectively.
- 5.5 "Effective Date" means the last of the dates on which the certified copies or the authenticated copies of the order of National Company Law Tribunal sectioning of the Scheme, are filed with the Respective Registrar of Companies by the Transferor Companies and the Transferee Company. Any reference to this scheme to the date of "coming into effect of this scheme" or "effective date of this scheme" or scheme taking effect, shall means the effective date;
- 5.6 **"Governmental Authority**" means any applicable central, state or local government, legislative body, regulatory or administrative authority, agency or commission or any court, tribunal, board, bureau or instrumentality thereof or arbitration or arbitral body having jurisdiction over the territory of India including but not limited to

Securities and Exchange of India, Stock Exchanges, Registrar of Companies, competition Commission of India, National Company Law Tribunal (to be constituted under Company Act, 2013)

- 5.7 **"Employees"** means the staff, workman and employees on the pay roles of the Transferor Companies.
- 5.8 "NCLT" means the National Company Law Tribunal, Mumbai Bench having jurisdiction in relation to the First Transferor Company and Second Transferor Company and The Transferee Company and National Company Law Tribunal, New Delhi Bench having jurisdiction in relation to Third Transferor Company and Fourth Transferor Company and as the context may admit;
- 5.9 **"Registrar of Companies"** means the Registrar of Companies, Delhi and / or Registrar of Companies, Mumbai.
- 5.10 **"SCHEME"** means this scheme of amalgamation, as amended or modified in accordance with the provisions thereof.
- 5.11 **"The First Transferor Company"** means **MIA INFRASTRUCTURE PRIVATE LIMITED**" (CIN: U74900MH2014PTC252385) a company incorporated under the Provisions of the Companies Act, 1956 and having its Registered Office at E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai – 400033.
- 5.12 "The Second Transferor Company" means "WELLKEPT FACILITY MANAGEMENT SERVICES PRIVATE LIMITED" (CIN: U93000MH2008PTC177346), a company incorporated under the Provisions of the Companies Act, 1956 and having its Registered Office at E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai – 400033.
- 5.13 **"The Third Transferor Company"** means **ACME LIVING SOLUTIONS PRIVATE LIMITED"** (CIN: U45209DL2008PTC178023), a company incorporated under the Provisions of the Companies Act, 1956 and having its Registered Office

at Tata Services Ltd., Jeevan Bharti, Tower - 1, 10th Floor,124, Connaught Circus, New Delhi - 110001.'

- 5.14 **"The Fourth Transferor Company"** means **"TRIF GURGAON HOUSING PROJECTS PRIVATE LIMITED"** (CIN: U74900DL2009PTC188404), a company incorporated under the Provisions of the Companies Act, 1956 and having its Registered Office at C/O Tata Services Ltd., Jeevan Bharti, Tower - 1, 10th Floor,124, Connaught Circus, New Delhi – 110001.'
- 5.15 "The Transferor Company" means each of the First Transferor Company, Second Transferor Company, Third Transferor Company and Fourth Transferor Company, referred individually.
- 5.16 **"The Transferor Companies"** means the First and Second and Third and Fourth Transferor Company, referred collectively.
- 5.17 "The Transferee Company" Means TATA REALTY AND INFRASTRUCTURE LIMITED" (CIN: U70102MH2007PLC168300), a company incorporated under the Provisions of the Companies Act, 1956 and having its Registered Office at E Block, Voltas Premises, T B Kadam Marg, Chinchpokli, Mumbai – 400033.
- 5.18 **"Undertaking"** means the whole undertaking and the entire business of the Transferor Company as going Concern, including (without limitation):
- a. All assets and properties of the Transferor Company as on the Appointed Date i.e. all the undertakings, the entire business, all the properties (tangible or intangible), offices, residential and other premises, capital work in progress, furniture, fixture, office equipment, investments of all kinds and in all forms, cash balances with banks, loans, advances, contingent rights or benefits, receivables, benefit of any deposits, financial assets, benefit of any security arrangements, reversions, powers, authorities, allotments, approvals, permissions, permits, rights, entitlements, guarantees, authorizations, approvals, agreements, contracts, licenses, registrations, tenancies, benefits of all taxes right to carry forward and set off unabsorbed losses and

depreciation, privileges and rights under State tariff regulations and under various laws, avail of telephones, telexes, facsimile, email, interest, electricity and other services, reserves, provisions, funds, benefits of all agreements, all records, files, papers, computer programs, manuals, data, and other records, and all other interests of whatsoever nature belonging to or in the ownership, power, possession or control of or vested in or granted in favour of or held for the benefit of or enjoyed by the Transferor Companies or which have accrued to the Transferor Companies as on the Appointed Date, whether in India or abroad, of whatsoever nature and wherever situated (hereinafter referred to as the "Assets");

- b. All liabilities including, without being limited to, secured and unsecured debts (whether in Indian rupees or foreign currency), sundry creditors, liabilities, bank loans, (including contingent liabilities), duties and obligations of the Transferor Company, of every kind, nature and description whatsoever and howsoever arising, raised or incurred or utilized;
- c. All agreements, rights, contracts, entitlements, permits, licenses, approvals, authorizations, concessions, consents, quota rights, engagements, arrangements, authorities, allotments, security arrangements (to the extent provided herein), benefits of any guarantees, reversions, powers and all other approvals of every kind, nature and description whatsoever relating to the business activities and operations of the Transferor Companies;
- d. All earnest monies and/or security or other deposits paid by the Transferor Company.
- e. All permanent employees engaged by the Transferor Company as on the Effective Date.
- f. All quotas, rights, entitlements, export/import incentives and benefits including advance licenses, bids, tenders (at any stage as it may be), letters of intent, expressions of interest, development rights (whatever vested or potential and whether under agreements or otherwise), subsidies, tenancies in relation to office, benefit of any deposits
privileges, all other rights, receivables, powers and facilities of every kind, nature and description whatsoever, rights to use and avail of telephones, telexes, facsimile connections and installations, utilities, electricity and other services, provisions and benefits of all agreements, contracts and arrangements, including technological licensing agreements, and all other interests in connection with or relating thereto;

- g. All intellectual property rights, trade marks created, developed or invented by employees concentrated on the research, development or marketing of products (including process development or enhancement) in connection with the Transferor Company;
- h. All benefits and privileges under letters of permission and letters, of approvals in respect all tax credits, including CENVAT credits, refunds; reimbursements, claims, exemptions, benefits under service tax laws, value added tax, purchase tax, sales tax, input credit under GST or any other duty or tax or cess or imposts under central or state law including sales tax deferrals, advance taxes, tax deducted at source, tax credits, right to carry forward and set-off unabsorbed losses, if any and depreciation, deductions and benefits under the Income-tax Act, 1961, as well as any recognition of the In-house Research and Development unit with the Department of Scientific & Industrial Research or any Government Authority;
- i. All terms and words not defined in this Scheme shall, unless repugnant or contrary to the context or meaning thereof, have the same meaning as ascribed to them under the Act and other applicable laws, rules, regulations and byelaws, as the case may be, or any statutory amendment(s) or re-enactment thereof, for the time being in force.

### 6. DATE OF TAKING EFFECT AND OPERATIVE DATE

The Scheme set out herein in its present form or with any modification(s) approved or imposed or directed by the NCLT or made as per Clause 20 of the Scheme, shall be effective from the Appointed Date but shall be operative from the Effective Date.

## 7. SHARE CAPITAL

A. The Authorised, Subscribed and Paid-up share capital of MIA
INFRASTRUCTURE PRIVATE LIMITED as on 31<sup>st</sup> March,
2018 was as under:

Particulars	Amount in Rupees
Authorised Share Capital	
28,00,000 Equity Shares of Rs.	2,80,00,000/-
10/- each	
Total	2,80,00,000/-
Issued, Subscribed and	
Paid-up Share Capital	
27,70,000 Equity Shares of Rs.	2,77,00,000/-
10/- each fully paid-up	
Total	2,77,00,000/-

B. The Authorised, Subscribed and Paid-up share capital of
WELLKEPT FACILITY MANAGEMENT SERVICES
PRIVATE LIMITED as on 31<sup>st</sup> March, 2018 was as under:

Particulars	Amount in Rupees
Authorised Share Capital	
10,00,000/- Equity Shares of	1,00,00,000/-
Rs. 10/- each.	
Total	1,00,00,000/-
Issued, Subscribed and	
Paid-up Share Capital	
4,00,000 Equity Shares of Rs.	40,00,000/-
10/- each fully paid-up	
Total	40,00,000/-

C. The Authorised, Subscribed and Paid-up share capital of ACME
LIVING SOLUTIONS PRIVATE LIMITED as on 31<sup>st</sup> March,
2018 was as under:

	· · _ 1
Particulars	Amount in Rupees
	r mite unit in runp etce

Authorised Share Capital	
50,000 Equity Shares of Rs.	5,00,000
10/- each	
Total	5,00,000/-
Issued, Subscribed and	
Paid-up Share Capital	
50,000 Equity Shares of Rs.	5,00,000
10/- each fully paid-up	
Total	5,00,000/-

 D. The Authorised, Subscribed and Paid-up share capital of TRIF
GURGAON HOUSING PROJECTS PRIVATE LIMITED as on 31<sup>st</sup> March, 2018 was as under:

Particulars	Amount in Rupees
Authorised Share Capital	
50,000 Equity Shares of Rs.	5,00,000/-
10/- each	
Total	5,00,000/-
Issued, Subscribed and	
Paid-up Share Capital	
50,000 Equity Shares of Rs.	5,00,000/-
10/- each fully paid-up	
Total	5,00,000/-

Subsequent to the above balance sheet date, there is no change in the Capital Structure of Transferor Companies. Transferor Companies are Wholly owned subsidiary of Transferee Company. Entire Paid up share capital of the Transferor Companies are held by the Transferee Company and itsjoint shareholders.

 E. The Authorised, Subscribed and Paid-up share capital of TATA REALTY AND INFRASTRUCTURE LIMITED as on 31<sup>st</sup> March, 2018 was as under:

Particulars	Amount in Rupees
-------------	------------------

Authorised Share Capital	
2,00,00,00,000 Equity Shares of Rs.	20,00,00,00,000/-
10/- each.; and	
1,00,00,00,000 5% Non-Cumulative	
Convertible Preference Shares of Rs.	10,00,00,00,000/-
10/- each.	
Total	30,00,00,00,000
Issued, Subscribed and Paid-up	
Share Capital	
1,01,73,07,692 Equity Shares of Rs.	10,17,30,76,920/-
10/- each fully paid-up	
Total	10,17,30,76,920/-

Subsequent to the above balance sheet date, there is change in the Capital Structure of Transferee Company. The Revised Capital Structure of Transferee Company as on 30th November, 2018 is as follows

Particulars	Amount in
	Rupees
Authorised Share Capital	
3,00,00,00,000 Equity Shares	30,00,00,00,000/-
of Rs. 10/- each.	
Total	30,00,00,00,000
Issued, Subscribed and	
Paid-up Share Capital	

1,01,73,07,692 Equity Shares	10,17,30,76,920/-
of Rs. 10/- each fully paid-	
up	
Total	10,17,30,76,920/-

Subsequent to the above, there is again change in the Capital Structure of Transferee Company. The Revised Capital Structure of Transferee Company as on 31<sup>st</sup> March, 2020 is as follows:

Particulars	Amount in
	Rupees
Authorised Share Capital	
8,00,00,00,000 Equity Shares	80,00,00,00,000/-
of Rs. 10/- each.	
Total	80,00,00,00,000
Issued, Subscribed and	
Paid-up Share Capital	
1,61,73,07,692 Equity Shares	16,17,30,76,920/-
of Rs. 10/- each fully paid-	
up	
Total	16,17,30,76,920/-

# PART II

### TRANSFER AND VESTING OF UNDERTAKING

## 8. TRANSFER OF UNDERTAKING

- 8.1 Upon coming into effect off this scheme and with the effect of the appointed date, the Undertaking, pursuant to the section of this scheme by NCLT under and accordance with the section of 230 to 232 and other applicable provisions, if any, of the Act, shall stand transferred and vested in and/or be deemed to have been transferred to and vested in the Transferee Company as a going concern without any further act, instrument, deed, matter or thing from the Appointed Date; so as to become estates, assets, rights, title, interests and authorities of the Transferee Company, by virtue of and in the manner provided in this scheme.
- 8.2 Without prejudice to the generality of Clause 8.1 above, upon coming into effect of this scheme and with effect from Appointed Date, all its properties and assets, (whether movable or immovable, tangible or intangible), land and building, plant and machinery, equipment, furniture, fixtures, vehicles, leasehold assets and other properties, real, in possession or reversion, present and contingent assets (whether tangible or intangible) of whatsoever nature, all the debts, liabilities, duties and obligations of the Transferor Companies of every description and also including, without limitation, all the movables and immovable properties and assets of the Transferor Companies comprising amongst other investments, vehicles, furniture and fixtures, computers, office equipment, electrical installations, water connections, telephones, telex, facsimile and other communication facilities and business licenses, permits, authorizations, if any, rights and benefits of all agreements (including all such contracts, scheme, deeds, bonds and other instruments of whatsoever nature to which the Transferor Companies are party) and all other interests, rights and powers of every kind, nature and description whatsoever, privileges, liberties, easements, advantages, benefits and approvals, advance and other taxes paid to the authorities, brand names, trademarks, copy rights, lease, tenancy rights, statutory permissions, consents and registrations, and all other Intellectual Property Rights and other intangible assets of whatsoever nature, pursuant to the Order of the

NCLT and pursuant to provisions of Sections 230 to 232 and other applicable provisions of the Companies Act, 2013 and without further act, instrument or deed, but subject to the charges affecting the same be transferred and/or deemed to be transferred to and vested in Transferee Company on a going concern basis so as to become the assets and liabilities of the Transferee Company.

- 8.3 In respect of all the movable assets of the Transferor Companies and the assets which are otherwise capable of transfer by physical delivery or endorsement and delivery, including cash on hand, shall be so transferred to Transferee Company and deemed to have been physically handed over by physical delivery or by endorsement and delivery, as the case may be, to Transferee Company to the end and intent that the property and benefit therein possess to Transferee Company with effect from the Appointed Date.
- 8.4 In respect of any assets of the Transferor Companies other than those mentioned in Clause 8.3 above, including actionable claims, sundry debtors, outstanding loans, advances recoverable in cash or kind or for value to be received and deposits with the Government, Semi-Government, local and other authorities and bodies and customers, The Transferee Company may, issue notices in such form as The Transferee Company may deem fit and proper stating that pursuant to the NCLT having sanctioned this Scheme between the Transferor Companies and the Transferee Company under Section 230 to 232 of the Companies Act, 2013, the relevant debt, loan, advance or other asset, to be or made good or held on account of the Transferee Company, as the person entitled thereto, to the end and intent that the right of the Transferor Companies to recover or realise the same stands transferred to the Transferee Company and that appropriate entries should be passed in their respective books to record the aforesaid changes.
- 8.5 With effect from the Appointed Date, all debts, liabilities, contingent liabilities, duties and obligations of the Transferor Companies, as on the Appointed Date whether provided for or not in the books of accounts of the Transferor Companies and all other liabilities which may accrue or arise after the Appointed date but which relates to the period on or upto the day of the Appointed

Date shall, pursuant to the Order of the NCLT under section 230 to 232 and other applicable provisions of the Companies Act, 2013, without any further act or deed, be transferred or deemed to be transferred to and vested in and assumed by the Transferee Company pursuant to the provisions of Sections 230 to 232 of the Companies Act, 2013 (or such other applicable provisions of Companies Act, 2013), so as to become the liabilities of the Transferee Company on the same terms and conditions as were applicable to Transferor Companies.

- 8.6 the existing securities, mortgages, charges, encumbrances or liens, if any, as on the Appointed Date and those created by the Transferor Companies after the Appointed Date over the assets of the Transferor Companies transferred to the Transferee Company shall, after the Effective Date, continue to relate and attach to such assets or any part thereof to which they are related or attached prior to the Effective Date.
- 8.7 All the licenses, permits, entitlements, approvals, permissions, registrations, incentives, tax deferrals, exemptions, benefits (including Sales Tax and Service Tax), subsidies, tenancy rights, liberties, special statues and other benefits or privileges enjoyed or conferred upon or held or availed of by the Transferor Companies and all rights and benefits that have accrued or which may accrue to the Transferor Companies whether on, or before or after the Appointed Date including income tax benefits and exemptions, shall under the provisions of Sections 230 to 232 of the Companies Act, 2013 (or such other applicable provisions of applicable law, if any without any further act, instrument or deed, cost or charge, be and stand transferred to the Transferee Company.
- 8.8 Without prejudice to the provisions of the forgoing clauses and upon the effectiveness of this scheme, the Transferor Companies and the Transferee Company shall execute all such instruments or documents or to do all such acts, deeds, as may be required including filing of necessary particulars and/or modification(s) of charge with the Registrar of Companies to give formal effect to the above provisions, if required.

8.9 This Scheme has been drawn up to comply with the conditions relating to "Amalgamation" as specified under Section 2(1B) of the Income Tax Act, 1961. If any terms or provisions of the scheme is/are inconsistent with the provisions of the said Section 2(1B), such provisions of the said Section 2(1B) shall prevail and the scheme shall stand modified to the extent necessary to comply with the said section. Such modification will, however not affect the other parts of the scheme.

### 9. CONTRACTS, DEEDS, ETC.

- 9.1 Upon the Coming into effect of the Scheme, all contracts, deeds, bonds, agreements, schemes, arrangements, assurances and other instrument of whatsoever nature to which the Transferor Companies are party or to the Benefit of which the Transferee Company may be eligible, and which are substituting or have effect immediately before the Effective Date, shall continue in full force and effected by, for or against or in favour of, as the case may be, the Transferee Company and may be enforced as fully, effectually as if, instead of the Transferor Company had been party or beneficiary or oblige or obligor thereto or thereunder.
- 9.2 Without prejudice to the provisions of this Scheme and notwithstanding the fact the vesting of Undertaking occurs by virtue of Scheme itself, the Transferee Company may, any time after coming into effect of this Scheme in accordance with the provisions hereof, if so required under any law or otherwise, take such actions and execute such deeds (including deeds of adherence), confirmations or other writings or arrangement to which the Transferor Companies are the Party or any writings as may be necessary in order to give formal effect to the provisions of this Scheme, be deemed to be authorised to execute any of such writings on behalf of the Transferor Company and to carry out or perform all such formalities or compliances referred to above on the part of the Transferee Company.
- 9.3 For the avoidance of doubt and without prejudice to the generality of the foregoing, it is clarified that upon the coming into effect of this Scheme, all consents, permissions, licences, certificates, clearances, authorities, power of attorney given by, issued to or executed in favour of the Transferor Company, shall without any

further act or deed, stand transferred to the Transferee Company, as if the same were originally given by, issued to or executed in favour of the Transferee Company, and the Transferee Company shall be bound by the terms thereof, the obligations and duties thereunder, and the rights and benefits under the same shall be available to the Transferee Company. The Transferee Company shall receive relevant approvals for the concerned Government Authorities as may be necessary in this behalf.

### 10. LEGAL PROCEEDINGS

On and form the Appointed Date, all suits, actions, claims and legal proceedings by or against the Transferor Company pending and/or arising on or before Effective Date shall be constituted and/or enforced as desired by the Transferee Company and on or from the Effective Date, shall be continued and/or enforced by or against the Transferee Company as effectually and in the same manner and to the same extent as if the same had been originally instituted and/or pending and/ or arising by or against the Transferee Company. On and From the Effective Date, the Transferee Company shall have the right to initiate, defend, compromise or otherwise deal with any legal proceedings relating to the Undertaking, in the same manner and to same extent as would or might have been initiated by the Transferor Company as the case may be, had the scheme not be made; If any suit, appeal or other proceedings relating to the Undertaking, of whatever nature by or against the Transferor Company be pending, the same shall not be abate or discontinued or in any way be prejudicially affected by reason of the amalgamation of the Undertaking or by anything contained in this Scheme but the proceedings may be continued, prosecuted and enforced by or against the Transferor Company as if the Scheme has not been made.

### **11. EMPLOYEES**

11.1 All the employees of the Transferor Companies, if any, who are in service on the date immediately preceding the Effective Date shall, on and from the Effective Date become and be engaged as the employees of the Transferee Company, without any break or interruption in service as a result of the transfer and on terms and conditions not less favourable than those on which they are engaged by the Transferor Companies immediately preceding the Effective Date. Services of the Employees of the Transferor Companies shall be taken into account from the date of their respective appointment with the Transferor Companies for the purposes of all retirement benefits and all other entitlements for which they may be eligible. Transferee Company further agrees that for the purpose of payment of any retrenchment compensation, if any, such past services with the Transferor Companies shall also be taken into account.

- **11.2** The services of such employees shall not be treated as being broken or interrupted for the purpose of Provident Fund or Gratuity or Superannuation or other statutory purposes and for all purposes will be reckoned from the date of their respective appointments with the Transferor Companies.
- It is provided that as far as the Provident Fund and pension 11.3 and/or Superannuation Fund or any other special fund created or existing for the benefit of the staff, workmen and other employees of the Transferor Companies are concerned, upon the Scheme becoming effective, Transferee Company shall stand substituted for the Transferor Companies in respect of the employees transferred for all purposes whatsoever relating to the administration or operation of such Funds or Trusts or in relation to the obligation to make contribution to the said Funds or Trust in accordance with the provisions of such Funds or Trust as provided in the respective Trust Deeds or other documents. It is the aim and the intent of the Scheme that all the rights, duties, powers and obligations of the Transferor Companies in relation to such Funds or Trusts shall become those of the Transferee Company. The Trustee including the Board of Directors of the Transferor Companies and the Transferee Company or through any Committee/person duly authorised by the Board of Directors in this regard shall be entitled to adopt such course of action in this regard as may be advised, provided however, that there shall be no discontinuation or breakage in the services of employees of the Transferor Companies.
- 11.4 With effect from the first of the dates of filing of this Scheme with NCLT and up to and including the Effective Date, the Transferor Companies shall not vary or modify the terms and conditions of

employment of any of its employees, except with the written consent of the Transferee Company.

## **12. CONDUCT OF BUSINESS UNTIL EFFECTIVE DATE**

- a. With effect from the Appointed Date and up to and including the Effective Date:
- 12.1 Transferor Company shall carry on and shall be deemed to have carried on all its business and activities as hitherto and shall hold and stand possessed of and shall be deemed to have held and stood possessed of the Undertaking on account of, and for the benefit of and in trust for, the Transferee Company.
- 12.2 All the profits or income accruing or arising to the Transferor Company, and all expenditure or losses arising or incurred (including all taxes, if any, paid or accruing in respect of any profits and income) by the Transferor Company shall, for all purposes, be treated and be deemed to be and accrue as the profits or income or as the case may be, expenditure or losses (including taxes) of the Transferee Company.
- 12.3 Any of the rights, powers, authorities and privileges attached or related or pertaining to and exercised by or available to the Transferor Company shall be deemed to have been exercised by the Transferor Company for and on behalf of and as agent for the Transferee Company.

Similarly, any of the obligations, duties and commitments attached, related or pertaining to the Undertaking that have been undertaken or discharged by the Transferor Company shall be deemed to have been undertaken or discharged for and on behalf of and as agent for the Transferee Company.

- b. With effect from the first of the date of filing of this Scheme with the NCLT and up to and including the Effective Date:
- 12.4 The Transferor Company shall preserve and carry on its business and activities with reasonable diligence and business prudence and shall not undertake any additional financial commitments of any nature whatsoever, borrow any amounts nor incur any other liabilities or expenditure, issue any additional guarantees,

indemnities, letters of comfort or commitments either for itself or on behalf of its group Company or any third party or sell, transfer, alienate, charge, mortgage or encumber or deal with the Undertaking or any part thereof save and except in each case in the following circumstances:

- i. if the same is in its ordinary course of business as carried on by it as on the date of filing this Scheme with NCLT; or
- ii. if the same is permitted by this Scheme; or
- iii. if consent of the Board of Directors of the Transferee Company has been obtained.
- 12.5 The Transferor Company shall not take, enter into, perform or undertake, as applicable (i) any material decision in relation to its business and operations other than decisions already taken prior to approval of the Scheme by the respective Board of Directors (ii) any agreement or transaction; and (iii) any new business, or discontinue any existing business or change the capacity of facilities.:(iv) such other matters as the Transferee Company may notify from time to time save and except in each case in the following circumstances:
  - a. if the same is in its ordinary course of business as carried onby it as on the date of filing this Scheme with NCLT; or
  - b. if the same is permitted by this Scheme; or
  - c. if consent of the Board of Directors of the Transferee Company has been obtained.

### 13. Treatment with Taxes

- 13.1 Tax liabilities under the Income-tax Act, 1961, Goods and Service Tax Act 2017, Wealth Tax Act, 1957, Customs Act, 1962, Central Excise Act, 1944, Maharashtra Value Added Tax Act, 2002, Central Sales Tax Act, 1956, any other state Sales Tax /Value Added Tax laws, service tax, luxury tax, stamp laws or other applicable laws/ regulations (hereinafter in this Clause referred to as "Tax Laws") dealing with taxes/ duties/ levies allocable or related to the business of the Transferor Company to the extent not provided for or covered by tax provision in the accounts made as on the date immediately preceding the Appointed Date shall be transferred to Transferee Company.
  - 13.2 All taxes (including income tax, Goods and Service Tax, wealth tax, sales tax, excise duty, customs duty, service tax, luxury tax, VAT, etc.) paid or payable by the Transferor Company in respect of the operations and/or the profits of the business on and from the Appointed Date, shall be on account of the Transferee Company and, insofar as it relates to the tax payment(including without limitation income tax, Goods and Service Tax, wealth tax, sales tax, excise duty, customs duty, service tax, luxury tax, VAT, etc.), whether by way of deduction at source, advance tax or otherwise howsoever, by the Transferor Company in respect of the profits or activities or operation of the business on and from the Appointed Date, the same shall be deemed to be the corresponding item paid by the Transferee Company, and shall, in all proceedings, be dealt with accordingly.
  - 13.3 Without prejudice to the generality of the above, all benefits including under the income tax, Goods and Service Tax, sales tax, excise duty, customs duty, service tax, luxury tax, VAT, etc., to which the Transferor Company are entitled to in terms of the applicable Tax Laws of the Union and State Governments, shall be available to and vest in the Transferee Company.
  - 13.4 During the Period between the Appointed Date and the Effective Date, the Transferor Companies shall continue to do all acts, necessary and in connection with all tax assessment proceedings/ appeals of whatsoever nature, pending and/or arising by or against

the transferor Companies. With Effect from the Effective Date, the Tax proceedings shall be continued and enforced by and against the Transferee Company in accordance with applicable law.

13.5 The Transferee Company shall be entitled to file/revise its income tax, Goods and Service Tax, sales tax, excise duty, customs duty, service tax, luxury tax, VAT, etc. returns if required, and shall have right to claim refunds, Goods and Service Tax Credit, CENVAT credit or any other credits etc. if any, as may be required, consequent to this Scheme becoming effective.

### 14. SAVING OF CONCLUDED TRANSACTIONS

Subject to the terms of this Scheme, the transfer and vesting of the Undertaking of the Transferor Company under Clause 8 of this Scheme shall not affect any transactions or proceedings already concluded by the Transferor Company on or before the Appointed Date or concluded after the Appointed Date till the Effective Date, to the end and intent that the Transferee Company accepts and adopts all acts, deeds and things made, done and executed by the Transferor Company as acts, deeds and things made, done and executed by or on behalf of the Transferee Company.

### PART III

## **15. CANCELLATION OF SHARES**

- 15.1 The entire issued, subscribed and paid-up share capital of the Transferor Company is held by the Transferee Companies along with itsJoint Shareholders. Upon the scheme becoming effective, no shares of the Transferee Company shall be allotted in lieu or exchange of its holding in the Transferor Companies and the Paid up share capital of the Transferor Companies shall stand cancelled.
- 15.2 The Shares or the Share Certificate of the Transferor Companies shall, without any further application, act, instrument, or deed, be deemed to have been automatically cancelled and be of no effect on and from the Effective Date.

## PART IV

## 16. ACCOUNTING TREATMENT AND DIVIDENDS

- 16.1 Upon the coming into effect of this Scheme and with effect from the Appointed Date, for the purpose of accounting for and dealing with the value of the assets and liabilities in the books of the Transferee Company, all assets and liabilities recorded in the books of the Transferor Company and transferred to and vested in the Transferee Company pursuant to this scheme shall be recorded by the Transferee Company as per the pooling of interest method specified in Indian Accounting Standards (Ind AS) 103 and other applicable Indian Accounting Standards, in terms of Section 133 of the Companies Act, 2013 for the purpose of accounting for amalgamation.
- 16.2 Subject to provisions of this Scheme, the Transferee Company shall abide by Indian Accounting Standards i.e. Ind AS 103 and others issued by the Ministry of Corporate Affairs.
- 16.3 The amalgamation of Transferor Company with the Transferee Company in terms of this Scheme shall take place with effect from the Appointed Date and shall be in accordance with the provisions of Section 2(1B) of the Income Tax Act, 1961.
- 16.4 Inter Company balances shall be cancelled.
- 16.5 Comparative accounting period presented in the financial statements of Transferee Company shall be restated for the accounting impact of amalgamation, as stated above.
- 16.6 The identity of reserves of the Transferor Companies, if any, shall be preserved and they shall appear in the financial statements of the Transferee Company in the same form and manner in which they appeared in the financial statements of the Transferor Company prior to this Scheme becoming effective.

### 17. DECLARATION OF DIVIDEND

- 17.1 During the period between the Appointed Date and up to and including the Effective Date, the Transferor Company shall not declare any dividend without the prior written consent of the Board of Directors of the Transferee Company.
- 17.2 For the avoidance of doubt, it is hereby declared that nothing in the Scheme shall prevent the Transferee Company from declaring and

paying dividends, whether interim or final, to its equity shareholders as on the Record Date for the purpose of dividend and the shareholders of the Transferor Company shall not be entitled to dividend, if any, declared by the Transferee Company prior to the Effective Date.

### PART V

# 18. DISSOLUTION OF TRANSFEROR COMPANIES AND GENERAL TERMS AND CONDITIONS

On the coming into effect of this Scheme, the Transferor Company shall stand dissolved without winding-up, and the Board of Directors and any committees thereof of the Transferor Company shall without any further act, instrument or deed be and stand dissolved.

### 19. VALIDITY OF EXISTING RESOLUTIONS, ETC.

Upon the coming into effect of this Scheme, the resolutions, if any, of the Transferor Company, which are valid and subsisting on the Effective Date, shall continue to be valid and subsisting and be considered as resolutions of the Transferee Company and if any such resolutions have any monetary limits approved under the provisions of the Act, or any other applicable statutory provisions, then the said limits shall be added to the limits, if any, under like resolutions passed by the Transferee Company and shall constitute the aggregate of the said limits in the Transferee Company.

### 20. MODIFICATION OF SCHEME

The Transferor Company and the Transferee Company through their respective Boards of Directors including Committees of Directors or other persons, duly authorised by the respective Boards in this regard, may make, or assent to, any alteration or modification to this Scheme or to any conditions or limitations, which the NCLT or any other Competent Authority may deem fit to direct, approve or impose and may give such directions including an order of dissolution of the Transferor Company without process of winding up as they may consider necessary, to settle any doubt, question or difficulty, arising under the scheme or in regard to its implementation or in any manner connected therewith and to do and to execute all such acts, deeds, matters and things necessary for putting this Scheme into effect, or to review the portion relating to the satisfaction of the conditions to this scheme and if necessary, to waive any of those (to the extent permitted under law) for bringing this scheme into effect.

## 21. FILING OF APPLICATIONS

The Transferor Company and the Transferee Company shall use their best efforts to make and file all applications and petitions under Sections 230 to 232 and other applicable provisions of the Act, before the respective NCLT having jurisdiction for sanction of this Scheme under the provisions of law, and shall apply for such approvals as may be required under law.

### 22. APPROVALS

The Transferee Company shall be entitled, pending the sanction of the Scheme, to apply to any Governmental Authority, if required, under any law for such consents and approvals which the Transferee Company may require to own the Undertaking and to carry on the business of the Transferor Company.

### 23. SCHEME CONDITIONAL UPON SANCTIONS, ETC.

- 23.1 This Scheme is conditional upon and subject to:
- 23.2 The Scheme being agreed to by the requisite majority of the respective classes of members and/or creditors of the Transferor Company and of the Transferee Company as required under the Act and the requisite orders of the Jurisdictional NCLT being obtained; and
- 23.3 The certified copies of the orders of the Jurisdictional NCLT sanctioning this Scheme being filed with the Registrar of Companies, Maharashtra, Mumbai and the Registrar of Companies, New Delhi.

## 24. COSTS, CHARGES, EXPENSES AND STAMP DUTY

All costs, charges and expenses (including any taxes and duties) incurred or payable by the Transferor Company and Transferee Company in relation to or in connection with this Scheme and incidental to the completion of the amalgamation of the Transferor Company with the Transferee Company in pursuance of this Scheme, including stamp duty on the orders of NCLT, if any, and to the extent applicable and payable, shall be borne and paid by the Transferee Company.

\*\*\*\*\*

Chartered Accountants Lotus Corporate Park 1<sup>st</sup> Floor, Wing A – G CTS No.185/A, Jay Coach Off Western Express Highway Goregaon (East) Mumbai – 400 063 Maharashtra, India

Tel: +91 22 6245 1000 Fax: +91 22 6245 1001

#### INDEPENDENT AUDITOR'S REPORT

To The Members of MIA Infrastructure Private Limited Report on the Audit of the Ind AS Financial Statements

### Opinion

We have audited the accompanying Ind AS financial statements of MIA Infrastructure Private Limited ("the Company"), which comprise the Balance Sheet as at March 31, 2019, and the Statement of Profit and Loss (including Other Comprehensive Income), the Statement of Cash Flows and the Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended, ("Ind AS") and other accounting principles generally accepted in India, of the state of affairs of the Company as at March 31, 2019, and its loss, its cash flows and the changes in equity for the year ended on that date.

### **Basis for Opinion**

SKINS

Chartered

Accountants

We conducted our audit of the Ind AS financial statements in accordance with the Standards on Auditing specified under section 143(10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Ind AS Financial Statements section of our report. We are independent of the Company in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the Ind AS financial statements.

### Material uncertainty related to Going Concern

We draw attention to Note 2(b) to the Ind As financial statements, which indicates that the Company has incurred a net loss of 25.72 lakhs during the year ended 31<sup>st</sup> March, 2019, the Company's current liabilities exceeded its current assets by 30.67 lakhs. The. These events or conditions, along with other matters as set forth in Note 2(b), indicate that a material uncertainty exists that may cast significant doubt on the Company's ability to continue as a going concern. However, the Ind As financial statements of the Company have been prepared on a going concern basis for the reasons stated in the said Note.

Our opinion is not modified in respect of this matter

### Information Other than the Financial Statements and Auditor's Report Thereon

The Company's Board of Directors is responsible for the other information. The other information comprises the Director's report which we obtained prior to the date of this auditor's report

Our opinion on the Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.

In connection with our audit of the Ind AS financial statements, our responsibility is to read the other information and, in doing so, consider whether the other information is materially

Regd. Office: Indiabulls Finance Centre, Tower 3, 27<sup>th</sup> - 32<sup>nd</sup> Floor, Senapati Bapat Marg, Elphinstone Road (West), Mumbai - 400 013, Maharashtra, India.

SKINS

Chartered Accountants

\*

inconsistent with the Ind AS financial statements or our knowledge obtained during the course of our audit or otherwise appears to be materially misstated.

• If, based on the work we have performed, we conclude that there is a material misstatement of this other information, we are required to report that fact. We have nothing to report in this regard.

## Management's Responsibility for the Ind AS Financial Statements

The Company's Board of Directors is responsible for the matters stated in section 134(5) of the Act with respect to the preparation of these Ind AS financial statements that give a true and fair view of the financial position, financial performance including other comprehensive income, cash flows and changes in equity of the Company in accordance with the Ind AS and other accounting principles generally accepted in India.

This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the Ind AS financial statement that give a true and fair view and are free from material misstatement, whether due to fraud or error.

In preparing the Ind AS financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those Board of Directors are also responsible for overseeing the Company's financial reporting process.

### Auditor's Responsibility for the Audit of the Ind AS Financial Statements

Our objectives are to obtain reasonable assurance about whether the Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these Ind AS financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

 Identify and assess the risks of material misstatement of the Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.

Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under section 143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Company has adequate internal financial controls system in place and the operating effectiveness of such controls.

- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to

events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.

 Evaluate the overall presentation, structure and content of the Ind AS financial statements, including the disclosures, and whether the Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Materiality is the magnitude of misstatements in the Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the Ind AS financial statements.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

### **Report on Other Legal and Regulatory Requirements**

- 1. As required by Section 143(3) of the Act, based on our audit we report, to the extent applicable that:
  - a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit.
  - b) In our opinion, proper books of account as required by law have been kept by the Company so far as it appears from our examination of those books
  - c) The Balance Sheet, the Statement of Profit and Loss including Other Comprehensive Income, the Statement of Cash Flows and Statement of Changes in Equity dealt with by this Report are in agreement with the relevant books of account.
  - d) In our opinion, the aforesaid Ind AS financial statements comply with the Ind AS specified under Section 133 of the Act.
  - e) The matter described in the Material uncertainty related to Going Concern section above, in our opinion, may have an adverse effect on the functioning of the Company.



On the basis of the written representations received from the directors as on March 31, 2019 taken on record by the Board of Directors, none of the directors is disqualified as on March 31, 2019 from being appointed as a director in terms of Section 164(2) of the Act.

- g) Reporting on the adequacy of the internal financial controls over financial reporting of the Company and the operating effectiveness of such controls is not applicable to the company in view of the exemption available to the company in terms of the notification no. G.S.R. 583(E) dated 13 June 2017 issued by the Ministry of Corporate Affairs, Government of India, read with general circular No. 08/2017 dated 25 July 2017.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of section 197(16) of the Act, as amended.

In our opinion and to the best of our information and according to the explanations given to us, the Company being a private company, section 197 of the Act related to managerial remuneration is not applicable.

- i) With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended in our opinion and to the best of our information and according to the explanations given to us:
  - i. The Company does not have any pending litigations which would impact its financial position.
  - ii. The Company did not have any long-term contracts including derivative contracts for which there were any material foreseeable losses.
  - iii. There were no amounts, which were required to be transferred to the Investor Education and Protection Fund by the Company
- 2. As required by the Companies (Auditor's Report) Order, 2016 ("the Order") issued by the Central Government in terms of Section 143(11) of the Act, we give in "Annexure A" a statement on the matters specified in paragraphs 3 and 4 of the Order.

For Deloitte Haskins & Sells LLP

Chartered Accountants (Firm's Registration No.117366W/W-100018)

Africalia

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai, April 5, 2019

### ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT

(Referred to in paragraph 2 under 'Report on Other Legal and Regulatory Requirements' Section of our report of even date)

(i) The Company does not have any property, plant and equipment and hence reporting under clause of the CARO 2016 is not applicable.

(ii) The Company does not have any inventory and hence reporting under clause (ii) of the CARO 2016 is not applicable.

(iii) The Company has not granted any loans, secured or unsecured, to companies, firms, Limited Liability Partnerships or other parties covered in the register maintained under Section 189 of the Companies Act, 2013

(iv) The Company has not granted any loans, made investments or provided guarantees and hence reporting under clause (iv) of the CARO 2016 is not applicable.

(v) According to the information and explanations given to us, the Company has not accepted any deposit during the year as provided under Section 73 to 76 or any other relevant provisions of the Companies Act, 2013. There are no unclaimed deposits any time during the year.

(vi) The maintenance of cost records has not been specified by the Central Government under Section 148(1) of the Companies Act, 2013.

(vii) According to the information and explanations given to us, in respect of statutory dues:

- (a) The Company has been regular in depositing undisputed statutory dues, including Incometax, Service Tax / Goods & Services Tax, cess and other material statutory dues applicable to it to the appropriate authorities.
- (b) There were no undisputed amounts payable in respect of Income-tax, Service Tax / Goods & Services Tax, cess and other material statutory dues in arrears as at March 31, 2019 for a period of more than six months from the date they became payable.
- (c) There are no dues of Income-tax, Service Tax / Goods & Services Tax as on March 31, 2019 on account of disputes.

(viii) In our opinion and according to the information and explanations given to us, the Company has not defaulted in the repayment of loans or borrowings to debenture holders (Bank). The Company has not taken any loans from banks, financial institutions and Government.

(ix) The Company has not raised moneys by way of initial public offer or further public offer (including debt instruments) or term loans and hence reporting under clause (ix) of the CARO 2016 Order is not applicable.

(x) To the best of our knowledge and according to the information and explanations given to us, no fraud by the Company and no material fraud on the Company by its officers or employees has been ASKIN Shoticed or reported during the year.

Chartere [x] In our opinion and according to the information and explanations given to us, the Company is Accountant of required to have and does not have any managerial personnel and hence the reporting under

clause (xi) of the CARO 2016 Order for payment of managerial remuneration under Section 197 read with Schedule V of the Companies Act 2013 is not applicable.

(xii) The Company is not a Nidhi Company and hence reporting under clause (xii) of the CARO 2016 Order is not applicable.

(xiii) In our opinion and according to the information and explanations given to us the Company is in compliance with Section 177 and 188 of the Companies Act, 2013, where applicable, for all transactions with the related parties and the details of related party transactions have been disclosed in the financial statements etc. as required by the applicable accounting standards.

(xiv) During the year the Company has not made any preferential allotment or private placement of shares or fully or partly convertible debentures and hence reporting under clause (xiv) of CARO 2016 is not applicable to the Company.

(xv) In our opinion and according to the information and explanations given to us, during the year the Company has not entered into any non-cash transactions with its directors or directors of its holding, subsidiary or associate company or persons connected with them and hence provisions of Section 192 of the Companies Act, 2013 are not applicable.

(xvi) The Company is not required to be registered under Section 45-IA of the Reserve Bank of India Act, 1934.

### For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No. 117366W/W-100018)

Africhta

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai, April 5, 2019

### MIA Infrastructure Private Limited Balance Sheet as at 31 March 2019

Particu	lars	Note No.	As at 31 March 2019	As at 31 March 2018
		Note No.	AS at 51 March 2019	As at 51 March 2018
ASSET				
	URRENT ASSETS			
(a)	Deferred Tax Assets (net)	11	-	-
	TOTAL NON-CURRENT ASSETS		•	
CURR	ENT ASSETS			
(a)	Financial assets			
	(i) Cash and Cash Equivalents	3	1.35	34.39
(b)	Other current assets	4	5.08	1.43
	TOTAL CURRENT ASSETS		6.43	35.82
	TOTAL ASSETS		6.43	35.82
EQUIT	Y AND LIABILITIES Y			
(a)	Equity share capital	5	277.00	277.00
(b)	Other equity	6	(307.67)	(281.95)
	TOTAL EQUITY		(30.67)	(4.95)
LIABIL	ITIES			
CU	RRENT LIABILITIES			
(a)	Financial liabilities			
	(i) Trade payables	7	3.26	6.64
	(ii) Other financial liabilities	8	33.84	33.51
(b)	Other current liabilities	9	and the second second	0.62
	TOTAL CURRENT LIABILITIES		37.10	40.77
	TOTAL EQUITY AND LIABILITIES		6.43	35.82
	ant accounting policies	2		
Notes t	to the Ind AS financial statements	3 - 21		

The accompanying notes 1 to 21 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants

Firm Registration No. 117366W/ W- 100018

la K

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019 For and on behalf of the Board of Directors of MIA Infrastructure Private Limited CIN: U74900MH2014PTC252385

S.S.r Inor

Santosh Mhadgut Director DIN - 08049549

Mumbai Dated : 05 April 2019

Prakash Patil Director DIN - 00049866

### MIA Infrastructure Private Limited Statement of Profit and Loss for the year ended 31 March 2019

				(Rs in lakhs)
Parti	culars	Note No.	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018
1	Revenue from Operations :			0.0
Ш	Other Income		a	-
Ш	Total Income (I +II)		Contraction of the second	-
IV	Expenses:			
	Other expenses	10	25.72	47.31
	Total Expenses		25.72	47.31
v	Loss before exceptional items and tax (III-IV)		(25.72)	(47.31)
VI	Exceptional items Income / (Expense)			(L)
VII	Loss before tax (V-VI)		(25.72)	(47.31)
VIII	Tax expenses			
	Current Tax		-	
	Deferred Tax			1.00
	Total tax expenses			
IX	LOSS FOR THE YEAR (VII-VIII)		(25.72)	(47.31)
Х	Other Comprehensive Income:			
	A. (i) Items that will not be reclassified to profit or loss		-	-
	(ii) Income tax on above		- 12 C	-
	B. (i) Items that will not be reclassified to profit or loss			-
	(ii) Income tax on above		-	-
XI	Total Comprehensive Loss for the Year (IX+X)		(25.72)	(47.31)
XII	Earnings per equity share	13		
	Basic		(0.93)	(1.88)
	Diluted		(0.93)	(1.88)
Sign	ificant accounting policies	2		
Note	Notes to the Ind AS financial statements			

The accompanying notes 1 to 21 form an integral part of these Ind AS financial statements. As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

melila 14

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019

For and on behalf of the **Board of Directors of MIA Infrastructure Private Limited** CIN: U74900MH2014PTC252385

SSMLO Locit

Santosh Mhadgut Director DIN - 08049549

Prakash Patil Director DIN - 00049866

Mumbai Dated : 05 April 2019

### **MIA Infrastructure Private Limited** Statement of Cash Flow for the year ended 31 March 2019

Particulars	For the Year Ende 2019	d 31 March	For the Year Ende	(Rs in lakhs) d 31 March
A CASH FLOW FROM OPERATING ACTIVITIES	2013		2010	
Loss Before Tax		(25.72)		(47.31)
Increase / (Decrease) in Trade Payables	(3.38)	(===)	5.62	(47.01)
(Decrease) in other Non Financial Assets	(3.65)		(1.06)	
Increase / (Decrease) in other financial liabilities	0.33		(78.50)	
Increase / (Decrease) in other non-financial liabilities	(0.62)	(7.32)	0.62	(73.32)
CASH USED IN OPERATIONS	(0.02)	(33.04)	0.02	(120.63)
Direct Taxes Paid		(00:01)		(120.00)
Net Cash Flow used in Operating Activities		(33.04)		(120.63)
B CASH FLOW FROM INVESTMENT ACTIVITIES				
Net Cash Flow from Investment Activities		-		•
C CASH FLOW FROM FINANCING ACTIVITIES				
Proceeds from Issue of Shares			147.00	
Net Cash Flow from Financing Activities				147.00
NET INCREASE / (DECREASE) IN CASH AND CASH EQUIVALENTS		(33.04)		26.37
Opening Balance		34.39		8.02
Closing Balance		1.35		34.39
NET INCREASE / (DECREASE) IN CASH AND CASH EQUIVALENTS		(33.04)		26.37
Components of Cash and Cash Equivalents				
Balances with Bank		1.35		34.39
Total Balance		1.35		34.39

#### Note:

The Cash flow statement has been prepared under the indirect method as set out in Indian Accounting Standard - 7 ('Ind AS 7') on Cash Flow Statement prescribed in Companies (Indian Accounting Standard) Rules, 2015, notified under Section 133 of the Companies Act, 2013.

The accompanying notes 1 to 21 form an integral part of these Ind AS financial statements.

As per our report of even date attached

For Deloitte Haskins & Sells LLP **Chartered Accountants** Firm Registration No. 117366W/ W- 100018

Kjuchla

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019 For and on behalf of the Board of Directors of **MIA Infrastructure Private Limited** CIN: U74900MH2014PTC252385

S.S.Mha

**Prakash Patil** Director

Santosh Mhadgut DIN - 08049549

DIN - 00049866

Mumbai Dated : 05 April 2019

Director

### MIA Infrastructure Private Limited

Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

### Statement of Changes in Equity for the year ended 31 March 2019

#### A Equity Share Capital

Destinutes	For the Year er 20'		For the Year ende	d 31 March 2018
Particulars	Number of Shares	Rs. in lakhs	Number of Shares	Rs. in lakhs
Subscribed and Fully Paid up Capital				
Equity shares of INR 10 each				
Opening Balance	27,70,000	277.00	13,00,000	130.00
Changes in equity share capital during the year			14,70,000	147.00
Closing Balance	27,70,000	277.00	27,70,000	277.00

#### **B** Other Equity

Particulars	Retained Earnings	Other Comprehensive Income	Total
Balance as at 01 April 2017	(234.64)	•	(234.64)
Loss for the year	(47.31)		(47.31)
Balance as at 31 March 2018	(281.95)		(281.95)
Loss for the year	(25.72)		(25.72)
Balance as at 31 March 2019	(307.67)		(307.67)

The accompanying notes 1 to 21 form an integral part of these Ind AS financial statements.

As per our report of even date

#### For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

Henchta

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019 For and on behalf of the **Board of Directors of MIA Infrastructure Private Limited** CIN: U74900MH2014PTC252385

madeut 9

Santosh Mhadgut Director DIN - 08049549

Mumbai Dated : 05 April 2019

b

Prakash Patil Director DIN - 00049866

#### 1 Background of the Company

- a) MIA Infrastructure Private Limited ('the Company') was incorporated to carry on the business of establishment, construction, development, conception, identification, operation, maintenance and management of airport and airport infrastructure and connected facilities. In addition, it can carry and conduct the business of infrastructure and develop all kind of facilities for commercial or other use. The company is a private limited company incorporated and domiciled in India and has its registered office at Mumbai, Maharashtra, India.
- b) "In order to achieve operating efficiency, the Board of Directors of the Company, in their meeting held on January 2, 2019, had approved a scheme of amalgamation between MIA Infrastructure Private Limited ("the Transferor Company") with its parent company viz. Tata Realty and Infrastructure Limited ("the Transferee Company") and their respective members and creditors w.e.f. April 1, 2019 ("Appointed date"). Consequently, on February 13, 2019, the companies have filed the said scheme of amalgamation with National Company Law Tribunal, Mumbai.

#### 2 SIGNIFICANT ACCOUNTING POLICIES

#### i) Basis of Preparation

#### a) Statement of compliance:

The financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 (the Act) and other relevant provisions of the Act.

These Ind AS financial statements were authorised for issue by the Company's Board of Directors on 5th April 2019

#### b) Going concern :

As at 31 March 2019, the Company's paid up capital and reserves (excluding the deficit in the profit and loss) were INR 277 lakhs (2018: INR 277 lakhs) and correspondingly, the Company's accumulated losses aggregated INR 307.67 lakhs (2018: INR 281.95 lakhs) which leads to material uncertainty related to Going Concern. However, in pursuance to Note No. 1 (b) above, the financial statements have been prepared on a going concern basis. Accordingly, these financial statements do not include any adjustments relating to the recoverability and classification of recorded labilities that might be necessary if the Company is unable to continue as a going concern.

#### c) Historical cost convention:

The financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

- 1 Financial instruments measured at fair value through profit or loss, if applicable
- 2 Financial instruments measured at fair value through other comprehensive income, if applicable

#### d) Key estimates and assumptions

In preparing these financial statements, management has made judgments, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates.

Estimates and assumptions are required in particular for:

· Measurement of fair values

The Company's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Company has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the group CFO.

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorised into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

· Level 1: quoted prices (unadjusted) in active markets for identical assets or liabilities.

• Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

· Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorised in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement

#### ii) Functional and presentation currency

The financial statements are presented in Indian rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian rupees (INR) and all amounts have been rounded-off to the nearest lakhs, except for share data and as otherwise stated. Due to rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.

# SKI Income tax

Chartered Accountants

directly in equity or in OCI.



#### iv) Cash and cash equivalents

For the purpose of presentation in the statement of cash flows, cash and cash equivalents includes cash on hand, deposits held at call with financial institutions, other short-term, highly liquid investments with original maturities of three months or less that are readily convertible to known amounts of cash and which are subject to an insignificant risk of changes in value, and bank overdrafts. Bank overdrafts are shown within borrowings in current liabilities in the balance sheet.

#### v) Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity. Financial instruments also include derivative contracts such as foreign currency foreign exchange forward contracts, interest rate swaps and currency options; and embedded derivatives in the host contract.

#### 1 Financial assets

#### Classification

The Company classifies financial assets as measured at amortised cost, at fair value through other comprehensive income (FVOCI) or at fair value through profit or loss (FVTPL) on the basis of its business model for managing the financial assets and the contractual cash flow characteristics of the financial asset.

#### Initial recognition and measurement

All financial assets are recognised initially at fair value plus, in the case of financial assets not recorded at fair value through profit or loss, transaction costs that are attributable to the acquisition of the financial asset. Purchases or sales of financial assets that require delivery of assets within a time frame established by regulation or convention in the market place (regular way trades) are recognised on the trade date, i.e., the date that the Company commits to purchase or sell the asset.

#### **Debt instruments**

A 'debt instrument' is measured at the amortised cost if both the following conditions are met:

a) The asset is held within a business model whose objective is to hold assets for collecting contractual cash flows, and

b) Contractual terms of the asset give rise on specified dates to cash flows that are solely payments of principal and interest (SPPI) on the principal amount outstanding.

After initial measurement, such financial assets are subsequently measured at amortised cost using the effective interest rate (EIR) method. Amortised cost is calculated by taking into account any discount or premium and fees or costs that are an integral part of the EIR. The EIR amortisation is included in finance income in the profit or loss.

Debt instruments included within the fair value through profit and loss (FVTPL) category are measured at fair value with all changes recognised in the statement of profit and loss.

#### **De-recognition**

A financial asset (or, where applicable, a part of a financial asset or part of a Company of similar financial assets) is primarily derecognised (i.e. removed from the Company's balance sheet) when:

The rights to receive cash flows from the asset have expired, or

The Company has transferred its rights to receive cash flows from the asset or has assumed an obligation to pay the received cash flows in full without material delay to a third party under a 'pass-through' arrangement; and either (a) the Company has transferred substantially all the risks and rewards of the asset, or (b) the Company has neither transferred nor retained substantially all the risks and rewards of the asset.

When the Company has transferred its rights to receive cash flows from an asset or has entered into a pass-through arrangement, it evaluates if and to what extent it has retained the risks and rewards of ownership. When it has neither transferred nor retained substantially all of the risks and rewards of the asset, nor transferred control of the asset, the Company continues to recognise the transferred asset to the extent of the Company's continuing involvement. In that case, the Company also recognises an associated liability. The transferred asset and the associated liability are measured on a basis that reflects the rights and obligations that the Company has retained.

Continuing involvement that takes the form of a guarantee over the transferred asset is measured at the lower of the original carrying amount of the asset and the maximum amount of consideration that the Company could be required to repay.

#### Impairment of financial assets

In accordance with Ind-AS 109, the Company applies expected credit loss (ECL) model for measurement and recognition of impairment loss on the following financial assets and credit risk exposure:

a) Financial assets that are debt instruments, and are measured at amortised cost e.g., loans, debt securities, deposits, and bank balance

#### 2 Financial liabilities

#### Classification

The Company classifies all financial liabilities as measured at amortised cost, except for financial liabilities at fair value through profit or loss. Such liabilities, including derivatives that are liabilities, shall be subsequently measured at fair value.

#### Initial recognition and measurement

Financial liabilities are classified, at initial recognition, as financial liabilities at fair value through profit or loss, loans and borrowings, payables, or as derivatives designated as hedging instruments in an effective hedge, as appropriate.

All financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

The Company's financial liabilities include trade and other payables, loans and borrowings including bank overdrafts, financial guarantee contracts and derivative financial instruments.



(Currency: Indian rupees in lakhs)

### Financial liabilities at fair value through profit or loss

Financial liabilities at fair value through profit or loss include financial liabilities held for trading and financial liabilities designated upon initial recognition as at fair value through profit or loss. Financial liabilities are classified as held for trading if they are incurred for the purpose of repurchasing in the near term. This category also includes derivative financial instruments entered into by the Company that are not designated as hedging instruments in hedge relationships as defined by Ind-AS 109. Separated embedded derivatives are also classified as held for trading unless they are designated as effective hedging instruments.

Gains or losses on liabilities held for trading are recognised in the profit or loss.

Financial liabilities designated upon initial recognition at fair value through profit or loss are designated at the initial date of recognition, and only if the criteria in Ind-AS 109 are satisfied. For liabilities designated as FVTPL, fair value gains/ losses attributable to changes in own credit risk are recognized in OCI. These gains/losses are not subsequently transferred to profit and loss. However, the Company may transfer the cumulative gain or loss within equity. All other changes in fair value of such liability are recognised in the statement of profit or loss. The Company has not designated any financial liability as at fair value through profit and loss.

#### Trade and other payables

Trade and other payables represent liabilities for goods and services provided to the company prior to the end of the financial year which are unpaid. The amounts are unsecured and are usually paid within 30-60 days of recognition or maybe payable on demand. Trade and other payables are represented as current liabilities unless payment is not due within 12 months after the reporting period. They are recognised initially at their fair value and subsequently recognised at amortised cost using the effective interest method.

#### Loans and borrowings

After initial recognition, interest-bearing loans and borrowings are subsequently measured at amortised cost using the EIR method. Gains and losses are recognised in profit or loss when the liabilities are derecognized.

Amortised cost is calculated by taking into account any discount or premium on acquisition and fees or costs that are an integral part of the EIR. The EIR amortisation is included as finance costs in the statement of profit and loss.

This category generally applies to interest-bearing loans and borrowings.

Preference shares, which are mandatorily redeemable on a specific date, are classified as liabilities. The dividends on these preference shares are recognised in profit or loss as finance costs.

The fair value of the liability portion of optionally convertible bonds is determined using a market interest rate for equivalent non-convertible bonds. This amount is recorded as a liability on an amortised cost basis until extinguished on conversion or redemption of the bonds. The remainder of the proceeds is attributable to the equity portion of the compound instrument. This is recognised and included in shareholders' equity, net of income tax effects, and not subsequently remeasured.

Where the terms of a financial liability are renegotiated and the entity issues equity instruments to a creditor to extinguish all or part of the liability (debt for equity swap), a gain or loss is recognised in profit or loss, which is measured as the difference between the carrying amount of the financial liability and the fair value of the equity instruments issued.

#### Derecognition

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit or loss.

#### Offsetting of financial instruments

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable right to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable right must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the company or the counterparty.

#### vi) Provisions

A provision is recognised in the balance sheet when the Company has a legal or constructive obligation as a result of a past event, and it is probable that an outflow of economic benefits will be required to settle the obligation. If the effect is material, provisions are determined by discounting the expected future cash flows at a pre-tax rate that reflects current market assessments of the time value of money and, where appropriate, the risks specific to the liability.

#### vii) Earnings per share

Basic earnings per share is computed by dividing the profit / (loss) after tax by the weighted average number of equity shares outstanding during the year. The weighted average number of equity shares outstanding during the year is adjusted for the events for bonus issue, bonus element in a rights issue to existing shareholders, share split and reverse share split (consolidation of shares).

Diluted earnings per share is computed by dividing the profit / (loss) after tax as adjusted for dividend, interest and other charges to expense or income (net of any attributable taxes) relating to the dilutive potential equity shares, by the weighted average number of equity shares considered for deriving basic earnings per share and the weighted average number of equity shares which could have been issued on conversion of all dilutive potential equity shares.





#### MIA Infrastructure Private Limited Notes to the Ind AS financial statements (Currency: Indian rupees in lakhs)

#### 3 Cash and Cash Equivalents

Particulars	As at 31 March 2019	As at 31 March 2018
Cash and Cash Equivalents		
Balances with Bank	1.35	34.39
Total	1.35	34.39

#### 4 Other Assets

Particulars	As at 31 March 2019		As at 31 March 2018	
	Non Current	Current	Non Current	Current
Balance with Tax Authority	-	5.08	-	1.43
Others Receivable		-		-
Total		5.08	-	1.43

#### 5 Equity Share Capital

Particulars	As at 31 March 2019		As at 31 March 2018	
	No of Shares	Amount	No of Shares	Amount
Authorised Capital :		and the states of		
Equity Shares of Rs.10/- each	28,00,000	280.00	28,00,000	280.00
ssued, Subscribed and Fully Paid up Capital : Issued Capital				
Equity Shares of Rs.10/- each, fully paid	27,70,000	277.00	27,70,000	277.00
Subscribed and Fully Paid up Capital				
Equity Shares of Rs.10/- each, fully paid	27,70,000	277.00	27,70,000	277.00
Total		277.00		277.00

Particulars	As at 31 Ma	rch 2019	As at 31 Ma	arch 2018
	No of Shares	Amount	No of Shares	Amount
As at the beginning of the year	27,70,000	277.00	13,00,000	130.00
Add: Issued during the year	-	-	14,70,000	147.00
As at the end of the year	27,70,000	277.00	27,70,000	277.00

#### (c) Terms and rights attached to the equity share

The Company has only one class of equity shares having par value of INR 10 per share. Each holder of equity shares is entitled to one vote per share. The Company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject the approval of shareholders in the ensuing Annual General Meeting.

During the year ended 31 March 2019, the amount of per share dividend recognised as distribution to equity shareholders is Rs Nil (2018: Rs Nil). In the event of liquidation of the Company, the holders of the equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the share holders.

#### (d) Shares of the company held by the Holding company

Name of Shareholder	As at 31 Mai	As at 31 March 2019		As at 31 March 2018	
	No of Shares	Amount	No of Shares	Amount	
Equity shares of INR 10 each, fully paid-up by	and the second second second second		- Contraction and the second		
Tata Realty and Infrastructure Limited	27,70,000	277.00	27,70,000	277.00	

Name of Shareholder	As at 31 Ma	rch 2019	As at 31 M	arch 2018
	No of Shares	% Holding	No of Shares	% Holding
Equity shares of INR 10 each, fully paid-up by				
Tata Realty and Infrastructure Limited	27,70,000	100%	27,70,000	100%

(f) Aggregate number of bonus shares issued and shares issued for consideration other than cash during the period of five years immediately During the five year period ended 31 March, 2019, the company has not issued any equity shares for consideration other than cash.

#### 6 Other Equity

1.

Particulars	As at 31 March 2019	As at 31 March 2018	
Retained earnings	(307.67)	(281.95)	
TOTAL	(307.67)	(281.9	
Retained earnings			
Particulars	As at 31 March 2019	As at 31 March 2018	
(Deficit) in statement of profit and loss			
Balance at the beginning of the year	(281.95)	(234.64)	
Add: (Loss) for the year	(25.72)	(47.31)	
Balance at the end of the Year	(307.67)	(281.95)	

#### 7 Current Financial Liabilities - Trade Payables

Particulars		As at 31 March 2019	As at 31 March 2018
Trade Payables		CARGE A LEGERAL MARCH	
SKINS Micro, Sma	II and Medium Enterprises		
POthers		3.26	6.64
1 John	Total	3.26	6.64
Chartered  -			
Accountants 5			



#### MIA Infrastructure Private Limited Notes to the Ind AS financial statements (Currency: Indian rupees in lakhs)

### 8 Other Current Financial Liabilities

Particulars	As at 31 March 2019	As at 31 March 2018
Other Payables		
- Related Party	33.25	33.25
- Others	0.59	0.26
	33	3.84 33.51
Total	33	.84 33.51

#### 9 Other Current Liabilities

Particulars	As at 31 March 2019	As at 31 March 2018
Duty & Taxes Payable	and the second	0.62
Total	The AVERT CONTRACT OF A DESCRIPTION OF A	0.62

#### 10 Other Expenses

Particulars	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018	
Audit Fees (refer note 10(a))	0.82	0.27	
Bank Charges & Guarantee Commission	0.04	0.12	
Fees & Consultations	24.11	44.49	
Rates & Taxes	0.03	1.60	
Sundry Expenses		0.83	
Travelling Expenses	0.72		
Total	25.72	47.31	

#### (a) Remuneration to Statutory Auditors

Particulars	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018	
Statutory audit	0.50	0.23	
Other Services	0.23	-	
Service tax / GST	0.09	0.04	
Total	0.82	0.27	

#### 11 Deferred Tax Assets

(a) Amounts recognised in profit and loss

Particulars	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018
Current income tax		
Origination and reversal of temporary differences	Second and second	
Reduction in tax rate		
Recognition of previously unrecognised tax losses	the CANASA SLAN	•
Change in recognised deductible temporary differences		
Total deferred tax expense/(benefit)		-
Tax expense for the year		
(b) Reconciliation of effective tax rate		
Loss before tax	(25.72)	(47.31)
Tax using the Company's domestic tax rate 25%	(6.43)	(11.83)
Reduction in tax rate		
Tax effect of:		
Current-year losses for which no deferred tax asset is recognised	6.43	11.83
Total	-	

#### (c) Unrecognised deferred tax assets

Deferred tax assets have not been recognised in respect of the following items, because of no virtual certainty of the profit in the future year against which the deferred tax asset created can be utilised.

The Company offsets tax assets and liabilities if and only if it has a legally enforceable right to set off current tax assets and current tax liabilities and the deferred tax assets and deferred tax liabilities relate to income taxes levied by the same tax authority.

Significant management judgement is required in determining provision for income tax, deferred income tax assets and liabilities and recoverability of deferred income tax assets. The recoverability of deferred income tax assets is based on estimates of taxable income and the period over which deferred income tax assets will be recovered. Any changes in future taxable income would impact the recoverability of deferred tax assets.





#### **MIA Infrastructure Private Limited** Notes to the Ind AS financial statements (Currency: Indian rupees in lakhs)

#### 12 Due to micro and small suppliers

Under the Micro, Small and Medium Enterprises Development Act, 2006, (MSMED) which came into force from 2 October 2006, certain disclosures are required to be made relating to Micro, Small and Medium enterprises. On the basis of the information and records available with the Management, there are no outstanding dues to the Micro, Small and Medium enterprises as defined in the Micro, Small and Medium Enterprises Development Act, 2006 as set out in following disclosure.

Particulars	As at 31 March 2019	As at 31 March 2018
Principal amount remaining unpaid to any supplier as at the year-end		
Interest due thereon		
The amount of interest paid by the buyer as per the Micro Small and Medium Enterprises Development Act, 2006 (MSMED Act, 2006)		
The amount of interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under MSMED Act, 2006		
The amount of interest accrued and remaining unpaid at the end of each accounting year		
The amount of further interest remaining due and payable even in the succeeding years, until such date when the interest dues as above are actually paid to the small enterprise for the purpose of disallowance as a deductible expenditure under the MSMED Act, 2006		

#### 13 Earning Per Share

Earnings Per Share (EPS) = Net Profit attributable to Shareholders / Weighted Number of Shares Outstanding

Particulars	For the Y	ear Ended 31 March 2019	For the Year Ende 2018	d 31 March
(Loss) after tax attributable to equity shareholders	A	(25.72)		(47.31)
Calculation of weighted average number of equity shares:				
Number of equity shares at the beginning of the year	с	27,70,000		27,70,000
Optionally convertible participatory debentures				-
Number of equity shares outstanding at the end of the year	Con PS	27,70,000		27,70,000
Weighted average number of equity shares outstanding during the year	В	27,70,000		25,19,615
Earning Per Share – Basic (Rs.)	(A / B)	(0.93)		(1.88)
Earning Per Share – Diluted (Rs.)		(0.93)		(1.88)

#### 14 Disclosure of transactions with Related Parties, as required by Indian Accounting Standard (Ind AS) - 24

(a) Related	Party	Disc	losure
-------------	-------	------	--------

Holding Company	Ultimate Holding Company	<u>IV</u>
Tata Realty and Infrastructure Limited	Tata Sons Limited	
(b) Nature of Transactions / relationship / major parties	Holding Co	Total
Issue of Equity Shares		
Tata Realty and Infrastructure Limited	277.00	277.00
	(277.00)	(277.00)
Outstanding Balances Payable		
Tata Realty and Infrastructure Limited	33.25	33.25
	(33.25)	(33.25)

Note: figures in bracket represent previous years figures.

#### 15 Segment Reporting as per IND AS108 " Operating Segments"

Operating segments are reported in a manner consistent with the internal reporting provided to the chief operating decision maker (CODM). Chief operating decision maker's function is to allocate the resources of the entity and access the performance of the operating segment of the entity. The Board of directors of the Company assesses the financial performance and position of the Company and makes strategic decisions. It is identified as being the chief operating decision maker for the company. The Company is operating in the real estate and infrastructure industry and has only domestic operations. The Company has only one operating segment, which is development of real estate and infrastructure facilities. All assets of the Company are domiciled in India and the Company has no revenue

#### 16 Other matters

Information with regard to other matters specified in Schedule III of the Act is either nil or not applicable to the Company for the year.

#### 17 Capital commitment, contingencies and other commitments

There are no capital commitments and contingent liabilities as at the balance sheet date. Estimated amount of other commitments remaining to be executed on capital account and not provided for (net of advances) Rs Nil (2018: Rs Nil).

#### 18 Financial instruments - Fair values and risk management

#### **Credit risk** A

S

2 4

0

Credit risk is the risk of financial loss to the Company if a customer or counterparty to a financial instrument fails to meet its contractual obligations, and arises principally from the Company's receivables from customers and investment securities. The carrying amounts of financial assets represent the maximum credit xposure

### SKIN Cash and cash equivalents

The Company held cash and cash equivalents with credit worthy banks and financial institutions of INR 1.35 lakhs and INR 34.39 lakhs as at 31 March 2019 and 31 March 2018 respectively. The credit worthiness of such banks and financial institutions is evaluated by the management on an ongoing basis and is Charteresons are done and the good. Accountants,

MUMBA

#### MIA Infrastructure Private Limited

Notes to the Ind AS financial statements

#### (Currency: Indian rupees in lakhs)

#### Liquidity risk в

Liquidity risk is the risk that the Company will not be able to meet its financial obligations as they become due. The Company manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Company's reputation.

The Company has obtained a finance facility from a bank. The Company also constantly monitors funding options available in the debt and capital markets with a view to maintaining financial flexibility.

As at 31 March, 2019, the Company had working capital of Rs(30.67) lakhs including short term borrowings of Rs Nil. As at 31 March, 2018, the Company had working capital of Rs (4.95) lakhs including short-term borrowings of Rs Nil.

#### C Exposure to liquidity risk

The table below analyses the Company's financial liabilities into relevant maturity groupings based on their contractual maturities for:

- all non derivative financial liabilities

Particulars	Carrying amount	Total	1 year or less	1-2 years	2-5 years
As at 31 March 2019					
Non-derivative financial liabilities					
Unsecured short-term borrowings					
Other financial liabilities	33.84	33.84	33.84		
	33.84	33.84	33.84	Self-Melkinss	I CARL CARL CARL
As at 31 March 2018					
Non-derivative financial liabilities					
Unsecured short-term borrowings		-			
Other financial liabilities	33.51	33.51	33.51		
	33.51	33.51	33.51		

#### D Market risk

Market risk is the risk that changes in market prices - such as foreign exchange rates, interest rates and equity prices - will affect the Company's income or the value of its holdings of financial instruments. The Company is domiciled in India and has its revenues and other transactions in its functional currency i.e. INR. Accordingly the Company is not exposed to any currency risk. Also the Company does not hold any equity investments, accordingly the Company is not exposed to any equity price risk.

#### E Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Company's exposure to market risk for changes in interest rates relates to borrowings from Holding Company.

#### Exposure to interest rate risk

The Company does not account for any fixed-rate financial assets or financial liabilities at fair value through profit or loss, and the Company does not have any designate derivatives (interest rate swaps).

Therefore, a change in interest rates at the reporting date would not affect profit or loss.

#### **Capital management**

The Company's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. Management monitors the return on capital as well as the level of dividends to ordinary shareholders.

The Company monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total borrowings, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity. Charge for the year on goodwill amortisation has been deducted while calculating total equity of the company since it represents a pure non-cash expense.

#### The Company's adjusted net debt to equity ratio at 31 March 2019 was as follows.

Particulars	As at 31 March 2019	As at 31 March 2018	
Total borrowings			
Less : Cash and cash equivalent	1.35	34.39	
Adjusted net debt	(1.35)	(34.39)	
Adjusted equity	(30.67)	(4.95)	
Adjusted net debt to adjusted equity ratio (Times)	0.04	6.94	

#### 19 Financial instruments - Fair values and risk management

#### Accounting classification and fair values

The following table shows the carrying amounts and fair values of financial assets and financial liabilities, including their levels in the fair value hierarchy. It does not include fair value information for financial assets and financial liabilities if the carrying amount is a reasonable approximation of fair value.

Particulars	FVTPL	FVTOCI	Amotised Cost	Total	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Total
31 March 2019								
Financial assets								
Cash and cash equivalents			1.35	1.35	-	-	-	-
Financial liabilities								
Interest-free loans from						20		
related parties		-			-	-	-	
Other current financial			33.84	33.84				
liabilities			55.04	55.04				
31 March 2018								
Financial assets								
Cash and cash equivalents			34.39	34.39	-	-		-
Financial liabilities								
Interest-free loans from								
related parties			-	-	-	-	-	-
Other current financial liabilities			33.51	33.51	-		-	

#### B **Financial risk management**

The Company has exposure to the following risks arising from financial instruments:

#### redit risk SKIN Liquidity risk ; and

· Market isk

# m

Chartered Risk management framework
#### MIA Infrastructure Private Limited Notes to the Ind AS financial statements (Currency: Indian rupees in lakhs)

#### D Measurement of fair values

### Valuation techniques and significant unobservable inputs

The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for financial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used.

Туре	Valuation technique	Significant	Inter-relationship between
Amortised cost: 1. Borrowings	Discounted cash flow approach: The valuation model considers the present value of expected payments, discounted using a risk adjusted discount rate.		Not applicable

### Transfers between Levels 1 and 2

There have been no transfers between Level 1 and Level 2 during the reporting periods

Level 3 fair values There are no items in Level 3 fair values

20 There are no offsetting of financial assets and financial liabilities during the year.

#### 21 Subsequent events

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

For and on behalf of the Board of Directors of MIA Infrastructure Private Limited CIN: U74900MH2014PTC252385

madaut S 5

Santosh Mhadgut Director DIN - 08049549 Prakash Patil Director DIN - 00049866

Mumbai Dated : 05 April 2019



### **Chartered Accountants**

Indiabulls Finance Centre Tower 3, 27<sup>th</sup> – 32<sup>nd</sup> Floor Senapati Bapat Marg Elphinstone Road (West) Mumbai – 400 013 Maharashtra, India Tel: +91 22 6185 4000 Fax: +91 22 6185 4001

### **INDEPENDENT AUDITOR'S REPORT**

### To The Members of MIA Infrastructure Private Limited Report on the Audit of the Ind AS Financial Statements

### Opinion

We have audited the accompanying Ind AS financial statements of MIA Infrastructure Private Limited ("the Company"), which comprise the Balance Sheet as at 31 March 2020, and the Statement of Profit and Loss (including Other Comprehensive Income), the Statement of Cash Flows and the Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under Section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended, ("Ind AS") and other accounting principles generally accepted in India, of the state of affairs of the Company as at 31 March 2020, and its loss, total comprehensive loss, its cash flows and the changes in equity for the year ended on that date.

### **Basis for Opinion**

We conducted our audit of the Ind AS financial statements in accordance with the Standards on Auditing specified under Section 143(10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Ind AS Financial Statements section of our report. We are independent of the Company in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the Ind AS financial statements.

### Material uncertainty related to Going Concern

We draw attention to Note 2(b) to the Ind AS financial statements, wherein the events or conditions, along with other matters as set forth in the said note, indicate that a material uncertainty exists that may cast significant doubt on the Company's ability to continue as a going concern. However, the Ind AS financial statements of the Company have been prepared on a going concern basis for the reasons stated in the said Note.

Our opinion is not modified in respect of this matter

### Information Other than the Financial Statements and Auditor's Report Thereon



The Company's Board of Directors is responsible for the other information. The other information comprises the Director's report which we obtained prior to the date of this auditor's report.

- Our opinion on the Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.
- In connection with our audit of the Ind AS financial statements, our responsibility is to read the other information and, in doing so, consider whether the other information is materially inconsistent with the Ind AS financial statements or our knowledge obtained during the course of our audit or otherwise appears to be materially misstated.
- If, based on the work we have performed, we conclude that there is a material misstatement of this other information, we are required to report that fact. We have nothing to report in this regard.

### Management's Responsibility for the Ind AS Financial Statements

The Company's Board of Directors is responsible for the matters stated in Section 134(5) of the Act with respect to the preparation of these Ind AS financial statements that give a true and fair view of the financial position, financial performance including other comprehensive income, cash flows and changes in equity of the Company in accordance with the Ind AS and other accounting principles generally accepted in India.

This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the Ind AS financial statements that give a true and fair view and are free from material misstatement, whether due to fraud or error.

In preparing the Ind AS financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those Board of Directors are also responsible for overseeing the Company's financial reporting process.

### Auditor's Responsibility for the Audit of the Ind AS Financial Statements

Our objectives are to obtain reasonable assurance about whether the Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these Ind AS financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

Identify and assess the risks of material misstatement of the Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.

RKH

- Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under Section143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Company has adequate internal financial controls system in place and the operating effectiveness of such controls.
- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.
- Evaluate the overall presentation, structure and content of the Ind AS financial statements, including the disclosures, and whether the Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Materiality is the magnitude of misstatements in the Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the Ind AS financial statements.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

### **Report on Other Legal and Regulatory Requirements**

- 1. As required by Section 143(3) of the Act, based on our audit we report, to the extent applicable that:
  - a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit.
  - b) In our opinion, proper books of account as required by law have been kept by the Company so far as it appears from our examination of those books
  - c) The Balance Sheet, the Statement of Profit and Loss including Other Comprehensive Income, the Statement of Cash Flows and Statement of Changes in Equity dealt with by this Report are in agreement with the relevant books of account.
  - d) In our opinion, the aforesaid Ind AS financial statements comply with the Ind AS specified under Section 133 of the Act.

 The matter described in the Material uncertainty related to Going Concern section above, in our opinion, may have an adverse effect on the functioning of the Company.

RKU

- f) On the basis of the written representations received from the directors as on 31 March 2020 taken on record by the Board of Directors, none of the directors is disqualified as on 31 March 2020 from being appointed as a director in terms of Section 164(2) of the Act.
- g) Reporting on the adequacy of the internal financial controls over financial reporting of the Company and the operating effectiveness of such controls is not applicable to the company in view of the exemption available to the company in terms of the notification no. G.S.R. 583(E) dated 13 June 2017 issued by the Ministry of Corporate Affairs, Government of India, read with general circular No. 08/2017 dated 25 July 2017.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of Section 197(16) of the Act, as amended.

In our opinion and to the best of our information and according to the explanations given to us, the Company being a private company, Section 197 of the Act related to managerial remuneration is not applicable.

- With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended in our opinion and to the best of our information and according to the explanations given to us:
  - i. The Company does not have any pending litigations which would impact its financial position.
  - ii. The Company did not have any long-term contracts including derivative contracts for which there were any material foreseeable losses.
  - iii. There were no amounts, which were required to be transferred to the Investor Education and Protection Fund by the Company
- 2. As required by the Companies (Auditor's Report) Order, 2020 ("the Order") issued by the Central Government in terms of Section 143(11) of the Act, we give in "Annexure A" a statement on the matters specified in paragraphs 3 and 4 of the Order.

For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No.117366W/W-100018)

pru.

Rajesh K. Hiranandani Partner (Membership No. 36920) UDIN: 20036920AAAABK1936

Mumbai, 17 June 2020

### ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT

(Referred to in paragraph 2 under 'Report on Other Legal and Regulatory Requirements' Section of our report of even date)

- (i) The Company does not have any fixed assets and hence reporting under clause (i) of the CARO 2016 is not applicable.
- (ii) The Company does not have any inventory and hence reporting under clause (ii) of the CARO 2016 is not applicable.
- (iii) The Company has not granted any loans, secured or unsecured, to companies, firms, Limited Liability Partnerships or other parties covered in the register maintained under Section 189 of the Companies Act, 2013
- (iv) The Company has not granted any loans, made investments or provided guarantees and hence reporting under clause (iv) of the CARO 2016 is not applicable.
- (v) According to the information and explanations given to us, the Company has not accepted any deposit during the year as provided under Section 73 to 76 or any other relevant provisions of the Companies Act, 2013. There are no unclaimed deposits any time during the year.
- (vi) The maintenance of cost records has not been specified by the Central Government under Section 148(1) of the Companies Act, 2013.
- (vii) According to the information and explanations given to us, in respect of statutory dues:
  - (a) The Company has been regular in depositing undisputed statutory dues, including Incometax, Goods & Services Tax, cess and other material statutory dues applicable to it to the appropriate authorities.
  - (b) There were no undisputed amounts payable in respect of Income-tax, Goods & Services Tax, cess and other material statutory dues in arrears as at 31 March 2020 for a period of more than six months from the date they became payable.
  - (c) There are no dues of Income-tax, Goods & Services Tax as on 31 March 2020 on account of disputes.
- (viii) In our opinion and according to the information and explanations given to us, The Company has not taken any loans or borrowings from financial institutions, banks and government or has not issued any debentures. Hence reporting under clause (viii) of CARO 2016 is not applicable to the Company.
- (ix) The Company has not raised moneys by way of initial public offer or further public offer (including debt instruments) or term loans and hence reporting under clause (ix) of the CARO 2016 Order is not applicable.
- (x) To the best of our knowledge and according to the information and explanations given to us, no fraud by the Company and no material fraud on the Company by its officers or employees has been noticed or reported during the year.
- (xi) In our opinion and according to the information and explanations given to us, the Company is not required to have and does not have any managerial personnel and hence the reporting under clause of the CARO 2016 Order for payment of managerial remuneration under Section 197 read with Schedule V of the Companies Act 2013 is not applicable.

RKH

- (xii) The Company is not a Nidhi Company and hence reporting under clause (xii) of the CARO 2016 Order is not applicable.
- (xiii) In our opinion and according to the information and explanations given to us the Company is in compliance with Section 177 and 188 of the Companies Act, 2013, where applicable, for all transactions with the related parties and the details of related party transactions have been disclosed in the financial statements etc. as required by the applicable accounting standards.
- (xiv) During the year the Company has not made any preferential allotment or private placement of shares or fully or partly convertible debentures and hence reporting under clause (xiv) of CARO 2016 is not applicable to the Company.
- (xv) In our opinion and according to the information and explanations given to us, during the year the Company has not entered into any non-cash transactions with its directors or directors of its holding, subsidiary or associate company or persons connected with them and hence provisions of Section 192 of the Companies Act, 2013 are not applicable.
- (xvi) The Company is not required to be registered under Section 45-IA of the Reserve Bank of India Act, 1934.

### For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No. 117366W/W-100018)

ptkine.

Rajesh K. Hiranandani Partner (Membership No. 36920) UDIN: 20036920AAAABK1936

Mumbai, 17 June 2020

### MIA Infrastructure Private Limited Balance Sheet as at 31 March 2020

				(Rs. in lakhs)
Particu	lars	Note No.	As at 31 March 2020	As at 31 March 2019
ASSET	S			
NON-C	URRENT ASSETS			
(a)	Deferred tax assets (net)	11		
	TOTAL NON-CURRENT ASSETS		<del>.</del> (	
CURR	ENT ASSETS			
(a)	Financial assets			1000
	(i) Cash and cash equivalents	3	1.33	1.35
(b)		4	5.08	5.08
	TOTAL CURRENT ASSETS		6.41	6.43
	TOTAL ASSETS		6.41	6.43
	Y AND LIABILITIES		19	
EQUIT				077.00
(a)	Equity share capital	5	277.00	277.00
(b)	Other equity	6	(308.28)	(307.67)
	TOTAL EQUITY		(31.28)	(30.67)
LIABI	LITIES			
CU	RRENT LIABILITIES			
(a)				
	(i) Borrowings	7 8 9	1.18	
	(ii) Trade payables	8	3.26	3.26
	(iii) Other financial liabilities	9	33.25	33.84
	TOTAL CURRENT LIABILITIES		37.69	37.10
	TOTAL EQUITY AND LIABILITIES		6.41	6.43
Signifi	cant accounting policies	2	and the second se	
Notes	to the Ind AS financial statements	3 - 21		

.

The accompanying notes 1 to 21 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants

Firm Registration No. 117366W/ W- 100018

prk. in

Rajesh K. Hiranandani Partner

	Mumba	i.
R	Dated :	17

Mumbai Dated : 17 June 2020 For and on behalf of the Board of Directors of MIA Infrastructure Private Limited CIN: U74900MH2014PTC252385

SSM

Santosh Mhadgut Director DIN - 08049549

Mumbai Dated : 17 June 2020 Ritesh Kamdar Director DIN - 00047995

A

### **MIA Infrastructure Private Limited**

### Statement of Profit and Loss for the year ended 31 March 2020

(Rs in lakhs)				
For the year ended 31 March 2019	For the year ended 31 March 2020	Note No.	culars	Partic
			Revenue from Operations :	1
1.5			Other Income	11
			Total Income (I +II)	ш
			Expenses:	IV
25.72	0.61	10	Other expenses	
25.72	0.61		Total Expenses	
(25.72)	(0.61)		Loss before tax (III-IV)	V
			Tax expenses	VI
(25.72)	(0.61)		Loss for the year (V-VI)	VII
-			Other Comprehensive Income:	VIII
(25.72)	(0.61)		Total Comprehensive Loss for the Year (VII+VIII)	IX
		13	Earnings per equity share	X
(0.93)	(0.02)		Basic	
(0.93)	(0.02)		Diluted	
		2	ificant accounting policies	Signi
		3 - 21	s to the Ind AS financial statements	
	(0.61)	2	Other Comprehensive Income: Total Comprehensive Loss for the Year (VII+VIII) Earnings per equity share Basic Diluted ificant accounting policies	VIII IX X

81

The accompanying notes 1 to 21 form an integral part of these Ind AS financial statements. As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

La P 12-t

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020

For and on behalf of the Board of Directors of MIA Infrastructure Private Limited CIN: U74900MH2014PTC252385

1 Sismhad

Santosh Mhadgut Director DIN - 08049549

Ritesh Kamdar Director DIN - 00047995

(De to laber)

### MIA Infrastructure Private Limited Statement of Cash Flow for the year ended 31 March 2020

Particulars	For the year ended March 2020	31	For the year ended 31 March 2019
A CASH FLOW FROM OPERATING ACTIVITIES	and the second second		105 70
Loss Before Tax	(	0.61)	(25.72)
(Decrease) in Trade Payables			(3.38)
(Decrease) in other Non Financial Assets		-	(3.65)
(Decrease)/ Increase in other financial liabilities	(	0.59)	0.33
(Decrease) in other non-financial liabilities		*	(0.62)
Cash (used in) Operations	(	1.20)	(33.04)
Direct Taxes Paid			•
Net Cash (used in) Operating Activities	(	1.20)	(33.04)
B CASH FLOW FROM INVESTMENT ACTIVITIES			
Net Cash Flow from Investment Activities		•	•
C CASH FLOW FROM FINANCING ACTIVITIES			
Proceeds from Short term Borrowings		1.18	*
Net Cash Flow from Financing Activities		1.18	
NET (DECREASE) IN CASH AND CASH EQUIVALENTS		0.02)	(33.04)
Opening Balance		1.35	34.39
Closing Balance		1.33	1.35
NET (DECREASE) IN CASH AND CASH EQUIVALENTS		(0.02)	(33.04)
Components of Cash and Cash Equivalents			
Balances with Bank		1.33	1.35
Total Balance	-	1.33	1.35

Note:

The Cash flow statement has been prepared under the indirect method as set out in Indian Accounting Standard - 7 ('Ind AS 7') on Cash Flow Statement prescribed in the Companies (Indian Accounting Standard) Rules, 2015. notified under Section 133 of the Companies Act, 2013.

The accompanying notes 1 to 21 form an integral part of these Ind AS financial statements.

As per our report of even date attached

### For Deloitte Haskins & Sells LLP

Chartered Accountants Firm Registration No. 117366W/ W- 100018

P

Rajesh K. Hiranandani Partner



Mumbai Dated : 17 June 2020 For and on behalf of the Board of Directors of MIA Infrastructure Private Limited CIN: U74900MH2014PTC252385\_

adell S.CMI

Santosh Mhadgut Director DIN - 08049549

Ritesh Kamdar Director DIN - 00047995

1

Statement of Changes in Equity for the year ended 31 March 2020

### A Equity Share Capital

Particulars	31 March 2020	31 March 2019
Subscribed and Fully Paid up Capital		
Equity shares of INR 10 each		
Opening Balance	277.00	277.00
Closing Balance	277.00	277.00

### **B** Other Equity

Particulars	Retained earnings	Other comprehensive income	Total
Balance as at 01 April 2018	(281.95)		(281.95)
Loss for the year	(25.72)		(25.72)
Balance as at 31 March 2019	(307.67)	•	(307.67)
Loss for the year	(0.61)		(0.61)
Balance as at 31 March 2020	(308.28)		(308.28)

The accompanying notes 1 to 21 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

P K 11

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020 For and on behalf of the **Board of Directors of MIA Infrastructure Private Limited** CIN: U74900MH2014PTC252385

S.S. Madget

Santosh Mhadgut Director DIN - 08049549

Ritesh Kamdar Director DIN - 00047995

MIA Infrastructure Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### 1 Background of the Company

- a) MIA Infrastructure Private Limited ('the Company') was incorporated on 22 January 2014 to carry on the business of establishment, construction, development, conception, identification, operation, maintenance and management of airport and airport infrastructure and connected facilities. In addition, it can carry and conduct the business of infrastructure and develop all kind of facilities for commercial or other use. The Company is a wholly owned subsidiary of Tata Realty and Infrastructure Limited, which is a wholly owned subsidiary of Tata Sons Private Limited, the ultimate holding Company. The Company is a private limited company incorporated and domiciled in India and has its registered office at Mumbai. Maharashtra. India.
- b) In order to achieve operating efficiency, the Board of Directors of the Company, in their meeting held on 2 January 2019, had approved a scheme of amalgamation between MIA Infrastructure Private Limited ("the Transferor Company") with its parent company viz. Tata Realty and Infrastructure Limited ("the Transferee Company") and their respective members and creditors w.e.f. 1 April 2019 ("Appointed date"). Consequently, on 13 February 2019, the companies have filed the said scheme of amalgamation with National Company Law Tribunal. Mumbai.

#### **2 SIGNIFICANT ACCOUNTING POLICIES**

### i) Basis of Preparation

#### a) Statement of compliance:

The financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 (the Act) and other relevant provisions of the Act.

These Ind AS financial statements were authorised for issue by the Company's Board of Directors on 17 June 2020

#### b) Going concern :

As at 31 March 2020, the Company's paid up capital and reserves (excluding the deficit in the profit and loss) were INR 277 lakhs (2019: INR 277 lakhs) correspondingly, the Company's accumulated losses aggregated INR 308.28 lakhs (2019: INR 307.67 lakhs) and the Company's current liabilities exceeded its current assets by INR 31.28 lakhs. These factors give rise to a material uncertainty whether the Company would be able to continue as a going concern. However, in pursuance to Note No. 1 (b) above, the financial statements have been prepared on a going concern basis. Accordingly, these financial statements do not include any adjustments relating to the recoverability and classification of recorded assets, or to amounts and classification of recorded liabilities that might be necessary if the Company is unable to continue as a going

#### c) Historical cost convention:

The financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

- 1 Financial instruments measured at fair value through profit or loss, if applicable
- 2 Financial instruments measured at fair value through other comprehensive income, if applicable

#### d) Key estimates and assumptions

In preparing these financial statements, management has made judgments, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates.

#### Estimates and assumptions are required in particular for:

· Measurement of fair values

The Company's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Company has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the group CFO.

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorised into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

Level 1: quoted prices (unadjusted) in active markets for identical assets or liabilities.

• Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorised in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement

#### ii) Functional and presentation currency

The financial statements are presented in Indian rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian rupees (INR) and all amounts have been rounded-off to the nearest lakhs, except for share data and as otherwise stated. Due to rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.

RKU

#### MIA Infrastructure Private Limited

Notes to the Ind AS financial statements for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

### iii) Income tax

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised directly in equity or in OCI.

#### a) Current tax

Current tax comprises the expected tax payable or receivable on the taxable income or loss for the year and any adjustment to the tax payable or receivable in respect of previous years. It is measured using tax rates enacted or substantively enacted at the reporting date. Current tax also includes any tax arising from dividends. Management periodically evaluates positions taken in tax returns with respect to situations in which applicable tax regulation is subject to interpretation. It establishes provisions where appropriate on the basis of amounts expected to be paid to the tax authorities.

Minimum Alternate Tax ('MAT') under the provisions of Income-tax Act, 1961 is recognised as current tax in the statement of profit and loss. MAT paid in accordance with the tax laws, which gives future economic benefits in the form of adjustment to future income tax liability, is considered as an asset if there is a convincing evidence that the Company will pay normal tax. Accordingly, MAT is recognised as an asset in the balance sheet when it is probable that the future economic benefit associated with it will flow to the Company.

Current tax assets and liabilities are offset only if, the Company:

- 1 has a legally enforceable right to set off the recognised amounts; and
- 2 intends either to settle on a net basis, or to realise the asset and settle the liability simultaneously.

#### b) Deferred tax

Deferred tax is recognised in respect of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for taxation purposes. However, deferred tax liabilities are not recognised if they arise from the initial recognition of goodwill. Deferred income tax is also not accounted for if it arises from initial recognition of an asset or liability in a transaction other than a business combination that at the time of the transaction affects neither accounting profit nor taxable profit (tax loss).

Deferred tax assets are recognised for unused tax losses, unused tax credits and deductible temporary differences to the extent that it is probable that future taxable profits will be available against which they can be used. Deferred tax assets are reviewed at each reporting date and are reduced to the extent that it is no longer probable that the related tax benefit will be realised; such reductions are reversed when the probability of future taxable profits improves. Unrecognised deferred tax assets are reassessed at each reporting date and recognised to the extent that it has become probable that future taxable profits will be available against which they can be used.

Deferred tax is measured at the tax rates that are expected to be applied to temporary differences when they reverse, using tax rates enacted or substantively enacted at the reporting date and are expected to apply when the related deferred income tax asset is realised or the deferred income tax liability is settled.

#### Deferred tax assets and liabilities are offset only if:

1 the entity has a legally enforceable right to set off current tax assets against current tax liabilities; and

2 the deferred tax assets and the deferred tax liabilities relate to income taxes levied by the same taxation authority on the same taxable entity.

REU



### **MIA Infrastructure Private Limited** Notes to the Ind AS financial statements for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

### iv) Cash and cash equivalents

For the purpose of presentation in the statement of cash flows, cash and cash equivalents includes cash on hand, deposits held at call with financial institutions, other short-term, highly liquid investments with original maturities of three months or less that are readily convertible to known amounts of cash and which are subject to an insignificant risk of changes in value, and bank overdrafts. Bank overdrafts are shown within borrowings in current liabilities in the balance sheet.

### v) Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity. Financial instruments also include derivative contracts such as foreign currency foreign exchange forward contracts, interest rate swaps and currency options; and embedded derivatives in the host contract.

### 1 Financial assets

### Classification

The Company classifies financial assets as measured at amortised cost, at fair value through other comprehensive income (FVOCI) or at fair value through profit or loss (FVTPL) on the basis of its business model for managing the financial assets and the contractual cash flow characteristics of the financial asset.

#### Initial recognition and measurement

All financial assets are recognised initially at fair value plus, in the case of financial assets not recorded at fair value through profit or loss, transaction costs that are attributable to the acquisition of the financial asset. Purchases or sales of financial assets that require delivery of assets within a time frame established by regulation or convention in the market place (regular way trades) are recognised on the trade date, i.e., the date that the Company commits to purchase or sell the asset.

#### Debt instruments

A 'debt instrument' is measured at the amortised cost if both the following conditions are met:

a) The asset is held within a business model whose objective is to hold assets for collecting contractual cash flows, and

b) Contractual terms of the asset give rise on specified dates to cash flows that are solely payments of principal and interest (SPPI) on the principal amount outstanding.

After initial measurement, such linancial assets are subsequently measured at amortised cost using the effective interest rate (EIR) method. Amortised cost is calculated by taking into account any discount or premium and fees or costs that are an integral part of the EIR. The EIR amortisation is included in finance income in the profit or loss.

Debt instruments included within the fair value through profit and loss (FVTPL) category are measured at fair value with all changes recognised in the statement of profit and loss.

#### **De-recognition**

A financial asset (or, where applicable, a part of a financial asset or part of a Company of similar financial assets) is primarily derecognised (i.e. removed from the Company's balance sheet) when:

The rights to receive cash flows from the asset have expired, or

The Company has transferred its rights to receive cash flows from the asset or has assumed an obligation to pay the received cash flows in full without material delay to a third party under a 'pass-through' arrangement; and either (a) the Company has transferred substantially all the risks and rewards of the asset, or (b) the Company has neither transferred nor retained substantially all the risks and rewards of the asset, but has transferred control of the asset.

When the Company has transferred its rights to receive cash flows from an asset or has entered into a pass-through arrangement, it evaluates if and to what extent it has retained the risks and rewards of ownership. When it has neither transferred nor retained substantially all of the risks and rewards of the asset, nor transferred control of the asset, the Company continues to recognise the transferred asset to the extent of the Company's continuing involvement. In that case, the Company also recognises an associated liability. The transferred asset and the associated liability are measured on a basis that reflects the rights and obligations that the Company has retained.

Continuing involvement that takes the form of a guarantee over the transferred asset is measured at the lower of the original carrying amount of the asset and the maximum amount of consideration that the Company could be required to repay.

#### Impairment of financial assets

In accordance with Ind-AS 109, the Company applies expected credit loss (ECL) model for measurement and recognition of impairment loss on the following financial assets and credit risk exposure:

a) Financial assets that are debt instruments, and are measured at amortised cost e.g., loans, debt securities, deposits, and bank balance

#### 2 Financial liabilities

#### Classification

The Company classifies all financial liabilities as measured at amortised cost, except for financial liabilities at fair value through profit or loss. Such liabilities, including derivatives that are liabilities, shall be subsequently measured at fair value.

#### Initial recognition and measurement

Financial liabilities are classified, at initial recognition, as financial liabilities at fair value through profit or loss, loans and borrowings, payables, or as derivatives designated as hedging instruments in an effective hedge, as appropriate.

All financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

The Company's financial liabilities include trade and other payables, loans and borrowings including bank overdrafts, financial guarantee contracts and derivative financial instruments.





#### MIA Infrastructure Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### Financial liabilities at fair value through profit or loss

Financial liabilities at fair value through profit or loss include financial liabilities held for trading and financial liabilities designated upon initial recognition as at fair value through profit or loss. Financial liabilities are classified as held for trading if they are incurred for the purpose of repurchasing in the near term. This category also includes derivative financial instruments entered into by the Company that are not designated as hedging instruments in hedge relationships as defined by Ind-AS 109. Separated embedded derivatives are also classified as held for trading unless they are designated as effective hedging instruments.

Gains or losses on liabilities held for trading are recognised in the profit or loss.

Financial liabilities designated upon initial recognition at fair value through profit or loss are designated at the initial date of recognition, and only if the criteria in Ind-AS 109 are satisfied. For liabilities designated as FVTPL, fair value gains/ losses attributable to changes in own credit risk are recognized in OCI. These gains/losses are not subsequently transferred to profit and loss. However, the Company may transfer the cumulative gain or loss within equity. All other changes in fair value of such liability are recognised in the statement of profit or loss. The Company has not designated any financial liability as at fair value through profit and loss.

#### Trade and other payables

Trade and other payables represent liabilities for goods and services provided to the company prior to the end of the financial year which are unpaid. The amounts are unsecured and are usually paid within 30-60 days of recognition or maybe payable on demand. Trade and other payables are represented as current liabilities unless payment is not due within 12 months after the reporting period. They are recognised initially at their fair value and subsequently recognised at amortised cost using the effective interest method.

#### Loans and borrowings

After initial recognition, interest-bearing loans and borrowings are subsequently measured at amortised cost using the EIR method. Gains and losses are recognised in profit or loss when the liabilities are derecognized.

Amortised cost is calculated by taking into account any discount or premium on acquisition and fees or costs that are an integral part of the EIR. The EIR amortisation is included as finance costs in the statement of profit and loss.

This category generally applies to interest-bearing loans and borrowings.

Preference shares, which are mandatorily redeemable on a specific date, are classified as liabilities. The dividends on these preference shares are recognised in profit or loss as finance costs.

The fair value of the liability portion of optionally convertible bonds is determined using a market interest rate for equivalent non-convertible bonds. This amount is recorded as a liability on an amortised cost basis until extinguished on conversion or redemption of the bonds. The remainder of the proceeds is attributable to the equity portion of the compound instrument. This is recognised and included in shareholders' equity, net of income tax effects, and not subsequently remeasured.

Where the terms of a financial liability are renegotiated and the entity issues equity instruments to a creditor to extinguish all or part of the liability (debt for equity swap), a gain or loss is recognised in profit or loss, which is measured as the difference between the carrying amount of the financial liability and the fair value of the equity instruments issued.

#### Derecognition

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit or loss.

#### Offsetting of financial instruments

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable right to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable right must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the company or the counterparty.

#### vi) Provisions

A provision is recognised in the balance sheet when the Company has a legal or constructive obligation as a result of a past event, and it is probable that an outflow of economic benefits will be required to settle the obligation. If the effect is material, provisions are determined by discounting the expected future cash flows at a pre-tax rate that reflects current market assessments of the time value of money and, where appropriate, the risks specific to the liability.

#### vii) Earnings per share

Basic earnings per share is computed by dividing the profit / (loss) after tax by the weighted average number of equity shares outstanding during the year. The weighted average number of equity shares outstanding during the year is adjusted for the events for bonus issue, bonus element in a rights issue to existing shareholders, share split and reverse share split (consolidation of shares).

Diluted earnings per share is computed by dividing the profit / (loss) after tax as adjusted for dividend, interest and other charges to expense or income (net of any attributable taxes) relating to the dilutive potential equity shares, by the weighted average number of equity shares considered for deriving basic earnings per share and the weighted average number of equity shares which could have been issued on conversion of all dilutive potential equity shares.

REU



#### MIA Infrastructure Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### 3 Cash and cash equivalents

Particulars	31 March 2020	31 March 2019
Cash and Cash Equivalents	17.141	
Balances with Bank	1.33	1.35
Total	1.33	1.35

#### 4 Other assets

Particulars	31 Marcl	1 2020	31 Marc	h 2019
	Non Current	Current	Non Current	Current
Balance with Tax Authority		5.08		5.08
Total	-	5.08		5.08

#### 5 Equity share capital

Particulars	31 March	2020	31 Marc	h 2019
Faresers, or	No of Shares	Amount	No of Shares	Amount
Authorised Capital :		10000000000		
Equity Shares of Rs.10/- each	28,00,000	280.00	28,00,000	280.00
ssued, Subscribed and Fully Paid up Capital : Issued Capital				
Equity Shares of Rs.10/- each, fully paid	27.70,000	277.00	27,70,000	277.00
Subscribed and Fully Paid up Capital		10,000	1.	
Equity Shares of Rs. 10/- each, fully paid	27,70,000	277.00	27,70.000	277.00
Total		277.00		277.00

Particulars	31 March	2020	31 Marc	h 2019
	No of Shares	Amount	No of Shares	Amount
As at the beginning of the year	27,70,000	277.00	27.70.000	277.00
As at the end of the year	27,70,000	277.00	27,70,000	277.00

#### (c) Terms and rights attached to the equity share

The Company has only one class of equity shares having par value of INR 10 per share. Each holder of equity shares is entitled to one vote per share. The Company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject the approval of shareholders in the ensuing Annual General Meeting.

During the year ended 31 March 2020, the amount of per share dividend recognised as distribution to equity shareholders is Rs Nil (2019; Rs Nil). In the event of liquidation of the Company, the holders of the equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the share holders.

### (d) Shares of the company held by the holding company

	Name of Shareholder	31 March	2020	31 Marc	h 2019
		No of Shares	Amount	No of Shares	Amount
	Equity shares of INR 10 each, fully paid-up by	10.000 000	10000	Construction of the	a alter
	Tata Realty and Infrastructure Limited	27,70,000	277.00	27,70,000	277.00
(9)	Details of shareholding more than 5% in the Co	mpany			
(e)	Details of shareholding more than 5% in the Co Name of Shareholder	mpany 31 March	2020	31 Marc	h 2019
(e)			2020 % Holding	31 Marc No of Shares	h 2019 % Holding
		31 March			

(f) Aggregate number of bonus shares issued and shares issued for consideration other than cash during the period of five years immediately preceding the reporting date :

During the five year period ended 31 March, 2020, the company has not issued any equity shares for consideration other than cash.

#### 6 Other equity

Particulars	31 March 2020	31 March 2019
Retained earnings	(308.28)	(307.67
TOTAL	(308.28)	(307.67
Retained earnings		
Particulars	31 March 2020	31 March 2019
(Deficit) in statement of profit and loss		
Balance at the beginning of the year	(307.67)	(281.95)
Add: (Loss) for the year	(0.61)	(25.72
Balance at the end of the Year	(308.28)	(307.67

Particulars	31 March 2020	31 March 2019	
Interest-free loans from related parties Tata Realty and Infrastructure Limited			
rata healty and initiastructure climited	1.18	17	
Total	1.18		
The above amount includes			
Secured Borrowings			
Unsecured Borrowings	1.18		



æ REU

MIA Infrastructure Private Limited Notes to the Ind AS financial statements

for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

8 Current Financial Liabilities - Trade Payables

Particulars	31 March 2020	31 March 2019
Trade Payables		
Micro, Small and Medium Enterprises		
Others	3.26	3.26
Total	3.26	3.26

#### 9 Other Current Financial Liabilities

articulars 31 March 2020		31 March 2019		
Other Payables				
- Related Party	33.25	33.25		
- Others		0.59		
	33.25	33.84		
Total	33.25	33.84		

#### 10 Other Expenses

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Audit Fees (refer note 10(a))	0.59	0.82
Bank Charges & Guarantee Commission		0.04
Fees & Consultations		24.11
Rates & Taxes	0.02	0.03
Travelling Expenses		0.72
Total	0.61	25.72

### (a) Remuneration to Statutory Auditors

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019	
Statutory audit	0.50	0.50	
Other Services	and the second se	0.23	
Goods and Services Tax	0.09	0.09	
Total	0.59	0.82	

#### 11 Deferred Tax Assets

#### (a) Amounts recognised in profit and loss

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019	
Current income tax			
Origination and reversal of temporary differences			
Reduction in tax rate		-	
Recognition of previously unrecognised tax losses	-		
Change in recognised deductible temporary differences		<u>*</u> :	
Total deferred tax expense/(benefit)			
Tax expense for the year			
(b) Reconciliation of effective tax rate			
Loss before tax	(0.61)	(25.72	
Tax using the Company's domestic tax rate 25%	(0.15)	(6.43	
Tax effect of:			
Current-year losses for which no deferred tax asset is recognised	0.15	6.43	
Total			

#### (c) Unrecognised deferred tax assets

Deferred tax assets have not been recognised in respect of the following items, because of no virtual certainty of the profit in the future against which the deferred tax asset created can be utilised.

The Company offsets tax assets and liabilities if and only if it has a legally enforceable right to set off current tax assets and current tax liabilities and the deferred tax assets and deferred tax liabilities relate to income taxes levied by the same tax authority.

Significant management judgement is required in determining provision for income tax, deferred income tax assets and liabilities and recoverability of deferred income tax assets. The recoverability of deferred income tax assets is based on estimates of taxable income and the period over which deferred income tax assets will be recovered. Any changes in future taxable income would impact the recoverability of deferred tax assets.

REU



#### 12 Due to micro and small suppliers

Under the Micro, Small and Medium Enterprises Development Act. 2006, (MSMED) which came into force from 2 October 2006, certain disclosures are required to be made relating to Micro, Small and Medium enterprises. On the basis of the information and records available with the Management, there are no outstanding dues to the Micro, Small and Medium enterprises as defined in the Micro, Small and Medium Enterprises Development Act. 2006 as set out in following disclosure.

Particulars	31 March 2020	31 March 2019	
Principal amount remaining unpaid to any supplier as			
at the year-end			
Interest due thereon		1.0	
The amount of interest paid by the buyer as per the Micro Small and Medium Enterprises Development Act, 2006 (MSMED Act, 2006)			
The amount of interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under MSMED Act. 2006	14		
The amount of interest accrued and remaining unpaid at the end of each accounting year			
The amount of further interest remaining due and payable even in the succeeding years, until such date when the interest dues as above are actually paid to the small enterprise for the purpose of disallowance as a deductible expenditure under the MSMED Act, 2006	*		

#### 13 Earning Per Share

Earnings Per Share (EPS) = Net Profit attributable to Shareholders / Weighted Number of Shares Outstanding

Particulars		For the year ended 31 March 2020	For the year ended 31 March 2019	
(Loss) after tax attributable to equity shareholders	A	(0.61)	(25.72)	
Calculation of weighted average number of equity shares:				
Number of equity shares at the beginning of the year	С	27,70,000	27,70.000	
Optionally convertible participatory debentures				
Number of equity shares outstanding at the end of the year		27.70.000	27.70.000	
Weighted average number of equity shares outstanding during the year	в	27,70,000	27.70.000	
Earning Per Share - Basic (Rs.)	(A / B)	- (0.02)	(0.93)	
Earning Per Share - Diluted (Rs.)		(0.02)	(0.93)	

### 14 Disclosure of transactions with Related Parties, as required by Indian Accounting Standard (Ind AS) - 24

Holding Company Tata Realty and Infrastructure Limited	Ultimate Holding Compar Tata Sons Private Limited	אַר
(b) Nature of Transactions / relationship / major parties	Holding Co	Total
Borrowings	and the second	
Tata Realty and Infrastructure Limited	1.18	1.18
Outstanding Balances Payable		
Tata Realty and Infrastructure Limited	34.43	34.43
	(33.25)	(33.25

### 15 Segment Reporting as per IND AS108 " Operating Segments"

Operating segments are reported in a manner consistent with the internal reporting provided to the chief operating decision maker (CODM). Chief operating decision maker's function is to allocate the resources of the entity and access the performance of the operating segment of the entity. The Board of directors of the Company assesses the financial performance and position of the Company and makes strategic decisions. It is identified as being the chief operating decision maker for the company. The Company is operating in the real estate and infrastructure industry and has only domestic operations. The Company has only one operating segment, which is development of real estate and infrastructure facilities. All assets of the Company are domiciled in India and the Company has no revenue.

#### 16 Other matters

Information with regard to other matters specified in Schedule III of the Act is either nil or not applicable to the Company for the year.

#### 17 Capital commitment, contingencies and other commitments

There are no capital commitments and contingent liabilities as at the balance sheet date. Estimated amount of other commitments remaining to be executed on capital account and not provided for (net of advances) Rs Nil (2019: Rs Nil).

#### 18 Financial instruments - Fair values and risk management

A Credit risk

Credit risk is the risk of financial loss to the Company if a customer or counterparty to a financial instrument fails to meet its contractual obligations, and arises principally from the Company's receivables from customers and investment securities. The carrying amounts of financial assets represent the maximum credit exposure.

#### Cash and cash equivalents

The Company held cash and cash equivalents with credit worthy banks of INR 1.33 lakhs and INR 1.35 lakhs as at 31 March 2020 and 31 March 2019 respectively. The credit worthiness of such banks is evaluated by the management on an ongoing basis and is considered to be good.

URE PRI

REU

#### (Currency: Indian rupees in lakhs) B Liquidity risk

Liquidity risk is the risk that the Company will not be able to meet its financial obligations as they become due. The Company manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Company's reputation.

The Company has obtained a finance facility from a bank. The Company also constantly monitors funding options available in the debt and capital markets with a view to maintaining financial flexibility.

As at 31 March, 2020, the Company had working capital of Rs (31.28) lakhs including short term borrowings of Rs 1.18 lakhs. As at 31 March, 2019, the Company had working capital of Rs (30.67) lakhs including short-term borrowings of Rs Nil.

#### C Exposure to liquidity risk

The table below analyses the Company's financial liabilities into relevant maturity groupings based on their contractual maturities for:

- all non derivative financial liabilitie

an non denvative interioral additions					
Particulars	Carrying amount	Total	1 year or less	1-2 years	2-5 years
As at 31 March 2020					
Non-derivative financial liabilities					
Unsecured short-term borrowings	1.18	1.18	1.18	-	
Other financial liabilities	33.25	33.25	33.25	14	
	34.43	34.43	34.43	-	-
As at 31 March 2019					
Non-derivative financial liabilities					
Unsecured short-term borrowings		-		14	-
Other financial liabilities	33.84	33.84	33.84	11	1
	33.84	33.84	33.84		

#### D Market risk

Market risk is the risk that changes in market prices – such as foreign exchange rates, interest rates and equity prices – will affect the Company's income or the value of its holdings of financial instruments. The Company is domiciled in India and has its revenues and other transactions in its functional currency i.e. INR. Accordingly the Company is not exposed to any currency risk. Also the Company does not hold any equity investments, accordingly the Company is not exposed to any equity price risk.

#### E Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Company's exposure to market risk for changes in interest rates relates to borrowings from Holding Company.

#### Exposure to interest rate risk

The Company does not account for any fixed-rate financial assets or financial liabilities at fair value through profit or loss, and the Company does not have any designate derivatives (interest rate swaps).

Therefore, a change in interest rates at the reporting date would not affect profit or loss.

#### F Capital management

The Company's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. Management monitors the return on capital as well as the level of dividends to ordinary shareholders.

The Company monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total borrowings, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity. Charge for the year on goodwill amortisation has been deducted while calculating total equity of the company since it represents a pure non-cash expense.

The Company's adjusted net debt to equity ratio at 31 March 2020 was as follows.

Particulars	31 March 2020	31 March 2019	
Total borrowings	1.18		
Less : Cash and cash equivalent	1.33	1.35	
Adjusted net debt	(0.15)	(1.35)	
Adjusted equity	(31.28)	(30.67)	
Adjusted net debt to adjusted equity ratio (Times)	0.00	0.04	

#### 19 Financial instruments - Fair values and risk management

#### A Accounting classification and fair values

The following table shows the carrying amounts and fair values of financial assets and financial liabilities, including their levels in the fair value hierarchy. It does not include fair value information for financial assets and financial liabilities if the carrying amount is a reasonable approximation of fair value.

Particulars	FVTPL	FVTOCI	Amotised Cost	Total	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Total
31 March 2020							1.41.1	
Financial assets								
Cash and cash equivalents		10.00	1.33	1.33				
Financial liabilities								
Interest-free loans from	22		1.18	1.18	0		12	102
related parties		1000	1.10	1.10				-
Other current financial	54	-		33.25				
liabilities			33.25	33.25				
31 March 2019								
Financial assets								
Cash and cash equivalents			1.35	1.35				
Financial liabilities			10.0000					
Interest-free loans from								
related parties							*/·	77.
Other current financial			C SPECIAL R	No. Contractor				
liabilities	1	200	33.84	33.84	0	200	*	

#### B Financial risk management

The Company has exposure to the following risks arising from financial instruments:

· Credit risk ;

· Liquidity risk ; and

· Market risk



The Company's board of directors has overall responsibility for the establishment and oversight of the Company's risk management framework.

REU

#### MIA Infrastructure Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

#### Measurement of fair values . D

Valuation techniques and significant unobservable inputs The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for financial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used.

Туре	Valuation technique	Significant	Inter-relationship between
Amortised cost: 1. Borrowings	Discounted cash flow approach: The valuation model considers the present value of expected payments, discounted using a risk adjusted discount rate.		Not applicable

Transfers between Levels 1 and 2

There have been no transfers between Level 1 and Level 2 during the reporting periods

Level 3 fair values

There are no items in Level 3 fair values

There are no offsetting of financial assets and financial liabilities during the year. 20

#### 21. Subsequent events

R

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

REU

For and on behalf of the Board of Directors of MIA Infrastructure Private Limited CIN: U74900MH2014PTC252385

Santosh Mhadgut Director DIN - 08049549

**Ritesh Kamdar** Director DIN - 00047995

0

Chartered Accountants Lotus Corporate Park 1<sup>st</sup> Floor, Wing A – G CTS No.185/A, Jay Coach Off Western Express Highway Goregaon (East) Mumbai – 400 063 Maharashtra, India

### INDEPENDENT AUDITOR'S REPORT

Tel: +91 22 6245 1000 Fax: +91 22 6245 1001

To The Members of Wellkept Facility Management Services Private Limited Report on the Audit of the Ind AS Financial Statements

### Opinion

We have audited the accompanying Ind AS financial statements of Wellkept Facility Management Services Private Limited ("the Company"), which comprise the Balance Sheet as at March 31, 2019, and the Statement of Profit and Loss (including Other Comprehensive Income), the Statement of Cash Flows and the Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended, ("Ind AS") and other accounting principles generally accepted in India, of the state of affairs of the Company as at March 31, 2019, and its loss, its cash flows and the changes in equity for the year ended on that date.

### **Basis for Opinion**

e

We conducted our audit of the Ind AS financial statements in accordance with the Standards on Auditing specified under section 143(10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Ind AS Financial Statements section of our report. We are independent of the Company in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the Ind AS financial statements.

### Material uncertainty related to Going Concern

We draw attention to Note 2(b) to the Ind As financial statements, which indicates that the Company has incurred a net loss of 1.06 lakhs during the year ended 31<sup>st</sup> March, 2019, the Company's current liabilities exceeded its current assets by 1.44 lakhs. The. These events or conditions, along with other matters as set forth in Note 2(b), indicate that a material uncertainty exists that may cast significant doubt on the Company's ability to continue as a going concern. However, the Ind As financial statements of the Company have been prepared on a going concern basis for the reasons stated in the said Note.

Our opinion is not modified in respect of this matter

### Information Other than the Financial Statements and Auditor's Report Thereon

The Company's Board of Directors is responsible for the other information. The other information comprises the Director's report which we obtained prior to the date of this auditor's report

• Our opinion on the Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.

Chartered Street information and, in doing so, consider whether the other information is materially

Regd. Office: Indiabulls Finance Centre, Tower 3, 27th - 32th Floor, Senapati Bapat Marg, Elphinstone Road (West), Mumbai - 400 013, Maharashtra, India.

SKINS

Chartered

Accountants

HO

inconsistent with the Ind AS financial statements or our knowledge obtained during the course of our audit or otherwise appears to be materially misstated.

• If, based on the work we have performed, we conclude that there is a material misstatement of this other information, we are required to report that fact. We have nothing to report in this regard.

### Management's Responsibility for the Ind AS Financial Statements

The Company's Board of Directors is responsible for the matters stated in section 134(5) of the Act with respect to the preparation of these Ind AS financial statements that give a true and fair view of the financial position, financial performance including other comprehensive income, cash flows and changes in equity of the Company in accordance with the Ind AS and other accounting principles generally accepted in India.

This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the Ind AS financial statement that give a true and fair view and are free from material misstatement, whether due to fraud or error.

In preparing the Ind AS financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those Board of Directors are also responsible for overseeing the Company's financial reporting process.

### Auditor's Responsibility for the Audit of the Ind AS Financial Statements

Our objectives are to obtain reasonable assurance about whether the Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these Ind AS financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

 Identify and assess the risks of material misstatement of the Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.

Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under section 143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Company has adequate internal financial controls system in place and the operating effectiveness of such controls.



- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to

events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.

• Evaluate the overall presentation, structure and content of the Ind AS financial statements, including the disclosures, and whether the Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Materiality is the magnitude of misstatements in the Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the Ind AS financial statements.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

### Report on Other Legal and Regulatory Requirements

- As required by Section 143(3) of the Act, based on our audit we report, to the extent applicable that:
  - a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit.
  - b) In our opinion, proper books of account as required by law have been kept by the Company so far as it appears from our examination of those books
  - c) The Balance Sheet, the Statement of Profit and Loss including Other Comprehensive Income, the Statement of Cash Flows and Statement of Changes in Equity dealt with by this Report are in agreement with the relevant books of account.
  - d) In our opinion, the aforesaid Ind AS financial statements comply with the Ind AS specified under Section 133 of the Act.
  - e) The matter described in the Material uncertainty related to Going Concern section above, in our opinion, may have an adverse effect on the functioning of the Company.
  - f) On the basis of the written representations received from the directors as on March 31, 2019 taken on record by the Board of Directors, none of the directors is disqualified as on March 31, 2019 from being appointed as a director in terms of Section 164(2) of the Act.



- g) Reporting on the adequacy of the internal financial controls over financial reporting of the Company and the operating effectiveness of such controls is not applicable to the company in view of the exemption available to the company in terms of the notification no. G.S.R. 583(E) dated 13 June 2017 issued by the Ministry of Corporate Affairs, Government of India, read with general circular No. 08/2017 dated 25 July 2017.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of section 197(16) of the Act, as amended.

In our opinion and to the best of our information and according to the explanations given to us, the Company being a private company, section 197 of the Act related to managerial remuneration is not applicable.

- With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended in our opinion and to the best of our information and according to the explanations given to us:
  - i. The Company does not have any pending litigations which would impact its financial position.
  - ii. The Company did not have any long-term contracts including derivative contracts for which there were any material foreseeable losses.
  - iii. There were no amounts, which were required to be transferred to the Investor Education and Protection Fund by the Company
- 2. As required by the Companies (Auditor's Report) Order, 2016 ("the Order") issued by the Central Government in terms of Section 143(11) of the Act, we give in "Annexure A" a statement on the matters specified in paragraphs 3 and 4 of the Order.

### For Deloitte Haskins & Sells LLP

Chartered Accountants (Firm's Registration No.117366W/W-100018)

Auchta

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai, April 5, 2019

Chartered Accountants

### ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT

(Referred to in paragraph 2 under 'Report on Other Legal and Regulatory Requirements' Section of our report of even date)

(i) The Company does not have any property, plant and equipment and hence reporting under clause of the CARO 2016 is not applicable.

(ii) The Company does not have any inventory and hence reporting under clause (ii) of the CARO 2016 is not applicable.

(iii) The Company has not granted any loans, secured or unsecured, to companies, firms, Limited Liability Partnerships or other parties covered in the register maintained under Section 189 of the Companies Act, 2013

(iv) The Company has not granted any loans, made investments or provided guarantees and hence reporting under clause (iv) of the CARO 2016 is not applicable.

(v) According to the information and explanations given to us, the Company has not accepted any deposit during the year as provided under Section 73 to 76 or any other relevant provisions of the Companies Act, 2013. There are no unclaimed deposits any time during the year.

(vi) The maintenance of cost records has not been specified by the Central Government under Section 148(1) of the Companies Act, 2013.

(vii) According to the information and explanations given to us, in respect of statutory dues:

- (a) The Company has been regular in depositing undisputed statutory dues, including Incometax, Service Tax / Goods & Services Tax, cess and other material statutory dues applicable to it to the appropriate authorities.
- (b) There were no undisputed amounts payable in respect of Income-tax, Service Tax / Goods & Services Tax, cess and other material statutory dues in arrears as at March 31, 2019 for a period of more than six months from the date they became payable.
- (c) There are no dues of Income-tax, Service Tax / Goods & Services Tax as on March 31, 2019 on account of disputes.

(viii) In our opinion and according to the information and explanations given to us, the Company has not defaulted in the repayment of loans or borrowings to debenture holders (Bank). The Company has not taken any loans from banks, financial institutions and Government.

(ix) The Company has not raised moneys by way of initial public offer or further public offer (including debt instruments) or term loans and hence reporting under clause (ix) of the CARO 2016 Order is not applicable.

(x) To the best of our knowledge and according to the information and explanations given to us, no fraud by the Company and no material fraud on the Company by its officers or employees has been noticed or reported during the year.

SKIN (xi) In our opinion and according to the information and explanations given to us, the Company is not required to have and does not have any managerial personnel and hence the reporting under

### Wellkept Facility Management Services Private Limited

### (formerly known as TRIL Hospitality Private Limited)

Balance Sheet as at 31 March 2019

	(Rs. in lakhs)		
Particulars	Note No.	As at 31 March 2019	As at 31 March 2018
ASSETS			
NON-CURRENT ASSETS			
(a) Deferred tax assets (net)	10		-
TOTAL NON-CURRENT ASSETS			-
CURRENT ASSETS			
(a) Financial assets			
(i) Cash and Cash Equivalents	3	5.65	0.28
(b) Other current assets	4		23.67
TOTAL CURRENT ASSETS		5.65	23.95
TOTAL ASSETS		5.65	23.95
EQUITY AND LIABILITIES			
EQUITY			
(a) Equity share capital	5	40.00	40.00
(b) Other equity	6	(41.44)	(40.38)
TOTAL EQUITY		(1.44)	(0.38)
LIABILITIES			
CURRENT LIABILITIES			
(a) Financial liabilities			
(i) Borrowings	7	6.50	6.50
(ii) Other financial liabilities	8	0.59	17.83
TOTAL CURRENT LIABILITIES		7.09	24.33
TOTAL EQUITY AND LIABILITIES		5.65	23.95
Significant accounting policies	2		

Notes to the Ind AS financial statements

The accompanying notes 1 to 20 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP

Chartered Accountants Firm Registration No. 117366W/ W- 100018

Huchla

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019 For and on behalf of the **Board of Directors of Wellkept Facility Management Services Private Limited** (formerly known as TRIL Hospitality Private Limited) CIN: U93000MH2008PTC177346

3 - 20

Nandlal Singh Director DIN: 07384037

Mumbai Dated : 05 April 2019

Amit Sheth Director DIN: 07997663

### Wellkept Facility Management Services Private Limited

(formerly known as TRIL Hospitality Private Limited)

### Statement of Profit and Loss for the year ended 31 March 2019

		_		(Rs in lakhs)
Parti	culars	Note No.	For the year ended March 31, 2019	For the year ended March 31, 2018
1	Revenue from Operations :			-
Ш	Other Income		-	-
Ш	Total Income (I +II)			
IV	Expenses:			
	Other expenses	9	1.06	0.55
	Total Expenses		1.06	0.55
V	Loss before exceptional items and tax (III-IV)		(1.06)	(0.55)
VI	Exceptional items Income / (Expense)		-	-
VII	Loss before tax (V-VI)		(1.06)	(0.55)
VIII	Tax expenses			
	Current Tax			( <b>.</b> .)
	Deferred Tax			-
	Total tax expenses			
IX	LOSS FOR THE YEAR (VII-VIII)		(1.06)	(0.55)
X	Other Comprehensive Income:			
	A. (i) Items that will not be reclassified to profit or loss		-	8 <b></b> 6
	(ii) Income tax onabove		A COLORIS DI	1.51
	B. (i) Items that will not be reclassified to profit or loss			-
	(ii) Income tax onabove			-
				-
XI	Total Comprehensive Loss for the Year (IX+X)		(1.06)	(0.55)
XII	Earnings per equity share	12		
	Basic		(0.27)	(0.14)
	Diluted		(0.27)	(0.14)
Sign	ficant accounting policies	2		
	s to the Ind AS financial statements	3 - 20		

The accompanying notes 1 to 20 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

Guchla

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019

For and on behalf of the **Board of Directors of** Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) CIN: U93000MH2008PTC177346

Nandlal Singh Director DIN: 07384037

Mumbai Dated : 05 April 2019

Amit Sheth Director DIN: 07997663

### Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) Statement of Cash Flow for the year ended 31 March 2019

					Rs in lakhs)
	Particulars	culars For the year ended March 31, 2019		For the year ended Marc 31, 2018	
A	CASH FLOW FROM OPERATING ACTIVITIES				
	Loss Before Tax	(1.	06)		(0.55)
	(Decrease) in other financial liabilities	(17.24)		(0.27)	
	Increase / (Decrease) in other non-financial liabilities	- 6.	44	(0.40)	(0.68)
•	CASH GENERATED FROM OPERATIONS Direct Taxes Paid	5.	37		(1.23)
3	Net Cash Flow Generated from Operating Activities	5.	37		(1.23)
в	CASH FLOW FROM INVESTMENT ACTIVITIES				
	Net Cash Flow Used in Investment Activities				-
C	CASH FLOW FROM FINANCING ACTIVITIES				
3	Net Cash Flow Used in Financing Activities	State of the second			-
	NET INCREASE / (DECREASE) IN CASH AND CASH EQUIVALENTS	5.	37		(1.23)
	Opening Balance	0.	28		1.51
0	Closing Balance	5.	65		0.28
	NET INCREASE / (DECREASE) IN CASH AND CASH EQUIVALENTS	5.	37		(1.23)
	Components of Cash and Cash Equivalents				
	Cash on Hand				
	Balances with Bank	5	65		0.28
	Total Balance		65		0.28

### Note:

The Cash flow statement has been prepared under the indirect method as set out in Indian Accounting Standard - 7 ('Ind AS 7') on Cash Flow Statement prescribed in Companies (Indian Accounting Standard) Rules, 2015, notified under Section 133 of the Companies Act, 2013.

The accompanying notes 1 to 20 form an integral part of these Ind AS financial statements.

As per our report of even date attached For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

Huehla

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019 For and on behalf of the **Board of Directors of Wellkept Facility Management Services Private Limited** (formerly known as TRIL Hospitality Private Limited) CIN: U93000MH2008PTC177346

Mumbai

Nandlal Singh Director DIN: 07384037

Dated : 05 April 2019



Director DIN: 07997663

### Statement of Changes in Equity for the year Ended 31 March 2019

### A Equity Share Capital

Particulars	For the year March 31,	For the year ended March 31, 2018		
	Number of Shares	Rs. in lakhs	Number of Shares	Rs. in lakhs
Subscribed and Fully Paid up Capital	THE TREAS			
Equity shares of INR 10 each				
Opening Balance	4,00,000	40.00	4.00.000	40.00
Changes in equity share capital during the year		-	-	-
Closing Balance	4,00,000	40.00	4,00,000	40.00

### **B** Other Equity

Particulars	Retained Earnings	Other Compreh ensive Income	Total
Balance as at 01 April 2017	(39.83)	•	(39.83)
Loss for the year	(0.55)		(0.55)
Balance as at 31 March 2018	(40.38)		(40.38)
Loss for the year	(1.06)	•	(1.06)
Balance as at 31 March 2019	(41.44)		(41.44)

The accompanying notes 1 to 20 form an integral part of these Ind AS financial statements.

As per our report of even date For Deloitte Haskins & Sells LLP Chartered Accountants

Firm Registration No. 117366W/ W- 100018

Sjuchta

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019

For and on behalf of the **Board of Directors of** Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) CIN: U93000MH2008PTC177346

Nandlal Singh Director DIN: 07384037

Amit Sheth

Director DIN: 07997663

Mumbai Dated : 05 April 2019 (formerly known as TRIL Hospitality Private Limited)

### Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

### 1 Background of the Company

- a) Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) ('the Company') was incorporated to carry on the business of development of real estate and infrastructure facilities. The Company is a wholly owned subsidiary of Tata Realty and Infrastructure Limited, which is a wholly owned subsidiary of Tata Sons Limited, the ultimate holding Company. The company is a private limited company incorporated and domiciled in India and has its registered office at Mumbai, Maharashtra, India.
- b) In order to achieve operating efficiency, the Board of Directors of the Company, in their meeting held on January 2, 2019, had approved a scheme of amalgamation between Wellkept Facility Management Services Private Limited ("the Transferor Company") with its parent company viz. Tata Realty and Infrastructure Limited ("the Transferee Company") and their respective members and creditors w.e.f. April 1, 2019 ("Appointed date"). Consequently, on February 13, 2019, the companies have filed the said scheme of amalgamation with National Company Law Tribunal, Mumbai.

### 2 SIGNIFICANT ACCOUNTING POLICIES

### i) Basis of Preparation

### a) Statement of compliance:

These Ind AS financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 (the Act) and other relevant provisions of the Act.

These Ind AS financial statements were authorised for issue by the Company's Board of Directors on 5th April 2019

### b) Going concern :

As at 31 March 2019, the Company's paid up capital and reserves (excluding the deficit in the profit and loss) were INR 40 lakhs (2018: INR 40 lakhs) and correspondingly, the Company's accumulated losses aggregated INR 41.44 lakhs (2018: INR 40.38 lakhs) which leads to material uncertainty related to Going Concern. However, in pursuance to Note No. 1 (b) above, the financial statements have been prepared on a going concern basis. Accordingly, these financial statements do not include any adjustments relating to the recoverability and classification of recorded liabilities that might be necessary if the Company is unable to continue as a going concern.

### c) Historical cost convention:

The financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

- 1 Financial instruments measured at fair value through profit or loss, if applicable
- 2 Financial instruments measured at fair value through other comprehensive income, if applicable

### d) Key estimates and assumptions

In preparing these financial statements, management has made judgments, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates.

Estimates and assumptions are required in particular for:

· Measurement of fair values

The Company's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Company has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the group CFO.

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorised into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

· Level 1: quoted prices (unadjusted) in active markets for identical assets or liabilities.

• Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

• Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorised in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement

### ii) Functional and presentation currency

The financial statements are presented in Indian rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian rupees (INR) and all amounts have been rounded-off to the nearest lakhs, except for share data and as otherwise stated in the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.



Borrowing cost Borrowing costs are expensed in the period in which they are incurred.

### (formerly known as TRIL Hospitality Private Limited)

Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

### iv) Income tax

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised directly in equity or in OCI.

### v) Cash and cash equivalents

For the purpose of presentation in the statement of cash flows, cash and cash equivalents includes cash on hand, deposits held at call with financial institutions, other short-term, highly liquid investments with original maturities of three months or less that are readily convertible to known amounts of cash and which are subject to an insignificant risk of changes in value, and bank overdrafts. Bank overdrafts are shown within borrowings in current liabilities in the balance sheet.

### vi) Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity. Financial instruments also include derivative contracts such as foreign currency foreign exchange forward contracts, interest rate swaps and currency options; and embedded derivatives in the host contract.

### 1 Financial assets

### Classification

The Company classifies financial assets as measured at amortised cost, at fair value through other comprehensive income (FVOCI) or at fair value through profit or loss (FVTPL) on the basis of its business model for managing the financial assets and the contractual cash flow characteristics of the financial asset.

### Initial recognition and measurement

All financial assets are recognised initially at fair value plus, in the case of financial assets not recorded at fair value through profit or loss, transaction costs that are attributable to the acquisition of the financial asset. Purchases or sales of financial assets that require delivery of assets within a time frame established by regulation or convention in the market place (regular way trades) are recognised on the trade date, i.e., the date that the Company commits to purchase or sell the asset.

### **Debt instruments**

A 'debt instrument' is measured at the amortised cost if both the following conditions are met:

a) The asset is held within a business model whose objective is to hold assets for collecting contractual cash flows, and

b) Contractual terms of the asset give rise on specified dates to cash flows that are solely payments of principal and interest (SPPI) on the principal amount outstanding.

After initial measurement, such financial assets are subsequently measured at amortised cost using the effective interest rate (EIR) method. Amortised cost is calculated by taking into account any discount or premium and fees or costs that are an integral part of the EIR. The EIR amortisation is included in finance income in the profit or loss.

Debt instruments included within the fair value through profit and loss (FVTPL) category are measured at fair value with all changes recognised in the statement of profit and loss.

### **De-recognition**

A financial asset (or, where applicable, a part of a financial asset or part of a Company of similar financial assets) is primarily derecognised (i.e. removed from the Company's balance sheet) when:

The rights to receive cash flows from the asset have expired, or

The Company has transferred its rights to receive cash flows from the asset or has assumed an obligation to pay the received cash flows in full without material delay to a third party under a 'pass-through' arrangement; and either (a) the Company has transferred substantially all the risks and rewards of the asset, or (b) the Company has neither transferred nor retained substantially all the risks and rewards of the asset.

When the Company has transferred its rights to receive cash flows from an asset or has entered into a pass-through arrangement, it evaluates if and to what extent it has retained the risks and rewards of ownership. When it has neither transferred nor retained substantially all of the risks and rewards of the asset, nor transferred control of the asset, the Company continues to recognise the transferred asset to the extent of the Company's continuing involvement. In that case, the Company also recognises an associated liability. The transferred asset and the associated liability are measured on a basis that reflects the rights and obligations that the Company has retained.

Continuing involvement that takes the form of a guarantee over the transferred asset is measured at the lower of the original carrying amount of the asset and the maximum amount of consideration that the Company could be required to repay.

### Impairment of financial assets

In accordance with Ind-AS 109, the Company applies expected credit loss (ECL) model for measurement and recognition of impairment loss on the following financial assets and credit risk exposure:

a) Financial assets that are debt instruments, and are measured at amortised cost e.g., loans, debt securities, deposits, and bank balance





### Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

### **2** Financial liabilities

### Classification

The Company classifies all financial liabilities as measured at amortised cost, except for financial liabilities at fair value through profit or loss. Such liabilities, including derivatives that are liabilities, shall be subsequently measured at fair value.

### Initial recognition and measurement

Financial liabilities are classified, at initial recognition, as financial liabilities at fair value through profit or loss, loans and borrowings, payables, or as derivatives designated as hedging instruments in an effective hedge, as appropriate.

All financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

The Company's financial liabilities include trade and other payables, loans and borrowings including bank overdrafts, financial guarantee contracts and derivative financial instruments.

### Trade and other payables

Trade and other payables represent liabilities for goods and services provided to the Company prior to the end of the financial year which are unpaid. The amounts are unsecured and are usually paid within 30-60 days of recognition or maybe payable on demand. Trade and other payables are represented as current liabilities unless payment is not due within 12 months after the reporting period. They are recognised initially at their fair value and subsequently recognised at amortised cost using the effective interest method.

### Loans and borrowings

After initial recognition, interest-bearing loans and borrowings are subsequently measured at amortised cost using the EIR method. Gains and losses are recognised in profit or loss when the liabilities are derecognized.

Amortised cost is calculated by taking into account any discount or premium on acquisition and fees or costs that are an integral part of the EIR. The EIR amortisation is included as finance costs in the statement of profit and loss.

This category generally applies to interest-bearing loans and borrowings.

Preference shares, which are mandatorily redeemable on a specific date, are classified as liabilities. The dividends on these preference shares are recognised in profit or loss as finance costs.

The fair value of the liability portion of an optionally convertible bonds is determined using a market interest rate for an equivalent non-convertible bonds. This amount is recorded as a liability on an amortised cost basis until extinguished on conversion or redemption of the bonds. The remainder of the proceeds is attributable to the equity portion of the compound instrument. This is recognised and included in shareholders' equity, net of income tax effects, and not subsequently remeasured.

Where the terms of a financial liability are renegotiated and the entity issues equity instruments to a creditor to extinguish all or part of the liability (debt for equity swap), a gain or loss is recognised in profit or loss, which is measured as the difference between the carrying amount of the financial liability and the fair value of the equity instruments issued.

### Derecognition

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit or loss.

### Offsetting of financial instruments

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable right to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable right must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the company or the counterparty.

### vii) Provisions

A provision is recognised in the balance sheet when the Company has a legal or constructive obligation as a result of a past event, and it is probable that an outflow of economic benefits will be required to settle the obligation. If the effect is material, provisions are determined by discounting the expected future cash flows at a pre-tax rate that reflects current market assessments of the time value of money and, where appropriate, the risks specific to the liability.

### viii) Earnings per share

The Company presents basic and diluted earnings per share (EPS) data for its ordinary shares. Basic EPS is calculated by dividing the profit or loss attributable to ordinary shareholders of the Company by the weighted average number of ordinary shares outstanding during the period, adjusted for own shares held. Diluted EPS is determined by adjusting the profit or loss attributable to ordinary shareholders and the weighted average number of ordinary shareholders and the weighted average number of ordinary shareholders and the weighted average number of ordinary shares outstanding, adjusted for own shares held, for the effects of all dilutive potential ordinary shares.





### Wellkept Facility Management Services Private Limited

(formerly known as TRIL Hospitality Private Limited)

Notes to the Ind AS financial statements

### (Currency: Indian rupees in lakhs)

#### 3 Cash and Cash Equivalents

Particulars	As at 31 March 2019	As at 31 March 2018	
Cash and Cash Equivalents			
Balances with Bank	5.65	0.28	
Total	5.65	0.28	

#### 4 Other Assets

Particulars	As at 31 Ma	As at 31 March 2018		
	Non Current	Current	Non Current	Current
Others Receivable - Related Party		-	-	23.67
Total		-	-	23.67

### 5 Equity Share Capital

(a) Authorised, Issued, Subscribed and Fully Paid up :

Particulars	As at 31 Ma	rch 2019	As at 31 March 2018	
	No of Shares	Amount	No of Shares	Amount
Authorised Capital :		A STATE OF A		
Equity Shares of Rs. 10/- each	10,00,000	100.00	10,00,000	100.00
Issued, Subscribed and Fully Paid up Capital : Issued Capital				
Equity Shares of Rs.10/- each, fully paid	4,00,000	40.00	4,00,000	40.00
Subscribed and Fully Paid up Capital	Martin Martin Constanting			
Equity Shares of Rs. 10/- each, fully paid	4,00,000	40.00	4,00,000	40.00
Total		40.00		40.00

### (b) Reconciliation of Number of Shares Outstanding

Particulars	As at 31 Ma	As at 31 March 2019		rch 2018
	No of Shares	Amount	No of Shares	Amount
As at the beginning of the year	4,00,000	40.00	4,00,000	40.00
Add: Issued during the year		-	-	-
As at the end of the year	4,00,000	40.00	4,00,000	40.00

#### (c) Terms and rights attached to the equity share

The Company has only one class of equity shares having par value of INR 10 per share. Each holder of equity shares is entitled to one vote per share. The Company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject the approval of shareholders in the ensuing Annual General Meeting.

During the year ended 31 March 2019, the amount of per share dividend recognised as distribution to equity shareholders is Rs Nil (2018: Rs Nil). In the event of liquidation of the Company, the holders of the equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the share holders.

#### (d) Shares of the company held by the Holding company

Name of Shareholder	As at 31 Ma	rch 2019	As at 31 March 2018		
	No of Shares	Amount	No of Shares	Amount	
Equity shares of INR 10 each, fully paid-up by Tata Realty and Infrastructure Limited, the holding company and its nominee	4,00,000	40.00	4,00,000	40.00	

Name of Shareholder	As at 31 March 2019		As at 31 March 2018	
	No of Shares	% Holding	No of Shares	% Holding
Equity shares of INR 10 each, fully paid-up by Tata Realty and Infrastructure Limited, the holding company and its nominee	4,00,000	100%	4,00,000	100%

# (f) Aggregate number of bonus shares issued and shares issued for consideration other than cash during the period of five years immediately preeceding the reporting date :

During the five year period ended 31 March, 2019, the company has not issued any equity shares for consideration other than cash.

### 6 Other Equity

(

Particulars	As at 31 March 2019	As at 31 March 2018
Retained earnings	(41.44)	(40.38)
TOTAL	(41.44)	(40.38)
NS e Retained earnings Particulais	As at 31 March 2019	As at 31 March 2018
Defielt induction	As at 51 March 2019	As at 51 March 2016
(Deficit) in statement of profit and loss Balance at the beginning of the year	(40.38)	(39.83)
Add: (Loss) for the year	(1.06)	(0.55)
Balance at the end of the Year	105 (41.44)	(40.38)



Notes to the Ind AS financial statements (Currency: Indian rupees in lakhs)

# 7 Current Financial Liabilities - Borrowings

Particulars	As at 31 March 2019	As at 31 March 2018
Interest-free loans from related parties recoverable on demand		
Tata Realty and Infrastructure Limited	6.50	6.50
TOTAL	6.50	6.5
The above amount includes		
Secured Borrowings		
Unsecured Borrowings	6.50	6.50

### 8 Other Current Financial Liabilities

Particulars	As at 31 March 2019	As at 31 March 2018
Other Payables		
- Related Party		17.53
- Others	0.59	0.30
	0.	59 17.83
Total	0.	59 17.83

#### 9 Other Expenses

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018
Fees & Consultations	0.08	0.21
Rates & Taxes ( incl indirect taxes)	0.06	0.07
Audit Fees (refer note 9 (a))	0.92	0.27
Total	1.06	0.55

### (a) Remuneration to Statutory Auditors

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018
Statutory audit	0.50	0.23
Other Services	0.33	( <b>-</b> )
Service tax / GST	0.09	0.04
Total	0.92	0.27

#### 10 Deferred Tax Assets

#### (a) Amounts recognised in profit and loss

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018
Current income tax		-
Origination and reversal of temporary differences		
Reduction in tax rate	and the second second second second	
Recognition of previously unrecognised tax losses		-
Change in recognised deductible temporary differences		1-
Total deferred tax expense/(benefit)		-
Tax expense for the year	in the second	-
(b) Reconciliation of effective tax rate		
Loss before tax	(1.06)	(0.55
Tax using the Company's domestic tax rate 25% (p.y. 25%)	(0.27)	(0.14
Reduction in tax rate		-
Tax effect of:		
Current-year losses for which no deferred tax asset is recognised	0.27	0.14
Total		

#### (c) Unrecognised deferred tax assets

Deferred tax assets have not been recognised in respect of the following items, because of no virtual certainty of the profit in the future year against which the deferred tax asset created can be utilised.

			Gross amount	Unrecognised tax effect	Gross amount	Unrecognised tax effect
	Tax losses		7.55	1.89	8.12	2.03
	(d) Tax losses carried Particulars	I forward Financial Year	Ac at 21 M	larch 2019	As at 31 M	arah 2019
	Farticulars	Financial fear	Gross amount	Expiry date	Gross amount	Expiry date
	Business loss	2010-11	-	2018-19	1.12	2018-19
HASKI	Business loss	2011-12	0.73	2019-20	0.73	2019-20
		2012-13	0.63	2020-21	0.63	2020-21
L' Chai	te Business loss	2013-14	1.23	2021-22	1.23	2021-22
Acco	unt Business loss	2014-15	0.99	2022-23	0.99	2022-23
5 (1000	Business loss	2015-16	2.54	2023-24	2.54	2023-24
30	Business loss	2016-17	0.88	2024-25	0.88	2024-25
6	* Business loss	2017-18	0.55	19625-26	-	2025-26
	Total		7.55		8.12	



### Wellkept Facility Management Services Private Limited

(formerly known as TRIL Hospitality Private Limited)

### Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

The Company offsets tax assets and liabilities if and only if it has a legally enforceable right to set off current tax assets and current tax liabilities and the deferred tax assets and deferred tax liabilities relate to income taxes levied by the same tax authority.

Significant management judgement is required in determining provision for income tax, deferred income tax assets and liabilities and recoverability of deferred income tax assets. The recoverability of deferred income tax assets is based on estimates of taxable income and the period over which deferred income tax assets will be recovered. Any changes in future taxable income would impact the recoverability of deferred tax assets.

### 11 Due to micro and small suppliers

Under the Micro, Small and Medium Enterprises Development Act, 2006, (MSMED) which came into force from 2 October 2006, certain disclosures are required to be made relating to Micro, Small and Medium enterprises. On the basis of the information and records available with the Management, there are no outstanding dues to the Micro, Small and Medium enterprises as defined in the Micro, Small and Medium Enterprises Development Act, 2006 as set out in following disclosure.

Particulars	As at 31 March 2019	As at 31 March 2018
Principal amount remaining unpaid to any supplier as at the year-end		
Interest due thereon	-	
The amount of interest paid by the buyer as per the Micro Small and Medium Enterprises Development Act, 2006 (MSMED Act, 2006)		
The amount of interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under MSMED Act, 2006		
The amount of interest accrued and remaining unpaid at the end of each accounting year		
The amount of further interest remaining due and payable even in the succeeding years, until such date when the interest dues as above are actually paid to the small enterprise for the purpose of disallowance as a deductible expenditure under the MSMED Act, 2006		

#### 12 Earning Per Share

(a

Та

Earnings Per Share (EPS) = Net Profit attributable to Shareholders / Weighted Number of Shares Outstanding

Particulars	For the year er	nded 31 March 2019	For the year ended 31 March 2018
(Loss) after tax attributable to equity shareholders	A	(1.06)	(0.55
Calculation of weighted average number of equity shares:			
Number of equity shares at the beginning of the year	с	4,00,000	4,00,000
Optionally convertible participatory debentures		A SHARE AND A SHARE AND	-
Number of equity shares outstanding at the end of the year		4,00,000	4,00,000
Weighted average number of equity shares outstanding during the year	В	4,00,000	4,00,000
Earning Per Share – Basic (Rs.)	(A / B)	(0.27)	(0.14
Earning Per Share - Diluted (Rs.)		(0.27)	(0.14

13 Disclosure of transactions with Related Parties, as required by Indian Accounting Standard (Ind AS) - 24

) Related Party Disclosure	
Ultimate Holding Company	
Tata Sons Limited	

Holding Company

Fellow subsidiary **TRIL Infopark Limited** 

ta	Realty	and	Infrastructure	Limited	

f Transactions / relationship / major parties Ultimate holding Holding Co Company		Fellow subsidiary	Total
-	-	-	-
(-)	(17.53)	(-)	(17.53
			6.1. W. B. B. B.
-	6.50	-	6.50
(-)	(6.50)	(-)	(6.50
			-
(-)	(0.35)	(-)	(0.35
	1.000		
2	-	-	-//
(-)	(-)	(23.67)	(23.67
	Company - (-)	Company (-) (17.53) - 6.50 (-) (6.50)	Company subsidiary   - -   - -   - -   - -   - -   - -   - -   - -   - -   - -   - -   - -   - -   - -

\*Note figures in bracket represent previous years figures.

### Wellkept Facility Management Services Private Limited

(formerly known as TRIL Hospitality Private Limited)

### Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

### 14 Segment Reporting as per IND AS108 " Operating Segments"

Operating segments are reported in a manner consistent with the internal reporting provided to the chief operating decision maker (CODM). Chief operating decision maker's function is to allocate the resources of the entity and access the performance of the operating segment of the entity. The Board of directors of the company assesses the financial performance and position of the Company and makes strategic decisions. It is identified as being the chief operating decision maker for the company. The Company is operating in the real estate and infrastructure industry and has only domestic operations. The Company has only one operating segment, which is development of real estate and infrastructure facilities. All assets of the Company are domiciled in India and the Company has no revenue.

#### 15 Other matters

Information with regard to other matters specified in Schedule III of the Act is either nil or not applicable to the Company for the year.

#### 16 Capital commitment, contingencies and other commitments

There are no capital commitments and contingent liabilities as at the balance sheet date. Estimated amount of other commitments remaining to be executed on capital account and not provided for (net of advances) Rs Nil (2018: Rs Nil).

#### 17 Financial instruments - Fair values and risk management

#### A Credit risk

Credit risk is the risk of financial loss to the Company if a customer or counterparty to a financial instrument fails to meet its contractual obligations, and arises principally from the Company's receivables from customers and investment securities. The carrying amounts of financial assets represent the maximum credit exposure.

#### Cash and cash equivalents

The Company held cash and cash equivalents with credit worthy banks and financial institutions of INR 5.73 lakhs and INR 0.28 lakhs as at 31 March 2019 and 31 March 2018 respectively. The credit worthiness of such banks and financial institutions is evaluated by the management on an ongoing basis and is considered to be good.

#### B Liquidity risk

Liquidity risk is the risk that the Company will not be able to meet its financial obligations as they become due. The Company manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Company's reputation.

The Company has obtained a finance facility from a bank. The Company also constantly monitors funding options available in the debt and capital markets with a view to maintaining financial flexibility.

As at 31 March, 2019, the Company had working capital of Rs (1.44) lakhs inculding short term borrowings of Rs 6.50 lakhs. As at 31 March, 2018, the Company had working capital of Rs (0.38) lakhs inculding short-term borrowings of Rs 6.50 lakhs.

#### C Exposure to liquidity risk

The table below analyses the Company's financial liabilities into relevant maturity groupings based on their contractual maturities for:

### - all non derivative financial liabilities

Particulars	Carrying amount	Total	1 year or less	1-2 years	2-5 years
As at 31 March 2019					
Non-derivative financial liabilities					
Unsecured short-term borrowings	6.50	6.50	6.50		
Other financial liabilities	0.59	0.59	0.59		
	7.09	7.09	7.09		
As at 31 March 2018					
Non-derivative financial liabilities					
Unsecured short-term borrowings	6.50	6.50	6.50		
Other financial liabilities	17.83	17.83	17.83		
	24.33	24.33	24.33		

#### D Market risk

Market risk is the risk that changes in market prices – such as foreign exchange rates, interest rates and equity prices – will affect the Company's income or the value of its holdings of financial instruments. The Company is domiciled in India and has its revenues and other transactions in its functional currency i.e. INR. Accordingly the Company is not exposed to any currency risk. Also the Company does not hold any equity investments, accordingly the Company is not exposed to any currency risk.

#### E Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Company's exposure to market risk for changes in interest rates relates to borrowings from Holding Company.

#### Exposure to interest rate risk

The interest rate profile of the company's interest-bearing financial instruments is as follows.					
Fixed-rate instruments	As at 31 March 2019	As at 31 March 2018			
Financial liabilities					
ICD from Tata Realty and Infrastructure Limited	6.50	6.50			

The Company does not account for any fixed-rate financial assets or financial liabilities at fair value through profit or loss, and the Company does not have any designate derivatives (interest rate swaps).

Therefore, a change in interest rates at the reporting date would not affect profit or loss.

#### Capital management

in

F

Chartered Accountants

The Company's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. Management monitors the return on capital as well as the level of dividends to ordinary shareholders.

#### The Company has obtained borrowings primarily from its holding company.

The Company monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total borrowings, SKIN sconterising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity. Charge for the year on goodwill amortisation has been deducted while calculating total equity of the company since it represents a pure non-cash expense.
## Wellkept Facility Management Services Private Limited

(formerly known as TRIL Hospitality Private Limited)

#### Notes to the Ind AS financial statements

## (Currency: Indian rupees in lakhs)

The Company's adjusted net debt to equity ratio at March 31, 2019 was as follows.

Paticulars	As at 31 March 2019	As at 31 March 2018
Total borrowings	6.50	6.50
Less : Cash and cash equivalent	5.65	0.28
Adjusted net debt	0.85	6.22
Adjusted equity	(1.44)	(0.38)
Adjusted net debt to adjusted equity ratio %	(0.59)	(16.57)

#### 18 Financial instruments - Fair values and risk management

## A Accounting classification and fair values

The following table shows the carrying amounts and fair values of financial assets and financial liabilities, including their levels in the fair value hierarchy. It does not include fair value information for financial assets and financial liabilities if the carrying amount is a reasonable approximation of fair value.

Particulars	FVTPL	FVTOCI	Amotised Cost	Total	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Total
31 March 2019				1 Et soleen				1.000
Financial assets								
Cash and cash equivalents	-	-	5.65	5.65		-	-	
Financial liabilities								
Interest-free loans from			6.50	6.50				
related parties	•	-	0.50	6.50	-	-	-	-
Other current financial			0.59	0.59				
liabilities		30 <del>7</del> 3	0.55	0.55	(			1.70
31 March 2018								
Financial assets								
Cash and cash equivalents	( <b>L</b> )(	-	0.28	0.28	-	-	-	
Financial liabilities								
Interest-free loans from	-		6.50	6.50				120
related parties	-	-	0.50	0.50		-	-	-
Other current financial	-		17.83	17.83	-		-	
liabilities			17:00	17.00				

#### B Financial risk management

The Company has exposure to the following risks arising from financial instruments:

Credit risk ;

· Liquidity risk ; and

Market risk

#### C Risk management framework

The Company's board of directors has overall responsibility for the establishment and oversight of the Company's risk management framework.

#### D Measurement of fair values

#### Valuation techniques and significant unobservable inputs

The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for financial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used.

<b>Financial instru</b>	ments measure	d at	fair value
-------------------------	---------------	------	------------

Туре	Valuation technique	Significant	Inter-relationship
Amortised cost: 1. Borrowings	Discounted cash flow approach: The valuation model considers the present value of expected payments, discounted using a risk adjusted discount rate.		Not applicable

Transfers between Levels 1 and 2

There have been no transfers between Level 1 and Level 2 during the reporting periods

Level 3 fair values

There are no items in Level 3 fair values

19 There are no offsetting of financial assets and financial liabilities during the year.

#### 20 Subsequent events

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

For and on behalf of the **Board of Directors of Wellkept Facility Management Services Private Limited** (formerly known as TRIL Hospitality Private Limited) CIN: U93000MH2008PTC177346

Nandlal Singh Director DIN: 07384037

Amit Sheth Director DIN: 07997663



### **Chartered Accountants**

Indiabulls Finance Centre Tower 3, 27th - 32nd Floor Senapati Bapat Marg Elphinstone Road (West) Mumbai - 400 013 Maharashtra, India Tel: +91 22 6185 4000 Fax: +91 22 6185 4001

## **INDEPENDENT AUDITOR'S REPORT**

## To The Members of Wellkept Facility Management Services Private Limited **Report on the Audit of the Ind AS Financial Statements**

## Opinion

We have audited the accompanying Ind AS financial statements of Wellkept Facility Management Services Private Limited ("the Company"), which comprise the Balance Sheet as at 31 March 2020, and the Statement of Profit and Loss (including Other Comprehensive Income), the Statement of Cash Flows and the Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under Section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended, ("Ind AS") and other accounting principles generally accepted in India, of the state of affairs of the Company as at 31 March 2020, and its loss, total comprehensive loss, its cash flows and the changes in equity for the year ended on that date.

## **Basis for Opinion**

We conducted our audit of the Ind AS financial statements in accordance with the Standards on Auditing specified under Section 143(10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Ind AS Financial Statements section of our report. We are independent of the Company in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the Ind AS financial statements.

## Material uncertainty related to Going Concern

We draw attention to Note 2(b) to the Ind AS financial statements, wherein the events or conditions, along with other matters as set forth in the said note, indicate that a material uncertainty exists that may cast significant doubt on the Company's ability to continue as a going concern. However, the Ind AS financial statements of the Company have been prepared on a going concern basis for the reasons stated in the said Note.

Our opinion is not modified in respect of this matter.

## Information Other than the Financial Statements and Auditor's Report Thereon

The Company's Board of Directors is responsible for the other information. The other information comprises the Director's report which we obtained prior to the date of this auditor's report.

- Our opinion on the Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.
- In connection with our audit of the Ind AS financial statements, our responsibility is to read the
  other information and, in doing so, consider whether the other information is materially
  inconsistent with the Ind AS financial statements or our knowledge obtained during the course
  of our audit or otherwise appears to be materially misstated.
- If, based on the work we have performed, we conclude that there is a material misstatement of this other information, we are required to report that fact. We have nothing to report in this regard.

## Management's Responsibility for the Ind AS Financial Statements

The Company's Board of Directors is responsible for the matters stated in Section 134(5) of the Act with respect to the preparation of these Ind AS financial statements that give a true and fair view of the financial position, financial performance including other comprehensive income, cash flows and changes in equity of the Company in accordance with the Ind AS and other accounting principles generally accepted in India.

This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the Ind AS financial statements that give a true and fair view and are free from material misstatement, whether due to fraud or error.

In preparing the Ind AS financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those Board of Directors are also responsible for overseeing the Company's financial reporting process.

## Auditor's Responsibility for the Audit of the Ind AS Financial Statements

Our objectives are to obtain reasonable assurance about whether the Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these Ind AS financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

• Identify and assess the risks of material misstatement of the Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.

RKU

- Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under Section143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Company has adequate internal financial controls system in place and the operating effectiveness of such controls.
- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.
- Evaluate the overall presentation, structure and content of the Ind AS financial statements, including the disclosures, and whether the Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Materiality is the magnitude of misstatements in the Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the Ind AS financial statements.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

## **Report on Other Legal and Regulatory Requirements**

- 1. As required by Section 143(3) of the Act, based on our audit we report, to the extent applicable that:
  - a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit.
  - b) In our opinion, proper books of account as required by law have been kept by the Company so far as it appears from our examination of those books
  - c) The Balance Sheet, the Statement of Profit and Loss including Other Comprehensive Income, the Statement of Cash Flows and Statement of Changes in Equity dealt with by this Report are in agreement with the relevant books of account.
  - d) In our opinion, the aforesaid Ind AS financial statements comply with the Ind AS specified under Section 133 of the Act.

e) The matter described in the Material uncertainty related to Going Concern section above, in our opinion, may have an adverse effect on the functioning of the Company.

RKH

- f) On the basis of the written representations received from the directors as on 31 March 2020 taken on record by the Board of Directors, none of the directors is disqualified as on 31 March 2020 from being appointed as a director in terms of Section 164(2) of the Act.
- g) Reporting on the adequacy of the internal financial controls over financial reporting of the Company and the operating effectiveness of such controls is not applicable to the company in view of the exemption available to the company in terms of the notification no. G.S.R. 583(E) dated 13 June 2017 issued by the Ministry of Corporate Affairs, Government of India, read with general circular No. 08/2017 dated 25 July 2017.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of Section 197(16) of the Act, as amended.

In our opinion and to the best of our information and according to the explanations given to us, the Company being a private company, Section 197 of the Act related to managerial remuneration is not applicable.

- With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended in our opinion and to the best of our information and according to the explanations given to us:
  - i. The Company does not have any pending litigations which would impact its financial position.
  - ii. The Company did not have any long-term contracts including derivative contracts for which there were any material foreseeable losses.
  - iii. There were no amounts, which were required to be transferred to the Investor Education and Protection Fund by the Company
- 2. As required by the Companies (Auditor's Report) Order, 2016 ("the Order") issued by the Central Government in terms of Section 143(11) of the Act, we give in "Annexure A" a statement on the matters specified in paragraphs 3 and 4 of the Order.

For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No.117366W/W-100018)

ptkine.

Rajesh K. Hiranandani Partner (Membership No. 36920) UDIN: 20036920AAAABM3474

Mumbai, 17 June 2020

## ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT

(Referred to in paragraph 2 under 'Report on Other Legal and Regulatory Requirements' Section of our report of even date)

- (i) The Company does not have any fixed assets and hence reporting under clause (i) of the CARO 2016 is not applicable.
- (ii) The Company does not have any inventory and hence reporting under clause (ii) of the CARO 2016 is not applicable.
- (iii) The Company has not granted any loans, secured or unsecured, to companies, firms, Limited Liability Partnerships or other parties covered in the register maintained under Section 189 of the Companies Act, 2013
- (iv) The Company has not granted any loans, made investments or provided guarantees and hence reporting under clause (iv) of the CARO 2016 is not applicable.
- (v) According to the information and explanations given to us, the Company has not accepted any deposit during the year as provided under Section 73 to 76 or any other relevant provisions of the Companies Act, 2013. There are no unclaimed deposits any time during the year.
- (vi) The maintenance of cost records has not been specified by the Central Government under Section 148(1) of the Companies Act, 2013.
- (vii) According to the information and explanations given to us, in respect of statutory dues:
  - (a) The Company has been regular in depositing undisputed statutory dues, including Incometax, Goods & Services Tax, cess and other material statutory dues applicable to it to the appropriate authorities.
  - (b) There were no undisputed amounts payable in respect of Income-tax, Goods & Services Tax, cess and other material statutory dues in arrears as at 31 March 2020 for a period of more than six months from the date they became payable.
  - (c) There are no dues of Income-tax, Goods & Services Tax as on 31 March 2020 on account of disputes.
- (viii) In our opinion and according to the information and explanations given to us, The Company has not taken any loans or borrowings from financial institutions, banks and government or has not issued any debentures. Hence reporting under clause (viii) of CARO 2016 is not applicable to the Company.
- (ix) The Company has not raised moneys by way of initial public offer or further public offer (including debt instruments) or term loans and hence reporting under clause (ix) of the CARO 2016 Order is not applicable.
- (x) To the best of our knowledge and according to the information and explanations given to us, no fraud by the Company and no material fraud on the Company by its officers or employees has been noticed or reported during the year.
- (xi) In our opinion and according to the information and explanations given to us, the Company is not required to have and does not have any managerial personnel and hence the reporting under clause of the CARO 2016 Order for payment of managerial remuneration under Section 197 read with Schedule V of the Companies Act 2013 is not applicable.

RKU

- (xii) The Company is not a Nidhi Company and hence reporting under clause (xii) of the CARO 2016 Order is not applicable.
- (xiii) In our opinion and according to the information and explanations given to us the Company is in compliance with Section 177 and 188 of the Companies Act, 2013, where applicable, for all transactions with the related parties and the details of related party transactions have been disclosed in the financial statements etc. as required by the applicable accounting standards.
- (xiv) During the year the Company has not made any preferential allotment or private placement of shares or fully or partly convertible debentures and hence reporting under clause (xiv) of CARO 2016 is not applicable to the Company.
- (xv) In our opinion and according to the information and explanations given to us, during the year the Company has not entered into any non-cash transactions with its directors or directors of its holding, subsidiary or associate company or persons connected with them and hence provisions of Section 192 of the Companies Act, 2013 are not applicable.
- (xvi) The Company is not required to be registered under Section 45-IA of the Reserve Bank of India Act, 1934.

## For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No. 117366W/W-100018)

etter. en

Rajesh K. Hiranandani Partner (Membership No. 36920) UDIN: 20036920AAAABM3474

Mumbai, 17 June 2020

## Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited)

Balance Sheet as at 31 March 2020

				(Rs. in lakhs
Particu	lars	Note No.	As at 31 March 2020	As at 31 March 2019
ASSET	S			
NON-C	URRENT ASSETS			
(a)	Deferred tax assets (net)	10	-	
	TOTAL NON-CURRENT ASSETS			<b>1</b> 3
CURR	ENT ASSETS			
(a)	Financial assets			
1.1	(i) Cash and Cash Equivalents	3	5.64	5.65
	TOTAL CURRENT ASSETS		5.64	5.65
	TOTAL ASSETS		5.64	5.65
EQUIT	Y AND LIABILITIES			
EQUIT	Ŷ			
(a)	Equity share capital	4	40.00	40.00
(b)	Other equity	5	(41.45)	(41.44
	TOTAL EQUITY		(1.45)	(1.44
LIABIL	ITIES			
CU	RRENT LIABILITIES			
(a)	Financial liabilities			
	(i) Borrowings	67	6.50	6.50
	(ii) Other financial liabilities	7	0.59	0.59
	TOTAL CURRENT LIABILITIES		7.09	7.09
	TOTAL EQUITY AND LIABILITIES	1.0	5.64	5.65
Signific	cant accounting policies	2		
Notes t	to the Ind AS financial statements	3 - 20		

The accompanying notes 1 to 20 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

For and on behalf of the Board of Directors of Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) CIN: U93000MH2008PTC177346

PE

Rajesh K. Hiranandani Partner

	Mumbai	
R	Dated : 17 June 2020	

Sunil Dhagat Director DIN: 03081163



Amit Sheth Director DIN: 07997663



## Wellkept Facility Management Services Private Limited

(formerly known as TRIL Hospitality Private Limited)

## Statement of Profit and Loss for the year ended 31 March 2020

				(NS III Idkiis)
Parti	culars	Note No.	For the year ended 31 March 2020	For the year ended 31 March 2019
1	Revenue from Operations :			
11	Other Income	8	0.01	
III	Total Income (I +II)		0.01	
IV	Expenses:			
	Other expenses	9	0.02	1.06
	Total Expenses		0.02	1.06
V	Loss before tax (III-IV)		(0.01)	(1.06)
VI	Tax expenses		-	
VII	Loss for the year (V-VI)		(0.01)	(1.06)
VIII	Other Comprehensive Income:	00		260
IX	Total Comprehensive Loss for the Year (VII+VIII)		(0.01)	(1.06)
x	Earnings per equity share	12		
	Basic		(0.00)	(0.27)
	Diluted		(0.00)	(0.27)
Sign	ificant accounting policies	2		
Note	s to the Ind AS financial statements	3 - 20		

The accompanying notes 1 to 20 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

pra. in

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020 For and on behalf of the Board of Directors of Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) CIN: U93000MH2008PTC177346

Ihag

Sunil Dhagat Director DIN: 03081163 Amit Sheth Director DIN: 07997663

(Rs in lakhs)

## Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited)

Statement of Cash Flow for the year ended 31 March 2020

				(Rs in lakhs)
	Particulars	For the year ended March 2020	31	For the year ended 31 March 2019
A	CASH FLOW FROM OPERATING ACTIVITIES			
	Loss Before Tax	9	(0.01)	(1.06)
	(Decrease) in Other Non Financial Assets		-	23.67
	(Decrease) in other financial liabilities		-	(17.24)
	Cash (used in)/ generated from operations	1	(0.01)	5.37
	Direct Taxes Paid		-	
	Net Cash Flow (used in)/ Generated from Operating Activities		(0.01)	5.37
в	CASH FLOW FROM INVESTMENT ACTIVITIES			
	Net Cash Flow Used in Investment Activities			1
С	CASH FLOW FROM FINANCING ACTIVITIES			
	Net Cash Flow Used in Financing Activities		•	
	NET (DECREASE)/ INCREASE IN CASH AND CASH EQUIVALENTS		(0.01)	5.37
	Opening Balance		5.65	0.28
	Closing Balance		5.64	5.65
	NET (DECREASE)/ INCREASE IN CASH AND CASH EQUIVALENTS		(0.01)	5.37
	Components of Cash and Cash Equivalents			
	Cash on Hand			
	Balances with Bank		5.64	5.65
	Total Balance		5.64	5.65

#### Note:

The Cash flow statement has been prepared under the indirect method as set out in Indian Accounting Standard - 7 ('Ind AS 7') on Cash Flow Statement prescribed in the Companies (Indian Accounting Standard) Rules, 2015, notified under Section 133 of the Companies Act, 2013.

The accompanying notes 1 to 20 form an integral part of these Ind AS financial statements.

As per our report of even date attached For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

pra. ...

Rajesh K. Hiranandani Partner

	Mumbai
R	Dated : 17 June 2020

For and on behalf of the Board of Directors of Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) CIN: U93000MH2008PTC177346

Sunil Dhagat

Director DIN: 03081163 Amit Sheth Director DIN: 07997663

Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### A Equity Share Capital

Particulars	31 March 2020	31 March 2019
Subscribed and Fully Paid up Capital		
Equity shares of INR 10 each		
Opening Balance	40.00	40.00
Changes in equity share capital during the year		-
Closing Balance	40.00	40.00

## **B** Other Equity

Particulars	Retained earnings	Other comprehensive income	Total
Balance as at 01 April 2018	(40.38)		(40.38)
Loss for the year	(1.06)		(1.06)
Balance as at 31 March 2019	(41.44)		(41.44)
Loss for the year	(0.01)		(0.01)
Balance as at 31 March 2020	(41.45)		(41.45)

The accompanying notes 1 to 20 form an integral part of these Ind AS financial statements.

As per our report of even date For Deloitte Haskins & Sells LLP Chartered Accountants

Firm Registration No. 117366W/ W- 100018

PF

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020 For and on behalf of the Board of Directors of Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited)

CIN: U93000MH2008PTC177346

B.

Sunil Dhagat Director DIN: 03081163

Amit Sheth Director DIN: 07997663

### Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited)

Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

## Background of the Company

- a) Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) ('the Company') was incorporated on 4 January 2008 to carry on the business of development of real estate and infrastructure facilities. The Company is a wholly owned subsidiary of Tata Realty and Infrastructure Limited, which is a wholly owned subsidiary of Tata Sons Private Limited, the ultimate holding Company. The company is a private limited company incorporated and domiciled in India and has its registered office at Mumbai, Maharashtra, India.
- b) In order to achieve operating efficiency, the Board of Directors of the Company, in their meeting held on 2 January 2019, had approved a scheme of amalgamation between Wellkept Facility Management Services Private Limited ("the Transferor Company") with its parent company viz. Tata Realty and Infrastructure Limited ("the Transferee Company") and their respective members and creditors w.e.f. 1 April 2019 ("Appointed date"). Consequently, on 13 February 2019, the companies have filed the said scheme of amalgamation with National Company Law Tribunal, Mumbai.

#### 2 SIGNIFICANT ACCOUNTING POLICIES

#### i) Basis of Preparation

#### a) Statement of compliance:

These Ind AS financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 (the Act) and other relevant provisions of the Act.

These Ind AS financial statements were authorised for issue by the Company's Board of Directors on 17 June 2020

#### b) Going concern :

As at 31 March 2020, the Company's paid up capital and reserves (excluding the deficit in the profit and loss) were INR 40 lakhs (2019: INR 40 lakhs) correspondingly, the Company's accumulated losses aggregated INR 41.45 lakhs (2019: INR 41.44 lakhs) and the Company's current liabilities exceeded its current assets by INR.1.45 lakhs. These factors give rise to a material uncertainty whether the Company would be able to continue as a going concern. However, in pursuance to Note No. 1 (b) above, the financial statements have been prepared on a going concern basis. Accordingly, these financial statements do not include any adjustments relating to the recoverability and classification of recorded assets, or to amounts and classification of recorded liabilities that might be necessary if the Company is unable to continue as a going concern.

#### c) Historical cost convention:

The financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

- 1 Financial instruments measured at fair value through profit or loss, if applicable
- 2 Financial instruments measured at fair value through other comprehensive income, if applicable

#### d) Key estimates and assumptions

In preparing these financial statements, management has made judgments, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates.

Estimates and assumptions are required in particular for:

· Measurement of fair values

The Company's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Company has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the group CFO.

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorised into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

· Level 1: quoted prices (unadjusted) in active markets for identical assets or liabilities.

• Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

· Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorised in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement

#### ii) Functional and presentation currency

The financial statements are presented in Indian rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian rupees (INR) and all amounts have been rounded-off to the nearest lakhs, except for share data and as otherwise stated. Due to rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.

#### iii) Borrowing cost

Borrowing costs are expensed in the period in which they are incurred.



Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) Notes to the Ind AS financial statements

for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### iv) Income tax

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised directly in equity or in OCI.

#### a) Current tax

Current tax comprises the expected tax payable or receivable on the taxable income or loss for the year and any adjustment to the tax payable or receivable in respect of previous years. It is measured using tax rates enacted or substantively enacted at the reporting date. Current tax also includes any tax arising from dividends. Management periodically evaluates positions taken in tax returns with respect to situations in which applicable tax regulation is subject to interpretation. It establishes provisions where appropriate on the basis of amounts expected to be paid to the tax authorities.

Minimum Alternate Tax ('MAT') under the provisions of Income-tax Act, 1961 is recognised as current tax in the statement of profit and loss. MAT paid in accordance with the tax laws, which gives future economic benefits in the form of adjustment to future income tax liability, is considered as an asset if there is a convincing evidence that the Company will pay normal tax. Accordingly, MAT is recognised as an asset in the balance sheet when it is probable that the future economic benefit associated with it will flow to the Company.

Current tax assets and liabilities are offset only if, the Company:

- 1 has a legally enforceable right to set off the recognised amounts; and
- 2 intends either to settle on a net basis, or to realise the asset and settle the liability simultaneously.

#### b) Deferred tax

Deferred tax is recognised in respect of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for taxation purposes. However, deferred tax liabilities are not recognised if they arise from the initial recognition of goodwill. Deferred income tax is also not accounted for if it arises from initial recognition of an asset or liability in a transaction other than a business combination that at the time of the transaction affects neither accounting profit nor taxable profit (tax loss).

Deferred tax assets are recognised for unused tax losses, unused tax credits and deductible temporary differences to the extent that it is probable that future taxable profits will be available against which they can be used. Deferred tax assets are reviewed at each reporting date and are reduced to the extent that it is no longer probable that the related tax benefit will be realised; such reductions are reversed when the probability of future taxable profits improves. Unrecognised deferred tax assets are reassessed at each reporting date and recognised to the extent that it has become probable that future taxable profits will be available against which they can be used.

Deferred tax is measured at the tax rates that are expected to be applied to temporary differences when they reverse, using tax rates enacted or substantively enacted at the reporting date and are expected to apply when the related deferred income tax asset is realised or the deferred income tax liability is settled.

Deferred tax assets and liabilities are offset only if:

- 1 the entity has a legally enforceable right to set off current tax assets against current tax liabilities; and
- 2 the deferred tax assets and the deferred tax liabilities relate to income taxes levied by the same taxation authority on the same taxable entity.

#### v) Cash and cash equivalents

For the purpose of presentation in the statement of cash flows, cash and cash equivalents includes cash on hand, deposits held at call with financial institutions, other short-term, highly liquid investments with original maturities of three months or less that are readily convertible to known amounts of cash and which are subject to an insignificant risk of changes in value, and bank overdrafts. Bank overdrafts are shown within borrowings in current liabilities in the balance sheet.

#### vi) Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity. Financial instruments also include derivative contracts such as foreign currency foreign exchange forward contracts, interest rate swaps and currency options; and embedded derivatives in the host contract.

### **1** Financial assets

#### Classification

The Company classifies financial assets as measured at amortised cost, at fair value through other comprehensive income (FVOCI) or at fair value through profit or loss (FVTPL) on the basis of its business model for managing the financial assets and the contractual cash flow characteristics of the financial asset.

#### Initial recognition and measurement

All financial assets are recognised initially at fair value plus, in the case of financial assets not recorded at fair value through profit or loss, transaction costs that are attributable to the acquisition of the financial asset. Purchases or sales of financial assets that require delivery of assets within a time frame established by regulation or convention in the market place (regular way trades) are recognised on the trade date, i.e., the date that the Company commits to purchase or sell the asset.

#### Debt instruments

A 'debt instrument' is measured at the amortised cost if both the following conditions are met:

a) The asset is held within a business model whose objective is to hold assets for collecting contractual cash flows, and

b) Contractual terms of the asset give rise on specified dates to cash flows that are solely payments of principal and interest (SPPI) on the principal amount outstanding.

After initial measurement, such financial assets are subsequently measured at amortised cost using the effective interest rate (EIR) method. Amortised cost is calculated by taking into account any discount or premium and fees or costs that are an integral part of the EIR. The EIR amortisation is included in finance income in the profit or loss.

Debt instruments included within the fair value through profit and loss (FVTPL) category are measured at fair value with all changes recognised in the statement of profit and loss. 121

1245163X

De-recognition REH

Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited)

## Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

A financial asset (or, where applicable, a part of a financial asset or part of a Company of similar financial assets) is primarily derecognised (i.e. removed from the Company's balance sheet) when:

The rights to receive cash flows from the asset have expired, or

The Company has transferred its rights to receive cash flows from the asset or has assumed an obligation to pay the received cash flows in full without material delay to a third party under a 'pass-through' arrangement; and either (a) the Company has transferred substantially all the risks and rewards of the asset, or (b) the Company has neither transferred nor retained substantially all the risks and rewards of the asset.

When the Company has transferred its rights to receive cash flows from an asset or has entered into a pass-through arrangement, it evaluates if and to what extent it has retained the risks and rewards of ownership. When it has neither transferred nor retained substantially all of the risks and rewards of the asset, nor transferred control of the asset, the Company continues to recognise the transferred asset to the extent of the Company's continuing involvement. In that case, the Company also recognises an associated liability. The transferred asset and the associated liability are measured on a basis that reflects the rights and obligations that the Company has retained.

Continuing involvement that takes the form of a guarantee over the transferred asset is measured at the lower of the original carrying amount of the asset and the maximum amount of consideration that the Company could be required to repay.

#### Impairment of financial assets

In accordance with Ind-AS 109, the Company applies expected credit loss (ECL) model for measurement and recognition of impairment loss on the following financial assets and credit risk exposure:

a) Financial assets that are debt instruments, and are measured at amortised cost e.g., loans, debt securities, deposits, and bank balance





### Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### **2** Financial liabilities

#### Classification

The Company classifies all financial liabilities as measured at amortised cost, except for financial liabilities at fair value through profit or loss. Such liabilities, including derivatives that are liabilities, shall be subsequently measured at fair value.

### Initial recognition and measurement

Financial liabilities are classified, at initial recognition, as financial liabilities at fair value through profit or loss, loans and borrowings, payables, or as derivatives designated as hedging instruments in an effective hedge, as appropriate.

All financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

The Company's financial liabilities include trade and other payables, loans and borrowings including bank overdrafts, financial guarantee contracts and derivative financial instruments.

#### Trade and other payables

Trade and other payables represent liabilities for goods and services provided to the Company prior to the end of the financial year which are unpaid. The amounts are unsecured and are usually paid within 30-60 days of recognition or maybe payable on demand. Trade and other payables are represented as current liabilities unless payment is not due within 12 months after the reporting period. They are recognised initially at their fair value and subsequently recognised at amortised cost using the effective interest method.

#### Loans and borrowings

After initial recognition, interest-bearing loans and borrowings are subsequently measured at amortised cost using the EIR method. Gains and losses are recognised in profit or loss when the liabilities are derecognized.

Amortised cost is calculated by taking into account any discount or premium on acquisition and fees or costs that are an integral part of the EIR. The EIR amortisation is included as finance costs in the statement of profit and loss.

This category generally applies to interest-bearing loans and borrowings.

Preference shares, which are mandatorily redeemable on a specific date, are classified as liabilities. The dividends on these preference shares are recognised in profit or loss as finance costs.

The fair value of the liability portion of an optionally convertible bonds is determined using a market interest rate for an equivalent non-convertible bonds. This amount is recorded as a liability on an amortised cost basis until extinguished on conversion or redemption of the bonds. The remainder of the proceeds is attributable to the equity portion of the compound instrument. This is recognised and included in shareholders' equity, net of income tax effects, and not subsequently remeasured.

Where the terms of a financial liability are renegotiated and the entity issues equity instruments to a creditor to extinguish all or part of the liability (debt for equity swap), a gain or loss is recognised in profit or loss, which is measured as the difference between the carrying amount of the financial liability and the fair value of the equity instruments issued.

#### Derecognition

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit or loss.

#### Offsetting of financial instruments

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable right to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable right must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the company or the counterparty.

#### vii) Provisions

A provision is recognised in the balance sheet when the Company has a legal or constructive obligation as a result of a past event, and it is probable that an outflow of economic benefits will be required to settle the obligation. If the effect is material, provisions are determined by discounting the expected future cash flows at a pre-tax rate that reflects current market assessments of the time value of money and, where appropriate, the risks specific to the liability.

#### viii) Earnings per share

The Company presents basic and diluted earnings per share (EPS) data for its ordinary shares. Basic EPS is calculated by dividing the profit or loss attributable to ordinary shareholders of the Company by the weighted average number of ordinary shares outstanding during the period, adjusted for own shares held. Diluted EPS is determined by adjusting the profit or loss attributable to ordinary shareholders and the weighted average number of ordinary shareholders and the weighted average number of ordinary shares outstanding, adjusted for own shares held, for the effects of all dilutive potential ordinary shares.

REH



Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

## 3 Cash and Cash Equivalents

4

Particulars	31 March	2020	31 March	2019	
Cash and Cash Equivalents		2010-000		10000	
Balances with Bank		5.64		5.65	
Total		5.64		5.65	
ity Share Capital Authorised, Issued, Subscribed and Fully Paid up					
Particulars	31 March	2020	31 March	2019	
	No of Shares	Amount	No of Shares	Amount	
Authorised Capital :					
Equity Shares of Rs.10/- each	10,00,000	100.00	10,00,000	100.00	
Issued, Subscribed and Fully Paid up Capital : Issued Capital					
Equity Shares of Rs.10/- each, fully paid	4,00,000	40.00	4,00,000	40.00	
Subscribed and Fully Paid up Capital	Contractor Contractor	C. Second		1.	
Equity Shares of Rs.10/- each, fully paid	4,00,000	40.00	4.00.000	40.00	
Total		40.00		40.00	
Reconciliation of Number of Shares Outstanding Particulars	31 March	2020	31 March	2019	
r universite s	No of Shares	Amount	No of Shares	Amount	
As at the beginning of the year	4,00,000	40.00	4,00,000	40.00	
Add: Issued during the year	200 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100		. 19	10	
As at the end of the year	4,00,000	40.00	4,00,000	40.00	
Terms and rights attached to the equity share The Company has only one class of equity shares hav Company declares and pays dividends in Indian rupe ensuing Annual General Meeting.					
During the year ended 31 March 2020, the amount of	per share dividend rec	ognised as distrib	oution to equity shar	eholders is Rs Nil	(2019: Rs Nil).
In the event of liquidation of the Company, the holders all preferential amounts. The distribution will be in pro					bany, after distrib
	any				
Shares of the company held by the Holding comp	ally				
Name of Shareholder	31 March	2020	31 March	2019	

Equity shares of INR 10 each, fully paid-up by Tata Realty and Infrastructure Limited, the holding company and its nominee 4,00,000 40.00 4,00,000 (e) Details of shareholding more than 5% in the Company

Name of Shareholder	31 Marci	n 2020	31 March 2019	
	No of Shares	% Holding	No of Shares	% Holding
Equity shares of INR 10 each, fully paid-up by Tata Realty and Infrastructure Limited, the holding company and its nominee	4.00.000	100%	4.00.000	100%

(f) Aggregate number of bonus shares issued and shares issued for consideration other than cash during the period of five years immediately preceding the reporting date :

During the five year period ended 31 March, 2020, the company has not issued any equity shares for consideration other than cash.

#### 5 Other Equity

Particulars	31 March 2020	31 March 2019
Retained earnings	(41.45)	(41.44)
TOTAL	(41.45)	(41.44)
Retained earnings		
Particulars	31 March 2020	31 March 2019
(Deficit) in statement of profit and loss		
Balance at the beginning of the year	(41.44)	(40.38)
Add: (Loss) for the year	(0.01)	(1.06)
Balance at the end of the Year	(41.45)	(41.44

& REU



40.00

Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) Notes to the Ind AS financial statements for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

## 6 Current Financial Liabilities - Borrowings

The borrowings are analysed as follows :		
Particulars	31 March 2020	31 March 2019
Interest-free loans from related parties		
recoverable on demand		
Tata Realty and Infrastructure Limited	6.50	6
TOTAL	6.50	6
The above amount includes		
Secured Borrowings		
Unsecured Borrowings	6.50	6

#### 7 Other Current Financial Liabilities

Particulars	31 March 2020	31 March 2019	1
Other Payables			
- Related Party		(0.03)	
- Others	0.59	0.62	
	0.59		0.59
Total	0.59		0.59

## 8 Other Income

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Miscellaneous Income	0.01	-
Total	0.01	

## 9 Other Expenses

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Fees & Consultations		0.08
Rates & Taxes ( incl indirect taxes)	0.02	0.06
Audit Fees (refer note 9 (a))		0.92
Total	0.02	1.06

## (a) Remuneration to Statutory Auditors

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Statutory audit	÷.	0.50
Other Services		0.33
Goods and Services Tax		0.09
Total	•	0.92

#### 10 Deferred Tax Assets (net)

<sup>(</sup>a) Amounts recognised in profit and loss

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Current income tax	91	14
Origination and reversal of temporary differences		
Reduction in tax rate	· · ·	
Recognition of previously unrecognised tax		64
Change in recognised deductible temporary differences		18
Total deferred tax expense/(benefit)		
Tax expense for the year	······································	
(b) Reconciliation of effective tax rate		
Loss before tax	(0.01)	(1.0
Tax using the Company's domestic tax rate 25% (p.y. 25%)	(0.00)	(0.2
Tax effect of:		
Current-year losses for which no deferred tax asset is recognised	0.00	0.2
Total	•	
REU		

## Wellkept Facility Management Services Private Limited

(formerly known as TRIL Hospitality Private Limited)

Notes to the Ind AS financial statements

for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

#### (c) Unrecognised deferred tax assets

Deferred tax assets have not been recognised in respect of the following items, because of no virtual certainty of the profit in the future year against which the deferred tax asset created can be utilised.

		Gross amount	Unrecognised tax effect	Gross amount	Unrecognised tax effect
Tax losses		8.61	2.15	7.55	1.89
(d) Tax losses carried	forward	and the second se	COULAS		0.166
Particulars	Financial Year	31 Mar	ch 2020	31 Marc	ch 2019
		Gross amount	Expiry date	Gross amount	Expiry date
Business loss	2011-12	0.73	2019-20	0.73	2019-20
Business loss	2012-13	0.63	2020-21	0.63	2020-21
Business loss	2013-14	1.23	2021-22	1.23	2021-22
Business loss	2014-15	0.99	2022-23	0.99	2022-23
Business loss	2015-16	2.54	2023-24	2.54	2023-24
Business loss	2016-17	0.88	2024-25	0.88	2024-25
Business loss	2017-18	0.55	2025-26	0.55	2025-26
Business loss	2018-19	1.06	2026-27		
Total		8.61		7.55	

The Company offsets tax assets and liabilities if and only if it has a legally enforceable right to set off current tax assets and current tax liabilities and the deferred tax assets and deferred tax liabilities relate to income taxes levied by the same tax authority.

Significant management judgement is required in determining provision for income tax, deferred income tax assets and liabilities and recoverability of deferred income tax assets. The recoverability of deferred income tax assets is based on estimates of taxable income and the period over which deferred income tax assets will be recovered. Any changes in future taxable income would impact the recoverability of deferred tax assets.

#### 11 Due to micro and small suppliers

Under the Micro, Small and Medium Enterprises Development Act, 2006, (MSMED) which came into force from 2 October 2006, certain disclosures are required to be made relating to Micro, Small and Medium enterprises. On the basis of the information and records available with the Management, there are no outstanding dues to the Micro, Small and Medium enterprises as defined in the Micro, Small and Medium Enterprises Development Act, 2006 as set out in following disclosure.

Particulars	31 March 2020	31 March 2019
Principal amount remaining unpaid to any supplier as at the year-end	(† 1	2
Interest due thereon	<b>.</b>	5
The amount of interest paid by the buyer as per the Micro Small and Medium Enterprises Development Act, 2006 (MSMED Act, 2006)	*	
The amount of interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under MSMED Act, 2006	8	×.
The amount of interest accrued and remaining unpaid at the end of each accounting year		8
The amount of further interest remaining due and payable even in the succeeding years, until such date when the interest dues as above are actually paid to the small enterprise for the purpose of disallowance as a deductible expenditure under the MSMED Act. 2006	<b>1</b> 1	12

#### 12 Earning Per Share

Earnings Per Share (EPS) = Net Profit attributable to Shareholders / Weighted Number of Shares Outstanding

Particulars		For the year ended 31 March 2020	For the year ended 31 March 2019
(Loss) after tax attributable to equity shareholders	A	(0.01)	(1.06)
Calculation of weighted average number of equity shares:			
Number of equity shares at the beginning of the year	C	4,00,000	4,00,000
Optionally convertible participatory debentures			-
Number of equity shares outstanding at the end of the year		4,00,000	4,00,000
Weighted average number of equity shares outstanding during the year	в	4,00,000	4,00,000
Earning Per Share - Basic (Rs.)	(A / B)	(0.00)	(0.27)
Earning Per Share – Diluted (Rs.)		(0.00)	(0.27

REU



Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited)		
Notes to the Ind AS financial statements		
for the year ended 31 March 2020		
(Currency: Indian rupees in lakhs)		
13 Disclosure of transactions with Related Parties, as required by India	n Accounting Standard (In	d AS) - 24
(a) Related Party Disclosure		
Ultimate Holding Company		
Tata Sons Private Limited		
Holding Company		
Tata Realty and Infrastructure Limited		
(b) Nature of Transactions / relationship / major parties	Holding Co	Total
Outstanding Balances Payable		
ICD Payable		
Tata Realty and Infrastructure Limited	6.50	6.50

(6.50) All the transactions with related parties are at arm's length except the interest free loan from Holding company and all the outstanding balances are unsecured. Note: figures in bracket represent previous years figures. KKH



#### Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited)

Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### 14 Segment Reporting as per IND AS108 " Operating Segments"

Operating segments are reported in a manner consistent with the internal reporting provided to the chief operating decision maker (CODM). Chief operating decision maker's function is to allocate the resources of the entity and access the performance of the operating segment of the entity. The Board of directors of the company assesses the financial performance and position of the Company and makes strategic decisions. It is identified as being the chief operating decision maker for the company. The Company is operating in the real estate and infrastructure industry and has only domestic operations. The Company has only one operating segment, which is development of real estate and infrastructure facilities. All assets of the Company are domiciled in India and the Company has no revenue.

#### 15 Other matters

Information with regard to other matters specified in Schedule III of the Act is either nil or not applicable to the Company for the year.

#### 16 Capital commitment, contingencies and other commitments

There are no capital commitments and contingent liabilities as at the balance sheet date.

Estimated amount of other commitments remaining to be executed on capital account and not provided for (net of advances) Rs Nil (2019: Rs Nil).

#### 17 Financial instruments - Fair values and risk management

#### A Credit risk

Credit risk is the risk of financial loss to the Company if a customer or counterparty to a financial instrument fails to meet its contractual obligations, and arises principally from the Company's receivables from customers and investment securities. The carrying amounts of financial assets represent the maximum credit exposure.

#### Cash and cash equivalents

The Company held cash and cash equivalents with credit worthy banks of INR 5.64 lakhs and INR 5.65 lakhs as at 31 March 2020 and 31 March 2019 respectively. The credit worthiness of such banks is evaluated by the management on an ongoing basis and is considered to be good.

B Liquidity risk

Liquidity risk is the risk that the Company will not be able to meet its financial obligations as they become due. The Company manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Company's reputation.

The Company has obtained a finance facility from a bank. The Company also constantly monitors funding options available in the debt and capital markets with a view to maintaining financial flexibility.

As at 31 March, 2020, the Company had working capital of Rs (1.45) lakhs including short term borrowings of Rs 6.50 lakhs. As at 31 March, 2019, the Company had working capital of Rs (1.44) lakhs including short-term borrowings of Rs 6.50 lakhs.

#### C Exposure to liquidity risk

The table below analyses the Company's financial liabilities into relevant maturity groupings based on their contractual maturities for:

- all non deri	vative final	ncial liabilities
----------------	--------------	-------------------

Particulars	Carrying amount	Total	1 year or less	1-2 years	2-5 years
As at 31 March 2020					
Non-derivative financial liabilities					
Unsecured short-term borrowings	6.50	6.50	6.50		
Other financial liabilities	0.59	0.59	0.59	-	
As at 31 March 2019	7.09	7.09	7.09		
Non-derivative financial liabilities					
Unsecured short-term borrowings	6.50	6.50	6.50		5
Other financial liabilities	0.59	0.59	0.59		1
	7.09	7.09	7.09	1	-

#### D Market risk

Market risk is the risk that changes in market prices – such as foreign exchange rates, interest rates and equity prices – will affect the Company's income or the value of its holdings of financial instruments. The Company is domiciled in India and has its revenues and other transactions in its functional currency i.e. INR. Accordingly the Company is not exposed to any currency risk. Also the Company does not hold any equity price risk.

#### E Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Company's exposure to market risk for changes in interest rates relates to borrowings from Holding Company.

#### Exposure to interest rate risk

31 March 2020	31 March 2019
	OT MATCH 2015
6.50	6.50

The Company does not account for any fixed-rate financial assets or financial liabilities at fair value through profit or loss, and the Company does not have any designate derivatives (interest rate swaps).

Therefore, a change in interest rates at the reporting date would not affect profit or loss.

### F Capital management

The Company's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. Management monitors the return on capital as well as the level of dividends to ordinary shareholders.

The Company has obtained borrowings primarily from its holding company.

The Company monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total borrowings, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity. Charge for the year on goodwill amortisation has been deducted while calculating total equity of the company since it represents a pure non-cash expense.

3M

REH

for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

The Company's adjusted net debt to equity ratio at March 31, 2020 was as follows.

Particulars	31 March 2020	31 March 2019
Total borrowings	6.50	6.50
Less : Cash and cash equivalent	5.64	5.65
Adjusted net debt	0.86	0.85
Adjusted equity	(1.45)	(1.44)
Adjusted net debt to adjusted equity ratio %	(0.59)	(0.59)

18 Financial instruments - Fair values and risk management

#### A Accounting classification and fair values

The following table shows the carrying amounts and fair values of financial assets and financial liabilities, including their levels in the fair value hierarchy. It does not include fair value information for financial assets and financial liabilities if the carrying amount is a reasonable approximation of fair value.

Particulars	FVTPL	FVTOCI	Amotised Cost	Total	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Tota
31 March 2020							123.70	
Financial assets								
Cash and cash equivalents	12	83	5.64	5.64	-			-
Financial liabilities								
Interest-free loans from			0.50	C 50				
related parties	•		6.50	6.50	•		•	
Other current financial			0.59	0.59		1		
liabilities			0.59	0.59		-		
31 March 2019								
Financial assets								
Cash and cash equivalents	20	*1	5.65	5.65	-	-	1200	
Financial liabilities								
Interest-free loans from	-		6.50	6.50				
related parties	<u>- 1</u>	5-	0.50	0.50	-	-	100	-
Other current financial	2	-	0.59	0.59				
liabilities	2.1		0.00	0.00		50		•

#### B Financial risk management

The Company has exposure to the following risks arising from financial instruments:

· Credit risk ;

- · Liquidity risk ; and
- · Market risk

#### C Risk management framework

The Company's board of directors has overall responsibility for the establishment and oversight of the Company's risk management framework.

#### D Measurement of fair values

#### Valuation techniques and significant unobservable inputs

The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for financial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used.

#### Financial instruments measured at fair value

Туре	Valuation technique	Significant	Inter-relationship
Amortised cost: 1. Borrowings	Discounted cash flow approach: The valuation model considers the present value of expected payments, discounted using a risk adjusted discount rate.		Not applicable

Transfers between Levels 1 and 2

There have been no transfers between Level 1 and Level 2 during the reporting periods

Level 3 fair values

There are no items in Level 3 fair values

19 There are no offsetting of financial assets and financial liabilities during the year.

#### 20 Subsequent events

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

REU

For and on behalf of the Board of Directors of Wellkept Facility Management Services Private Limited (formerly known as TRIL Hospitality Private Limited) CIN: U93000MH2008PTC177346

S.B.

Sunil Dhagat Director DIN: 03081163

Mumbai Dated : 17 June 2020 Amit Sheth Director DIN: 07997663

D

## INDEPENDENT AUDITOR'S REPORT

To The Members of Acme Living Solutions Private Limited Report on the Audit of the Ind AS Financial Statements

## Opinion

We have audited the accompanying Ind AS financial statements of Acme Living Solutions Private Limited ("the Company"), which comprise the Balance Sheet as at March 31, 2019, and the Statement of Profit and Loss (including Other Comprehensive Income), the Statement of Cash Flows and the Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended, ("Ind AS") and other accounting principles generally accepted in India, of the state of affairs of the Company as at March 31, 2019, and its loss, its cash flows and the changes in equity for the year ended on that date.

## **Basis for Opinion**

ASKINS

Chartered

SE

We conducted our audit of the Ind AS financial statements in accordance with the Standards on Auditing specified under section 143(10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Ind AS Financial Statements section of our report. We are independent of the Company in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the Ind AS financial statements.

## Material uncertainty related to Going Concern

We draw attention to Note 2(b) to the Ind As financial statements, which indicates that the Company has incurred a net loss of 1.04 lakhs during the year ended 31<sup>st</sup> March, 2019, the Company's current liabilities exceeded its current assets by 7.94 lakhs. The. These events or conditions, along with other matters as set forth in Note 2(b), indicate that a material uncertainty exists that may cast significant doubt on the Company's ability to continue as a going concern. However, the Ind As financial statements of the Company have been prepared on a going concern basis for the reasons stated in the said Note.

Our opinion is not modified in respect of this matter

## Information Other than the Financial Statements and Auditor's Report Thereon

The Company's Board of Directors is responsible for the other information. The other information comprises the Director's report which we obtained prior to the date of this auditor's report

• Our opinion on the Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.

In connection with our audit of the Ind AS financial statements, our responsibility is to read the other information and, in doing so, consider whether the other information is materially

Regul Office: Indiabulls Finance Centre, Tower 3, 27<sup>th</sup> - 32<sup>rd</sup> Floor, Senapati Bapat Marg, Elphinstone Road (West), Mumbai - 400 013, Maharashtra, India. (LLP Identification No. AAB-8737)

Chartered Accountants Lotus Corporate Park 1<sup>st</sup> Floor, Wing A – G CTS No.185/A, Jay Coach Off Western Express Highway Goregaon (East) Mumbai – 400 063 Maharashtra, India

Tel: +91 22 6245 1000 Fax: +91 22 6245 1001

Chartered Accountants inconsistent with the Ind AS financial statements or our knowledge obtained during the course of our audit or otherwise appears to be materially misstated.

If, based on the work we have performed, we conclude that there is a material misstatement
of this other information, we are required to report that fact. We have nothing to report in this
regard.

## Management's Responsibility for the Ind AS Financial Statements

The Company's Board of Directors is responsible for the matters stated in section 134(5) of the Act with respect to the preparation of these Ind AS financial statements that give a true and fair view of the financial position, financial performance including other comprehensive income, cash flows and changes in equity of the Company in accordance with the Ind AS and other accounting principles generally accepted in India.

This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the Ind AS financial statement that give a true and fair view and are free from material misstatement, whether due to fraud or error.

In preparing the Ind AS financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those Board of Directors are also responsible for overseeing the Company's financial reporting process.

## Auditor's Responsibility for the Audit of the Ind AS Financial Statements

Our objectives are to obtain reasonable assurance about whether the Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these Ind AS financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

 Identify and assess the risks of material misstatement of the Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.

• Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under section 143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Company has adequate internal financial controls system in place and the operating effectiveness of such controls.

- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to

events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.

• Evaluate the overall presentation, structure and content of the Ind AS financial statements, including the disclosures, and whether the Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Materiality is the magnitude of misstatements in the Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the Ind AS financial statements.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

## **Report on Other Legal and Regulatory Requirements**

- 1. As required by Section 143(3) of the Act, based on our audit we report, to the extent applicable that:
  - a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit.
  - b) In our opinion, proper books of account as required by law have been kept by the Company so far as it appears from our examination of those books
  - c) The Balance Sheet, the Statement of Profit and Loss including Other Comprehensive Income, the Statement of Cash Flows and Statement of Changes in Equity dealt with by this Report are in agreement with the relevant books of account.
  - d) In our opinion, the aforesaid Ind AS financial statements comply with the Ind AS specified under Section 133 of the Act.
  - e) The matter described in the Material uncertainty related to Going Concern section above, in our opinion, may have an adverse effect on the functioning of the Company.
  - f) On the basis of the written representations received from the directors as on March 31, 2019 taken on record by the Board of Directors, none of the directors is disqualified as on March 31, 2019 from being appointed as a director in terms of Section 164(2) of the Act.



- g) Reporting on the adequacy of the internal financial controls over financial reporting of the Company and the operating effectiveness of such controls is not applicable to the company in view of the exemption available to the company in terms of the notification no. G.S.R. 583(E) dated 13 June 2017 issued by the Ministry of Corporate Affairs, Government of India, read with general circular No. 08/2017 dated 25 July 2017.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of section 197(16) of the Act, as amended.

In our opinion and to the best of our information and according to the explanations given to us, the Company being a private company, section 197 of the Act related to managerial remuneration is not applicable.

- With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended in our opinion and to the best of our information and according to the explanations given to us:
  - i. The Company does not have any pending litigations which would impact its financial position.
  - ii. The Company did not have any long-term contracts including derivative contracts for which there were any material foreseeable losses.
  - iii. There were no amounts, which were required to be transferred to the Investor Education and Protection Fund by the Company
- 2. As required by the Companies (Auditor's Report) Order, 2016 ("the Order") issued by the Central Government in terms of Section 143(11) of the Act, we give in "Annexure A" a statement on the matters specified in paragraphs 3 and 4 of the Order.

## For Deloitte Haskins & Sells LLP

Chartered Accountants (Firm's Registration No.117366W/W-100018)

Aferichia

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai, April 5, 2019

4

Chartered Accountants

## ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT

(Referred to in paragraph 2 under 'Report on Other Legal and Regulatory Requirements' Section of our report of even date)

(i) The Company does not have any property, plant and equipment and hence reporting under clause of the CARO 2016 is not applicable.

(ii) The Company does not have any inventory and hence reporting under clause (ii) of the CARO 2016 is not applicable.

(iii) The Company has not granted any loans, secured or unsecured, to companies, firms, Limited Liability Partnerships or other parties covered in the register maintained under Section 189 of the Companies Act, 2013

(iv) The Company has not granted any loans, made investments or provided guarantees and hence reporting under clause (iv) of the CARO 2016 is not applicable.

(v) According to the information and explanations given to us, the Company has not accepted any deposit during the year as provided under Section 73 to 76 or any other relevant provisions of the Companies Act, 2013. There are no unclaimed deposits any time during the year.

(vi) The maintenance of cost records has not been specified by the Central Government under Section 148(1) of the Companies Act, 2013.

(vii) According to the information and explanations given to us, in respect of statutory dues:

- (a) The Company has been regular in depositing undisputed statutory dues, including Incometax, Service Tax / Goods & Services Tax, cess and other material statutory dues applicable to it to the appropriate authorities.
- (b) There were no undisputed amounts payable in respect of Income-tax, Service Tax / Goods & Services Tax, cess and other material statutory dues in arrears as at March 31, 2019 for a period of more than six months from the date they became payable.
- (c) There are no dues of Income-tax, Service Tax / Goods & Services Tax as on March 31, 2019 on account of disputes.

(viii) In our opinion and according to the information and explanations given to us, the Company has not defaulted in the repayment of loans or borrowings to debenture holders (Bank). The Company has not taken any loans from banks, financial institutions and Government.

(ix) The Company has not raised moneys by way of initial public offer or further public offer (including debt instruments) or term loans and hence reporting under clause (ix) of the CARO 2016 Order is not applicable.

(x) To the best of our knowledge and according to the information and explanations given to us, no fraud by the Company and no material fraud on the Company by its officers or employees has been noticed or reported during the year.

SKIN(XI) In our opinion and according to the information and explanations given to us, the Company is not required to have and does not have any managerial personnel and hence the reporting under

clause (xi) of the CARO 2016 Order for payment of managerial remuneration under Section 197 read with Schedule V of the Companies Act 2013 is not applicable.

(xii) The Company is not a Nidhi Company and hence reporting under clause (xii) of the CARO 2016 Order is not applicable.

(xiii) In our opinion and according to the information and explanations given to us the Company is in compliance with Section 177 and 188 of the Companies Act, 2013, where applicable, for all transactions with the related parties and the details of related party transactions have been disclosed in the financial statements etc. as required by the applicable accounting standards.

(xiv) During the year the Company has not made any preferential allotment or private placement of shares or fully or partly convertible debentures and hence reporting under clause (xiv) of CARO 2016 is not applicable to the Company.

(xv) In our opinion and according to the information and explanations given to us, during the year the Company has not entered into any non-cash transactions with its directors or directors of its holding, subsidiary or associate company or persons connected with them and hence provisions of Section 192 of the Companies Act, 2013 are not applicable.

(xvi) The Company is not required to be registered under Section 45-IA of the Reserve Bank of India Act, 1934.

For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No. 117366W/W-100018)

Fuchla

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai, April 5, 2019

## Acme Living Solutions Private Limited Balance Sheet as at 31st March 2019

Particulars	Note No.	As at 31 March 2019	(Rs. in lakhs As at 31 March 2018
ASSETS			
NON-CURRENT ASSETS			
(a) Deferred Tax Assets (net)	9		
TOTAL NON-CURRENT ASSETS			
CURRENT ASSETS			
(a) Financial assets			
(i) Cash and Cash Equivalents	3	0.19	0.47
TOTAL CURRENT ASSETS		0.19	0.47
TOTAL ASSETS		0.19	0.47
EQUITY AND LIABILITIES			
EQUITY			
(a) Equity share capital	4	5.00	5.00
(b) Other equity	5	(12.94)	(11.90)
TOTAL EQUITY		(7.94)	(6.90)
LIABILITIES			
CURRENT LIABILITIES			
(a) Financial liabilities			
(i) Borrowings	6	7.00	7.00
(ii) Other financial liabilities	7	1.13	0.37
TOTAL CURRENT LIABILITIES		8.13	7.37
TOTAL EQUITY AND LIABILITIES	-	0.19	0.47
Significant accounting policies	2		
Notes to the Ind AS financial statements The accompanying notes 1 to 19 form an integral part of the	3 - 19		

As per our report of even date

## For Deloitte Haskins & Sells LLP

Chartered Accountants Firm Registration No. 117366W/ W- 100018

web UH

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019

5.

For and on behalf of the **Board of Directors of** Acme Living Solutions Private Limited CIN: U45209DL2008PTC178023

R. B.

Sunil Dhagat Director DIN: 03081163

Vaidehi Modi Director DIN: 06967259

## Acme Living Solutions Private Limited Statement of Profit and Loss for the year ended 31st March 2019

Particulars		Note No.	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018	
I	Revenue from Operations :		-		
Ш	Other Income		-	-	
III	Total Income (I +II)		-	-	
IV	Expenses:				
	Other expenses	8	1.04	0.59	
	Total Expenses		1.04	0.59	
v	Loss before exceptional items and tax (III-IV)		(1.04)	(0.59)	
VI	Exceptional items Income / (Expense)		-	-	
VII	Loss before tax (V-VI)		(1.04)	(0.59)	
VIII	Tax expenses			9 <sup>1</sup> .	
	Current Tax			-	
	Excess / (Short) Provision of Earlier years		-	4.36	
	Deferred Tax			-	
	Total tax expenses		-	4.36	
IX	PROFIT/ (LOSS) FOR THE YEAR (VII-VIII)		(1.04)	3.77	
X	Other Comprehensive Income:				
	A. (i) Items that will not be reclassified to profit or loss			-	
	(ii) Income tax on above			-	
	B. (i) Items that will not be reclassified to profit or loss			-	
	(ii) Income tax on above		-	-	
XI	Total Comprehensive Income / (Loss) For The Year (IX+X)		(1.04)	3.77	
XII	Earnings per equity share	11			
	Basic		(2.08)	7.54	
	Diluted		(2.08)	7.54	
Signi	ficant accounting policies	2			

Notes to the Ind AS financial statements

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date

## For Deloitte Haskins & Sells LLP

Chartered Accountants Firm Registration No. 117366W/ W- 100018

Aguella

Kalpesh J. Mehta Partner

Mumbai Dated: 05 April 2019 For and on behalf of the Board of Directors of Acme Living Solutions Private Limited CIN: U45209DL2008PTC178023

3 - 19

Sunil Dhagat Director DIN: 03081163

Vaidehi Modi Director DIN: 06967259

## Acme Living Solutions Private Limited Statement of Cash Flows for the year ended 31st March 2019

		(Rs in lakhs)
Particulars	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018
A CASH FLOW FROM OPERATING ACTIVITIES Loss Before Tax Increase/ (Decrease) in other financial liabilities	( <b>1.04</b> )	
(Decrease) in other non-financial liabilities	- 0.76	(0.18) (0.05) (0.23)
CASH USED IN OPERATIONS Direct Taxes Paid	(0.28)	
Net Cash Flow used in Operating Activities	(0.28)	(0.82)
B CASH FLOW FROM INVESTMENT ACTIVITIES Net Cash used in Investment Activities		
C CASH FLOW FROM FINANCING ACTIVITIES Proceeds from Short term Borrowing		
Net Cash Flow from Financing Activities		1.00
NET INCREASE / (DECREASE) IN CASH AND CASH EQUIVALENTS	(0.28)	0.18
Opening Balance	0.47	0.29
Closing Balance	0.19	0.47
NET INCREASE / (DECREASE) IN CASH AND CASH EQUIVALENTS	(0.28)	0.18
Components of Cash and Cash Equivalents		
Cash on Hand	-	
Balances with Bank	0.19	0.47
Total Balance	0.19	0.47

## Note:

The Cash flow statement has been prepared under the indirect method as set out in Indian Accounting Standard - 7 ('Ind AS 7') on Cash Flow Statement prescribed in Companies (Indian Accounting Standard) Rules, 2015, notified under Section 133 of the Companies Act, 2013.

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date attached

## For Deloitte Haskins & Sells LLP

Chartered Accountants Firm Registration No. 117366W/ W- 100018

Spinchia

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019 For and on behalf of the **Board of Directors of** Acme Living Solutions Private Limited CIN: U45209DL2008PTC178023

S.B.

Sunil Dhagat Director DIN: 03081163 Vaidehi Modi Director DIN: 06967259

## Acme Living Solutions Private Limited

Statement of Changes in Equity for the year ended 31st March 2019 (Currency: Indian rupees in lakhs)

## Statement of Changes in Equity for the year ended 31st March 2019

## A Equity Share Capital

Particulars	For the Year March 2	For the Year Ended 31 March 2018		
	Number of Shares	Rs. in lakhs	Number of Shares	Rs. in lakhs
Subscribed and Fully Paid up Capital				
Equity shares of INR 10 each				
Opening Balance	50.000	5.00	50,000	5.00
Changes in equity share capital during the year		-	-	0.00
Closing Balance	50,000	5.00	50,000	5.00

## **B** Other Equity

Particulars	Retained Earnings	Other Compre hensive Income	Total
Balance as at 01 April 2017	(15.67)	-	(15.67)
Profit for the year	3.77		3.77
Balance as at 31 March 2018	(11.90)	•	(11.90)
Loss for the year	(1.04)	÷	(1.04)
Balance as at 31 March 2019	(12.94)		(12.94)

The accompanying notes 1 to 19 are an integral part of these Ind AS financial statements.

As per our report of even date

## For Deloitte Haskins & Sells LLP

Chartered Accountants Firm Registration No. 117366W/ W- 100018

Auch la

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019 For and on behalf of the Board of Directors of Acme Living Solutions Private Limited CIN: U45209DL2008PTC178023

Sunil Dhagat Director DIN: 03081163

Vaidehi Modi Director DIN: 06967259

#### Acme Living Solutions Private Limited Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

#### 1 General information

- a) Acme Living Solutions Private Limited ('the Company') was incorporated to carry on the business of development of real estate and infrastructure facilities. The Company is a wholly owned subsidiary of Tata Realty and Infrastructure Limited ('the holding company'), which is a wholly owned subsidiary of Tata Sons Limited, the ultimate holding Company. The company is a private limited company incorporated and domiciled in India and has its registered office at New Delhi, India.
- b) In order to achieve operating efficiency, the Board of Directors of the Company, in their meeting held on January 2, 2019, had approved a scheme of amalgamation between Acme Living Solutions Private Limited ("the Transferor Company") with its parent company viz. Tata Realty and Infrastructure Limited ("the Transferee Company") and their respective members and creditors w.e.f. April 1, 2019 ("Appointed date"). Consequently, on March 29, 2019, the companies have filed the said scheme of amalgamation with National Company Law Tribunal, Delhi.

## 2 SIGNIFICANT ACCOUNTING POLICIES

### i) Basis of Preparation

#### a) Statement of compliance:

The financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 (the Act) and other relevant provisions of the Act.

The financial statements were authorized for issue by the Company's Board of Directors on 5th April 2019

#### b) Going concern :

As at 31 March 2019, the Company's paid up capital and reserves (excluding the deficit in the profit and loss) were INR 5 lakhs (2018: INR 5 Lakhs) and correspondingly, the Company's accumulated losses aggregated INR 12.94 lakhs (2018: INR 11.90 lakhs) which leads to material uncertainty related to Going Concern. However, in pursuance to Note No. 1 (b) above, the financial statements have been prepared on a going concern basis. Accordingly, these financial statements do not include any adjustments relating to the recoverability and classification of recorded labilities that might be necessary if the Company is unable to continue as a going concern.

#### c) Historical cost convention:

The financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

1 Financial instruments measured at fair value through profit or loss, if applicable

2 Financial instruments measured at fair value through other comprehensive income, if applicable

#### d) Key estimates and assumptions

In preparing these financial statements, management has made judgments, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates.

Estimates and assumptions are required in particular for:

Measurement of fair values

The Company's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Company has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the group CFO.

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorised into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

Level 1: quoted prices (unadjusted) in active markets for identical assets or liabilities.

• Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

· Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorised in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement

#### ii) Functional and presentation currency

The financial statements are presented in Indian rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian rupees (INR) and all amounts have been rounded-off to the nearest lakhs, except for share data and as otherwise stated. Due to rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.

#### iii) Income tax

Chartered Accountants

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised



(Currency: Indian rupees in lakhs)

## a) Current tax

Current tax comprises the expected tax payable or receivable on the taxable income or loss for the year and any adjustment to the tax payable or receivable in respect of previous years. It is measured using tax rates enacted or substantively enacted at the reporting date. Current tax also includes any tax arising from dividends. Management periodically evaluates positions taken in tax returns with respect to situations in which applicable tax regulation is subject to interpretation. It establishes provisions where appropriate on the basis of amounts expected to be paid to the tax authorities.

Minimum Alternate Tax ('MAT') under the provisions of Income-tax Act, 1961 is recognised as current tax in the statement of profit and loss. MAT paid in accordance with the tax laws, which gives future economic benefits in the form of adjustment to future income tax liability, is considered as an asset if there is a convincing evidence that the Company will pay normal tax. Accordingly, MAT is recognised as an asset in the balance sheet when it is probable that the future economic benefit associated with it will flow to the Company.

Current tax assets and liabilities are offset only if, the Company:

- 1 has a legally enforceable right to set off the recognised amounts; and
- 2 intends either to settle on a net basis, or to realise the asset and settle the liability simultaneously.

#### b) Deferred tax

Deferred tax is recognised in respect of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for taxation purposes. However, deferred tax liabilities are not recognised if they arise from the initial recognition of goodwill. Deferred income tax is also not accounted for if it arises from initial recognition of an asset or liability in a transaction other than a business combination that at the time of the transaction affects neither accounting profit nor taxable profit (tax loss).

Deferred tax assets are recognised for unused tax losses, unused tax credits and deductible temporary differences to the extent that it is probable that future taxable profits will be available against which they can be used. Deferred tax assets are reviewed at each reporting date and are reduced to the extent that it is no longer probable that the related tax benefit will be realised; such reductions are reversed when the probability of future taxable profits improves. Unrecognised deferred tax assets are reassessed at each reporting date and recognised to the extent that it has become probable that future taxable profits will be available against which they can be used.

Deferred tax is measured at the tax rates that are expected to be applied to temporary differences when they reverse, using tax rates enacted or substantively enacted at the reporting date and are expected to apply when the related deferred income tax asset is realised or the deferred income tax liability is settled.

Deferred tax assets and liabilities are offset only if:

1 the entity has a legally enforceable right to set off current tax assets against current tax liabilities; and

2 the deferred tax assets and the deferred tax liabilities relate to income taxes levied by the same taxation authority on the same taxable entity.

#### iv) Cash and cash equivalents

For the purpose of presentation in the statement of cash flows, cash and cash equivalents includes cash on hand, deposits held at call with financial institutions, other short-term, highly liquid investments with original maturities of three months or less that are readily convertible to known amounts of cash and which are subject to an insignificant risk of changes in value, and bank overdrafts. Bank overdrafts are shown within borrowings in current liabilities in the balance sheet.

#### v) Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity. Financial instruments also include derivative contracts such as foreign currency foreign exchange forward contracts, interest rate swaps and currency options; and embedded derivatives in the host contract.

### **Financial liabilities**

#### Classification

The Company classifies all financial liabilities as measured at amortised cost, except for financial liabilities at fair value through profit or loss. Such liabilities, including derivatives that are liabilities, shall be subsequently measured at fair value.

#### Initial recognition and measurement

Financial liabilities are classified, at initial recognition, as financial liabilities at fair value through profit or loss, loans and borrowings, payables, or as derivatives designated as hedging instruments in an effective hedge, as appropriate.

All financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

The Company's financial liabilities include trade and other payables, loans and borrowings including bank overdrafts, financial guarantee contracts and derivative financial instruments.

#### Trade and other payables

Trade and other payables represent liabilities for goods and services provided to the Company prior to the end of the financial year which are unpaid. The amounts are unsecured and are usually paid within 30-60 days of recognition or maybe payable on demand. Trade and other payables are represented as current liabilities unless payment is not due within 12 months after the reporting period. They are recognised initially at their fair value and subsequently recognised at amortised cost using the effective interest method.

#### Loans and borrowings

2

After initial recognition, interest-bearing loans and borrowings are subsequently measured at amortised cost using the EIR method. Gains and losses are recognised in profit or loss when the liabilities are derecognized.

Amortised cost is calculated by taking into account any discount or premium on acquisition and fees or costs that are an integral part of the EIR. The EIR amortisation is included as finance costs in the statement of profit and loss.

This category generally applies to interest-bearing loans and borrowings.

Preference shares, which are mandatorily redeemable on a specific date, are classified as liabilities. The dividends on these preference shares are recognised in profit or loss as finance costs.

The fair value of the liability portion of an optionally convertible bonds is determined using a market interest rate for an equivalent non-convertible bonds. This amount is recorded as a liability on an amortised cost basis until extinguished on conversion or redemption of the bonds. The remainder of the proceeds is attributable to the equity portion of the compound instrument. This is recognised and included in shareholders' equity net of income tax effects, and not subsequently remeasured.

Solur

Charter where the terms of a financial liability are renegotiated and the entity issues equity instruments to a creditor to extinguish all or part of the liability Accountance of the equity swap), a gain or loss is recognised in profit or loss, which is measured as the difference between the carrying amount of the financial liability and the fair value of the equity instruments issued.

## Acme Living Solutions Private Limited Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

#### Derecognition

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit or loss.

## Offsetting of financial instruments

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable right to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable right must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the company or the counterparty.

#### vi) Provisions

A provision is recognised in the balance sheet when the Company has a legal or constructive obligation as a result of a past event, and it is probable that an outflow of economic benefits will be required to settle the obligation. If the effect is material, provisions are determined by discounting the expected future cash flows at a pre-tax rate that reflects current market assessments of the time value of money and, where appropriate, the risks specific to the liability.

## vii) Earning Per Share

Basic earnings per share is computed by dividing the profit / (loss) after tax by the weighted average number of equity shares outstanding during the year. The weighted average number of equity shares outstanding during the year is adjusted for the events for bonus issue, bonus element in a rights issue to existing shareholders, share split and reverse share split (consolidation of shares).

Diluted earnings per share is computed by dividing the profit / (loss) after tax as adjusted for dividend, interest and other charges to expense or income (net of any attributable taxes) relating to the dilutive potential equity shares, by the weighted average number of equity shares considered for deriving basic earnings per share and the weighted average number of equity shares which could have been issued on conversion of all dilutive potential equity shares.





#### 3 Cash and Cash Equivalents

Particulars	As at 31 March 2019	As at 31 March 2018
Cash and cash equivalents Balances with Bank		
	0.19	0.47
Total	0.19	0.47

## 4 Equity Share Capital

(a) Authorised, Issued, Subscribed and Fully Paid up :

Particulars	As at 31 March 2019		As at 31 March 2018	
	No of Shares	Amount	No of Shares	Amount
Authorised Capital :				
Equity Shares of Rs.10/- each	50,000	5.00	50,000	5.00
Issued, Subscribed and Fully Paid up Capital : Issued Capital				
Equity Shares of Rs.10/- each, fully paid	50,000	5.00	50,000	5.00
Subscribed and Fully Paid up Capital				
Equity Shares of Rs.10/- each, fully paid	50,000	5.00	50,000	5.00
Total		5.00		5.00

#### (b) Reconciliation of Number of Shares Outstanding

Particulars	As at 31 March 2019		As at 31 March 2018	
	No of Shares	Amount	No of Shares	Amount
As at the beginning of the year	50,000	5.00	50,000	5.00
Add: Issued during the year		-	-	2
As at the end of the year	50,000	5.00	50,000	5.00

## (c) Terms and rights attached to the equity share

The Company has only one class of equity shares having par value of INR 10 per share. Each holder of equity shares is entitled to one vote per share. The Company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject the approval of shareholders in the ensuing Annual General Meeting.

During the year ended 31 March 2019, the amount of per share dividend recognised as distribution to equity shareholders is Rs Nil (2018: Rs Nil).

In the event of liquidation of the Company, the holders of the equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the share holders.

## (d) Shares of the company held by the Holding company

	Name of Shareholder	As at 31 Ma	rch 2019	As at 31 Ma	rch 2018
		No of Shares	Amount	No of Shares	Amount
	Equity shares of INR 10 each, fully paid-up by Tata Realty and Infrastructure Limited, the holding company and its nominee	50,000	5.00	50,000	5.00
e) _	Details of shareholding more than 5% in the Com Name of Shareholder	pany As at 31 Ma	rch 2019	As at 31 Ma	arch 2018
e) _			rch 2019 % Holding	As at 31 Ma No of Shares	rch 2018 % Holding

(f) Aggregate number of bonus shares issued and shares issued for consideration other than cash during the period of five years immediately preeceding the reporting date :

During the five year period ended 31 March, 2019, the company has not issued any equity shares for consideration other than cash.

### 5 Other Equity

Chartered Accountants

Particulars	As at 31 March 2019	As at 31 March 2018	
Retained earnings	(12.94)	(11.90)	
TOTAL	(12.94)	(11.90)	

Particulars	As at 31 March 2019	As at 31 March 2018
(Deficit) in statement of profit and loss		
Balance at the beginning of the year	(11.90)	(15.67)
Add: Profit / (Loss) for the year	(1.04)	3.77
Balance at the end of the Year	(12.94)	(11.90)



# Acme Living Solutions Private Limited Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

Particulars	As at 31 March 2019	As at 31 March 2018
Interest-free loans from related parties recoverable on demand		As at of March 2010
Tata Realty and Infrastructure Limited	7.00	7.0
TOTAL	7.00	7.0
The above amount includes		
Secured Borrowings		
Unsecured Borrowings	7.00	7.0

## 7 Other Current Financial Liabilities

Particulars	As at 31 March 2019	As at 31 March 2018
Other Payables		As at of march 2010
- Related Party		
- Others	1.13	0.37
	1	.13 0.37
Total	1	.13 0.37

## 8 Other Expenses

Particulars	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018
Fees & Consultations	0.11	0.27
Rates & Taxes ( incl indirect taxes)	0.01	0.03
Bank Charges & Guarantee Commission	0.01	0.02
Audit Fees (refer note 10 (a))	0.91	0.27
Total	1.04	0.59

## (a) Remuneration to Statutory Auditors

Particulars	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018
Statutory audit	0.50	0.23
Other Services	0.32	0.20
Service tax / GST	0.09	0.04
Total	0.91	0.27

## 9 Deferred Tax Assets

(a) Amounts recognised in profit and loss

Particulars	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018
Current income tax		5 <u>4</u> 5
Origination and reversal of temporary differences		2 <b>4</b> 2
Reduction in tax rate		
Recognition of previously unrecognised tax losses		
Change in recognised deductible temporary differences		
Total deferred tax expense/(benefit)		
Tax expense for the year		-
(b) Reconciliation of effective tax rate		
Loss before tax	(1.04)	(0.59
Tax using the Company's domestic tax rate 25% ( 2018 25%)	(0.26)	(0.15
Reduction in tax rate		
Tax effect of:		
Current-year losses for which no deferred tax asset is recognised	0.26	0.15
Total		-




# Acme Living Solutions Private Limited Notes to the Ind AS financial statements (Currency: Indian rupees in lakhs)

## (c) Unrecognised deferred tax assets

Deferred tax assets have not been recognised in respect of the following items, because of no virtual certainty of the profit in the future year against which the deferred tax asset created can be utilised.

		Gross amount	Unrecognised tax effect	Gross amount	Unrecognised tax effect
Tax losses		34.93	8.73	34.88	8.72
(d) Tax losses carried	forward				
Particulars	Financial Year	As at 31 M	Aarch 2019	As at 31 M	larch 2018
		Gross amount	Expiry date	Gross amount	Expiry date
Business loss	2010-11		2018-19	0.54	2018-19
Business loss	2011-12	32.19	2019-20	32,19	2019-20

Total		34.93		34.88	
Business loss	2017-18	- 0.58	2025-26		
Business loss	2016-17	0.67	2024-25	0.67	2024-25
Business loss	2015-16	0.79	2023-24	0.79	2023-24
Business loss	2014-15	0.37	2022-23	0.37	2022-23
Business loss	2013-14	0.31	2021-22	0.31	2021-22
Business loss	2012-13	0.01	2020-21	0.01	2020-21
Business loss	2011-12	32.19	2019-20	32.19	2019-20
			2010 10	0.04	2010-13

The Company offsets tax assets and liabilities if and only if it has a legally enforceable right to set off current tax assets and current tax liabilities and the deferred tax assets and deferred tax liabilities relate to income taxes levied by the same tax authority.

Significant management judgement is required in determining provision for income tax, deferred income tax assets and liabilities and recoverability of deferred income tax assets. The recoverability of deferred income tax assets is based on estimates of taxable income and the period over which deferred income tax assets will be recovered. Any changes in future taxable income would impact the recoverability of deferred tax assets.

## 10 Due to micro and small suppliers

Under the Micro, Small and Medium Enterprises Development Act, 2006, (MSMED) which came into force from 2 October 2006, certain disclosures are required to be made relating to Micro, Small and Medium enterprises. On the basis of the information and records available with the Management, there are no outstanding dues to the Micro, Small and Medium enterprises as defined in the Micro, Small and Medium Enterprises Development Act, 2006 as set out in following disclosure.

Particulars	As at 31 March 2019	As at 31 March 2018	
Principal amount remaining unpaid to any supplier as at the year-end			
Interest due thereon The amount of interest paid by the buyer as per the Micro Small and Medium Enterprises Development Act, 2006 (MSMED Act, 2006)			
The amount of interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under MSMED Act, 2006			
The amount of interest accrued and remaining unpaid at the end of each accounting year			
The amount of further interest remaining due and payable even in the succeeding years, until such date when the interest dues as above are actually paid to the small enterprise for the purpose of disallowance as a deductible expenditure under the MSMED Act, 2006			

#### 11 Earning Per Share

Earnings Per Share (EPS) = Net Profit attributable to Shareholders / Weighted Number of Shares Outstanding

Particulars	For the Year Ended 31 March 2019		For the Year Ended 31 March 2018
(Loss) after tax attributable to equity shareholders	A	(1.04)	3.77
Calculation of weighted average number of equity shares:			
Number of equity shares at the beginning of the year	с	50,000	50,000
Number of equity shares outstanding at the end of the year		50,000	50,000
Weighted average number of equity shares outstanding during the year	В	50,000	50,000
Earning Per Share – Basic (Rs.)	(A / B)	(2.08)	7.54
Earning Per Share – Diluted (Rs.)		(2.08)	7.54





# 12 Disclosure of transactions with Related Parties, as required by Indian Accounting Standard (Ind AS) - 24

(a) Related Party Disclosure <u>Ultimate Holding Company</u> Tata Sons Limited <u>Holding Company</u> Tata Recite and Lefonders Line

Tata Realty and Infrastructure Limited

(b) Nature of Transactions / relationship / major parties	Ultimate holding Company	Holding Co	Total
Finance provided for Loans, expenses & on a/c payments Tata Realty and Infrastructure Limited			
and must detaile climited	1.	1.00	1.00
Outstanding Balances Payable	(-)	(1.00)	(1.00)
Tata Realty and Infrastructure Limited	-	7.00	7.00
If the transactions with related parties are at arm's length and a	(-)	(7.00)	(7.00)

company.

Note: figures in bracket represent previous years figures.

# 13 Segment Reporting as per IND AS108 " Operating Segments"

Operating segments are reported in a manner consistent with the internal reporting provided to the chief operating decision maker (CODM). Chief operating decision maker's function is to allocate the resources of the entity and access the performance of the operating segment of the entity. The Board assesses the financial performance and position of the Company and makes strategic decisions. It is identified as being the chief operating decision maker for the company. The Company is operating in the real estate and infrastructure industry and has only domestic operations. The Company has only one operating segment, which is development of real estate and infrastructure facilities. All assets of the Company are domiciled in India and the Company has no revenue.

## 14 Other matters

Information with regard to other matters specified in Schedule III of the Act is either nil or not applicable to the Company for the year.

# 15 Capital commitment, contingencies and other commitments

There are no capital commitments and contingent liabilities as at the balance sheet date.

Estimated amount of other commitments remaining to be executed on capital account and not provided for (net of advances) Rs Nil (2018: Rs Nil).

# 16 Financial instruments - Fair values and risk management

#### A Credit risk

Credit risk is the risk of financial loss to the Company if a customer or counterparty to a financial instrument fails to meet its contractual obligations, and arises principally from the Company's receivables from customers and investment securities. The carrying amounts of financial assets represent the maximum credit exposure.

# Cash and cash equivalents

The Company held cash and cash equivalents with credit worthy banks and financial institutions of INR 0.19 lakhs and INR 0.47 lakhs as at 31 March 2019 and 31 March 2018 respectively. The credit worthiness of such banks and financial institutions is evaluated by the management on an ongoing basis and is considered to be good.

#### B Liquidity risk

Liquidity risk is the risk that the Company will not be able to meet its financial obligations as they become due. The Company manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Company's reputation.

The Company has obtained a finance facility from a bank. The Company also constantly monitors funding options available in the debt and capital markets with a view to maintaining financial flexibility.

As at 31 March, 2019, the Company had working capital of Rs (7.94) lakhs inculding short term borrowings of Rs 7 Lakhs. As at 31 March, 2018, the Company had working capital of Rs (6.90) lakhs inculding short-term borrowings of Rs 7 Lakhs.

# C Exposure to liquidity risk

The table below analyses the Company's financial liabilities into relevant maturity groupings based on their contractual maturities for:

# - all non derivative financial liabilities

Particulars	Carrying amount	Total	1 year or less	1-2 years	2-5 years
As at 31 March 2019					
Non-derivative financial liabilities					
Unsecured short-term borrowings	7.00	7.00	7.00		
Other financial liabilities	1.13	1.13	1.13		
	8.13	8.13	8.13		
As at 31 March 2018					
Non-derivative financial liabilities					
Unsecured short-term borrowings	7.00	7.00	7.00		
Other financial liabilities	0.37	0.37	0.37		
	7.37	7.37	7.37		

#### Market risk

(c)

\* 0

S Market nek is the risk that changes in market prices – such as foreign exchange rates, interest rates and equity prices – will affect the Company's income or the value on its holdings of financial instruments. The Company is domiciled in India and has its revenues and other transactions in its functional currency i.e. INR. Accordingly the Company is not exposed to any currency risk. Also the Company does not hold any equity investments, accordingly the Company is not charposed to any equity price risk.

# Acme Living Solutions Private Limited

Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

#### F Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Company's exposure to market risk for changes in interest rates relates to borrowings from Holding Company.

## Exposure to interest rate risk

The interest rate profile of the company's interest-bear Fixed-rate instruments		
	As at 31 March 2019	As at 31 March 2018
inancial liabilities	West and the gradient and the second s	the second second second
CD from Tata Realty and Infrastructure Limited		
tere receive and impast doture Limited	7.00	

The Company does not account for any fixed-rate financial assets or financial liabilities at fair value through profit or loss, and the Company does not have any designate derivatives (interest rate swaps).

Therefore, a change in interest rates at the reporting date would not affect profit or loss.

#### Capital management F

The Company's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. Management monitors the return on capital as well as the level of dividends to ordinary shareholders.

The Company has obtained borrowings primarily from its holding company.

The Company monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total borrowings, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity. Charge for the year on goodwill amortisation has been deducted while calculating total equity of the company since it represents a pure non-cash expense.

The Company's adjusted net debt to equity ratio at 31 March 2019 was as follows.

Paticulars	As at 31 March 2019	As at 31 March 2018
Total borrowings	7.00	7.00
Less : Cash and cash equivalent Adjusted net debt	0.19	0.47
Adjusted equity	6.81	6.53
Adjusted net debt to adjusted equity ratio (%)	(7.94)	(6.90)
regulated her debr to adjusted equity ratio (%)	(0.86)	(0.95)

#### 17 Financial instruments – Fair values and risk management Α

# Accounting classification and fair values

The following table shows the carrying amounts and fair values of financial assets and financial liabilities, including their levels in the fair value hierarchy. It does not include fair value information for financial assets and financial liabilities if the carrying amount is a reasonable approximation of fair value.

Particulars	FVTPL	FVTOCI	Amotised Cost	Total	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Total
31 March 2019								
Financial assets								
Cash and cash equivalents			0.19	0.19				
Financial liabilities			0.15	0.19	-	-	8 <del></del> 65	-
Interest-free loans from								
related partie	-	-	7.00	7.00	-	-	-	-
Other current financial								
liabilities	-	Q23	1.13	1.13		-	-	
31 March 2018								
Financial assets								
Cash and cash equivalents	2 <u>-</u>	-	0.47	0.47				
Financial liabilities							-	
Interest-free loans from				22070-01				
related parties		1	7.00	7.00	-	-	2	•
Other current financial			0.27	0.07				
liabilities		-	0.37	0.37	-	-	-	

Financial risk management в

The Company has exposure to the following risks arising from financial instruments:

· Credit risk

· Liquidity risk ; and

Market risk

#### C Risk management framework

The Company's board of directors has overall responsibility for the establishment and oversight of the Company's risk management framework.

Sol



(Currency: Indian rupees in lakhs)

#### Measurement of fair values D

# Valuation techniques and significant unobservable inputs

The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for financial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used.

# Financial instruments measured at fair value

Туре	Valuation technique	Significant	Inter-relationship between
Amortised cost: 1. Borrowings	Discounted cash flow approach: The valuation model considers the present value of expected payments, discounted using a risk adjusted discount rate.		Not applicable

#### Transfers between Levels 1 and 2

There have been no transfers between Level 1 and Level 2 during the reporting periods

Level 3 fair values

There are no items in Level 3 fair values

There are no offsetting of financial assets and financial liabilities during the year except provision for tax and advance tax. 18

# 19 Subsequent events

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

For and on behalf of the Board of Directors of Acme Living Solutions Private Limited CIN: U45209DL2008PTC178023 S.B

Sunil Dhagat Director

Vaidehi Modi Director DIN: 06967259



DIN: 03081163 Mumbai Dated: 05 April 2019

## **Chartered Accountants**

Indiabulls Finance Centre Tower 3, 27th - 32nd Floor Senapati Bapat Marg Elphinstone Road (West) Mumbai - 400 013 Maharashtra, India Tel: +91 22 6185 4000 Fax: +91 22 6185 4001

# **INDEPENDENT AUDITOR'S REPORT**

# To The Members of Acme Living Solutions Private Limited **Report on the Audit of the Ind AS Financial Statements**

# Opinion

We have audited the accompanying Ind AS financial statements of Acme Living Solutions Private Limited ("the Company"), which comprise the Balance Sheet as at 31 March 2020, and the Statement of Profit and Loss (including Other Comprehensive Income), the Statement of Cash Flows and the Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under Section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended, ("Ind AS") and other accounting principles generally accepted in India, of the state of affairs of the Company as at 31 March 2020, and its loss, total comprehensive loss, its cash flows and the changes in equity for the year ended on that date.

# **Basis for Opinion**

We conducted our audit of the Ind AS financial statements in accordance with the Standards on Auditing specified under Section 143(10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Ind AS Financial Statements section of our report. We are independent of the Company in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the Ind AS financial statements.

# Material uncertainty related to Going Concern

We draw attention to Note 2(b) to the Ind AS financial statements, wherein the events or conditions, along with other matters as set forth in the said note, indicate that a material uncertainty exists that may cast significant doubt on the Company's ability to continue as a going concern. However, the Ind AS financial statements of the Company have been prepared on a going concern basis for the reasons stated in the said Note.

Our opinion is not modified in respect of this matter

# Information Other than the Financial Statements and Auditor's Report Thereon



The Company's Board of Directors is responsible for the other information. The other information comprises the Director's report which we obtained prior to the date of this auditor's report.

- Our opinion on the Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.
- In connection with our audit of the Ind AS financial statements, our responsibility is to read the other information and, in doing so, consider whether the other information is materially inconsistent with the Ind AS financial statements or our knowledge obtained during the course of our audit or otherwise appears to be materially misstated.
- If, based on the work we have performed, we conclude that there is a material misstatement of this other information, we are required to report that fact. We have nothing to report in this regard.

# Management's Responsibility for the Ind AS Financial Statements

The Company's Board of Directors is responsible for the matters stated in Section 134(5) of the Act with respect to the preparation of these Ind AS financial statements that give a true and fair view of the financial position, financial performance including other comprehensive income, cash flows and changes in equity of the Company in accordance with the Ind AS and other accounting principles generally accepted in India.

This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the Ind AS financial statements that give a true and fair view and are free from material misstatement, whether due to fraud or error.

In preparing the Ind AS financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those Board of Directors are also responsible for overseeing the Company's financial reporting process.

# Auditor's Responsibility for the Audit of the Ind AS Financial Statements

Our objectives are to obtain reasonable assurance about whether the Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these Ind AS financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

 Identify and assess the risks of material misstatement of the Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.

RKU

- Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under Section143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Company has adequate internal financial controls system in place and the operating effectiveness of such controls.
- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.
- Evaluate the overall presentation, structure and content of the Ind AS financial statements, including the disclosures, and whether the Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Materiality is the magnitude of misstatements in the Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the Ind AS financial statements.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

# **Report on Other Legal and Regulatory Requirements**

- 1. As required by Section 143(3) of the Act, based on our audit we report, to the extent applicable that:
  - a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit.
  - b) In our opinion, proper books of account as required by law have been kept by the Company so far as it appears from our examination of those books
  - c) The Balance Sheet, the Statement of Profit and Loss including Other Comprehensive Income, the Statement of Cash Flows and Statement of Changes in Equity dealt with by this Report are in agreement with the relevant books of account.
  - d) In our opinion, the aforesaid Ind AS financial statements comply with the Ind AS specified under Section 133 of the Act.

e) The matter described in the Material uncertainty related to Going Concern section above, in our opinion, may have an adverse effect on the functioning of the Company.

CV OUT Opinien,

- f) On the basis of the written representations received from the directors as on 31 March 2020 taken on record by the Board of Directors, none of the directors is disqualified as on 31 March 2020 from being appointed as a director in terms of Section 164(2) of the Act.
- g) Reporting on the adequacy of the internal financial controls over financial reporting of the Company and the operating effectiveness of such controls is not applicable to the company in view of the exemption available to the company in terms of the notification no. G.S.R. 583(E) dated 13 June 2017 issued by the Ministry of Corporate Affairs, Government of India, read with general circular No. 08/2017 dated 25 July 2017.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of Section 197(16) of the Act, as amended.

In our opinion and to the best of our information and according to the explanations given to us, the Company being a private company, Section 197 of the Act related to managerial remuneration is not applicable.

- With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended in our opinion and to the best of our information and according to the explanations given to us:
  - i. The Company does not have any pending litigations which would impact its financial position.
  - ii. The Company did not have any long-term contracts including derivative contracts for which there were any material foreseeable losses.
  - iii. There were no amounts, which were required to be transferred to the Investor Education and Protection Fund by the Company
- 2. As required by the Companies (Auditor's Report) Order, 2016 ("the Order") issued by the Central Government in terms of Section 143(11) of the Act, we give in "Annexure A" a statement on the matters specified in paragraphs 3 and 4 of the Order.

For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No.117366W/W-100018)

pro.

Rajesh K. Hiranandani Partner (Membership No. 36920) UDIN: 20036920AAAABI5387

Mumbai, 17 June 2020

# ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT

(Referred to in paragraph 2 under 'Report on Other Legal and Regulatory Requirements' Section of our report of even date)

- (i) The Company does not have any property plant and equipment hence reporting under clause (i) of the CARO 2016 is not applicable.
- (ii) The Company does not have any inventory and hence reporting under clause (ii) of the CARO 2016 is not applicable.
- (iii) The Company has not granted any loans, secured or unsecured, to companies, firms, Limited Liability Partnerships or other parties covered in the register maintained under Section 189 of the Companies Act, 2013
- (iv) The Company has not granted any loans, made investments or provided guarantees and hence reporting under clause (iv) of the CARO 2016 is not applicable.
- (v) According to the information and explanations given to us, the Company has not accepted any deposit during the year as provided under Section 73 to 76 or any other relevant provisions of the Companies Act, 2013. There are no unclaimed deposits any time during the year.
- (vi) The maintenance of cost records has not been specified by the Central Government under Section 148(1) of the Companies Act, 2013.
- (vii) According to the information and explanations given to us, in respect of statutory dues:
  - (a) The Company has been regular in depositing undisputed statutory dues, including Incometax, Goods & Services Tax, cess and other material statutory dues applicable to it to the appropriate authorities.
  - (b) There were no undisputed amounts payable in respect of Income-tax, Goods & Services Tax, cess and other material statutory dues in arrears as at 31 March 2020 for a period of more than six months from the date they became payable.
  - (c) There are no dues of Income-tax, Goods & Services Tax as on 31 March 2020 on account of disputes.
- (viii) In our opinion and according to the information and explanations given to us, The Company has not taken any loans or borrowings from financial institutions, banks and government or has not issued any debentures. Hence reporting under clause (viii) of CARO 2016 is not applicable to the Company.
- (ix) The Company has not raised moneys by way of initial public offer or further public offer (including debt instruments) or term loans and hence reporting under clause (ix) of the CARO 2016 Order is not applicable.
- (x) To the best of our knowledge and according to the information and explanations given to us, no fraud by the Company and no material fraud on the Company by its officers or employees has been noticed or reported during the year.
- (xi) In our opinion and according to the information and explanations given to us, the Company is not required to have and does not have any managerial personnel and hence the reporting under clause of the CARO 2016 Order for payment of managerial remuneration under Section 197 read with Schedule V of the Companies Act 2013 is not applicable.

RKH

- (xii) The Company is not a Nidhi Company and hence reporting under clause (xii) of the CARO 2016 Order is not applicable.
- (xiii) In our opinion and according to the information and explanations given to us the Company is in compliance with Section 177 and 188 of the Companies Act, 2013, where applicable, for all transactions with the related parties and the details of related party transactions have been disclosed in the financial statements etc. as required by the applicable accounting standards.
- (xiv) During the year the Company has not made any preferential allotment or private placement of shares or fully or partly convertible debentures and hence reporting under clause (xiv) of CARO 2016 is not applicable to the Company.
- (xv) In our opinion and according to the information and explanations given to us, during the year the Company has not entered into any non-cash transactions with its directors or directors of its holding, subsidiary or associate company or persons connected with them and hence provisions of Section 192 of the Companies Act, 2013 are not applicable.
- (xvi)The Company is not required to be registered under Section 45-IA of the Reserve Bank of India Act, 1934.

# For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No. 117366W/W-100018)

ptkine.

Rajesh K. Hiranandani Partner (Membership No. 36920) UDIN: 20036920AAAABI5387

Mumbai, 17 June 2020

# Acme Living Solutions Private Limited Balance Sheet as at 31st March 2020

Denti	(	· · · · · · · · · · · · · · · · · · ·		
Particu	lars	Note No.	As at 31 March 2020	As at 31 March 2019
ASSET				
NON-C	CURRENT ASSETS			
(a)	Deferred Tax Assets (net)	9		
	TOTAL NON-CURRENT ASSETS			
CURRI	ENT ASSETS			
(a)	Financial assets			
	(i) Cash and Cash Equivalents	3	0.19	0.19
	TOTAL CURRENT ASSETS		0.19	0.19
	TOTAL ASSETS		0.19	0.19
EQUITY	Y AND LIABILITIES			
EQUIT	Ŷ			
(a)	Equity share capital	4	5.00	5.00
(b)	Other equity	5	(13.53)	(12.94)
	TOTAL EQUITY		(8.53)	(7.94)
LIABIL	ITIES			
CU	RRENT LIABILITIES			
(a)	Financial liabilities			
	(i) Borrowings	6	8.72	7.00
	(ii) Other financial liabilities	7		1.13
	TOTAL CURRENT LIABILITIES		8.72	8.13
	TOTAL EQUITY AND LIABILITIES		0.19	0.19
Signific	cant accounting policies	2		
Notes t	to the Ind AS financial statements	3 - 19		

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

PP

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020

For and on behalf of the Board of Directors of Acme Living Solutions Private Limited CIN: U45209DL2008PTC178023

Sunil Dhagat Director DIN: 03081163

Mumbai Dated : 17 June 2020

Vaidehi Modi Director DIN: 06967259

# Acme Living Solutions Private Limited

# Statement of Profit and Loss for the year ended 31st March 2020

				(Rs in lakhs)
Partie	culars	Note No.	For the year ended 31 March 2020	For the year ended 31 March 2019
1	Revenue from Operations :			
11	Other Income			
Ш	Total Income (I +II)		89	•
IV	Expenses:			
	Other expenses	8	0.59	1.04
	Total Expenses	000	0.59	1.04
v	Loss before tax (III-IV)		(0.59)	(1.04
VI	Tax expenses			-
VII	Loss for the year (V-VI)		(0.59)	(1.04
VIII	Other Comprehensive Income:			
IX	Total Comprehensive Loss for the year (VII+VIII)		(0.59)	(1.04
X	Earnings per equity share	11		
	Basic		(1.18)	(2.08
	Diluted		(1.18)	(2.08

 Significant accounting policies
 2

 Notes to the Ind AS financial statements
 3 - 19

 The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

P

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020



For and on behalf of the Board of Directors of Acme Living Solutions Private Limited CIN: U45209DL2008PTC178023

B.

Sunil Dhagat Director DIN: 03081163

tring

Vaidehi Modi Director DIN: 06967259

Mumbai Dated : 17 June 2020

# Acme Living Solutions Private Limited Statement of Cash Flows for the year ended 31st March 2020

(Rs in lakhs)					
Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019			
A CASH FLOW FROM OPERATING ACTIVITIES	visition.	0.566.5			
Loss Before Tax	(0.59)	(1.04)			
(Decrease)/ Increase in other financial liabilities	(1.13)	0.76			
Cash (used in) Operations	(1.72)	(0.28			
Direct Taxes Paid	•	*/			
Net Cash (used in) Operating Activities	(1.72)	(0.28)			
B CASH FLOW FROM INVESTMENT ACTIVITIES					
Net Cash from Investment Activities		0			
C CASH FLOW FROM FINANCING ACTIVITIES					
Proceeds from Short term Borrowing	1.72				
Net Cash from Financing Activities	1.72	10 A			
NET INCREASE / (DECREASE) IN CASH AND CASH EQUIVALENTS		(0.28)			
Opening Balance	0.19	0.47			
Closing Balance	0.19	0.19			
NET INCREASE / (DECREASE) IN CASH AND CASH		(0.28			
EQUIVALENTS		(0.20			
Components of Cash and Cash Equivalents					
Cash on Hand	S. S. San	-			
Balances with Bank	0.19	0.19			
Total Balance	0.19	0.19			

Note:

The Cash flow statement has been prepared under the indirect method as set out in Indian Accounting Standard - 7 ('Ind AS 7') on Cash Flow Statement prescribed in the Companies (Indian Accounting Standard) Rules, 2015, notified under Section 133 of the Companies Act, 2013.

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date attached

#### For Deloitte Haskins & Sells LLP Chartered Accountants

Firm Registration No. 117366W/ W- 100018

P

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020

For and on behalf of the Board of Directors of Acme Living Solutions Private Limited CIN: U45209DL2008PTC178023

Sunil Dhagat Director DIN: 03081163

Vaidehi Modi Director DIN: 06967259

Mumbai Dated : 17 June 2020

157

# A Equity share capital

Particulars	31 March 2020	31 March 2019
Subscribed and Fully Paid up Capital	1	
Equity shares of INR 10 each		
Opening Balance	5.00	5.00
Changes in equity share capital during the year		
Closing Balance	5.00	5.00

## **B** Other equity

Particulars	Retained earnings	Other compreh ensive income	Total
Balance as at 01 April 2018	(11.90)		(11.90)
Loss for the year	(1.04)		(1.04)
Balance as at 31 March 2019	(12.94)		(12.94)
Loss for the year	(0.59)	8	(0.59)
Balance as at 31 March 2020	(13.53)	N 19	(13.53)

The accompanying notes 1 to 19 are an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

in the 4 P

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020

For and on behalf of the **Board of Directors of** Acme Living Solutions Private Limited CIN: U45209DL2008PTC178023

Sunil Dhagat Director DIN: 03081163

Vaidehi Modi Director DIN: 06967259

Mumbai Dated : 17 June 2020

.

Acme Living Solutions Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### 1 General information

- a) Acme Living Solutions Private Limited ('the Company') was incorporated on 13 May 2008 to carry on the business of development of real estate and infrastructure facilities. The Company is a wholly owned subsidiary of Tata Realty and Infrastructure Limited ('the holding company'), which is a wholly owned subsidiary of Tata Sons Private Limited, the ultimate holding Company. The company is a private limited company incorporated and domiciled in India and has its registered office at New Delhi. India.
- b) In order to achieve operating efficiency, the Board of Directors of the Company, in their meeting held on 2 January 2019, had approved a scheme of amalgamation between Acme Living Solutions Private Limited ("the Transferor Company") with its parent company viz. Tata Realty and Infrastructure Limited ("the Transferee Company") and their respective members and creditors w.e.f. 1 April 2019 ("Appointed date"). Consequently, on 29 March 2019, the companies nave thied the said scheme of amalgamation with National Company Law Tribunal, Delh.

#### 2 SIGNIFICANT ACCOUNTING POLICIES

#### i) Basis of Preparation

#### a) Statement of compliance:

The financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 (the Act) and other relevant provisions of the Act.

The financial statements were authorized for issue by the Company's Board of Directors on 17 June 2020

#### b) Going concern :

As at 31 March 2020, the Company's paid up capital and reserves (excluding the deficit in the profit and loss) were INR 5 lakhs (2019: INR 5 Lakhs) correspondingly, the Company's accumulated losses aggregated INR 13.53 lakhs (2019: INR 12.94 lakhs) and the Company's current liabilities exceeded its current assets by INR 8.53 lakhs. These factors give rise to a material uncertainty whether the Company would be able to continue as a going concern. However, in pursuance to Note No. 1 (b) above, the financial statements have been prepared on a going concern basis. Accordingly, these financial statements do not include any adjustments relating to the recoverability and classification of recorded assets, or to amounts and classification of recorded liabilities that might be necessary if the Company is unable to continue as a going concern.

#### c) Historical cost convention:

The financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

- 1 Financial instruments measured at fair value through profit or loss, if applicable
- 2 Financial instruments measured at fair value through other comprehensive income, if applicable

#### d) Key estimates and assumptions

In preparing these financial statements, management has made judgments, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates.

Estimates and assumptions are required in particular for:

Measurement of fair values

The Company's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Company has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the group CFO.

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance learn assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorised into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

. Level 1: guoted prices (unadjusted) in active markets for identical assets or liabilities.

Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

. Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fail into different levels of the fair value hierarchy, then the fair value measurement is categorised in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement

#### ii) Functional and presentation currency

The financial statements are presented in Indian rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian rupees (INR) and all amounts have been rounded-off to the nearest lakhs, except for share data and as otherwise stated. Due to rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.

#### iii) Income tax

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised directly in equity or in OCI.

RKH



Acme Living Solutions Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### a) Current tax

Current tax comprises the expected tax payable or receivable on the taxable income or loss for the year and any adjustment to the tax payable or receivable in respect of previous years. It is measured using tax rates enacted or substantively enacted at the reporting date. Current tax also includes any tax arising from dividends. Management periodically evaluates positions taken in tax returns with respect to situations in which applicable tax regulation is subject to interpretation. It establishes provisions where appropriate on the basis of amounts expected to be paid to the tax authorities.

Minimum Alternate Tax ('MAT') under the provisions of Income-tax Act, 1961 is recognised as current tax in the statement of profit and loss. MAT paid in accordance with the tax laws, which gives future economic benefits in the form of adjustment to future income tax flability, is considered as an asset if there is a convincing evidence that the Company will pay normal tax. Accordingly, MAT is recognised as an asset in the balance sheet when it is probable that the future economic benefit associated with it will flow to the Company.

Current tax assets and liabilities are offset only if, the Company: 1 has a legally enforceable right to set off the recognised amounts; and

2 intends either to settle on a net basis, or to realise the asset and settle the liability simultaneously.

#### b) Deferred tax

Deferred tax is recognised in respect of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for taxation purposes. However, deferred tax liabilities are not recognised if they arise from the initial recognition of goodwill. Deferred income tax is also not accounted for if it arises from initial recognition of an asset or liability in a transaction other than a business combination that at the time of the transaction affects neither accounting profit nor taxable profit (tax loss).

Deferred tax assets are recognised for unused tax losses, unused tax credits and deductible temporary differences to the extent that it is probable that future taxable profits will be available against which they can be used. Deferred tax assets are reviewed at each reporting date and are reduced to the extent that it is no longer probable that the related tax benefit will be realised; such reductions are reversed when the probability of future taxable profits improves. Unrecognised deferred tax assets are reassessed at each reporting date extent that it has become probable that future taxable profits will be available against which they can be used.

Deferred tax is measured at the tax rates that are expected to be applied to temporary differences when they reverse, using tax rates enacted or substantively enacted at the reporting date and are expected to apply when the related deferred income tax asset is realised or the deferred income tax liability is settled.

Deferred tax assets and liabilities are offset only if:

- 1 the entity has a legally enforceable right to set off current tax assets against current tax liabilities: and
- 2 the deferred tax assets and the deferred tax liabilities relate to income taxes levied by the same taxation authority on the same taxable entity.

#### iv) Cash and cash equivalents

For the purpose of presentation in the statement of cash flows, cash and cash equivalents includes cash on hand, deposits held at call with financial institutions, other short-term, highly liquid investments with original maturities of three months or less that are readily convertible to known amounts of cash and which are subject to an insignificant risk of changes in value, and bank overdrafts. Bank overdrafts are shown within borrowings in current liabilities in the balance sheet.

#### v) Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity. Financial instruments also include derivative contracts such as foreign currency foreign exchange forward contracts, interest rate swaps and currency options; and embedded derivatives in the host contract.

## **Financial liabilities**

Classification

The Company classifies all financial liabilities as measured at amortised cost, except for financial liabilities at fair value through profit or loss. Such liabilities, including derivatives that are liabilities, shall be subsequently measured at fair value.

#### Initial recognition and measurement

Financial liabilities are classified, at initial recognition, as financial liabilities at fair value through profit or loss, loans and borrowings, payables, or as derivatives designated as hedging instruments in an effective hedge, as appropriate.

All financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

The Company's financial liabilities include trade and other payables, loans and borrowings including bank overdrafts, financial guarantee contracts and derivative financial instruments.

#### Trade and other payables

Trade and other payables represent liabilities for goods and services provided to the Company prior to the end of the financial year which are unpaid. The amounts are unsecured and are usually paid within 30-60 days of recognition or maybe payable on demand. Trade and other payables are represented as current liabilities unless payment is not due within 12 months after the reporting period. They are recognised initially at their fair value and subsequently recognised at amortised cost using the effective interest method.

#### Loans and borrowings

After initial recognition, interest bearing loans and borrowings are subsequently measured at amortised cost using the EIR method. Gains and losses are recognised in profit or loss when the liabilities are derecognized.

Amortised cost is calculated by taking into account any discount or premium on acquisition and fees or costs that are an integral part of the EIR. The EIR amortisation is included as finance costs in the statement of profit and loss.

#### This category generally applies to interest-bearing loans and borrowings

Preference shares, which are mandatorily redeemable on a specific date, are classified as liabilities. The dividends on these preference shares are recognised in profit or loss as finance costs.

The fair value of the liability portion of an optionally convertible bonds is determined using a market interest rate for an equivalent nonconvertible bonds. This amount is recorded as a liability on an amortised cost basis until extinguished on conversion or redemption of the bonds. The remainder of the proceeds is attributable to the equity portion of the compound instrument. This is recognised and included in shareholders' equity, net of income tax effects, and not subsequently remeasured.

Where the terms of a financial liability are renegotiated and the entity issues equity instruments to a creditor to extinguish all or part of the liability (debt for equity swap), a gain or loss is recognised in profit or loss, which is measured as the difference between the carrying amount of the financial liability and the fair value of the equity instruments issued.

PKH



Acme Living Solutions Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### Derecognition

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit or loss.

#### Offsetting of financial instruments

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable right to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable right must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the company or the counterparty.

#### vi) Provisions

A provision is recognised in the balance sheet when the Company has a legal or constructive obligation as a result of a past event, and it is probable that an outflow of economic benefits will be required to settle the obligation. If the effect is material, provisions are determined by discounting the expected future cash flows at a pre-tax rate that reflects current market assessments of the time value of money and, where appropriate, the risks specific to the liability.

#### vii) Earning Per Share

Basic earnings per share is computed by dividing the profit / (loss) after tax by the weighted average number of equity shares outstanding during the year. The weighted average number of equity shares outstanding during the year is adjusted for the events for bonus issue, bonus element in a rights issue to existing shareholders, share split and reverse share split (consolidation of shares).

Diluted earnings per share is computed by dividing the profit / (loss) after tax as adjusted for dividend, interest and other charges to expense or income (net of any attributable taxes) relating to the dilutive potential equity shares, by the weighted average number of equity shares considered for deriving basic earnings per share and the weighted average number of equity shares which could have been issued on conversion of all dilutive potential equity shares.

REU



#### 3 Cash and cash equivalents

	Particulars	31 March	2020	31 Marc	h 2019
	Cash and cash equivalents	Contrastit allow			
	Balances with Bank		0.19		0.19
	Total		0.19		0.19
Eau	ity Share Capital				
-	Authorised, Issued, Subscribed and Fully Paid up				
10.18	Particulars	31 March	2020	31 Marc	h 2019
		No of Shares	Amount	No of Shares	Amount
	Authorised Capital : Equity Shares of Rs. 10/- each	50,000	5.00	50,000	5.00
	Issued, Subscribed and Fully Paid up Capital : Issued Capital				
	Equity Shares of Rs.10/- each, fully paid	50,000	5.00	50.000	5.00
	Subscribed and Fully Paid up Capital Equity Shares of Rs.10/- each, fully paid	50,000	5.00	50,000	5.00
	Total		5.00		5.00
(b)	Reconciliation of Number of Shares Outstanding				
	Particulars	31 March	2020	31 Marc	h 2019
		No of Shares	Amount	No of Shares	Amount
	As at the beginning of the year	50,000	5.00	50,000	5.00
	Add: Issued during the year		<b>5</b> .		
	As at the end of the year	50.000	5.00	50.000	5.00

#### (c) Terms and rights attached to the equity share

The Company has only one class of equity shares having par value of INR 10 per share. Each holder of equity shares is entitled to one vote per share. The Company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject the approval of shareholders in the ensuing Annual General Meeting.

During the year ended 31 March 2020, the amount of per share dividend recognised as distribution to equity shareholders is Rs Nil (2019; Rs Nil).

In the event of liquidation of the Company, the holders of the equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the share holders.

#### (d) Shares of the company held by the holding company

Name of Shareholder	31 March	2020	31 March 2019		
	No of Shares	Amount	No of Shares	Amount	
Equity shares of INR 10 each, fully paid-up by Tata Realty and Infrastructure Limited, the holding company and its nominee	50,000	5.00	50.000	5.00	

#### (e) Details of shareholding more than 5% in the Company

Name of Shareholder	31 March 2020		31 March 2019	
and the permit walk of the	No of Shares	% Holding	No of Shares	% Holding
Equity shares of INR 10 each, fully paid-up by				
Tata Realty and Infrastructure Limited, the holding company and its nominee	50,000	100%	50,000	100%

# (f) Aggregate number of bonus shares issued and shares issued for consideration other than cash during the period of five years immediately preceding the reporting date :

During the five year period ended 31 March, 2020, the company has not issued any equity shares for consideration other than cash.

#### 5 Other Equity

Particulars	31 March 2020 31 March 2019	
Retained earnings	(13.53)	(12.94)
TOTAL	(13.53)	(12.94)

# Retained earnings31 March 202031 March 2019Particulars31 March 202031 March 2019(Deficit) in statement of profit and loss(12.94)(11.90)Balance at the beginning of the year(0.59)(1.04)Add: (Loss) for the year(0.59)(12.94)Balance at the end of the Year(13.53)(12.94)



REU

# Acme Living Solutions Private Limited Notes to the Ind AS financial statements

#### for the year ended 31 March 2020

(Currency: Indian rupees in lakhs) 6 Current Financial Liabilities - Borrowings

Current	Financial	Liabilities	-	Borrowings	
Th	a to a set a set of	and the second second	1	and the second second second	

Particulars	31 March 2020	31 March 2019	
nterest-free loans from related parties ecoverable on demand			
fata Realty and Infrastructure Limited	8.72	7.00	
TOTAL	8.72	7.00	
The above amount includes			
Secured Borrowings		#7	
Unsecured Borrowings	8.72	7.00	

## 7 Other Current Financial Liabilities

Particulars	31 March 2020	31 March 2019
Other Payables		
- Related Party	*	
- Others	· · · · · · · · · · · · · · · · · · ·	1.13
	1.8	1.13
Total		1.13

# 8 Other Expenses

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019	
Fees & Consultations		0.11	
Rates & Taxes (incl indirect taxes)		0.01	
Bank Charges & Guarantee Commission		0.01	
Audit Fees (refer note 8 (a))	0.59	0.91	
Total	0.59	1.04	

# (a) Remuneration to Statutory Auditors

Particulars	For the year ended	For the year ended	
	31 March 2020	31 March 2019	
Statutory audit	0.50	0.50	
Other Services		0.32	
Goods and Services Tax	0.09	0.09	
Total	0.59	0.91	

## 9 Deferred Tax Assets (net)

# (a) Amounts recognised in profit and loss

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Current income tax		
Origination and reversal of temporary differences		
Reduction in tax rate		
Recognition of previously unrecognised tax losses Change in recognised deductible temporary		
differences		
Total deferred tax expense/(benefit)		-
Tax expense for the year		
(b) Reconciliation of effective tax rate		
Loss before tax	(0.59)	(1.04
Tax using the Company's domestic tax rate 25% ( 2019 25%)	(0.15)	(0.26
Tax effect of:		
Current-year losses for which no deferred tax asset is recognised	0.15	0.26
Total		

R Total



# (c) Unrecognised deferred tax assets

Deferred tax assets have not been recognised in respect of the following items, because of no virtual certainty of the profit in the future against which the deferred tax asset created can be utilised.

	Gross amount	Unrecognised tax effect	Gross amount	Unrecognised tax effect
Tax losses	35.97	8.99	34.93	8.73

Particulars	Financial Year	31 March 2020		31 March 2019	
		Gross amount	Expiry date	Gross amount	Expiry date
Business loss	2011-12	32.19	2019-20	32.19	2019-20
Business loss	2012-13	0.01	2020-21	0.01	2020-21
Business loss	2013-14	0.31	2021-22	0.31	2021-22
Business loss	2014-15	0.37	2022-23	0.37	2022-23
Business loss	2015-16	0.79	2023-24	0.79	2023-24
Business loss	2016-17	0.67	2024-25	0.67	2024-25
Business loss	2017-18	0.58	2025-26	0.58	2025-26
Business loss	2018-19	1.04	2026-27		Brich Brock Step
Total		35.97		34.93	

The Company offsets tax assets and liabilities if and only if it has a legally enforceable right to set off current tax assets and current tax liabilities and the deferred tax assets and deferred tax liabilities relate to income taxes levied by the same tax authority.

Significant management judgement is required in determining provision for income tax, deferred income tax assets and liabilities and recoverability of deferred income tax assets. The recoverability of deferred income tax assets is based on estimates of taxable income and the period over which deferred income tax assets will be recovered. Any changes in future taxable income would impact the recoverability of deferred tax assets.

#### 10 Due to micro and small suppliers

Under the Micro, Small and Medium Enterprises Development Act, 2006, (MSMED) which came into force from 2 October 2006, certain disclosures are required to be made relating to Micro, Small and Medium enterprises. On the basis of the information and records available with the Management, there are no outstanding dues to the Micro, Small and Medium enterprises as defined in the Micro, Small and Medium Enterprises Development Act, 2006 as set out in following disclosure.

Particulars	31 March 2020	31 March 2019
Principal amount remaining unpaid to any supplier as at the year-end		
Interest due thereon		*
The amount of interest paid by the buyer as per the Micro Small and Medium Enterprises Development Act, 2006 (MSMED Act, 2006)	e	5
The amount of interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under MSMED Act, 2006		•
The amount of interest accrued and remaining unpaid at the end of each accounting year	\$1. \$1	80
The amount of further interest remaining due and payable even in the succeeding years, until such date when the interest dues as above are actually paid to the small enterprise for the purpose of disallowance as a deductible expenditure under the MSMED Act, 2006		-

#### 11 Earning Per Share

Earnings Per Share (EPS) = Net Profit attributable to Shareholders / Weighted Number of Shares Outstanding

Particulars		year ended irch 2020	For the year ended 31 March 2019	
(Loss) after tax attributable to equity shareholders Calculation of weighted average number of equity shares:	Α	(0.59)	(1.04)	
Number of equity shares at the beginning of the year	с	50.000	50,000	
Number of equity shares outstanding at the end of the year		50,000	50.000	
Weighted average number of equity shares outstanding during the year	в	50,000	50.000	
Earning Per Share – Basic (Rs.) Earning Per Share – Diluted (Rs.)	(A / B)	(1.18) (1.18)	(2.08 (2.08	
REA				



#### 12 Disclosure of transactions with Related Parties, as required by Indian Accounting Standard (Ind AS) - 24

(a) Related Party Disclosure Ultimate Holding Company Tata Sons Private Limited Holding Company Tata Realty and Infrastructure Limited

(b) Nature of Transactions / relationship / major parties	Holding Co	Total
Finance provided for expenses & on a/c payments		
Tata Realty and Infrastructure Limited	1.72	1.72
	(1.00)	(1.00
Outstanding Balances Payable		14-55
Tata Realty and Infrastructure Limited	8.72	8.72
	(7.00)	(7.00

All the transactions with related parties are at arm's length except the interest free loan from Holding company and all the outstanding balances are unsecured

Note: figures in bracket represent previous years figures.

#### 13 Segment Reporting as per IND AS108 " Operating Segments"

Operating segments are reported in a manner consistent with the internal reporting provided to the chief operating decision maker (CODM). Chief operating decision maker's function is to allocate the resources of the entity and access the performance of the operating segment of the entity. The Board assesses the financial performance and position of the Company and makes strategic decisions. It is identified as being the chief operating decision maker for the company. The Company is operating in the real estate and infrastructure industry and has only domestic operations. The Company has nol revenue, which is development of real estate and infrastructure facilities. All assets of the Company are domiciled in India and the Company has no revenue.

#### 14 Other matters

Information with regard to other matters specified in Schedule III of the Act is either nil or not applicable to the Company for the year.

#### 15 Capital commitment, contingencies and other commitments

There are no capital commitments and contingent liabilities as at the balance sheet date.

Estimated amount of other commitments remaining to be executed on capital account and not provided for (net of advances) Rs Niil (2019: Rs Niil).

#### 16 Financial instruments - Fair values and risk management

#### A Credit risk

Credit risk is the risk of financial loss to the Company if a customer or counterparty to a financial instrument fails to meet its contractual obligations, and arises principally from the Company's receivables from customers and investment securities. The carrying amounts of financial assets represent the maximum credit exposure.

#### Cash and cash equivalents

The Company held cash and cash equivalents with credit worthy banks of INR 0.19 lakhs and INR 0.19 lakhs as at 31 March 2020 and 31 March 2019 respectively. The credit worthiness of such banks is evaluated by the management on an ongoing basis and is considered to be good.

#### B Liquidity risk

Liquidity risk is the risk that the Company will not be able to meet its financial obligations as they become due. The Company manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Company's reputation.

The Company has obtained a finance facility from a bank. The Company also constantly monitors funding options available in the debt and capital markets with a view to maintaining financial flexibility.

As at 31 March, 2020, the Company had working capital of Rs (8.53) lakhs including short term borrowings of Rs 8.72 Lakhs. As at 31 March, 2019, the Company had working capital of Rs (7.94) lakhs including short-term borrowings of Rs 7 Lakhs.

#### C Exposure to liquidity risk

The table below analyses the Company's financial liabilities into relevant maturity groupings based on their contractual maturities for:

all non derivative financial liabilities	Carrying amount	Total	1 year or less	1-2 years	2-5 years
As at 31 March 2020					
Non-derivative financial liabilities					
Unsecured short-term borrowings	8.72	8.72	8.72	200	
Other financial liabilities				140	
	8.72	8.72	8.72		1.1
As at 31 March 2019					
Non-derivative financial liabilities					
Unsecured short-term borrowings	7.00	7.00	7.00		
Other financial liabilities	1.13	1.13	1.13	1.00	
	8.13	8.13	8.13		

D Market risk

Market risk is the risk that changes in market prices – such as foreign exchange rates, interest rates and equity prices – will affect the Company's income or the value of its holdings of financial instruments. The Company is domiciled in India and has its revenues and other transactions in its functional currency i.e. INR. Accordingly the Company is not exposed to any currency risk. Also the Company does not hold any equity investments, accordingly the Company is not exposed to any equity price risk.

REU



# Acme Living Solutions Private Limited

Notes to the Ind AS financial statements

for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

# E Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Company's exposure to market risk for changes in interest rates relates to borrowings from Holding Company.

#### Exposure to interest rate risk

Fixed-rate instruments	31 March 2020	31 March 2019
Financial liabilities		
ICD from Tata Realty and Infrastructure Limited	8.72	7.00

The Company does not account for any fixed-rate financial assets or financial liabilities at fair value through profit or loss, and the Company does not have any designate derivatives (interest rate swaps).

Therefore, a change in interest rates at the reporting date would not affect profit or loss.

#### F Capital management

The Company's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. Management monitors the return on capital as well as the level of dividends to ordinary shareholders.

The Company has obtained borrowings primarily from its holding company.

The Company monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total borrowings, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity. Charge for the year on goodwill amortisation has been deducted while calculating total equity of the company since it represents a pure non-cash expense.

The Company's adjusted net debt to equity ratio at 31 March 2020 was as follows.

Particulars	31 March 2020	
Total borrowings	8.72	7.00
Less : Cash and cash equivalent	0.19	0.19
Adjusted net debt	8.53	6.81
Adjusted equity	(8.53)	(7.94)
Adjusted net debt to adjusted equity ratio (%)	(1.00)	(0.86)

## 17 Financial instruments - Fair values and risk management

#### A Accounting classification and fair values

The following table shows the carrying amounts and fair values of financial assets and financial liabilities, including their levels in the fair value hierarchy. It does not include fair value information for financial assets and financial liabilities if the carrying amount is a reasonable approximation of fair value.

Particulars	FVTPL	FVTOCI	Amotised Cost	Total	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Tota
31 March 2020								
Financial assets								
Cash and cash equivalents	-	1.5	0.19	0.19				
Financial liabilities								
Interest-free loans from	-		8.72	8.72				
related parties			0.72	0.12				
Other current financial			14	08				
liabilities								
31 March 2019								
Financial assets								
Cash and cash equivalents			0.19	0.19			-	-
Financial liabilities								
Interest-free loans from		2	7.00	7.00				24
related parties	1.0	8	1.00	7.00				
Other current financial	÷		1.13	1.13				-
liabilities	075		1110				74-	- 10

#### B Financial risk management

The Company has exposure to the following risks arising from financial instruments:

Credit risk ;

· Liquidity risk ; and

Market risk

#### Risk management framework

The Company's board of directors has overall responsibility for the establishment and oversight of the Company's risk management framework.

REU



## D Measurement of fair values

#### Valuation techniques and significant unobservable inputs

The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for financial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used.

Туре	Valuation technique	Significant	Inter-relationship between
Amortised cost:	Discounted cash flow approach: The	Not applicable	Not applicable
1. Borrowings	valuation model considers the present value of expected payments, discounted using a risk adjusted discount rate.		

#### Transfers between Levels 1 and 2

There have been no transfers between Level 1 and Level 2 during the reporting periods

Level 3 fair values

There are no items in Level 3 fair values

18 There are no offsetting of financial assets and financial liabilities during the year except provision for tax and advance tax.

#### 19 Subsequent events

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

RKU

For and on behalt of the Board of Directors of Acme Living Solutions Private Limited CIN: U45209DL2008PTC178023

3

Sunil Dhagat Director DIN: 03081163

Mumbai Dated : 17 June 2020



Director DIN: 06967259



Chartered Accountants Lotus Corporate Park 1<sup>st</sup> Floor, Wing A – G CTS No.185/A, Jay Coach Off Western Express Highway Goregaon (East) Mumbai – 400 063 Maharashtra, India

Tel: +91 22 6245 1000 Fax: +91 22 6245 1001

# **INDEPENDENT AUDITOR'S REPORT**

To The Members of TRIF Gurgaon Housing Projects Private Limited Report on the Audit of the Ind AS Financial Statements

# Opinion

We have audited the accompanying Ind AS financial statements of TRIF Gurgaon Housing Projects Private Limited ("the Company"), which comprise the Balance Sheet as at March 31, 2019, and the Statement of Profit and Loss (including Other Comprehensive Income), the Statement of Cash Flows and the Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended, ("Ind AS") and other accounting principles generally accepted in India, of the state of affairs of the Company as at March 31, 2019, and its loss, its cash flows and the changes in equity for the year ended on that date.

# **Basis for Opinion**

SKINS

¢

We conducted our audit of the Ind AS financial statements in accordance with the Standards on Auditing specified under section 143(10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Ind AS Financial Statements section of our report. We are independent of the Company in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the Ind AS financial statements.

# Material uncertainty related to Going Concern

We draw attention to Note 2(b) to the Ind As financial statements, which indicates that the Company has incurred a net loss of 1.17 lakhs during the year ended 31<sup>st</sup> March, 2019, the Company's current liabilities exceeded its current assets by 3.01 lakhs. The. These events or conditions, along with other matters as set forth in Note 2(b), indicate that a material uncertainty exists that may cast significant doubt on the Company's ability to continue as a going concern. However, the Ind As financial statements of the Company have been prepared on a going concern basis for the reasons stated in the said Note.

Our opinion is not modified in respect of this matter

# Information Other than the Financial Statements and Auditor's Report Thereon

The Company's Board of Directors is responsible for the other information. The other information comprises the Director's report which we obtained prior to the date of this auditor's report

• Our opinion on the Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.

In connection with our audit of the Ind AS financial statements, our responsibility is to read the other information and, in doing so, consider whether the other information is materially

SKINS

Chartered Accountants inconsistent with the Ind AS financial statements or our knowledge obtained during the course of our audit or otherwise appears to be materially misstated.

If, based on the work we have performed, we conclude that there is a material misstatement
of this other information, we are required to report that fact. We have nothing to report in this
regard.

# Management's Responsibility for the Ind AS Financial Statements

The Company's Board of Directors is responsible for the matters stated in section 134(5) of the Act with respect to the preparation of these Ind AS financial statements that give a true and fair view of the financial position, financial performance including other comprehensive income, cash flows and changes in equity of the Company in accordance with the Ind AS and other accounting principles generally accepted in India.

This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the Ind AS financial statement that give a true and fair view and are free from material misstatement, whether due to fraud or error.

In preparing the Ind AS financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those Board of Directors are also responsible for overseeing the Company's financial reporting process.

# Auditor's Responsibility for the Audit of the Ind AS Financial Statements

Our objectives are to obtain reasonable assurance about whether the Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these Ind AS financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

 Identify and assess the risks of material misstatement of the Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.

 Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under section 143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Company has adequate internal financial controls system in place and the operating effectiveness of such controls.



- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to

events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.

• Evaluate the overall presentation, structure and content of the Ind AS financial statements, including the disclosures, and whether the Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Materiality is the magnitude of misstatements in the Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the Ind AS financial statements.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

# **Report on Other Legal and Regulatory Requirements**

- As required by Section 143(3) of the Act, based on our audit we report, to the extent applicable that:
  - a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit.
  - b) In our opinion, proper books of account as required by law have been kept by the Company so far as it appears from our examination of those books
  - c) The Balance Sheet, the Statement of Profit and Loss including Other Comprehensive Income, the Statement of Cash Flows and Statement of Changes in Equity dealt with by this Report are in agreement with the relevant books of account.
  - d) In our opinion, the aforesaid Ind AS financial statements comply with the Ind AS specified under Section 133 of the Act.
  - e) The matter described in the Material uncertainty related to Going Concern section above, in our opinion, may have an adverse effect on the functioning of the Company.



On the basis of the written representations received from the directors as on March 31, 2019 taken on record by the Board of Directors, none of the directors is disqualified as on March 31, 2019 from being appointed as a director in terms of Section 164(2) of the Act.

- g) Reporting on the adequacy of the internal financial controls over financial reporting of the Company and the operating effectiveness of such controls is not applicable to the company in view of the exemption available to the company in terms of the notification no. G.S.R. 583(E) dated 13 June 2017 issued by the Ministry of Corporate Affairs, Government of India, read with general circular No. 08/2017 dated 25 July 2017.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of section 197(16) of the Act, as amended.

In our opinion and to the best of our information and according to the explanations given to us, the Company being a private company, section 197 of the Act related to managerial remuneration is not applicable.

- With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended in our opinion and to the best of our information and according to the explanations given to us:
  - i. The Company does not have any pending litigations which would impact its financial position.
  - ii. The Company did not have any long-term contracts including derivative contracts for which there were any material foreseeable losses.
  - iii. There were no amounts, which were required to be transferred to the Investor Education and Protection Fund by the Company
- 2. As required by the Companies (Auditor's Report) Order, 2016 ("the Order") issued by the Central Government in terms of Section 143(11) of the Act, we give in "Annexure A" a statement on the matters specified in paragraphs 3 and 4 of the Order.

For Deloitte Haskins & Sells LLP

Chartered Accountants (Firm's Registration No.117366W/W-100018)

Gavella

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai, April 5, 2019

4

Accountants.

# ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT

(Referred to in paragraph 2 under 'Report on Other Legal and Regulatory Requirements' Section of our report of even date)

(i) The Company does not have any property, plant and equipment and hence reporting under clause of the CARO 2016 is not applicable.

(ii) The Company does not have any inventory and hence reporting under clause (ii) of the CARO 2016 is not applicable.

(iii) The Company has not granted any loans, secured or unsecured, to companies, firms, Limited Liability Partnerships or other parties covered in the register maintained under Section 189 of the Companies Act, 2013

(iv) The Company has not granted any loans, made investments or provided guarantees and hence reporting under clause (iv) of the CARO 2016 is not applicable.

(v) According to the information and explanations given to us, the Company has not accepted any deposit during the year as provided under Section 73 to 76 or any other relevant provisions of the Companies Act, 2013. There are no unclaimed deposits any time during the year.

(vi) The maintenance of cost records has not been specified by the Central Government under Section 148(1) of the Companies Act, 2013.

(vii) According to the information and explanations given to us, in respect of statutory dues:

- (a) The Company has been regular in depositing undisputed statutory dues, including Incometax, Service Tax / Goods & Services Tax, cess and other material statutory dues applicable to it to the appropriate authorities.
- (b) There were no undisputed amounts payable in respect of Income-tax, Service Tax / Goods & Services Tax, cess and other material statutory dues in arrears as at March 31, 2019 for a period of more than six months from the date they became payable.
- (c) There are no dues of Income-tax, Service Tax / Goods & Services Tax as on March 31, 2019 on account of disputes.

(viii) In our opinion and according to the information and explanations given to us, the Company has not defaulted in the repayment of loans or borrowings to debenture holders (Bank). The Company has not taken any loans from banks, financial institutions and Government.

(ix) The Company has not raised moneys by way of initial public offer or further public offer (including debt instruments) or term loans and hence reporting under clause (ix) of the CARO 2016 Order is not applicable.

(x) To the best of our knowledge and according to the information and explanations given to us, no fraud by the Company and no material fraud on the Company by its officers or employees has been noticed or reported during the year.

(xi) In our opinion and according to the information and explanations given to us, the Company is Chartered of required to have and does not have any managerial personnel and hence the reporting under

clause (xi) of the CARO 2016 Order for payment of managerial remuneration under Section 197 read with Schedule V of the Companies Act 2013 is not applicable.

(xii) The Company is not a Nidhi Company and hence reporting under clause (xii) of the CARO 2016 Order is not applicable.

(xiii) In our opinion and according to the information and explanations given to us the Company is in compliance with Section 177 and 188 of the Companies Act, 2013, where applicable, for all transactions with the related parties and the details of related party transactions have been disclosed in the financial statements etc. as required by the applicable accounting standards.

(xiv) During the year the Company has not made any preferential allotment or private placement of shares or fully or partly convertible debentures and hence reporting under clause (xiv) of CARO 2016 is not applicable to the Company.

(xv) In our opinion and according to the information and explanations given to us, during the year the Company has not entered into any non-cash transactions with its directors or directors of its holding, subsidiary or associate company or persons connected with them and hence provisions of Section 192 of the Companies Act, 2013 are not applicable.

(xvi) The Company is not required to be registered under Section 45-IA of the Reserve Bank of India Act, 1934.

# For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No. 117366W/W-100018)

Kulehla

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai, April 5, 2019

# TRIF Gurgaon Housing Projects Private Limited Balance Sheet as at 31 March 2019

				(Rs. in lakhs)
Particu	lars	Note No.	As at 31 March 2019	As at 31 March 2018
ASSET	S			
NON-C	URRENT ASSETS			
(a)	Deferred tax assets (net)	9	A STATE OF A	-
	TOTAL NON-CURRENT ASSETS		•	2
CURR	ENT ASSETS			
(a)	Financial assets			
	(i) Cash and Cash Equivalents	3	0.12	0.46
	TOTAL CURRENT ASSETS		0.12	0.46
	TOTAL ASSETS		0.12	0.46
EQUIT	AND LIABILITIES			
EQUIT	Y			
(a)	Equity share capital	4	5.00	5.00
(b)	Other equity	5	(8.01)	(6.84)
	TOTAL EQUITY		(3.01)	(1.84)
LIABIL	ITIES			
CU	RRENT LIABILITIES			
(a)	Financial liabilities			
	(i) Borrowings	6	2.00	2.00
	(ii) Other financial liabilities	7	1.13	0.30
	TOTAL CURRENT LIABILITIES		3.13	2.30
	TOTAL EQUITY AND LIABILITIES		0.12	0.46
	ant accounting policies	2		
Notes t	o the Ind AS financial statements	3 - 19		

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date

# For Deloitte Haskins & Sells LLP Chartered Accountants

Firm Registration No. 117366W/ W- 100018

Djuchta

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019

For and on behalf of the **Board of Directors of TRIF Gurgaon Housing Projects Private Limited** CIN: U74900DL2009PTC188404

Amit Sheth Director DIN - 07997663

S.S. Madget

Santosh Mhadgut Director DIN - 08049549

# TRIF Gurgaon Housing Projects Private Limited Statement of Profit and Loss for the year ended 31 March 2019

		131		(Rs in lakhs)
Parti	culars	Note No.	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018
- 1	Revenue from Operations :		-	
11	Other Income		- 1.5	
III	Total Income (I +II)		-	•
IV	Expenses:			
	Other expenses	8	1.17	0.66
	Total Expenses		1.17	0.66
v	Loss before exceptional items and tax (III-IV)		(1.17)	(0.66)
VI	Exceptional items Income / (Expense)			( <del>-</del> )
VII	Loss before tax (V-VI)		(1.17)	(0.66)
VIII				
	Current Tax		244 (D) (D) (D)	
	Deferred Tax		-	-
	Total tax expenses			9 <b>.</b>
IX	LOSS FOR THE YEAR (VII-VIII)		(1.17)	(0.66)
Х	Other Comprehensive Income:			
	A. (i) Items that will not be reclassified to profit or loss			-
	(ii) Income tax onabove			121 121
	B. (i) Items that will not be reclassified to profit or loss			2 <b>-</b> 1
	(ii) Income tax onabove		-	-
				•
XI	Total Comprehensive Loss For The Year (IX+X)		(1.17)	(0.66)
XII	Earnings per equity share	11		
	Basic		(2.34)	(1.33)
	Diluted		(2.34)	(1.33)
Sign	ificant accounting policies	2		
Note	s to the Ind AS financial statements	3 - 19		

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

ehle the

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019 For and on behalf of the **Board of Directors of TRIF Gurgaon Housing Projects Private Limited** CIN: U74900DL2009PTC188404

Amit Sheth Director DIN - 07997663

S. Mhadgent

Santosh Mhadgut Director DIN - 08049549

# TRIF Gurgaon Housing Projects Private Limited Statement of Cash Flow for the year ended 31 March 2019

Particulars	For the Year Ended 31 March 2019	For the Year Endeo 2018	(Rs in lakhs) d 31 March
A CASH FLOW FROM OPERATING ACTIVITIES	and the second sec		
Loss Before Tax	(1.17)		(0.66)
Increase / (Decrease) in other financial liabilities	0.83	(0.83)	
Increase / (Decrease) in other non-financial liabilities	- 0.83	(0.11)	(0.94)
CASH USED IN OPERATIONS	(0.34)		(1.60)
Direct Taxes Paid			51 <u>1</u> 3
Net Cash Flow used in Operating Activities	(0.34)		(1.60)
B CASH FLOW FROM INVESTMENT ACTIVITIES			
Net Cash Flow from Investment Activities			
C CASH FLOW FROM FINANCING ACTIVITIES			
Proceeds from Short term Borrowings	The second s	2.00	
Net Cash Flow from Financing Activities			2.00
NET INCREASE / (DECREASE) IN CASH AND CASH EQUIVALENTS	(0.34)		0.39
EQUIVALENTS			
Opening Balance	0.46		0.07
Closing Balance	0.12		0.46
NET INCREASE / (DECREASE) IN CASH AND CASH EQUIVALENTS	(0.34)		0.39
Components of Cash and Cash Equivalents			
Balances with Bank	0.12		0.46
Total Balance	0.12	2.	0.46

# Note:

The Cash flow statement has been prepared under the indirect method as set out in Indian Accounting Standard - 7 ('Ind AS 7') on Cash Flow Statement prescribed in Companies (Indian Accounting Standard) Rules, 2015, notified under Section 133 of the Companies Act, 2013.

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date attached

# For Deloitte Haskins & Sells LLP

Chartered Accountants Firm Registration No. 117366W/ W- 100018

Aquichta

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019 For and on behalf of the **Board of Directors of TRIF Gurgaon Housing Projects Private Limited** CIN: U74900DL2009PTC188404

**Amit Sheth** 

Director DIN - 07997663

Sampadout

Santosh Mhadgut Director DIN - 08049549

# **TRIF Gurgaon Housing Projects Private Limited**

Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

# Statement of Changes in Equity for the year ended 31 March 2019

# A Equity Share Capital

Particulars	For the Year Ended 31 March 2019		For the Year Ended 31 March 2018	
	Number of Shares	Rs. in lakhs	Number of Shares	Rs. in lakhs
Subscribed and Fully Paid up Capital				
Equity shares of INR 10 each				
Opening Balance	50,000	5.00	50,000	5.00
Changes in equity share capital during the year		-	-	-
Closing Balance	50,000	5.00	50,000	5.00

## **B** Other Equity

Particulars	Retained Earnings	Other Comprehensi ve Income	Total
Balance as at 1 April 2017	(6.18)	-	(6.18)
Loss for the year	(0.66)		(0.66)
Balance as at 31 March 2018	(6.84)	•	(6.84)
Loss for the year	(1.17)	576	(1.17)
Balance as at 31 March 2019	(8.01)	•	(8.01)

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date

# For Deloitte Haskins & Sells LLP

Chartered Accountants Firm Registration No. 117366W/ W- 100018

we la

Kalpesh J. Mehta Partner

Mumbai Dated : 05 April 2019 For and on behalf of the **Board of Directors of TRIF Gurgaon Housing Projects Private Limited** CIN: U74900DL2009PTC188404

Amit Sheth Director DIN - 07997663

S.S. Madget

Santosh Mhadgut Director DIN - 08049549

# 1 Background of the Company

- a) TRIF Gurgaon Housing Projects Private Limited ('the Company') was incorporated to carry on the business of development of Real Estate & infrastructural facilities. The Company is a wholly owned subsidiary of Tata Realty and Infrastructure Limited, which is a wholly owned subsidiary of Tata Sons Limited, the ultimate holding Company. The company is a private limited company incorporated and domiciled in India and has its registered office at New Delhi, India.
- b) In order to achieve operating efficiency, the Board of Directors of the Company, in their meeting held on January 2, 2019, had approved a scheme of amalgamation between TRIF Gurgaon Housing Projects Private Limited ("the Transferor Company") with its parent company viz. Tata Realty and Infrastructure Limited ("the Transferee Company") and their respective members and creditors w.e.f. April 1, 2019 ("Appointed date"). Consequently, on March 29, 2019, the companies have filed the said scheme of amalgamation with National Company Law Tribunal, Delhi.

# **2 SIGNIFICANT ACCOUNTING POLICIES**

# i) Basis of Preparation

#### a) Statement of compliance:

The financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 (the Act) and other relevant provisions of the Act.

These Ind AS financial statements were authorised for issue by the Company's Board of Directors on 5th April 2019

#### b) Going concern :

As at 31 March 2019, the Company's paid up capital and reserves (excluding the deficit in the profit and loss) were INR 5 lakhs (2018: INR 5 lakhs) and correspondingly, the Company's accumulated losses aggregated INR 8.01 lakhs (2018: INR 6.84 lakhs) which leads to material uncertainty related to Going Concern. However, in pursuance to Note No. 1 (b) above, the financial statements have been prepared on a going concern basis. Accordingly, these financial statements do not include any adjustments relating to the recoverability and classification of recorded labilities that might be necessary if the Company is unable to continue as a going concern.

#### c) Historical cost convention:

The financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

- 1 Financial instruments measured at fair value through profit or loss, if applicable
- 2 Financial instruments measured at fair value through other comprehensive income, if applicable

#### d) Use of estimates and judgements

In preparing these Ind AS financial statements, management has made judgements, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates.

Estimates and assumptions are required in particular for:

• Determination of the estimated useful lives of tangible assets (including Investment Property) and the assessment as to which components of the cost may be capitalised.

Useful lives of tangible assets are based on the life prescribed in Schedule II of the Act. In cases, where the useful lives are different from that prescribed in Schedule II of the Act, they are based on management estimate, taking into account the nature of the asset, the estimated usage of the asset, the operating conditions of the asset, past history of replacement, anticipated technological changes, manufacturers' warranties and maintenance support. Assumptions also need to be made, when the Company assesses, whether an asset may be capitalised and which components of the cost of the asset may be capitalised.

# · Measurement of fair values

The Company's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Company has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the holding company General Manager- Finance.

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorised into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

Level 1: quoted prices (unadjusted) in active markets for identical assets or liabilities.

• Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorised in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement.

## ii) Functional and presentation currency

The financial statements are presented in Indian rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian rupees (INR) and all amounts have been rounded-off to the nearest lakhs, except for share data and as otherwise stated. Charber do rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not perc



## iii) Borrowing cost

Borrowing costs directly attributable to the acquisition, construction or production of an asset that necessarily takes a substantial period of time to get ready for its intended use or sale are capitalised as part of the cost of the asset. All other borrowing costs are expensed in the period in which they occur. Borrowing costs consist of interest and other costs that an entity incurs in connection with the borrowing of funds. Borrowing cost also includes exchange differences to the extent regarded as an adjustment to the borrowing costs.

## iv) Income tax

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised directly in equity or in OCI.

## v) Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity.

# Non-derivative financial liabilities

A financial liability is (i) a contractual obligation to deliver cash or another financial asset to another entity; or to exchange financial instruments under potentially unfavourable conditions; (ii) or a contract that will or may be settled in the entity's own equity instruments and is a non-derivative for which the entity is or may be obliged to deliver a variable number of its own equity instruments; or a derivative that will or may be settled other than by the exchange of a fixed amount of cash or another financial asset for a fixed number of the entity's own equity instruments.

#### Recognition, measurement and classification

A financial liability is recognised in the balance sheet only when the Company becomes party to the contractual provisions to the instrument. The Company initially recognises debt securities issued and subordinated liabilities on the date that they are originated. All other financial liabilities (including liabilities designated at fair value through profit or loss) are recognised initially on the trade date at which the Company becomes a party to the contractual provisions of the instrument.

Financial liabilities are classified as either held at a) fair value through profit or loss, or b) at amortised cost. Management determines the classification of its financial liabilities at the time of initial recognition or, where applicable, at the time of reclassification.

The Company has the following non-derivative financial liabilities: loans and borrowings and trade and other payables. Such financial liabilities are recognised initially at fair value plus any directly attributable transaction costs. Subsequent to initial recognition these financial liabilities are measured at amortised cost using the effective interest method.

# De-recognition and offsetting

The Company derecognises a financial liability when its contractual obligations are discharged or cancelled or expire.

#### vi) Provisions

A provision is recognised in the balance sheet when the Company has a legal or constructive obligation as a result of a past event, and it is probable that an outflow of economic benefits will be required to settle the obligation. If the effect is material, provisions are determined by discounting the expected future cash flows at a pre-tax rate that reflects current market assessments of the time value of money and, where appropriate, the risks specific to the liability.

#### vii) Earnings per share

The Company presents basic and diluted earnings per share (EPS) data for its ordinary shares. Basic EPS is calculated by dividing the profit or loss attributable to ordinary shareholders of the Company by the weighted average number of ordinary shares outstanding during the period, adjusted for own shares held. Diluted EPS is determined by adjusting the profit or loss attributable to ordinary shareholders and the weighted average number of ordinary shareholders and the weighted average number of ordinary shares outstanding, adjusted for own shares held, for the effects of all dilutive potential ordinary shares.

#### viii) Cash and cash equivalents

Cash and cash equivalent comprise of cash on hand and at banks including short-term deposits with an original maturity of three months or less, which are subject to an insignificant risk of changes in value.





#### TRIF Gurgaon Housing Projects Private Limited Notes to the Ind AS financial statements (Currency: Indian rupees in lakhs)

#### 3 Cash and Cash Equivalents

Particulars	As at 31 March 2019	As at 31 March 2018	
Cash and Cash Equivalents	A Designation of the second se		
Balances with Bank	0.12	0.46	
Total	0.12	0.46	

#### 4 Equity Share Capital (a) Authorised, Iss

a) _	Authorised, Issued, Subscribed and Fully Paid up	:			
Particulars	Particulars	As at 31 Ma	rch 2019	As at 31 Ma	rch 2018
		No of Shares	Amount	No of Shares	Amount
	Authorised Capital :				
	Equity Shares of Rs.10/- each	50,000	5.00	50,000	5.0
	Issued, Subscribed and Fully Paid up Capital :				
	Issued Capital				
	Equity Shares of Rs.10/- each, fully paid	50,000	5.00	50,000	5.0
	Subscribed and Fully Paid up Capital		Contraction of the second		
	Equity Shares of Rs.10/- each, fully paid	50,000	5.00	50,000	5.0
	Total		5.00		5.0

#### (b) Reconciliation of Number of Shares Outstanding

Particulars	As at 31 Ma	As at 31 March 2019		As at 31 March 2018	
	No of Shares	Amount	No of Shares	Amount	
As at the beginning of the year	50,000	5.00	50,000	5.00	
Add: Issued during the year					
As at the end of the year	50 000	5.00	50 000	5.00	

## (c) Terms and rights attached to the equity share

The Company has only one class of equity shares having par value of INR 10 per share. Each holder of equity shares is entitled to one vote per share. The Company During the year ended 31 March 2019, the amount of per share dividend recognised as distribution to equity shareholders is Rs Nil (2018: Rs Nil). In the event of liquidation of the Company, the holders of the equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the share holders.

5.00

5.00

5.00 5.00

#### (d) Shares of the company held by the Holding company

Name of Shousholder	As at 31 Ma	As at 31 March 2019		As at 31 March 2018	
Name of Shareholder	No of Shares	Amount:	No of Shares	Amount	
Equity shares of INR 10 each, fully paid-up by					
Tata Realty and Infrastructure Limited	50.000	5.00	50,000	5.00	
	As at 31 March 2019		As at 31 March 2018		
		1011 2019	AS AL ST MA	rch 2018	
Name of Shareholder	No of Shares	% Holding	No of Shares	rch 2018 % Holding	
Name of Shareholder Equity shares of INR 10 each, fully paid-up by	No of Shares				

(f) Aggregate number of bonus shares issued and shares issued for consideration other than cash during the period of five years immediately preceding the reporting date :

During the five year period ended 31 March 2019, the company has not issued any equity shares for consideration other than cash.

#### 5 Other Equity

6 C

(e

Particulars	As at 31 March 2019	As at 31 March 2018
Retained earnings	(8.01)	(6.84
TOTAL	(8.01)	(6.84
Retained earnings		
Particulars	As at 31 March 2019	As at 31 March 2018
(Deficit) in statement of profit and loss		
Balance at the beginning of the year	(6.84)	(6.18
Add (I and March 1	(1.17)	(0.66
Add: (Loss) for the year	(1.17)	
Balance at the end of the Year	(8.01)	and the second se
	· · · · · · · · · · · · · · · · · · ·	
Balance at the end of the Year ent Financial Liabilities - Borrowings The borrowings are analysed as follows :	(8.01)	(6.84
Balance at the end of the Year ent Financial Liabilities - Borrowings The borrowings are analysed as follows : Particulars	(8.01)	(6.84
Balance at the end of the Year ent Financial Liabilities - Borrowings The borrowings are analysed as follows : Particulars Interest-free loans from related parties	(8.01) As at 31 March 2019	(6.84 As at 31 March 2018 2.00
Balance at the end of the Year ent Financial Liabilities - Borrowings The borrowings are analysed as follows : Particulars Interest-free loans from related parties Tata Realty and Infrastructure Limited	(8.01) As at 31 March 2019 2.00	(6.84 As at 31 March 2018
Balance at the end of the Year ent Financial Liabilities - Borrowings The borrowings are analysed as follows : Particulars Interest-free loans from related parties Tata Realty and Infrastructure Limited TOTAL	(8.01) As at 31 March 2019 2.00	(6.84 As at 31 March 2018 2.00

#### 7 Other Current Financial Liabilities

Particulars	As at 31 March 2019	As at 31 March 2018
Other Payables		
- Related Party		-
- Others	1.13	0.30
and the second se	1.13	0.30
Total	1.13	0.30




## (a) Remuneration to Statutory Auditors

Particulars	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018	
Statutory audit	0.50	0.23	
Other Services	0.32		
Service tax / GST	0.09	0.04	
Total	0.91	0.27	

## 9 Deferred Tax Assets

(a) Amounts recognised in profit and loss

Particulars	For the Year Ended 31 March 2019	For the Year Ended 31 March 2018
Current income tax		
Origination and reversal of temporary differences		2
Reduction in tax rate		
Recognition of previously unrecognised tax losses		
Change in recognised deductible temporary differences		
Total deferred tax expense/(benefit)	-	
Tax expense for the year		
(b) Reconciliation of effective tax rate		
Loss before tax	(1.17)	(0.66)
Tax using the Company's domestic tax rate 25% (p.y.25%)	(0.29)	(0.17)
Reduction in tax rate		
Tax effect of:		
Current-year losses for which no deferred tax asset is recognised	0.29	0.17
Total		

(c) Unrecognised deferred tax assets

Deferred tax assets have not been recognised in respect of the following items, because of no virtual certainty of the profit in the future year against which the deferred tax asset created can be utilised.

		Gross amount	Unrecognised tax effect	Gross amount	Unrecognised tax effect
Tax losses		8.59	2.15	8.56	2.14
(d) Tax losses carried	forward				
Particulars	Financial Year	As at 31 N	larch 2019	As at 31 M	arch 2018
		Gross amount	Expiry date	Gross amount	Expiry date
Business loss	2010-11	-	2018-19	0.63	2018-19
Business loss	2011-12	0.36	2019-20	0.36	2019-20
Business loss	2012-13	0.30	2020-21	0.30	2020-21
Business loss	2013-14	4.12	2021-22	4.12	2021-22
Business loss	2014-15	0.39	2022-23	0.39	2022-23
Business loss	2015-16	1.42	2023-24	1.42	2023-24
Business loss	2016-17	1.33	2024-25	1.33	2024-25
Business loss	2017-18	0.66	2025-26		
Total		8.59		8.56	

The Company offsets tax assets and liabilities if and only if it has a legally enforceable right to set off current tax assets and current tax liabilities and the deferred tax assets and deferred tax liabilities relate to income taxes levied by the same tax authority.

Significant management judgement is required in determining provision for income tax, deferred income tax assets and liabilities and recoverability of deferred income tax assets. The recoverability of deferred income tax assets is based on estimates of taxable income and the period over which deferred income tax assets will be recovered. Any changes in future taxable income would impact the recoverability of deferred tax assets.

## 10 Due to micro and small suppliers

Under the Micro, Small and Medium Enterprises Development Act, 2006, (MSMED) which came into force from 2 October 2006, certain disclosures are required to be made relating to Micro, Small and Medium enterprises. On the basis of the information and records available with the Management, there are no outstanding dues to the Micro, Small and Medium enterprises as defined in the Micro, Small and Medium Enterprises Development Act, 2006 as set out in following disclosure.

Particulars	As at 31 March 2019	As at 31 March 2018
Principal amount remaining unpaid to any supplier as at the year-end		
Interest due thereon	and the second sec	
The amount of interest paid by the buyer as per the Micro Small and Medium Enterprises Development Act, 2006 (MSMED Act, 2006)		
The amount of interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under MSMED Act, 2006		
The amount of interest accrued and remaining unpaid at the end of each accounting year		
The amount of further interest remaining due and payable even in the succeeding years, until such date when the interest dues as above are actually paid to the small enterprise for the purpose of disallowance as a teductible expenditure under the MSMED Act, 2006		
hartered m countants		



### TRIF Gurgaon Housing Projects Private Limited Notes to the Ind AS financial statements (Currency: Indian rupees in lakhs)

### 11 Earning Per Share

Earnings Per Share (EPS) = Net Profit attributable to Shareholders / Weighted Number of Shares Outstanding

Particulars	For the Year Ended 31 March 2019		the Year Ended 31 March 2018
(Loss) after tax attributable to equity shareholders	А	(1.17)	(0.66)
Calculation of weighted average number of equity shares:			
Number of equity shares at the beginning of the year	с	50,000	50,000
Optionally convertible participatory debentures			
Number of equity shares outstanding at the end of the year	State and	50,000	50,000
Weighted average number of equity shares outstanding during the year	В	50,000	50,000
Earning Per Share – Basic (Rs.)	(A / B)	(2.34)	(1.33)
Earning Per Share – Diluted (Rs.)		(2.34)	(1.33)

12 Disclosure of transactions with Related Parties, as required by Indian Accounting Standard (Ind AS) - 24

a) Related Party Disclosure	
Ultimate holding company	1
Tata Sons Limited	
Holding Company	

Tata Realty and Infrastructure Limited

b) Nature of Transactions / relationship / major parties	Holding Co	Company having significant influence	Total
Finance provided for Loans, expenses & on a/c payments			
Tata Realty and Infrastructure Limited	3.63	2.00	2.00
	(-)	(2.00)	(2.00
Outstanding Balances Payable			
Tata Realty and Infrastructure Limited	-	2.00	2.00
	(-)	(2.00)	(2.00

All the transactions with related parties are at arm's length and all the outstanding balances are unsecured except the interest free loan from Holding company.

Note: figures in bracket represent previous years figures.

## 13 Segment Reporting as per IND AS108 " Operating Segments"

Operating segments are reported in a manner consistent with the internal reporting provided to the chief operating decision maker (CODM). Chief operating decision maker's function is to allocate the resources of the entity and access the performance of the operating segment of the entity. The Board of directors of company assesses the financial performance and position of the Company and makes strategic decisions. It is identified as being the chief operating decision maker for the company.

The company is operating in the real estate and infrastructure industry and has only domestic opearations. The Company has only one reportable business segment, which is development of real estate and infrastructure facilities and only one reportable geographical segment. All assets of the Company are domiciled in India and the Company has no revenue from operation.

### 14 Other matters

Information with regard to other matters specified in Schedule III of the Act is either nil or not applicable to the Company for the year.

### 15 Capital commitment, contingencies and other commitments

There are no capital commitments and contingent liabilities as at the balance sheet date.

Estimated amount of other commitments remaining to be executed on capital account and not provided for (net of advances) Rs Nil (2018: Rs Nil).

## 16 Financial instruments - Fair values and risk management

## A Credit risk

Currently, the Company is not exposed to any credit risk from trade receivables.

Cash and cash equivalents

The Company held cash and cash equivalents with credit worthy banks and financial institustions of INR 0.17 and INR 0.46 as at 31 March 2019 and 31 March 2018 respectively. The credit worthiness of such banks and financial institutions is evaluated by the management on an ongoing basis and is considered to be good.

### B Liquidity risk

Liquidity risk is the risk that the Company will not be able to meet its financial obligations as they become due. The Company manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Company's reputation.

As at 31 March 2019, the Company had working capital of INR (1.77). As at 31 March 2018, the Company had working capital of INR (1.77).

C Exposure to liquidity risk

The table below analyses the Company's financial liabilities into relevant maturity groupings based on their contractual maturities for: - all non derivative financial liabilities

Particulars	Carrying amount	Total	1 year or less	1-2 years	2-5 years
As at 31 March 2019					
Non-derivative financial liabilities					
Unsecured short-term borrowings	2.00	2.00	2.00		
e other financial liabilities	1.13	1.13	1.13		
101	3.13	3.13	3.13		
d As at 31 March 2018					
nts Non-derivative financial liabilities					
Consecured short-term borrowings	2.00	2.00	2.00		
Other financial liabilities	0.30	0.30	0.30		11
8//	2.30	2.30	2.30		11 -
Market risk					11 11

Market risk is the risk that changes in market prices – such as foreign exchange rates, interest rates and equity prices – will affect the Company's income or the value of its holdings of financial instruments. The Company is not exposed to any sign 82 nt currency risk and equity price risk.

## TRIF Gurgaon Housing Projects Private Limited Notes to the Ind AS financial statements

(Currency: Indian rupees in lakhs)

## E Interest rate risk

17

Currently, the Company is not exposed to any interest rate risk on any financial assets and liabilities.

## F Capital management

The Company's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. It sets the amount of capital required on the basis of annual business and long-term operating plans which include capital and other strategic investments.

The Company has adequate cash and bank balances and continues to remain debt-free. The company monitors its capital and makes a regular assessment of any debt requirements.

The Company monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total borrowings, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity other than amounts accumulated in the hedging reserve.

Paticulars	As at 31 March 2019	As at 31 March 2018
Total borrowings	2.00	2.00
Less : Cash and cash equivalent	0.12	0.46
Adjusted net debt	1.88	1.54
Adjusted equity	(3.01)	(1.84)
Adjusted net debt to adjusted equity ratio %	(0.62)	(0.84)

Financial instruments – Fair values and risk management

## A Accounting classification and fair values

The following table shows the carrying amounts and fair values of financial assets and financial liabilities, including their levels in the fair value hierarchy. It does not include fair value information for financial assets and financial liabilities if the carrying amount is a reasonable approximation of fair value.

Particulars	FVTPL	FVTOCI	Amotised Cost	Total	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Total
31 March 2019							in parts	
Financial assets								
Cash and cash equivalents			0.12	0.12			2	
Financial liabilities								
Interest-free loans from			2.00					
related parties	•	0.72	2.00	2.00		-	-	
Other current financial			1.13	1.13				
liabilities	57.5	. 73	1.10	1.15		-		-
31 March 2018								
Financial assets								
Cash and cash equivalents	•	1.41	0,46	0.46		18	<u> </u>	-
Financial liabilities								
Interest-free loans from			2.00	2.00				
related parties	•	•	2.00	2.00		-	5	
Other current financial			0.30	0.30				
liabilities	•		0.30	0.50		-	5	•

### B Financial risk management

The Company has exposure to the following risks arising from financial instruments:

· Credit risk ;

· Liquidity risk ; and

Market risk

### C Risk management framework

The Company's board of directors has overall responsibility for the establishment and oversight of the Company's risk management framework.

## D Measurement of fair values

## Valuation techniques and significant unobservable inputs

The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for financial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used. Financial instruments measured at fair value

Туре	Valuation technique	Significant	Inter-relationship between
Amortised cost: 1. Borrowings	Discounted cash flow approach: The valuation model considers the present value of expected payments, discounted using a risk adjusted discount rate.		Not applicable

### Transfers between Levels 1 and 2

There have been no transfers between Level 1 and Level 2 during the reporting periods Level 3 fair values

There are no items in Level 3 fair values

18 There are no offsetting of financial assets and financial liabilities during the year.

## 19 Subsequent events

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.



For and on behalf of the Board of Directors of TRIF Gurgaon Housing Projects Private Limited CIN: U74900DL2009PTC188404

Amit Sheth Director DIN - 07997663

Mumbai Dated : 05 April 2019

Smhadget

Santosh Mhadgut Director DIN - 08049549

## **Chartered Accountants**

Indiabulls Finance Centre Tower 3, 27th - 32nd Floor Senapati Bapat Marg Elphinstone Road (West) Mumbai - 400 013 Maharashtra, India Tel: +91 22 6185 4000 Fax: +91 22 6185 4001

## **INDEPENDENT AUDITOR'S REPORT**

## To The Members of TRIF Gurgaon Housing Projects Private Limited **Report on the Audit of the Ind AS Financial Statements**

## Opinion

We have audited the accompanying Ind AS financial statements of TRIF Gurgaon Housing Projects Private Limited ("the Company"), which comprise the Balance Sheet as at 31 March 2020, and the Statement of Profit and Loss (including Other Comprehensive Income), the Statement of Cash Flows and the Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under Section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended, ("Ind AS") and other accounting principles generally accepted in India, of the state of affairs of the Company as at 31 March 2020, and its loss, total comprehensive loss, its cash flows and the changes in equity for the year ended on that date.

## **Basis for Opinion**

We conducted our audit of the Ind AS financial statements in accordance with the Standards on Auditing specified under Section 143(10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Ind AS Financial Statements section of our report. We are independent of the Company in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the Ind AS financial statements.

## Material uncertainty related to Going Concern

We draw attention to Note 2(b) to the Ind AS financial statements, wherein the events or conditions, along with other matters as set forth in the said note, indicate that a material uncertainty exists that may cast significant doubt on the Company's ability to continue as a going concern. However, the Ind AS financial statements of the Company have been prepared on a going concern basis for the reasons stated in the said Note.

Our opinion is not modified in respect of this matter

## Information Other than the Financial Statements and Auditor's Report Thereon



The Company's Board of Directors is responsible for the other information. The other information comprises the Director's report which we obtained prior to the date of this auditor's report.

- Our opinion on the Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.
- In connection with our audit of the Ind AS financial statements, our responsibility is to read the other information and, in doing so, consider whether the other information is materially inconsistent with the Ind AS financial statements or our knowledge obtained during the course of our audit or otherwise appears to be materially misstated.
- If, based on the work we have performed, we conclude that there is a material misstatement of this other information, we are required to report that fact. We have nothing to report in this regard.

## Management's Responsibility for the Ind AS Financial Statements

The Company's Board of Directors is responsible for the matters stated in Section 134(5) of the Act with respect to the preparation of these Ind AS financial statements that give a true and fair view of the financial position, financial performance including other comprehensive income, cash flows and changes in equity of the Company in accordance with the Ind AS and other accounting principles generally accepted in India.

This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the Ind AS financial statements that give a true and fair view and are free from material misstatement, whether due to fraud or error.

In preparing the Ind AS financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those Board of Directors are also responsible for overseeing the Company's financial reporting process.

## Auditor's Responsibility for the Audit of the Ind AS Financial Statements

Our objectives are to obtain reasonable assurance about whether the Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these Ind AS financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

 Identify and assess the risks of material misstatement of the Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.



- Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under Section143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Company has adequate internal financial controls system in place and the operating effectiveness of such controls.
- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.
- Evaluate the overall presentation, structure and content of the Ind AS financial statements, including the disclosures, and whether the Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Materiality is the magnitude of misstatements in the Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the Ind AS financial statements.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

## **Report on Other Legal and Regulatory Requirements**

- 1. As required by Section 143(3) of the Act, based on our audit we report, to the extent applicable that:
  - a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit.
  - b) In our opinion, proper books of account as required by law have been kept by the Company so far as it appears from our examination of those books
  - c) The Balance Sheet, the Statement of Profit and Loss including Other Comprehensive Income, the Statement of Cash Flows and Statement of Changes in Equity dealt with by this Report are in agreement with the relevant books of account.
  - d) In our opinion, the aforesaid Ind AS financial statements comply with the Ind AS specified under Section 133 of the Act.

e) The matter described in the Material uncertainty related to Going Concern section above, in our opinion, may have an adverse effect on the functioning of the Company.

RKH

- f) On the basis of the written representations received from the directors as on 31 March 2020 taken on record by the Board of Directors, none of the directors is disqualified as on 31 March 2020 from being appointed as a director in terms of Section 164(2) of the Act.
- g) Reporting on the adequacy of the internal financial controls over financial reporting of the Company and the operating effectiveness of such controls is not applicable to the company in view of the exemption available to the company in terms of the notification no. G.S.R. 583(E) dated 13 June 2017 issued by the Ministry of Corporate Affairs, Government of India, read with general circular No. 08/2017 dated 25 July 2017.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of Section 197(16) of the Act, as amended.

In our opinion and to the best of our information and according to the explanations given to us, the Company being a private company, Section 197 of the Act related to managerial remuneration is not applicable.

- With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended in our opinion and to the best of our information and according to the explanations given to us:
  - i. The Company does not have any pending litigations which would impact its financial position.
  - ii. The Company did not have any long-term contracts including derivative contracts for which there were any material foreseeable losses.
  - iii. There were no amounts, which were required to be transferred to the Investor Education and Protection Fund by the Company
- 2. As required by the Companies (Auditor's Report) Order, 2016 ("the Order") issued by the Central Government in terms of Section 143(11) of the Act, we give in "Annexure A" a statement on the matters specified in paragraphs 3 and 4 of the Order.

For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No.117366W/W-100018)

ptkine.

Rajesh K. Hiranandani Partner (Membership No. 36920) UDIN: 20036920AAABL8765

Mumbai, 17 June 2020

## ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT

(Referred to in paragraph 2 under 'Report on Other Legal and Regulatory Requirements' Section of our report of even date)

- (i) The Company does not have any fixed assets and hence reporting under clause (i) of the CARO 2016 is not applicable.
- (ii) The Company does not have any inventory and hence reporting under clause (ii) of the CARO 2016 is not applicable.
- (iii) The Company has not granted any loans, secured or unsecured, to companies, firms, Limited Liability Partnerships or other parties covered in the register maintained under Section 189 of the Companies Act, 2013
- (iv) The Company has not granted any loans, made investments or provided guarantees and hence reporting under clause (iv) of the CARO 2016 is not applicable.
- (v) According to the information and explanations given to us, the Company has not accepted any deposit during the year as provided under Section 73 to 76 or any other relevant provisions of the Companies Act, 2013. There are no unclaimed deposits any time during the year.
- (vi) The maintenance of cost records has not been specified by the Central Government under Section 148(1) of the Companies Act, 2013.
- (vii) According to the information and explanations given to us, in respect of statutory dues:
  - (a) The Company has been regular in depositing undisputed statutory dues, including Incometax, Goods & Services Tax, cess and other material statutory dues applicable to it to the appropriate authorities.
  - (b) There were no undisputed amounts payable in respect of Income-tax, Goods & Services Tax, cess and other material statutory dues in arrears as at 31 March 2020 for a period of more than six months from the date they became payable.
  - (c) There are no dues of Income-tax, Goods & Services Tax as on 31 March 2020 on account of disputes.
- (viii) In our opinion and according to the information and explanations given to us, The Company has not taken any loans or borrowings from financial institutions, banks and government or has not issued any debentures. Hence reporting under clause (viii) of CARO 2016 is not applicable to the Company.
- (ix) The Company has not raised moneys by way of initial public offer or further public offer (including debt instruments) or term loans and hence reporting under clause (ix) of the CARO 2016 Order is not applicable.
- (x) To the best of our knowledge and according to the information and explanations given to us, no fraud by the Company and no material fraud on the Company by its officers or employees has been noticed or reported during the year.
- (xi) In our opinion and according to the information and explanations given to us, the Company is not required to have and does not have any managerial personnel and hence the reporting under clause of the CARO 2016 Order for payment of managerial remuneration under Section 197 read with Schedule V of the Companies Act 2013 is not applicable.

RKH

- (xii) The Company is not a Nidhi Company and hence reporting under clause (xii) of the CARO 2016 Order is not applicable.
- (xiii) In our opinion and according to the information and explanations given to us the Company is in compliance with Section 177 and 188 of the Companies Act, 2013, where applicable, for all transactions with the related parties and the details of related party transactions have been disclosed in the financial statements etc. as required by the applicable accounting standards.
- (xiv) During the year the Company has not made any preferential allotment or private placement of shares or fully or partly convertible debentures and hence reporting under clause (xiv) of CARO 2016 is not applicable to the Company.
- (xv) In our opinion and according to the information and explanations given to us, during the year the Company has not entered into any non-cash transactions with its directors or directors of its holding, subsidiary or associate company or persons connected with them and hence provisions of Section 192 of the Companies Act, 2013 are not applicable.
- (xvi) The Company is not required to be registered under Section 45-IA of the Reserve Bank of India Act, 1934.

## For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No. 117366W/W-100018)

ptkine.

Rajesh K. Hiranandani Partner (Membership No. 36920) UDIN: 20036920AAABL8765

Mumbai, 17 June 2020

## TRIF Gurgaon Housing Projects Private Limited Balance Sheet as at 31 March 2020

				(Rs. in lakhs)
Particu	lars	Note No.	As at 31 March 2020	As at 31 March 2019
ASSET	S			
NON-C	URRENT ASSETS			
(a)	Deferred tax assets (net)	9		
	TOTAL NON-CURRENT ASSETS			
CURR	ENT ASSETS			
(a)	Financial assets			
	(i) Cash and Cash Equivalents	3	0.12	0.12
	TOTAL CURRENT ASSETS		0.12	0.12
	TOTAL ASSETS		0.12	0.12
EQUIT	Y AND LIABILITIES			
EQUIT	Y			
(a)	Equity share capital	4	5.00	5.00
(b)	Other equity	5	(8.60)	(8.01)
	TOTAL EQUITY		(3.60)	(3.01)
LIABIL	ITIES			
CU	RRENT LIABILITIES			
(a)	Financial liabilities			
	(i) Borrowings	6	3.72	2.00
	(ii) Other financial liabilities	7		1.13
	TOTAL CURRENT LIABILITIES	<i></i>	3.72	3.13
	TOTAL EQUITY AND LIABILITIES		0.12	0.12
	cant accounting policies	2		
Notes	to the Ind AS financial statements	3 - 19		

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

pru.u.

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020

For and on behalf of the Board of Directors of TRIF Gurgaon Housing Projects Private Limited CIN: U74900DL2009PTC188404

Q Amit Sheth

Amit Sheth Director DIN - 07997663

Mumbai Dated : 17 June 2020

S.S. Mhadget

Santosh Mhadgut Director DIN - 08049549

¥

## TRIF Gurgaon Housing Projects Private Limited

## Statement of Profit and Loss for the year ended 31 March 2020

				(Rs in lakhs)
Parti	culars	Note No.	For the year ended 31 March 2020	For the year ended 31 March 2019
1	Revenue from Operations :		-	
11	Other Income		-	
ш	Total Income (I +II)		•	2 <u>0</u> 7
IV	Expenses:			
	Other expenses	8	0.59	1.17
	Total Expenses		0.59	1.17
v	Loss before tax (III-IV)		(0.59)	(1.17)
VI	Tax expenses		-	-
VII	Loss for the year (V-VI)		(0.59)	(1.17)
VIII	Other Comprehensive Income:			
IX	Total Comprehensive Loss For The Year (VII+VIII)		(0.59)	(1.17)
X	Earnings per equity share	11		
	Basic		(1.18)	(2.34)
	Diluted		(1.18)	(2.34)
Signi	ficant accounting policies	. 2		
Note	s to the Ind AS financial statements	3 - 19		

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date

## For Deloitte Haskins & Sells LLP

Chartered Accountants Firm Registration No. 117366W/ W- 100018

pru-

Rajesh K. Hiranandani Partner

	Mumbai
a	Dated : 17 June 2020

For and on behalf of the **Board of Directors of TRIF Gurgaon Housing Projects Private Limited** CIN: U74900DU2009PTC188404

0 Amit Sheth Director

S.S. Mhadgut

Amit Sheth Director • DIN - 07997663

Santosh Mhadgut Director DIN - 08049549

## TRIF Gurgaon Housing Projects Private Limited Statement of Cash Flow for the year ended 31 March 2020

ANALY IN CONTRACTOR STOCK	(Rs in lakhs)
For the year ended 31 March 2020	For the year ended 31 March 2019
(0.59)	(1.17)
(1.13)	0.83
(1.72)	(0.34)
(1.72)	(0.34)
	20 20
1.72	
1.72	<b>7</b> 2
	(0.34)
0.12	0.46
0.12	0.12
	(0.34)
0.12	0.12
0.12	0.12
	31 March 2020 (0.59) (1.13) (1.72) (1.72) (1.72) - - - - - - - 0.12 0.12 - - - - - - - - - - - - - - - - - - -

### Note:

The Cash flow statement has been prepared under the indirect method as set out in Indian Accounting Standard - 7 ('Ind AS 7') on Cash Flow Statement prescribed in the Companies (Indian Accounting Standard) Rules, 2015, notified under Section 133 of the Companies Act, 2013.

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date attached

## For Deloitte Haskins & Sells LLP

Chartered Accountants Firm Registration No. 117366W/ W- 100018

P

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020

For and on behalf of the **Board of Directors of TRIF Gurgaon Housing Projects Private Limited** CIN: U74900DL2009PTC188404

Amit Sheth Director DIN - 07997663

S.S. Mhadget Santosh Mhadgut

Director DIN - 08049549



TRIF Gurgaon Housing Projects Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

## A Equity Share Capital

Particulars	31 March 2020	31 March 2019	
Subscribed and Fully Paid up Capital			
Equity shares of INR 10 each			
Opening Balance	5.00	5.00	
Changes in equity share capital during the year		(A.	
Closing Balance	5.00	5.00	

## **B** Other Equity

Particulars	Retained earnings	Other comprehensive income	Total
Balance as at 1 April 2018	(6.84)	1	(6.84)
Loss for the year	(1.17)	15	(1.17)
Balance as at 31 March 2019	(8.01)	68	(8.01)
Loss for the year	(0.59)	17	(0.59)
Balance as at 31 March 2020	(8.60)	-	(8.60)

The accompanying notes 1 to 19 form an integral part of these Ind AS financial statements.

As per our report of even date

For Deloitte Haskins & Sells LLP Chartered Accountants Firm Registration No. 117366W/ W- 100018

K He P

Rajesh K. Hiranandani Partner

Mumbai Dated : 17 June 2020 For and on behalf of the **Board of Directors of TRIF Gurgaon Housing Projects Private Limited** CIN: U74900D\_2009PTC188404

Q Amit Sheth

Amit Sheth Director DIN - 07997663

S.S. Mhadget

Santosh Mhadgut Director DIN - 08049549

TRIF Gurgaon Housing Projects Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

## 1 Background of the Company

- a) TRIF Gurgaon Housing Projects Private Limited ('the Company') was incorporated on 13 March 2009 to carry on the business of development of Real Estate & infrastructural facilities. The Company is a wholly owned subsidiary of Tata Realty and Infrastructure Limited, which is a wholly owned subsidiary of Tata Sons Private Limited, the ultimate holding Company. The company is a private limited company incorporated and domiciled in India and has its registered office at New Delhi, India.
- b) In order to achieve operating efficiency, the Board of Directors of the Company, in their meeting held on 2 January 2019, had approved a scheme of amalgamation between TRIF Gurgaon Housing Projects Private Limited ("the Transferor Company") with its parent company viz. Tata Realty and Infrastructure Limited ("the Transferee Company") and their respective members and creditors w.e.f. 1 April 2019 ("Appointed date"). Consequently, on 29 March 2019, the companies have filed the said scheme of amalgamation with National Company Law Tribunal, Delhi.

### 2 SIGNIFICANT ACCOUNTING POLICIES

### i) Basis of Preparation

### a) Statement of compliance:

The financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 (the Act) and other relevant provisions of the Act.

These Ind AS financial statements were authorised for issue by the Company's Board of Directors on 17 June 2020

### b) Going concern :

As at 31 March 2020, the Company's paid up capital and reserves (excluding the deficit in the profit and loss) were INR 5 lakhs (2019: INR 5 lakhs) correspondingly, the Company's accumulated losses aggregated INR 8.60 lakhs (2019: INR 8.01 lakhs) and the Company's current liabilities exceeded its current assets by INR 3.60 lakhs. These factors give rise to a material uncertainty whether the Company would be able to continue as a going concern. However, in pursuance to Note No. 1 (b) above, the financial statements have been prepared on a going concern basis. Accordingly, these financial statements do not include any adjustments relating to the recoverability and classification of recorded assets, or to amounts and classification of recorded liabilities that might be necessary if the Company is unable to continue as a going concern.

## c) Historical cost convention:

The financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

1 Financial instruments measured at fair value through profit or loss, if applicable

2 Financial instruments measured at fair value through other comprehensive income, if applicable

### d) Use of estimates and judgements

In preparing these Ind AS financial statements, management has made judgements, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates.

Estimates and assumptions are required in particular for:

Determination of the estimated useful lives of tangible assets (including Investment Property) and the assessment as to which components of the cost may be capitalised.

Useful lives of tangible assets are based on the life prescribed in Schedule II of the Act. In cases, where the useful lives are different from that prescribed in Schedule II of the Act, they are based on management estimate, taking into account the nature of the asset, the estimated usage of the asset, the operating conditions of the asset, past history of replacement, anticipated technological changes, manufacturers' warranties and maintenance support. Assumptions also need to be made, when the Company assesses, whether an asset may be capitalised and which components of the cost of the asset may be capitalised.

· Measurement of fair values

The Company's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Company has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the holding company General Manager- Finance.

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorised into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

· Level 1: quoted prices (unadjusted) in active markets for identical assets or liabilities.

Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

. Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorised in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement.

## ii) Functional and presentation currency

The financial statements are presented in Indian rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian rupees (INR) and all amounts have been rounded-off to the nearest lakhs, except for share data and as otherwise stated. Due to rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.



RKH

## iii) Borrowing cost

Borrowing costs directly attributable to the acquisition, construction or production of an asset that necessarily takes a substantial period of time to get ready for its intended use or sale are capitalised as part of the cost of the asset. All other borrowing costs are expensed in the period in which they occur. Borrowing costs consist of interest and other costs that an entity incurs in connection with the borrowing of funds. Borrowing cost also includes exchange differences to the extent regarded as an adjustment to the borrowing costs.

### iv) Income tax

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised directly in equity or in OCI.

### a) Current tax

Current tax comprises the expected tax payable or receivable on the taxable income or loss for the year and any adjustment to the tax payable or receivable in respect of previous years. It is measured using tax rates enacted or substantively enacted at the reporting date. Current tax also includes any tax arising from dividends. Management periodically evaluates positions taken in tax returns with respect to situations in which applicable tax regulation is subject to interpretation. It establishes provisions where appropriate on the basis of amounts expected to be paid to the tax authorities.

Minimum Alternate Tax ('MAT') under the provisions of Income-tax Act, 1961 is recognised as current tax in the statement of profit and loss. MAT paid in accordance with the tax laws, which gives future economic benefits in the form of adjustment to future income tax liability, is considered as an asset if there is a convincing evidence that the Company will pay normal tax. Accordingly, MAT is recognised as an asset in the balance sheet when it is probable that the future economic benefit associated with it will flow to the Company.

Current tax assets and liabilities are offset only if, the Company:

- 1 has a legally enforceable right to set off the recognised amounts; and
- 2 intends either to settle on a net basis, or to realise the asset and settle the liability simultaneously.

### b) Deferred tax

Deferred tax is recognised in respect of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for taxation purposes. However, deferred tax liabilities are not recognised if they arise from the initial recognition of goodwill. Deferred income tax is also not accounted for if it arises from initial recognition of an asset or liability in a transaction other than a business combination that at the time of the transaction affects neither accounting profit nor taxable profit (tax loss).

Deferred tax assets are recognised for unused tax losses, unused tax credits and deductible temporary differences to the extent that it is probable that future taxable profits will be available against which they can be used. Deferred tax assets are reviewed at each reporting date and are reduced to the extent that it is no longer probable that the related tax benefit will be realised; such reductions are reversed when the probability of future taxable profits improves. Unrecognised deferred tax assets are reassessed at each reporting date and recognised to the extent that it has become probable that future taxable profits will be available against which they can be used.

Deferred tax is measured at the tax rates that are expected to be applied to temporary differences when they reverse, using tax rates enacted or substantively enacted at the reporting date and are expected to apply when the related deferred income tax asset is realised or the deferred income tax liability is settled.

Deferred tax assets and liabilities are offset only if:

1 the entity has a legally enforceable right to set off current tax assets against current tax liabilities; and

2 the deferred tax assets and the deferred tax liabilities relate to income taxes levied by the same taxation authority on the same taxable entity.

### v) Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity.

## Non-derivative financial liabilities

A financial liability is (i) a contractual obligation to deliver cash or another financial asset to another entity; or to exchange financial instruments under potentially unfavourable conditions; (ii) or a contract that will or may be settled in the entity's own equity instruments and is a non-derivative for which the entity is or may be obliged to deliver a variable number of its own equity instruments; or a derivative that will or may be settled other than by the exchange of a fixed amount of cash or another financial asset for a fixed number of the entity's own equity instruments.

### Recognition, measurement and classification

A financial liability is recognised in the balance sheet only when the Company becomes party to the contractual provisions to the instrument. The Company initially recognises debt securities issued and subordinated liabilities on the date that they are originated. All other financial liabilities (including liabilities designated at fair value through profit or loss) are recognised initially on the trade date at which the Company becomes a party to the contractual provisions of the instrument.

Financial liabilities are classified as either held at a) fair value through profit or loss. or b) at amortised cost. Management determines the classification of its financial liabilities at the time of initial recognition or, where applicable, at the time of reclassification.

The Company has the following non-derivative financial liabilities: loans and borrowings and trade and other payables. Such financial liabilities are recognised initially at fair value plus any directly attributable transaction costs. Subsequent to initial recognition these financial liabilities are measured at amortised cost using the effective interest method.

### De-recognition and offsetting

The Company derecognises a financial liability when its contractual obligations are discharged or cancelled or expire.

### vi) Provisions

A provision is recognised in the balance sheet when the Company has a legal or constructive obligation as a result of a past event, and it is probable that an outflow of economic benefits will be required to settle the obligation. If the effect is material, provisions are determined by discounting the expected future cash flows at a pre-tax rate that reflects current market assessments of the time value of money and, where appropriate, the risks specific to the liability.

vii) Earnings per share

RKH

TRIF Gurgaon Housing Projects Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

The Company presents basic and diluted earnings per share (EPS) data for its ordinary shares. Basic EPS is calculated by dividing the profit or loss attributable to ordinary shareholders of the Company by the weighted average number of ordinary shares outstanding during the period, adjusted for own shares held. Diluted EPS is determined by adjusting the profit or loss attributable to ordinary shareholders and the weighted average number of ordinary shareholders and the weighted average number of ordinary shares outstanding, adjusted for own shares held, for the effects of all dilutive potential ordinary shares.

## viii) Cash and cash equivalents



Cash and cash equivalent comprise of cash on hand and at banks including short-term deposits with an original maturity of three months or less, which are subject to an insignificant risk of changes in value.





TRIF Gurgaon Housing Projects Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### 3 Cash and cash equivalents

Particulars	31 March 2020	31 March 2019	
Cash and Cash Equivalents			
Balances with Bank	0.12	0.12	
Total	0.12	0.12	

## 4 Equity Share Capital

(a

up:			
31 March	31 March 2020		2019
No of Shares	Amount	No of Shares	Amount
50.000	5.00	50,000	5.00
50,000	5.00	50,000	5.00
and the second s		and the second	
50,000	5.00	50,000	5.00
	5.00		5.00
	31 March No of Shares 50,000 50,000	31 March 2020           No of Shares         Amount           50,000         5.00           50,000         5.00           50,000         5.00           50,000         5.00	31 March 2020         31 March           No of Shares         Amount         No of Shares           50,000         5.00         50,000           50,000         5.00         50,000           50,000         5.00         50,000           50,000         5.00         50,000           50,000         5.00         50,000

### (b) Reconciliation of Number of Shares Outstanding

Particulars	31 March	2020	31 March 2019	
	No of Shares	Amount	No of Shares	Amount
As at the beginning of the year	50,000	5.00	50,000	5.00
Add: Issued during the year				
As at the end of the year	50,000	5.00	50,000	5.00

### (c) Terms and rights attached to the equity share

The Company has only one class of equity shares having par value of INR 10 per share. Each holder of equity shares is entitled to one vote per share. The Company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject the approval of shareholders in the ensuing Annual General Meeting.

During the year ended 31 March 2020, the amount of per share dividend recognised as distribution to equity shareholders is Rs Nil (2019; Rs Nil). In the event of liquidation of the Company, the holders of the equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the share holders.

### (d) Shares of the company held by the Holding company

	Name of Shareholder	31 March	31 March 2020		31 March 2019	
	Name of Shareholder	No of Shares	Amount	No of Shares	Amount	
	Equity shares of INR 10 each, fully paid-up by	12.1.102	T.C.S.	517.8471	100	
	Tata Realty and Infrastructure Limited	50,000	5.00	50.000	5.00	
(e)	Details of shareholding more than 5% in the Co	mpany	2019			
(e)		mpany 31 March	2020	31 March	2019	
(e)	Details of shareholding more than 5% in the Co Name of Shareholder		1 2020 % Holding	31 March No of Shares	2019 % Holding	
		31 March		COLOR DE LA CALENCE COLOR DE LA CALENCIA DE LA CALE		

### (f) Aggregate number of bonus shares issued and shares issued for consideration other than cash during the period of five years immediately preceding the reporting date :

During the five year period ended 31 March 2020, the company has not issued any equity shares for consideration other than cash.

### 5 Other equity

6

7

Particulars	31 March 2020	31 March 2019	
Retained earnings	(8.60)		(8.01
TOTAL	(8.60)		(8.01)
Retained earnings			
Particulars	31 March 2020	31 March 2019	
(Deficit) in statement of profit and loss			
Balance at the beginning of the year	(8.01)		(6.84)
Add: (Loss) for the year	(0.59)		(1.17
Balance at the end of the Year	(8.60)		(8.01)
rrent Financial Liabilities - Borrowings			
The borrowings are analysed as follows :			
Particulars	31 March 2020	31 March 2019	
Interest-free loans from related parties			
Tata Realty and Infrastructure Limited	3.72		2.00
TOTAL	3.72		2.00
The above amount includes	and the second		10
Secured Borrowings			1948
Unsecured Borrowings	3.72		2.00
her Current Financial Liabilities			
Particulars	31 March 2020	31 March 2019	
Other Payables			_
- Related Party	2	-	
- Others	2	1.13	
The second s			1.13
Total	1747		1.13

## 8 Other Expenses

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Audit Fees (refer note 8(a))	0.59	0.91
Bank Charges		0.01
Fees & Consultations		0.20
Rates & Taxes ( incl indirect taxes)		0.05
Total	0.59	1,17
RKU	197	



## TRIF Gurgaon Housing Projects Private Limited Notes to the Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### (a) Remuneration to Statutory Auditors

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019		
Statutory audit	0.50	0.50		
Other Services			0.32	
Goods and Services Tax	0.09		0.09	
Total	0.59		0.91	

## 9 Deferred Tax Assets

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Current income tax		
Origination and reversal of temporary differences	-	
Reduction in tax rate		-
Recognition of previously unrecognised tax losses	1	-
Change in recognised deductible temporary differences		Ŷ.
Total deferred tax expense/(benefit)		
Tax expense for the year		
(b) Reconciliation of effective tax rate		
Loss before tax	(0.59)	(1.17
Tax using the Company's domestic tax rate 25% (p.y.25%)	(0.15)	(0.29
Tax effect of:		
Current-year losses for which no deferred tax asset is recognised	0.15	0.29
Total		
(c) Unrecognised deferred tax assets		

Deferred tax assets have not been recognised in respect of the following items, because of no virtual certainty of the profit in the future against which the deferred tax asset created can be utilised.

		Gross amount	Unrecognised tax effect	Gross amount	Unrecognised tax effect
Tax losses		9.76	2.44	8.59	2.15
(d) Tax losses carried	I forward				
Particulars	Financial Year	31 Mar	ch 2020	31 Marc	ch 2019
		Gross amount	Expiry date	Gross amount	Expiry date
Business loss	2011-12	0.36	2019-20	0.36	2019-20
Business loss	2012-13	0.30	2020-21	0.30	2020-21
Business loss	2013-14	4.12	2021-22	4.12	2021-22
Business loss	2014-15	0.39	2022-23	0.39	2022-23
Business loss	2015-16	1,42	2023-24	1.42	2023-24
Business loss	2016-17	1.33	2024-25	1.33	2024-25
Business loss	2017-18	0.66	2025-26	0.66	2025-26
Business loss	2018-19	1,17			0.00
Total		9.76		8.59	

The Company offsets tax assets and liabilities if and only if it has a legally enforceable right to set off current tax assets and current tax liabilities and the deferred tax assets and deferred tax liabilities relate to income taxes levied by the same tax authority.

Significant management judgement is required in determining provision for income tax, deterred income tax assets and liabilities and recoverability of deferred income tax assets. The recoverability of deferred income tax assets is based on estimates of taxable income and the period over which deferred income tax assets will be recovered. Any changes in future taxable income would impact the recoverability of deferred tax assets.

10 Due to micro and small suppliers Under the Micro. Small and Medium Enterprises Development Act. 2006, (MSMED) which came into force from 2 October 2006, certain disclosures are required to be made relating to Micro. Small and Medium enterprises. On the basis of the information and records available with the Management, there are no outstanding dues to the Micro, Small and Medium enterprises as defined in the Micro, Small and Medium Enterprises Development Act, 2006 as set out in following disclosure.

Particulars	31 March 2020	31 March 2019
Principal amount remaining unpaid to any supplier as at the year-end		•
Interest due thereon		
The amount of interest paid by the buyer as per the Micro Small and Medium Enterprises Development Act. 2006 (MSMED Act. 2006)		1
The amount of interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under MSMED Act, 2006 The amount of interest accrued and remaining unpaid at the end of each accounting year		
The amount of further interest remaining due and payable even in the succeeding years, until such date when the interest dues as above are actually paid to the small enterprise for the purpose of disallowance as a deductible expenditure under the MSMED Act, 2006		





### 11 Earning Per Share

Earnings Per Share (EPS) = Net Profit attributable to Shareholders / Weighted Number of Shares Outstanding

Particulars		For the year ended 31 March 2020	For the year ended 31 March 2019	
(Loss) after tax attributable to equity shareholders	A	(0.59)		(1.17)
Calculation of weighted average number of equity shares:				
Number of equity shares at the beginning of the year	C	50,000		50,000
Optionally convertible participatory debentures		¥		
Number of equity shares outstanding at the end of the year		50,000		50,000
Weighted average number of equity shares outstanding during the year	в	50,000		50.000
Earning Per Share - Basic (Rs.)	(A / B)	(1.18)		(2.34)
Earning Per Share - Diluted (Rs.)		(1.18)		(2.34)
sclosure of transactions with Related Parties, as requ (a) Related Party Disclosure <u>Ultimate holding company</u> Tata Sons Private Limited <u>Holding Company</u>	lired by Ir	idian Accounting Standard (Ind	AS) - 24	

b) Nature of Transactions / relationship / major parties	Holding Co	Total
Borrowings		
Tata Realty and Infrastructure Limited	1.72	1.72
	(2.00)	(2.00
Outstanding Balances Payable		1.4
Tata Realty and Infrastructure Limited	3.72	3.72
	(2.00)	12.00

(2.00) (2.00) All the transactions with related parties are at arm's length except the interest free loan from Holding company and all the outstanding balances are unsecured.

Note: figures in bracket represent previous years figures.

13 Segment Reporting as per IND AS108 " Operating Segments" Operating segments are reported in a manner consistent with the internal reporting provided to the chief operating decision maker (CODM). Chief operating decision maker's function is to allocate the resources of the entity and access the performance of the operating segment of the entity. The Board of directors of company assesses the linancial performance and position of the Company and makes strategic decisions. It is identified as being the chief operating decision maker for the company.

The company is operating in the real estate and infrastructure industry and has only domestic operations. The Company has only one reportable business segment. which is development of real estate and infrastructure facilities and only one reportable geographical segment. All assets of the Company are domiciled in India and the Company has no revenue from operation.

### 14 Other matters

Information with regard to other matters specified in Schedule III of the Act is either nil or not applicable to the Company for the year.

### 15 Capital commitment, contingencies and other commitments

There are no capital commitments and contingent liabilities as at the balance sheet date. Estimated amount of other commitments remaining to be executed on capital account and not provided for (net of advances) Rs Nil (2019: Rs Nil).

### 16 Financial instruments - Fair values and risk management

### A Credit risk

Currently, the Company is not exposed to any credit risk from trade receivables. Cash and cash equivalents

The Company held cash and cash equivalents with credit worthy banks of INR 0,12 and INR 0,12 as at 31 March 2020 and 31 March 2019 respectively. The credit worthiness of such banks is evaluated by the management on an ongoing basis and is considered to be good.

### **B** Liquidity risk

Liquidity risk is the risk that the Company will not be able to meet its financial obligations as they become due. The Company manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Company's reputation.

As at 31 March 2020, the Company had working capital of INR (3.60). As at 31 March 2019, the Company had working capital of INR (3.01).

### C Exposure to liquidity risk

The table below analyses the Company's financial liabilities into relevant maturity groupings based on their contractual maturities for:

all non derivative financial liabilities					
Particulars	Carrying amount	Total	1 year or less	1-2 years	2-5 years
As at 31 March 2020					
Non-derivative financial liabilities					
Unsecured short-term borrowings	3.72	3.72	3.72		
Other financial liabilities					1
	3.72	3.72	3.72		1
As at 31 March 2019					
Non-derivative financial liabilities					
Unsecured short-term borrowings	2.00	2.00	2.00	22	
Other financial liabilities	1.13	1.13	1.13	41	
	3.13	3.13	3.13		

Market risk D

Market risk is the risk that changes in market prices - such as foreign exchange rates, interest rates and equily prices - will affect the Company's income or the value of its holdings of financial instruments. The Company is not exposed to any significant currency risk and equity price risk.

REU



TRIF Gurgaon Housing Projects Private Limited Notes to the Ind AS financial statements

for the year ended 31 March 2020

(Currency: Indian rupees in lakhs) E Interest rate risk

Currently, the Company is not exposed to any interest rate risk on any financial assets and liabilities.

#### F Capital management

The Company's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. It sets the amount of capital required on the basis of annual business and long-term operating plans which include capital and other strategic investments.

The Company has adequate cash and bank balances and continues to remain debt-free. The company monitors its capital and makes a regular assessment of any debt requirements.

The Company monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total borrowings, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity other than amounts accumulated in the hedging reserve

The Company's adjusted net debt to equity ratio at March 31, 2020 was as follows.

Particulars	31 March 2020	31 March 2019
Total borrowings	3.72	2.00
Less : Cash and cash equivalent	0.12	0.12
Adjusted net debt	3.60	1.88
Adjusted equity	(3.60)	(3.01)
Adjusted net debt to adjusted equity ratio %	(1.00)	(0.62)

#### 17 Financial instruments - Fair values and risk management Accounting classification and fair values A

The following table shows the carrying amounts and fair values of financial assets and financial liabilities, including their levels in the fair value hierarchy. It does not include fair value information for financial assets and financial liabilities if the carrying amount is a reasonable approximation of fair value.

Particulars	FVTPL	FVTOCI	Amotised Cost	Totai	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Total
31 March 2020								
Financial assets								
Cash and cash equivalents	17		0.12	0.12			17.1	
Financial liabilities								
Interest-free loans from		24	3.72	3.72	1.1			
related parties			3.72	3.12				
Other current financial					6	100	10	
liabilities								
31 March 2019								
Financial assets								
Cash and cash equivalents	21	-	0.12	0.12		(e)		
Financial liabilities								
Interest-free loans from			2020					
related parties		<b>.</b>	2.00	2.00		•		
Other current financial								
liabilities		•	1.13	1.13				100

#### Financial risk management B

The Company has exposure to the following risks arising from financial instruments:

· Credit risk :

- Liquidity risk ; and Market risk

### C Risk management framework

The Company's board of directors has overall responsibility for the establishment and oversight of the Company's risk management framework.

#### Measurement of fair values D

### Valuation techniques and significant unobservable inputs

The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for financial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used.

Financial instruments measured at fair value

Туре	Valuation technique	Significant	Inter-relationship between
Amortised cost: 1. Borrowings	Discounted cash flow approach: The valuation model considers the present value of expected payments, discounted using a risk adjusted discount rate.		Not applicable

### Transfers between Levels 1 and 2

There have been no transfers between Level 1 and Level 2 during the reporting periods Level 3 fair values

There are no items in Level 3 fair values

18 There are no offsetting of financial assets and financial liabilities during the year.

### **19** Subsequent events

æ

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

REU

For and on behalf of the Board of Directors of TRIF Gurgaon Housing Projects Private Limited CIN: U74900DL2009PTC188404

N Amit Sheth Director

S.S. Mbadget

DIN - 07997663

Santosh Mhadgut Director DIN - 08049549

A

Chartered Accountants Lotus Corporate Park 1<sup>st</sup> Floor, Wing A – G CTS No.185/A, Jay Coach Off Western Express Highway Goregaon (East) Mumbai – 400 063 Maharashtra, India

Tel: +91 22 6245 1000 Fax: +91 22 6245 1001

## INDEPENDENT AUDITOR'S REPORT To The Members of Tata Realty and Infrastructure Limited Report on the Standalone Ind AS Financial Statements

## Opinion

We have audited the accompanying standalone Ind AS financial statements of **Tata Realty and Infrastructure Limited** ("the Company"), which comprise the Balance Sheet as at March 31, 2019, and the Statement of Profit and Loss (including Other Comprehensive Income), the Statement of Cash Flows and the Statement of Changes in Equity for the year then ended, and a summary of the significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid standalone Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended, ("Ind AS") and other accounting principles generally accepted in India, of the state of affairs of the Company as at March 31, 2019, and its loss, total comprehensive income, its cash flows and the changes in equity for the year ended on that date.

## **Basis for Opinion**

CHARTERED

è

We conducted our audit of the standalone Ind AS financial statements in accordance with the Standards on Auditing specified under section 143(10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the standalone Ind AS Financial Statements section of our report. We are independent of the Company in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the standalone Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the standalone Ind AS financial statements.

## Information Other than the Financial Statements and Auditor's Report Thereon

- The Company's Board of Directors is responsible for the other information. The other information comprises of Director's report, which we obtained prior to the date of this auditor's report.
- Our opinion on the standalone Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.

Regd. Office: Indiabulls Finance Centre, Tower 3, 27th - 32th Floor, Senapati Bapat Marg, Elphinstone Road (West), Mumbai - 400 013, Maharashtra, India. (LLP Identification No. AAB-8737)

- In connection with our audit of the standalone Ind AS financial statements, our responsibility is to read the other information and, in doing so, consider whether the other information is materially inconsistent with the standalone Ind AS financial statements or our knowledge obtained during the course of our audit or otherwise appears to be materially misstated.
- If, based on the work we have performed, we conclude that there is a material misstatement of this other information, we are required to reports this fact. We have nothing to report in this regard.

# Management's Responsibility for the Standalone Ind AS Financial Statements

The Company's Board of Directors is responsible for the matters stated in Section 134(5) of the Companies Act, 2013 (the "Act") with respect to the preparation of these standalone Ind AS financial statements that give a true and fair view of the financial position, financial performance including other comprehensive income, cash flows and changes in equity of the Company in accordance with the Ind AS and other accounting principles generally accepted in India.

This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the standalone Ind AS financial statements that give a true and fair view and are free from material misstatement, whether due to fraud or error.

In preparing the standalone Ind AS financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those Board of Directors are also responsible for overseeing the Company's financial reporting process.

# Auditor's Responsibility for the Audit of the Standalone Ind AS Financial Statements

Our objectives are to obtain reasonable assurance about whether the standalone Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these standalone Ind AS financial statements.



As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

- Identify and assess the risks of material misstatement of the standalone Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.
- Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under section 143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Company has adequate internal financial controls system in place and the operating effectiveness of such controls.
- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the standalone Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.
- Evaluate the overall presentation, structure and content of the standalone Ind AS financial statements, including the disclosures, and whether the standalone Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Materiality is the magnitude of misstatements in the standalone Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the standalone Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the standalone Ind AS financial statements.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

## Report on Other Legal and Regulatory Requirements

1. As required by Section 143(3) of the Act, based on our audit we report that:



- a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit.
- b) In our opinion, proper books of account as required by law have been kept by the Company so far as it appears from our examination of those books.
- c) The Balance Sheet, the Statement of Profit and Loss including Other Comprehensive Income, the Statement of Cash Flows and Statement of Changes in Equity dealt with by this Report are in agreement with the books of account.
- d) In our opinion, the aforesaid standalone Ind AS financial statements comply with the Indian Accounting Standards prescribed under Section 133 of the Act.
- e) On the basis of the written representations received from the directors of the Company as on March 31, 2019 taken on record by the Board of Directors, none of the directors is disqualified as on March 31, 2019 from being appointed as a director in terms of Section 164(2) of the Act.
- f) With respect to the adequacy of the internal financial controls over financial reporting of the Company and the operating effectiveness of such controls, refer to our separate Report in "Annexure A". Our report expresses an unmodified opinion on the adequacy and operating effectiveness of the Company's internal financial controls over financial reporting.
- g) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of section 197(16) of the Act, as amended, In our opinion and to the best of our information and according to the explanations given to us, the remuneration paid by the Company to its directors during the year is in accordance with the provisions of section 197 of the Act.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended, in our opinion and to the best of our information and according to the explanations given to us:
  - i. The Company has disclosed the impact of pending litigations on its financial position in Note 40 to its standalone Ind AS financial statements;
  - ii. The Company has made provision, as required under the applicable law or accounting standards, for material foreseeable losses, if any, on long-term contracts including derivative contracts;
  - iii. There were no amounts which were required to be transferred to the Investor Education and Protection Fund by the Company.



 As required by the Companies (Auditor's Report) Order, 2016 ("the Order") issued by the Central Government in terms of Section 143(11) of the Act, we give in "Annexure B" a statement on the matters specified in paragraphs 3 and 4 of the Order.

> For DELOITTE HASKINS & SELLS LLP Chartered Accountants (Firm's Registration No. 117366W/W-100018)

Ginchia

Kalpesh J. Mehta Partner (Membership No.48791)

Mumbai: May 08, 2019

## ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT (Referred to in paragraph 1(f) under 'Report on Other Legal and Regulatory Requirements' section of our report of even date)

# Report on the Internal Financial Controls Over Financial Reporting under Clause (i) of Sub-section 3 of Section 143 of the Companies Act, 2013 ("the Act")

We have audited the internal financial controls over financial reporting of Tata Realty and Infrastructure Limited ("the Company") as of March 31, 2019 in conjunction with our audit of the Ind AS financial statements of the Company for the year ended on that date.

## Management's Responsibility for Internal Financial Controls

The Company's management is responsible for establishing and maintaining internal financial controls based on the internal control over financial reporting criteria established by the Company considering the essential components of internal control stated in the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting issued by the Institute of Chartered Accountants of India. These responsibilities include the design, implementation and maintenance of adequate internal financial controls that were operating effectively for ensuring the orderly and efficient conduct of its business, including adherence to the company's policies, the safeguarding of its assets, the prevention and detection of frauds and errors, the accuracy and completeness of the accounting records, and the timely preparation of reliable financial information, as required under the Companies Act, 2013.

## Auditor's Responsibility

Our responsibility is to express an opinion on the Company's internal financial controls over financial reporting of the Company and its joint operations companies incorporated in India (retain as applicable) based on our audit. We conducted our audit in accordance with the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting (the "Guidance Note") issued by the Institute of Chartered Accountants of India and the Standards on Auditing prescribed under Section 143(10) of the Companies Act, 2013, to the extent applicable to an audit of internal financial controls. Those Standards and the Guidance Note require that we comply with ethical requirements and plan and perform the audit to obtain reasonable assurance about whether adequate internal financial controls over financial reporting was established and maintained and if such controls operated effectively in all material respects.

Our audit involves performing procedures to obtain audit evidence about the adequacy of the internal financial controls system over financial reporting and their operating effectiveness. Our audit of internal financial controls over financial reporting included obtaining an understanding of internal financial controls over financial reporting, assessing the risk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. The procedures selected depend on the auditor's judgement, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error.

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion on the Company's internal financial controls system over financial reporting.



# Meaning of Internal Financial Controls Over Financial Reporting

A Company's internal financial control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal financial control over financial reporting includes those policies and procedures that:

- pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company;
- b. provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorisations of management and directors of the company; and
- c. provide reasonable assurance regarding prevention or timely detection of unauthorised acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

# Inherent Limitations of Internal Financial Controls Over Financial Reporting

Because of the inherent limitations of internal financial controls over financial reporting, including the possibility of collusion or improper management override of controls, material misstatements due to error or fraud may occur and not be detected. Also, projections of any evaluation of the internal financial controls over financial reporting to future periods are subject to the risk that the internal financial control over financial reporting may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

## Opinion

In our opinion, to the best of our information and according to the explanations given to us, the Company has, in all material respects, an adequate internal financial controls system over financial reporting and such internal financial controls over financial reporting were operating effectively as at March 31, 2019, based on the criteria for internal financial control over financial reporting established by the Company considering the essential components of internal control stated in the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting issued by the Institute of Chartered Accountants of India.

## For Deloitte Haskins & Sells LLP

Chartered Accountants (Firm's Registration No.117366W/W-100018)

officiella

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai: May 08, 2019

# ANNEXURE B TO THE INDEPENDENT AUDITOR'S REPORT

(Referred to in paragraph 2 under 'Report on Other Legal and Regulatory Requirements' Section of our report of even date)

- (i) (a) The Company has maintained proper records showing full particulars, including quantitative details and situation of property, plant and equipment.
  - (b) The Company performs physical verification of its property, plant and equipment annually. In our opinion, the periodicity of physical verification is reasonable having regard to the size of the Company and the nature of its assets. According to the information and explanation given to us, no material discrepancies were noticed on such verification.
  - (c) According to the information and explanations given to us and on the basis of our examination of the records of the Company, the title deeds of immovable properties, as disclosed in Note 4 to the standalone Ind AS financial statements, are held in the name of the Company as at the balance sheet date.
- (ii) As explained to us, the inventories were physically verified during the year by the Management at reasonable intervals and no material discrepancies were noticed on physical verification between physical stock and the books of accounts.
- (iii) According to the information and explanations given to us, the Company has granted unsecured loans to eleven companies covered in the register maintained under Section 189 of the Companies Act, 2013, in respect of which:
  - (a) The terms and conditions of the grant of such loans are, in our opinion, prima facie, not prejudicial to the Company's interest.
  - (b) According to the information and explanations given to us, in respect of four unsecured loans, interest along with principal is repayable on demand and seven unsecured loans are interest free and the principal is repayable on demand. The Company has not demanded any loan during the year.
  - (c) There is no amount overdue for more than 90 days as at March 31, 2019.
- (iv) In our opinion and according to the information and explanations given to us, the Company has complied with the provisions of Sections 185 and 186 of the Companies Act, 2013 in respect of grant of loans, making investments and providing guarantees and securities, as applicable.
- (v) According to the information and explanations given to us, the Company has not accepted any deposit during the year as provided under Section 73 to 76 or any other relevant provisions of the Companies Act 2013. There are no unclaimed deposits any time during the year.



(vi) We have broadly reviewed the cost records maintained by the Company pursuant to the Companies (Cost Records and Audit) Rules, 2014, as amended prescribed by the Central Government under sub-section (1) of Section 148 of the Companies Act, 2013, and are of the opinion that, prima facie, the prescribed cost records have been made and maintained. We have, however, not made a detailed examination of the cost records with a view to determine whether they are accurate or complete.

(vii)According to the information and explanations given to us, in respect of statutory dues:

- (a) The Company has generally been regular in depositing undisputed statutory dues, including Income-tax, professional tax, provident fund, work contracts tax, labour cess, Goods & Services Tax, cess and other material statutory dues applicable to it to the appropriate authorities.
- (b) There were no undisputed amounts payable in respect of Income-tax, professional tax, provident fund, work contracts tax, labour cess, Goods & Services Tax, cess and other material statutory dues in arrears as at March 31, 2019 for a period of more than six months from the date they became payable.

Name of the Statute	Nature of Dues	Forum where dispute is pending	Period to which the amount relates	Amount involved	Amount Unpaid
Income Tax Act, 1961	Income Tax	Commissioner of Income Tax (Appeals) – Mumbai	FY 2014-2015	126,858,923	126,858,923
Finance Act, 1994	Service Tax	Commissioner of CGST & Central Excise – Mumbai	FY 2010-2011	26,721,775	26,721,775
Finance Act, 1994	Service Tax	Commissioner of CGST & Central Excise – Nagpur	FY 2010-11, FY 2011-12, FY 2012-13	8,837,820	79,54,038
Finance Act, 1994	Service Tax	Commissioner of CGST & Central Excise - Kochi	FY 2010-11, FY 2011-12, FY 2012-13	18,207,459	16,386,713
Finance Act, 1994	Kerala VAT	Commissioner of CGST & Central Excise - Kochi	FY 2014-2015	20,010,000	20,010,000

(c) Details of dues of Income-tax and Service Tax, which have not been deposited as on March 31, 2019 on account of disputes are given below:

(viii) In our opinion and according to the information and explanations given to us, the Company has not defaulted in the repayment of loans or borrowings from financial institutions and dues to debenture holders during the year. The Company did not have any outstanding dues to banks and government.



- (ix) The Company has not raised moneys by way of initial public offer or further public offer (including debt instruments) or term loans and hence reporting under clause (ix) of the CARO 2016 Order is not applicable.
- (x) To the best of our knowledge and according to the information and explanations given to us, no fraud by the Company and no material fraud on the Company by its officers or employees has been noticed or reported during the year.
- (xi) In our opinion and according to the information and explanations given to us, the Company has paid / provided managerial remuneration in accordance with the requisite approvals mandated by the provisions of Section 197 read with Schedule V to the Companies Act, 2013.
- (xii) The Company is not a Nidhi Company and hence reporting under clause (xii) of the CARO 2016 Order is not applicable.
- (xiii) In our opinion and according to the information and explanations given to us the Company is in compliance with Section 177 and 188 of the Companies Act, 2013, where applicable, for all transactions with the related parties and the details of related party transactions have been disclosed in the financial statements etc. as required by the applicable accounting standards.
- (xiv) According to the information and explanation given to us, during the year the Company has not made any preferential allotment or private placement of shares or fully or partly convertible debentures and hence reporting under clause (xiv) of CARO 2016 is not applicable to the Company.
- (xv) In our opinion and according to the information and explanations given to us, during the year the Company has not entered into any non-cash transactions with its directors or directors of its holding, subsidiary or associate company or persons connected with them and hence provisions of Section 192 of the Companies Act, 2013 are not applicable.
- (xvi) The Company is not required to be registered under Section 45-IA of the Reserve Bank of India Act, 1934.

## For Deloitte Haskins & Sells LLP

Chartered Accountants (Firm's Registration No. 117366W/W-100018)

Auchla

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai: May 08, 2019

## Tata Realty and Infrastructure Limited

Balance Sheet as at 31 March 2019 (Currency: Indian rupees in lakhs)

Particulars	Note No.	As at 31 March 2019	As at 31 March 2018
ASSETS			
NON-CURRENT ASSETS			
(a) Property, plant and equipment (PPE)	4	1,743.39	4 000 4
(b) Capital work-in-progress	4		1,396.4
(c) Intangible assets	4	3.76	
(d) Financial assets	4	50.43	58.5
(i) Investments	5		2 100 2000 0
(ii) Loans and advances	5	4,06,035.76	3,47,753.0
(iii) Others		52,124.48	37,553.9
(e) Current tax assets (net)	8	0.75	0.7
(f) Other non-current assets	9	6,012.90	5,370.8
TOTAL NON-CURRENT ASSETS	10 _	10,968.46 4,76,939.93	9,019.2 4,01,152.8
CURRENT ASSETS			4,01,102.0
(a) Inventories			
(b) Financial assets	11	41,011.06	41,741.53
(i) Investments			
	5	311.42	6,966.1
(ii) Trade and other receivables	6	1,397.81	3,132.9
(iii) Cash and cash equivalents	12	445.25	955.9
<ul><li>(iv) Other bank balances</li><li>(v) Loans and advances</li></ul>	13	444.69	361.6
	7	2,349.77	13,369.5
(vi) Others financial assets	8	2,861.19	2,908.0
(c) Other current assets	10	1,086.29	979.7
TOTAL CURRENT ASSETS		49,907.48	70,415.5
TOTAL ASSETS		5,26,847.41	4,71,568.38
EQUITY AND LIABILITIES			
(a) Equity share capital	14	1,01,730.77	1,01,730.77
(b) Other equity	15	96,955.18	88,015.4
TOTAL EQUITY		1,98,685.95	1,89,746.1
IABILITIES			
NON-CURRENT LIABILITIES			
(a) Financial liabilities			
(i) Long-term borrowings	16	72,460,15	1,82,310.9
(ii) Other financial liabilities	18	11,425.58	4,931.2
(b) Long-term provisions	19	685.48	4,951.2
<ul><li>(c) Current tax liabilities (net)</li></ul>	20	1,751.88	1,751.88
(d) Deferred tax liabilities (net)	21	13,258.38	10,470.12
TOTAL NON-CURRENT LIABILITIES		99,581.47	2,00,319.99
CURRENT LIABILITIES			
(a) Financial liabilities			
(i) Short-term borrowings	16	94,685,15	59,694,74
(ii) Trade and other payables other than MSME	17	3,458.03	
(iii) Other financial liabilities	17		3,025.61
(b) Other current liabilities	22	1,25,092.20	14,407.41
(c) Short term provisions	19	5,119.48	4,281.18
TOTAL CURRENT LIABILITIES	19 -	225.13 2,28,579.99	93.28
TOTAL EQUITY AND LIABILITIES			
	-	5,26,847.41	4,71,568.38
ignificant accounting policies	1-3		
otes to the standalone Ind AS financial statements	4 4-7		

Notes to the standalone Ind AS financial statements

4-47 The accompanying notes 1 to 47 form an integral part of these standalone Ind AS financial statements.

As per our report of even date attached For Deloitte Haskins & Sells LLP **Chartered Accountants** (Firm's Registration No. 117366W / W-100018)

Kalpesh J. Mehta Partner

Mumbai Dated : 8 May, 2019

I. N. huped F. N. Subedar Director

2CM Neera Saggi

DIN - 00501029

211

Director

DIN - 00028428

Banmuli Agrawala Chairman DIN - 00120029 Sab

For and on behalf of the Board of Directors of

Tata Realty and Infrastructure Limited

CIN No: U70102MH2007PLC168300

Rajiv Sabharwal Director DIN - 00057333

608

N an Sanjay Sharma

Chief Financial Officer

Sanjay Dutt Managing Director DIN - 05251670

1 Aull V en

S. Santhanakrishnan Director DIN - 00032049

Vinay Gaokar **Company Secretary** Membership No: A6120

Mumbai Dated : 8 May, 2019

# Tata Realty and Infrastructure Limited

Statement of Profit and Loss for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

Part	iculars	Note No.	For the year ended 31 March 2019	For the year ended 31 March 2018
1	Revenue from operations			
п	Other income	23	14,493.29	13,202.18
Ш	Total Income (I +II)	24	12,753.95	5,423.20
			27,247.24	18,625.38
IV	Expenses:			
	Cost of sale of flats			
	Employee benefits expense	25	6,122.91	4,496.96
	Finance costs	26	5,585.39	5,031.40
	Depreciation and amortization expense	27	22,988.91	19,671.90
	Loss on fair valuation of derivative contracts	28	140.30	145.87
	Other expenses	29	440.00	2,854.76
	Amounts written off during the year	30	3,883.25	3,744.21
	Total Expenses	31	5,414.38	117.24
			44,575.14	36,062.34
۷	(Loss) before tax (III-IV)		(17 007 00)	
VI	Tax expenses	32	(17,327.90)	(17,436.96
	Current Tax	52		
	Deferred Tax charge		(1 000 000	1.1.1.1.1.1
	Current tax expenses relating to prior years		(1,298.87)	(331.59)
	Total tax expenses		(1,298.87)	(265.61)
	0		(1,290.07)	(597.20)
VII	(Loss) for the year (V-VI)	-	(18,626.77)	(18,034.16)
VIII	Other Comprehensive Income:		(10,020.77)	(10,034.10)
	A. (i) Items that will not be reclassified to profit or loss			
	Remeasurements of defined benefit liability		52.65	21.64
	Equity instruments fair valued through OCI		32,334.60	1.624.18
	Income tax relating to items that will not be reclassified to profit or loss		(3,516.56)	(175.95)
	B. (i) Items that will not be reclassified to profit or loss		(0,010.00)	(175.95)
	Income tax relating to items that will be reclassified to profit or loss			
IX	Total Comprohensive Draft / //		28,870.69	1,469.87
	Total Comprehensive Profit / (Loss) for the Year (VII+VIII)		10,243.92	(16,564.29)
х	Earnings per equity share (Face value of INR 10 each)	24		
	Basic	34	1. 194	
	Diluted		(1.83)	(1.77)
gnif	icant accounting policies		(1.83)	(1.77)
otes	to the standalone Ind AS financial statements	1-3		
ne ac	companying notes 1 to 47 form an integral part of these standalone lad AS frameial	4-47		

The accompanying notes 1 to 47 form an integral part of these standalone Ind AS financial statements.

As per our report of even date attached For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No. 117366W / W-100018)

Q CB

Kalpesh J. Mehta Partner

Mumbai Dated : 8 May, 2019

Banmali Agrawala Chairman DIN - 00120029

**Rajiv Sabharwal** 

DIN - 00057333

Sabl

For and on behalf of the Board of Directors of

Tata Realty and Infrastructure Limited

CIN No: U70102MH2007PLC168300

Sanjay Dutt Managing Director DIN - 05251670

mbedal I M 1

F. N. Subedar Director DIN - 00028428

P

Na

Δ

Director

Neera Saggi Director DIN - 00501029

Sanjay Sharma **Chief Financial Officer** 

Nalu

S. Santhanakrishnan Director DIN - 00032049

Vinay Gaokar Company Secretary Membership No: A6120

Mumbai Dated : 8 May, 2019

# Tata Realty and Infrastructure Limited

Statement of Cash Flow for the Year ended 31 March 2019

(Currency: )	ndian	rupees	in	lakt
				and the second se

Particulars		For the Year Ended 31 March 2019		For the Year Ended 31 March	
A CASH FLOW FROM OPERATING ACTIVITIES				201	8
Loss before tax			147 207 001		and the best
Adjustments for :			(17,327.90)		(17,436.96
Depreciation and amortisation expense		440.00			
Loss on sale of PPE		140.30		145.87	
(Gain) on sale of current investments		23.17		11.26	
(Gain) / loss on fair valuation of investments and derivative instruments		(263.43)		(160.97)	
Interest Income		(5,685.50)		2,854,76	
Unwinding of advances		(5,537.52)		(4,102.00)	
Finance costs		(540.44)		(597.13)	
		22 988 91		19,671.90	
Amounts written off during the year		5 414.38			
Provision for employee benefits		14.15	10 55 4 00	117.24	
Operating (loss) / profit before working capital changes		14.15	16,554.02 (773.88)	102.10	18,043.03
Changes in working capital			(		000.07
Decrease in trade receivables					
Decrease in inventories		430.97		555.47	
Decrease in all ender		730.47		1,921,41	
Decrease in advances, other current assets and other non-current assets		1,418.20		72.93	
(Increase) / Decrease in trade payables, other financial liabilities and other		1.530.62			
financial liabilities		1,000.02		(2,118.38)	
Cash flows generated from operating activities			4,110.26		431.43
Tax refund / (paid) during the year (net of payments)			3,336.38	_	1.037.50
Net cash flows cancented further year (net of payments)	10.5		(642.09)		496.21
Net cash flows generated from operating activities	Α.		2,694.29		1,533.71
3 CASH FLOW FROM INVESTMENT ACTIVITIES					
Payment for purchase of Property, plant & equipment					
Proceeds on sale of Property, plant & equipments		(538.52)		(126.20)	
Investment in Final days in the equipments		32.51		3.09	
Investment in Fixed deposits under lien with maturity		(00.00)		0.00	
less than 12 months (net)		(83.02)		(21.66)	
Investment in subsidiaries and joint venture companies		(14,072.95)		(23,906.45)	
Investment in mutual fund		(1,45,427.98)			
Proceeds from sale of investments in mutual fund		1,52,366.17		(22,336.09)	
Inter-corporate deposit refunded				35,184.84	
Inter-corporate deposit given		7,490.33		21,885.65	
Interest Received		(24,717.17)		(16,080.00)	
Net cash flows (used in) Investing Activities	в	3,115.67	101 00 1 001	5,195.22	_
	-		(21,834.96)		(201.60)
CASH FLOW FROM FINANCING ACTIVITIES					
Proceeds from long-term borrowings				32,500,00	
Proceeds from short-term borrowings		35,245.66		1,39,500.00	
Repayment of short-term borrowings					
Finance costs paid		(16.615.70)		(1,58,500.00) (14,861.63)	
Net cash flows generated from Financing Activities		(		(14,001.03)	
5 Annual Annual Annual Annual	c		18,629.96		(1,361.63)
Net decrease in cash and cash equivalents (A	+B+C)				
			(510.71)		(29.52)
Cash and cash equivalents at the beginning of the year			955.96		985.48
Cash and cash equivalents at the end of year	-		445.25		955.96
Cash and bank balances at the end of the year comprise of:					000.00
Cash on Hand					
					0.61
Balances with Bank			445.25		
Deposit Accounts with less than or equal to 3 months maturity			440.20		755.35
Total Balance	-				200.00
			445.25		955.96

The Cash flow statement has been prepared under the indirect method as set out in Indian Accounting Standard - 7 ('Ind AS 7') on Cash Flow Statement prescribed in Companies (Indian Accounting Standard) Rules, 2015, notified under Section 133 of the Companies Act, 2013.

Significant accounting policies Notes to the standalone Ind AS financial statements

4-47 The accompanying notes 1 to 47 form an integral part of these standalone Ind AS financial statements.

As per our report of even date attached For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No. 117366W / W-100018)

Kalpesh J. Mehta Partner

Mumbai Dated : 8 May, 2019

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

Law

hickedar

F. N. Subedar

DIN - 00028428

DIN - 00501029

213

Director

pà Neera Saggi

Director

Banmali Agrawala Chairman DIN - 00120029

1-3

Sabhar day

Rajiv Sabharwal Director

DIN - 00057333

njay Sharma Chief Financial Officer

Vinay Gaokar Company Secretary Membership No: A6120

Sanjay Dutt

S

Managing Director DIN - 05251670

a

DIN - 00032049

Director

S. Santhanakrishnan

Mumbai Dated : 8 May, 2019

## Tata Realty and Infrastructure Limited Statement of Changes in Equity (SOCIE)

for the year ended 31 March 2019

(Currency: Indian rupees in lakhs)

## A Equity Share Capital

Particulars	31 March	2019	31 March 2018		
	Number of Shares	Amount	Number of Shares	Amount	
Subscribed and Fully Paid up Capital			ondres		
Equity shares of INR 10 each					
Opening Balance	4 04 70 07 000	111000-00-00			
Changes in equity share capital during the year	1,01,73,07,692	1,01,730,77	1,01,73,07,692	1,01,730.77	
Closing Balance			and the second		
citesting bulance	1,01,73,07,692	1,01,730.77	1,01,73,07,692	1,01,730.77	

## B Other equity

Particulars	Reserves and surplus			Items of Other comprehensive income		
	Capital Reserve	Securities Premium	Retained Earnings	Defined benefit plan adjustment	Equity Instruments through Other Comprehensive Income reserve	Total equity
Balance as at 01 April 2017	4,783.49	15,769.23	(500.87)	(31.49)	84,559,33	1,04,579.69
Loss for the year	e.	÷	(18,034.16)		-	(18,034.16
Other comprehensive income for the year Equity Instruments fair valued through Other Comprehensive Income	n chan	-			1,624.18	1,624.18
Remeasurements of defined benefit liability (asset)				21.64		
Income tax relating on above	1.					21.64
Conversion of 5% non-cumulative convertible preference shares into equity shares.			806.49	(8.93)	(167.02) (806.49)	(175.95
Balance as at 31 March 2018	4,783.49	15.769.23	(17,728.54)	110		
IND AS 115 Adjustment under transition (Refer Note 23)		10,100.20	(11,120.34)	(18,78)	85,210.00	88,015.40
			(1,304.14)			(1,304.14
Loss for the year		-	(18,626.77)			(18,626.77
Other comprehensive income for the year						(
Equity Instruments fair valued through Other Comprehensive Income	4	÷.				
Remeasurements of defined benefit liability (asset)					32,334.60	32,334.60
ncome tax relating on above			+	52.65		52.65
Balance as at 31 March 2019	4,783.49	15,769.23	(37,659.45)	(13.68) 20.19	(3,502.88)	(3,516.56)

 Significant accounting policies
 1-3

 Notes to the standalone Ind AS financial statements
 4-47

 The accompanying poles 1 to 47 form on interval
 4-47

The accompanying notes 1 to 47 form an integral part of these standalone Ind AS financial statements.

As per our report of even date attached For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No. 117366W / W-100018)

Kalpesh J. Mehta Partner

Mumbai Dated : 8 May, 2019 For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

40

Banmali Agrawala Chairman

Lazon

utt

Sanjay Dutt Managing Director DIN - 05251670

Rajiv Sabharwal Director DIN - 00057333

DIN - 00 20029

NUS N O

Sanjay Sharma Chief Financial Officer

S. Santhanakrishnan Director DIN - 00032049

al

Vinay Gaokar Company Secretary Membership No: A6120

Mumbai Dated : 8 May, 2019

F. N. Subedar Director DIN - 00028428

1. N hubedal

Neera Saggi Director DIN - 00501029

Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

#### 1 Background of the Company

Tata Realty and Infrastructure Limited ('the Company') was incorporated on 2 March 2007 to carry on the business of investment advisory services, project management consultancy services and real estate and infrastructure development. The Company is a wholly owned subsidiary of Tata Sons Private Limited.

#### 2 Basis of preparation

## (a) Statement of compliance

These standalone Ind AS financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 ('the Act') and other relevant provisions of the Act.

## (b) Going Concern

- As at March 31, 2019, the Company is having :
- · accumulated losses of INR 37,659.45 Lakhs.
- current liabilities exceeding its current assets by INR 68,672.52 Lakhs.
- repayment of INR. 1,10,000 Lakhs pertaining to the current portion of the long term debt (NCD) due in next 12 months.
- cash losses during the current year and previous year.

All of the above may indicate doubt about the Company's ability to continue as a going concern.

The Board of Directors have assessed going concern ability considering the following factors:

· Negative working capital is on account of management decision to borrow short-term funds through commercial papers considering benefits of interest arbitrage under current economic scenario.

 The Company has credit ratings of AA and A1+ from two credit rating agencies which supports raising of long-term funds or refinance short-term loans on a need basis respectively. Based on the credit rating, the Company has a long-term borrowing limit of INR 80,000 lakhs and short-term borrowing limit of INR 1,00,000 lakhs. The Company has a term sheet from a bank for long term funding upto INR

 The Board of Directors has approved rights issue of INR 3,60,000 lakhs approximately through equity capital from the parent i.e. Tata Sons Private Limited. The Board of Directors are confident that the same will be fully subscribed and thereby the Authorized Capital is realigned to accommodate the above rights issue.

 The Board of Directors of the Company are actively considering disposal of certain projects which are not strategically important to the business and yield positive cashflows in the next 12 months.

Hence in the opinion of the Board of Directors, the Company is able to continue as going concern. Accordingly, the financial statements have been prepared on a going concern basis.

## (c) Functional and presentation currency

These financial statements are presented in Indian Rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian Rupess (INR) and all amounts have been rounded-off to the nearest lakhs with two decimals, except for share data and as otherwise stated. Due to rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.

## (d) Basis of measurement

The standalone Ind AS financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

- Certain financial assets and liabilities (including derivative instruments)
- 2 Defined benefit plans plan assets measured at fair value

## (e) Critical accounting judgements and key sources of estimation of uncertainty

In preparing these standalone Ind AS financial statements, management has made judgements, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates

Estimates and underlying assumptions are reviewed on an ongoing basis. Revisions to accounting estimates are recognised prospectively

## (i) Judgements

Information about judgements made in applying accounting policies that have the most significant effects on the amounts recognised in the standalone Ind AS financial statements is included in the following notes:

## (ii) Assumptions and estimation uncertainties

Information about assumptions and estimation uncertainties that have a significant risk of resulting in a material adjustment in the year ended 31 March 2019 is included in the following notes:

Note 41 - measurement of defined benefit obligations: key actuarial assumptions;

Note 42 - determining the fair value of investments on the basis of significant unobservable inputs.





## Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements for the year ended 31 March 2019

(Currency: Indian rupees in lakhs)

## (f) Measurement of fair values

The Company's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Company has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorized into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

Level 1: quoted prices (unadjusted) in active markets for identical assets or liabilities.

Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorized in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the

## (g) Foreign currency transactions

Foreign currency transactions are translated into the functional currency using the exchange rates at the dates of the transactions. Foreign exchange gains and losses resulting from the settlement of such transactions and from the translation of monetary assets and liabilities denominated in foreign currencies at year end exchange rates are generally recognised in profit or loss.

Non-monetary items that are measured at fair value in a foreign currency are translated using the exchange rates at the date when the fair value was determined. Translation differences on assets and liabilities carried at fair value are reported as part of the fair value gain or loss. The gain or loss arising on translation of non-monetary items measured at fair value is treated in line with the recognition of the gain or loss on the change in fair value of the item (i.e., translation differences on items whose fair value gain or loss is recognised in OCI or profit or loss are also recognised in OCI or profit or loss, respectively).

#### 3 Significant accounting policies

## 3.01 Revenue recognition

Revenue is recognised to the extent that it is probable that the economic benefits will flow to the Company and the revenue can be reliably measured

## Sale of completed property.

Revenue from sale of completed property (residential and commercial) is recognised when:

- 1. The Company has transferred to the buyer signicificant risk and rewards of ownership of the completed property;
- 2. The Company retains neither continuing managerial involvement to the degree usually associated with the ownership nor effective
- control over the completed property sold;
- 3. The amount of revenue can be measured reliably;
- 4. It is probable that the ecomonic benefit associated with the transaction will flow to the Company; and
- 5. Cost incurred or to be incurred in respect of the transaction can be measured reliably.

Asset management fees and Project management consultancy fees are recognized in accordance with terms of agreement with customers

A dividend is recognised as revenue when the right to receive payment has been established. For all debt instruments measured either at amortised cost or at fair value through other comprehensive income, interest income is recorded using the effective interest rate (EIR). EIR is the rate that discounts the estimated future cash payments or receipts over the expected life of the financial instrument or a shorter period, where appropriate, to the gross carrying amount of the financial asset or to the amortised cost of a financial liability.

## 3.02 Property, plant and equipment

## (i) Recognition and measurement

Freehold land is carried at historical cost. All other items of property, plant and equipment are measured at cost less accumulated depreciation and any accumulated impairment losses.

The cost of an item of property, plant and equipment comprises:

- a) its purchase price, including import duties and non-refundable purchase taxes, after deducting trade discounts and rebates.
- b) any costs directly attributable to bringing the asset to the location and condition necessary for it to be capable of operating in the manner intended by management.
- c) the initial estimate of the costs of dismantling and removing the item and restoring the site on which it is located, the obligation for which an entity incurs either when the item is acquired

Income and expenses related to the incidental operations, not necessary to bring the item to the location and condition necessary for it to be capable of operating in the manner intended by management, are recognised in profit or loss

If significant parts of an item of property, plant and equipment have different useful lives, then they are accounted for as separate items (major components) of property, plant and equipment. Any gain or loss on disposal of an item of property, plant and equipment is recognised in profit or loss.




#### (ii) Subsequent expenditure

Subsequent expenditure is capitalised only if it is probable that the future economic benefits associated with the expenditure will flow to the Company.

#### (iii) Depreciation

Depreciable amount for assets is the cost of an asset, or other amount substituted for cost, less its estimated residual value

Depreciation is provided using the straight line method in the manner and at the rates prescribed by Part 'C' of Schedule II of the Act. Depreciation is charged on a monthly pro-rata basis for assets purchased or sold during the year.

In the following cases, the useful life is less than the corresponding useful life prescribed in Part 'C' of Schedule II of the Act, based on internal technical evaluation, taking into account the nature of the assets, the estimated usage of the asset, the operating conditions of the asset, past history of replacement, anticipated technological changes etc.

Data Processing Natural - D	
Data Processing Networks - Servers and Networks Motor Car	5 years
Motor Car	5 years

Leasehold improvements are amortised over the primary period of the lease

Depreciation methods, useful lives and residual values are reviewed at each reporting date and adjusted if appropriate.

Gains and losses on disposals are determined by comparing proceeds with carrying amount. These are included in profit or loss within other gains/(losses).

Expenditure incurred on acquisition /construction of fixed assets which are not ready for their intended use at balance sheet date are disclosed under capital work-in-progress. Capital work in progress is stated at cost less impairment losses, if any. Cost comprises of expenditures incurred in respect of capital projects under development and includes any attributable / allocable cost and other incidental expenses. Revenues earned, if any, before capitalization from such capital project are adjusted against the capital work in progress.

Borrowing costs relating to acquisition / construction / development of tangible assets, intangible assets and capital work in progress which takes substantial period of time to get ready for its intended use are also included to the extent they relate to the period till such assets are ready to be put to use.

#### 3.03 Intangible assets

Intangible assets comprise application software purchased / developed, which are not an integral part of the related hardware and are amortised using the straight line method over a period of the software license, which in the Management's estimate represents the period during which the economic benefits will be derived from their use. Subsequent expenditure is capitalised only when it increases the future economic benefits embodied in the specific to which it relates.

#### 3.04 Impairment of non-financial assets

Goodwill and intangible assets that have an indefinite useful life are not subject to amortisation and are tested annually for impairment, or more frequently if events or changes in circumstances indicate that they might be impaired. Other assets are tested for impairment whenever events or changes in circumstances indicate that the carrying amount may not be recoverable. Management periodically assesses using, external and internal sources, whether there is an indication that an asset may be impaired.

The recoverable amount is higher of the asset's net selling price or value in use, which means the present value of future cash flows expected to arise from the continuing use of the asset and its eventual disposal. An impairment loss for an asset is reversed if, and only if, the reversal can be related objectively to an event occurring after the impairment loss was recognized. The carrying amount of an asset is increased to its revised recoverable amount, provided that this amount does not exceed the carrying amount that would have been determined (net of any accumulated amortization or depreciation) had no impairment loss been recognized for the asset in prior years.

#### 3.05 Borrowing cost

General and specific borrowing costs that are directly attributable to the acquisition, construction or production of a qualifying asset are capitalised during the period of time that is required to complete and prepare the asset for its intended use or sale. Qualifying assets are assets that necessarily take a substantial period of time to get ready for their intended use or sale.

Investment income earned on the temporary investment of specific borrowings pending their expenditure on qualifying assets is deducted from the borrowing costs eligible for capitalisation.

Other borrowing costs are expensed in the period in which they are incurred.

#### 3.06 Income-tax

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised directly in equity or in OCI.





#### (i) Current tax

Current tax comprises the expected tax payable or receivable on the taxable income or loss for the year and any adjustment to the tax payable or receivable in respect of previous years. It is measured using tax rates enacted or substantively enacted at the reporting date. Current tax also includes any tax arising from dividends. Management periodically evaluates positions taken in tax returns with respect to situations in which applicable tax regulation is subject to interpretation. It establishes provisions where appropriate on the basis of amounts expected to be paid to the tax authorities.

Minimum Alternate Tax ('MAT') under the provisions of Income-tax Act, 1961 is recognised as current tax in the statement of profit and loss. MAT paid in accordance with the tax laws, which gives future economic benefits in the form of adjustment to future income tax liability, is considered as an asset if there is a convincing evidence that the Company will pay normal tax. Accordingly, MAT is recognized as an asset in the balance sheet when it is probable that the future economic benefit associated with it will flow to the Company.

Current tax assets and liabilities are offset only if, the Company:

a) has a legally enforceable right to set off the recognised amounts; and

b) intends either to settle on a net basis, or to realise the asset and settle the liability simultaneously.

#### (ii) Deferred tax

Deferred tax is recognised in respect of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for taxation purposes. However, deferred tax liabilities are not recognised if they arise from the initial recognition of goodwill. Deferred income tax is also not accounted for if it arises from initial recognition of an asset or liability in a transaction other than a business combination that at the time of the transaction affects neither accounting profit nor taxable profit (tax loss).

Deferred tax assets are recognised for unused tax losses, unused tax credits and deductible temporary differences to the extent that it is probable that future taxable profits will be available against which they can be used. Deferred tax assets are reviewed at each reporting date and are reduced to the extent that it is no longer probable that the related tax benefit will be realised; such reductions are reversed when the probability of future taxable profits improves. Unrecognised deferred tax assets are reassessed at each reporting date and recognised to the extent that it has become probable that future taxable profits will be available against which they can be used.

Deferred tax is measured at the tax rates that are expected to be applied to temporary differences when they reverse, using tax rates enacted or substantively enacted at the reporting date and are expected to apply when the related deferred income tax asset is realised or the deferred income tax liability is settled.

Deferred tax liabilities are not recognised for temporary differences between the carrying amount and tax bases of investments in subsidiaries, branches and associates and interest in joint arrangements where the company is able to control the timing of the reversal of the temporary differences and it is probable that the differences will not reverse in the foreseeable future.

Deferred tax assets are not recognised for temporary differences between the carrying amount and tax bases of investments in subsidiaries, branches and associates and interest in joint arrangements where it is not probable that the differences will reverse in the foreseeable future and taxable profit will not be available against which the temporary difference can be utilised.

#### Deferred tax assets and liabilities are offset only if:

a) the entity has a legally enforceable right to set off current tax assets against current tax liabilities; and

b) the deferred tax assets and the deferred tax liabilities relate to income taxes levied by the same taxation authority on the same taxable entity.

#### (iii) Current and deferred tax for the year

Current and deferred tax are recognised in profit or loss, except when they relate to items that are recognised in other comprehensive income or directly in equity / deemed equity, in which case, the current and deferred tax are also recognised in other comprehensive income or directly in equity / deemed equity respectively. Where current tax or deferred tax arises from the initial accounting for a business combination, the tax effect is included in the accounting for the business combination.

#### 3.07 Cash and cash equivalents

For the purpose of presentation in the statement of cash flows, cash and cash equivalent includes cash on hand, deposits held at call with financial institutions, other short-term, highly liquid investments with original maturities of three months or less that are readily convertible to known amount of cash and which are subject to an insignificant risk of changes in value, and bank overdrafts. Bank overdrafts are shown within borrowings in current liabilities in the balance sheet.

#### 3.08 Inventories

Direct expenses like land cost, development rights, site labour cost, material used for project construction, cost of borrowing, project management consultancy, costs for moving the plant and machinery to the site and general expenses incurred specifically for the respective project like insurance, design and technical assistance, and construction overheads are taken as the cost of the project work in progress and cost of unsold flats.

Material at site comprise of building material, components, stores and spares, consumables

Inventories are valued at lower of cost or net realizable value, cost is determined on weighted average basis. Net realizable value is the estimated selling price in the ordinary course of business, less estimated costs of completion and estimated costs necessary to make the sale.





#### 3.09 Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity. Financial instruments also include derivative contracts such as put options, call options; and forward contracts.

#### (i) Financial assets

#### Classification

The Company shall classify financial assets as subsequently measured at amortised cost, fair value through other comprehensive income (FVOCI) or fair value through profit or loss (FVTPL) on the basis of its business model for managing the financial assets and the contractual cash flow characteristics of the financial asset.

#### Initial recognition and measurement

Financial assets are initially measured at fair value. Transaction costs that are directly attributable to the acquisition or issue of financial assets (other than fair valued through profit or loss) are added to the fair value of the financial assets, as appropriate, on initial recognition. Transaction costs directly attributable to the acquisition of financial assets fair valued through profit or loss are recognised immediately in profit or loss.

All regular way purchases or sales of financial assets are recognised and derecognised on trade date basis. Regular way purchases or sales are purchases or sales of financial assets that require delivery of assets within the time frame established by regulation or convention in the market place.

All recognised financial assets are subsequently measured in their entiretly at either amortised cost or fair value, depending on the classification of the financial assets.

#### **Debt instruments**

1, A 'debt instrument' is measured at the amortised cost if both the following conditions are met:

- a) The asset is held within a business model whose objective is to hold assets for collecting contractual cash flows, and
- b) Contractual terms of the asset give rise on specified dates to cash flows that are solely payments of principal and interest (SPPI) on the principal amount outstanding.

2. After initial measurement, such financial assets are subsequently measured at amortised cost using the effective interest rate (EIR) method. Amortised cost is calculated by taking into account any discount or premium and fees or costs that are an integral part of the EIR. The EIR amortisation is included in finance income in the profit or loss.

3, Debt instruments included within the fair value through profit and loss (FVTPL) category are measured at fair value with all changes recognized in the statement of profit and loss.

#### Equity instruments

The Company measures its equity investments in equity shares of subsidiaries, joint ventures and associates at fair value through other comprehensive income.

Equity investments in companies other than equity investments in subsidiaries, joint ventures and associates are measured at fair value through profit and loss account.

Where the company's management has elected to present fair value gains and losses on equity investments in other comprehensive income, there is no subsequent reclassification of fair value gains and losses to profit or loss. Dividends from such investments are recognised in profit or loss as other income when the company's right to receive payments is established.

#### **De-recognition**

A financial asset (or, where applicable, a part of a financial asset or part of a Company of similar financial assets) is primarily derecognised (i.e. removed from the Company's balance sheet) when:

The rights to receive cash flows from the asset have expired, or

The Company has transferred its rights to receive cash flows from the asset or has assumed an obligation to pay the received cash flows in full without material delay to a third party under a 'pass-through' arrangement; and either (a) the Company has transferred substantially all the risks and rewards of the asset, or (b) the Company has neither transferred nor retained substantially all the risks and rewards of the asset.

When the Company has transferred its rights to receive cash flows from an asset or has entered into a pass-through arrangement, it evaluates if and to what extent it has retained the risks and rewards of ownership. When it has neither transferred nor retained substantially all of the risks and rewards of the asset, nor transferred control of the asset, the Company continues to recognise the transferred asset to the extent of the Company's continuing involvement. In that case, the Company also recognises an associated liability. The transferred asset and the associated liability are measured on a basis that reflects the rights and obligations that the Company has retained.

Continuing involvement that takes the form of a guarantee over the transferred asset is measured at the lower of the original carrying amount of the asset and the maximum amount of consideration that the Company could be required to repay.

#### Impairment of financial assets

In accordance with Ind-AS 109, the Company applies expected credit loss (ECL) model for measurement and recognition of impairment loss on the following financial assets and credit risk exposure:

a) Financial assets that are debt instruments, and are measured at amortised cost e.g., loans, debt securities, deposits, and bank

balance

b) Lease receivables

c) Trade receivables

The Company follows 'simplified approach' for recognition of impairment loss allowance on: Trade receivables which do not contain a significant financing component.

All lease receivables resulting from transactions.





The application of simplified approach does not require the Company to track changes in credit risk. Rather, it recognises impairment loss allowance based on lifetime ECLs at each reporting date, right from its initial recognition.

For recognition of impairment loss on other financial assets and risk exposure, the Company determines that whether there has been a significant increase in the credit risk since initial recognition. If credit risk has not increased significantly, 12-month ECL is used to provide for impairment loss. However, if credit risk has increased significantly, lifetime ECL is used. If, in a subsequent period, credit quality of the instrument improves such that there is no longer a significant increase in credit risk since initial recognition, then the entity reverts to recognising impairment loss allowance based on 12-month ECL.

#### (ii) Financial liabilities

#### Classification

The Company classifies all financial liabilities as subsequently measured at amortised cost, except for financial liabilities at fair value through profit or loss. Such liabilities, including derivatives that are liabilities, shall be subsequently measured at fair value.

Debt and equity instruments issued by the Company are classified as either financial liabilities or equity in accordance with the substance of the contractual arrangements and the definitions of a financial liability and an equity instrument.

#### Initial recognition and measurement

Financial liabilities are classified, at initial recognition, as financial liabilities at fair value through profit or loss, loans and borrowings, payables, or as derivatives designated as hedging instruments in an effective hedge, as appropriate.

All financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

The Company's financial liabilities include trade and other payables, loans and borrowings including bank overdrafts and derivative financial instruments.

#### Financial liabilities at fair value through profit or loss

Financial liabilities at fair value through profit or loss include financial liabilities held for trading and financial liabilities designated upon initial recognition as at fair value through profit or loss. Financial liabilities are classified as held for trading if they are incurred for the purpose of repurchasing in the near term. This category also includes derivative financial instruments entered into by the Company that are not designated as hedging instruments in hedge relationships as defined by Ind-AS 109. The company does not have any separated embedded derivatives

Gains or losses on liabilities held for trading are recognised in the profit or loss.

Financial liabilities designated upon initial recognition at fair value through profit or loss are designated at the initial date of recognition, and only if the criteria in Ind-AS 109 are satisfied. For liabilities designated as FVTPL, fair value gains/ losses attributable to changes in own credit risk are recognized in OCI. These gains/loss are not subsequently transferred to profit and loss. However, the Company may transfer the cumulative gain or loss within equity. All other changes in fair value of such liability are recognised in the statement of profit or loss. The Company has not designated any financial liability as at fair value through profit and loss.

#### Loans and borrowings

After initial recognition, interest-bearing loans and borrowings are subsequently measured at amortised cost using the EIR method. Gains and losses are recognised in profit or loss when the liabilities are derecognized.

Amortised cost is calculated by taking into account any discount or premium on acquisition and fees or costs that are an integral part of the EIR. The EIR amortisation is included as finance costs in the statement of profit and loss. This category generally applies to interest-bearing loans and borrowings.

Preference shares, which are mandatorily redeemable on a specific date, are classified as liabilities. The dividends on these preference shares are recognised in profit or loss as finance costs

The fair value of the liability portion of an optionally convertible bonds is determined using a market interest rate for an equivalent nonconvertible bonds. This amount is recorded as a liability on an amortised cost basis until extinguished on conversion or redemption of the bonds. The remainder of the proceeds is attributable to the equity portion of the compound instrument. This is recognised and included in shareholders' equity, net of income tax effects, and not subsequently remeasured.

Where the terms of a financial liability are renegotiated and the entity issues equity instruments to a creditor to extinguish all or part of the liability (debt for equity swap), a gain or loss is recognised in profit or loss, which is measured as the difference between the carrying amount of the financial liability and the fair value of the equity instruments issued.

#### **De-recognition**

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit or loss.

#### Embedded derivatives

If the hybrid contract contains a host that is a financial asset within the scope of Ind-AS 109, the Company does not separate embedded derivatives. The Company applies the classification requirements contained in Ind AS 109 to the entire hybrid contract. Derivatives embedded in all other host contracts are accounted for as separate derivatives and recorded at fair value if their economic characteristics and risks are not closely related to those of the host contracts and the host contracts are not held for trading or designated at fair value through profit or loss. These embedded derivatives are measured at fair value with changes in fair value recognised in profit or loss, unless designated as effective hedging instruments. Reassessment only occurs if there is either a change in the terms of the contract that significantly modifies the cash flows.





#### Offsetting of financial instruments

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable right to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable right must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the company or the counterparty.

#### Derivative financial instruments

The Company has entered into derivative financial instruments, such as put and call option contracts and forward purchase contracts to acquire stake from Non-controlling interests. Such derivative financial instruments are initially recognised at fair value on the date on which a derivative contract is entered into and are subsequently re-measured at fair value through profit or loss account. Derivatives are carried as financial assets when the fair value is positive and as financial liabilities when the fair value is negative.

The company has not designated its derivatives as hedging instruments.

#### Financial guarantee contracts

A financial guarantee contract is a contract that requires issuer to make specified payments to reimburse the holder for a loss it incurs because a specified debtor fails to make payments when due in accordance with the terms of a debt instrument.

Financial guarantee contracts issued by the Company are initially measured at their fair values and, if not designated as at FVTPL, are subsequently measured at the higher of :

(i) the amount of loss allowance determined in accordance with impairment requirements of Ind AS 109; and

(ii)the amount initially recognised less, when appropriate, the cumulative amount of income recognised in accordance with the principles of Ind AS 18.

#### Commitments to provide a loan at a below-market interest rate

Commitments to provide a loan at a below-market interest rate are initially measured at their fair values and, if not designated as at FVTPL, are subsequently measured at the higher of :

(i) the amount of loss allowance determined in accordance with the impairment requirements of Ind AS 109; and

(ii)the amount initially recognised less, when appropriate, the cumulative amount of income recognised in accordance with the principles of Ind AS 18.

#### 3.10 Employee benefits

#### (i) Short term employee benefits

Short-term employee benefits are expensed as the related service is provided. A liability is recognised for the amount expected to be paid if the Company has a present legal or constructive obligation to pay this amount as a result of past service provided by the employee and the obligation can be estimated reliably.

#### (ii) Compensated absences

The liabilities for earned leave and sick leave are not expected to be settled wholly within 12 months after the end of the period in which the employees render the related service. They are therefore measured as the present value of expected future payments to be made in respect of services provided by employees up to the end of the reporting period using the projected unit credit method. The benefits are discounted using the market yields at the end of the reporting period that have terms approximating to the terms of the related obligation. Re-measurements as a result of experience adjustments and changes in actuarial assumptions are recognised in profit or loss.

#### (iii) Defined contribution plans

Obligations for contributions to defined contribution plans are expensed as the related service is provided. Prepaid contributions are recognised as an asset to the extent that a cash refund or a reduction in future payments is available.

#### (iv) Defined benefit plans

The Company's net obligation in respect of defined benefit plans is calculated separately for each plan by estimating the amount of future benefit that employees have earned in the current and prior periods, discounting that amount and deducting the fair value of any plan assets.

The calculation of defined benefit obligations is performed annually by a qualified actuary using the projected unit credit method. When the calculation results in a potential asset for the Company, the recognised asset is limited to the present value of economic benefits available in the form of any future refunds from the plan or reductions in future contributions to the plan. To calculate the present value of economic benefits, consideration is given to any applicable minimum funding requirements.

Re-measurement of the net defined benefit liability, which comprise actuarial gains and losses, the return on plan assets (excluding interest) and the effect of the asset ceiling (if any, excluding interest), are recognised immediately in other comprehensive income (OCI). Net interest expense (income) on the net defined liability (assets) is computed by applying the discount rate, used to measure the net defined liability (asset), to the net defined liability (asset) at the start of the financial year after taking into account any changes as a result of contribution and benefit payments during the year. Net interest expense and other expenses related to defined benefit plans are recognised in profit or loss.

When the benefits of a plan are changed or when a plan is curtailed, the resulting change in benefit that relates to past service or the gain or loss on curtailment is recognised immediately in profit or loss. The Company recognises gains and losses on the settlement of a defined benefit plan when the settlement occurs.

#### (v) Other long-term employee benefits

The Company's net obligation in respect of long-term employee benefits is the amount of future benefit that employees have earned in return for their service in the current and prior periods. That benefit is discounted to determine its present value. Re-measurement are recognised in profit or loss in the period in which they arise.

These plans typically expose the Company to actuarial risks such as : Investment risk, interest rate risk, longevity risk and salary risk:





> (i) Investment risk : The present value of the defined benefit plan liability is calculated using the discount rate which is determined by reference to market yields at the end of the reporting period on government bonds. For other defined benefit plans, the discount rate is determined by reference to market yields at the end of the reporting period on high quality corporate bonds when there is a deep market for such bonds; if the return on plan asset is below this rate, it will create a plan deficit. Currently, for the plan in India, it has a relatively balanced mix of investments in government securities and other debt instruments. Further, the overseas plan has a relatively balanced investment in equity securities, debt instruments and real estates. Due to the long-term nature of the plan liabilities, the board of the overseas fund considers it appropriate that a reasonable portion of the plan assets should be invested in equity securities and in real estate to leverage the return generated by the Fund.

> (ii) Interest risk : A decrease in the bond interest rate will increase the plan liability; however, this will be partially offset by an increase in the return on the plan's debt investments.

> (iii) Longevity risk : The present value of the defined benefit plan liability is calculated by reference to the best estimate of the mortality of plan participants both during and after their employment. An increase in the life expectancy of the plan participants will increase the plan's liability.

> (iv) Salary risk : The present value of the defined benefit plan liability is calculated by reference to the future salaries of plan participants. An increase in the Salary of the plan participants will increase the plan's liability.

#### 3.11 Earnings per share

Basic earnings per share is computed by dividing the profit / (loss) after tax by the weighted average number of equity shares outstanding during the year. The weighted average number of equity shares outstanding during the year is adjusted for the events for bonus issue, bonus element in a rights issue to existing shareholders, share split and reverse share split (consolidation of shares).

Diluted earnings per share is computed by dividing the profit / (loss) after tax as adjusted for dividend, interest and other charges to expense or income (net of any attributable taxes) relating to the dilutive potential equity shares, by the weighted average number of equity shares considered for deriving basic earnings per share and the weighted average number of equity shares which could have been issued on conversion of all dilutive potential equity shares.

#### 3.12 Recent Indian Accounting Standards (Ind AS)

Ministry of Corporate Affairs ("MCA"), through Companies (Indian Accounting Standards) Amendment Rules, 2019 and Companies (Indian Accounting Standards) Second Amendment Rules, has notified the following new and amendments to Ind ASs which the Group has not applied as they are effective from April 1, 2019:

#### Ind AS - 116

Ind AS 116 will replace the existing leases standard, Ind AS 17 Leases. Ind AS 116 sets out the principles for the recognition, measurement, presentation and disclosure of leases for both lessees and lessors. It introduces a single, on-balance sheet lessee accounting model for lessees. A lessee recognises right-of-use asset representing its right to use the underlying asset and a lease liability representing its obligation to make lease payments. The standard also contains enhanced disclosure requirements for lessees. Ind AS 116 substantially carries forward the lessor accounting requirements in Ind AS 17

The Company will adopt Ind AS 116, effective annual reporting period beginning April 1, 2019.

#### Ind AS 12 Income taxes (amendments relating to income tax consequences of dividend and uncertainty over income tax treatments)

The amendment relating to income tax consequences of dividend clarify that an entity shall recognise the income tax consequences of dividends in profit or loss, other comprehensive income or equity according to where the entity originally recognised those past transactions or events. The Company does not expect any impact from this pronouncement.

### Ind AS 109 – Prepayment Features with Negative Compensation

The amendments relate to the existing requirements in Ind AS 109 regarding termination rights in order to allow measurement at amortised cost (or, depending on the business model, at fair value through other comprehensive income) even in the case of negative The Compensation payments. The Company does not expect this amendment to have any impact on its financial statements.

#### Ind AS 19 - Plan Amendment, Curtailment or Settlement

The amendments clarify that if a plan amendment, curtailment or settlement occurs, it is mandatory that the current service cost and the net interest for the period after the re-measurement are determined using the assumptions used for the re-measurement. In addition, amendments have been included to clarify the effect of a plan amendment, curtailment or settlement on the requirements regarding the asset ceiling. The Company does not expect this amendment to have any significant impact on its financial statements.

#### Ind AS 23 - Borrowing Costs

The amendments clarify that if any specific borrowing remains outstanding after the related asset is ready for its intended use or sale, that borrowing becomes part of the funds that an entity borrows generally when calculating the capitalisation rate on general borrowings. The Company does not expect any impact from this amendment.





# 4 Property, Plant and Equipment , Capital work-in-progress and Intangible assets a Tangible Assets

Particulars	Freehold Land	Leasehold Improvements	Furniture & Fixtures	Data Processing Equipments	Office Equipments	Motor Vehicles	Total
COST / DEEMED COST				Equipmento			
As at 01 April 2017	1,144.90	-	157.78	280.95	166.15	74 42	4 004 04
Additions		8.97	17.64	18.74	100100	74.43	1,824.21
Disposals/Adjustments	1	0.01	(30.06)		55.19	6.91	107.45
As at 31 March 2018	1,144.90	8.97	145.36	(1.77) 297.92	(24.42) 196.92	(10.62)	(66.87)
		0.01	145.00	251.52	190.92	70.72	1,864.79
Additions		466.93	7.41	15.62	39.09		529.05
Disposals/Adjustments			(21.89)	(1.59)	(15.70)	(44.04)	
As at 31 March 2019	1,144.90	475.90	130.88	311.95	220.31	26.68	(83.22)
DEPRECIATION				011.00	220.31	20.00	2,310.62
As at 01 April 2017			77.11	199.52	113.15	19.15	408.93
Charge for the Year		1.51	23.74	41.86	27.56	0.000	
Disposals/Adjustments			(18.81)	(1.00)		17.23	111.90
As at 31 March 2018		1.51	82.04	240.38	(22.13)	(10.59) 25.79	(52.53)
			02.04	240.00	110.50	25.79	468.30
Charge for the Year		30.56	15.81	35.69	37.17	7.24	126.47
Disposals/Adjustments		00.00	(4.06)	(1.09)			
As at 31 March 2019		32.07	93.79		(4.77)	(17.62)	(27.54)
NET BLOCK		52.07	33.19	274.98	150.98	15.41	567.23
As at 31 March 2018	1,144.90	7.46	63.32	57.54	78.34	44.93	1 206 40
As at 31 March 2019	1,144.90	443.83	37.09	36.97	69.33	11.27	1,396.49

#### b Intangible Assets

Particulars	Computer Software	Total
GROSS BLOCK		
As at 01 April 2017	340.94	340.94
Additions	18.75	18.75
Disposals/Adjustments	(9.95)	(9.95)
As at 31 March 2018	349.74	349.74
Additions	5.72	5.72
Disposals/Adjustments	0.12	5.12
As at 31 March 2019	355.46	355.46
AMORTISATION	000.10	000.40
As at 01 April 2017	267.17	267.17
Charge for the Year	33.97	33.97
Disposals/Adjustments	(9.94)	(9.94)
As at 31 March 2018	291.20	291.20
Charge for the Year	13.83	13.83
Disposals/Adjustments	10.00	15.05
As at 31 March 2019	305.03	305.03
NET BLOCK		000.00
As at 31 March 2019	58.54	58.54
As at 31 March 2019	50.43	50.43
Capital Work in Progress		
Particulars		Total
As at 01 April 2017		

	iotai
As at 01 April 2017	
Additions	
As at 31 March 2018	
Additions	3.76
As at 31 March 2019	3.76



c



5 Investments

Particulars	March 31, 2019		March 31, 2018	
1.1	Quantity	Amount	Quantity	Amount
) Non-current Investments			a second	Fillouit
(i) Fair valued through Other Comprehensive Income:				
Unquoted Equity shares of INR 10 each, fully paid-up:				
I) Investment in subsidiary companies:				
Acme Living Solutions Private Limited	50,000		50,000	
Arrow Infra Estates Private Limited	1,02,300	6,362.00	1,02,300	6,792.7
Gurgaon Constructwell Private Limited	65,700	8,599.00	65,700	
Deemed Equity Investments in Gurgaon Constructwell Private Limited	00,700	619.68	65,700	11,552.6
Gurgaon Realtech Limited	66,500			0.154.0
TRIL Roads Private Limited*	77,45,012	6,907.00	66,500	8,154.6
Deemed Equity Investments in TRIL Roads Private Limited	11,45,012	34,723.00	77,45,012	18,572.8
TRIF Gurgaon Housing Projects Private Limited	-	2,421.61		
TRIL Urban Transport Private Limited*	50,000	-	50,000	
Deemed Equity Investments in TRIL Urban Transport Private Limited	2,68,38,000	2,334.00	43,38,000	304.3
Wellkept Facility Management Services Private Limited		240.43		
TRIL Constructions Limited	4,00,000		4,00,000	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1
TRIL Infopark Limited (Refer Foot Note 1)	2,44,00,000	2,451.51	2,44,00,000	3,111.2
TOLL Integrate Limited (Kener Pot Note 1)	62,89,00,000	1,03,922.85	62,89,00,000	89,568.1
TRIL Amritsar Projects Limited*	32,43,08,408	3,492.00	32,43,08,408	5,935.9
Deemed Equity Investments in TRIL Amritsar Projects Limited		2,487.94		
TRIF Real Estate Development Limited*	38,78,843	7,615.00	38,78,843	6,989.0
HV Farms Private Limited	10,000	1.00	10,000	7.9
MIA Infratech Private Limited	13,57,300	100	13,57,300	5
International Infrabuild Private Limited*	26,000	150.28	26,000	1,577.5
III Internet and the test of the second s	- entrease		20,000	1,011.0
II) Investment in joint ventures:				
TRIL IT 4 Private Limited*	7,40,000	24,487.00	7,40,000	21,636.5
Mikado Realtors Private Limited.*	1,99,87,400	33,228.20	1,99,87,400	31,258.73
Industrial Minerals and Chemicals Company Private Limited*	3,256	22,270.60	3,256	21,919.7
Taj Air Limited	1,90,00,000	-	1,90,00,000	-
Unserviced Performance above to the state of the			are and showing the	
Unquoted Preference shares, fully paid-up (Compound financial instruments)				
I) Investment in subsidiary companies:				
0.001% Compulsory Convertible Preference shares of INR 10 each in TRIL Constructions Limited	5,92,80,000	5,970.40	5,92,80,000	7,559.00
0% Compulsory Convertible Preference shares of INR 100 each in TRIL Infopark Limited	3,00,00,000	49,578.00	3,00,00,000	42,729.00
n and a second se				
Unquoted Debentures, fully paid-up:				
I) Investment in joint venture:				
Compulsorily Convertible Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private	4 04 70 000	10 000 70		
Limited	1,04,73,960	10,636.70	-	ň.,
(ii) Fair valued through Profit and Loss:				
Unquoted Debentures of INR 10 each, fully paid-up:				
I) Investment in subsidiary companies:				
0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited.	1,13,40,400	1,242.00	1,09,10,000	1,314.00
Compulsorily Convertible Debentures in TRIL Urban Transport Private Limited	3,11,53,750	2,918.00	2,85,43,750	3,024.01
0% Optionally Convertible Debentures in TRIL Roads Private Limited	24,57,00,000	22,565.00	22,12,00,000	20,544.00
0.01% Unsecured Compulsorily Convertible Debentures in TRIL Infopark Limited.	25,00,00,000	41,273.00	25,00,00,000	35.608.00
0% Optionally Convertible Debentures in HV Farms Private Limited	10,00,000	1,188.00	10,00,000	1,148.00
0% Optionally Convertible Debentures in International Infrabuilds Private Limited	27,80,000	211,56	27,80,000	199.70
Deemed investment in 0% Optionally Convertible Debentures in International Infrabuilds Private Limited		211.00	21,00,000	105.38
				100.00
(iii) Quoted Debentures fully paid-up:				
Investment in joint venture:				
Redeemable Non-convertible Debentures of INR 687,500 each in TRIL IT4 Private Limited	1,184	8,140.00	1,184	8 1 40 00
	1,104	0,140.00	1,104	8,140.00
		4,06,035.76		3,47,753.08
Aggregate value of quoted investments				-,-,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
Aggregate book value		8,140.00		8,140.00
Aggregate fair value,		8,140.00		8,140.00
		0,140.00		0,140.00

Foot note:
1. 50,000,000 (2018: 50,000,000) equity shares have been pledged with Tamilnadu Industrial Development Corporation Limited for a period upto 09th December 2019.
+ The Company has provided Non Disposal Undertakings to the lenders / Investors of its subsidiaries and joint ventures for the minimum shareholding that the Company needs to maintain until the final settlement date of the Ioan.

) Current Investments	March 31	March 31, 2018		
Investment in mutual funds:	Quantity	Amount	Quantity	Amount
TATA money market fund - Direct Plan - Growth NAV per unit (in INR) 2,944.44 (2018 : 2,738.28)	10,576.43	311.42	2,54,398.32	6,966.15
		311.42		6,966.15





#### 6 Financial Assets - Trade Receivables

Particulars	March 31	March 31, 2019		1, 2018
Insecured, considered good	Non Current	Current	Non Current	Current
Trade receivables from related parties				
Outstanding for a period exceeding six months Outstanding for a period less than six months	+	660.21 670.66	1	1,301.10
Trade receivables from others		070.00	-	700.00
Outstanding for a period exceeding six months				
Outstanding for a period less than six months (Refer Note No. 44) Total		66.94	-	1,043.83
iviai		1,397.81		3,132.93

#### 7 Financial Assets: Loans and Advances

articulars	March 31	2019	March 31, 2018	
	Non Current	Current	Non Current	Current
(Unsecured, considered good)				Guitein
Advances recoverable from related parties	-	2,015.66		4,142.15
Inter corporate deposits to related parties	52,124.48		36,190,00	4,137.53
Security deposits		38.41	00,100.00	700.15
Other advances	-	295,70		339.69
(Unsecured, considered doubtful)				000.00
Advances recoverable from related parties	-	35.00		35.00
Inter corporate deposits to related parties		95.00	-	95.00
Less : Provisions		(130.00)		(130.00)
(Secured, considered good)		()		(100.00)
Other advances			1.363.93	4,050.00
Total	52,124.48	2,349.77	37,553,93	13,369,52

#### 8 Other Financial Assets

Particulars	March 31	March 31, 2019		2018
	Non Current	Current	Non Current	Current
nterest Accrued Receivable:				- and only
Considered Good		2,861,19		2.806.27
Considered Doubtful		10.00		10.00
Less Provision for Doubtful Interest	-	(10.00)		(10.00
Fixed deposits with more than 12 months maturity (Refer Foot Note) Jnbilled revenue	0.75		0.75	
				101.78
Total Foot Note:	0.75	2,861.19	0.75	2,908.05

Bank deposit of INR 75,000 (2018: INR 75,000) is having lien in favour of Commercial Tax Officer, KVAT Works Contract Ernakulam.

#### 9 Current Tax Assets (net)

Particulars	March 31, 2019	March 31, 2018
Advance Payment of taxes		
Provision for tax	10,407.62	9,765.53
	(4,394.72)	(4.394.72)
Total	6,012.90	5,370.81

#### 10 Other Assets

Particulars	March 31	, 2019	March 31, 2018	
	Non Current	Current	Non Current	Current
Capital Advances	159.91		159.91	
Call Option Premium (Refer Foot note)	5.725.23		5,205.20	
Derivative asset	5,079.00		3,654.00	
Balance with Tax Authority		1,008.66	0,004.00	872.7
Prepaid Expenses	4.32	70.54	0.16	106.9
Other Receivables	-	7.09	0.10	100.5
Total	10,968.46	1.086.29	9,019.27	979.7

#### Foot note:

The Company had paid an interest free advance of INR 7,110/- lakhs to Indian Hotels Company Limited (IHCL) vide MOU dated 11th July, 2011 and MOU dated 23 February, 2010. The consideration for the advance is with an option to acquire the equity investment of TRIL Infopark Limited amounting to INR 7,110/- lakhs representing fair value of shares. The shares will be transferred on or before 10th July, 2021.

#### 11 Inventories

Particulars	March 31, 2019	March 31, 2018
Bought out construction materials	52.15	256.31
Work In Progress	12,710.70	10,983.89
Finished Goods	28,248.21	30,501.33
Total	41,011.06	41,741.53





#### CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements

for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

12 Cash and Bank Balance

Particulars	March 31, 2019	March 31, 2018
Cash and Cash Equivalents		
Cash on Hand	÷	0.61
Balances with Banks		
- in current accounts	445.25	755.35
Deposit Accounts with less than or equal to 3 months original maturity		200.00
Total	445.25	955.96

#### 13 Other Bank Balances

Particulars	March 31, 2019	March 31, 2018	
Deposit Accounts with less than 12 months maturity	444.69	361.67	
Total	444.69	361.6	

#### 14 Equity Share Capital

(a)	Authorised, Issued,	Subscribed and Fully	Paid up :

Particulars	March 31	, 2019	March 31	, 2018
	No of Shares	Amount	No of Shares	Amount
Authorised Capital :				
3,00,00,000 (2018: 2,00,00,00,000) equity shares of INR 10 each	30000,00,000	3,00,000.00	20000,00,000	2,00,000.00
NIL (2018: 1,00,00,00,000) 5% Non cumulative convertible preference shares of INR 10 each.			10000,00,000	1,00,000.00
Issued, Subscribed and Fully Paid up Capital :				
1,01,73,07,692 (2018: 1,01,73,07,692) equity shares of INR 10 each	10173,07,692	1,01,730.77	10173.07.692	1.01.730.77
Total	10173,07,692	1,01,730.77	10173,07,692	1,01,730.77
Reconciliation of Number of Shares Outstanding				
Particulars	March 31	, 2019	March 31	, 2018
	No of Shares	Amount	No of Shares	Amount
As at the beginning of the year	10173,07,692	1,01,730,77	10173,07,692	1.01.730.77
Add: Issued during the year		and the second second		
As at the end of the year	10173,07,692	1,01,730.77	10173,07,692	1,01,730.77

#### (c) Terms and rights attached to the equity share

The Company has only one class of equity shares having par value of INR 10 per share. Each holder of equity shares is entitled to one vote per share. The Company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject to the approval of shareholders in the ensuing Annual General Meeting. In the event of liquidation of the Company, the holders of the equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the shareholders.

(d)	Shares	of the	company	held by	/ the Holding	company	_

Name of Shareholder	March 31, 2019		March 31, 2018	
	No of Shares	Amount	No of Shares	Amount
Equity shares of INR 10 each fully paid-up by		All a series of a		
Tata Sons Private Limited	10173,07,692	1,01,730.77	10173,07,692	1.01,730.77

#### (e) Details of shareholding more than 5% in the Company Name of Shareholder

No of Shares	% Holding	No of Shares	% Holding
		and the second second	
10173.07,692	100%	10173,07,692	100%

March 31, 2019

March 31, 2018

#### 15 Other Equity

Particulars	March 31, 2019	March 31, 2018	
eserves and surplus			
Securities Premium Account	15,769.23	15,769.23	
Capital reserve	4,783.49	4,783.49	
Retained earnings	(37,659.45)	(17,728.54)	
tems of Other comprehensive income			
FVOCI - equity instruments	1,14,041.72	85.210.00	
Defined benefit plan adjustment	20.19	(18.78)	
TOTAL	96,955.18	88,015.40	

Securities premium		
Particulars	March 31, 2019	March 31, 2018
Opening balance	15,769.23	15,769.23
Balance at the end of the Year	15,769.23	15,769.23
Capital reserve		
Particulars	March 31, 2019	March 31, 2018
Opening balance	4,783.49	4,783.49
Balance at the end of the Year	4,783.49	4,783.49





### Tata Realty and Infrastructure Limited

CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements

for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

Particulars	March 31, 2019	March Dd. DDd.D
Balance at the beginning of the year		March 31, 2018
IND AS 115 Adjustment (refer note 23)	(17,728.54)	(500.87
Transferred from other comprehensive income	(1,304.14)	
Add: (Loss) for the year	and the second se	806.49
Balance at the end of the Year	(18,626.77)	(18,034.16)
	(37,659.45)	(17,728.54
Equity Instruments through Other Comprehensive Income (OCI) reserve	(37,659.45) March 31, 2019	
Equity Instruments through Other Comprehensive Income (OCI) reserve	March 31, 2019	March 31, 2018
Equity Instruments through Other Comprehensive Income (OCI) reserve Particulars	March 31, 2019 85,210.00	March 31, 2018 84,559.33
Equity Instruments through Other Comprehensive Income (OCI) reserve Particulars Opening balance	March 31, 2019	March 31, 2018 84,559.33 1,624.18
Equity Instruments through Other Comprehensive Income (OCI) reserve Particulars Opening balance Equity instruments fair value through OCI (FVOCI)	March 31, 2019 85,210.00	84,559.33

Other comprehensive income Particulars March 31, 2019 March 31, 2018 Opening balance (18.78) (31.49) Remeasurements of defined benefit liability (asset) 52.65 21.64 Income tax relating to items that will not be reclassified to profit or loss (13.68) (8.93) Balance at the end of the Year 20.19 (18.78) Nature and purpose of the reserve

#### Capital reserve

Capital reserve was created to record excess of net assets taken over pursuant to scheme of merger sanctioned by the Bombay High Court in the year 2015-16 between Tata Realty and Infrastructure Limited, Mara Builder Private Limited and TRIF Real Estate and Development Limited.

#### Debenture redemption reserve

The Company has not created debenture redemption reserve as per Section 71 of the Companies Act, 2013 due to losses incurred post issuance of debentures.

#### Equity Instruments through Other Comprehensive Income reserve

The Company has elected to recognise changes in the fair value, of investments in equity and preference securities of subsibaries in other comprehensive income. These changes are accumulated within the FVOCI equity investments reserve within equity.

The Company transfers amounts from this reserve to retained earnings when the relevant equity securities are derecognised.

#### 16 Financial Liabilities - Borrowings

Particulars	March 31	, 2019	March 31,	2018
	Non Current	Current	Non Current	Current
Non Convertible Debentures - Unsecured, Unlisted		COLOR ACCONT.		
18,250 (2018: 18,250) 8.25% - 9.25% Non convertible debentures (NCD) @ INR 10 Lacs	72,500.00		1,82,500.00	
Less . Unexpired issuance costs	(39.85)		(189.09)	
Bank Overdraft	(00.00)	4,117.86	(100.00)	
Commercial Papers from Mutual funds (Refer Foot Note No. 2 below)		77,500.00		60,500.00
Less Unexpired discount		(932.71)		(805.26
Inter Corporate Deposits (Refer Foot Note No. 3 below)		14.000.00		(000.20
TOTAL	72,460.15	94,685.15	1,82,310.91	59,694.74
The above amount includes				
Secured Borrowings				
Unsecured Borrowings	72,460.15	94,685.15	1,82,310.91	59,694.74

#### Foot Note:

1) Terms of repayment and Interest of Unsecured Non Convertible Debentures:

Particulars	March 31	, 2019	March 31	, 2018
	Non Current	Current	Non Current	Current
Yes Bank LtdInterest @ 9.25 % payable annually. Principal on Bullet repayment on maturity, on 23-May-2019.		35,000.00	35,000.00	-
Yes Bank LtdInterest @ 9.25 % payable annually. Principal on Bullet repayment on maturity, on 23-July-2019.		25,000.00	25,000.00	
Kotak Mahindra Bank -Interest @ 9.10 % payable annually. Principal on Bullet repayment on maturity, on 25-June-2019.	4	24,000.00	24,000.00	
Kotak Mahindra Bank -Interest @ 9.10 % payable annually. Principal on Bullet repayment on naturity, on 23-Aug-2019.		26,000.00	26,000.00	
Kotak Mahindra Bank -Interest @ 8.25 % payable on maturity. Principal on Bullet repayment on naturity, on 20-Apr-2020.	10,000.00		10,000.00	
Kotak Mahindra Bank -Interest @ 8.25 % payable on maturity. Principal on Bullet repayment on naturity, on 17-Aug-2020.	10,000.00	-	10,000.00	
Kotak Mahindra Bank -Interest @ 8.25 % payable on maturity. Principal on Bullet repayment on maturity, on 20-May-2020.	20,000.00	-	20,000.00	
Kotak Mahindra Bank -Interest @ 8.57 % payable on maturity. Principal on Bullet repayment on naturity, on 20-Apr-2021.	32,500.00	-	32,500.00	-
TOTAL	72,500.00	1,10,000.00	1,82,500.00	

2) Commercial paper issued to mutual funds are at a discount rate ranging from 7.20% - 9.10 % per annum (2018: 7.05% - 7.50% per annum), and the same are repayable within one year at the agreed upon full face value.

3) Inter Corporate Deposit is obtained from a group company at interest rate of 9.00 % per annum and the same is repayable within 90 days.





#### CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements for the year ended 31 March 2019

(Currency: Indian rupees in lakhs)

#### 17 Financial Liabilities - Trade Payable

Particulars	March 31, 2019		March 31, 2018	
	Non Current	Current	Non Current	Current
Trade Payables				
Micro, Small and Medium Enterprises	-		-	
Other than Micro, Small and Medium Enterprises		3,458.03	•)	3,025,61
Retention / Deposits				-
Total		3,458.03	-	3,025.6

Based on information received by the Company from its vendors, the amount of principal outstanding in respect of MSME as at Balance Sheet date covered under the Micro, Small and Medium Enterprises Development Act, 2006 is INR Nil. There were no delays in the payment of dues to Micro and Small Enterprises.

#### 18 Other financial liabilities

Particulars	Narch 31, 2019		March 31, 2018	
	Non Current	Current	Non Current	Current
Interest accrued but not due on borrowings	11,425.58	7,731.74	4,931.25	7,725.41
Current Maturity of Unsecured Long term borrowings	-	1,10,000.00		
Less : Unexpired issuance costs		(21.44)		
Derivatives - Put option (Refer Foot Note)	21	7,122.00		6.682.00
Creditors for Capital Goods		259.90		
	11,425.58	1,25,092.20	4,931.25	14,407.41

#### Foot Note:

Foot Note: As per the terms of agreement of the Company with Tamil Nadu Industrial Development Corporation (TIDCO) dated 24th March 2008 and as per the Company's letter dated 8th March, 2017 for extending the period of put option for a year upto 31 December, 2019, TIDCO has an option to sell its investments in TRIL Infopark Limited for an aggregate consideration of INR 14,583 lakhs. As a security for the above transaction, the Company has pledged its investment in TRIL Infopark Limited with TIDCO, (5,00,00,000 equity shares of INR 10 each, fully paid) and also placed postdated cheque of INR 14,583 lakhs.

#### 19 Provisions

March 31, 2019		March 31, 2018	
Non-Current Provisions	Current Provisions	Non-Current Provisions	Current Provisions
		1.1	100 100 100
292.38	75.55	387.77	19.81
393.10	149.58	468.06	73.47
685.48	225.13	855.83	93.28
	Non-Current Provisions 292.38 393.10	Non-Current Provisions Provisions 292.38 75.55 393.10 149.58	Non-Current Provisions         Current Provisions         Non-Current Provisions           292.38         75.55         387.77           393.10         149.58         468.06

#### 20 Current Tax Liabilities

Particulars	March 31, 2019	March 31, 2018
Provision for taxation	8,965.86	8,965.86
Advance Payment of taxes	(7,213.98)	(7,213.98)
Total	1,751.88	1,751.88

#### 21 Deferred Tax (Liabilities) / Assets (Net)

Particulars	As at March 31, 2018	Movement Recognised in Statement of Profit and Loss	Movement Recognised in Other comprehensive income	Movement Others (Footnote 1)	As at March 31, 2019
Deferred Tax Assets					
Property, plant and equipments and intangible assets	1,410.43	(619.92)			790.51
Fair valuation of derivatives at FVTPL	881.76	(350.57)	-		531.19
Defined benefit obligation	276.38	(25.94)	(13.68)		236.76
Deemed Investment on ICD discounting	-	(615.40)		2,027,17	1,411.77
Deferred Tax Liabilities					
Fair valuations of Equity investments at FVOCI	(10,986.83)		(3,502.88)		(14,489.71
Fair valuations of other financial assets at FVTPL	(2,051.86)	312.96			(1,738.90
Total	(10,470.12)	(1,298.87)	(3,516.56)	2,027.17	(13,258.38

Particulars	As at March 31, 2017	Movement Recognised in Statement of Profit and Loss	Movement Recognised in Other comprehensive income	Movement Others (Footnote 1)	As at March 31, 2018
Deferred Tax Assets			the second s		
Property, plant and equipments and intangible assets	1,112.60	297.83			1,410.43
Fair valuation of derivatives at FVTPL	849.54	32.22	· · ·		881.76
Defined benefit obligation	1.482.69	(1,197.38)	(8.93)		276.38
Deferred Tax Liabilities	1000				
Fair valuations of Equity investments at FVOCI	(10,819.81)		(167.02)		(10,986.83)
Fair valuations of other financial assets at FVTPL	(2,587.60)	535.74			(2,051.86)
Total	(9,962.58)	(331.59)	(175.95)	1.4	(10,470.12)

Footnote:

1. Movement in Deemed Equity Investments (Note 5).





#### Notes to Ind AS financial statements

for the year ended 31 March 2019

(Currency: Indian rupees in lakhs)

#### 22 Other Current Liabilities

Particulars	March 31, 2019	March 31, 2018
Advances from customers	3,939,16	2,956.42
Statutory dues including provident fund and tax deducted at source	82.02	340.3
Compensation on delayed possession payable	10.88	54.20
Corpus fund collection	676.09	622.62
Advance maintenance charges	95.06	243.87
Security deposits from customers	63.70	63.70
Other Payable	252.57	03.70
Total	5,119.48	4,281.11

#### 23 Revenue from Operations

Particulars	For the year ended 31 March 2019		For the year ended 31 March 2018	
Sale of residential flats		9,463.27		7,081.06
Sale of services				
Project management consultancy fees	3,281,49		4,598.46	
Asset management fees	1,120.39		998.78	
Maintenance and other receipts	628.14	5,030.02	523.88	6,121.12
Total revenue from operations	1	14,493.29		13,202.18

Effective 1 April 2018, the Company has transited to Ind AS 115 and has availed exemption from retrospective restatement of revenue by adopting the modified approach as permissible under the transition provisions of the standard. The effect of initially applying this standard is recognised at the date of initial application (i.e. 1 April 2018). The standard is applied retrospectively only to contracts that are not completed as at the date of initial application and the comparative information in the statement of profit and loss is not restated (Refer Note 43).

#### 24 Other Income

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018	
Interest Income on:			
Inter corporate deposits / non convertible debentures*	5.441.88	4,082.93	
Income-tax refund	67.40	331.26	
Fixed deposits	95.64	19.07	
Unwinding of advances	540.44	597.13	
Profit on sale of Current Investments	263.43	160.97	
Gain on fair valuations of investments	4,680,48	100.01	
Gain on fair valuations of call options	1.425.00		
Mark to Market gain on current investment in Mutual funds	20.02		
Other income from residential projects	161.62	225.23	
Miscellaneous Income	58.04	6.61	
Total	12,753.95	5,423.20	

\* Includes Unwinding of interest amounting to INR 2,366.93 lakhs (2018; Nil)

#### 25 Cost of sale of flats

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018	
Material consumed			
Opening Stock	256.31	394.46	
Add : Purchases of materials			
Less : Closing inventories (Refer Note 11)	(52.15)	(256.31)	
Total cost of materials consumed	204.16	138.15	
Expenditure during the year			
Opening Stock of Inventories	41,485.22	43,268,47	
Addition during the year		10,200.11	
Professional fees and technical fees	10.00	21.53	
Project management consultancy charges	37.26	18.82	
Approval and permission expenses	70.09	27.64	
Construction cost	3,318.50	2,453.33	
Other expenses	86.21	54.24	
	45,007.28	45,844.03	
	45,211.44	45,982.18	
Less: Closing Stock of Inventories (Refer Note 11)	(40,958.91)	(41,485.22)	
Add: Reversal of cost of IND AS 115	1,870,38		
Total	6,122.91	4,496.96	

#### 26 Employee Benefits

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018
Salaries, wages and bonus	6,598.04	5,226.91
Less: Deputation charges recovered	(1,566,48)	(837.43)
Gratuity charges and Contributions to Provident and pension funds	222.41	217.64
Staff welfare expenses	252.97	380.48
Compensated absences	78.45	43.80
Total	5,585.39	5,031.40





27 Finance Cost

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018
Interest Expense	01 March 2013	ST March 2018
- on NCD	16,594,33	13,798.82
- on bank overdraft	33.48	60.59
- on commercial paper	6,140.06	5,505.06
Finance charges	221.04	307.43
Total	22,988.91	19.671.90

#### 28 Depreciation & Amortisation

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018
Depreciation	126.48	111.90
Amortisation	13.82	33.97
Total	140.30	145.87

#### 29 Loss on fair valuations through profit or loss

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018
Loss on fair valuation of investments		2,388.76
Loss on fair valuation of derivatives		
- on call options	and the second se	364.00
- on put options	440.00	102.00
Total	440.00	2,854.76

#### 30 Other Expenses

articulars	For the year ended 31 March 2019	For the year ended 31 March 2018
Advertisement and business promotion expenses	734.02	1,012.25
Audit Fees (Refer Foot Note)	23.65	18.20
Brokerage	93.49	79.75
Telephone and Communication expenses	90.35	97.43
Compensation to Customers	98.67	7.10
Directors Sitting Fees	47.60	42.13
Fees & Consultations	885.27	484.90
Insurance	33.09	24.31
Loss on Sale of PPE	23.17	11.20
Miscellaneous expenses	46.35	16.20
Office and common area maintenance charges	840.73	771.5
Power & Utilities	29.96	42.38
Printing, courier and stationery	25.03	45.3
Rates & Taxes	66.62	249.47
Recruitment and conference expenses	109.54	75.70
Rent	364.34	395.90
Repairs and maintenance	138.33	123.03
Security charges	41.56	34.17
Travelling and conveyance	191.48	213.12
Total	3,883.25	3,744.21

Foot Note Remuneration to Statutory Auditors:

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018	
Statutory Audit	21.00	13.00	
Other Services	1.15		
Taxation Matters	1.50	5.20	
Total	23.65	18.20	

#### 31 Amounts written off during the year

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018	
Advances written off	5,414.38	15.50	
Service tax input credit written off	-	101.74	
Total	5,414.38	117.24	





(Currency: Indian rupees in lakhs)

#### 32 Tax Expense

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018
a) Amounts recognised in profit and loss Current income tax		
	(1,298.87)	(331.59)
Deferred Tax Tax expense for the year	(1,298.87)	(331.59)
(b) Income tax recognised in other comprehensive income Items that will not be reclassified to profit or loss		
Remeasurements of the defined benefit plans	(13.68)	(8.93)
Equity Instruments through Other Comprehensive Income	(3,502.88)	(167.02)
Tax expense for the year	(3,516.56)	(175.95
(c) Income tax expense for the year can be reconciled to the accounting profit as follows		
Loss before tax	(17,275.25)	(17,415.32
Tax using the Company's domestic tax rate 26.00% (2018: 26.00%)	-	-
Tax effect of:		
Reduction in tax rate		
Deferred tax on fair valuation through profit or loss	(678.95)	(629.42)
Deferred tax on business expenses	(619.92)	297.83
Income tax expense / (benefit) recognised in Statement of profit and loss	(1,298.87)	(331.59

#### (d) Movement in deferred tax balances

Particulars	Net balance April 1, 2018	Recognised in profit or loss	Recognised in OCI	Recognised in Deemed Equity Investments	Net balance March 31, 2019	Deferred tax asset	Deferred tax liability
Deferred tax asset	-		-	-		-	and the second second
Deferred tax liability	(10.470.11)	(1,298.87)	(3,516.56)	2,027.17	(13,258.37)	14	(13,258.37)
Deterror	(10,470.11)	(1,298.87)	(3,516.56)	2,027.17	(13,258.37)		(13,258.37)
Particulars	Net balance April 1, 2017	Recognised in profit or loss	Recognised in OCI	Recognised in Deemed Equity Investments	Net balance March 31, 2018	Deferred tax asset	Deferred tax liability
Deferred tax asset						- 1+1-	
Deferred tax liability	(9,962.57)	(331.59)	(175.95)		(10,470.11)		(10,470.11)
Deletion tax hability	(9,962.57)	(331.59)	(175.95)		(10,470.11)		(10,470.11)

The Company offsets tax assets and liabilities if and only if it has a legally enforceable right to set off current tax assets and current tax liabilities and the deferred tax assets and deferred tax liabilities relate to income taxes levied by the same tax authority.

Significant management judgement is required in determining provision for income tax, deferred income tax assets and liabilities and recoverability of deferred income tax assets. The recoverability of deferred income tax assets is based on estimates of taxable income and the period over which deferred income tax assets will be recovered. Any changes in future taxable income would impact the recoverability of deferred tax assets.

#### Tax losses carried forward

Deferred tax assets have not been recognised in respect of unabsorbed business losses, because it is not probable that future capital gains profit will be available against which the Company can use the benefits therefrom.

#### 33 Expenditure in foreign currency (on accrual basis)

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018
Professional fees	-	23.56
Training and conference expenses	1.21	61.88
Membership & Subscription Expenses	11.56	100
Travelling expenses	0.95	6.83

#### 34 Earning Per Share

Earnings Per Share (EPS) = Net Profit attributable to Shareholders / Weighted Number of Shares Outstanding

Particulars		the year ended March 2019	For the year ended 31 March 2018
(Loss) after tax attributable to equity shareholders	A	(18,626.77)	(18,034.16)
Calculation of weighted average number of equity shares: Number of equity shares at the beginning of the year	с	1,01,73,07,692	1,01,73,07,692
Number of equity shares outstanding at the end of the year		1,01,73,07,692	1,01,73,07,692
Weighted average number of equity shares outstanding during the year	В	1,01,73,07,692	1,01,73,07,692
Earning Per Share – Basic (INR)	(A / B)	(1.83)	(1.77)
Earning Per Share - Diluted (INR)		(1.83)	(1.77)





35 Contingencies and commitments

1) Continuents 1) Control to Company that Situate from Company 1, 200, 55 1, 100, 55 1,	Particulars		March 31, 2019	March 31, 2018
- Incone tax demands contested by the Company     - Indivest tax demands     - Indivest tax demands contested by the Company     - Indivest tax     - Indivestax     - Indive	(i) Contingent Liabilities (Refer footnote 1)			March 51, 2010
- Incone tax demands contested by the Company     - Indivest tax demands     - Indivest tax demands contested by the Company     - Indivest tax     - Indivestax     - Indive	Claims against the Company not acknowledg	ed as debts		
- Indirect tax demands contracted by the Company 497 30 107 30 107 407 30 100 107 400 107 400 107 400 100 100 100 100 100 100 100 100 100	<ul> <li>Income tax demands contested by the Com</li> </ul>	pany	1,268,59	2,300,83
Commission     Construction     Commission     Commissi     Commission     Commission     Commission     Commission     C	<ul> <li>Indirect tax demands contested by the Com</li> </ul>	pany		
1. The Company does not expect any outflow of economic resources in respect of line above and therefore no provision is made in respect thereof.          (ii) Commany has issued letter of comfort to banks in respect of loans availed by a few of its subsidiaries :         Name of Subsidiaries       Nature of Confort given         Mited Realizors PV. Lid       Shortfall undertaking to meet any shortfall during the tenure of facility:         International Infrabulid Private Limited       To ensure payment to debenture holdes in the event of famination of the concession agreement.         TRIF Fiel Electro Development Limited       Undertake to meet concern to the extent of 10% of project Cost.         (b) The Company has issued financial support lotter to following subsidiaries :       ). A famina famination of the consession agreement.         (c) Multified Finality Management Services Private Limited (Previously known as TRIL Hospitality Private Limited)       9. Wilding Finality Management Services Private Limited (Previously known as TRIL Hospitality Private Limited)         (c) Oronomithents       Barch 31, 2019       March 31, 2019         Particulars       March 31, 2019       March 31, 2019         Particulars       1/3.053.05       7.935.85	- Claims made by contractors		1,179.00	
1. The Company does not expect any outflow of economic resources in respect of line above and therefore no provision is made in respect thereof.          (ii) Commany has issued letter of comfort to banks in respect of loans availed by a few of its subsidiaries :         Name of Subsidiaries       Nature of Confort given         Mited Realizors PV. Lid       Shortfall undertaking to meet any shortfall during the tenure of facility:         International Infrabulid Private Limited       To ensure payment to debenture holdes in the event of famination of the concession agreement.         TRIF Fiel Electro Development Limited       Undertake to meet concern to the extent of 10% of project Cost.         (b) The Company has issued financial support lotter to following subsidiaries :       ). A famina famination of the consession agreement.         (c) Multified Finality Management Services Private Limited (Previously known as TRIL Hospitality Private Limited)       9. Wilding Finality Management Services Private Limited (Previously known as TRIL Hospitality Private Limited)         (c) Oronomithents       Barch 31, 2019       March 31, 2019         Particulars       March 31, 2019       March 31, 2019         Particulars       1/3.053.05       7.935.85	Foot Note			
(a) The Company has issued letter of comfort to banks in respect of leans evalled by a few of its subsidiaries :          Name of Subsidiaries       Nature of Comfort given         Midado Realtors Pvi. L1d.       Shortfall undertaking to meet any shortfall during the tenure of facility         International Infrabulied Private Limited       Undertaking to meet cost overruin to the event of 10% of project Cost.         (b) The Company has issued financial support letter to following subsidiaries :       1) Arme Lwing Solutions Private Limited         1) Mare Limit Support letter to following subsidiaries :       1) Arme Lwing Solutions Private Limited (Previously known as TRIL Hospitality Private Limited)         1) TRIE Grappion Houng Private Limited       Previously known as TRIL Hospitality Private Limited)         1) TRIE Grappion Houng Private Limited       March 31, 2019         March 31, 2019       March 31, 2019         Particulars       March 31, 2019         Particulars       March 31, 2019         Particulars       March 31, 2019         Particulars       March 31, 2019         Commitments       7, 935, 85         TRIL Constructions L1d       Trivate Private Private Limited         Company has taken various residential premises under competial account and not provided for       7, 935, 85         TRIL Constructions L1d       Trivate Private Preference shares of       7		economic resources in respect of the above and t	herefore no provision is made in respect	t thereof.
Madd Realtors Pvt. Ltd. Shortfall undersking to meet any shortfall during the tenure of facility International Infrabulid Private Limited To ensure payment to debenture holders in the event of termination of the concession agreement. TRIF Roal Estate Development Limited Undertake to meet cost overrun to the extent of 10% of project Cost. (b) The Company has issued financial support letter to following subsidiaries : 1) Acre Living Solutions Private Limited 2) MAI Infrastructure Private Limited 3) Welkept Teality Management Services Private Limited (Previously known as TRIL Hospitality Private Limited) 4) TRIF Gurgaon Housing Projects Private Limited 6) Gurgaon Constructivell Private Limited 7, 335, 85		banks in respect of loans availed by a few of its s	ubsidiaries :	
International Infrabulid Private Limited To ensure payment to debanture holders in the event of termination of the concession agreement. TRIF Real Estate Development Limited Undertake to meet cost overrun to the extent of 10% of project Cost. (b) The Company has issued financial support letter to following subsidiaries : 1) Advin Lings Okulions Private Limited 2) MW Mithough Facility Management Services Private Limited (Previously known as TRIL Hospitality Private Limited) 3) MW Mithough Previously Management Services Private Limited (Previously known as TRIL Hospitality Private Limited) 4) TRIE Amritaer Projects Private Limited 5) TRIL Amritaer Projects Private Limited 6) Grug on Constructivel Private Limited 7) TRIC Constructions Ltd. Amrical commitments Particulars Commitments Iowards investment in 0.001% Non Cumulative Convertible Preference shares of 7, 935.85 TRIL Constructions Ltd. 6) The Company has taken various residential premises under cancellable operating leases. (b) The Company has taken various residential premises under cancellable operating leases. (b) The Company has taken various residential premises under cancellable operating leases. (c) The Company has taken various residential premises under cancellable operating leases. The future minimum lease payments in respect of Particulars Particulars Particulars 102 The Company has taken various residential premises under cancellable operating leases. The future minimum lease payments in respect of Particulars 202 Ad 202 B 202 Ad 202 B 2	Name of Subsidiaries	Nature of Comfort given		
THIF Real Estate Development Limited       Undertake to meet cost overrun to the extent of 10% of project Cost         (b) The Company has issued financial support letter to following subsidiaries :       1) Acre Living Solutions Private Limited         2) MW Infrastructure Private Limited       1) Multicet Facility Management Services Private Limited (Previously known as TRIL Hospitality Private Limited)         3) TBIL Amritar Projects Limited       1) TRIC Marge molessing Private Limited         3) Gurgaon Constructivel Private Limited       12.16         Particulars       March 31, 2019       March 31, 2018         Commitments       12.16         Particulars       March 31, 2019       March 31, 2018         Commitments       7.935.85       7.935.85         TRIL Constructions Lid.       7.935.85       7.935.85         (a) The Company has laken various residential premises under cancellable operating leases.       1       7.935.85         (b) The Company has laken various residential premises under cancellable operating leases.       1       228.34         (b) The Company has laken various residential premises under cancellable operating leases.       1       228.34         (c) The Company has laken various residential premises under cancellable operating leases.       1       228.34         (b) The Company has laken various residential premises under cancellable operating leases.       1       228.34	Mikado Realtors Pvt. Ltd.	Shortfall undertaking to meet any shortfall	during the tenure of facility	
TRIF Real Estate Development Limited       Undertake to meet cost overrun to the extent of 10% of project Cost.         (b) The Company has issued financial support letter to following subsidiaries :       1) After Lining Solutions Private Limited         3) MW linity Facility Management Services Private Limited (Previously known as TRIL Hospitality Private Limited)       1) TRI Amritas Projects Private Limited         6) Grug con Constructivel Private Limited       10 Rar for any point Services Private Limited (Previously known as TRIL Hospitality Private Limited)         6) Grug con Constructivel Private Limited       12 16         Particulars       March 31, 2019       March 31, 2018         Commitments       12 16       12 16         Particulars       March 31, 2019       March 31, 2018         Commitments       7,935.85       7,935.85         TRIL Constructions Ltd.       7,935.85       7,935.85         Company has taken various residential premises under cancellable operating leases.       10 Adreh 31, 2019       March 31, 2018         Particulars       March 31, 2019       March 31, 2019       March 31, 2018         Commitments       242.85       242.85       242.85         Particulars       March 31, 2019       March 31, 2018       242.85         (c) The Company has taken various residential premises under cancellable operating leases.       10 Adreh 31, 2018       242	International Infrabuild Private Limited	To ensure payment to debenture holders	in the event of termination of the conce	ession agreement.
1) Acme Living Solutions Private Limited 2) MA Infrastructure Provide Limited 3) Welkinger Facility Management Services Private Limited (Previously known as TRIL Hospitality Private Limited) 4) TRIE (aprican Housing Projects Private Limited 6) Gurgaon Constructivell Private Limited 9) TRIL Anntias Projects Limited 6) Gurgaon Constructivell Private Limited Particulars Particulars Estimated amount of contracts remaining to be executed on capital account and not provided for (net of advances) March 31, 2019 March 31, 2019 March 31, 2019 March 31, 2018 Commitments Particulars Commitments Particulars March 31, 2019 March 31, 2019 March 31, 2018 Commitments ID Particulars (a) The Company has taken various residential premises under cancellable operating leases. (b) The Company has taken various residential premises under cancellable operating leases. (c) The Company has taken various residential premises under cancellable operating leases. (b) The Company has taken various residential premises under cancellable operating leases. (c) The Company has taken various residential premises under cancellable operating leases. (b) The Company has taken various residential premises under cancellable operating leases. (c) The Company has taken various residential premises under cancellable operating leases. (c) The Company has taken various residential premises under cancellable operating leases. (c) The Company has taken various residential premises under cancellable operating leases. (c) The Company has taken various residential premises under cancellable operating leases. (c) The Company has taken various residential premises under cancellable operating leases. (c) There are no exceptionalizes frictive covenants in the lease agreements. The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in sub- section 4 of Section 186 of the Act in respect of loans given, investment made o	TRIF Real Estate Development Limited			
Particulars         March 31, 2019         March 31, 2018           Estimated amount of contracts remaining to be executed on capital account and not provided for (net of advances)         12.16           nancial commitments         March 31, 2019         March 31, 2018           Particulars         March 31, 2019         March 31, 2018           Commitments towards investment in 0.001% Non Cumulative Convertible Preference shares of TRIL Constructions Ltd.         7,935.85         7,935.85           ating lease (a) The Company has taken various residential premises under cancellable operating leases.         (b) The Company has taken various residential premises under cancellable operating leases.         (b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases.         (b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases.         Not later than one year         328.34           Later than one year         328.34         242.86         242.86           Later than one year but not later than five years         384.24         429.72           (c)There are no exceptional/restrictive covenants in the lease agreements.         384.24         429.72           (c)There are no exceptional/restrictive covenants in the lease agreements.         The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in subsecloin 146 of the Act in r	<ol> <li>Acme Living Solutions Private Limited</li> <li>MIA Infrastructure Private Limited</li> <li>Wellkept Facility Management Services</li> <li>TRIF Gurgaon Housing Projects Private</li> <li>TRIL Amritsar Projects Limited</li> <li>Gurgaon Constructwell Private Limited</li> </ol>	Private Limited (Previously known as TRIL Hospi	tality Private Limited)	
Estimated amount of contracts remaining to be executed on capital account and not provided for (net of advances)  anacial commitments  Particulars  March 31, 2019 March 31, 2019 March 31, 2019 March 31, 2019 March 31, 2018  TRU Constructions Ltd.  ating lease (a) The Company has taken various residential premises under cancellable operating leases. (b) The Company has taken various residential premises under cancellable operating leases. (c) The Company has taken various residential premises under cancellable operating leases. (b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. (c) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. The future minimum lease payments in respect of these properties on 31 March 2019 March 31, 2018 Not later than one year Later than one year Later than five years Later than five year (includes recorded under Schedule VI to the Act.	pital commitments			
Estimated amount of contracts remaining to be executed on capital account and not provided for (net of advances)  ancial commitments  Particulars  March 31, 2019 March 31, 2019 March 31, 2018  TRIL Constructions Ltd.  ating lease (a) The Company has taken various residential premises under cancellable operating leases. (b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. (c) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. (b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. (c) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. The future minimum lease payments in respect of these properties as on 31 March 2019 March 31, 2019 March 31, 2019 March 31, 2019 March 31, 2018 (c) There are no exceptional/restrictive covenants in the lease agreements. The operations of the Company are classified as infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in sub- section 405 of the Act in respect of fonse given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company. Details of investments made by Company as on 31 March 2019 (including investments made during the year).				
In the original set of advances investments in 0.001% Non Cumulative Convertible Preference shares of 7.935.85	Particulars		March 31, 2019	March 31, 2018
Particulars       March 31, 2019       March 31, 2018         Commitments towards investment in 0.001% Non Cumulative Convertible Preference shares of TRIL Constructions Ltd.       7.935.85       7,935.85         ating lease       (a) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases.       The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases.       The future minimum lease payments in respect of these properties as on 31 March 2019 is as follows:         Particulars       March 31, 2019       March 31, 2018         Not later than one year       328.34       328.34         Later than one year but not later than five years       328.34       328.34         Payments of lease rentals during the year (includes recorded under personnel costs INR 19.89       384.24       429.72         (c)There are no exceptional/restrictive covenants in the lease agreements.       384.24       429.72         The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in subsection 4 of Section 186 of the Act in respect of loans, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.         Details of investments made by Company as on 31 March 2019 (including investments made during the year)       Investments made during the year)         N		executed on capital account and not provided for	and a second	March 31, 2018
Commitments towards investment in 0.001% Non Cumulative Convertible Preference shares of TRIL Constructions Ltd.       7,935.85       7,935.85         ating lease       (a) The Company has taken various residential premises under cancellable operating leases.       (b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases.       The future minimum lease payments in respect of these properties as on 31 March 2019 is as follows:         Particulars       March 31, 2019       March 31, 2018         Not later than one year       328.34         (c) There are no exceptional/restrictive covenants in the lease agreements.       384.24       429.72         (c) There are no exceptional flow of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.       Sale of Investments made         Details of investments made by Company as on 31 March 2019 (including investments made during the year)       Investments made       Sale of Investments         Name of the entity       31 March 2018       Investments made       Investments       Ind AS adjustments       31 March 2019	Estimated amount of contracts remaining to be e	executed on capital account and not provided for	and a second	March 31, 2018
Commitments towards investment in 0.001% Non Cumulative Convertible Preference shares of 7,935.85 7,93	Estimated amount of contracts remaining to be e (net of advances)	executed on capital account and not provided for	and a second	March 31, 2018
TRIL Constructions Ltd.         ating lease         (a) The Company has taken various residential premises under cancellable operating leases.         (b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. The future minimum lease payments in respect of these properties as on 31 March 2019 is as follows:         Particulars       March 31, 2019       March 31, 2019         Not later than one year       328.34         (c) There are no exceptional/restrictive covenants in the lease agreements.       384.24         The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in subsection 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.         Details of investments made by Company as on 31 March 2019 (including investments made during the year)         Name of the entity       31 March 2018       Investments made       Sale of investments in dAS adjustments       31 March 2019	Estimated amount of contracts remaining to be e (net of advances) rancial commitments	executed on capital account and not provided for	12.16	*
(a) The Company has taken various residential premises under cancellable operating leases.       (b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. The future minimum lease payments in respect of these properties as on 31 March 2019 is as follows:         March 31, 2019       March 31, 2019       March 31, 2018         Particulars       March 31, 2019       March 31, 2018         Not later than one year         Later than one years         Later than one years         August the sear entals during the year (includes recorded under personnel costs INR 19.89         akk, (2018. 47.88 lakh)         (c) There are no exceptional/restrictive covenants in the lease agreements.         The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act, Accordingly, the disclosure requirements specified in subsection 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.         Details of investments made by Company as on 31 March 2019 (including investments made during the year)         Name of the entity       31 March 2018         Name of the entity       31 March 2018	Estimated amount of contracts remaining to be e (net of advances) ancial commitments Particulars Commitments towards investment in 0.001% No		12.16 March 31, 2019	March 31, 2018
(a) The Company has taken various residential premises under cancellable operating leases.       (b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. The future minimum lease payments in respect of these properties as on 31 March 2019 is as follows:         March 31, 2019       March 31, 2019       March 31, 2018         Particulars       March 31, 2019       March 31, 2018         Not later than one year         Later than one years         Later than one years         August the sear entals during the year (includes recorded under personnel costs INR 19.89         akk, (2018. 47.88 lakh)         (c) There are no exceptional/restrictive covenants in the lease agreements.         The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act, Accordingly, the disclosure requirements specified in subsection 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.         Details of investments made by Company as on 31 March 2019 (including investments made during the year)         Name of the entity       31 March 2018         Name of the entity       31 March 2018	Estimated amount of contracts remaining to be e (net of advances) ancial commitments Particulars Commitments towards investment in 0.001% No		12.16 March 31, 2019	March 31, 2018
(b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. The future minimum lease payments in respect of these properties as on 31 March 2019 is as follows:         Particulars       March 31, 2019       March 31, 2018         Not later than one year       328.34         Later than one year but not later than five years       242.86         Later than five years       384.24       429.72         Payments of lease rentals during the year (includes recorded under personnel costs INR 19.89)       384.24       429.72         (c)There are no exceptional/restrictive covenants in the lease agreements.       The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in subsection 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.         Details of investments made by Company as on 31 March 2019 (including investments made during the year)       Sale of Investments ind AS adjustments 31 March 2019         Name of the entity       31 March 2018       Investments made during the year)       Sale of Investments ind AS adjustments 31 March 2019	Estimated amount of contracts remaining to be e (net of advances) ancial commitments Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd.		12.16 March 31, 2019	March 31, 2018
these properties as on 31 March 2019 is as follows:         Particulars       March 31, 2019       March 31, 2018         Not later than one year       328.34         Later than one year but not later than five years       242.86         Later than five years       384.24         Payments of lease rentals during the year (includes recorded under personnel costs INR 19.89       384.24         Iakh (2018. 47.88 lakh)       384.24         (c)There are no exceptional/restrictive covenants in the lease agreements.       384.24         The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in subsection 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.         Details of investments made by Company as on 31 March 2019 (including investments made during the year)         Name of the entity       31 March 2018       Investments made during the year)         Investments made       Sale of Investments       Ind AS adjustments       31 March 2018	Estimated amount of contracts remaining to be e (net of advances) Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. ating lease	n Cumulative Convertible Preference shares of	12.16 March 31, 2019	March 31, 2018
Not later than one year Later than one year but not later than five years Later than one year but not later than five years Later than one year but not later than five years Later than one year but not later than five years Payments of lease rentals during the year (includes recorded under personnel costs INR 19.89 lakh (2018. 47.88 lakh) (c)There are no exceptional/restrictive covenants in the lease agreements. The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in sub- section 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company. Details of investments made by Company as on 31 March 2019 (including investments made during the year) Name of the entity 31 March 2018 Investments made during the year	Estimated amount of contracts remaining to be e (net of advances) Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. ating lease (a) The Company has taken various residential (	n Cumulative Convertible Preference shares of premises under cancellable operating leases.	12.16 March 31, 2019 7,935.85	March 31, 2018 7,935.85
Not later than one year Later than one year but not later than five years Later than one year but not later than five years Later than one year but not later than five years Payments of lease rentals during the year (includes recorded under personnel costs INR 19.89 lakh (2018. 47.88 lakh) (c)There are no exceptional/restrictive covenants in the lease agreements. The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in sub- section 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company. Details of investments made by Company as on 31 March 2019 (including investments made during the year) Name of the entity 31 March 2018 Investments made during the year	Estimated amount of contracts remaining to be e (net of advances) ancial commitments Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. Iting lease (a) The Company has taken various residential ( b) The Company has also taken a commercial	n Cumulative Convertible Preference shares of premises under cancellable operating leases. property and certain residential premises on non-	12.16 March 31, 2019 7,935.85	March 31, 2018 7,935.85
Later than five years       Payments of lease rentals during the year (includes recorded under personnel costs INR 19.89       384.24       429.72         Jakh (2018. 47.88 lakh)       384.24       429.72         (c)There are no exceptional/restrictive covenants in the lease agreements.       section 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.         Details of investments made by Company as on 31 March 2019 (including investments made during the year)         Name of the entity       31 March 2018       Investments made during the year       Sale of Investments       Ind AS adjustments       31 March 2018	Estimated amount of contracts remaining to be e (net of advances) ancial commitments Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. ating lease (a) The Company has taken various residential to b) The Company has also taken a commercial hese properties as on 31 March 2019 is as folic	n Cumulative Convertible Preference shares of premises under cancellable operating leases. property and certain residential premises on non-	12.16 March 31, 2019 7.935.85 cancellable operating leases. The future	March 31, 2018 7,935.85 e minimum lease payments in respect of
Payments of lease rentals during the year (includes recorded under personnel costs INR 19.89 384.24 429.72 lakh (2018.47.88 lakh) 429.72 (c)There are no exceptional/restrictive covenants in the lease agreements. The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in subsection 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company as on 31 March 2019 (including investments made during the year) Name of the entity 31 March 2018 Investments made Sale of Investments Ind AS adjustments 31 March 2019	Estimated amount of contracts remaining to be e (net of advances) Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. ating lease (a) The Company has taken various residential ( (b) The Company has also taken a commercial hese properties as on 31 March 2019 is as follo Particulars Not later than one year	n Cumulative Convertible Preference shares of premises under cancellable operating leases. property and certain residential premises on non-	12.16 March 31, 2019 7.935.85 cancellable operating leases. The future	March 31, 2018 7,935.85 e minimum lease payments in respect of March 31, 2018
lakh (2018.47.88 lakh)       304.24       429.72         (c)There are no exceptional/restrictive covenants in the lease agreements.       429.72         The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in subsection 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.         Details of investments made by Company as on 31 March 2019 (including investments made during the year)         Name of the entity       31 March 2018       Investments made during the year investments and during the year	Estimated amount of contracts remaining to be e (net of advances) Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. ating lease (a) The Company has taken various residential (b) The Company has also taken a commercial hese properties as on 31 March 2019 is as folic Particulars Not later than one year Later than one year but not later than five years	n Cumulative Convertible Preference shares of premises under cancellable operating leases. property and certain residential premises on non-	12.16 March 31, 2019 7.935.85 cancellable operating leases. The future	March 31, 2018 7,935.85 e minimum lease payments in respect of March 31, 2018 328.34
The operations of the Company are classified as 'infrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in sub- section 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company. Details of investments made by Company as on 31 March 2019 (including investments made during the year) Name of the entity 31 March 2018 Investments made Sale of Investments Ind AS adjustments 31 March 2019	Estimated amount of contracts remaining to be e (net of advances) ancial commitments Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. ating lease (a) The Company has taken various residential to b) The Company has also taken a commercial hese properties as on 31 March 2019 is as folic Particulars Not later than one year Later than one years	n Cumulative Convertible Preference shares of premises under cancellable operating leases. property and certain residential premises on non- ws:	12.16 March 31, 2019 7.935.85 cancellable operating leases. The future	March 31, 2018 7,935.85 e minimum lease payments in respect of March 31, 2018 328.34 242.86
Section 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.  Details of investments made by Company as on 31 March 2019 (including investments made during the year)  Mame of the entity  31 March 2018  Investments made Sale of Investments Ind AS adjustments 31 March 2019	Estimated amount of contracts remaining to be e (net of advances) ancial commitments Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. ating lease (a) The Company has taken various residential ) (b) The Company has also taken a commercial hese properties as on 31 March 2019 is as folic Particulars Not later than one year Later than one year but not later than five years atter than five years Payments of lease rentals during the year (inclu	n Cumulative Convertible Preference shares of premises under cancellable operating leases. property and certain residential premises on non- ws:	12.16 March 31, 2019 7.935.85 cancellable operating leases. The future March 31, 2019	March 31, 2018 7,935.85 e minimum lease payments in respect of March 31, 2018 328.34 242.86
Section 4 of Section 186 of the Act in respect of loans given, investment made or guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.  Details of investments made by Company as on 31 March 2019 (including investments made during the year)  Mame of the entity  31 March 2018  Investments made Sale of Investments Ind AS adjustments 31 March 2019	Estimated amount of contracts remaining to be e (net of advances) Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. ating lease (a) The Company has taken various residential ( b) The Com	n Cumulative Convertible Preference shares of premises under cancellable operating leases. property and certain residential premises on non- ws: des recorded under personnel costs INR 19.89	12.16 March 31, 2019 7.935.85 cancellable operating leases. The future March 31, 2019	March 31, 2018 7,935.85 e minimum lease payments in respect of March 31, 2018 328.34 242.86
Name of the entity 31 March 2018 Investments made Sale of Investments Ind AS adjustments 31 March 2019	Estimated amount of contracts remaining to be e (net of advances) ancial commitments Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. atting lease (a) The Company has taken various residential ) (b) The Company has also taken a commercial hese properties as on 31 March 2019 is as follo Particulars Not later than one year Later than one year but not later than five years Later than five years Payments of lease rentals during the year (inclu akh (2018. 47.88 lakh) c)There are no exceptional/restrictive covenant	n Cumulative Convertible Preference shares of premises under cancellable operating leases. property and certain residential premises on non- ws: des recorded under personnel costs INR 19.89 s in the lease agreements.	12.16 March 31, 2019 7.935.85 cancellable operating leases. The future March 31, 2019	March 31, 2018 7,935.85 e minimum lease payments in respect of March 31, 2018 328.34 242.86 - 429.72
Name of the entity 31 March 2018 investments made investments ind AS adjustments 31 March 2019	Estimated amount of contracts remaining to be e (net of advances) <b>Particulars</b> Commitments towards investment in 0.001% No TRIL Constructions Ltd. <b>ating lease</b> (a) The Company has taken various residential ) (b) The Company has also taken a commercial hese properties as on 31 March 2019 is as folice <b>Particulars</b> Not later than one year .ater than one years but not later than five years .ater than one years but not later than five years .ater than five years Payments of lease rentals during the year (inclu akh (2018. 47.88 lakh) (c) There are no exceptional/restrictive covenant The operations of the Company are classified section 4 of Section 186 of the Act in respect of	n Cumulative Convertible Preference shares of premises under cancellable operating leases, property and certain residential premises on non- ws: des recorded under personnel costs INR 19.89 s in the lease agreements, as 'infrastructure facilities' as defined under Sch of loans given, investment made or guarantee giv	12.16 March 31, 2019 7.935.85 cancellable operating leases. The future March 31, 2019 - - - - - - - - - - - - -	March 31, 2018 7,935.85 e minimum lease payments in respect of March 31, 2018 328.34 242.86 429.72 sclosure requirements specified in sub-
31 March 2018 during the year Investments Ind AS adjustments 31 March 2019	Estimated amount of contracts remaining to be e (net of advances) ancial commitments Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. ating lease (a) The Company has also taken a commercial hese properties as on 31 March 2019 is as folic Particulars Not later than one year Later than one year Later than one year Later than five years Payments of lease rentals during the year (inclu akh (2018. 47.88 lakh) (c)There are no exceptional/restrictive covenant The operations of the Company are classified section 4 of Section 186 of the Act in respect of recipient companies, are not applicable to the C	n Cumulative Convertible Preference shares of premises under cancellable operating leases. property and certain residential premises on non- ws: des recorded under personnel costs INR 19.89 s in the lease agreements. as 'infrastructure facilities' as defined under Sch of loans given, investment made or guarantee giv ompany.	12.16 March 31, 2019 7.935.85 cancellable operating leases. The future March 31, 2019	March 31, 2018 7,935.85 e minimum lease payments in respect of March 31, 2018 328.34 242.86 429.72 sclosure requirements specified in sub-
	Estimated amount of contracts remaining to be e (net of advances) ancial commitments Particulars Commitments towards investment in 0.001% No TRIL Constructions Ltd. Atting lease (a) The Company has taken various residential ( b) The Company has also taken a commercial hese properties as on 31 March 2019 is as follo Particulars Not later than one year .ater than one year but not later than five years .ater than one year .ater than one year .ater than five years .ater than five yea	n Cumulative Convertible Preference shares of premises under cancellable operating leases. property and certain residential premises on non- ws: des recorded under personnel costs INR 19.89 s in the lease agreements. as 'infrastructure facilities' as defined under Scho of loans given, investment made or guarantee giv ompany. on 31 March 2019 (including investments made	12.16 March 31, 2019 7.935.85 cancellable operating leases. The future March 31, 2019  384.24 edule VI to the Act. Accordingly, the diren or security provided and the related te during the year) Investments made Sale of	March 31, 2018 7,935.85 e minimum lease payments in respect of March 31, 2018 328.34 242.86 429.72 sclosure requirements specified in sub- disclosures on purposes/ utilisation by

INS	1,52,567.00	2,250.00	-	27,510.30	1,82,327.30
International Infrabuild Private Limited	1,577.51			(1,427.23)	150.28
HV Farms Private Limited	7.94		-	(6.94)	1.00
MIA Infrastructure Private Limited					
TRIF Real Estate Development Limited	6,989.09		-	625,91	7,615.00
Deemed Investment in TRIL Amritsar Projects Limited		-	-	2,487.94	2,487.9
TRIL Amritsar Projects Limited	5,935.92		5	(2,443.92)	3,492.0
TRIL Infopark Limited	89,568.11			14,354.74	1,03,922.8
TRIL Constructions Limited	3,111.24			(659.72)	2,451.5
Wellkept Facility Management Services Private Limited	Sec. Sec.				
Deemed Investment in TRIL Urban Transport Private Limited	4		-	240.43	240.4
TRIL Urban Transport Private Limited	304.33	2,250.00		(220.33)	2,334.0
TRIF Gurgaon Housing Projects Private Limited	1				
Deemed Investment in TRIL Roads Private Limited				2,421.61	2,421.6
TRIL Roads Private Limited	18,572.86	-		16,150.14	34,723.0
Gurgaon Realtech Limited	8,154.62	-		(1,247.62)	6,907.0
Deemed Investment in Gurgaon Constructwell Private Limited		-		619.68	619.6
Gurgaon Constructwell Private Limited	11,552.66			(2,953.66)	8,599.0
Arrow Infra Estates Private Limited	6,792.72			(430.72)	6,362.0
Acme Living Solutions Private Limited					
estment in equity shares					
vestment in subsidiaries					





#### Tata Realty and Infrastructure Limited

#### CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements

for the year ended 31 March 2019

(Currency: Indian rupees in lakhs)

Name of the entity	31 March 2018	Investments made during the year	Sale of Investments during the year	Ind AS adjustments	31 March 2019
Investment in Joint venture companies			during the year		
Investment in equity shares					
TRIL IT4 Private Limited	21,636,53			2,850,47	24 407 00
Industrial Minerals and Chemicals Company Private Limited	21,919,74			350.86	24,487.00
Mikado Realtors Pvt. Ltd.	31,258.72			1,969,48	22,270.60
	74,814.99			5,170.82	79,985.81
Investment in Preference shares					
TRIL Constructions Limited	7 550 00				
TRIL Infopark Limited	7,559.00			(1,588.60)	5,970.40
The mopula childe	42,729.00	in the second se		6,849.00	49,578.00
Investment in Debentures	50,288.00			5,260.40	55,548.40
TRIL Urban Transport Private Limited					
TRIL Roads Private Limited	4,338.00	304.04		(482.04)	4,160.00
TRIL IT4 Private Limited	20,544.00	2,450.00		(429.00)	22,565.00
TRIL Infopark Limited	8,140.00		÷		8,140.00
HV Farms Private Limited	35,608.00			5,665.00	41,273.00
	1,148.00	*		40.00	1,188.00
International Infrabuilds Private Limited	305,08			(93.52)	211.56
Industrial Minerals and Chemicals Company Private Limited		10,47'3.96		162.74	10,636.70
	70,083.08	13,228.00		4,863.18	88,174.26
B. Trade Investments					
nvestment in Mutual Funds					
Tata money market mutual funds	6,966.16	1,45,427.98	1,52,102.74	20.02	311.42
	6,966.16	1,45,427.98	1,52,102.74	20.02	311.42
Total	3.54.719.23	1.60.905.98	1 52 102 74	42 824 72	4 06 347 19

1,52,102.74

42,824.72

4,06,347.18

39 Disclosure of transactions with Related Parties, as required by Indian Accounting Standard (Ind AS) - 24 "Related Party Disclosures" has been set out below:

(a) List of Related Parties:

Holding Company:

Tata Sons Private Limited

Subsidiary Companies:

Acme Living Solutions Private Limited

Arrow Infraestate Private Limited

Dharamshala Ropeway Limited (subsidiary of TRIL Urban Transport Private Limited)

Durg Shivnath Expressways Private Limited (Formerly known as SMS Shivnath Infrastructure Pvt Ltd) (wholly owned subsidiary of TRPL Roadways Private Limited) Gurgaon Constructwell Private Limited

Gurgaon Realtech Limited

Hampi Expressways Private Limited (wholly owned subsidiary of TRIL Roads Private Limited)

HV Farms Private Limited

International Infrabuild Private Limited

Manali Ropeway Private Limited (subsidiary of TRIL Urban Transport Private Limited)

Matheran Rope-Way Private Limited (subsidiary of TRIL Urban Transport Private Limited)

MIA Infrastructure Private Limited

TRIF Gurgaon Housing Projects Private Limited

TRIF Real Estate And Development Limited

TRIL Amritsar Projects Limited (formerly known as TRIF Amritsar Projects Limited)

**TRIL Constructions Limited** 

TRIL Infopark Limited

TRIL Roads Private Limited

TRIL Urban Transport Private Limited

TRPL Roadways Private Limited (wholly-owned subsidiary of TRIL Roads Private Limited w.e.f. March 20, 2017)

Uchit Expressways Private Limited (wholly owned subsidiary of TRIL Roads Private Limited)

Wellkept Facility Management Services Private Limited (Previously known as TRIL Hospitality Private Limited)

#### Joint Venture:

A & T Road Construction Management and Operation Private Limited (Investment is held by TRIL Roads Private Limited, which is 100% subsidiary of Tata Realty and Infrastructure Limited)

Industrial Minerals and Chemicals Private Limited Mikado Realtors Private Limited

Pune Solapur Expressways Private Limited (Investment is held by TRIL Roads Private Limited, which is 100% subsidiary of Tata Realty and Infrastructure Limited) TRIL IT4 Private Limited (Previously known as Albrecht Builder Private Limited)

#### Other related parties with whom transactions have taken place during the year:

Fellow Subsidiaries:

**Ewart Investments Limited** Infinity Retail Limited

One Colombo Project (Private) Limited

- Tata AIA Life Insurance Company Limited (Formerly known as Tata AIG Life Insurance Company Limited) Tata AIG General Insurance Limited
- Tata Asset Management Limited
- Tata Business Excellence Group (A Division of Tata Sons Limited)
- Tata Capital Financial Services Limited
- Tata Consultancy Services Limited

Tata Consulting Engineers Limited

Tata Housing Development Company Limited

- Tata Quality Management Services (A Division of Tata Sons Limited)
- TC Travel and Services Limited (up to October 31, 2017)

#### Key Managerial Personnel:

Sanjay Dutt Sanjay Ubale Sanjay Sharma wind Chokhany

in

d

HASKING

Chartered Accountants

4

Managing Director - upto 31st March 2018 Chief Financial Officer - w.e.f. 10th September 2018 Chief Financial Officer - upto 28th February 2019 Company Secretary

Managing Director & CEO - w.e.f. 1st April 2018



-	Nature of Transactions / relationship / major parties	Holding company	Subsidiary companies	Fellow subsidiaries	Associates / Joint Ventures	Total
1	Unsecured loan taken			14,000.00		44.000.00
		(-)	(-)	(-)	()	14,000.00
1	Tata Housing Development Company Limited		(-)	14,000.00	(-)	(-
		(-)	(-)	(-)	(-)	
ľ	Interest expenses on unsecured loans			20.71	(-)	(-
		(-)	(-)	(-)		
	Tata Housing Development Company Limited			20.71	(-)	(-
		(-)	(-)	(-)	()	12 C 1 C 1 C 1 C 1 C 1 C 1 C 1 C 1 C 1 C
Ĩ	Purchase of fixed assets		(-)	(-)	(-)	(-
1		(-)	(-)	(7.12)	(-)	(7.1)
	Infinity Retail Limited	-		-	-	
		(-)	(-)	(7.12)	(-)	(7.12
1	Investments in Equity,CCD,OCD during the year		5,004.04	-	10,473,96	15,478.00
		(-)	(15,922.56)	(-)	(7,983.46)	(23,906.02
1	Mikado Realtors Private Limited		-	-	(1,000.10)	(20,000.01
		(-)	(-)	(-)	(6,239.28)	(6,239.28
1	TRIL Roads Private Limited		2,450.00		(0,200,20)	2,450.00
		(-)	(12,815.00)	(-)	(-)	(12,815.00
1	TRIL Urban Transport Private Limited	-	2,554.04		(-)	2,554.04
		(-)	(3,035.38)	(-)	(-)	(3,035.38
ij	Industrial Minerals and Chemicals Private Limited		(0,000.00)	(-)	10,473.96	10,473.96
		(-)	(-)	(-)	(1,744.18)	(1,744.18
b	Others		(7	(-)	(1,744,10)	(1,744.10
		(-)	(72.19)	(-)	(-)	(72.19
	Sale of fixed assets	28.32	(12.10)	2.19	(-)	30.50
		(-)	(-)	(-)	(-)	
1	Tata Sons Private Limited	28,32		(-)	(-)	28.32
		(-)	(-)	(-)	(-)	
i	Ewart Investments Limited	(-)	(-)	2.19	(-)	(-) 2.19
		(-)	(-)	(-)	(-)	2.15
1	nter Corporate Deposit Given		24,717.17	(-)	(-)	24,717.17
		(-)	(16,080.00)	()	()	
5	Arrow Infraestate Private Limited	17	2,253.00	(-)	(-)	(16,080.00
	anow mindestate i mate climited					2,253.00
h	Gurgaon Constructwell Private Limited	(-)	(3.00)	(-)	(-)	(3.00
	Surgaon Constructiven Private Ennited	11	4,664.79	-		4,664.79
ĥ	nternational Infrabuild Private Limited	(-)	(16.00)	(-)	(-)	(16.00
1	international infrabulid Private Limited		1,540.00	-	-	1,540.00
1	Curroon Dooltook Linded	(-)	(1,662.00)	(-)	(-)	(1,662.00
1	Gurgaon Realtech Limited		2,657.00	5. A.	-	2,657.00
1	TPU Deade Brinste United	(-)	(160.00)	(-)	(-)	(160.00
	TRIL Roads Private Limited		8,363.00	1 - 195 A.	-	8,363.00
4		(-)	(12,616.00)	(-)	(-)	(12,616.00
	TRIF Real Estate And Development Limited		2,500.00		-	2,500.00
		(-)	(500.00)	(-)	(-)	(500.00
	TRIL Urban Transport Private Limited		1,193.48	÷	-	1,193.48
		(-)	(-)	(-)	(-)	(-)
1	Others		1,545.90			1,545.90
1		(-)	(1,123.00)	(-)	(-)	(1,123.00
			7,490,33	-	¥	7,490.33
	nter Corporate Deposit Refunded					104 005 05
1		- (-)	(21,885.65)	(-)	(-)	(21,885.65
1	nter Corporate Deposit Refunded Arrow Infraestate Private Limited			(-)	(-)	
1		(-)	(21,885.65)			2,453.52
1		(-)	(21,885.65) 2,453.52			2,453.52 (1,092.88
1	Arrow Infraestate Private Limited Gurgaon Realtech Limited	(-) - (-)	(21,885.65) 2,453.52 (1,092.88)	(-)	- (-)	2,453.52 (1,092.88 3,228.81
1	Arrow Infraestate Private Limited	(-) - (-)	(21,885.65) 2,453.52 (1,092.88) 3,228.81	(-)	(-)	2,453.52 (1,092.88 3,228.81 (992.77
1	Arrow Infraestate Private Limited Gurgaon Realtech Limited	(-) - (-) - (-) -	(21,885.65) 2,453.52 (1,092.88) 3,228.81 (992.77)	(-)	(-) (-)	2,453.52 (1,092.88 3,228.81 (992.77 1,030.00
1 1	Arrow Infraestate Private Limited Gurgaon Realtech Limited	(-) - (-)	(21,885.65) 2,453.52 (1,092.88) 3,228.81 (992.77) 1,030.00	(-)	- (-)	2,453.52 (1,092.88 3,228.81 (992.77 1,030.00
1 1	Arrow Infraestate Private Limited Gurgaon Realtech Limited FRIL Amritsar Projects Limited	(-) (-) (-)	(21,885.65) 2,453.52 (1,092.88) 3,228.81 (992.77) 1,030.00 (9,300.00)	(-) (-) (-)	(-) (-) (-)	2,453.52 (1,092.88 3,228.81 (992.77 1,030.00 (9,300.00
1	Arrow Infraestate Private Limited Gurgaon Realtech Limited FRIL Amritsar Projects Limited FRIL Roads Private Limited	(-) - (-) - (-) -	(21,885.65) 2,453.52 (1,092.88) 3,228.81 (992.77) 1,030.00 (9,300.00) (8,500.00)	(-)	(-) (-)	2,453.52 (1,092.88 3,228.81 (992.77 1,030.00 (9,300.00 - (8,500.00
1	Arrow Infraestate Private Limited Gurgaon Realtech Limited FRIL Amritsar Projects Limited	(-) (-) (-) (-)	(21,885.65) 2,453.52 (1,092.88) 3,228.81 (992.77) 1,030.00 (9,300.00) (8,500.00) 765.00	(-) (-) (-)	(-) (-) (-) (-)	2,453.52 (1,092.88 3,228.81 (992.77 1,030.00 (9,300.00 (8,500.00 765.00
1 1 1 1 1 1	Arrow Infraestate Private Limited Gurgaon Realtech Limited FRIL Amritsar Projects Limited FRIL Roads Private Limited	(-) (-) (-)	(21,885.65) 2,453.52 (1,092.88) 3,228.81 (992.77) 1,030.00 (9,300.00) (8,500.00)	(-) (-) (-)	(-) (-) (-)	(21,885.65 2,453.52 (1,092.88 3,228.81 (992.77 1,030.00 (9,300.00 (9,300.00 (8,500.00 765.00 (-) 13.00





)	Nature of Transactions / relationship / major parties	Holding company	Subsidiary companies	Fellow subsidiaries	Associates / Joint Ventures	Total
1.1	Project Management Consultancy fees		2,158,91		1,118.49	3,277.40
		(-)	(3.016.70)	(251.24)	(1,313.72)	(4,581.66)
	Gurgaon Realtech Limited	-	1,045.46	(10 112 1)	(1,010.72)	1,045.46
		(-)	(1,402.34)	(-)	(-)	(1,402.34)
	International Infrabuild Private Limited		14.35			14.35
		(-)	(470.61)	(-)	(-)	(470.61)
	Mikado Realtors Private Limited			- 10	1,118.49	1,118.49
		(-)	(-)	(-)	(1,313.72)	(1,313.72)
	TRIF Real Estate And Development Limited	-	208.18	-	-	208.18
		(-)	(299.56)	(-)	(-)	(299.56)
	TRIL Infopark Limited		422.92	-	-	422.92
		(-)	(116.26)	(-)	(-)	(116.26
	Uchit Expressways Private Limited	•	378.00			378.00
		(-)	(-)	(-)	(-)	(-)
	Tata Consultancy Services Limited	-	-	-	-	
	12070	(-)	(-)	(251.24)	(-)	(251.24)
	Others	÷	90.00	-		90.00
		(-)	(727.93)	(-)	(-)	(727.93)
	Asset Management Fees	-	972.88	-	148.40	1,121.28
		(-)	(855.37)	(-)	(143.40)	(998.78)
	TRIL Infopark Limited	-	882.78		-	882.78
		(-)	(769.33)	(-)	(-)	(769.33)
	TRIL IT4 Private Limited	÷			148.40	148.40
		(-)	(-)	(-)	(143.40)	(143.40
	Others		90.10			90.10
		· (-)	(86.05)	(-)	(-)	(86.05)
	Interest Income	2	1,609.75	-	1,465.20	3,074.95
		(-)	(2,617.73)	(-)	(1,465.20)	(4,082.93
	TRIL Roads Private Limited		1,164.20	-	-	1,164.20
		(-)	(1,074.60)	(-)	(-)	(1,074.60)
	TRIL Amritsar Projects Limited				- 1	-
		(-)	(1,463.34)	(-)	(-)	(1,463.34)
	TRIL IT4 Private Limited				1,465.20	1,465.20
		(-)	(-)	(-)	(1,465.20)	(1,465.20)
	Others	-	445.55		-	445.55
	Automatic	(-)	(79.80)	(-)	(-)	(79.80)
	Other Expenses	2.20	1.4	530.21	-	532.41
		(99.08)	(51.03)	(407.33)	(-)	(557.43)
	Ewart Investments Limited			175.30	-	175.30
		(-)	(-)	(237.55)	(-)	(237.55)
	Tata Sons Private Limited	2.20		-		2.20
		(99.08)	(-)	(-)	(-)	(99.08)
	Tata AIG General Insurance Limited	-		151.46	-1	151.46
		(-)	(-)	(93.99)	(-)	(93.99
	TC Travel and Services Limited	1.00	1.1		- 4° °	
		(-)	(-)	(60.84)	(-)	(60.84)
	Tata Consulting Engineers Limited	-		150.03		150.03
		(-)	(-)	(-)	(-)	-
	Others		-	53.42	-	53.42
ŝ		(-)	(51.03)	(14.95)	(-)	(65.98)
	Reimbursement of expenses	7.61	1,093.47	604.61	29.38	1,735.08
		(55.00)	(1,114.37)	(-)	(38.02)	(1,207.39)
	Arrow Infraestate Private Limited		0.02		- Fil	0.02
		(-)	(-)	(-)	(-)	(-)
	Gurgaon Realtech Limited		214.17		-	214.17
		(-)	(174.98)	(-)	(-)	(174.98)
	Gurgaon Constructwell Private Limited	-	0.04		-	0.04
		(-)	(-)	(-)	(-)	(-)
	International Infrabuild Private Limited	-	68.47		-	68.47
		(-)	(124.85)	(-)	(-)	(124.85)
	TRIL Roads Private Limited		425.53		- 1	425.53
		(-)	(360.02)	(-)	(-)	(360.02
	TRIL Urban Transport Private Limited		187.53	-	-	187.53
		(-)	(173.39)	(-)	(-)	(173.39
	Tata Housing Development Co. Limited			604.61	3.0	604.61
		(-)	(-)	(-)	(-)	(-)
	Others	7.61	197.71	100	29.38	234.71
		(55.00)	(281.13)	(-)	(38.02)	(374.15)





AP 4 OITTA

-	Nature of Transactions / relationship / major parties	Holding company	Subsidiary companies	Fellow subsidiaries	Associates / Joint Ventures	Total
	Deposit Refund Received	150.00	-	171	-	150.
2	Tata Sons Private Limited	(-)	(-)	(-)	(-)	
	Tata Sons Private Limited	150.00 (-)	(-)	(-)	- ()	150.
1	Deposit Given	(-)		34.90	(-)	34.
		(-)	(-)	(-)	(-)	-
	Tata Consulting Engineers Limited	-	-	34.90	-	34.
1	Employee Benefit Transfer	(-)	(-)	(-)	(-)	56.
		(-)	(-)	(-)	(-)	50.
1	Tata Sons Private Limited	56.50	-	-	-	56.
-	Prepaid Expenses	(-)	(-)	(-)	(-)	
	repaid Expenses	(-)	(-)	(61.30)	(-)	(61.
Ĩ	Tata AIG General Insurance Limited		-	-	-	(01.
1	Others	(-)	(-)	(52.93)	(-)	(52.
1	Others	- (-)	- 15	-	-	
-	Outstanding Balances Receivables	(-)	(-)	(8.37)	(-)	(8.
	Inter Corporate Deposit - Current - Unsecured			-	4.0	
		(-)	(4,232.53)	(-)	(-)	(4,232
1	Arrow Infraestate Private Limited	-	-	1.	÷.,	
1	Gurgaon Realtech Limited	(-)	(440.52)	(-)	(-)	(440.
		(-)	(871.81)	(-)	(-)	(871.
1	International Infrabuild Private Limited	-			-	
1	TRIF Real Estate And Development Limited	(-)	(1,462.00)	(-)	(-)	(1,462.
	This root Educe And Development Einned	(-)	(500.00)	(-)	(-)	(500
1	TRIL Constructions Limited	-	-		-	(000
		(-)	(670.00)	(-)	(-)	(670
	TRIL Roads Private Limited	-	-	-		
i	TRIL Amritsar Projects Limited	(-)	(-)	(-)	(-)	
		(-)	(-)	(-)	(-)	
(	Others		-	-	-	
ī	nter Corporate Deposit - Non-Current - Unsecured	(-)	(288.21) 57,649.38	(-)	(-)	(288.
		(-)	(36,190.00)	(-)	(-)	(36,190.
1	TRIL Amritsar Projects Limited	-	19,372.00	1.4		19,372.
-	TRIL Roads Private Limited	(-)	(19,752.00)	(-)	(-)	(19,752
	The Roads I mate Limited	(-)	24,801.00 (16,438.00)	(-)	(-)	24,801 (16,438
(	Others		13,476.38	-	-	13,476
-		(-)	(-)	(-)	(-)	
1	Interest Accrued but not due		2,806.97	-	64.23	2,871
1	TRIL Amritsar Projects Limited	(-)	(2,736.05)	(-)	(64.23)	(2,800
		(-)	(1,317.01)	(-)	(-)	(1,317
1	TRIL Roads Private Limited		1,047.78		1	1,047
(	Others	(-)	(1,367.14) 442.18	(-)	(-) 64.23	(1,367 506
		(-)	(51.91)	(-)	(64.23)	(116
(	Other Recoverable		1,429.19	652.98	14.46	2,096
	A	(5.90)	(4,137.12)	(4.37)	(11.20)	(4,158.
1	Arrow Infraestate Private Limited	(-)	0.02 (801.58)	()	- 15	0.
(	Gurgaon Realtech Limited	(-)	(001.50)	(-)	(-)	(801.
		(-)	(1,083.20)	(-)	(-)	(1,083.
(	Gurgaon Constructwell Private Limited	-	0.04		÷	0
-	TRIL Urban Transport Private Limited	(-)	(1,359.38) 759.28	(-)	(-)	(1,359 759
		(-)	(572.36)	(-)	(-)	(572
1	TRIL Constructions Limited		199.17	-	-	199
	TPIL Panda Drivata Licellard	(-)	(113.42)	(-)	(-)	(113
	TRIL Roads Private Limited	(-)	198.77	(-)	- ()	198
100	Tata Housing Development Co. Ltd.		(-)	652.98	(-)	652
		(-)	(-)	(-)	(-)	
1	others	(5.90)	271.91 (207.17)	(4.37)	14.46	286
S					(11.20)	(228.



\_

	Nature of Transactions / relationship / major parties	Holding company	Subsidiary companies	Fellow subsidiaries	Associates / Joint Ventures	Total
1	Security Deposits - Non-Current - Unsecured	(150.00)	- (-)	(300.00)		-
17	Tata Sons Private Limited	(150.00)	(-)	(300.00)	(-)	(450.0
i	Ewart Investments Limited	(150.00)	(-)	(-)	(-)	(150.00
		(-)	(-)	(300.00)	(-)	(300.00
7	Trade Receivable		749.16	300.00	287.69	1,336.8
		(-)	(985.43)	(300.00)	(788.72)	(2,074.1
1	Gurgaon Realtech Limited	-	289.37	(	(100:12)	289.3
		(-)	(252.12)	(-)	(-)	(252.1
1	TRIF Real Estate And Development Limited	-	175.79			175.7
		(-)	(-)	(-)	(-)	(
1	TRIL Infopark Limited		234.40			234.4
		(-)	(665.20)	(-)	(-)	(665.2
1	Mikado Realtors Private Limited	-	(		272.83	272.8
		(-)	(-)	(-)	(788.72)	(788.7
1	nternational Infrabuild Private Limited	( )	1.67	(-)	(100.12)	1.6
		(-)	(17.02)	(-)		
1	Tata Consultancy Services Limited	(-)	(17.02)	300.00	(-)	(17.0
	Tale ophotically dervices Linited	1	-		-	300.0
(	Others	(-)	(-)	(300.00)	(-)	(300.0
1	Stiers		47.94		14.86	62.8
7	Provision for later Comparts Devent	(-)	(51.09)	(-)	(-)	(51.0
1	Provision for Inter Corporate Deposit	5.0	95.00	-	-	95.0
1		(-)	(95.00)	(-)	(-)	(95.0
	TRIL Urban Transport Private Limited	-	95.00			95.0
-		(-)	(95.00)	(-)	(-)	(95.0
F	Provision for Interest Accrued but not due		10.00	4	24	10.0
_		(-)	(10.00)	(-)	(-)	(10.0
1	TRIL Urban Transport Private Limited		10.00	-	-	10.0
ć		(-)	(10.00)	(-)	(-)	(10.0
F	Provision for Advances recoverable		35.00	-	-	35.0
Ľ.		(-)	(35.00)	(-)	(-)	(35.0
1	TRIL Urban Transport Private Limited	-	35.00	-	-	35.0
2		(-)	(35.00)	(-)	(-)	(35.0
1	Amount Written off - Interest Accrued but not due	-	-	-	-	
1		(-)	(6.00)	(-)	(-)	(6.0
F	Acme Living Solutions Private Limited	-		-	-	-
		(-)	(6.00)	(-)	(-)	(6.0
7	Amount Written off - Inter Corporate Deposit		-			(0.0
	and a second	(-)	(9.50)	(-)	(-)	(9.5
7	Acme Living Solutions Private Limited		(0.00)	-	(-)	(5.5
		(-)	(1.00)			
1	<b>TRIF</b> Gurgaon Housing Projects Private Limited	(-)	(1.00)	(-)	(-)	(1.0
	The suggest floating flogests findet Einned	1.5	(2.00)			(2.0
L	Welkept Facility Management Services Private Limited	(-)	(2.00)	(-)	(-)	(2.0
1	remept r denty management dervices i rivate Linited	(-)	(6.50)	(-)	- (-)	10 5
0	Dutstanding Balances Payable	(-)	(0.50)	(-)	(-)	(6.5
	owards unsecured loans	4		14,000.00		14,000.0
1		(-)	(-)	(-)	-	
1	Tata Housing Development Co. Ltd.	(-)	(-)	14,000.00	(-)	14 000 0
1	and housing borolopinoin oo. Eld.		(-)			14,000.0
t	owards interest on unsecured loans	(-)	(-)	(-) 18.64	(-)	(18.6
1		15			-	
1	Tata Housing Development Co. Ltd.	(-)	(-)	(-)	(-)	(
1	ata notaing Development CO. Ltd.	-	5.	18.64	-	18.6
-	Advances as a fired	(-)	(-)	(-)	(-)	(
+	Advances received	-		100		1.0.0
-		(-)	(59.08)	(66.12)	(-)	(125.2
1	RIF Real Estate And Development Limited		2.00		-	-
ļ		(-)	(58.67)	(-)	(-)	(58.6
1	Tata Consultancy Services Limited		-		2	
		(-)	(-)	(66.12)	(-)	(66.1
				a second		
¢	Others		-	-	· ·	





	Nature of Transactions / relationship / major parties	Holding company	Subsidiary companies	Fellow subsidiaries	Associates / Joint Ventures	Total
1	Trade Payable		-	6.54		6.54
	and the second se	(-)	(36.89)	(2.90)	(-)	(39.79)
1	Tata Consultancy Services Limited		-1	6.54	÷ .	6.54
	and president claim sound towns	(-)	(-)	(2.90)	(-)	(2.90)
	TRIF Real Estate And Development Limited			-	· · · · ·	-
	The second second second second second second	(-)	(36.89)	(-)	(-)	(36.89)
	guarantee given outstanding	,	÷P	-		1.00
		(-)	(-)	(-)	(-)	(-)
	Managerial remuneration					1,521.32 (754.64
	Sanjay Dutt*					1,007.97
	Sanjay Ubale					-
	Sanjay Obale					(414.61
	Sanjay Sharma					179.44
	Ganjay Ghanna					(-)
	Arvind Chokhany					224.49
						(236.83
	Vinay Gaokar					109.41
	and a second second					(103.20

Figures in brackets pertains to previous year.
 \* Recovery of managerial remuneration from fellow subsidiary is not netted off for this disclosure.





#### 40 Segment Reporting

The Company is engaged in development of real estate and infrastructure facilities for residential use and project management consultancy services for real estate and infrastructure development. Thus, the Company is engaged in three business segments viz. development of residential property for outright sale, project management and consultancy services and Investment and Lending services. Further, the Company is engaged in providing services in domestic market only. Hence, there are no separate reportable geographical segments.

	Development of resident of resident of resident of the second sec		Project managemen servio		Investment and L	ending services	Total		
	31 March 2019	31 March 2018	31 March 2019	31 March 2018	31 March 2019	31 March 2018	31 March 2019	31 March 2018	
REVENUE									
Net sales									
Segment Revenue	10,253.03	7,830.17	4,401.88	5,597.24	12,466,89	4,860,10	27,121.80	18,287,51	
inter segment sales	4						21,121.00	10,201.01	
Total revenue	10,253.03	7,830.17	4,401.88	5,597.24	12,466.89	4,860.10	27,121.80	18,287.51	
Segment Expenses	8,236.33	6,789.14	1,614,15	188.48	10,663.02	8,641.58	20,513.50	15,619.20	
RESULT									
Segment Result	2,016.70	1,041.03	2,787.73	5,408.76	1,803.87	(3,781.48)	6,608.30	2,668.31	
Unallocated income			-				125.44	337.87	
Unallocated expenses	4						932.43	625.37	
Operating Profit / (Loss)	2,016.70	1,041.03	2,787.73	5,408.76	1,803.87	(3,781.48)	5,801.31	2,380.81	
Finance costs allocated	4	- C.		- C.	18,799.22	15,068.21	18,799.22	15,068.21	
Finance cost unallocated	-				tri) trian		4,189.69	4,603,69	
Depreciation / Amortisation	13.84	21.46	63,23	62,20	63.23	62.20	140.30	145.87	
Net Profit / (Loss )	2,002.87	1,019.57	2,724.50	5,346.56	(17,058.58)	(18.911.90)	(17,327.90)	(17,436.96)	
OTHER INFORMATION									
ASSETS									
Segment Assets	43,382.01	45,369,70	1.308.00	2,014,44	4,76,144,51	4,18,813,43	5,20,834,51	4.66.197.57	
Unallocated assets	-					4,10,010,40	6,012.90	5,370.81	
Total Assets	43,382.01	45,369.70	1,308.00	2,014.44	4,76,144.51	4,18,813.43	5,26,847.41	4,71,568.38	
LIABILITIES									
Segment Liabilities	6,691,48	5,695.44		2.4	2,26,618.52	1,85,370.61	2.33.310.00	1,91,066.05	
Unallocated Liabilities	-	-	-			1,00,070,01	94,851.43	90,756.16	
Total Liabilities	6,691.48	5,695.44		-	2,26,618.52	1,85,370.61	3,28,161.43	2,81,822.21	
Cost incurred to acquire Segment		de tra							
ixed assets during the year	1.71	8.66	268.41	58.77	268,41	58.77	538.52	126.20	

#### Segment accounting policies

Segment accounting policies are in line with accounting policies of the Company. In addition, the following specific accounting policies have been followed for segment reporting: Segment revenue includes income directly identifiable with the segments.

Expenses that are directly identifiable with the segments are considered for determining the segment result. Expenses which relate to the Company as a whole and not allocable to segments and expenses which relate to the operating activities of the segment but are impracticable to allocate to the segment, are included under "Unallocable expenses".

Income which relates to the Company as a whole and not allocable to segments is included in Unallocable Income and netted off from Unallocable expenses.

Segment assets and liabilities include those directly identifiable with the respective segments. Unallocable corporate assets and liabilities represent the assets and liabilities that relate to the Company as a whole and not allocable to any segment.

#### 41 Employee Benefits:

(i) The Company has adopted Ind AS 19 on "Employee Benefits" as prescribed by the Companies (Accounting Standards) Rules, 2006 issued by the Central Government.

#### (ii) Contribution to Provident fund

·····································	31 March 2019	31 March 2018
Contribution to provident fund recognised as an expense under "Employee" benefits".	134.56	120.41





## Tata Realty and Infrastructure Limited

#### CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements

for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

(iii) Defined Benefit Plans

#### Gratuity

The Company has a defined benefit gratuity plan. Every employee who has completed five years or more of service gets a gratuity on death or resignation or retirement calculated as per the Payment of Gratuity Act, 1972 with no ceiling.

a server to a state for an internet of the server of the s	31 March 2019	31 March 2018
I Change in the defined benefit obligation		
Liability at the beginning of the year	407.58	351.62
Interest cost	31.97	25.91
Current service cost	70.58	70.83
Benefits paid	(89.55)	(19.61
Actuarial loss on obligations	(52.65)	(21.17
Liability acquired on acquisition / (settled on Divestiture)		
Liability at the end of the year	367.93	407.58
Il Amount Recognised in the Balance Sheet		
Liability at the end of the year	367.93	407.58
Fair Value of Plan Assets at the end of the year	-	-
Difference	367.93	407.58
Amount recognised in the Balance Sheet	367.93	407.58
Expenses Recognised in the statement of profit and loss		
Current Service Cost	70.58	70.83
Interest Cost	31.97	25.91
Expected Return on Plan Assets	10111	-
Net Actuarial (Gain) / Loss To Be Recognised	(52.65)	(21.17
Past service cost		-
Expense Recognised in the statement of profit and loss	49.90	75.57
Balance Sheet Reconciliation		
Opening net liability	407.58	351.62
Expense as above	49.90	75.57
Employer's contribution received / (paid)	(89.55)	(19.61
Liability acquired on acquisition / (settled on Divestiture)		1
Amount recognised in Balance Sheet	367.93	407.58
Actuarial Assumptions :		
Discount Rate	7.05%	7.85%
Salary escalation	7.00%	7.00%

Attrition Rate: Directors - Nil, Age 21-30 years - 5%, Age 31-40 years - 3%, Age 41-59 years -2%

Estimates of future salary increases, considered in actuarial valuation, take account of inflation, seniority, promotion and other relevant factors, such as supply and demand in the employment market.

The Company's liability on account of gratuity is not funded and hence, the disclosures relating to the planned assets are not applicable.

Experience Adjustments:	31 March 2019	31 March 2018
Defined benefit obligation	367.93	407.58
Plan assets	-	-
Surplus / (Deficit)	(367.93)	(407.58)
Experience adjustment on plan liabilities		
Experience adjustment on plan assets		

#### (iv) Other long term employment benefits

Compensated absences

The liability towards compensated absences for the year ended 31 March 2019 recognised in the Balance Sheet based on actuarial valuation using the projected unit credit method amounted to INR 542.68 lakhs (2018: INR 541.53 lakhs) and the charge to the Statement of profit and loss amounted to INR 1.15 lakhs (2018: INR 24.51 lakhs).

#### Sensitivity analysis

Reasonably possible changes at the reporting date to one of the relevant actuarial assumptions, holding other assumptions constant, would have affected the defined benefit obligation by the amounts shown below.

int coto / 0/ m

Discount rate ( % movement)				
Salary escalation (% movement)	31 March 2019	31 March 2019	31 March 2018	31 March 2018
	Increase	Decrease	Increase	Decrease
Discount rate ( % movement)	(2.6)	2.7	(4.9)	5.3
Salary escalation (% movement)	2.7	(2.6)	5.3	(4.9)
Additional Rate (% movement)	(5.7)	8.6		





#### Tata Realty and Infrastructure Limited

#### CIN No U70102MH2007PLC168300 Notes to Ind AS financial statements

for the year ended 31 March 2019

(Currency: Indian rupees in lakhs)

42 Financial instruments – Fair values and risk management A. Accounting classification and fair values The following table shows the carrying amounts and fair values of financial assets and financial liabilities, including their levels in the fair value hierarchy. It does not include fair value information for financial assets and financial liabilities if the carrying amount is a reasonable approximation of fair value. The Company's secured loan under current maturities has been contracted at floating rates of interest, which are reset at short intervals. Accordingly, the carrying value of such long-term debt approximates fair value.

		Carrying an	nount		Fair value				
31 March 2019	FVTPL	FVTOCI	Amortised Cost	Tota	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Total	
Non-current financial assets									
- Equity instruments in subsidiaries, joint ventures and associate	τ.	2,62,313.10	*	2,62,313,10			2,62,313.10	2,62,313.11	
- Preference shares		55,548.40	- C - C -	55,548,40			55,548,40	55,548,40	
<ul> <li>Bonds and debentures</li> </ul>	69,397.56	10,636.70	8,140.00	88,174,26			88,174.26	88,174.20	
Long-term loans and advances	52,124.48		-	52,124,48			52,124.48	52,124.4	
Other non-current financial assets			0.75	0.75			0.75	52,124.40	
Current financial assets									
Current investments	311.42	-		311.42	311.42			311.42	
Trade receivables	1.1.1		1,397.81	1,397.81			1,397.81	1,397.8	
Cash and cash equivalents			445.25	445.25			445.25	445.25	
Other Bank Balances		-	444.69	444.69			444.69	444.69	
Short-term loans and advances			2,349.77	2,349.77			2,349.77	2,349.77	
Other financial assets	4		2,861.19	2,361.19			2,861,19	2,861.19	
Total Financial Assets	1,21,833.46	3,28,498.20	15,639.46	4,65,971.12	311.42		4,65,659.70	4,65,971.12	
Financial liabilities Non-current Financial liabilities									
Long-term borrowings	4		72,460.15	72,460.15			72,460,15	72,460.15	
Interest accrued but not due on borrowings Current Financial liabilities		-	11,425.58	11,425.58	-	6	11,425.58	11,425.58	
Short-term borrowings	-		94,685.15	94,685,15			94,685,15	94,685.15	
Trade and other payables other than MSME	-	÷.	3,458.03	3,458.03	-		3,458.03	3,458.03	
Other financial liability	7,122.00		1,17,970.20	1,25,092.20			1,25,092.20	1,25,092.20	
Total Financial Liabilities	7,122.00	1.	2,99,999.11	3,07,121.11			3,07,121.11	3,07,121.11	

		Carrying an	nount			Fair valu	le	
31 March 2018	FVTPL	FVTOCI	Amortised Cost	Total	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Total
Financial assets								
Non-current financial assets								
<ul> <li>Equity instruments in subsidiaries, joint ventures and associate</li> </ul>		2,27,381.99		2,27,381.99		-	2,27,381.99	2,27,381.99
<ul> <li>Preference shares</li> </ul>	1	50,288.00		50,288.00	-		50,288.00	50,288.00
<ul> <li>Bonds and debentures</li> </ul>	61,943.09		8,140.00	70,083.09			70,083.08	70,083.08
Long-term loans and advances	37,553.93	-		37,553.93		1.1	37,553,93	37,553.93
Other non-current financial assets	-		0.75	0.75			0.75	0.75
Current financial assets								
Current investments	6,966.15			6,966.15	6,966.15			6,966.15
Trade receivables	-		3,132.93	3,132.93	-		3,132.93	3,132.93
Cash and cash equivalents	-	-	955,96	955.96			955.96	955.96
Other Bank Balances	-		361,67	361.67	÷		361.67	361.67
Short-term loans and advances			13,369.52	13,369.52			13,369.52	13,369.52
Other financial assets			2,908.05	2,908.05		÷	2,908.05	2,908.05
Total Financial Assets	1,06,463.17	2,77,669.99	28,868.88	4,13,002.04	6,966.15		4,06,035.88	4,13,002.03
Financial liabilities								
Non-current Financial liabilities								
Long-term borrowings		-	1,82,310.91	1,82,310.91			1,82,310.91	1,82,310.91
interest accrued but not due on borrowings Current Financial liabilities		1	4,931.25	4,931.25	4	÷.	4,931.25	4,931.25
Short-term borrowings			59,694,74	59,694,74			59,694,74	59,694,74
Trade and other payables other than MSME	-		3,025.61	3,025.61	-		3,025.61	3,025.61
Other financial liability	6,682.00		7,725.41	14,407.41			14,407,41	14,407.41
	6,682.00		2,57,687.92	2,64,369.92	1		2,64,369.92	2,64,369.92





#### Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

### Notes to Ind AS financial statements

for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

#### FVTPL : Fair valued through profit or loss

FVTOCI : Fair valued through other comprehensive income B. Measurement of fair values

Valuation techniques and significant unobservable inputs The Company has appointed independent valuer to determine the fair value of each of its investments. The Company has given certain unobservable inputs to the valuer to compute the valuation. Considering micro market and industry scenario the valuer has derived valuation by using appropriate technique for valuation. The Company has accounted fair valuation gain / loss in value of its investments using this report.

The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for financial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used. 

Туре	Valuation technique	Significant	Inter-relationship between
		unobservable inputs	significant unobservable inputs and fair value measurement
Inquoted equity shares (Infopark)	Discounted cash flow technique: DCF method/analysis is a financial modelling technique based on explicit assumptions regarding the prospective income arising out of the subject property. In case of a	* Rent growth every three years (%) 2019: (2018: 12% - 18%). (2018: 12% - 18%).	Estimated fair value would increase (decrease) if expected lease rent were higher (lower).
	valuation of a large land parcel, where the development potential is realized over a period of time (i.e. time value of money comes into the picture) and also where there are no or few immediate similar	* Occupancy (%) 2019: 100 % (2018: 97.2%). * Capitalization Rate (%) 2019: 8.5% -	Estimated fair value would increase (decrease) if expected lease escalation were higher (lower).
	properties (i.e. comparable) available for comparison, DCF method considering relevant potential developments of the project is used.	9.0% (2018: 8.4% - 9.4%)	Estimated fair value would decrease if occupancy is lower.
Unquoted equity shares (Nagpur)	considening relevant potential developments of the project is used.	* Rent growth every years (%) 2019: 4.77% (2018: 4.77%), * Occupancy (%) 2019: 90% (2018: 90%), * Capitalization Rate (%) 2019: 9.0% (2018: 9.5%)	Estimated fair value would increase (decrease) if expected lease escalation were lower (higher).
Unquoted equity shares (IT 4)		* Rent growth every three years (%) 2019: 12% - 18% (2018: 12% - 18%). * Stabilised Occupancy (%) 2019: 93% (2018: 95%). * Capitalization Rate (%) 2019: 9.0% - 10.0% (2018: 9.5% - 10.5%)	•
Unquoted equity shares - Amritsar		* Rent growth every three years (%) 2019: 12% - 18% (2018: 12% - 18%), * Stabilised Occupancy (%) 2019: 80 % (2018: 90%). * Capitalization Rate (%) 2019: 9.0% - 10.0% (2018: 9.0% - 10.0%)	
Unquoted equity shares (IIPL)		* Rent growth every three years (%) 2019: 12% - 18% (2018: 12% - 18%) * Occupancy (%) 2019: 95% (2018: 90%). * Capitalization Rate (%) - Not applicable	
Unquoted equity shares (Gurgaon R/Arrow/Gurgaon C)		* Rent growth every three years (%) 2019: 12% - 18% (2018: 12% - 18%). * Occupancy (%) 2019: 95% (2018: 90%). * Capitalization Rate (%) 2019: 9.0% - 10.0% (2018: 5.5% - 10.5%)	
Unquoted equity shares & Computsory Convertible Debentures (IMCC)	Discounted cash flow technique: DCF method/analysis is a financial modelling technique based on explicit assumptions regarding the prospective income arising out of the subject property. In case of a valuation of a large land parcel, where the development potential is realized over a period of time (i.e. time value of money comes into the picture) and also where there are no or few immediate similar properties (i.e. comparable) available for companison.	Technique - Land Comparision Method has been used for valuation.	Estimated fair value would increase/ (decrease) if fair value of land increases/ decreases.
Investments in unlisted corporat			
Optionaliy & Fully Convertible Debentures-TRPL	The value of the plain vanilla debenture is generally estimated as present value of future expected cash flows. The Company has used Monte Carlo Simulation to estimate the fair value of the compulsarily convertible debentures and optionally convertible debentures.	2019: * Volatility of share price of comparable companies (21% of volatility) (2018 :44.7%)	2019: *If Equity value of unlisted company increases/(decreases) by 10%, then fair value would (decrease)/increase. * If there is an Increase / (decrease) of 10% volatility in shar prices of comparable companies , then the fair value would increase/(decrease)
Investments in unlisted corporat	e debt instruments		
Optionally Convertible Debentures-HV farms	The value of the plain vanilla debenture is generally estimated as present value of future expected cash flows. The Company has used Monte Carlo Simulation to estimate the fair value of the compulsarily convertible debentures and optionally convertible debentures.	2019: * Volatility of share price of comparable companies (37 5% of volatility) (2018 - 37.5 %)	2019: *If Equity value of unlisted company increases/(decreases) by 10%, then fair value would (decrease)/increase. *If there is an Increase / (decrease) of 10% volatility in shar prices of comparable companies , then the fair value would increase/(decrease)
Land and the second			
Optionally Convertible Debentures- TUTPL	The value of the plain vanilla debenture is generally estimated as present value of future expected cash flows. The Company has used Monte Carlo Simulation to estimate the fair value of the compulsarily convertible debentures and optionally	2019 : * Volatility of share price of comparable companies (45% of volatility) (2018 : 41%)	2019 : 'If Equity value of unlisted company increases/(decreases) by 10%, then fair value would (decrease)/increase. ' If there is an Increase / (decrease) of 10% volatility in shar
Compulsorily Convertible Debentures- Infopark	convertible debentures.	2019: * Volatility of share price of comparable companies (21% of volatility) (2018: 45%)	prices of comparable companies , then the fair value would increase/(decrease).





#### Tata Realty and Infrastructure Limited

CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements

for the year ended 31 March 2019

(Currency: Indian rupees in lakhs)

Туре	Valuation technique		Inter-relationship between	
Investments in preference share	15			
Fully Convertible Preference Shares- TRIL Infopark	The value is generally estimated as present value of future expected cash flows. The Company has used Monte Carlo Simulation to estimate the fair value of the compulsarily convertible preference and optionally convertible preference shares	2019: * Volatility of share price of comparable companies (21% of volatility) (2018: 45%)	2019: "If Equity value of unlisted company increases/(decreases) by 10%, then fair value would increase/(decrease). "If there is an increase / (decrease) of 10% volatility in share prices of comparable companies, then the fair value would decrease.	
Fully Convertible Preference Shares-TCL	The value is generally estimated as present value of future expected cash flows. The Company has used Monte Carlo Simulation to estimate the fair value of the compulsarily convertible preference and optionally convertible preference shares	Technique - Land Comparision Method has been used for valuation.	2019: *If Equity value of unlisted company increases/(decreases) by 10%, then fair value would (decrease)/increase.	
Derivative -Put option				
Tamil Nadu Industrial Corporation Limited ("TIDCO") Put option	The Company has used Monte Carlo Simulation to estimate the fair value of the options.	2019: * Volatility of share price of comparable companies (21% of volatility) (2018: 45%)	2019: *If Equity value of unlisted company increases/(decreases) by 10%, then fair value would (decrease)/increase. *If there is an Increase / (decrease) of 10% volatility in share prices of comparable companies, then the fair value would increase/(decrease).	
Put / call options				
The Indian Hotels Company Limited ("IHCL") Call option	The Company has used Monte Carlo Simulation to estimate the fair value of the options.	* Volatility of share price of comparable	2019: *If Equity value of unlisted company increases/(decreases) by 10%, then fair value would (decrease)/increase. *If there is an Increase (decrease) of 10% volatility in share prices of comparable companies, then the fair value would increase/(decrease).	
Amortised cost type items (including security deposits, loans , cash , trade receivables and payables )	Discounted cash flow approach: The valuation model considers the present value of expected payment, discounted using a risk adjusted discount rate.	Not applicable	Not applicable	

43 Disclosure in respect of Sale of Residential Flats
 a) Disclosure with respect to transition adjustment of IndAS 115

	Amount
Opening Retained Earnings (before Ind AS 115)	(17,728.54)
Reversal of revenue	(3,174,43)
Reversal of Cost of sale	1.870.38
Opening Retained Earning (After Ind AS 115	(19,032.59)
Increase in Inventory	1,870.38
Decrease in Trade Receivable (Other than related Party)	(3,174.43)

b) Disclosure in respect of Construction Contacts

Particulars	For the year ended 31 March 2019	For the year ended 31 March 2018	
Contract revenue recognised as revenue during the year	9,463.27	7,081.06	
Particulars	March 31, 2019	March 31, 2018	
Cumulative revenue recognised (net of Ind AS 115 adjustment INR 3,174.42 Lakhs,(2018: INR Nil))	1,07,331.21	1,01,042.35	
Cumulative costs recognised (net of Ind AS 115 adjustment INR 1,870.38 Lakhs,(2018: INR Nil))	72,606.23	68,156.2	
Cumulative margins accounted	34,724.98	32,886.1	
Advances paid	83.30	193.68	
Retention money payable	227.54	275.2	

#### 44 Financial instruments - Fair values and risk management

(i) Financial risk management The Company has exposure to the following risks arising from financial instruments: A. Credit risk B. Liquidity risk C. Market risk





#### **Risk management framework**

The Company's board of directors has overall responsibility for the establishment and oversight of the Company's risk management framework. The Company manages market risk through a treasury department, which evaluates and exercises independent control over the entire process of market risk management. The treasury department recommends risk management objectives and policies, which are approved by Board of Directors. The activities of this department include management of cash resources, borrowing strategies, and ensuring compliance with market risk limits and policies.

The Company's risk management policies are established to identify and analyse the risks faced by the Company, to set appropriate risk limits and controls and to monitor risks and adherence to limits. Risk management policies and systems are reviewed regularly to reflect changes in market conditions and the Company's activities. The Company, through its training and management standards and procedures, aims to maintain a disciplined and constructive control environment in which all employees understand their roles and obligations.

The audit committee oversees how management monitors compliance with the company's risk management policies and procedures, and reviews the adequacy of the risk management framework in relation to the risks faced by the Company. The audit committee is assisted in its oversight role by internal audit. Internal audit undertakes both regular and ad hoc reviews of risk management controls and procedures, the results of which are reported to the audit committee.

#### A. Credit Risk

Credit risk is the risk of financial loss to the Company if a customer or counterparty to a financial instrument fails to meet its contractual obligations, and arises principally from the Company's receivables from customers, loans and advances to related parties and investments at amortised cost. Credit risk is managed through credit approvals, establishing credit limits and continuously monitoring the creditworthiness of customers to which the Company grants credit terms in the normal course of business. The Company establishes an allowance for doubtful debts and impairment that represents its estimate of incurred losses in respect of trade and other receivables, loans and advances and investments.

#### Trade and other receivables

The Company's exposure to credit risk is influenced mainly by the individual characteristics of each customer. The demographics of the customer, including the default risk of the industry and country in which the customer operates, also has an influence on credit risk assessment. Credit risk is managed through credit approvals, establishing credit limits and continuously monitoring the creditworthiness of customers to which the Company grants credit terms in the normal course of business.

#### Credit risk for receivables pertaining to residential business

The risk for trade receivables pertaining to residential business is considered nil as the possession of the residential property is transferred only after the receipt of payment from the customer.

Particulars	March 31, 2019	March 31, 2018
Past due but not impaired	initian en 2010	Warch 51, 2016
Past due 1-90 days	643.12	
Past due 91-180 days		1,584.51
Past due 181-270 days	94.48	247.32
		710.93
Past due 271-365 days	96.93	277.71
Past due more than 365	563.28	312.46

#### Cash and cash equivalents

The Company held cash and bank balances with credit worthy banks and financial institutions of INR 890,69 lakhs and INR 1,318.38 lakhs as at 31 March 2019 and 31 March 2018 respectively. The credit worthiness of such banks and financial institutions is evaluated by the management on an ongoing basis and is considered to be good.

1,397.81

#### Derivatives

The derivatives are entered into with credit worthy counterparties. The credit worthiness is evaluated by the management on an ongoing basis and is considered to be good.

#### Security deposits given to lessors

The security deposit majorly pertains to rent deposit amounting to INR 22.64 The Company does not expect any losses from non-performance by theses counter-parties.

#### B. Liquidity risk

Liquidity risk is the risk that the Company will not be able to meet its financial obligations as they become due. The Company manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Company's reputation.

The Company has obtained fund and non-fund based working capital lines from banks, commercial papers issued to Mutual funds and through issue of debentures. The Company also constantly monitors funding options available in the debt and capital markets with a view to maintaining financial flexibility.

#### Exposure to liquidity risk

The table below analyses the Company's financial liabilities into relevant maturity groupings based on their contractual maturities for: \* all non derivative financial liabilities

\* net and gross settled derivative financial instruments for which the contractual maturities are essential for the understanding of the timing of the cash flows,

			Contr	actual cash flows		
As at 31 March, 2019	Carrying amount	Total	1 year or less	1-2 years	2-5 years	More than 5 years
Non-derivative financial liabili	ties					
Commercial paper issued to mutual funds	76,567.29	77,500.00	77,500.00	-	-	,
Inter Corporate Deposits	14,000.00	14,000.00	14,000,00			
Trade and other payables	3,458.03	3,458.03	3,458.03			
Bank overdraft	4,117.86	4,117.86	4,117.86		100	
Other financial liabilities	1,29,395.78	1,29,135,88	1,17,710.30	11,425.58		
Secured loan from Banks		Series a series	A lost serves			
Non convertible debentures	72,460.15	72,460.15		72,460.15		
	2,99,999.11	3,00,671.92	2,16,786.19	83,885.73		
Derivative financial liabilities						
Derivativės liabilities at fair value	7,122.00	7,122.00	7,122.00	•	14	- 1
	7,122.00	7,122.00	7,122.00	•		





3,132.93

#### Tata Realty and Infrastructure Limited

CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements

for the year ended 31 March 2019

(Currency: Indian rupees in lakhs)

An	And the state		Contr	actual cash flows		
As at 31 March, 2018	Carrying amount	Total	1 year or less	ess 1-2 years	2-5 years	More than
Non-derivative financial liabili	ties		-			5 years
Commercial paper issued to mutual funds	59,694.74	60,500.00	60,500.00	A	-	
Trade and other payables Bank overdraft	3,025.61	3,025.61	3,025.61	4		
			100 C 100 C	-	+	
Other financial liabilities	12,656.66	12,656.66	7,725.41		4,931,25	
Secured loan from Banks		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			1,001.20	
Non convertible debentures	1,82,310.91	2,12,497.61	2,361.90	1.20.357.29	89,778,42	
	2,57,687.92	2,88,679.88	73,612.92	1,20,357.29	94,709.67	
Derivative financial liabilities						
Derivatives liabilities at fair value	6,682.00	6,682.00	6,682.00			
	6,682.00	6.682.00	6,682.00	11		

The inflows/(outflows) disclosed in the above table represent the contractual undiscounted cash flows relating to derivative financial liabilities held for risk management purposes and which are not usually closed out before contractual maturity. The disclosure shows net cash flow amounts for derivatives that are net cash-settled and gross cash inflow and outflow amounts for derivatives that have simultaneous gross cash settlement.

#### C. Market risk

Market risk is the risk that changes in market prices - such as foreign exchange rates, interest rates and equity prices - will affect the Company's income or the value of its holdings of financial instruments. The Company is domiciled in India and has its revenues and other transactions in its functional currency i.e. INR. Accordingly the Company is not exposed to any currency risk.

#### (ii) Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Company's exposure to market risk for changes in interest rates relates to fixed deposits and borrowings from financial institutions

#### Exposure to interest rate risk

The interest rate profile of the Company's interest-bearing financial instruments is as follows:

Particulars	March 31, 2019	March 31, 2018	
Zero-rate instruments		March 51, 2010	
Financial assets	3,63,988,20	3,32,501.61	
Financial liabilities			
	1,39,975.81	22,364.27	
Fixed-rate instruments			
Financial assets	1.01.982.92	80,500.42	
Financial liabilities			
	1,67,145.30	2,42,005.65	
Variable-rate instruments			
Financial assets			
Financial liabilities		-	
r indiricial habilities		-	

#### Interest rate sensitivity - fixed rate instruments

The Company does not account for any fixed-rate financial assets or financial liabilities at fair value through profit or loss, and the Company does not have any designate derivatives (interest rate swaps) except for the following:

Financial liability - 0.01% Compulsorily convertible debentures subscribed by Tata Sons Private Limited which are carried at fair value through profit or loss
 Financial asset - 0.01% Compulsorily convertible debentures invested in TRIL Infopark Limited which are carried at fair value through profit or loss

Since both the instruments are compulsorily convertible in nature, there is no redemption value. Further sensitivity pertaining to risk free rate will not have any impact on fair values due to monte carlo simulation techniques used. Refer Note 42 for valuation techniques used to determine fair value. Therefore, a change in interest rates at the reporting date would not affect profit or loss for any of these fixed interest bearing financial instruments accounted at fair value through profit or loss

#### Interest rate sensitivity - variable rate instruments

The Company is having only fixed rate borrowings and fixed rate bank deposits which are carried at amortised cost. They are therefore not subject to interest rate risk as defined in IND AS 107, since neither the carrying amount nor the future cash flow will fluctuate because of a change in market interest rates.

#### 45 Capital Management

The Company's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. It sets the amount of capital required on the basis of annual business and long-term operating plans which include capital and other strategic investments.

The funding requirements are met through a mixture of equity, convertible debt securities, and other borrowings. The Group's policy is to use short-term and long-term borrowings to meet anticipated funding requirements

The Company monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total liabilities, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity other than amounts accumulated in the hedging reserve

#### The Company's adjusted net debt to equity ratio is as follows:

Particulars	March 31, 2019	March 31, 2018
Total liabilities (comprising of interest bearing borrowings and interest accrued thereon)	1,86,302.62	2.54.662.31
Less : Cash and cash equivalent	890.69	1,318.38
Adjusted net debt	1.85,411,93	2.53.343.93
Adjusted equity	1,98,685,95	1.89,746.17
Adjusted net debt to adjusted equity ratio		
	0.93	1.34





### 46 Events after the balance sheet date

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

#### 47 Previous Year's Figures

Previous year figures have been regrouped / reclassified to conform to current year presentation, wherever considered necessary.

Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 N'hiberla tt Banmal Agrawala Sanjay Dutt F. N. Subedar Chairmad DIN - 00120029 Managing Director Director Saphoenue 3 DIN - 05251670 DIN - 00028428 Basin

For and on behalf of the Board of Directors of

ASKINS \$ 4 SEL EL OIT Chartered Accountants + an é 9

Rajiv Sabharwal Director DIN - 00057333

S. Santhanakrishnan Director DIN - 00032049

Neera Saggi Director DIN - 00501029

anno

Sanjay Sharma Chief Financial Officer

Juna

Vinay Gaoka

Membership No: A6120 Mumbai Dated : 8 May, 2019

Company Secretary

246

Chartered Accountants Indiabulls Finance Centre Tower 3, 27<sup>th</sup> – 32<sup>nd</sup> Floor Senapati Bapat Marg Elphinstone Road (West) Mumbai – 400 013 Maharashtra, India Tel: +91 22 6185 4000 Fax: +91 22 6185 4001

#### INDEPENDENT AUDITOR'S REPORT

#### To The Members of Tata Realty and Infrastructure Limited Report on the Audit of the Standalone Financial Statements

#### Opinion

We have audited the accompanying standalone financial statements of **Tata Realty and Infrastructure Limited** ("the Company"), which comprise the Balance Sheet as at 31 March 2020, and the Statement of Profit and Loss (including Other Comprehensive Income), the Statement of Cash Flows and the Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid standalone financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended, ("Ind AS") and other accounting principles generally accepted in India, of the state of affairs of the Company as at 31 March 2020, and its loss, total comprehensive loss, its cash flows and the changes in equity for the year ended on that date.

#### **Basis for Opinion**

We conducted our audit of the standalone financial statements in accordance with the Standards on Auditing specified under section 143(10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Standalone Financial Statements section of our report. We are independent of the Company in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the standalone financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the standalone financial statements.

#### **Key Audit Matters**

Key audit matters are those matters that, in our professional judgment, were of most significance in our audit of the standalone financial statements of the current period. These matters were addressed in the context of our audit of the standalone financial statements as a whole, and in forming our opinion thereon, and we do not provide a separate opinion on these matters. We have determined the matters described below to be the key audit matters to be communicated in our report.

RECO

r. No.	Key Audit Matter	Auditor's Response
	The determination of the fair value	We assessed the Company's process fo
	of investments requires significant judgement, due to various intricate	the valuation of non-curren
	assumptions / estimates such as	investments carried at Fair Value.
	market rent levels, toll revenues,	Our audit approach consisted testing o
	expenditure to be incurred, vacancy	the design and operating effectiveness
	factors, prevailing market yields	of the internal controls and substantive
	and market transactions, cash flows	testing as follows:
	as well as impact due to COVID 19.	<ul> <li>Evaluated the design of the interna</li> </ul>
	as well as impact due to COVID 19.	controls relating to the valuation o
	As at 31 March 2020, the Company	non-current investments at Fai
	had investments in various	Value.
	subsidiaries and joint ventures	<ul> <li>Tested the operating effectiveness</li> </ul>
	which have been accounted for at	of controls for the review o
	fair value amounting to Rs.	assumptions and estimates used in
	4231,81.34 lakhs (Refer note 5 and	evaluation of inputs for the purpose
	2(f) to the financial statements).	of fair valuation. We carried out a
	z() to the manela statements).	combination of procedures involving
	The valuation of unquoted	enquiry and observation, and
	investments is considered to be a	inspection of evidence in respect of
	key audit matter as this amount	operation of these controls.
	represents a very significant portion	
	of the total assets of the Company	Principal audit procedures performed:
	included in the standalone financial	
	statements, combined with the	<ul> <li>Assessed the management's maker</li> </ul>
	competence of management's	/ checker controls over preparation
	expert and the level of judgement	of the discounted cash flow mode
	exercised for determining the fair	for the valuation of investments and
	values.	controls over management's
		analysis of the variances in values
		in comparison with previous year.
		<ul> <li>Ascertained whether the fair value</li> </ul>
		of investments has been
		determined by external,
		independent valuer, having
		appropriate recognised professional
		qualifications and recent experience
		in the location and category of the
		property underlying the
		investments being valued based on information and explanations
		information and explanations provided by the management. We
		그는 그는 것 같아. 안 이렇는 것 같아. 것 같아. 그는 것 같아. 나는 것 같아. ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ?
		assessed their competence, independence and integrity.
		<ul> <li>The audit team included Fair Value</li> </ul>
		Specialists for reviewing the
		assumptions of WACC,
		capitalisation rate and market rent
		levels, attended meetings with the
		management team and the
		valuation experts appointed by the
		Company's management to understand the methodology
		applied, the main assumptions underlying their valuations and
		more particularly, amongst other
		more particularly, amongst other

3 pt

	<ul> <li>inputs, WACC, capitalisation rate, market rental levels, toll road traffic growth/decline, vacancy factor, impact due to COVID 19.</li> <li>Documented the understanding from the meetings for the basis on which WACC rate were determined</li> <li>Checked the arithmetic accuracy of the cash flow models.</li> <li>Performed procedures to reconcile the valuations concluded by the management expert and the recognition done by the management in the financial statements.</li> <li>Performed analytical procedures by comparing assumptions and fair values on a year-on-year basis and obtained reasons for the variations.</li> </ul>
--	---

### Information Other than the Financial Statements and Auditor's Report Thereon

- The Company's Board of Directors is responsible for the other information. The other information comprises the information included in the Board of Directors report, but does not include the standalone financial statements and our auditor's report thereon.
- Our opinion on the standalone financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.
- In connection with our audit of the standalone financial statements, our responsibility is to read the other information and, in doing so, consider whether the other information is materially inconsistent with the standalone financial statements or our knowledge obtained during the course of our audit or otherwise appears to be materially misstated.
- If, based on the work we have performed, we conclude that there is a material misstatement of this other information, we are required to report that fact. We have nothing to report in this regard.

### Management's Responsibility for the Standalone Financial Statements

The Company's Board of Directors is responsible for the matters stated in Section 134(5) of the Act with respect to the preparation of these standalone financial statements that give a true and fair view of the financial position, financial performance including other comprehensive loss, cash flows and changes in equity of the Company in accordance with the Ind AS and other accounting principles generally accepted in India. This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the standalone financial statements that give a true and fair view and are free from material misstatement, whether due to fraud or error.

PRH

In preparing the standalone financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those Board of Directors are also responsible for overseeing the Company's financial reporting process.

#### Auditor's Responsibility for the Audit of the Standalone Financial Statements

Our objectives are to obtain reasonable assurance about whether the standalone financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these standalone financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

- Identify and assess the risks of material misstatement of the standalone financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.
- Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under section 143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Company has adequate internal financial controls system in place and the operating effectiveness of such controls.
- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the standalone financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.
- Evaluate the overall presentation, structure and content of the standalone financial statements, including the disclosures, and whether the standalone financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Materiality is the magnitude of misstatements in the standalone financial statements that, individually or in aggregate, makes it probable that the economic decisions of a

REKU

reasonably knowledgeable user of the standalone financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the standalone financial statements.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

From the matters communicated with those charged with governance, we determine those matters that were of most significance in the audit of the standalone financial statements of the current period and are therefore the key audit matters. We describe these matters in our auditor's report unless law or regulation precludes public disclosure about the matter or when, in extremely rare circumstances, we determine that a matter should not be communicated in our report because the adverse consequences of doing so would reasonably be expected to outweigh the public interest benefits of such communication.

### Report on Other Legal and Regulatory Requirements

- 1. As required by Section 143(3) of the Act, we report that:
  - a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit.
  - b) In our opinion, proper books of account as required by law have been kept by the Company so far as it appears from our examination of those books.
  - c) The Balance Sheet, the Statement of Profit and Loss including Other Comprehensive Loss, the Statement of Cash Flows and Statement of Changes in Equity dealt with by this Report are in agreement with the books of account.
  - d) In our opinion, the aforesaid standalone financial statements comply with the Ind AS specified under Section 133 of the Act.
  - e) On the basis of the written representations received from the directors as on 31 March 2020 taken on record by the Board of Directors, none of the directors is disqualified as on 31 March 2020 from being appointed as a director in terms of Section 164(2) of the Act.
  - f) With respect to the adequacy of the internal financial controls over financial reporting of the Company and the operating effectiveness of such controls, refer to our separate Report in "Annexure A". Our report expresses an unmodified opinion on the adequacy and operating effectiveness of the Company's internal financial controls over financial reporting.
  - g) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of section 197(16) of the Act, as amended, in

REKE

our opinion and to the best of our information and according to the explanations given to us, the remuneration paid by the Company to its directors during the year is in accordance with the provisions of section 197 of the Act.

- h) With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended, in our opinion and to the best of our information and according to the explanations given to us:
  - i. The Company has disclosed the impact of pending litigations on its financial position in its standalone financial statements;
  - ii. The Company has made provision, as required under the applicable law or accounting standards, for material foreseeable losses, if any, on long-term contracts including derivative contracts;
  - iii. There were no amounts which were required to be transferred to the Investor Education and Protection Fund by the Company.
- 2. As required by the Companies (Auditor's Report) Order, 2016 ("the Order") issued by the Central Government in terms of Section 143(11) of the Act, we give in "Annexure B" a statement on the matters specified in paragraphs 3 and 4 of the Order.

For **Deloitte Haskins & Sells LLP** Chartered Accountants (Firm's Registration No. 117366W/W-100018)

11

Rajesh K. Hiranandani (Partner) (Membership No. 36920)

UDIN: 20036920AAAACA1696

Place: Mumbai Date: 6 July 2020
## ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT (Referred to in paragraph 1(g) under 'Report on Other Legal and Regulatory Requirements' section of our report of even date)

## Report on the Internal Financial Controls Over Financial Reporting under Clause (i) of Sub-section 3 of Section 143 of the Companies Act, 2013 ("the Act")

We have audited the internal financial controls over financial reporting of Tata Realty and Infrastructure Limited ("the Company") as of 31 March 2020 in conjunction with our audit of the standalone Ind AS financial statements of the Company for the year ended on that date.

## Management's Responsibility for Internal Financial Controls

The Company's management is responsible for establishing and maintaining internal financial controls based on the internal control over financial reporting criteria established by the Company considering the essential components of internal control stated in the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting issued by the Institute of Chartered Accountants of India. These responsibilities include the design, implementation and maintenance of adequate internal financial controls that were operating effectively for ensuring the orderly and efficient conduct of its business, including adherence to company's policies, the safeguarding of its assets, the prevention and detection of frauds and errors, the accuracy and completeness of the accounting records, and the timely preparation of reliable financial information, as required under the Companies Act, 2013.

### Auditor's Responsibility

Our responsibility is to express an opinion on the Company's internal financial controls over financial reporting of the Company based on our audit. We conducted our audit in accordance with the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting (the "Guidance Note") issued by the Institute of Chartered Accountants of India and the Standards on Auditing prescribed under Section 143(10) of the Companies Act, 2013, to the extent applicable to an audit of internal financial controls. Those Standards and the Guidance Note require that we comply with ethical requirements and plan and perform the audit to obtain reasonable assurance about whether adequate internal financial controls over financial reporting was established and maintained and if such controls operated effectively in all material respects.

Our audit involves performing procedures to obtain audit evidence about the adequacy of the internal financial controls system over financial reporting and their operating effectiveness. Our audit of internal financial controls over financial reporting included obtaining an understanding of internal financial controls over financial reporting, assessing the risk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. The procedures selected depend on the auditor's judgement, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error.

## Meaning of Internal Financial Controls Over Financial Reporting

A company's internal financial control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal financial control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of

RECO

records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorisations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorised acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

### Inherent Limitations of Internal Financial Controls Over Financial Reporting

Because of the inherent limitations of internal financial controls over financial reporting, including the possibility of collusion or improper management override of controls, material misstatements due to error or fraud may occur and not be detected. Also, projections of any evaluation of the internal financial controls over financial reporting to future periods are subject to the risk that the internal financial control over financial reporting may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

## Opinion

In our opinion, to the best of our information and according to the explanations given to us, the Company has, in all material respects, an adequate internal financial controls system over financial reporting and such internal financial controls over financial reporting were operating effectively as at 31 March 2020, based on the criteria for internal financial control over financial reporting established by the Company considering the essential components of internal control stated in the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting issued by the Institute of Chartered Accountants of India.

> For **Deloitte Haskins & Sells LLP** Chartered Accountants (Firm's Registration No. 117366W/W-100018)

eru

Rajesh K. Hiranandani (Partner) (Membership No. 36920)

UDIN: 20036920AAAACA1696

Place: Mumbai Date: 6 July 2020

8 PKH

## ANNEXURE B TO THE INDEPENDENT AUDITOR'S REPORT

(Referred to in paragraph 2 under 'Report on Other Legal and Regulatory Requirements' Section of our report of even date)

- (i) (a) The Company has maintained proper records showing full particulars, including quantitative details and situation of property, plant and equipment.
  - (b) The Company performs physical verification of its property, plant and equipment annually. In our opinion, the periodicity of physical verification is reasonable having regard to the size of the Company and the nature of its assets. According to the information and explanation given to us, no material discrepancies were noticed on such verification.
  - (c) According to the information and explanations given to us and on the basis of our examination of the records of the Company, the title deeds of immovable properties, as disclosed in Note 4 to the standalone financial statements, are held in the name of the Company as at the balance sheet date.
- (ii) As explained to us, the inventories were physically verified during the year by the Management at reasonable intervals and no material discrepancies were noticed on physical verification between physical stock and the books of accounts.
- (iii) According to the information and explanations given to us, the Company has granted unsecured loans to eleven companies covered in the register maintained under Section 189 of the Companies Act, 2013, in respect of which:
  - (a) The terms and conditions of the grant of such loans are, in our opinion, prima facie, not prejudicial to the Company's interest.
  - (b) According to the information and explanations given to us, in respect of four unsecured loans, interest along with principal is repayable on demand and seven unsecured loans are interest free and the principal is repayable on demand. The Company has not demanded any loan during the year.
  - (c) There is no amount overdue for more than 90 days as at 30 March 2020.
- (iv) In our opinion and according to the information and explanations given to us, the Company has complied with the provisions of Sections 185 and 186 of the Companies Act, 2013 in respect of grant of loans, making investments and providing guarantees and securities, as applicable.
- (v) According to the information and explanations given to us, the Company has not accepted any deposit during the year as provided under Section 73 to 76 or any other relevant provisions of the Companies Act 2013. There are no unclaimed deposits any time during the year.

& PKB

- (vi) We have broadly reviewed the cost records maintained by the Company pursuant to the Companies (Cost Records and Audit) Rules, 2014, as amended prescribed by the Central Government under sub-section (1) of Section 148 of the Companies Act, 2013, and are of the opinion that, prima facie, the prescribed cost records have been made and maintained. We have, however, not made a detailed examination of the cost records with a view to determine whether they are accurate or complete.
- (vii)According to the information and explanations given to us, in respect of statutory dues:
  - (a) The Company has generally been regular in depositing undisputed statutory dues, including Income-tax, professional tax, provident fund, work contracts tax, labour cess, Goods & Services Tax, cess and other material statutory dues applicable to it to the appropriate authorities.
  - (b) There were no undisputed amounts payable in respect of Income-tax, professional tax, provident fund, work contracts tax, labour cess, Goods & Services Tax, cess and other material statutory dues in arrears as at 31 March 2020 for a period of more than six months from the date they became payable.

Name of the Statute	Nature of Dues	Forum where dispute is pending	Period to which the amount relates	Amount involved	Amount Unpaid
Income Tax Act, 1961	Income Tax	Income Tax Appellate Tribunal - Mumbai	FY 2014-2015	4,43,65,426	4,43,65,426
Finance Act, 1994	Service Tax	Commissioner of CGST & Central Excise – Mumbai	FY 2010-2011	2,67,21,775	2,67,21,775
Finance Act, 1994	Service Tax	Commissioner of CGST & Central Excise – Nagpur	FY 2010-11, FY 2011-12, FY 2012-13	88,37,820	79,54,038
Finance Act, 1994	Service Tax	Commissioner of CGST & Central Excise - Kochi	FY 2010-11, FY 2011-12, FY 2012-13	1,82,07,459	1,63,86,713
Finance Act, 1994	Kerala VAT	Commissioner of CGST & Central Excise - Kochi	FY 2014-2015	2,00,10,000	2,00,10,000

(c) Details of dues of Income-tax and Service Tax, which have not been deposited as on 31 March 2020 on account of disputes are given below:

(viii) In our opinion and according to the information and explanations given to us, the Company has not defaulted in the repayment of loans or borrowings from financial institutions and dues to debenture holders during the year. The Company did not have any outstanding dues to banks and government.

FY 2014-2015

& Central Excise - Kochi

PKG

- (ix) The Company has not raised moneys by way of initial public offer or further public offer (including debt instruments) or term loans and hence reporting under clause (ix) of the CARO 2016 Order is not applicable.
- (x) To the best of our knowledge and according to the information and explanations given to us, no fraud by the Company and no material fraud on the Company by its officers or employees has been noticed or reported during the year.
- (xi) In our opinion and according to the information and explanations given to us, the Company has paid / provided managerial remuneration in accordance with the requisite approvals mandated by the provisions of Section 197 read with Schedule V to the Companies Act, 2013.
- (xii) The Company is not a Nidhi Company and hence reporting under clause (xii) of the CARO 2016 Order is not applicable.
- (xiii) In our opinion and according to the information and explanations given to us the Company is in compliance with Section 177 and 188 of the Companies Act, 2013, where applicable, for all transactions with the related parties and the details of related party transactions have been disclosed in the standalone financial statements etc. as required by the applicable accounting standards.
- (xiv) According to the information and explanation given to us, during the year the Company has not made any preferential allotment or private placement of shares or fully or partly convertible debentures and hence reporting under clause (xiv) of CARO 2016 is not applicable to the Company.
- (xv) In our opinion and according to the information and explanations given to us, during the year the Company has not entered into any non-cash transactions with its directors or directors of its holding, subsidiary or associate company or persons connected with them and hence provisions of Section 192 of the Companies Act, 2013 are not applicable.
- (xvi) The Company is not required to be registered under Section 45-IA of the Reserve Bank of India Act, 1934.

#### For Deloitte Haskins & Sells LLP

Chartered Accountants (Firm's Registration No. 117366W/W-100018)

11 er

Rajesh K. Hiranandani Partner (Membership No. 36920) UDIN: 20036920AAAACA1696

Place: Mumbai Date: 6 July 2020

& PCG

## Tata Realty and Infrastructure Limited Balance Sheet as at 31 March 2020

(Currency: Indian rupees in lakhs)

	culars	Note No.	As at 31 March 2020	As at 31 March 2019
ASSE				- a at or march 2019
	-CURRENT ASSETS			
(a	<ul> <li>Property, plant and equipment (PPE)</li> </ul>	4		
(b	) Intangible assets	4	1,594.84	1,743.3
(C	) Right to use an asset		77.70	50.4
(d		4	146.38	
(e	) Financial assets	4	-	3.7
	(i) Investments			
	(ii) Loans and advances	5	4,30,771.34	4,06,035.7
	(iii) Others	7	39,709.19	52,124.4
(f)	Current tax assets (net)	8	0.75	0.7
(g)		9	6,934.68	6,012.90
101	TOTAL NON-CURRENT ASSETS	10	12,779.98	10,968.46
	A COLORED AND A COLOR		4,92,014.86	4,76,939.93
CURF	RENT ASSETS			
	Inventories			
	Financial assets	11	30,957.33	41,011.06
(~)	(i) Investments			
	(ii) Trade and other receivables	5	71,179.33	311.42
	(iii) Cash and cash equivalents	6	1,878.42	1,397.81
	(iv) Other bank balances	12	25,580.30	445.25
	(v) Loans and advances	13	419.25	444.69
	(vi) Others financial assets	7	2,464.51	2,349.77
(c)	Other current assets	8	1,428.91	2,861.19
10)	TOTAL CURRENT ASSETS	10	431.31	1,086.29
	TOTAL CORRENT ASSETS	-	1,34,339.36	49,907.48
	TOTAL ASSETS		6,26,354.22	5,26,847.41
	AND LIABILITIES	-		
EQUIT				
	Equity share capital	14	1 01 700 77	a subserver.
(b)	Other equity	15	1,01,730.77	1,01,730.77
	TOTAL EQUITY	19 -	1,91,695.85 2,93,426.62	96,955.18 1,98,685.95
IABIL	TIES			1,00,000.00
	-CURRENT LIABILITIES			
(a)	Financial liabilities			
(4)	(i) Long-term borrowings			
	(ii) Other financial liabilities	16	1,39,419.04	72,460,15
(b)	Long-term provisions	18	6,071.09	11,425.58
(c)	Current tax liabilities (net)	19	626.53	685.48
(d)	Deferred tax liabilities (net)	20	1,751.88	1,751.88
	TOTAL NON-CURRENT LIABILITIES	21	19,168.35	13,258.38
	The men connent EIABIETTES		1,67,036.90	99,581.47
CUR	RENT LIABILITIES			
	Financial liabilities			
	(i) Short-term borrowings			
	(ii) Trade and other payables from Micro and Small Enterprises	16	95,303.21	94,685.15
	(iii) Trade and other payables other than MSME	17		-
	iv) Other financial liabilities	17	3,185.65	3,458.03
	Other current liabilities	18	64,132.53	1,25,092.20
(c) s	Short term provisions	22	3,084.76	5,119.48
1	TOTAL CURRENT LIABILITIES	19	184.56	225.13
			1,65,890.71	2,28,579.99
7	OTAL EQUITY AND LIABILITIES		6,26,354.22	5,26,847,41
nifica	nt accounting policies			0,20,047,41
tes to	the standalone Ind AS financial statements	1-3		
nificat tes to	nt accounting policies the standalone Ind AS financial statements	1-3		0,20,0

Notes to the standalone Ind AS financial statements 4-48 The accompanying notes 1 to 48 form an integral part of these standalone Ind AS financial statements.

As per our report of even date attached For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No. 117366W / W-100018)

Rajesh K. Hiranandani Partner

Mumbai Dated : 06 July 2020

J.M. huba 1-8

F. N. Subedar Director DIN - 00028428

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

Aav in Banmali Agrawala

Chairman DIN - 00120029

Ne 0

Sanjay Sharma Chief Financial Officer

utt

Sanjay Dutt Managing Director DIN - 05251670

Sudhakar Shetty Company Secretary Membership No: A13200

Mumbai Dated : 06 July 2020

## Tata Realty and Infrastructure Limited

Statement of Profit and Loss for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

For the year For the year Particulars Note ended ended No. 31 March 2020 31 March 2019 Revenue from operations 1 23 17,584.54 14,493.29 I Other income 24 14,966.08 12,753,95 ш Total Income (I +II) 32,550.62 27,247,24 IV Expenses Cost of sale of flats 25 12,118.10 6.122.91 Employee benefits expense 26 5,469.96 5.585.39 Finance costs 27 28,496,17 22,988.91 Depreciation and amortization expense 28 191.80 140.30 Loss on fair valuation of derivative contracts 29 440.00 Other expenses 30 6.441.84 3,883.25 Amounts written off during the year 31 5,414.38 **Total Expenses** 52.717.87 44,575.14 v (Loss) before tax (III-IV) (20,167.25) (17,327.90) VI Tax expenses 32 Current Tax Deferred Tax charge (2.417.51) (1,298.87) Total tax expenses (2,417.51)(1,298.87) VII (Loss) for the year (V-VI) (22,584.76) (18,626.77) VIII Other Comprehensive Income: A. Items that will not be reclassified to profit or loss Remeasurements of defined benefit (asset) / liability (17.31) 52.65 Equity instruments fair valued through OCI 835.21 32,334.60 Income tax relating to items that will not be reclassified to profit or loss 21 (3,492.47) (3,516.56) B. Items that will not be reclassified to profit or loss (2,674.57) 28,870.69 IX Total Comprehensive Income / (Loss) for the Year (VII+VIII) (25,259.33) 10,243.92 x Earnings per equity share (Face value of INR 10 each) 34 Basic (2.22) (1.83)Diluted (2.22) (1.83)\* Diluted EPS for the year ended 31 March 2020 is anti-dilutive hence not considered Significant accounting policies 1-3 Notes to the standalone Ind AS financial statements

The accompanying notes 1 to 48 form an integral part of these standalone Ind AS financial statements.

As per our report of even date attached For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No. 117366W / W-100018)

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

Banmali Agrav Chairman DIN - 00120029

0

Sanjay Sharma Chief Financial Officer

Sanjay Dutt Managing Director DIN - 05251670

Sudhakar Shetty Company Secretary Membership No: A13200

Mumbai Dated : 06 July 2020

Rajesh K. Hiranandani Partner

Mumbai Dated : 06 July 2020

F. N hubertar

F. N. Subedar

DIN - 00028428

Director

(Currency: Indian rupees in lakhs) Particulars A CABLELOW EDOM ODERATING ACTIVITIES		For the year ended		For the year	
A CASH FLOW FROM OPERATING ACTIVITIES		31 Marci	1 2020	31 March	2019
Loss before tax			(20,167.25)		(17,327.90
Adjustments for :			(		(11,021.00
Depreciation and amortisation expense		191.80		140.30	
(Gain) / Loss on sale of PPE		(0.33)		23.17	
(Gain) on sale of current investments		(818.52)		(263,43)	
(Gain) on fair valuation of investments and derivative instruments		(6,737.53)		(5,685.50)	
Interest Income		(6,779.71)		(5,537.52)	
Unwinding of call option premium		(574.09)		(540.44)	
Finance costs		28,496.17		22,988,91	
Amounts written off during the year				5,414.38	
Provision for credit impaired Trade Receivables		306.00		0,11,100	
Provision for impairment of inter corporate deposits		516.62			
Provision for employee benefits		(116.84)	14,483.57	14,15	16,554.02
Operating (loss) before working capital changes		(110.04)	(5,683.68)	14,13	(773.88
Changes in working capital					
(Increase) / Decrease in trade receivables		(700.04)		1000	
Decrease in inventories		(786.61)		430.97	
Decrease in advances, other current assets and other non-current assets		10,053.73		730.47	
(Decrease) / Increase in trade payables, other financial liabilities and other		1,216.81		1,418.20	
financial liabilities		(2,418.72)		1,530.62	
A L A			8,065.21		4,110.26
Cash flows generated from operating activities			2,381.53		3,336.38
Tax refund / (paid) during the year (net)	1.1.1.4		(921.78)		(642.09
Net cash flows generated from operating activities	A		1,459.75		2,694.29
CASH FLOW FROM INVESTMENT ACTIVITIES					
Payment for purchase of Property, plant & equipment and intangible asset		(219.48)		(538,52)	
Proceeds on sale of Property, plant & equipment		6.69		32.51	
Investment in Fixed deposits under lien with maturity less					
than 12 months (net)		25.44		(83.02)	
Investment in subsidiaries and joint venture companies		(18,985.50)		(14,072.95)	
Proceeds from sale of investments in subsidiary companies		9,177.38			
Investment in mutual fund		(4,46,529,96)		(1,45,427.98)	
Proceeds from sale of investments in mutual fund		3,76,915.78		1,52,366.17	
Inter-corporate deposit refunded		48,799,48		7,490.33	
Inter-corporate deposit given		(43,086.52)		(24,717.17)	
Interest Received		4,692.75		3,115.67	
Net cash flows used in Investing Activities	в		(69,203.94)		(21,834.96)
CASH FLOW FROM FINANCING ACTIVITIES					
Rights Issue Application Money Received		1,20,000.00			
Proceeds from long-term borrowings		1,07,000.00			
Repayment of long-term borrowings		(1,10,000.00)			
Proceeds from short-term borrowings		16,356.61		35.245.66	
Repayment of ICD taken from related parties		(14,000.00)		30,240.00	
Finance costs paid		(26,477.37)		(10 CHE 70)	
Net cash flows from Financing Activities	с -	(20,477.37)	92,879.24	(16,615.70)	18,629,96
Net increase/ (decrease) in cash and cash equivalents	(A+B+C)		25,135.05		1510 74
	(41010)_				(510.71)
Cash and cash equivalents at the beginning of the year			445.25		955.96
Cash and cash equivalents at the end of year		_	25,580.30		445.25
Cash and bank balances at the end of the year comprise of: Cash on Hand			0.06		
					in ar
Balances with Bank Deposit Accounts with less than or equal to 3 months maturity			970.24 24,610.00		445.25

#### Note:

The Cash flow statement has been prepared under the indirect method as set out in Indian Accounting Standard - 7 ('Ind AS 7') on Cash Flow Statement prescribed in Companies (Indian Accounting Standard) Rules, 2015, notified under Section 133 of the Companies Act, 2013.

 Significant accounting policies
 1-3

 Notes to the standalone Ind AS financial statements
 4-48

 The accompanying notes 1 to 48 form an integral part of these standalone Ind AS financial statements.

As per our report of even date attached For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No. 117366W / W-100018)

H

Rajesh K. Hiranandani Partner

Mumbai Dated : 06 July 2020

3 N habredo

F. N. Subedar Director DIN - 00028428

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

10

Sanjay Dutt Managing Director DIN - 05251670

7 Sudhakar Shetty Company Secretary Membership No: A13200

Mumbai Dated : 06 July 2020

Sanjay Sharma

130

Banmali Agrawala Chairman DIN - 00120029

Chief Financial Officer

A

2 'on

260

# Tata Realty and Infrastructure Limited Statement of Changes in Equity (SOCIE) for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### A Equity Share Capital

	31 March	2020	31 March 2019		
Particulars	Number of Shares	Amount	Number of Shares	Amount	
Subscribed and Fully Paid up Capital					
Equity shares of INR 10 each					
Opening Balance	1,01,73,07,692	1.01.730.77	1.01.73.07.692	1.01.730.77	
Changes in equity share capital during the year			10110001002	1,01,700.77	
Closing Balance	1,01,73,07,692	1,01,730.77	1,01,73,07,692	1,01,730.77	

B Other equity

	Re	serves and surplu	IS		Items of Other comprehensive income		
Particulars	Capital reserve	Retained earnings	Securities premium reserve	Share application money pending allotment	Defined benefit plan adjustment	Equity Instruments through Other Comprehensive Income reserve	Total other equity
Balance as at 01 April 2018	4,783.49	(17,728.54)	15.769.23		(18.78)	85,210.00	88,015.40
IND AS 115 Adjustment under transition		(1,304.14)			(101.0)	05,210.00	(1,304.14
Loss for the year	4	(18,626.77)		+			(18,626.77
Other comprehensive income for the year Equity Instruments fair valued through Other Comprehensive Income	4		(*)	-		32,334.60	32,334.60
Remeasurements of defined benefit liability			-	-	52.65	12	52.65
Income tax relating on above			· · · · · · · · · · · · · · · · · · ·		(13.68)	(3,502.88)	(3,516.56
Balance as at 31 March 2019	4,783.49	(37,659.45)	15,769.23		20.19	1,14,041.72	96,955.18
Loss for the year Share application money pending allotment	÷	(22,584.76)			c	3	(22,584.76)
received during the year Other comprehensive income for the year				1,20,000.00			1,20,000.00
Equity Instruments fair valued through Other Comprehensive Income	-	-				835.21	835.21
Remeasurements of defined benefit (asset)	-	-	-		(17.31)		(17.31)
ncome tax relating on above	÷	û.	1.20		4.50	(3,496,97)	(3,492.47)
ransferred from OCI to Retained eqrnings on lerecogntion of equity instruments		(13,307.84)				13,307.84	
Balance as at 31 March 2020	4,783.49	(60,244.21)	15,769.23	1,20,000.00	7.38	1,11,379.96	1,91,695.85

Significant accounting policies Notes to the standalone Ind AS financial statements The accompanying notes 1 to 48 form an integral part of these standalone Ind AS financial statements.

As per our report of even date attached For Deloitte Haskins & Sells LLP Chartered Accountants (Firm's Registration No. 117366W / W-100018)

2

Rajesh K. Hiranandani Partner

Mumbai Dated : 06 July 2020

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

120 w D Banmali Agrawala Chairman DIN - 00120029

for

Sanjay Sharma Chief Financial Officer

4-48

1

Sanjay Dutt Managing Director DIN - 05251670

1

tt

F. N. Subedar Director DIN - 00028428

1 dubdor

deg Sudhakar Shetty

Company Secretary Membership No: A13200

Mumbai Dated : 06 July 2020

#### 1 Background of the Company

Tata Realty and Infrastructure Limited ('the Company') was incorporated on 2 March 2007 to carry on the business of investment advisory services, project management consultancy services and real estate and infrastructure development. The Company is a wholly owned subsidiary of Tata Sons Private Limited.

#### 2 Basis of preparation

#### (a) Summary of Significant Accounting Policies

These standalone Ind AS financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 ('the Act') and other relevant provisions of the Act.

#### (b) Going Concern

The Company is primarily engaged in development of commercial and Infrastructure projects to generate stable cashflows and capital appreciation over the life of the assets through investments in various project SPVs.

The Company has incurred losses amounting to Rs. 22,585 lakhs in the current year (previous year Rs. 18,627 lakhs), As at 31 March 2020 the Company has a net current liability position of Rs. 31,552 lakhs where the current liabilities at Rs. 1,65,891 lakhs exceed the current assets at Rs. 1,34,339 lakhs. Based on scheduled repayment Rs 2,29,000 lakhs is due for repayment within 12 months from the approval of these financial statements. The Company has also agreed to provide financial support of Rs. 33,800 lakhs to its subsidiary companies.

Assessment: The Board of Directors have assessed the above operational conditions and indicators and have come to the conclusion that no material uncertainty exists that may cast significant doubt on the Company's ability to continue as a going concern based on cashflow forecasts and the plan management has put in place.

Mitigating factors: In spite of difficult market conditions over a period, the superior nature of portfolio of the Company's developments has increased the Fair Value of these assets to Rs 456,000 Lakhs over its historical cost of Rs 284,400 Lakhs.

During FY 2020-21 based on projections, the SPV's are expected to generate operational net cash flow of more than Rs 450,00 Lakhs which will increase the value of investments of the Company. The management is evaluating possibility of divestment of selected assets and change in capital structure in its project SPVs' which is expected to generate more than Rs 1,50,000 Lakhs as equity value. Also, the free cashflow from sale of Ready to Move in (RTMI) residential inventory will support its operations during coming financial years.

Negative working capital is on account of management decision to borrow short-term funds through commercial papers to take advantage of interest arbitrage. However, management has modified the strategy to replace, to the extent possible, short term funding with long term funding arrangement going forward.

The equity capital from the parent i.e. Tata Sons Private Limited, of an amount of Rs. 1,20,000 Lakhs received during the year has improved the company's net worth allowing the company further ability for additional borrowing in future and is reflected in the ratings of the Company

Conclusion: The Board of Directors based on cash flow forecasts and management plans have concluded on ability of the Company to continue as going concern and the financial statements have been prepared on that basis.

#### (c) Functional and presentation currency

These financial statements are presented in Indian Rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian Rupees (INR) and all amounts have been rounded-off to the nearest lakhs with two decimals, except for share data and as otherwise stated. Due to rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.



#### (d) Basis of measurement

The standalone Ind AS financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

- 1 Certain financial assets and liabilities (including derivative instruments)
- 2 Defined benefit plans plan assets measured at fair value

#### (e) Critical accounting judgements and key sources of estimation of uncertainty

In preparing these standalone Ind AS financial statements, management has made judgements, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates.

Estimates and underlying assumptions are reviewed on an ongoing basis. Revisions to accounting estimates are recognised prospectively,

#### (i) Judgements

Information about judgements made in applying accounting policies that have the most significant effects on the amounts recognised in the standalone Ind AS financial statements is included in the following notes:

Note 41 - measurement of defined benefit obligations: key actuarial assumptions;

Note 42 - determining the fair value of investments on the basis of significant unobservable inputs.

#### (ii) Assumptions and estimation uncertainties

Information about assumptions and estimation uncertainties that have a significant risk of resulting in a material adjustment in the year ended 31 March 2020is included in the following notes:

#### Note 41 - measurement of defined benefit obligations: key actuarial assumptions;

Note 42 - determining the fair value of investments on the basis of significant unobservable inputs.

#### (f) Measurement of fair values

The Company's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Company has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the CFO.

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorized into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

Level 1: quoted prices (unadjusted) in active markets for identical assets or liabilities.

- Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorized in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement.

#### (g) Foreign currency transactions

Foreign currency transactions are translated into the functional currency using the exchange rates at the dates of the transactions. Foreign exchange gains and losses resulting from the settlement of such transactions and from the translation of monetary assets and liabilities denominated in foreign currencies at year end exchange rates are generally recognised in profit or loss.

Non-monetary items that are measured at fair value in a foreign currency are translated using the exchange rates at the date when the fair value was determined. Translation differences on assets and liabilities carried at fair value are reported as part of the fair value gain or loss. The gain or loss arising on translation of non-monetary items measured at fair value is treated in line with the recognition of the gain or loss on the change in fair value of the item (i.e., translation differences on items whose fair value gain or loss is recognised in OCI or profit or loss are also recognised in OCI or profit or loss, respectively).

#### 3 Significant accounting policies

#### 3.01 Revenue recognition

Revenue is recognised to the extent that it is probable that the economic benefits will flow to the Company and the revenue can be reliably measured.

#### Sale of completed property.

Revenue from sale of completed property (residential and commercial) is recognised when:

1. The Company has transferred to the buyer significant risk and rewards of ownership of the completed property:

2. The Company retains neither continuing managerial involvement to the degree usually associated with the ownership nor effective control over the completed property sold;

- 3. The amount of revenue can be measured reliably;
- 4. It is probable that the economic benefit associated with the transaction will flow to the Company; and
- 5. Cost incurred or to be incurred in respect of the transaction can be measured reliably.

Asset management fees and Project management consultancy fees are recognized in accordance with terms of agreement with customers.

A dividend is recognised as revenue when the right to receive payment has been established. For all debt instruments measured either at amortised cost or at fair value through other comprehensive income, interest income is recorded using the effective interest rate (EIR). EIR is the rate that discounts the estimated future cash payments or receipts over the expected life of the financial instrument or a shorter period, where appropriate, to the gross carrying amount of the financial asset or to the amortised cost of a financial liability.

#### 3.02 Property, plant and equipment

#### (i) Recognition and measurement

Freehold land is carried at historical cost. All other items of property, plant and equipment are measured at cost less accumulated depreciation and any accumulated impairment losses.

The cost of an item of property, plant and equipment comprises:

- a) its purchase price, including import duties and non-refundable purchase taxes, after deducting trade discounts and rebates.
- b) any costs directly attributable to bringing the asset to the location and condition necessary for it to be capable of operating in the manner intended by management.
- c) the initial estimate of the costs of dismantling and removing the item and restoring the site on which it is located, the obligation for which an entity incurs either when the item is acquired

Income and expenses related to the incidental operations, not necessary to bring the item to the location and condition necessary for it to be capable of operating in the manner intended by management, are recognised in profit or loss.

If significant parts of an item of property, plant and equipment have different useful lives, then they are accounted for as separate items (major components) of property, plant and equipment. Any gain or loss on disposal of an item of property, plant and equipment is recognised in profit or loss.

#### (ii) Subsequent expenditure

Subsequent expenditure is capitalised only if it is probable that the future economic benefits associated with the expenditure will flow to the Company.

#### (iii) Depreciation

Depreciable amount for assets is the cost of an asset, or other amount substituted for cost, less its estimated residual value.

Depreciation is provided using the straight line method in the manner and at the rates prescribed by Part 'C' of Schedule II of the Act. Depreciation is charged on a monthly pro-rate basis for assets purchased or sold during the year.

In the following cases, the useful life is less than the corresponding useful life prescribed in Part 'C' of Schedule II of the Act, based on internal technical evaluation, taking into account the nature of the assets, the estimated usage of the asset, the operating conditions of the asset, past history of replacement, anticipated technological changes etc.:

Data Processing Networks - Servers and Networks	5 years	
Motor Car	5 years	

Leasehold improvements are amortised over the primary period of the lease.

Depreciation methods, useful lives and residual values are reviewed at each reporting date and adjusted if appropriate.

Gains and losses on disposals are determined by comparing proceeds with carrying amount. These are included in profit or loss within other gains/(losses).

Expenditure incurred on acquisition /construction of property, plant and equipment which are not ready for their intended use at balance sheet date are disclosed under capital work-in-progress. Capital work in progress is stated at cost less impairment losses, if any. Cost comprises of expenditures incurred in respect of capital projects under development and includes any attributable / allocable cost and other incidental expenses. Revenues earned, if any, before capitalization from such capital project are adjusted against the capital work in progress.

Borrowing costs relating to acquisition / construction / development of tangible assets, intangible assets and capital work in progress which takes substantial period of time to get ready for its intended use are also included to the extent they relate to the period till such assets are ready to be put to use.



#### 3.03 Intangible assets

Intangible assets comprise application software purchased / developed, which are not an integral part of the related hardware and are amortised using the straight line method over a period of the software license, which in the Management's estimate represents the period during which the economic benefits will be derived from their use. Subsequent expenditure is capitalised only when it increases the future economic benefits embodied in the specific to which it relates.

#### 3.04 Impairment of non-financial assets

Goodwill and intangible assets that have an indefinite useful life are not subject to amortisation and are tested annually for impairment, or more frequently if events or changes in circumstances indicate that they might be impaired. Other assets are tested for impairment whenever events or changes in circumstances indicate that the carrying amount may not be recoverable. Management periodically assesses using, external and internal sources, whether there is an indication that an asset may be impaired.

The recoverable amount is higher of the asset's net selling price or value in use, which means the present value of future cash flows expected to arise from the continuing use of the asset and its eventual disposal. An impairment loss for an asset is reversed if, and only if, the reversal can be related objectively to an event occurring after the impairment loss was recognized. The carrying amount of an asset is increased to its revised recoverable amount, provided that this amount does not exceed the carrying amount that would have been determined (net of any accumulated amortization or depreciation) had no impairment loss been recognized for the asset in prior years.

#### 3.05 Borrowing cost

General and specific borrowing costs that are directly attributable to the acquisition, construction or production of a qualifying asset are capitalised during the period of time that is required to complete and prepare the asset for its intended use or sale. Qualifying assets are assets that necessarily take a substantial period of time to get ready for their intended use or sale.

Investment income earned on the temporary investment of specific borrowings pending their expenditure on qualifying assets is deducted from the borrowing costs eligible for capitalisation. Other borrowing costs are expensed in the period in which they are incurred.

3.06 Income-tax

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised directly in equity or in OCI.

#### (i) Current tax

Current tax comprises the expected tax payable or receivable on the taxable income or loss for the year and any adjustment to the tax payable or receivable in respect of previous years. It is measured using tax rates enacted or substantively enacted at the reporting date. Current tax also includes any tax arising from dividends. Management periodically evaluates positions taken in tax returns with respect to situations in which applicable tax regulation is subject to interpretation. It establishes provisions where appropriate on the basis of amounts expected to be paid to the tax authorities.

Minimum Alternate Tax ('MAT') under the provisions of Income-tax Act, 1961 is recognised as current tax in the statement of profit and loss. MAT paid in accordance with the tax laws, which gives future economic benefits in the form of adjustment to future income tax liability, is considered as an asset if there is a convincing evidence that the Company will pay normal tax. Accordingly, MAT is recognized as an asset in the balance sheet when it is probable that the future economic benefit associated with it will flow to the Company.

Current tax assets and liabilities are offset only if, the Company:

a) has a legally enforceable right to set off the recognised amounts; and

b) intends either to settle on a net basis, or to realise the asset and settle the liability simultaneously.

#### (ii) Deferred tax

Deferred tax is recognised in respect of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for taxation purposes. However, deferred tax liabilities are not recognised if they arise from the initial recognition of goodwill. Deferred income tax is also not accounted for if it arises from initial recognition of an asset or liability in a transaction other than a business combination that at the time of the transaction affects neither accounting profit nor taxable profit (tax loss).

Deferred tax assets are recognised for unused tax losses, unused tax credits and deductible temporary differences to the extent that it is probable that future taxable profits will be available against which they can be used. Deferred tax assets are reviewed at each reporting date and are reduced to the extent that it is no longer probable that the related tax benefit will be realised; such reductions are reversed when the probability of future taxable profits improves. Unrecognised deferred tax assets are reassessed at each reporting date and recognised to the extent that it has become probable that future taxable profits will be available against which they can be used.

Deferred tax is measured at the tax rates that are expected to be applied to temporary differences when they reverse, using tax rates enacted or substantively enacted at the reporting date and are expected to apply when the related deferred income tax asset is realised or the deferred income tax liability is settled.

Deferred tax liabilities are not recognised for temporary differences between the carrying amount and tax bases of investments in subsidiaries, branches and associates and interest in joint arrangements where the company is able to control the timing of the reversal of the temporary differences and it is probable that the differences will not reverse in the foreseeable future.

Deferred lax assets are not recognised for temporary differences between the carrying amount and tax bases of investments in subsidiaries, branches and associates and interest in joint arrangements where it is not probable that the differences will reverse in the foreseeable future and taxable profit will not be available against which the temporary difference can be utilised,

Deferred tax assets and liabilities are offset only if:

a) the entity has a legally enforceable right to set off current tax assets against current tax liabilities; and

b) the deferred tax assets and the deferred tax liabilities relate to income taxes levied by the same taxation authority on the same taxable entity.

#### (iii) Current and deferred tax for the year

Current and deferred tax are recognised in profit or loss, except when they relate to items that are recognised in other comprehensive income or directly in equity / deemed equity, in which case, the current and deferred tax are also recognised in other comprehensive income or directly in equity / deemed equity respectively. Where current tax or deferred tax arises from the initial accounting for a business combination, the tax effect is included in the accounting for the business combination.

#### 3.07 Cash and cash equivalents

For the purpose of presentation in the statement of cash flows, cash and cash equivalent includes cash on hand, deposits held at call with financial institutions, other short-term, highly liquid investments with original maturities of three months or less that are readily convertible to known amount of cash and which are subject to an insignificant risk of changes in value, and bank overdrafts. Bank overdrafts are shown within borrowings in current liabilities in the balance sheet.

#### 3.08 Inventories

Direct expenses like land cost, development rights, site labour cost, material used for project construction, cost of borrowing, project management consultancy, costs for moving the plant and machinery to the site and general expenses incurred specifically for the respective project like insurance, design and technical assistance, and construction overheads are taken as the cost of the project work in progress and cost of unsold flats.

Material at site comprise of building material, components, stores and spares, consumables

Inventories are valued at lower of cost or net realizable value, cost is determined on weighted average basis. Net realizable value is the estimated selling price in the ordinary course of business, less estimated costs of completion and estimated costs necessary to make the sale.

#### 3.09 Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity. Financial instruments also include derivative contracts such as put options, call options; and forward contracts.

#### (i) Financial assets

#### Classification

The Company shall classify financial assets as subsequently measured at amortised cost, fair value through other comprehensive income (FVOCI) or fair value through profit or loss (FVTPL) on the basis of its business model for managing the financial assets and the contractual cash flow characteristics of the financial asset.

#### Initial recognition and measurement

Financial assets are initially measured at fair value. Transaction costs that are directly attributable to the acquisition or issue of financial assets (other than fair valued through profit or loss) are added to the fair value of the financial assets, as appropriate, on initial recognition. Transaction costs directly attributable to the acquisition of financial assets fair valued through profit or loss are recognised immediately in profit or loss.

All regular way purchases or sales of financial assets are recognised and derecognised on trade date basis. Regular way purchases or sales are purchases or sales of financial assets that require delivery of assets within the time frame established by regulation or convention in the market place.

All recognised financial assets are subsequently measured in their entirety at either amortised cost or fair value, depending on the classification of the financial assets.

#### **Debt instruments**

- 1, A 'debt instrument' is measured at the amortised cost if both the following conditions are met:
- a) The asset is held within a business model whose objective is to hold assets for collecting contractual cash flows, and
- b) Contractual terms of the asset give rise on specified dates to cash flows that are solely payments of principal and interest (SPPI) on the principal amount outstanding.

2. After initial measurement, such financial assets are subsequently measured at amortised cost using the effective interest rate (EIR) method. Amortised cost is calculated by taking into account any discount or premium and fees or costs that are an integral part of the EIR. The EIR amortisation is included in finance income in the profit or loss.

3, Debt instruments included within the fair value through profit and loss (FVTPL) category are measured at fair value with all changes recognized in the statement of profit and loss.



#### Equity instruments

The Company measures its equity investments in equity shares of subsidiaries, joint ventures and associates at fair value through other comprehensive income.

Equity investments in companies other than equity investments in subsidiaries, joint ventures and associates are measured at fair value through profit and loss account.

Where the company's management has elected to present fair value gains and losses on equity investments in other comprehensive income, there is no subsequent reclassification of fair value gains and losses to profit or loss. Dividends from such investments are recognised in profit or loss as other income when the company's right to receive payments is established.

#### **De-recognition**

A financial asset (or, where applicable, a part of a financial asset or part of a Company of similar financial assets) is primarily derecognised (i.e. removed from the Company's balance sheet) when:

The rights to receive cash flows from the asset have expired, or

The Company has transferred its rights to receive cash flows from the asset or has assumed an obligation to pay the received cash flows in full without material delay to a third party under a 'pass-through' arrangement; and either (a) the Company has transferred substantially all the risks and rewards of the asset, or (b) the Company has neither transferred nor retained substantially all the risks and rewards of the asset, but has transferred control of the asset.

When the Company has transferred its rights to receive cash flows from an asset or has entered into a pass-through arrangement, it evaluates if and to what extent it has retained the risks and rewards of ownership. When it has neither transferred nor retained substantially all of the risks and rewards of the asset, nor transferred control of the asset, the Company continues to recognise the transferred asset to the extent of the Company's continuing involvement. In that case, the Company also recognises an associated liability. The transferred asset and the associated liability are measured on a basis that reflects the rights and obligations that the Company has retained.

Continuing involvement that takes the form of a guarantee over the transferred asset is measured at the lower of the original carrying amount of the asset and the maximum amount of consideration that the Company could be required to repay.

#### Impairment of financial assets

In accordance with Ind-AS 109, the Company applies expected credit loss (ECL) model for measurement and recognition of impairment loss on the following financial assets and credit risk exposure:

a) Financial assets that are debt instruments, and are measured at amortised cost e.g., loans, debt securities, deposits, and bank balance

#### b) Lease receivables

c) Trade receivables

The Company follows 'simplified approach' for recognition of impairment loss allowance on:

Trade receivables which do not contain a significant financing component.

All lease receivables resulting from transactions.

The application of simplified approach does not require the Company to track changes in credit risk. Rather, it recognises impairment loss allowance based on lifetime ECLs at each reporting date, right from its initial recognition.

For recognition of impairment loss on other financial assets and risk exposure, the Company determines that whether there has been a significant increase in the credit risk since initial recognition. If credit risk has not increased significantly, 12-month ECL is used to provide for impairment loss. However, if credit risk has increased significantly, lifetime ECL is used. If, in a subsequent period, credit quality of the instrument improves such that there is no longer a significant increase in credit risk since initial recognition, then the entity reverts to recognising impairment loss allowance based on 12-month ECL.

#### (ii) Financial liabilities

#### Classification

The Company classifies all financial liabilities as subsequently measured at amortised cost, except for financial liabilities at fair value through profit or loss. Such liabilities, including derivatives that are liabilities, shall be subsequently measured at fair value.

Debt and equity instruments issued by the Company are classified as either financial liabilities or equity in accordance with the substance of the contractual arrangements and the definitions of a financial liability and an equity instrument.

#### Initial recognition and measurement

Financial liabilities are classified, at initial recognition, as financial liabilities at fair value through profit or loss, loans and borrowings, payables, or as derivatives designated as hedging instruments in an effective hedge, as appropriate.

All financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

The Company's financial liabilities include trade and other payables, loans and borrowings including bank overdrafts and derivative financial instruments.



Financial liabilities at fair value through profit or loss

Financial liabilities at fair value through profit or loss include financial liabilities held for trading and financial liabilities designated upon initial recognition as at fair value through profit or loss. Financial liabilities are classified as held for trading if they are incurred for the purpose of repurchasing in the near term. This category also includes derivative financial instruments entered into by the Company that are not designated as hedging instruments in hedge relationships as defined by Ind-AS 109. The company does not have any separated embedded derivatives.

Gains or losses on liabilities held for trading are recognised in the profit or loss.

Financial liabilities designated upon initial recognition at fair value through profit or loss are designated at the initial date of recognition, and only if the criteria in Ind-AS 109 are satisfied. For liabilities designated as FVTPL, fair value gains/ losses attributable to changes in own credit risk are recognized in OCI. These gains/loss are not subsequently transferred to profit and loss. However, the Company may transfer the cumulative gain or loss within equity. All other changes in fair value of such liability are recognised in the statement of profit or loss. The Company has not designated any financial liability as at fair value through profit and loss.

#### Loans and borrowings

After initial recognition, interest-bearing loans and borrowings are subsequently measured at amortised cost using the EIR method. Gains and losses are recognised in profit or loss when the liabilities are derecognized.

Amortised cost is calculated by taking into account any discount or premium on acquisition and fees or costs that are an integral part of the EIR. The EIR amortisation is included as finance costs in the statement of profit and loss.

This category generally applies to interest-bearing loans and borrowings.

Preference shares, which are mandatorily redeemable on a specific date, are classified as liabilities. The dividends on these preference shares are recognised in profit or loss as finance costs.

The fair value of the liability portion of an optionally convertible bonds is determined using a market interest rate for an equivalent nonconvertible bonds. This amount is recorded as a liability on an amortised cost basis until extinguished on conversion or redemption of the bonds. The remainder of the proceeds is attributable to the equity portion of the compound instrument. This is recognised and included in shareholders' equity, net of income tax effects, and not subsequently remeasured.

Where the terms of a financial liability are renegotiated and the entity issues equity instruments to a creditor to extinguish all or part of the liability (debt for equity swap), a gain or loss is recognised in profit or loss, which is measured as the difference between the carrying amount of the financial liability and the fair value of the equity instruments issued.

#### De-recognition

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit or loss.

#### Embedded derivatives

If the hybrid contract contains a host that is a financial asset within the scope of Ind-AS 109, the Company does not separate embedded derivatives. The Company applies the classification requirements contained in Ind AS 109 to the entire hybrid contract. Derivatives embedded in all other host contracts are accounted for as separate derivatives and recorded at fair value if their economic characteristics and risks are not closely related to those of the host contracts and the host contracts are not held for trading or designated at fair value through profit or loss. These embedded derivatives are measured at fair value with changes in fair value recognised in profit or loss, unless designated as effective hedging instruments. Reassessment only occurs if there is either a change in the terms of the contract that significantly modifies the cash flows.

#### Offsetting of financial instruments

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable right to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable right must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the company or the counterparty.

#### Derivative financial instruments

The Company has entered into derivative financial instruments, such as put and call option contracts and forward purchase contracts to acquire stake from Non-controlling interests. Such derivative financial instruments are initially recognised at fair value on the date on which a derivative contract is entered into and are subsequently re-measured at fair value through profit or loss account. Derivatives are carried as financial assets when the fair value is positive and as financial liabilities when the fair value is negative.

The company has not designated its derivatives as hedging instruments.

#### Financial guarantee contracts

A financial guarantee contract is a contract that requires issuer to make specified payments to reimburse the holder for a loss it incurs because a specified debtor fails to make payments when due in accordance with the terms of a debt instrument.

Financial guarantee contracts issued by the Company are initially measured at their fair values and, if not designated as at FVTPL, are subsequently measured at the higher of :

(i) the amount of loss allowance determined in accordance with impairment requirements of Ind AS 109; and

(ii)the amount initially recognised less, when appropriate, the cumulative amount of income recognised in accordance with the principles of Ind AS 18.

#### Commitments to provide a loan at a below-market interest rate

Commitments to provide a loan at a below-market interest rate are initially measured at their fair values and, if not designated as at FVTPL, are subsequently measured at the higher of :

(i) the amount of loss allowance determined in accordance with the impairment requirements of Ind AS 109; and

(ii)the amount initially recognised less, when appropriate, the cumulative amount of income recognised in accordance with the principles of Ind AS 18.

#### 3.10 Employee benefits

#### (i) Short term employee benefits

Short-term employee benefits are expensed as the related service is provided. A liability is recognised for the amount expected to be paid if the Company has a present legal or constructive obligation to pay this amount as a result of past service provided by the employee and the obligation can be estimated reliably.

#### (ii) Compensated absences

The liabilities for earned leave and sick leave are not expected to be settled wholly within 12 months after the end of the period in which the employees render the related service. They are therefore measured as the present value of expected future payments to be made in respect of services provided by employees up to the end of the reporting period using the projected unit credit method. The benefits are discounted using the market yields at the end of the reporting period that have terms approximating to the terms of the related obligation. Re-measurements as a result of experience adjustments and changes in actuarial assumptions are recognised in profit or loss.

#### (iii) Defined contribution plans

Obligations for contributions to defined contribution plans are expensed as the related service is provided. Prepaid contributions are recognised as an asset to the extent that a cash refund or a reduction in future payments is available.

#### (iv) Defined benefit plans

The Company's net obligation in respect of defined benefit plans is calculated separately for each plan by estimating the amount of future benefit that employees have earned in the current and prior periods, discounting that amount and deducting the fair value of any plan assets.

The calculation of defined benefit obligations is performed annually by a qualified actuary using the projected unit credit method. When the calculation results in a potential asset for the Company, the recognised asset is limited to the present value of economic benefits available in the form of any future refunds from the plan or reductions in future contributions to the plan. To calculate the present value of economic benefits, consideration is given to any applicable minimum funding requirements.

Re-measurement of the net defined benefit liability, which comprise actuarial gains and losses, the return on plan assets (excluding interest) and the effect of the asset ceiling (if any, excluding interest), are recognised immediately in other comprehensive income (OCI). Net interest expense (income) on the net defined liability (asset) is computed by applying the discount rate, used to measure the net defined liability (asset), to the net defined liability (asset) at the start of the financial year after taking into account any changes as a result of contribution and benefit payments during the year. Net interest expense and other expenses related to defined benefit plans are recognised in profit or loss.

When the benefits of a plan are changed or when a plan is curtailed, the resulting change in benefit that relates to past service or the gain or loss on curtailment is recognised immediately in profit or loss. The Company recognises gains and losses on the settlement of a defined benefit plan when the settlement occurs.

#### (v) Other long-term employee benefits

The Company's net obligation in respect of long-term employee benefits is the amount of future benefit that employees have earned in return for their service in the current and prior periods. That benefit is discounted to determine its present value. Re-measurement are recognised in profit or loss in the period in which they arise.

These plans typically expose the Company to actuarial risks such as : Investment risk, interest rate risk, longevity risk and salary risk:

(i) Investment risk : The present value of the defined benefit plan liability is calculated using the discount rate which is determined by reference to market yields at the end of the reporting period on government bonds. For other defined benefit plans, the discount rate is determined by reference to market yields at the end of the reporting period on high quality corporate bonds when there is a deep market for such bonds; if the return on plan asset is below this rate, it will create a plan deficit. Currently, for the plan in India, it has a relatively balanced mix of investments in government securities and other debt instruments. Further.

(ii) Interest risk : A decrease in the bond interest rate will increase the plan liability; however, this will be partially offset by an increase in the return on the plan's debt investments.

(iii) Longevity risk : The present value of the defined benefit plan liability is calculated by reference to the best estimate of the mortality of plan participants both during and after their employment. An increase in the life expectancy of the plan participants will increase the plan's liability.

(iv) Salary risk : The present value of the defined benefit plan liability is calculated by reference to the future salaries of plan participants. An increase in the Salary of the plan participants will increase the plan's liability.

#### 3.11 Earnings per share

Basic earnings per share is computed by dividing the profit / (loss) after tax by the weighted average number of equity shares outstanding during the year. The weighted average number of equity shares outstanding during the year is adjusted for the events for bonus issue, bonus element in a rights issue to existing shareholders, share split and reverse share split (consolidation of shares).

Diluted earnings per share is computed by dividing the profit / (loss) after tax as adjusted for dividend, interest and other charges to expense or income (net of any attributable taxes) relating to the dilutive potential equity shares, by the weighted average number of equity shares considered for deriving basic earnings per share and the weighted average number of equity shares which could have been issued on conversion of all dilutive potential equity shares.

#### 3.12 Leases

The Company evaluates each contract or arrangement, whether it qualifies as lease as defined under Ind AS 116. The Company as a lessee

The Company assesses, whether the contract is, or contains, a lease. A contract is, or contains, a lease if the contract involves:

- (a) the use of an identified asset,
- (b) the right to obtain substantially all the economic benefits from use of the identified asset, and
- (c) the right to direct the use of the identified asset.

The Company at the inception of the lease contract recognizes a Right-of-Use (RoU) asset at cost and corresponding lease liability, except for leases with term of less than twelve months (short term) and low-value assets.

The cost of the right-of-use assets comprises the amount of the initial measurement of the lease liability, any lease payments made at or before the inception date of the lease plus any initial direct costs, less any lease incentives received. Subsequently, the right of-use assets is measured at cost less any accumulated depreciation and accumulated impairment losses, if any. The right-of-use assets is depreciated using the straight-line method from the commencement date over the shorter of lease term or useful life of right-of-use assets.

The Company applies Ind AS 36 to determine whether a Right-of-Use asset is impaired and accounts for any identified impairment loss in the Statement of Profit and Loss.

For lease liabilities at inception, the Company measures the lease liability at the present value of the lease payments that are not paid at that date. The lease payments are discounted using the interest rate implicit in the lease, if that rate is readily determined, if that rate is not readily determined, the lease payments are discounted using the incremental borrowing rate.

The Company recognizes the amount of the re-measurement of lease liability as an adjustment to the right-of-use assets. Where the carrying amount of the right-of-use assets is reduced to zero and there is a further reduction in the measurement of the lease liability, the Company recognizes any remaining amount of the re-measurement in the Statement of Profit and Loss.

Lease payments (other than short term, low value leases) have been classified as cash used in Financing activities in the Statement of Cash Flows.

Lease payments for short-term and low value leases, have been classified as cash used in Operating activities in the Statement of Cash Flows.

The Company has no assets given on lease to others.

# 4 Property, Plant and Equipment , Intangible assets, Right to use an asset and Capital work-in-progress a Property, plant and equipment (PPE)

Particulars	Freehold Land	Leasehold Improvements	Plant & Equipments	Furniture & Fixtures	Data Processing Equipments	Office Equipments	Motor Vehicles	Total
COST / DEEMED COST								
As at 01 April 2018	1,144.90	8.97		145.36	297.92	196.92	70.72	1,864.79
Additions		466.93		7.41	15.62	39.09		529.05
Disposals/Adjustments				(21.89)	(1.59)	(15.70)	(44.04)	(83.22)
As at 31 March 2019	1,144.90	475.90	•	130.88	311.95	220.31	26.68	2,310.62
Additions			18.11	- Q.	11.34	0.89	1.22	30.34
Disposals/Adjustments	1	0.25	2.12	(2.01)	(21.91)	(45.81)	(13.68)	(81.04)
As at 31 March 2020	1,144.90	476.15	20,23	128.87	301.38	175.39	13.00	2,259,92
DEPRECIATION								
As at 01 April 2018		1.51		82.04	240.38	118.58	25.79	468.30
Charge for the Year		30.57		15.81	35.69	37.17	7.24	126.48
Disposals/Adjustments				(4.07)	(1.09)	(4.77)	(17.62)	(27.55)
As at 31 March 2019	1.90	32.08		93.78	274.98	150.98	15.41	567.24
Charge for the Year		108.72	3.39	9,49	21.92	24.65	4.60	172.77
Disposals/Adjustments	-		0.37	(0.75)	(19.78)	(45.15)	(9.61)	(74.93)
As at 31 March 2020	•	140.80	3.76	102.51	277.12	130.48	10.40	665.08
NET BLOCK			100 C			2		
As at 31 March 2019	1,144.90	443.82		37.10	36.97	69.33	11.27	1,743.39
As at 31 March 2020	1,144.90	335.35	16.47	26.36	24.26	44.92	2.60	1,594.84

#### b Intangible Assets

Particulars	Computer Software	Total
GROSS BLOCK		
As at 01 April 2018	349.74	349.74
Additions	5.72	5.72
Disposals/Adjustments		-
As at 31 March 2019	355.46	355.46
Additions	42.24	42.24
Disposals/Adjustments	(33.22)	(33.22
As at 31 March 2020	364.48	364.48
AMORTISATION	001110	001.10
As at 01 April 2018	291.20	291.20
Charge for the Year	13.82	13.82
Disposals/Adjustments	10.02	-
As at 31 March 2019	305.02	305.02
Charge for the Year	14.75	14.75
Disposals/Adjustments	(32.98)	(32.98
As at 31 March 2020	286.78	286.78
NET BLOCK	200.78	200.70
As at 31 March 2019	50.43	50.43
As at 31 March 2020	77.70	77,70
	11.10	11.10
Right to use an asset Particulars		*
GROSS BLOCK		Total
As at 01 April 2018		
Additions		
As at 31 March 2019		
Additions		150.66
As at 31 March 2020		150.66
AMORTISATION		
Charge for the Year		
As at 31 March 2019		
Charge for the Year		4.28
As at 31 March 2020		4.28
NET BLOCK		
As at 31 March 2019		
As at 31 March 2020		146.38
Capital Work in Progress		
Particulars		Total
As at 01 April 2018		3.76
Additions/Transfer to PPE		
As at 31 March 2019		3.76
Additions/Transfer to PPE		(3.76)

As at 3

Outerity         Amount         Outerity         Amount         Outerity         Amount           Imported Explor barrier of MP to ach, MP pais-up:         1)         1         50.000         -         24.41         -         24.41         -         -         24.41         -         -         24.41         -         -         24.40         - <t< th=""><th>Particulars</th><th>31 March</th><th>2020</th><th>31 Marcl</th><th>2019</th></t<>	Particulars	31 March	2020	31 Marcl	2019
1 Pri Valaditi Stoph Oline Comprehensive Acone:		Quantity	Amount		
Ungunder Equity interes of INIT 10 acch, May paisage:         50.000         1.50.000         50.000         50.000           Answer int, States Private Limited         1.82.200         9.84.800         1.02.200         9.58.8           Degraps of Relative Private Limited         6.50.00         5.51.66         6.570         8.56.60         5.52.60         9.55.70         8.56.60         5.52.60         9.55.70         8.56.70         8.56.70         8.56.70         8.56.70         8.56.70         8.56.70         8.56.70         8.56.70         8.55.70         7.55.70         7.55.70         7.55.70         7.55.70         7.55.70         7.55.70         7.55.70         7.55.70         7.55.70         7.55.70         7.55.70         7.55.70         7.55.70         7.55.70 <td></td> <td></td> <td></td> <td></td> <td></td>					
I)Investment in ubsklary comparise:         50,000         5.000         2.000         7.0000         7.000         7.0000					
Ame         9,000         -         9,000           Answ Intra Stanton Private Limited         9,000         5,000         5,000         5,000           Auge Inter Stanton Private Limited         9,000         5,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1					
Arrow thm Estates Private Limited         1.02.000         3.88.000         1.02.000         6.58.00           Cargen Constructivel Private Limited         5.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         1.64.00         6.57.00         7.67.00         6.57.00					
Gurgen Constructivel Private Limited         65.700         55.700         7		50,000		50,000	
Defined Early Investments in Outgano Constructed Private Limited         55.50         12.41 <td></td> <td>1,02,300</td> <td>3.583.00</td> <td>1,02,300</td> <td>6,362.</td>		1,02,300	3.583.00	1,02,300	6,362.
Detend Equity Netameta Indiana         -         518.68         -         517           Cargion Ralabia Limitad         77.45.01         25.242,74         77.45.01         23.22           Detend Equity Investments In TRL Icass Prevase Limited         77.45.01         23.22         23.22           Detend Equity Investments In TRL Urban Transport Prevase Limited         77.45.01         24.93.00         2.33           Denned Equity Investments In TRL Urban Transport Prevase Limited         7.128.400         2.44.3         0.40.00           TRL Urban Transport Prevase Limited         7.128.400         2.45.18         2.40.000         2.45.18           Denned Equity Investments In TRL Aminas Projects Limited         7.128.400         2.45.18         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.84.56         2.40.000         1.02.85         7.61.56         2.40.00         1.62.55         7.61.50         2.40.00         1.62.55	Gurgaon Constructwell Private Limited	65,700	8,164.00	65.700	8,599.
Gurgan Rester: Limited         66.500         5.564.00         6.500         5.500         5.500         5.500           TRII. Roads Private Limited         774.5112         55.247.4         774.5012         55.27.012         52.247.0         774.5012         55.200         2.247.0         50.000         2.247.0         50.000         2.247.0         50.000         2.247.0         77.50.12         55.200         2.245.10         2.245.80.00         5.000         77.000	Deemed Equity Investments in Gurgaon Constructwell Private Limited		619.68		619.
Thill, Rods Private Limited         77,45012         28,243,74         77,45012         28,243,74           The Floating Tripped Private Limited         50,000         -         2,000           Deamed Equity Investments in Floating Private Limited         50,000         2,000         2,000           The Floating Private Private Limited         71,12,80,000         2,800,000         2,800,000         2,800,000         2,800,000         2,800,000         2,800,000         2,800,000         2,800,000         2,800,000         2,800,000         2,800,000         1,02,800         2,800,000         1,03,800         2,800,000         1,03,800         2,800,000         1,03,800         2,800,000         1,03,800         2,800,000         1,03,800         2,800,000         1,03,800         2,800,000         1,03,800         2,800,000         1,03,800         2,800,000         1,03,800         2,800,000         1,000,000         7,000,000         2,800,000         1,000,000         1,000,000         1,000,000         1,000,000         7,000,000         2,800,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000         1,000,000	Gurgaon Realtech Limited	66,500	5,568,00	66,500	6.907.
Desmed Suply Prestments in THL Reads Provas Limited         2.421.61         -         2.421.61           THF Grugos Notional Project Provas Limited         7.123.800         6.212.00         2.633.000         2.33           THL Ubban Transport Provas Limited         7.123.800         6.212.00         2.633.000         2.63           THL Ubban Transport Provas Limited         7.123.800         2.455.15         2.440.000         2.455           THL Indigat Umbarding Project Summed         2.450.000         1.12.824.65         2.800.000         1.02.824.65         2.800.000         1.02.824.65         2.800.000         1.02.824.65         2.800.000         1.02.824.65         2.800.000         1.02.824.65         2.800.000         1.02.824.65         2.800.000         1.02.824.65         2.800.000         1.02.824         3.426         2.800.000         1.02.824         3.426         2.800.000         1.02.824         3.426         2.810.00         3.426         2.817         1.82.836         3.427.00         3.426.820	TRIL Roads Private Limited*				34,723
THE         Elugianer Housing Progets Private Limited         50.000         -         50.000           Demonst Europi Proves Limited         7.12.83.000         6.212.00         2.63.50.00         2.33           THE         Using Transport Private Limited         7.12.83.000         6.212.00         2.63.50.00         2.34           THE         Ortson Strategy         4.40.000         2.64.50.00         2.64.00.00 </td <td>Deemed Equity Investments in TRIL Roads Private Limited</td> <td></td> <td></td> <td>refretere.</td> <td></td>	Deemed Equity Investments in TRIL Roads Private Limited			refretere.	
THU Uban Transport Private Limited*         7,12,80,00         6,242,00         2,69,300         2,69,300           Demmed Suphy Vestimets in Fills Uban Transport Private Limited         40,000         -         40,000         2,64,000           THU: Indipate Vestimets in Fills Ubans the Youse Private Limited         2,44,000         2,88,18         2,44,000         2,88,18         2,44,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         2,64,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         1,000         2,64,000         4,64,000         4,64,000         4,64,000         4,64,000         1,0	TRIF Gurgaon Housing Projects Private Limited	50.000		50,000	
Deemed Equity Investments In TRU, Urban Transpot Private Limited         24.4.3.3         24.4.0.0.00         2.4.0.0.00         2.4.0.	TRIL Urban Transport Private Limited*		6 212 00		2 334
Weiker Facily Management Services Press Limited         4,00,000         -         4,00,000           TRIL Cristical Controls Limited         2,440,000         2,441,0000         2,441,0000           TRIL Indegrate Limited         2,440,000         2,441,0000         2,441,0000         2,441,0000           TRIL Amings Trained (Face Foot Rise 1)         -         2,840,000         3,485         2,440,000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         2,441,0000         1,457,300         1,457,300         1,457,300         1,457,300         1,457,300         1,457,300         1,457,300         1,457,300         1,457,300         1,457,300         1,457,300         1,457,300         1,457,300         1,457,300         1,457,400         2,457,000         2,449,00         1,17,	Deemed Equity Investments in TRIL Urban Transport Private Limited	1112,001000		2100,001000	
THL         2.44.00.000         2.88.19         2.44.00.000         2.88.19         2.44.00.000         1.03.22           THL         Indigate TRL         82.89.0.000         1.12.824.56         82.89.0.000         1.03.22           THL         Indigate TRL         82.89.0.000         1.03.22         2.44.08.48         3.48           THL         Indigate TRL         82.89.0.000         1.03.23         2.48         7.65           THL         Indigate TRL         1.00.01         1.00		4 00 000	2.40.40	4 00 000	240
TTIL. Indegraf. Limited (Pater Foot Note 1)         \$2,89,00.000         1,12,83,456         \$2,80,0000         1,22,83,456         \$2,80,0000         3,44,054,00         3,45,00         1,50,00         3,44,054,00         1,50,00         2,4,497,00         3,256,00         7,40,00         2,4,497,00         3,256,00         3,22,207,00         2,4,497,00         3,256,00         3,250,00         2,24,00         3,256,00         3,250,00         2,24,00         3,256,00         3,250,00         2,4,497,00         3,256,00         3,250,00         3,250,00         3,250,00         3,250,00         3,250,00         3,250,00         3,250,00         3,250,00         3,250,00         3,250,00         3,250,00			2 895 10		0.451
TTIL.         Anistar Project         38:49:40         32:49:40					
Deemed Equity Investments in TRL Annutary Projects Limited         2.43           TAPR Peak Extend to evolption it unteet:         38.78.4.3           IV Parse Private Limited         10.000           MA Intrate-Private Limited         13.57.300           IV Parse Private Limited         13.57.300           IV Private Limited         14.863.55           IV Private Limited         14.863.55           IV Private Limited         14.863.55           IV Private Limited         14.863.55           IV Private Limited         14.987.400           IV Private Limited         14.988.740           IV Private Limited         14.988.740           IV Private Limited         14.988.740		62,63,00,000	1,12,024.00		
THF Paral Existic Development Limited       -				32,43,08,408	
HV Parts Protect Limited         10.000         1.00         10.000         10.000           MA Instacted Private Limited         13.57.300         13.57.300         150.000           Dermet Equity Investment In Instantial Intrabulat Private Limited         26.000         150.000         26.000         150.000           Mactor Relations Private Limited*         7.40.000         26.000         7.40.000         26.467           Mactor Relations Private Limited*         1.968.7400         25.60.75         1.98.7400         33.268         22.200           Industrial Minerals and Daminosk Company Private Limited*         3.268         22.00.77         3.268         22.200         33.268         22.200         3.268         22.200         05.328.000         5.92.80.000         1.04.73.960         1.02.87           Unquoted Determines of INR 100 each in Industrial Minerals and Chamicals Company Private Limited         1.17.87.460         1.21.4.85         1.04.73.9					
MA Infratech Private Limited       13.57.300       11.9.57.300       28.000       19.000         Deemed Equity investment in hieraational Infrabuld Private Limited       1.80.07       28.000       19.000         IT RHL TT APRIVE Limited       1.80.57.400       28.000       7.40.000       24.467         IT RHL TT APRIVE Limited       1.9.87.400       28.000       7.40.000       24.467         Insustait Minosite and Deminication Company Private Limited       3.286       28.000       7.40.000       24.467         Insustait Minosite and Deminication Company Private Limited       3.286       28.000       7.40.000       24.467         Unquoted Preference shares, NUY paid-up (Compound financial instruments)       10.987.400       3.286       28.000       5.92.80.000					
Interactional Infrabule Private Limited         26.000         -         26.000         150           If investment in joint ventures:         -         1.680.35         -         1.680.30         1.680.30<					1
Deamed Equity investment in International Infrabult Private Limited         1,880.35         1,880.35           10 Instant in joint ventures: TAIL IT 4 Private Limited'         7,40.000         40,358.00         7,40.000         24,497           Mikado Realtors Private Limited'         1.98,67.400         25,805.75         1.98,87.400         33,256           Industrial Mentals Company Private Limited'         3.256         20,000         7,019.22         5,928.000         5,979           Investment in sublidiary companies: Origo Computory Convertible Preference shares of INR 100 each in TRIL Inforant Limited         5,928.0000         7,019.22         5,928.0000         48,576           Unquicted Detentures, Iuly paid-up (Compound Intrabults Minerals and Chemicals Company Private Limited         1,17,87,460         12,314.95         1.04,73,960         10,836           Dravisord Detentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited         1,17,87,460         1,17,800         1,18,000         1,420,000         1,94,000					
I)         Investment in joint ventures:         7.40,000         40,358,000         7.40,000         24.497           TRUE 17 4 Private Limited*         1.99,97,400         25.807,76         3.256         22.200           Industrial Minerals and Chamicals Companies:         3.256         22.057,79         3.256         22.200           Industrial Minerals and Chamicals Companies:         0.001% Companies:         3.256         22.000,000         5.970           0.001% Companies:         0.001% Companies:         0.000,000         5.92,80,000         7.018.32         5.92,80,000         5.970           0.001% Companies:         0.000,000         5.92,80,000         7.018.32         5.92,80,000         5.970           0.001% Companies:         0.000,000         5.970         0.000,000         5.970         0.000,000         5.970           0.011% Computionity Convertible Potentures of INR 100 each in IRU transport Private Limited         1,17,87,460         12,314,95         1.04,73,960         1.980           10 Investiment in aubidary companies:         0.113,40,400         1,17,600         1.13,40,400         1.282           10 Investiment in aubidary companies:         0.113,40,400         1.13,40,400         1.13,40,400         1.282           10 Investiment in aubidary companies:         1.13,40,400         <		26,000	· · · · · ·	26,000	150
TRUE 14 Private Limited*       7.40.000       24.437         Mikada Realizations Private Limited*       1.99.87.400       25.806.76       1.99.87.400       23.226         Unquoted Preference shares, fully paid-up (Compound financial instruments)       1       1.99.87.400       25.92.80.000       5.97.9         Directioned in subsidiary companies:       0.001% Computory Conventible Preference shares of INR 10 each in TRIL Constructions Limited       3.000.0000       5.92.80.000       5.97.9         Over Computory Conventible Preference shares of INR 100 each in TRIL Infoark Limited       3.000.0000       5.92.80.000       5.97.9         Over Computory Conventible Preference shares of INR 100 each in ITRIL Undoark Limited       3.00.00.000       5.92.80.000       7.019.32       5.92.80.000       4.95.78         Unquoted Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited       1.17.87.460       12.314.95       1.04.73.960       10.636         Fair valued through Profit and Loss:       Unquoted Debentures in ITRIL Undan Transport Private Limited       3.757.26       24.57.00.000       22.85.00         Oblis Computory Conventible Debentures in TRIL Undan Transport Private Limited       3.766.20.000       37.577.26       24.57.00.000       22.85.00         Oblis Computory Conventible Debentures in TRIL Intraptort Private Limited       2.780.000       1.157.50       11.34.04.00       14	Deemed Equity investment in International Infrabuild Private Limited	8	1,680.35	· •	
TTIL 17 4 Private Limited*       7.40.000       24.437         Mikado Realizarios Private Limited*       1.99.87.400       25.406.76       33.226         Industal Minerals and Chamicals Companies:       3.256       23.057.79       3.266       22.270         Directional in associativy companies:       0.001% Computatory Conventible Proference shares of INR 10 each in TRIL Constructions Limited       5.92.80.000       7.019.32       5.92.80.000       4.95.79         Directional in associativy companies:       0.001% Computatory Conventible Proference shares of INR 100 each in ITRIL Infocark Limited       3.000.0000       5.92.80.000       4.95.79         Unquoted Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited       1.17.87.460       12.314.95       1.04.73.960       10.636         Fair valued through Profit and Loss:       Industrial Minerals and Schemed       3.75.72.8       2.45.700.000       22.85.00         Unquoted Debentures of INR 10 each, fully paid-up:       1       1.13.40.400       1.17.87.460       1.13.40.400       1.42.39.60       1.02.85         Undustrial Minerals       3.256.20.0000       37.577.28       2.45.700.000       22.85.70       2.85.00       2.25.000.000       4.12.73       2.95.00.000       4.12.73       2.95.00.000       4.12.73       2.95.00.000       4.12.73.000       1.95.75.00       2.15.05.	In International In Industry				
Mikado Realtors Private Limited*         1.9.87.400         55.806.76         1.9.87.400         55.22.20           Industrial Minerais and Chemicals Company Private Limited*         3.256         22.007.79         3.256         22.200           Diquoted Peterance shares, fully paid-up (Compound financial instruments)         0.001% Compulsory Conventible Preference shares of INR 100 each in TRIL Infogark Limited         5.92.80.000         7.019.32         5.92.80.000         45.576           O.001% Compulsory Conventible Preference shares of INR 100 each in TRIL Infogark Limited         3.00.00.000         58.835.00         3.00.00.000         49.578           Unquoted Debentures, fully paid-up:         1) Instantion in joint venture:         1.04.73.960         1.04.60         2.00.00         2.00.00         2.00.00			and the second se	10 m to 10 m	
Industrial Minerals and Chemicals Company Private Limited         3.256         24,057,79         3.256         22,220           Unquoted Preference shares, fully paid-up (Compound financial instruments)         1)         1)         10 each in TRIL Constructions Limited         5,52,80,000         7,019,32         5,52,80,000         45,578           0.001% Computiory Conventible Preference shares of INR 100 each in TRIL Incoark Limited         3,000,000         20,835,00         3,000,000         48,578           Unquoted Detentures, fully paid-up:         1)         Nestment in joint venture:         Compulsory Conventible Detentures of INR 100 each in TRIL Incoark Limited         1,17,87,480         12,314,95         1,04,73,960         10,836           IF ar valued through Profit and Loss:         Unquoted Detentures in TRIL Urban Transport Private Limited         1,13,40,400         1,17,87,480         12,314,95         1,04,73,960         10,836           0 //s Compulsory Conventible Detentures in TRIL Urban Transport Private Limited         3,235,756         2,881,00         1,154,0400         1,242           0 //s Compulsory Conventible Detentures in TRIL Urban Transport Private Limited         3,235,756         2,881,00         1,154,0400         1,242           0 //s Compulsory Conventible Detentures in TRIL Intogent Limited         3,235,756         2,881,00         1,154,020         1,242           0 //s Compulsory Con					
Unquoted Preference shares, fully paid-up (Compound financial instruments)           1) Investment in subsidiary companies:           0.001% Compulsory Conventible Preference shares of INR 100 each in TRIL Constructions Limited         5.92,80,000         7,019,32         5.92,80,000         48,578           Unquoted Debentures, fully paid-up:         1         3.00,00,000         58,855,00         3.00,00,000         58,855,00         3.00,00,000         48,578           Unquoted Debentures, fully paid-up:         1         1.13,40,400         12,314,95         1.04,73,960         10,866           1 Pair stantin in point venture:         Compulsority Conventible Preference shares of INR 100 each in Industrial Minerals and Chemicals Company Private Limited         1,17,87,460         12,314,95         1.04,73,960         10,866           1 Provestment in subsidiary companies:         0         9% Optionally Conventible Debentures in TRIL Unan Transport Private Limited         3.28,5750         2.881,00         3.11,53,750         2.918           0.01% Compulsority Conventible Debentures in TRIL Indea Private Limited         3.28,63,750         2.861,00         3.11,53,750         2.916           0.01% Compulsority Conventible Debentures in TRIL Indea Private Limited         3.28,63,750         2.45,700,000         2.17,80,00         1.94,40,00         1.94,80,00           0.% Optionally Conventible Debentures of INR 687,500 each in TRIL IT4					
I) Investment in subsidiary companies:         5.92.80.000         7.019.32         5.92.80.000         5.92.80.000         5.97.80.000         4.95.78           Unquoted Debentures, fully paid-up:         0.901% Compulsory Convertible Preference shares of INR 100 each in TRIL Infogank Limited         3.00,00.000         53.835.00         3.00,00.000         49.578           Unquoted Debentures, fully paid-up:         0.191% Compulsority Convertible Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited         1,17.87.460         12.314.95         1.04.73.960         10.636           Fair valued through Profit and Loss:         Unquoted Debentures of INR 10 each, fully paid-up:         1         1.13.40,400         1,176.00         1.13.40,400         1.244.95         1.04.73.960         10.636           Ontig Compulsority Convertible Debentures in TRIL Urban Transport Private Limited.         1,13.40,400         1,176.00         1.13.40,400         1.242.95         2.97.87.26         24.57.00.000         24.57.00         2.97.87.26         24.57.00.000         24.57.00         2.97.87.26         24.57.00.000         1.13.40.400         1.234.95         1.04.73.960         1.98.60.000         1.98.60.000         1.98.60.000         1.98.60.0000         1.98.60.0000         1.98.60.0000         1.98.60.0000         1.98.60.0000         1.98.60.0000         1.98.60.0000         1.98.60.0000         1.98.6	Industrial Minerals and Chemicals Company Private Limited*	3,256	23,057,79	3,256	22,270.
0.001% Compulsory Convertible Preference shares of INR 109 each in TRIL Infoark Limited         5.82.80.000         7.013.22         5.32.80.000         49.579           Orgonalos Optimizer, Sully paid-up:         0.001% Compulsory Convertible Detentures of INR 100 each in TRIL Infoark Limited         3.00.000         53.835.00         3.00.000         49.579           Orgonalo Debentures, Nully paid-up:         0.01% Compulsority Convertible Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited         1,17.87.460         12.314.95         1.04.73.960         10.656           Fair valued through Profit and Loss:         0.000 optimizers         1.13.40.400         1,17.87.460         1.34.0.400         1.14.40.400         1.14.40.400         1.14.40.400         1.14.40.400         1.14.40.400         1.2314.95         1.04.73.960 <td></td> <td></td> <td></td> <td></td> <td></td>					
0% Compulsory Convertible Preference shares of INR 100 each in TRIL Intopark Limited         3,00,00,000         53,835.00         3,00,00,000         49,578           Unquoted Debentures, (IU) paid-up: () investment in plot venture: Compulsorily Convertible Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited         1,17,87,460         12,314,35         1,04,73,660         10,636           Fair valued through Profit and Loss: Unquoted Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited         1,13,40,400         1,17,6,00         113,40,400         1,242           ON% Optionally Convertible Debentures in TRIL Urban Transport Private Limited. 00.01% Computabol Debentures in TRIL Index Transport Private Limited. 00.01% Computabol Debentures in TRIL Reads Private Limited. 0% Optionally Convertible Debentures in TRIL Index Transport Private Limited. 0% Optionally Convertible Debentures in INPL Index Transport Private Limited. 0% Optionally Convertible Debentures in INPL Index Transport Private Limited. 0% Optionally Convertible Debentures in INPL Index Transport Private Limited. 0% Optionally Convertible Debentures in INPL Index Transport Private Limited. 0% Optionally Convertible Debentures of INR 687,500 each in TRIL IT4 Private Limited. 1,184         1,184         8,140,00         1,184         8,140,00         1,184         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00 <td< td=""><td></td><td>5 92 80 000</td><td>7 010 22</td><td>5 92 80 000</td><td>5 070</td></td<>		5 92 80 000	7 010 22	5 92 80 000	5 070
Unquoted Debentures, fully paid-up: 1) investment in joint venture: Compulsority Convertible Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited 1,17,87,460 12,314,95 1,04,73,960 10,656 Fair valued through Profit and Loss: Unquoted Debentures of INR 100 each, fully paid-up: 1) Investment in subsidiary companies: 0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited 0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited 0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited 0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited 0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited 0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited 0% Optionally Convertible Debentures in TRIL Infogark Limited 0% Optionally Convertible Debentures in TRIL Infogark Limited 10,00,000 1,064,00 1,064,00 1,064,00 1,064,00 1,184 8,140,00 1,184 8,140,00 8,140 8,140,00					
I) Investment in joint venture: Computsoftly Convertible Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited         1,17,87,460         12,314,95         1,04,73,960         10,656           Fair valued through Profit and Loss:         Unquoted Debentures of INR 100 each, fully paid-up:         1,13,40,400         1,13,40,400         1,242           0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited         3,25,53,756         2,821,00         3,11,53,750         2,916           0% Optionally Convertible Debentures in TRIL IDran Transport Private Limited         3,26,820,000         3,7,877,26         24,57,000,000         24,256           0% Optionally Convertible Debentures in TRIL Indext United         10,00,000         1,964,00         10,00,000         1,964,00         10,00,000         1,184           0% Optionally Convertible Debentures of INR 687,500 each in TRIL ITA Private Limited         27,80,000         191,60         27,80,000         211.           Duoted Debentures of INR 687,500 each in TRIL ITA Private Limited         1,184         6,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         1,1184	on banpasery banaraban relations of the role agents that the mopel's Emiles	3,00,00,000	53,635.00	3,00,00,000	49,578.
I) Investment in joint venture: Computsoftly Convertible Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited         1,17,87,460         12,314,95         1,04,73,960         10,656           Fair valued through Profit and Loss:         Unquoted Debentures of INR 100 each, fully paid-up:         1,13,40,400         1,13,40,400         1,242           0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited         3,25,53,756         2,821,00         3,11,53,750         2,916           0% Optionally Convertible Debentures in TRIL IDran Transport Private Limited         3,26,820,000         3,7,877,26         24,57,000,000         24,256           0% Optionally Convertible Debentures in TRIL Indext United         10,00,000         1,964,00         10,00,000         1,964,00         10,00,000         1,184           0% Optionally Convertible Debentures of INR 687,500 each in TRIL ITA Private Limited         27,80,000         191,60         27,80,000         211.           Duoted Debentures of INR 687,500 each in TRIL ITA Private Limited         1,184         6,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         8,140,00         1,1184	Unguoted Debegtures, fully paid-up-				
Compulsority Convertible Debentures of INR 100 each in Industrial Minerals and Chemicals Company Private Limited         1,17,87,460         12,314,95         1,04,73,960         10,836           IF air valued through Profit and Loss:         Unquoted Debentures of INR 100 each, fully paid-up:         1         1,13,40,400         1,176,00         1,13,40,400         1,242           0% Optionally Convertible Debentures in TRIL, Urban Transport Private Limited         3,23,53,750         2,821,00         3,11,53,750         2,916           0.01% Compulsority Convertible Debentures in TRIL, Integark Limited         3,23,53,750         2,821,00         3,11,53,750         2,916           0.01% Compulsority Convertible Debentures in TRIL, Integark Limited         3,26,50,00,000         44,852,00         25,500,00,000         44,22,500,00,000         44,22,500,00,000         44,22,500,00,000         44,22,500,00,000         44,22,500,00,000         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,22,500,00,00         44,20,771,34         4,006,00,55           Pripate value					
Fair valued through Profit and Loss:       1,17,400       1,231,530       10,47,3,500       10,558         Unquoted Debentures of INR 10 each, fully paid-up:       1) Investment in subsidiary companies:       1,13,40,400       1,16,00       1,13,40,400       1,242         0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited       3,23,53,750       2,821,00       3,11,53,750       2,918         0.01% Compulsority Convertible Debentures in TRIL Integark Limited       3,23,53,750       2,821,00       3,11,53,750       2,918         0.01% Compulsority Convertible Debentures in TRIL Integark Limited       3,23,53,750       2,821,00       3,11,53,750       2,918         0% Optionally Convertible Debentures in TRIL Integark Limited       10,00,000       1,064,00       10,00,000       1,184         0% Optionally Convertible Debentures in INE Intrabuid Private Limited       27,80,000       191,80       27,80,000       211         2uoted Debentures (INP paid-up:       1       11,84       8,140,00       8,140       8,140,00       8,140         19// gregate tai' value.       figregate value of quoted investments       4,22,631,34       3,97,895       3,97,895         urrent Investments       4,22,631,34       3,97,895       3,84,66       10,513,10       -       -         NAV per unit (in NR) 33,93,761 (2015 · NA) <td></td> <td></td> <td></td> <td></td> <td></td>					
Unquoted Debentures of INR 10 each, fully paid-up:       1, investment in subsidiary companies:       1, 13, 40, 400       1, 13, 40, 400       1, 242         0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited       3, 25, 53, 750       2, 821, 00       3, 11, 53, 750       2, 918         0% Optionally Convertible Debentures in TRIL Intopark Limited       3, 26, 20, 00       4, 62, 00       24, 550, 00, 000       44, 662, 00       25, 60, 00, 000       44, 662, 00       25, 60, 00, 000       44, 662, 00       25, 60, 00, 000       44, 662, 00       25, 60, 00, 000       44, 662, 00       26, 00, 00, 00       44, 662, 00       26, 00, 00, 00       44, 662, 00       26, 00, 00, 00       44, 662, 00       26, 00, 00, 00       44, 662, 00       27, 80, 000       111, 184       8, 140, 00       111, 184       8, 140, 00       111, 184       8, 140, 00       111, 184       8, 140, 00	Sumparanty convertice destinates of the rockast in mustania minerals and chemicals company envale crimes	1,17,87,460	12,314.95	1.04,73,960	10,636.
Unquoted Debentures of INR 10 each, fully paid-up:       1, investment in subsidiary companies:       1, 13, 40, 400       1, 13, 40, 400       1, 242         0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited       3, 25, 53, 750       2, 821, 00       3, 11, 53, 750       2, 918         0% Optionally Convertible Debentures in TRIL Intopark Limited       3, 26, 20, 00       4, 62, 00       24, 550, 00, 000       44, 662, 00       25, 60, 00, 000       44, 662, 00       25, 60, 00, 000       44, 662, 00       25, 60, 00, 000       44, 662, 00       25, 60, 00, 000       44, 662, 00       26, 00, 00, 00       44, 662, 00       26, 00, 00, 00       44, 662, 00       26, 00, 00, 00       44, 662, 00       26, 00, 00, 00       44, 662, 00       27, 80, 000       111, 184       8, 140, 00       111, 184       8, 140, 00       111, 184       8, 140, 00       111, 184       8, 140, 00					
I) Investment in subsidiary companies:         0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited.         1,13,40,400         1,134,0,400         1,242           0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited.         3,23,53,750         2,821,00         3,11,53,750         2,916           0% Optionally Convertible Debentures in TRIL Indeak Private Limited.         3,76,820,000         37,577,26         24,57,00,000         41,273           0% Optionally Convertible Debentures in TRIL Indeak Itribacik Umited.         10,00,000         1,064,00         10,00,000         1,184         25,00,000         41,823         25,00,000         41,273           0% Optionally Convertible Debentures in TRIL Indeak Umited.         10,00,000         1,064,00         10,00,000         1,184         27,80,000         11,184         8,140,00         1,184           Quest Debentures of INR 687,500 each in TRIL IT4 Private Limited         1,184         8,140,00         1,184         8,140,00         8,140           Intestment in joint venture:         13         8,140,00         1,184         8,140,00         8,140           Ipregate value of quoted investments         8,140,00         8,140,00         8,140         1,184         3,97,895           urrent Investments         8,140,00         1,184         3,140,00         3,146,2019 </td <td></td> <td></td> <td></td> <td></td> <td></td>					
0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited.         1,13,40,400         1,13,40,400         1,242           0.01% Compulsority Convertible Debentures in TRIL Intopark Limited.         3,23,53,750         2,821,00         3,11,53,750         2,916           0.01% Compulsority Convertible Debentures in TRIL Intopark Limited.         3,23,53,750         2,821,00         3,11,53,750         2,916           0.01% Compulsority Convertible Debentures in TRIL Intopark Limited.         25,00,00,000         44,882,00         25,00,00,000         41,273           0% Optionally Convertible Debentures in International Intrabuild Private Limited         10,00,000         1,064,00         10,00,000         1,184         8,140,00         1,184         8,140,00         1,184         8,140,00         1,184         8,140,00 <td>Unquoted Debentures of INR 10 each, fully paid-up:</td> <td></td> <td></td> <td></td> <td></td>	Unquoted Debentures of INR 10 each, fully paid-up:				
0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited.         1,13,40,400         1,13,40,400         1,242           0.01% Compulsority Convertible Debentures in TRIL Intopark Limited.         3,23,53,750         2,821,00         3,11,53,750         2,916           0.01% Compulsority Convertible Debentures in TRIL Intopark Limited.         3,23,53,750         2,821,00         3,11,53,750         2,916           0.01% Compulsority Convertible Debentures in TRIL Intopark Limited.         25,00,00,000         44,882,00         25,00,00,000         41,273           0% Optionally Convertible Debentures in International Intrabuild Private Limited         10,00,000         1,064,00         10,00,000         1,184         8,140,00         1,184         8,140,00         1,184         8,140,00         1,184         8,140,00 <td></td> <td></td> <td></td> <td></td> <td></td>					
0.01% Compulsorily Convertible Debentures in TRIL Urban Transport Private Limited         3,21,53,750         2,821,00         3,11,53,750         2,916           0% Optionally Convertible Debentures in TRIL Intogark Limited         37,68,20,000         37,577,26         24,57,00,000         22,565           0% Optionally Convertible Debentures in TRIL Intogark Limited         37,68,20,000         37,577,26         24,57,00,000         21,525           0% Optionally Convertible Debentures in TRIL Intogark Limited         10,00,000         1,064,00         10,00,000         1,168           0% Optionally Convertible Debentures in TRIL Integark Limited         27,80,000         191,60         27,80,000         211,12           0 Newsteinent in joint venture:         18         8,140,00         1,184         8,140,00         8,140           10 Newsteinents         8,140,00         8,140         1,184         8,140,00         8,140           gregate value of unquoted investments         8,140,00         8,140         8,140         3,97,855           urrent Investments         3,163,750         2,98,73,36         10,513,10         -         -           NAV per unit (InNR) 3,915,553 (2019 : NA)         3,85,24,66         15,051,57         -         -           10// Claud Fund - Growth Option         3,85,28,466         15,051,57         <	0% Optionally Convertible Debentures in TRIL Urban Transport Private Limited.	1,13,40,400	1 176 00	1 13 40 400	1 242
0% Optionally Convertible Debentures in TRIL Roads Private Limited         37,68,20,000         37,577,26         24,57,00,000         22,565           0.01% Compulsorily Convertible Debentures in TRIL Intopark Limited         25,00,00,000         44,862,00         25,00,00,000         44,862,00         25,00,00,000         44,862,00         25,00,00,000         44,862,00         10,00,000         1,184           0% Optionally Convertible Debentures in TNP Tarms Private Limited         27,80,000         191,60         27,80,000         211.           Quarted Debentures in TNP Tarms Private Limited         27,80,000         1,184         8,140,00         1,184         8,140,00         1,184         8,140,00         8,140           pregate value of quoted investments         gregate tair value.         8,140,00         8,140         9,78,95         10,513,10         10,78,93         10,513,10         10,78,93         10,513,10         10,78,93         10,751,34         4,22,631,34         3,37,895         10,513,10         10,92         10,92 </td <td></td> <td></td> <td></td> <td></td> <td></td>					
0.01% Computedity Convertible Detertures in TRIL Integrate Limited.         25,00,00,000         44,262.00         25,00,00,000         41,273           0% Optionally Convertible Detertures in International Intrabuild Private Limited         10,00,000         1,064.00         10,00,000         1,184           0% Optionally Convertible Detertures in International Intrabuild Private Limited         27,80,000         191.60         27,80,000         111           Duest Detertures in International Intrabuild Private Limited         1,184         8,140,00         1,184         8,140,00         1,184         8,140,00         1,184         8,140,00         1,02,63,51,31         10,02,019         0,02,019         0,02,019         0,02,019         0,02,019         0,02,019         0,02,019         0,02,019         0,02,019         0,02,019	0% Optionally Convertible Depentures in TBIL Boads Private Limited				
0% Optionally Convertible Debentures in HV Parms Private Limited         10,00,000         1,064.00         10,00,000         1,188           0% Optionally Convertible Debentures in International Intrabuild Private Limited         27,80,000         191.80         27,80,000         191.80         27,80,000         111.80           Quoted Debentures (INP setment in joint venture:         18% Redeemable Non-convertible Debentures of INR 687,500 each in TRIL IT4 Private Limited         1,184         8,140,00         1,184         8,140,00         1,184         8,140,00         1,184         8,140,00         9,97,895,95         9,97,895,95         9,97,895,95         9	0.01% Compulsarily Convertible Deheatures in TBU Infonerk Limited				
0% Optionally Convertible Debentures in International Intrabuild Private Limited     27,80,000     191.60     27,80,000     211.       Duoted Debentures fully paid-up:     1) Investment in joint venture:     1,184     8,140,00     1,184     8,140,00     1,184     8,140,00     1,184     8,140,00     1,184     3,37,895     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,513,10     10,51	1% Ontingally Convertible Debestures in HV Forme Brivite Limited				
Quoted Debentures (IIII) paid-up: 1) Investment in joint venture: 18% Redeemable Non-convertible Debentures of INR 687.500 each in TRIL IT4 Private Limited       1,184       8,140.00       1,184       8,140.00         rregate value of quoted investments gregate fair value, gregate value of unquoted investments       8,140.00       8,140.00       8,140.00         urrent Investments Negternet In mutual funds (Fair valued through Profit and Loss): ABSL Liquid Fund - Growth-Direct       31 March 2020       31 March 2019         NAV per unit (in INR) 319,5593 (2019 : NA) HDFC Liquid Fund - Direct Plan - Growth NAV per unit (in INR) 328,9,874.36       15,051.57       -         NAV per unit (in INR) 328,570.04       30,220.05       -       -         NAV per unit (in INR) 328,771.34       10,576.43       211					
1) Investment in joint venture: 18% Redeemable Non-convertible Debentures of INR 687,500 each in TRIL IT4 Private Limited       1,184       8,140,00       1,184       8,140,00         regate value of quoted investments gregate book value gregate fair value.       8,140,00       8,140       8,140,00       8,140         urrent Investments       8,140,00       8,140       0       8,140       0       8,140         urrent Investments       8,140,00       8,140       0       8,140       0       8,140         urrent Investments       4,22,631,34       3,97,895       8,140       0       8,140       0       8,140         urrent Investments       4,22,631,34       3,97,895       10,513,10       -	United Determine Dependences in International Intrabulid Private Limited	27,80,000	191.60	27,80,000	211.
18% Redeemable Non-convertible Debentures of INR 667.500 each in TRIL IT4 Private Limited       1,184       8,140.00       1,184       8,140.00         pregate value of quoted investments       3,140.00       8,140.00       8,140.00       8,140.00         gregate value of quoted investments       8,140.00       8,140.00       8,140.00       8,140.00         gregate value of unquoted investments       8,140.00       8,140.00       8,140.00       8,140.00         urrent investments       4,22,631.34       3,97,895.       3,97,895.         urrent investment in mutual funds (Fair valued through Profit and Loss):       Amount       Quantity       Amount         ASEL Liquid Fund - Growth-Direct       31 March 2020       31 March 2019       Amount       Amount         NAV per unit (in INR) 319.5593 (2019 - NA)       0.01ext Plan - Growth       3.85,284.66       15,051.57       -         NAV per unit (in INR) 32.89,674.36       10,226.570.04       30,220.05       -       -         NAV per unit (in INR) 323.7616 (2019 - NA)       1,02.86,570.04       30,220.05       -       -         NAV per unit (in INR) 323.7616 (2019 - NA)       1,05.76.43       211.       -       -         NAV per unit (in INR) 323.7616 (2019 - NA)       4,91,528,95       15,394,61       10,576,43       211. <td></td> <td></td> <td></td> <td></td> <td></td>					
4,30,771.34         4,06,035           gregate book value         8,140,00         8,140           igregate book value         8,140,00         8,140           igregate tair value,         8,140,00         8,140           igregate tair value,         8,140,00         8,140           igregate tair value,         8,140,00         8,140           igregate value of unquoted investments         4,22,631.34         3,97,895           urrent Investments         4,22,631.34         3,97,895           investment in mutual funds (Fair valued through Profit and Loss):         Amount         Quantity         Amount           ABSL Liquid Fund - Growth-Direct         32,88,874,36         10,513.10         -         -           NAV per unit (in INR) 319,5593 (2019 : NA)         38,52,84,66         15,051.57         -         -           NAV per unit (in INR) 32,06,6111 (2019 : NA)         38,52,84,66         15,051.57         -         -           ICIC I Prudentai Liquid Fund - Direct Plan - Growth         1,02,86,570.04         30,220.05         -         -           NAV per unit (in INR) 32,37916 (2019 : NA)         -         -         -         -         -           NAV per unit (in INR) 32,36,95         10,576,43         211.         -         -					
31 March 2020         31 March 2020         31 March 2019           gregate value of unquoted investments         4,22,631.34         3,97,895           urrent Investments         4,22,631.34         3,97,895           urrent Investment In mutual funds (Fair valued through Profit and Loss):         31 March 2020         31 March 2019           ABSL Liquid Fund - Growth-Direct         32,89,874.36         10,513.10         -           NAV per unit (in INR) 319,5593 (2019 - NA)         3,85,284.66         15,051.57         -           ICICI Prudential Liquid Fund - Direct Plan - Growth         1,02,86,570.04         30,220.05         -           NAV per unit (in INR) 393.7616 (2019 - NA)         1,05,76,43         211.         -	18% Hedeemable Non-convertible Debentures of INR 687,500 each in TRIL IT4 Private Limited	1,184	8,140.00	1,184	8.140.0
gregate book value         8,140,00         8,10,00         8,10,0			4,30,771.34		4,06,035.
State         State <th< td=""><td></td><td></td><td>an an</td><td></td><td></td></th<>			an an		
Stress         Stres         Stres         Stres <td></td> <td></td> <td></td> <td></td> <td></td>					
31 March 2020         31 March 2019           Outstment in mutual funds (Fair valued through Profit and Loss):         Quantity         Amount         Quantity         Amount           ABSL Liquid Fund - Growth-Direct         32.89,874.36         10,513.10         -         -           NAV per unit (in INR) 319,5593 (2019 - NA)         32.89,874.36         10,513.10         -         -           HDFC Liquid Fund - Growth Option         3.85,284.66         15,051.57         -         -           NAV per unit (in INR) 390.55111 (2019 - NA)         1.02.86,570.04         30,220.05         -         -           NAV per unit (in INR) 293.7616 (2019 - NA)         1.02.86,570.04         30,220.05         -         -           NAV per unit (in INR) 293.7616 (2019 - NA)         4.91,528,95         15,394,61         10.576,43         211.	Igregate fair value,		8,140.00		8,140.
urrent Investments 31 March 2020 31 March 2019  Ouantity Amount Ouantity Amount Ouantity Amount Amount Amount 32,89,874.36 10,513.10	regate value of unquoted investments		4.22.631.34		3 97 895
Quantity         Amount         Quantity         Amount           ABSL Liquid Fund - Growth-Direct         32,89,874,36         10,513,10         -           NAV per unit (in INR) 319,5593 (2019 - NA)         105C Liquid Fund - Oirect Plan - Growth Option         3,85,284,66         15,051,57         -           NAV per unit (in INR) 3,906,6111 (2019 - NA)         3,85,284,66         15,051,57         -         -           NAV per unit (in INR) 3,906,6111 (2019 - NA)         1,02,86,570,04         30,220,05         -         -           NAV per unit (in INR) 23,915 (2019 - NA)         10,576,43         211,         -         -         -           TATA liquid fund - Direct Plan - Growth         4,91,528,95         15,394,61         10,576,43         211,	a a george a series a de l'Alles de la California a				
Quantity         Amount         Quantity         Amount           ABSL Liquid Fund - Growth-Direct         32,89,874,36         10,513,10         -           NAV per unit (in INR) 319,5593 (2019 · NA)         BDFC Liquid Fund - Oirect Plan - Growth Option         3,85,284,66         15,051,57         -           NAV per unit (in INR) 3,906.6111 (2019 · NA)         0.6111 (2019 · NA)         1,02 86,570.04         30,220.05         -           NAV per unit (in INR) 29,37616 (2019 · NA)         1,02 86,570.04         30,220.05         -         -           TATA liquid fund - Direct Plan - Growth         4,91,528,95         15,394,61         10,576,43         211.	urrent Investments	31 March 2	2020	31 March	2019
ABSL Liquid Fund - Growth-Direct         32,89,874.36         10,513.10         -           NAV per unit (in INR) 319,5593 (2019 - NA)         -         -         -         -           DPC Liquid Fund - Oirect Plan - Growth Option         3,85,284.66         15,051.57         -         -           NAV per unit (in INR) 3,906.6111 (2019 - NA)         1.02.86,570.04         30,220.05         -         -           NAV per unit (in INR) 93.7816 (2019 - NA)         1.02.86,570.04         30,220.05         -         -           TATA liquid fund - Direct Plan - Growth         1.0.576,43         211.         -         -	nvestment in mutual funds (Fair valued through Profit and Loss):				
NAV per unit (in INR) 319,5593 (2019: NA)         3,85,284.66         15,051.57           HDFC Liquid Fund - Direct Plan - Growth Option         3,85,284.66         15,051.57           NAV per unit (in INR) 3,906.5111 (2019: NA)         1.02.86,570.04         30,220.05           ICICI Prudential Liquid Fund - Direct Plan - Growth         1.02.86,570.04         30,220.05           NAV per unit (in INR) 293.7616 (2019: NA)         10.576,43         211.				Guaraty	Andun
HDFC Liquid Fund - Direct Plan - Growth Óption         3.85,284.66         15,051.57           NAV per unit (in INR) 3,906.6111 (2019: NA)         1.02.86,570.04         30,220.05           ICICI Prudential Liquid Fund - Direct Plan - Growth         1.02.86,570.04         30,220.05           NAV per unit (in INR) 23,7816 (2019: NA)         1.05.76,43         211.           TATA liquid fund - Direct Plan - Growth         4.91,528,95         15,394.61         10,576.43         211.		52,03,074,30	10,010,10		
NAV per unit. (in INR) 3,906.6111 (2019: NA)         1,02 86,570.04         30,220.05           ICICI Prudential Liquid Fund - Direct Plan - Growth         1,02 86,570.04         30,220.05           NAV per unit. (in INR) 239.7816 (2019: NA)         10,576.43         211.           TATA liquid fund - Direct Plan - Growth         4,91,528.95         15,394.61         10,576.43         211.		3.05 304 00			
ICICI Prudential Liquid Fund - Direct Plan - Growth         1,02,86,570,04         30,220,05           NAV per unit (in INR) 293.75016 (2019 - NA)         TATA fliquid fund - Direct Plan - Growth         4,91,528,95         15,394,61         10,576,43         211.		3,85,284.66	15,051.57		τ.
NAV per unit (in INP) 293.7816 (2019 : NA) TATA liquid fund - Direct Plan - Growth 4.91,528.95 15,394.61 10.576.43 311.					
TATA liquid fund - Direct Plan - Growth 4,91,528,95 15,394.61 10,576,43 311.		1.02 86,570.04	30,220.05		
		Cale Cale Com-	Ast Sec		
	TATA liquid fund - Direct Plan - Growth NAV per unit (in INR) 3,131,9848 (2019 ( 2,944,44)	4,91,528.95	15,394.61	10,576.43	311.4
			71 170 33		211

Foot note: 1, 50,000,000 (2019: 50,000,000) equity shares have been pledged with Tamilnadu Industrial Development Corporation Limited for a period upto 31 March 2021, \* The Company has provided Non Disposal Undertakings to the lenders / Investors of its subsidiaries and joint ventures for the minimum shareholding that the Company needs to maintain until the final settlement date of the loan.

71,179.33

311.42

#### 6 Financial Assets - Trade and other receivables

Particulars	31 Marc	31 March 2019		
	Non Current	Current	Non Current	Current
rade receivables from related parties				
Unsecured, considered good				
Outstanding for a period exceeding six months		445.51	10 C	660.21
Outstanding for a period less than six months		1,274,91	-	670.66
Unsecured, considered doutful				
Outstanding for a period exceeding six months		306.00		
Provision for credit impaired Trade Receivables	-	(306.00)		
rade receivables from others		descently.		
Unsecured, considered good				
Outstanding for a period exceeding six months			-	
Outstanding for a period less than six months		158.00		66.94
Total		1,878,42		1.397.81

#### 7 Financial Assets: Loans and Advances

Particulars	31 Marc	31 March 2019			
	Non Current	Current	Non Current	Current	
(Unsecured, considered good)					
Advances recoverable from related parties		1,029,21		2,015.66	
Inter corporate deposits to related parties	39,709.19		52,124,48		
Inter corporate deposits to other parties		675.00	-		
Security deposits		515.33	-	38.41	
Other advances	1	244.97		295.70	
(Unsecured, considered doubtful)					
Advances recoverable from related parties	14	35.00		35.00	
Inter corporate deposits to related parties	1 H I	95.00		95.00	
Less : Provisions for credit impaired advances and ICDs		(130.00)		(130.00	
Inter corporate deposits to other parties		516.62	-		
Less : Provisions for credit impaired ICDs		(516.62)			
Total	39,709.19	2,464.51	52,124.48	2.349.77	

#### 8 Financial Assets: Others

Particulars	31 Marc	31 March 2019		
	Non Current	Current	Non Current	Current
Interest Accrued Receivable:				
Considered Good		1,428.91	4.1	2.861.15
Considered Doubtful	÷	10.00		10.00
Less : Provision for Doubtful Interest		(10.00)		(10.00
Fixed deposits with more than 12 months maturity (Refer Foot Note)	0.75		0.75	
Total	0.75	1,428,91	0.75	2,861,19

Foot Note: Bank deposit of INR 0.75 lakhs (2019: INR 0.75 lakhs) is having lien in favour of Commercial Tax Officer, KVAT Works Contract Ernakulam.

#### 9 Current Tax Assets (net)

Particulars	31 March 2020	31 March 2019	
Advance Payment of taxes	11,329,40	10,407.62	
Provision for tax	(4.394.72)	(4.394.72)	
Total	6,934.68	6,012.90	

#### 10 Other Assets

Particulars	31 March 2020		31 March 2019	
	Non Current	Current	Non Current	Current
Capital Advances	159.91	-	159.91	
Call Option Premium (Refer Foot note)	6,299,32		5,725,23	
Derivative asset (Refer Foot note)	6,318.00		5.079.00	
Balance with Tax Authority		5.62		1.008.66
Prepaid Expenses	2.75	132.74	4,32	70.54
Other Receivables		292.95		7.09
Total	12,779.98	431.31	10,968.46	1,086.29

Foot note: The Company had paid an interest free advance of INR 7,110/- lakhs to Indian Hotels Company Limited (IHCL) vide MOU dated 23 February. 2010 and MOU dated 11 July. 2011. The consideration for the advance is with an option to acquire the equity investment of TRIL Intopark Limited for an amount of INR 7,110/- lakhs. The fair value of these shares are disclosed above as Call option premium and Derivative Asset appropriately. The shares will be transferred on or before 10 July. 2021.

#### 11 Inventories (Lower of cost and net realisable value)

Particulars	31 March 2020	31 March 2019
Bought out construction materials	52.15	52.15
Work In Progress		12,710,70
Finished Goods (Refer Foot note)	30,905.18	28.248.21
Total	30,957.33	41.011.06

Foot note: Represent value of residential units. Net off NRV provision on inventories INR 1,420.78 Lakhs (2019: NIL)

#### 12 Cash and cash equivalents

	Particulars	31 Mar	ch 2020	31 Marc	h 2019
	Gash on Hand		0.06		
	Balances with Banks - In current accounts		970.24		445.
	- in deposit accounts with less than or equal to 3 months original maturity Total		24,610.00 25,580.30		445.
Oth	er Bank Balances				
	Particulars	31 Mar	ch 2020	31 Marc	
	Deposit Accounts with less than 12 months maturity Total		419.25 419.25		444.
	Total		419.25	-	444.
	ity Share Capital Authorised, Issued, Subscribed and Fully Paid up :				
	Particulars	31 Mare No of Shares	ch 2020 Amount	31 Marc No of Shares	h 2019 Amount
	Authorised Capital : 8.00.00.00.000 (2019: 3.00.00.00,000) equity shares of INR 10 each	8,00,00,00,000	8,00,000.00	3.00,00,00,000	3,00,000.
	Issued, Subscribed and Fully Paid up Capital :				
	1.01,73,07,692 (2018: 1.01,73,07,692) equity shares of INR 10 each	1,01,73,07,692	1,01,730.77	1,01,73,07,692	1,01,730.
	Total	1,01,73,07,692	1,01,730.77	1,01,73,07,692	1,01,730.
(b)	Reconciliation of Number of Shares Outstanding Particulars	24.14-	L 0000	04 19 cm	- 0040
	Faturulars	31 Marc No of Shares	Amount	31 Marci No of Shares	h 2019 Amount
	As at the beginning of the year	1,01,73,07,692	1,01,730.77	1,01,73,07,692	1.01,730.
	Add: Issued during the year As at the end of the year	1 01 72 07 000	1 04 700 77	1.01.73.07.692	1.04 700
	As at the end of the year	1,01,73,07,692	1,01,730.77	1,01,73,07,692	1,01,730.
(d)	liquidation of the Company, the holders of the equity shares will be entitled to recent distribution will be in proportion to the number of equity shares held by the shareholders. Shares of the company held by the Holding company				
	Name of Shareholder	31 Marc No of Shares	h 2020 Amount	31 March No of Shares	h 2019 Amount
	Equity shares of INR 10 each, fully paid-up	and the		and an and the	
	Tata Sons Private Limited	1,01,73,07,692	1,01,730,77	1,01,73,07,692	1,01,730.
	Details of shareholding more than 5% in the Company				
		31 Marc	h 2020	31 March	h 2019
(e)	Details of shareholding more than 5% in the Company				
(e)	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited	31 Marc No of Shares	h 2020 % Holding	31 March No of Shares	h 2019 % Holding
(e)	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited r Equity	31 Marc No of Shares 1.01,73.07,692	h 2020 % Holding 100%	31 Marct No of Shares 1,01.73.07,692	h 2019 % Holding 100
(e) Othe	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited r Equity Particulars	31 Marc No of Shares	h 2020 % Holding 100% h 2020	31 March No of Shares	h 2019 % Holding 100
(e) Othe	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited r Equity Particulars Share application money pending allotment	31 Marc No of Shares 1.01,73.07,692	h 2020 % Holding 100%	31 Marct No of Shares 1,01.73.07,692	h 2019 % Holding 100
(e) Othe	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited r Equity Particulars Share application money pending allotment Reserves and surplus	31 Marc No of Shares 1.01,73.07,692	h 2020 % Holding 100% h 2020 1.20,000.00	31 Marct No of Shares 1,01.73.07,692	n 2019 % Holding 100 n 2019
(e) Othe	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited r r Equity Particulars Share application money pending allotment Reserves and surplus Securities Premium reserve Capital reserve	31 Marc No of Shares 1.01,73.07,692	h 2020 % Holding 100% h 2020 1,20,000.00 15,769.23 4,783.49	31 Marct No of Shares 1,01.73.07,692	h 2019 % Holding 104 h 2019 15,769.3 4,783.4
(e) Othe	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited r Equity Particulars Share application money pending allotment Reserves and surplus Securities Premium reserve	31 Marc No of Shares 1.01,73.07,692	h 2020 % Holding 100% h 2020 1.20,000.00 15,769.23	31 Marct No of Shares 1,01.73.07,692	h 2019 % Holding 100 h 2019 15,769,2 4,783,4
(e) Othe	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited r Equity Particulars Share application money pending allotment Reserves and surplus Securities Premium reserve Capital reserve Retained earnings Items of Other comprehensive income	31 Marc No of Shares 1.01,73.07,692	h 2020 % Holding 100% h 2020 1.20,000.00 15,769.23 4,783.49 (73,552.05)	31 Marct No of Shares 1,01.73.07,692	n 2019 % Holding 100 n 2019 15,769,2 (37,659,4
(e) Othe	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited requity Particulars Share application money pending allotment Reserves and surplus Securities Premium reserve Capital reserve Retained earnings Items of Other comprehensive income FVOCI - equity instruments	31 Marc No of Shares 1.01,73.07,692	h 2020 % Holding 100% h 2020 1.20,000.00 15,769,23 4,783,49 (73,552.05) 1,24,687,80	31 Marct No of Shares 1,01.73.07,692	15,769.2 (37,659.4 (37,659.4 (37,659.4 (1,14,041.7
(e) Othe	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited r Equity Particulars Share application money pending allotment Reserves and surplus Securities Premium reserve Capital reserve Retained earnings Items of Other comprehensive income	31 Marc No of Shares 1.01,73.07,692	h 2020 % Holding 100% h 2020 1.20,000.00 15,769.23 4,783.49 (73,552.05)	31 Marct No of Shares 1,01.73.07,692	h 2019 % Holding 100 h 2019 15,769,2 4,783,4 (37,659,4 1.14,041,7 20,1
(e) Othe	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited r Equity Particulars Share application money pending allotment Reserves and surplus Securities Premium reserve Capital reserve Retained earnings Items of Other comprehensive income FVOCI - equity instruments Defined benefit plan adjustment TOTAL	31 Marc No of Shares 1.01,73.07,692	h 2020 % Holding 100% h 2020 1,20,000.00 15,769.23 4,783.49 (73,552.05) 1,24,687.80 7.38	31 Marct No of Shares 1,01.73.07,692	h 2019 % Holding 100 h 2019 15,769,2 4,783,4 (37,659,4 1.14,041,7 20,1
(e) Othe	Details of shareholding more than 5% in the Company           Name of Shareholder           Equity shares of INR 10 each, fully paid-up           Tata Sons Private Limited           r Equity           Particulars           Share application money pending allotment           Reserves and surplus           Securities Premium reserve           Capital reserve           Ratined earnings           Items of Other comprehensive income           PVCCI - equity instruments           Defined benefit plan adjustment           TOTAL           Share application money pending allotment	31 Marc No of Shares 1.01,73.07,692	h 2020 % Holding 100% h 2020 1.20,000.00 15,769.23 4,783.49 (73,552.05) 1.24,687.80 7.38 1,91,695.85	31 Marct No of Shares 1,01.73.07,692	h 2019 % Holding 100 1 2019 15,769,2 4,783,4 (37,659,4 1,14,041,7 20,1 96,955,1
(e) Othe	Details of shareholding more than 5% in the Company Name of Shareholder Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited  r Equity Particulars Share application money pending allotment Reserves and surplus Securities Premium reserve Capital reserve Retained earnings tems of Other comprehensive income FVOCI - equity instruments Defined benefit plan adjustment TOTAL Share application money pending allotment Particulars Balance at the beginning of the year	31 Marc No of Shares 1,01,73,07,692 31 Marc	h 2020 % Holding 100% h 2020 1,20,000.00 15,769.23 4,783.49 (73,552.05) 1,24,687.80 7,38 1,91,695.85 h 2020	31 Marct No of Shares 1,01,73,07,692 31 March	h 2019 % Holding 100 1 2019 15,769,2 4,783,4 (37,659,4 1,14,041,7 20,1 96,955,1
(e) Othe	Details of shareholding more than 5% in the Company           Name of Shareholder           Equity shares of INR 10 each, fully paid-up           Tata Sons Private Limited           r Equity           Particulars           Share application money pending allotment           Reserves and surplus           Securities Premium reserve           Capital reserve           Ratined earnings           Items of Other comprehensive income           PVCCI - equity instruments           Defined benefit plan adjustment           TOTAL           Share application money pending allotment	31 Marc No of Shares 1,01,73,07,692 31 Marc	h 2020 % Holding 100% h 2020 1.20,000.00 15,769.23 4,783.49 (73,552.05) 1.24,687.80 7.38 1,91,695.85	31 Marct No of Shares 1,01,73,07,692 31 March	h 2019 % Holding 100 1 2019 15,769,2 4,783,4 (37,659,4 1,14,041,7 20,1 96,955,1
(e) Othe	Details of shareholding more than 5% in the Company         Name of Shareholder         Equity shares of INR 10 each, fully paid-up         Tata Sons Private Limited         r Equity         Particulars         Share application money pending allotment         Reserves and surplus         Securities Premium reserve         Capital reserve         Retained earnings         Items of Other comprehensive income         FVOCI - equity instruments         Defined benefit plan adjustment         TOTAL         Share application money pending allotment         Particulars         Balance at the beginning of the year         Add: received during the year         Balance at the end of the Year         Securities premium reserve	31 Marc No of Shares 1.01,73,07,692 31 Marc 31 Marc	h 2020 % Holding 100% h 2020 1.20,000.00 15,769.23 4,783.49 (73,552.05) 1,24,687,80 7.38 1,91,695.85 h 2020 1,20,000.00 1,20,000.00	31 March No of Shares 1,01,73,07,692 31 March 31 March	10019 % Holding 10019 12019 15,769.2 4,783.4 (37,659.4 1.14,041.7 20.1 96,955.1 96,955.1 -
(e) Othe	Details of shareholding more than 5% in the Company         Name of Shareholder         Equity shares of INR 10 each, fully paid-up         Tata Sons Private Limited         r Equity         Particulars         Share application money pending allotment         Reserves and surplus         Securities Premium reserve         Capital reserve         Retined earnings         Items of Other comprehensive income         FVCCI - equity instruments         Defined benefit plan adjustment         TOTAL         Share application money pending allotment         Particulars         Balance at the beginning of the year         Add: received during the year         Balance at the dot the Year         Securities premium reserve         Particulars	31 Marc No of Shares 1,01,73,07,692 31 Marc	h 2020 % Holding 100% h 2020 1.20,000.00 15,769,23 4,783,49 (73,552,05) 1,24,687,80 7.38 1,91,695,85 h 2020 1,20,000.00 1,20,000.00	31 Marct No of Shares 1,01,73,07,692 31 March	h 2019 % Holding 100 100 100 100 100 100 100 100 100 10
(e) Othe	Details of shareholding more than 5% in the Company         Name of Shareholder         Equity shares of INR 10 each, fully paid-up         Tata Sons Private Limited         r Equity         Particulars         Share application money pending allotment         Reserves and surplus         Securities Premium reserve         Capital reserve         Retained earnings         Items of Other comprehensive income         FVOCI - equity instruments         Defined benefit plan adjustment         TOTAL         Share application money pending allotment         Particulars         Balance at the beginning of the year         Add: received during the year         Balance at the end of the Year         Securities premium reserve	31 Marc No of Shares 1.01,73,07,692 31 Marc 31 Marc	h 2020 % Holding 100% h 2020 1.20,000.00 15,769.23 4,783.49 (73,552.05) 1,24,687,80 7.38 1,91,695.85 h 2020 1,20,000.00 1,20,000.00	31 March No of Shares 1,01,73,07,692 31 March 31 March	h 2019 % Holding 100 100 100 100 100 100 100 100 100 10
(e) Othe	Details of shareholding more than 5% in the Company         Name of Shareholder         Equity shares of INR 10 each, fully paid-up         Tata Sons Private Limited         r Equity         Particulars         Share application money pending allotment         Reserves and surplus         Securities Premium reserve         Capital reserve         Retained earnings         Items of Other comprehensive income         FVOCI - equity instruments         Defined benefit plan adjustment         TOTAL         Share application money pending allotment         Particulars         Share application money pending allotment         Particulars         Balance at the end of the year         Add: reserved during the year         Balance         Securities premium reserve         Particulars         Balance         Balance at the end of the Year	31 Marc No of Shares 1,01,73,07,692 31 Marc 31 Marc 31 Marc	h 2020 % Holding 100% h 2020 1.20,000.00 15,769,23 4,783,49 (73,552,05) 1,24,687,80 7,38 1,91,695,85 h 2020 1,20,000.00 1,20,000.00 1,20,000.00 1,20,000.00 1,20,000.00 1,5,769,23	31 March	15,769.2 10( 10( 10( 10( 10( 10( 10( 10(
(e) Othe	Details of shareholding more than 5% in the Company         Name of Shareholder         Equity shares of INR 10 each, fully paid-up         Tata Sons Private Limited         r Equity         Particulars         Share application money pending allotment         Reserves and surplus         Securities Premium reserve         Capital reserve         Ratined earnings         Items of Other comprehensive income         FVCC1 - equity instruments         Defined benefit plan adjustment         TOTAL         Share application money pending allotment         Particulars         Balance at the baginning of the year         Add. reserve         Particulars         Balance at the end of the Year         Securities premium reserve         Particulars         Opening balance         Balance at the end of the Year         Capital reserve         Particulars         Opening balance         Balance at the end of the Year         Capital reserve         Particulars	31 Marc No of Shares 1.01,73,07,692 31 Marc 31 Marc	h 2020 % Holding 100% h 2020 1.20,000.00 15,769.23 4,783.49 (73,552.05) 1,24,687.80 7.38 1,91,695.85 h 2020 1,20,000.00 1,20,000.00 1,20,000.00 1,20,000.00 1,20,000.00 1,20,000.00 1,20,000.00 1,20,000.00	31 March No of Shares 1,01,73,07,692 31 March 31 March	12019 % Holding 10( 12019 15,769,2 15,769,2 1,14,041,7 20,19 - - - - - - - - - - - - - - - - - - -
(e) Othe	Details of shareholding more than 5% in the Company         Name of Shareholder         Equity shares of INR 10 each, fully paid-up         Tata Sons Private Limited         r Equity         Particulars         Share application money pending allotment         Reserves and surplus         Securities Premium reserve         Capital reserve         Retained earnings         Items of Other comprehensive income         FVOCI - equity instruments         Defined benefit plan adjustment         TOTAL         Share application money pending allotment         Particulars         Share application money pending allotment         Particulars         Balance at the end of the year         Add: reserved during the year         Balance         Securities premium reserve         Particulars         Balance         Balance at the end of the Year	31 Marc No of Shares 1,01,73,07,692 31 Marc 31 Marc 31 Marc	h 2020 % Holding 100% h 2020 1.20,000.00 15,769,23 4,783,49 (73,552,05) 1,24,687,80 7,38 1,91,695,85 h 2020 1,20,000.00 1,20,000.00 1,20,000.00 1,20,000.00 1,20,000.00 1,5,769,23	31 March	15,769.2 10( 10( 10( 10( 10( 10( 10( 10(

## Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements

for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

Particulars	31 March 2020	31 March 2019
Balance at the beginning of the year	(37,659.45)	(17.728.54)
IND AS 115 Adjustment	(	(1.304.14)
Transferred from OCI to Retained eqrnings on derecognition of equity instruments	(13,307,84)	11100 1111
Add: (Loss) for the year	(22,584,76)	(18,626,77
Balance at the end of the Year	(73,552.05)	(37,659.45)
Equity Instruments through Other Comprehensive Income (OCI)		
Particulars	31 March 2020	31 March 2019
Opening balance	1,14,041.72	85,210.00
Equity instruments fair value through OCI (FVOCI)	835.21	32,334,60
Income tax relating to items that will not be reclassified to profit or loss	(3,496.97)	(3,502,88)
Transferred from OCI to Retained eqrnings on derecogniton of equity instruments	13,307,84	
Balance at the end of the Year	1,24,687.80	1,14,041.72
Other comprehensive income - Defined benefit plan adjustment		
Particulars	31 March 2020	31 March 2019
Opening balance	20.19	(18.78)
Remeasurements of defined benefit (asset) / liability	(17.31)	52.65
Income tax relating to items that will not be reclassified to profit or loss	4.50	(13.68)
Balance at the end of the Year	7.38	20.19

#### Nature and purpose of the reserve

Securities premium reserve 0.01% Compulsorily convertible debentures were compulsorily convertible into equity shares by 25 August 2016 or before at the option of investor. During the Financial year 2016-17 (on 24 August 2016), these debentures were converted into 192,307,692 Equity shares of INR 10 each at a premium of INR 3 each.

Capital reserve Capital reserve was created to record excess of net assets taken over pursuant to scheme of merger sanctioned by the Bombay High Court in the year 2015-16 between Tata Realty and Infrastructure Limited, Mara Builder Private Limited and TRIF Real Estate and Development Limited.

Debenture redemption reserve The Company has not created debenture redemption reserve as per Section 71 of the Companies Act, 2013 due to losses incurred post issuance of debentures.

Equity Instruments through Other Comprehensive Income The Company has elected to recognise changes in the fair value, of investments in equity and preference securities of subsidiaries in other comprehensive income. These changes are accumulated within the FVOCI equity investments reserve within equity. The Company transfers amounts from this reserve to retained earnings when the relevant equity securities are derecognised.

#### 16 Financial Liabilities - Borrowings

Particulars	31 Marc	h 2020	31 March	2019
	Non Current	Current	Non Current	Current
Von Convertible Debentures - Unsecured, Unlisted 17,950 (2019: 18,250) 8,25% - 9.50% Non convertible debentures (NCD) @ INR 10 Lacs each, fully paid up (Refer Foot Note No. 1 below)	1,39,500.00		72,500.00	~
Less : Unexpired Issuance costs Bank Overdraft	(80.96)		(39.85)	4,117,86
Commercial Papers from Mutual funds (Refer Foot Note No. 2 below) Less : Unexpired discount		78,000.00 (2,696,79)		77,500.00
Inter Corporate Deposits (Refer Foot Note No. 3 below)	2	1-1-1-1-1	2	14,000,00
Short Term Loan from Bank		20.000.00		
TOTAL	1,39,419.04	95,303.21	72,460.15	94,685.15
The above amount includes				
Secured Barrowings				

#### Unsecured Borrowings 1,39,419.04 94,685.15 72,460,15 95,303,21 Foot Note: 1) Terms of repayment and Interest of Unsecured Non Convertible Debentures:

Particulars	31 March 2020		31 March	1 2019
	Non Current	Current	Non Current	Current
Yes Bank LtdInterest @ 9.25 % payable annually. Principal on Bullet repayment on maturity, on 23-May-2019.				35,000.00
Yes Bank LtdInterest @ 9.25 % payable annually. Principal on Bullet repayment on maturity, on 23-July-2019.	14			25,000.00
Kotak Mahindra Bank -Interest @ 9.10 % payable annually. Principal on Bullet repayment on maturity, on 25-June-2019.	e	د ا		24.000.00
Kotak Mahindra Bank -Interest @ 9.10 % payable annually. Principal on Bullet repayment on maturity, on 23-Aug-2019.	-	1	4	26,000.00
Kotak Mahindra Bank -Interest @ 8.25 % payable on maturity, Principal on Bullet repayment on maturity, on 20-Apr-2020.		10,000.00	10.000.00	1
Kotak Mahindra Bank -Interest @ 8.25 % payable on maturity. Principal on Bullet repayment on maturity, on 17-Aug-2020.		10,000.00	10,000.00	
Kotak Mahindra Bank -Interest @ 8.25 % payable on maturity. Principal on Bullet repayment on maturity, on 20-May-2020.		20,000.00	20,000.00	
Kotak Mahindra Bank -Interest @ 8.57 % payable on maturity. Principal on Bullet repayment on maturity, on 20-Apr-2021.	32,500.00		32,500.00	
Kotak Mahindra Bank -Interest @ 9.50 % payable on maturity. Principal on Bullet repayment on maturity, on 04-Jun-2021.	40.000.00		÷	
CICI Bank LtdInterest @ 9 % payable annually. Principal on Bullet repayment on maturity, on 8-November-2022.	19,500.00			
DBI trusteeship Services LtdInterest @ 8.68 % payable annually. Principal on Bullet epayment on maturity, on 29-April-2022.	20,000.00			
DBI trusteeship Services LtdInterest @ 8.40 % payable annually. Principal on Bullet epayment on maturity, on 06-June-2022.	27,500.00	4		
TOTAL	1,39,500.00	40,000.00	72,500.00	1,10,000.00

2) Commercial paper issued to mutual funds are at a discount rate ranging from 6.10% -9.00% per annum (2019: 7.20% - 9.10 % per annum), and the same are repayable within one year at the agreed upon full face value.

3) Inter Corporate Deposit is obtained from a group company at interest rate ranging from 7.95% to 9% per annum (2019: 9% per annum) and the same is repayable within 90 days.

#### 17 Financial Liabilities - Trade and other payables

Particulars	31 Marc	31 March 2019		
	Non Current	Current	Non Current	Current
Trade Payables			and the second second	
Micro and Small Enterprises (Refer Foot Note)				
Other than Micro and Small Enterprises	÷	3,185.65		3,458.03
Total		3,185.65		3,458.03

Based on information received by the Company from its vendors, the amount of principal outstanding in respect of MSME as at Balance Sheet date covered under the Micro, Small and Medium Enterprises Development Act, 2006 is INR Nil. There were no delays in the payment of dues to Micro and Small Enterprises.

#### 18 Financial Liabilities - Others

Particulars	31 Marcl	31 March 2019		
	Non Current	Current	Non Current	Current
Interest accrued but not due on borrowings	5,932.57	17,007.63	11,425.58	7,731.74
Current Maturity of Unsecured Long term borrowings	-	40,000.00	-	1,10.000.00
Less : Unexpired issuance costs		(5.86)	-	(21.44
Derivatives - Put option (Refer Foot Note)		7,121,00	-	7,122.00
Creditors for Capital Goods				259.90
Lease Liabilities	138.52	9.76		
Dat Nata-	6,071.09	64,132.53	11,425.58	1,25,092.20

Foot Note: Agreement of the Company with Tamil Nadu Industrial Development Corporation (TIDCO) dated 24 March 2008 and supplementary agreements / arrangements entered into between the parties, TIDCO has an option, exercisable until 31 March 2021, to sell its investments in TRIL Infopark Limited, comprising 5,00,00,000 equity shares of INR10 each, representing 6,67% holding in TRIL Infopark Limited, to the Company. The consideration is to be computed at an agreed IRR, on the basis of which the consideration, as at 31 March 2020, Is INR 16,640 lakhs. As a security for the above transaction, the Company has pledged its investment in TRIL Infopark Limited with TIDCO, (5,00,00,000 equity shares of INR 10 each, fully paid) and also placed post-dated cheque of INR 18,640/- lakhs.

#### 19 Provisions

Particulars	31 Marc	31 March 2019		
	Non-Current	Current	Non-Current	Current
Provision for Employee Benefits:				
Gratuity	312.48	68.81	292.38	75.55
Compensated absences	314.05	115.75	393.10	149.58
Total	626.53	184.56	685.48	225.13

#### 20 Current tax liabilities (net)

Particulars	31 March 2020	31 March 2019	
Provision for taxation	8,965.86	8.965.86	
Advance Payment of taxes	(7,213.98)	(7,213,98)	
Total	1,751.88	1,751.88	

Particulars	As at 31 March 2019	Movement Recognised in Statement of Profit and Loss	Movement Recognised in Other comprehensive income	Movement Others (Footnote 1)	As at 31 March 2020
Deferred Tax Assets				-	
Property, plant and equipment and intangible assets	790.51	(529.04)	-	~	261.4
Fair valuation of derivatives at FVTPL	531.19	(322.40)	36.1		208.7
Defined benefit obligation	236.76	(29,39)	4.50		211.8
Deemed Investment on ICD discounting (Refer Foot Note No.1)	1,411.77	(915.00)	(437,25)	1.1	59.5
Deferred Tax Liabilities					
Fair valuations of Equity investments at FVOCI (Refer Foot Note No.2)	(14,489.71)	-	(3,059,72)		(17,549,4
Fair valuations of other financial assets at FVTPL	(1,738.90)	(621.68)			(2,360.5
Total	(13,258,38)	(2,417,51)	(3,492,47)	· · ·	(19,168.3

Particulars	As at 31 March 2018	Movement Recognised in Statement of Profit and Loss	Movement Recognised in Other comprehensive income	Movement Others (Footnote 1)	As at 31 March 2019
Deferred Tax Assets					
Property, plant and equipment and intangible assets	1.410.43	(619.92)	- L.	-	790.51
Fair valuation of derivatives at FVTPL	881.76	(350,57)			531.19
Defined benefit obligation	276.38	(25,94)	(13.68)		236,76
Deemed Investment on ICD discounting		(615,40)	-	2.027.17	1.411.77
Deferred Tax Liabilities			1		
Fair valuations of Equity investments at FVOCI	(10,986.83)		(3,502.88)		(14,489,71)
Fair valuations of other financial assets at FVTPL	(2,051.86)	312.96			(1.738.90)
Total	(10,470.12)	(1,298.87)	(3,516.56)	2,027.17	(13,258,38)

Foot Note:

1. Movement in Deamed Equity Investments (Note 5). 2. The gain of INR 835 lakhs arising on Equity Investments fair valued through OCI is net of loss of INR 11,767 lakhs. Deferred Tax Liability of INR 1,758 lakhs has been recognised on the gross gain of INR 12,753 lakhs and INR 1,302 lakhs of reversal of Deferred Tax Asset in respect of equity investments sold.

#### 22 Other Current Liabilities

Particulars	31 March 2020	31 March 2019
Advances from customers	1,863.18	3,939.16
Statutory dues including provident fund and tax deducted at source	285.42	82.02
Compensation on delayed possession payable	10.88	10.88
Corpus fund collection	780.65	676.09
Advance maintenance charges		95.06
Security deposits from customers	43.49	63.70
Other Payable	101.14	252.57
Total	3,084.76	5,119.48

#### 23 Revenue from Operations

Particulars	For the year 31 March 2		For the year 31 March 2	
Sale of residential flats		13,050.27		9,463.27
Sale of services				
Project management consultancy fees	2,857,43		3.281.49	
Asset management fees	1,168.64		1,120.39	
Maintenance and other receipts	508.20	4,534.27	628.14	5,030.02
Total		17,584.54		14,493.29

#### 24 Other Income

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Interest Income on:		
Inter corporate deposits / non convertible debentures"	6,608.51	5,441.88
Income-tax refund		67.40
Fixed deposits with bank	171.20	95.64
Unwinding of call option premium	574.09	540.44
Profit on sale of Current Investments	818.52	263.43
Gain on fair valuations of investments	5,062.30	4,680,48
Gain on fair value changes		
- on put options	1.00	10 A
- on call options	1,239.00	1,425.00
Mark to Market gain on current investment in Mutual funds	435.23	20.02
Other income from residential projects	50.20	161.62
Profit on Sale of Property Plant & Equipment and intangible assets	0.33	
Miscellaneous Income	5.70	58.04
Total	14,966.08	12,753.95

\* Includes Unwinding of interest amounting to INR 3,519.24 lakns (2019: 2,366.93 lakhs)

#### 25 Cost of sale of flats

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Material consumed		
Opening Stock of construction material	52.15	256.31
Less . Closing inventorles of construction material(Refer Note 11)	(52.15)	(52.15
Total cost of materials consumed		204.16
Expenditure during the year		
Opening Stock of Inventories	40,958.91	41,485.22
Add: Reversal of cost of IND AS 115		1.870.38
Addition during the year		
Professional fees and technical fees	31.04	10.00
Project management consultancy charges	37.68	37.26
Approval and permission expenses	25.97	70.09
Construction cost	1,950.13	3,318.50
Other expenses	19.55	86.21
	43,023.28	46,877.66
	43,023.28	47,081.82
Less: Closing Stock of Inventories (Refer Note 11)	(30,905.18)	(40,958.91
Total	12,118.10	6,122.91

#### 26 Employee benefits expense

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Salaries, wages and bonus	6,193.06	5,868.99
Less: Deputation charges recovered	(993.65)	(837.43)
Gratuity charges and Contributions to Provident and pension funds	152.58	222.41
Staff welfare expenses	150.25	252.97
Compensated absences	(32.28)	78.45
Total	5,469.96	5,585.39

#### 27 Finance Cost

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Interest Expense		
- on NCD from banks and financial intitution	14,050.68	16,594,33
- on bank overdraft	141.96	33.48
- on commercial paper	11,249.66	6,140.06
- on term loan from bank	1,547.47	
- on ICD taken	1,001.79	20.71
- on Lease liabilities	3,58	
Finance charges	501.03	200.33
Total	28,496.17	22,988.91

#### 28 Depreciation & Amortisation

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Depreciation on Property plant & equipment	172.77	126.48
Amortisation of intangible asset	14.75	13.82
Amortisation of right to use assets	4.28	
Total	191.80	140.30

#### 29 Loss on fair valuation of derivative contracts

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Loss on fair valuation		
- on put options	and the second	440.00
Total		440.00

#### 30 Other Expenses

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Advertisement and business promotion	872.84	734.02
Audit Fees	41.98	23.65
Brokerage	206.39	93.49
Arbitration Award paid	1,120,46	
Telephone and Communication	1.17	90.35
Compensation to Customers		98.67
Directors Sitting Fees	45.00	47.60
Fees & Consultations	1.999.53	885.27
Insurance	38.80	33.09
Loss on Sale of Property plant & equipment		23,17
Office and common area maintenance charges	732.05	840.73
Power & Utilities	22.08	29.96
Printing, courier and stationery	3.34	25.03
Provision for credit impaired Inter corporate deposits	516.62	
Provision for credit impaired Trade Receivables	306.00	
Rates & Taxes	116.81	66.62
Recruitment and conference expenses	12.72	109.54
Rent	109.55	364.34
Repairs and maintenance	92.27	138.33
Security charges	22.85	41.56
Travelling and conveyance	132.81	191,48
Miscellaneous expenses	48.57	46.35
Total	6.441.84	3,883.25

#### Foot Note: Remuneration to Statutory Auditors:

Particulars	For the year ended	For the year ended
	31 March 2020	31 March 2019
Statutory Audit	34.00	21.00
Other Services	3.96	0.26
Taxation Matters	3.00	1.50
Out of Pocket Expenses	1.02	0.89
Total	41.98	23.65

#### 31 Amounts written off during the year

1 March 2020	31 March 2019
	5,414.38
	5,414.38
	-

#### 32 Tax Expense

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
(a) Amounts recognised in profit and loss		
Current income tax		
Deferred Tax	(2,417,51)	(1,298.87)
Tax expense for the year	(2,417.51)	(1,298.87)
(b) Income tax recognised in other comprehensive income		
Items that will not be reclassified to profit or loss		
Remeasurements of the defined benefit plans	4.50	(13.68)
Equity Instruments through Other Comprehensive Income	(3,496,97)	(3,502.88)
Tax expense for the year	(3,492.47)	(3,516.56)
(c) Income tax expense for the year can be reconciled to the accounting profit as follows		
Loss before tax	(20,184,56)	(17,275.25)
Tax using the Company's domestic tax rate 26.00% (2019: 26.00%)		
Tax effect of:		
Reduction in tax rate		
Deferred tax on fair valuation through profit or loss	(1,859.08)	(678.95)
Deferred tax on business expenses	(558.43)	(619.92)
Income tax expense / (benefit) recognised in Statement of profit and loss	(2,417.51)	(1,298.87)

#### (d) Movement in deferred tax balances

Net balance 1 April 2019	Recognised in profit or loss	Recognised in OCI	Recognised in Deemed Equity Investments	Net balance 31 March 2020	Deferred tax asset	Deferred tax liability
		1.5	-			
(13,258.37)	(2,417.51)	(3,492.47)	A	(19,168.35)	-	(19,168.35)
(13,258.37)	(2,417.51)	(3,492.47)		(19,168.35)		(19,168.35)
Net balance 1 April 2018	Recognised in profit or loss	Recognised in OCI	Recognised in Deemed Equity Investments	Net balance 31 March 2019	Deferred tax asset	Deferred tax liability
	-					
(10.470.11)	(1,298.87)	(3,516.56)	2,027.17	(13,258.37)		(13,258.37)
(10,470.11)	(1,298.87)	(3,516.56)	2,027.17	(13,258.37)		(13,258.37)
	1 April 2019 (13,258.37) (13,258.37) Net balance 1 April 2018 (10.470.11)	1 April 2019         profit or loss           (13,258.37)         (2,417.51)           (13,258.37)         (2,417.51)           Net balance 1 April 2018         Recognised in profit or loss           (10,470,11)         (1,298.87)	1 April 2019         profit or loss         In OCI           (13,258.37)         (2,417.51)         (3,492.47)           (13,258.37)         (2,417.51)         (3,492.47)           Net balance 1 April 2018         Recognised in profit or loss         Recognised in OCI           (10,470.11)         (1,298.87)         (3,516.56)	Net balance         Recognised in profit or loss         Recognised in OCI         Deemed Equity Investments           (13.258.37)         (2.417.51)         (3.492.47)         -           (13.258.37)         (2.417.51)         (3.492.47)         -           Net balance 1 April 2018         Recognised in profit or loss         Recognised in OCI         Recognised in Deemed Equity Investments           (10.470.11)         (1.298.87)         (3.516.56)         2.027.17	Net balance         Recognised in profit or loss         Recognised in OCI         Deemed Equity Investments         Net balance 31 March 2020           (13,258.37)         (2,417.51)         (3,492.47)         -         (19,168.35)           (13,258.37)         (2,417.51)         (3,492.47)         -         (19,168.35)           Net balance         Recognised in profit or loss         Recognised in n OCI         Recognised in Deemed Equity Investments         Net balance           (10.470.11)         (1,298.87)         (3,516.56)         2,027.17         (13,258.37)	Net balance         Recognised in profit or loss         Recognised in OCI         Deemed Equity Investments         Net balance 31 March 2020         Deferred tax asset           (13,258.37)         (2,417.51)         (3,492.47)         -         (19,168.35)         -           (13,258.37)         (2,417.51)         (3,492.47)         -         (19,168.35)         -           Net balance 1 April 2018         Recognised in profit or loss         Recognised in OCI         Recognised in Deemed Equity Investments         Net balance 31 March 2019         Deferred tax asset           (10.470.11)         (1,298.87)         (3,516.56)         2,027.17         (13,258.37)

The Company offsets tax assets and liabilities if and only if it has a legally enforceable right to set off current tax assets and current tax liabilities and the deferred tax assets and deferred tax liabilities relate to income taxes levied by the same tax authority.

Significant management judgement is required in determining provision for income tax, deferred income tax assets and liabilities and recoverability of deferred income tax assets. The recoverability of deferred income tax assets is based on estimates of taxable income and the period over which deferred income tax assets will be recovered. Any changes in future taxable income would impact the recoverability of deferred tax assets.

Tax losses carried forward Deferred tax assets have not been recognised in respect of unabsorbed business losses, because it is not probable that future capital gains profit will be available against which the Company can use the benefits therefrom.

#### 33 Expenditure in foreign currency (on accrual basis)

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Professional fees	13.73	*
Training and conference expenses		1.21
Membership & Subscription Expenses	14.02	11.56
Travelling expenses	11.36	0.95

#### 34 Earning Per Share

Earnings Per Share (EPS) = Net Profit attributable to Shareholders / Weighted Number of Shares Outstanding

Particulars		For the year ended 31 March 2020	For the year ended 31 March 2019
(Loss) after tax attributable to equity shareholders	A	(22,584.76)	(18,626.77
Calculation of weighted average number of equity shares:			
Number of equity shares at the beginning of the year		1,01,73,07,692	1,01,73,07,692
Equity shares issued during the year		100 C C C C C C C C C C C C C C C C C C	
Number of equity shares outstanding at the end of the year		1,01,73,07,692	1,01.73,07,692
Weighted average number of equity shares outstanding during the year	В	1,01,73,07,692	1,01.73,07,692
Rights Shares - application money received during the year - pending allotment		60,00,00,000	
Weighted average number of equity shares outstanding during the year	C	1,61,73,07,692	1,01,73,07,692
Earning Per Share – Basic (INR)	(A / B)	(2.22)	(1.83
Earning Per Share - Diluted (INR) *	(A / C)	(2.22)	(1.83
* Anti-dilutive - hence not considered			

#### 35 Contingencies and commitments

Particulars	Jars 31 March 2020	
i) Contingent Liabilities (Refer footnote 1)		
Claims against the Company not acknowledged as debts		
- Income tax demands contested by the Company	443.65	1,268,59
- Indirect tax demands contested by the Company	467.30	467.30
- Claims made by contractors (Refer footnote 2)		1,179.00
- Other Legal Claims	602.49	

Foot Note

1. The Company does not expect any outflow of economic resources in respect of the above and therefore no provision is made in respect thereof

2. Claim made by contractors has been paid during the current year, and it is debited to the Statement of Profit and Loss (Refer Note 30)

#### (ii) Commitments

Particulars	31 March, 2020	31 March, 2019
(a) Indemnity for representations and warranties for disinvestment in retail business	1,350.00	

(b) Bank guarantee issued on behalf of the Company and its Subsidiaries and Joint Ventures out of the overall non fund based limits of the Company INR 7,222.31 Lakns.

Name of Subsidiaries / Joint ventures	Nature of Comfort given	
Mikado Realtors Pvt. Ltd.	Shortfall undertaking to meet any shortfall during the tenure of facility	
Arrow Infra Estates Private Limited	Shortfall undertaking to meet any shortfall during the tenure of facility	
Gurgaon Constructwell Private Limited	Shortfall undertaking to meet any shortfall during the tenure of facility	
Gurgaon Realtech Limited	Shortfall undertaking to meet any shortfall during the tenure of facility	
International Intrabuild Private Limited	To ensure payment to borrowers in the event of termination of the concession agreement.	

(d) The Company has issued financial support letter to following subsidiaries

(1) The Company has issued financial support letter to following subsidiaries :
 1) Acme Living Solutions Private Limited
 2) MIA Intrastructure Private Limited
 2) WIA Intrastructure Private Limited
 3) Walkept Facility Wanagement Services Private Limited (Previously known as TRIL Hospitality Private Limited)
 4) TRIF Gurgoan Housing Projects Private Limited
 5) Gurgaon Constructivell Private Limited
 6) HV Farms Private Limited
 7) TRIL Roads Private Limited
 8) TRIL Urban Transport Private Limited
 (Private Limited
 )
 (Private Limited
 )

#### 36 a) Capital commitments

Particulars	31 March 2020	31 March 2019	
Estimated amount of contracts remaining to be executed on capital account and not provided for (net of advances)	2.70	12.16	
Financial commitments	The second s	and a second second	
Financial commitments Particulars	31 March 2020	31 March 2019	

#### 37 IND AS 116 Disclosure

b)

As The Disclosure Ministry of Corporate Affairs ("MCA") through Companies (Indian Accounting Standards) Amendment Rules, 2019 and Companies (Indian Accounting Standards) Second Amendment Rules, has notified Ind AS 116 Leases which replaces the existing lease standard, Ind AS 17 leases, and other interpretations. Ind AS 116 sets out the principles for the recognition, measurement, presentation and disclosure of leases for both lessees and lessors. It introduces a single, on-balance sheet lease accounting model for lessees.

Effective 1 April 2019, the company adopted IndAS 116 "Leases" and applied the standard to all lease contracts using the modified retrospective method, Consequently, the company recorded the lease liability at the present value of the lease payments discounted at the incremental borrowing rate and the right of use asset at its carrying amount as if the standard had been applied since the commercement date of the lease, but discounted at the lease is incremental borrowing rate at the date of initial application. On transition, the adoption of the new standard resulted in recognition of 'flight of Use sastet of R.1.506 Lakhs and a lease liabilities of R.1.48.28 Lakhs. The cumulative effect of applying the standard, amounting to R.1.30 Lakhs was debited to retained earnings, net of taxes. Ind AS 116 will result in an increase in cash inflows from operating activities and an increase in cash outflows from financing activities on account of lease payments.

The Company had no finance leases as on 31 March 2019 and accordingly no remeasurment was done. On application of Ind AS 116, the nature of expenses has changed from lease rent in previous periods to depreciation cost for the right to-use asset, and finance cost for interest accrued on lease liability.

38

The operations of the Company are classified as intrastructure facilities' as defined under Schedule VI to the Act. Accordingly, the disclosure requirements specified in sub-section 4 of Section 186 of the Act in respect of loans given, investment made of guarantee given or security provided and the related disclosures on purposes/ utilisation by recipient companies, are not applicable to the Company.

Details of investments made by Company as on 31 March 2020 (including investments made during the year)

Name of the entity	31 March 2019	Investments made during the year	Sale of Investments during the year	Ind AS adjustments	31 March 2020
A Non-current investments					
Investment in subsidiaries					
Investment in equity shares					
Acme Living Solutions Private Limited			1.00		
Arrow Infra Estates Private Limited	5.362.00	1. S. C.		(2,679.00)	3.683.
Gurgaon Constructwell Private Limited	8,599,00		Q.	(435.00)	8,164.
Deemed Investment in Gurgaon Constructwell Private Limited	619.68			[433.00]	619.
Gurgaon Realtech Limited	6,907.00			(1,339.00)	5,568.
TRIL Roads Private Limited	34,722,99			(6,479.26)	28,243.
Deemed Investment in TRIL Roads Private Limited	2.421.61			(0,473,20)	2,421.
TRIF Gurgaon Housing Projects Private Limited					2,421.
TRIL Urban Transport Private Limited	2.334.00	4.440.00		(562,00)	6,212.
Deemed Investment in TRIL Urban Transport Private Limited	240.43	11440100		(302,00)	240.
Wellkept Facility Management Services Private Limited	2.14/15				240.
TRIL Constructions Limited	2.451.52			437,68	2,889.
TRIL Infopark Limited	1.03.922.85			8,901,81	
TRIL Infopark Limited (Deemed Investment - Call option)	Haplaction			9,901,81	1,12,824.
TRIL Amritsar Projects Limited	3,492.00		3,492,00		
Deemed Investment in TRIL Amritsar Projects Limited	2.457.94		2,487,94	1	
TRIF Real Estate Development Limited	7,615.00		7.615.00		
MIA Infrastructure Private Limited	1,013,00		7,615.00		
HV Farms Private Limited	1.00				1.2
International Infrabuild Private Limited	150.28			1100 000	1.0
Deemed Investment in International Infrabuild Private Limited	100.20	0.000.07		(150.28)	
Soonida integrinent in mendendirenti initabalia Physic Entitied	1,82,327,30	9,029,95	10 501.01	(7,349.60)	1,680.3
TAL S LOOP	1,82,327,30		13,594.94	(9,654.65)	1,72,547.0
Name of the entity	31 March 2019	Investments made during the year	Sale of Investments during the year	Ind AS adjustments	31 March 2020
nvestment in Joint venture companies				_	
nvestment in equity shares					
TRIL IT4 Private Limited	24,487.00			15.871.00	40,358.0
Industrial Minerals and Chemicals Company Private Limited	22.270,60	-		787.19	23,057.7
Mikado Realtors Pvt. Ltd.	33,228,20			(7,421.44)	25,806.7
Construction of the second	79,985,80			9,236.75	89,222,5
	10,000,00			9,230.75	89,222.5
vestment in Preference shares					
TRIL Constructions Limited	5.970.40			1 0 10 00	
TRIL Intopark Limited	49,578,00	-	-	1,048.92	7,019.3
	55,548,40			4,257.00	53,835.0
vestment in Debentures	55,548,40			5,305,92	60,854.3
TRIL Urban Transport Private Limited	- Value	14.00			
	4,160.00	120.00		(283.00)	3,997.0
TRIL Roads Private Limited	22,565.00	13,112.00		1,900.26	
TRIL Roads Private Limited TRIL IT4 Private Limited	8,140.00		1		
TRIL Roads Private Limited TRIL IT4 Private Limited TRIL Intopark Limited	8,140.00 41,273.00		1	1,900,26	8,140.0
TRIL Roads Private Limited TRIL If 4 Private Limited TRIL Integrark Limited HV Farms Private Limited	8,140.00 41,273.00 1,188.00		-		8,140.0 44,862.0
TRIL Roads Private Limited TRIL Tr4 Private Limited TRIL Intopark Limited Wy Farms Private Limited International Intrabuild Private Limited	8,140.00 41,273.00 1,188.00 211.56	13,112.00 - - -	-	3,589.00	8,140.0 44,862.0 1,064.0
TRIL Roads Private Limited TRIL Tr4 Private Limited TRIL Intopark Limited Wy Farms Private Limited International Intrabuild Private Limited	8,140.00 41,273.00 1,188,00 211.56 10,636.70	13,112,00		3,589.00 (124.00) (19.96) 364,75	8,140.0 44,862.0 1,064.0 191.6
TRIL Roads Private Limited TRIL Tr4 Private Limited TRIL Inforat Limited HV Farms Private Limited International Infrabulia Private Limited Industrial Minerals and Chemicals Company Private Limited	8,140.00 41,273.00 1,188.00 211.56	13,112.00 - - -		3,589.00 (124.00) (19.96)	8,140.0 44,862.0 1,064.0 191.6 12,314.9
TRIL Roads Private Limited TRIL Tr4 Private Limited TRIL Intopark Limited HV Farms Private Limited International Infrabuild Private Limited Industrial Minerals and Chemicals Company Private Limited .Trade Investments	8,140.00 41,273.00 1,188,00 211.56 10,636.70	13,112,00		3,589.00 (124.00) (19.96) 364,75	8,140.0 44,862.0 1,064.0 191.6 12,314.9
TRIL Roads Private Limited TRIL IT4 Private Limited TRIL Intopark Limited HV Farms Private Limited International Infrabuild Private Limited Industrial Minerals and Chemicals Company Private Limited <u>Trade Investments</u> vestment in Mutual Funds	8,140.00 41,273.00 1,188,00 211.56 10,636.70	13,112.00 		3,589.00 (124.00) (19.96) 364,75	8,140.0 44,862.0 1,064.0 191.6 12,314.9
TRIL Roads Private Limited TRIL Tr4 Private Limited TRIL Informat Limited HV Farms Private Limited International Infrabula Private Limited Industrial Minerals and Chemicals Company Private Limited  Trade Investments vestiment in Mutual Funds ABSL Liquid Fund - Growth-Direct	8,140.00 41,273.00 1,188,00 211.56 10,636.70	13,112,00	75,496.22	3,589.00 (124.00) (19.96) 364,75	8,140.0 44,862.0 1,064.0 191.6 12,314.9 1,08,146.8
TRIL Roads Private Limited TRIL Infoart Limited TRIL Infoart Limited HV Farms Private Limited Industrial Minerals and Chemicals Company Private Limited Industrial Minerals and Chemicals Company Private Limited <u>Trade Investments</u> <u>vestment in Mutual Funds</u> ABSL Liquid Fund - Growth-Direct HDFC Liquid Fund - Direct Plan - Growth Option	8,140.00 41,273.00 1,188,00 211.56 10,636.70	13,112.00 	5.000	3,589.00 (124.00) (19.96) 384.75 5,427.05	8,140.0 44,862.0 1,064.0 191.6 12,314.9 1,08,146.8
TRIL Treads Private Limited TRIL T14 Private Limited TRIL Infopati Limited HV Farms Private Limited International Infrabuild Private Limited Industrial Minerals and Chemicals Company Private Limited <u>Artade Investments</u> <u>vestment in Mutual Funds</u> ABSL Liquid Fund - Growth-Direct HDFC Liquid Fund - Direct Plan - Growth Option IGCI Prudential Liquid Fund - Direct Plan - Growth	8,140.00 41,273.00 1,188,00 211.56 10,636.70	13,112.00 	75,496.22	3,589.00 (124.00) (19.96) 384.75 5,427.05 64.33	8,140.0 44,862.0 1,064.0 191.6 12,314.9 1,08,146.8 10,513.1 15,051.5
TRIL Roads Private Limited TRIL T4 Private Limited TRIL T4 Private Limited HV Farms Private Limited International Infrabulat Private Limited Industrial Minerals and Chemicals Company Private Limited 	8,140.00 41,273.00 1,188,00 211.56 10,636.70	13,112.00 	75,496.22 79,980.28	3,589.00 (124.00) (19.96) 384.75 5,427.05 64.33 53.85	37,577,2/ 8,140,0/ 1,064,0/ 191,6/ 12,314,9/ 1,08,146,8/ 10,513,11 15,051,57 30,220,06/ 15,394,67
TRIL Treads Private Limited TRIL T14 Private Limited TRIL Infopati Limited HV Farms Private Limited International Infrabuild Private Limited Industrial Minerals and Chemicals Company Private Limited <u>Artade Investments</u> <u>vestment in Mutual Funds</u> ABSL Liquid Fund - Growth-Direct HDFC Liquid Fund - Direct Plan - Growth Option IGCI Prudential Liquid Fund - Direct Plan - Growth	8,140.00 41,273.00 1,188,00 211.56 10,636,70 <b>88,174.25</b>	13,112.00 	75,496.22 79,980.28 46,502.36	3,589,00 (124,00) (19,96) 384,75 5,427,05 5,427,05 54,33 53,85 194,41	8,140.0 44,862.0 1,064.0 191.6 12,314.9 1,08,146.8 10,513.1 15,051.5 30,220.0

Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements

for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### 39 Disclosure of transactions with Related Parties, as required by Indian Accounting Standard (Ind AS) • 24 "Related Party Disclosures" has been set out below: (a) List of Related Parties where control exists:

Holding Company: Tata Sons Private Limited

Subsidiary Companies: Acree Living Solutions Private Limited Arrow Infraestate Private Limited

- Dharamshala Ropeway Limited (subsidiary of TRIL Urban Transport Private Limited) Durg Shivnath Expressways Private Limited (Formerly known as SMS Shivnath Infrastructure Pvt Ltd) (wholly owned subsidiary of TRPL Roadways Private Limited) Gurgano Constructivell Private Limited
- Gurgaon Realtech Limited

Hampi Expressways Private Limited (wholly owned subsidiary of TRIL Roads Private Limited) HV Farms Private Limited International Infrabuld Private Limited

Manafi Ropeway Private Limited (subsidiary of TRIL Urban Transport Private Limited) Matheran Rope-Way Private Limited (subsidiary of TRIL Urban Transport Private Limited) MIA Infrastructure Private Limited

- TRIF Gurgan Housing Projects Private Limited TRIF Gurgan Housing Projects Private Limited TRIF Real Estate And Development Limited (upto 9th December, 2019) TRIL Amritsar Projects Limited (formerly known as TRIF Amritsar Projects Limited) (upto 9th December, 2019)
- TRIL Constructions Limited
- TRIL Infopark Limited TRIL Roads Private Limited
- TRIL Urban Transport Private Limited

TREL Roadways Private Limited (wholly-owned subsidiary of TRIL Roads Private Limited) Uchit Expressways Private Limited (wholly owned subsidiary of TRIL Roads Private Limited) Wellkept Facility Management Services Private Limited (Previously known as TRIL Höspitality Private Limited)

Joint Venture: A & T Road Construction Management and Operation Private Limited (Investment is held by TRIL Roads Private Limited, which is 100% subsidiary of Tata Reality and Infrastructure Limited)

Industrial Minerals and Chemicals Private Limited Mikado Realtors Private Limited

Pune Solapur Expressways Private Limited (Investment is held by TRIL Roads Private Limited, which is 100% subsidiary of Tata Realty and Infrastructure Limited) TRIL IT4 Private Limited (Previously known as Albrecht Builder Private Limited) Pune IT City Metro Rail Limited

#### Other related parties with whom transactions have taken place during the year:

Other related parties with whom transactions have taken place during the y Fellow Subsidiaries: Ewart Investments Limited Tata AIG General Insurance Limited Tata Business Excellence Group (A Division of Tata Sons Private Limited) Tata Consultancy Services Limited Tata Consulting Engineers Limited Tata Housing Development Company Limited Tata Quality Management Services (A Division of Tata Sons Privat Limited)

#### Key Managerial Personnel:

Sanjay Dutt Sanjay Sharma Managing Director & CEO Company Scretary - w.e.t. 10th September 2018 Chief Financial Officer - w.e.t. 10th September 2018 Chief Financial Officer - upto 28th February 2019 Company Secretary - upto 30th November 2019 Company Secretary - w.e.t. 1st December 2019 Arvind Chokhany Vinay Gaokar Sudhakar Shetty

	Nature of Transactions / relationship / major parties	Holding company	Subsidiary companies	Fellow subsidiaries	Joint Ventures	Total
1.1.1	Share Application Pending Allotment	1,20,000.00				1,20,000.00
Ļ						-
1	Tata Sons Private Limited	1,20,000.00	-		<b>T</b>	1,20,000.00
5	Unsecured loan taken	· · · ·				
1	Unsecured loan taken	· · ·		42,840.00	1+1	42,840.00
-	Tata Housing Development Company Limited	×	1.1	(14,000.00)	1	(14,000.0
	rata nousing Development Company Limited	*	-	42,840.00	· ·	42,840.00
ī	Unsecured loan repaid	-		(14,000.00)		(14,000.00
1		-		56,840.00		56,840.00
7	Tata Housing Development Company Limited	2		56,840.00		56,840.00
						50,040.00
0	A CONTRACTOR OF			-		
h	nterest expenses on unsecured loans	- 1	-	838.76		838.76
_		-		(20.71)		(20.71
T	ata Housing Development Company Limited	•	1.4	838.76	(-1	838.76
-				(20.71)		(20.71
h	nvestments in Equity,CCD,OCD during the year	•	17,672.00	9	1,313.50	18,985.50
-			(5,004.04)		(10,473.96)	(15,478.00
1	RIL Roads Private Limited		13,112.00	•	1. A. C.	13,112.00
Ŧ	Dil Ultra Transi Di La la val		(2,450.00)	2	6	(2,450.00
ľ	RIL Urban Transport Private Limited		4,560.00	-		4,560.00
Ir	ndustrial Minerals and Chemicals Private Limited	-	(2,554.04)	÷.	1	(2,554.04
	ioustrial minerals and Chemicals Private Limited	-	•		1,313.50	1,313.50
S	ale of property, plant and equipment	14			(10,473.96)	(10,473.96
~	are of property, plant and equipment	(28.32)	2	(2.10)		-
T	ata Sons Private Limited	(20.02)	-	(2.19)	-	(30.50
		(28.32)	_			(28.32)
E	wart Investments Limited	-	1	2	-	(20.52
		4	-	(2.19)		(2.19)
In	ter Corporate Deposit Given		43,086.52		-	43,086.52
_			(24,717.17)			(24,717.17)
A	rrow Infraestate Private Limited		180.00	÷		180.00
i.			(2,253.00)	12		(2,253.00)
G	urgaon Constructwell Private Limited		477.00		-	477.00
		-	(4,664.79)	~	-	(4,664.79)
III	ternational Infrabuild Private Limited	10	9,937.95	÷.	19	9,937.95
3	urgaon Realtech Limited		(1,540.00)	-		(1,540.00)
	ligaon realiech Limited	-	1,060.00	-	-	1,060.00
FF	RIL Roads Private Limited	18	(2,657.00)		1	(2,657.00)
1		3	3,925.00 (8,363.00)		-	3,925.00
FF	RIF Real Estate And Development Limited		(8,363.00) 22,837.00	-		(8,363.00) 22,837.00
			(2,500.00)			(2,500.00)
FF	RIL Urban Transport Private Limited	14	3,895.18			3,895.18
			(1,193.48)			(1,193.48)
	RIL Amritsar Projects Limited	1.000	516.62		1	516.62
P		1.1	(650,00)	1.	-	(650.00)
			257.77			257.77
	hers		201.11			

R

283

Inter Corporate Deposit Refunded Arrow Infraestate Private Limited Gurgaon Realtech Limited TRIL Amritsar Projects Limited TRIF Real Estate And Development Limited International Infrabuild Private Limited		48,799.48 (7,490.33) - (2,453.52) 200.00 (3,228.81)			48,799.44 (7,490.3
Gurgaon Realtech Limited TRIL Amritsar Projects Limited TRIF Real Estate And Development Limited		- (2,453.52) 200.00			(7,490.3
Gurgaon Realtech Limited TRIL Amritsar Projects Limited TRIF Real Estate And Development Limited		200.00	1		
TRIL Amritsar Projects Limited TRIF Real Estate And Development Limited		200.00			1.1.1.1.
TRIL Amritsar Projects Limited TRIF Real Estate And Development Limited					(2,453.5
TRIF Real Estate And Development Limited		(3,228.81)			200.0
TRIF Real Estate And Development Limited		1		1.00	(3,228.8
	-	19,372.00			19,372.0
	5	(1,030.00)	÷	-	(1,030.0
International Infrabuild Private Limited		25,162.00	-	÷	25,162.0
International Infrabuild Private Limited	(÷	1 ÷		100	1.04
	12	2,595.00	-	-	2,595.0
	1.000	(765.00)		-	(765.0
TRIL Urban Transport Private Limited	÷	1,462.48		- ÷	1,462.4
		(13.00)			(13.0
Others		8.00			8.0
	-				14
NCD Subscription		20,100.00	-		20,100.0
Nob Subscription		20,100,00			20,100.0
TRIL Amritsar Projects Limited		12,000.00	-	-	12,000.0
The Annabi Projecta Ennied		12,000.00			12,000.0
TRIF Real Estate And Development Limited		8,100.00			8,100.0
This heat Estate And Development Emilied	-	8,100.00			0,100.0
Desired Names at Occurring the		1 000 15			
Project Management Consultancy fees	-	1,232.45		1,624.99	2,857.4
		(2,158.91)		(1,118.49)	(3,277.4
Gurgaon Realtech Limited		493.98			493.9
	.x=1.	(1,045.46)	<u>^</u>		(1,045.4
International Infrabuild Private Limited	1.5.1	44.11			44.1
		(14.35)	-	1	(14.3
Mikado Realtors Private Limited				1,485.21	1,485.2
		7	-	(1,118.49)	(1,118.4
TRIL IT4 Private Limited				139.77	139.7
				-	
TRIF Real Estate And Development Limited	1	-	-	-	
	- ÷	(208.18)	-		(208.1
TRIL Infopark Limited		425.37	-	1.4	425.3
		(422.92)			(422.9
Uchit Expressways Private Limited	1.1	269.00	- C.		269.0
		(378.00)		4	(378.0
Hampi Expressways Private Limited		(0/0.00)			(010.0
hampi Expressways i fivate Elinted		(90.00)			(90.0
Asset Management Fees		1.032.89	-	135.75	1,168.6
Asset management rees	-				
TDU Informatic Limited		(972.88)		(148.40)	(1,121.2
TRIL Infopark Limited		974.92			974.9
	-	(882.78)		and the second se	(882.7
TRIL IT4 Private Limited	~		~	135.75	135.7
	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1.00		(148.40)	(148.4
Others	÷	57.97	-	*	57.9
		(90.10)	-	-	(90.1
Interest Income	•	1,561.64		1,465.20	3,026.8
		(1,609.75)		(1,465.20)	(3,074.9
TRIL Roads Private Limited		1,164.20			1,164.2
		(1,164.20)	- G. 1		(1,164.2
TRIL IT4 Private Limited		1.1		1,465.20	1,465.2
				(1,465.20)	(1,465.2
TRIF Real Estate And Development Limited	1.16	223.27			223.2
the second state of the participation climited		(183.04)		2	(183.04
Others		174.17			174.1
Oliford		(262.51)			(262.5

R

Nature of Transactions / relat	tionship / major parties	Holding company	Subsidiary companies	Fellow subsidiaries	Joint Ventures	Total
Other Expenses		6.61		627.47	1	634.08
		(2.20)		(530.21)		(532.41
Ewart Investments Limited			-			
		3. ·	4	(175.30)		(175.30
Tata Sons Private Limited		6.61		1.5	-	6.61
		(2.20)		-	-	(2.20
Tata AIG General Insurance Limited				117.41		117.41
		2		(151.46)	-	(151.46
TC Travel and Services Limited		-				
		4	1.21		-	
Tata Consulting Engineers Limited				491.79		491.79
		4	1.00	(150.03)		(150.03
Others			1.25	18.27		18.27
		-		(53.42)		(53.42
Reimbursement of expenses		19,11	794.42	294.38	36.31	1,144.22
		(7.61)	(1,093.47)	(604.61)	(29.38)	(1,735.08
Arrow Infraestate Private Limited		(7:01)	0.02	(004.01)	(23.30)	0.02
			(0.02)			(0.02
Gurgaon Realtech Limited			190.38		2	190.38
adigatin neareon Emilio		100	(214.17)			(214.17
Gurgaon Constructwell Private Limite	ad	-	(214.17)			
Curgaon Constructiven Private Linne				-	-	0.07
International Infrabuild Private Limite	d		(0.04)		-	(0.04
international finadulu Private Linite	a	•	20.45		-	20.45
TRIL Roads Private Limited		-	(68.47)		7	(68.47
TAIL HOADS FIIVale Limited			257.26			257.26
Tata Sons Private Limited		-	(425.53)	-		(425.53
Tata Sons Private Limited		19.11	-		1.5	19.11
TPIL Lichan Transport Drivets Limiter		(7.61)	-	-		(7.61
TRIL Urban Transport Private Limited	1		233.07			233.07
TOU An item Desired Linited			(187.53)	-	2	(187.53
TRIL Amritsar Projects Limited		-	7.64	·	÷	7.64
		-	(16.64)		-	(16.64
TRIF Real Estate And Development I	Limited		27.46	· ~	1.5	27.46
			(57.30)			(57.30
Tata Housing Development Co. Limit	ed	-		294.38		294.38
		~		(604.61)		(604.61
Others		-	58.06		36.31	94.36
			(123.77)	· · · ·	(29.38)	(153.16
Deposit Refund Received						1.11
		(150.00)				(150.00
Tata Sons Private Limited			~	-		•
		(150.00)		÷		(150.00
Deposit Given				153.83	8	153.83
			· · · ·	(34.90)		(34.90
Fata Consulting Engineers Limited		-[+]	1	153.83	~	153.83
				(34.90)		(34.90
Employee Benefit Transfer			- 14 -	(H)	(+	
		(56.50)	· ·	· · · · · · · · · · · · · · · · · · ·		(56.50
Tata Sons Private Limited						
		(56,50)				(56.50)



AV

Nature of Transactions / relationship / major parties	Holding company	Subsidiary companies	Fellow subsidiaries	Joint Ventures	Total
Outstanding Balances Receivables					
Inter Corporate Deposit - Current - Unsecured	- C	19,713.90	4	-	19,713.9
		-			
International Infrabuild Private Limited		9,579.95			9,579.9
		5,57 5,55			5,075.0
Gurgaon Realtech Limited		1,060.00	(* D.	C	1,060.0
		1,000,00			1,000.0
TRIF Real Estate And Development Limited		675.00		1	675.0
The field Estate find Development Elimited		075.00			075.0
Gurgaon Constructwell Private Limited		477.00			477.0
aligaen eenenaathen mata Ennied		477,00			4//.
TRIL Urban Transport Private Limited		3,739.18		-	3,739.1
The orban hansport invate climited		3,739.10		-	3,739.
TOIL Doods Drivets Limited	- C.	-	5		
TRIL Roads Private Limited		3,925.00		-	3,925.0
Others		-			
Others	(C)	257.77		-	257.
		· · · · ·		•	-
Inter Corporate Deposit - Non-Current - Unsecured		31,705.90			31,705.9
		(57,649.38)			(57,649.
TRIL Amritsar Projects Limited		÷ .		*	•
		(19,372.00)	-	-	(19,372.0
TRIL Roads Private Limited	-	24,801.00		-	24,801.
		(24,801.00)	-		(24,801.)
International Infrabuild Private Limited				-	
	÷	(2,237.00)			(2,237.0
Gurgaon Constructwell Private Limited		4.817.00		-	4,817.0
		(4,825.00)		2	(4,825.0
Others	-	2,087.90		-	2,087.9
		(6,414.38)		-	(6,414.3
Interest Accrued but not due		1,287.86		64.05	1,351.9
		(2,806.97)		(64.23)	(2,871.1
TRIL Amritsar Projects Limited		(2,000.97)		(04.23)	(2,0/1.
	-	(1,317.01)			
TRIL Roads Private Limited	-				(1,317.0
The hodds thivate elitited		1,047.78		-	1,047.7
TRIL Constructions Limited		(1,047.78)	-		(1,047.7
TRIE Constructions Limited		227.80			
Others					
Others		12.28		64.05	76.3
All . B		(442.18)		(64.23)	(506.4
Other Recoverable		801.84	320.39	34.27	1,156.5
		(1,429.19)	(652.98)	(14.46)	(2,096.6
Arrow Infraestate Private Limited		0.02			0.0
		(0.02)		÷.	(0.0
Gurgaon Realtech Limited	11 m	173.09	-	- 14 - 1	173.0
			-	-	
Gurgaon Constructwell Private Limited	-	0.04	-		0.0
	-	(0.04)	-	1.1	(0.0
TRIL Urban Transport Private Limited		21.93	-	-	21.9
	-	(759.28)	-		(759.2
TRIL Constructions Limited	14	204.73	1.1	6	204.7
		(199.17)			(199.1
TRIL Roads Private Limited		275.68			275.6
	1.1	(198.77)	5	-	(198.7
Tata Housing Development Co. Ltd.	100	(150.77)	317.93		
the stand betalaphiant of the	8				317.9
Others		100.00	(652.98)	01.07	(652.9
201013		402.03	2.46	34.27	438.7
	-	(271.91)		(14.46)	(286.3

R

	Nature of Transactions / relationship / major parties	Holding company	Subsidiary companies	Fellow subsidiaries	Joint Ventures	Total
T	rade Receivable	22.55	801.92	300.00	734.51	1,858.9
			(749.16)	(300.00)	(287.69)	(1,336.8
G	Gurgaon Realtech Limited		391.34	-	-	391.3
			(289.37)			(289.3
Т	RIF Real Estate And Development Limited					
		2	(175.79)		-	(175.7
Т	RIL Infopark Limited	-	354.54	-		354.5
			(234.40)	-		(234.4
N	fikado Realtors Private Limited		(204.40)	2	734.51	734.5
			-	-	(272.83)	(272.8
Ir	nternational Infrabuild Private Limited		8.40		(2/2.00)	8.4
"	Remandria minabalia i mate climitea					
т	ata Consultancy Services Limited		(1.67)			(1.6
1	ata consulancy services Limited			300.00		300.0
Ŧ	ate Case Dévets Limited	-		(300.00)		(300.0
3	ata Sons Private Limited	22.55	~	-	-	22.5
0					-	
0	thers		47.65	-	÷	47.6
_			(47.94)		(14.86)	(62.8
P	rovision for Inter Corporate Deposit	- C	611.62	-		611.6
			(95.00)	-	-	(95.0
TRIL Urban Trans	RIL Urban Transport Private Limited	-	95.00			95.0
		-	(95.00)	-	÷.	(95.0
TRIL Amritsar Projects Limited	RIL Amritsar Projects Limited	÷.	516.62	-	2,0	516.6
		*	-	2		
Provision for Inte	rovision for Interest Accrued but not due		10.00	-		10.00
			(10.00)			(10.00
TRIL Urban Transpor	RIL Urban Transport Private Limited		10.00		-	10.00
			(10.00)	-		(10.00
International I	ternational Infrabuild Private Limited		(10:00)			(10.0.
	to mail office in the condition of the condition	1				- 2
Provis	rovision for Advances recoverable		35.00			
	ovision for Advances recoverable					35.00
Т	DIL Lithon Transport Drivets Limited		(35.00)			(35.00
I RIL Urban	RIL Urban Transport Private Limited	-	35.00	×		35.00
-			(35.00)	-		(35.00
	utstanding Balances Payable					-
to	wards unsecured loans			C. S. Star		
_				(14,000.00)		(14,000.00
Ta	ata Housing Development Co. Ltd.				~	
_				(14,000.00)		(14,000.00
to	wards interest on unsecured loans					•
		1.5	4	(18.64)		(18.64
Та	ata Housing Development Co. Ltd.	(÷)	8	-	+	
			-	(18,64)	-	(18.64
٢r	ade Payable	(J-)	1	165.74		165.74
		141		(6.54)		(6.54
Га	ta Consultancy Services Limited			5.36	1-	5.36
-			1	(6.54)		(6.54
Ta	ta Consulting Engineers Limited			(6.54)		159.71
. a	a concerning Engineers Enning		100	159.71		
<b>F</b> -	ta Teleservices Ltd.					-
d	Ita Teleservices Liu.			0.66		0.66
						-

Managerial remuneration	1,132.76
	(1,521.32)
Sanjay Dutt*	717.64
	(1,007.97)
Sanjay Sharma	271.28
	(179.44)
Arvind Chokhany	
	(224.49)
Vinay Gaokar	111.99
	(109.41)
Sudhakar Shetty	31.86

Figures in brackets pertains to previous year.
 \* Recovery of managerial remuneration from fellow subsidiary is not netted off for this disclosure.

40 Segment Reporting The Company is engaged in development of real estate and infrastructure facilities for residential use and project management consultancy services for real estate and infrastructure development. Thus, the Company is engaged in three business segments viz, development of residential property for outright sale, project management and consultancy services and Investment and Lending services. Further, the Company is engaged in providing services in domestic market only. Hence, there are no separate reportable geographical segments.

	Development of residential property for outright sale		Project management and consultancy services		Investment and Lending services		Total	
	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019
REVENUE								
Net sales								
Segment Revenue	13,608.67	10,253.03	4,026.07	4,401,88	14,909.85	12,466,89	32,544,59	27,121.80
Inter segment sales		1	100 C	2				1.1.1.1
Total revenue	13,608.67	10,253.03	4.026.07	4,401.88	14,909.85	12,466.89	32.544,59	27,121.80
Segment Expenses	14,061.87	8,236.33	1,720.02	1,614.15	5,043,53	10,663.02	20,825,42	20,513.50
RESULT								
Segment Result	(453.20)	2,016.70	2,306.05	2,787.73	9,866.32	1,803.87	11.719.17	6.608.30
Unallocated income	1.0			-			6.02	125.44
Unallocated expenses	14		-	6		÷	3.204.47	932.43
Operating Profit / (Loss)	(453.20)	2,016.70	2,306.05	2.787.73	9,866.32	1,803.87	8,520.72	5.801.31
Finance costs allocated				4	28,496.17	18,825.06	28,496,17	18,825.06
Finance cost unallocated			e.					4,163,85
Depreciation / Amortisation	7.52	13.84	92.14	63.23	92.14	63.23	191,80	140.30
Net Profit / (Loss )	(460.72)	2,002.87	2,213.91	2,724.50	(18,721.99)	(17,084.42)	(20,167.25)	(17,327.90
OTHER INFORMATION								
Segment Assets	34,803,18	43,382.01	3.254.02	1,308,00	5.81.362.35	4.76.144.50	6 19.419.55	5.20.834.51
Unallocated assets	-	40,002,01	3,204.02	1,000,00			6,934.67	6,012,90
Total Assets	34,803.18	43,382.01	3,254,02	1,308.00	5,81,362,35	4,76,144.50	6 26 354,22	5,26,847,41
LIABILITIES								
Segment Liabilities	4,483.98	6,691,48			2 99,472.07	2,26,929,94	3.03.956.05	2,33,621.42
<b>Unallocated Liabilities</b>	2.00		-				28.971.55	94,540.04
Fotal Liabilities	4,483.95	6.691.48		4	2,99,472.07	2.26.929.94	3,32,927.60	3,28,161,46
Cost incurred to acquire Segment property, plant and equipment during the year		1.71	36.29	268.41	36,29	268.41	72.58	538.52

Segment accounting policies Segment accounting policies are in line with accounting policies of the Company. In addition, the following specific accounting policies have been followed for segment reporting. Segment revenue includes income directly identifiable with the segments. Expenses that are directly identifiable with the segments are considered for determining the segment result. Expenses which relate to the Company as a whole and not allocable to segments and expenses which relate to the operating activities of the segment but are impracticable to allocate to the segment, are included under "Unallocable expenses". Income which relates to the Company as a whole and not allocable to segments is included in Unallocable Income and netted off from Unallocable expenses. Segment assets and liabilities include those directly identifiable with the respective segments. Unallocable corporate assets and liabilities represent the assets and liabilities that relate to the Company as a whole and not allocable to any segment.
Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

#### 41 Employee Benefits:

(i) The Company has adopted Ind AS 19 on "Employee Benefits" as prescribed by the Companies (Indian Accounting Standards) Rules, 2015 issued by the Central Government

(ii) Contribution to Provident fund

Contract Section 1. And Advanced Section 2. A section of the se	31 March 2020	31 March 2019
Contribution to provident lund recognised as an expense under "Employee benefits".	128.79	134.56

#### (iii) Defined Benefit Plans

Gratuity
The Company has a defined benefit gratuity plan. Every employee who has completed five years or more of service gets a gratuity on
death or resignation or retirement calculated as per the Payment of Gratuity Act, 1972 with no ceiling. 1 March 2020
31 March 2019

	31 March 2020	31 March 2019
I Change in the defined benefit obligation		
Liability at the beginning of the year	367.93	407.58
Interest cost	25.99	31.97
Current service cost	57.08	70.58
Benefits paid	(87.02)	(89.55)
Actuarial loss on obligations	17.31	(52.65)
Liability acquired on acquisition / (settled on Divestiture)		
Liability at the end of the year	381.29	367.93
II Amount Recognised in the Balance Sheet		
Liability at the end of the year	381.29	367.93
Fair Value of Plan Assets at the end of the year		
Difference	381.29	367.93
Amount recognised in the Balance Sheet	381.29	367.93
Expenses Recognised in the statement of profit and loss		
Current Service Cost	57.08	70.58
Interest Cost	25.99	31.97
Expected Return on Plan Assets		
Net Actuarial (Gain) / Loss To Be Recognised	17.31	(52.65)
Past service cost		
Expense Recognised in the statement of profit and loss	100.38	49.90
Balance Sheet Reconciliation		
Opening net liability	367.93	407.58
Expense as above	100.38	49.90
Employer's contribution received / (paid)	(87.02)	(89.55)
Liability acquired on acquisition / (settled on Divestiture)		
Amount recognised in Balance Sheet	361.29	367.93
Actuarial Assumptions :		
Discount Rate	5.55%	7.05%
Salary escalation	7.00%	7.00%

Attrition Rate: Directors - Nil, Age 21-30 years - 5%. Age 31-40 years - 3%, Age 41-59 years -2% Estimates of future salary increases, considered in actuarial valuation, take account of inflation, seniority, promotion and other relevant factors, such as supply and demand in the employment market. The Company's liability on account of gratuity is not funded and mence, the disclosures relating to the planned assets are not applicable.

Experience Adjustments:	31 March 2020	31 March 2019
Defined benefit obligation	381.29	367.93
Plan assets		
Surplus / (Deficit)	(381.29)	(367.93
Experience adjustment on plan llabilities	-	-
Experience adjustment on plan assets		

## (iv) Other long term employment benefits

Compensated absences The liability towards compensated absences for the year ended 31 March 2020 recognised in the Balance Sheet based on actuarial valuation using the projected unit credit method amounted to INR 429.80 Takhs (2019: INR 542.68 Takhs).

Sensitivity analysis Reasonably possible changes at the reporting date to one of the relevant actuanal assumptions, holding other assumptions constant, would have affected the defined benefit obligation by the amounts shown below.

Salary escalation (% movement)	31 March 2020	31 March 2020	31 March 2019	31 March 2019	
	Increase	Decrease	Increase	Decrease	
Discount rate ( % movement)	(2.1)	2.2	(2.6)	2.7	
Salary escalation (% movement)	2.2	(2.1)	2.7	(2.6)	1
Additional Rate (% movement)	(7.4)	13.3	(5.7)	8.6	
					UX
					5
					V

# Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements (or the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

....

42 Financial instruments – Fair values and risk management
 A. Accounting classification and fair values
 The following table shows the carrying amounts and fair values of financial assets and financial liabilities, including their levels in the fair value hierarchy. It does not include fair value information for
 financial assets and financial liabilities if the carrying amount is a reasonable approximation of fair value. The Company's secured loan under current matunties has been contracted at floating rates of
 interest, which are reset at short intervals. Accordingly, the carrying value of such long-term debt approximates fair value.
 Fair value
 Carrying amount
 Carrying amount

		Carrying a	mount		Fair value				
31 March 2020	FVTPL	FVTOCI	Amortised Cost	Total	Level 1 - Quoted price in active markets		Level 3 - Significant unobservable inputs	Total	
Financial assets		-							
Non-current financial assets									
<ul> <li>Equity instruments in subsidiaries, joint ventures and</li> </ul>		2,61,770,21	-	2,61,770.21		-	2,61,770.21	2,61,770.3	
associate - Preference shares		60,854,33		60,854,33			60,854,33	60.854.3	
- Bonds and debentures	87.691.86	12.314.95	8,140,00	1.08.146.81			1.08.146.81		
Long-term loans and advances	39,709,19	12,314,95	8,140,00	39,709,19			39,709,19	1,08,146.8	
Other non-current financial assets	23,(031)3		0.75	0.75	1.1.1	1.1	34,709.19	39,709.1	
				0.70			0.00	0.1	
Current financial assets									
Current investments	71,179,33		1.1.81	71,179.33	71,179.33	8		71,179.3	
Trade receivables	-	-	1,878.42	1,878.42			1,878,42	1,878.4	
Cash and cash equivalents	141	1.1	25,580.30	25,580.30			25,580.30	25,580.3	
Other Bank Balances	÷1	3	419.25	419.25			419,25	419.3	
Short-term loans and advances	5	-	2,464,51	2,464.51	-	~	2,464,51	2,464.5	
Other financial assets		1.6	1,428,91	1.428.91	A	10	1,428.91	1,428,9	
Total Financial Assets	1,98,580.38	3,34,939.49	39,912.14	5,73,432.01	71,179.33	¥1	5,02,252.68	5,73,432.0	
Financial flabilities									
Non-current Financial liabilities									
ong-term borrowings	-	-	1,39,419,04	1,39,419,04			1.39.419.04	1,39,419,0	
nterest accrued but not due on porrowings Current Financial liabilities		14	6,071.09	6,071.09			6,071,09	6,071.0	
Short-term borrowings	1.1	1.2	95.303.21	95,303,21			95.303.21	95,303.2	
rade and other payables other han MSME		-	3,185.65	3,185.65			3,185.65	3,185.6	
Other financial liability	7,121.00	~	57,011.53	64,132,53			64,132.53	64,132.5	
fotal Financial Liabilities	7.121.00		3,00,990.52	3,08,111.52			3,08,111.52	3,08,111.5	
		Carrying an	iount			Fair va	lue		

		1.5				580.4		
31 March 2019	FVTPL	FVTOCI	Amortised Cost	Total	Level 1 - Quoted price In active markets		Level 3 - Significant unobservable inputs	Total
Financial assets								
Non-current financial assets								
- Equity instruments in subsidiaries, joint ventures and associate	<u>^</u>	2,62,313.10		2,62,313.10	~		2,62,313,10	2,62,313,10
- Preference shares		55.548.40	1	55,548,40			55.548.40	55,548,40
- Bonds and debentures	69.397.56	10.636.70	8,140.00	88,174,26			88.174.26	88,174,26
Long-term loans and advances	52.124.48			52,124,48			52,124,48	52,124.48
Other non-current financial assets			0.75	0.75		-	0.75	0.75
Current financial assets								
Current investments	311.42	1.4		311.42	311.42	- Go.	14	311.42
Trade receivables	1.1		1.397.81	1.397.81		1.1	1.397.81	1,397,81
Cash and cash equivalents	~		445,25	445.25			445.25	445.25
Other Bank Balances			444.69	444,69			444,69	444.65
Short-term loans and advances	-		2,349.77	2,349.77	142	1.00	2,349.77	2,349.77
Other financial assets			2,861.19	2,861.19			2,861.19	2,861,19
Total Financial Assets	1,21,833.46	3,28,498.20	15,639.46	4,65,971.12	311.42	94.	4,65,659.70	4,65,971.12
Financial liabilities								
Non-current Financial liabilities								
Long-term borrowings	-		72,460,15	72,460,15			72,460,15	72,460.15
Interest accrued but not due on borrowings Current Financial (labilities		2	11.425.58	11,425.58	-	-	11,425,58	11,425.58
Short-term borrowings	1.1	1	94,685,15	94,685,15			94.685.15	94,685,15
Frade and other payables other han MSME		-	3,458.03	3,458.03		~	3,458.03	3,458.03
Other financial liability	7,122.00		1,17,970.20	1,25,092.20			1,25,092,20	1.25.092.20
	7.122.00		2,99,999,11	3,07,121.11			3.07.121.11	3,07,121.11

R

290

# Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

FVTPL: Fair valued through profit or loss FVTOCI : Fair valued through other comprehensive income B. Measurement of fair values Valuation techniques and significant unobservable inputs The Company has given certain unobservable inputs to the valuer to determine the fair value of each of its investments. The Company has given certain unobservable inputs to the valuer to compute the valuation. Considering micro market and industry scenario the valuer has derived valuation by using appropriate technique for valuation. The Company has accounted fair valuering ain / loss in value of its investments using this report.

The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for financial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used.

Туре	Valuation technique	Significant unobservable inputs	Inter-relationship between significant unobservable inputs and fair value measurement
Infopark Limited )	Discounted cash flow technique: DCF method/analysis is a financial modelling technique based on arplicit assumptions regarding the prospective income arising out of the subject property. In case of a valuation of a large land parcel, where the development potential is realized over a period of time (i.e. time value of money comes into the picture) and also where there are no or few immediate similar properties (i.e. comparable) available for comparison, DCF method considering relevant potential developments of the project is used.	Rent growth every three years (%) 2020: 12% - 18%). (2019: 12% - 18%).     Stabilsed Occupancy (%) 2020: 100 %. (2019: 100%).     Capatalization Rate (%) 2020: 8,5% - 9,0% (2019: 8,5% - 9,0%)     Rent growth every three years (%) 2020:     Rent growth every three years (%) 2020:	Estimated fair value would increase (decrease) if expected lease rent were higher (lower), Estimated fair value would increase (decrease) if expected lease escalation were higher (lower). Estimated fair value would increase if occupancy is lower, Estimated fair value would increase (decrease) if expected lease
Privale Limied)		12% - 18% (2019: 12% - 18%). * Stabilised Occupancy (%) 2020: 94% (2019: 93%). * Capitalization Rate (%) 2020: 8.5% - 9.5% (2019: 9.0% - 10.0%)	escalation ware lower (higher).
Unquoted equity shares (Internationa) Intrabuild Private Limited)		* Rent growth every three years (%) 2020 12% - 18% (2019: 12% - 18%). * Occupancy (%) 2020: 95% (2019: 95%), * Capitalization Rate (%) - Not applicable	
Unquoted equity shares (Mikado Realtors Private Limited)		* Rent growth every three years (%) 2020: 12% - 18% (2019: NA). * Stabilised Occupancy (%) 2020: 94% (2019: NA%). * Capitalization Rate (%) 2020: 8.5% - 9.5% (2019: NA)	
Unquoted equity shares (Gurgaon Realtech Limited / Arrow Infraestate Private Limited / Gurgaon Constructwell Private Limited)		* Rent growth every three years (%) 2020: 12% - 18% (2019: 12% - 18%). * Occupancy (%) 2020: 94% (2019: 95%). * Capitalization Rate (%) 2020: 9.0% - 10.0%. (2019: 9.0% - 10.0%)	
Compulsory Convertible Debentures (Industrial Minerals and Chemicals Private Limited)	Discounted cash flow technique: DCF method/analysis is a financial modelling technique based on explicit assumptions regarding the prospective income arising out of the subject property, in case of a valuation of a large land parcel, where the development potential is realized over a period of time (i.e. time value of money comes into the picture) and also where there are no or few immediate similar properties (i.e. comparable) available for comparison.	Technique - Land Comparison Method has been used for valuation.	Estimated fair value would increase' (decrease) if fair value of land increases' decreases,
nvestments in unlisted corporate			
Debentures- TRIL Roads Private Limited	The value of the plain vanilla debenture is generally estimated as present value of future expected cash flows. The Company has used Monte Carlo Simulation to estimate the fair value of the compulsority convertible debentures and optionally convertible debentures.	2020: * Volatility of share price of comparable companies (57,17% of volatility) (2019-21% )	2019: 'If Equity value of unlisted company increases/(docreases) by 10%, then fair value would (docrease)/increase, 'If there is an increase / (docrease) of 10% volatify in share prices of comparable companies, then the fair value would increase/(docrease)
nvestments in unlisted corporate lebt instruments			(adardisa)
Optionally Convertible Debentures IV Farms Private Limited	The value of the plain vanilla debenture is generally astimated as present value of future expected cash flows. The Company has used Monte Carlo Simulation to estimate the fair value of the compulsority convertible debentures and optionally convertible debentures.	2020: *Volatility of share price of comparable companies : Volatility :(Not applicable (2019 .37.5 %)	2020: 'If Equity value of unlisted company increases/(decreases) by 10%, then fair value would (decrease)/increase. 'If there is an increase / (decrease) of 10% volatility in share prices of comparable companies, then the fair value would increase/(decrease)
vestments in unlisted corporate ebt instruments			
ptionally Convertible Debentures RIL Urban Transport Private imited	present value of future expected cash flows. The Company has used Monte Carlo Simulation to estimate the fair value of the compulsorily convertible debentures and optionally	2020 : * Volatility of share price of comparable companies (59,49% of volatility) (2019 -45 %)	2020 ; 'If Equity value of unlisted company increases/(decreases) by 10%, then lair value would (decrease)/increase. 'If there is an Increase / (decrease) of 10% volatility in share
compulsorily Convertible lebentures- TRIL Intopark imited	convertible debentures.	2020: * Volatility of share price of comparable companies (45.72% of volatility) (2019: 21%)	prices of comparable companies , then the fair value would increase/(decrease).

R

Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

Туре	Valuation technique	Significant	Inter-relationship between
Investments in preference shar	es		
Fully Convertible Prefarance Shares- TRIL Intopark Limited	The value is generally estimated as present value of future expected cash flows. The Company has used Monte Carlo Simulation to estimate the fair value of the compulsority convertible preference and optionally convertible preference shares	2020: • Volatility of share price of comparable companies (45.72% of volatility) (2019: 21%)	2020: <sup>1</sup> H Equity value of unlisted company increases(decreases) by 10%, then fair value would increase/(decrease). <sup>1</sup> If there is an increase / (decrease) of 10% volatify in share prices of comparable companies, then the fair value would decrease.
Fully Convertible Preference Shares- TRIL Constructions Limited	The value is generally estimated as present value of future expected cash flows. The Company has used Monte Carlo Simulation to estimate the fair value of the compulsonly convertible preference and optionally convertible preference shares	Technique - Land Comparison Method has been used for valuation.	2020: "It Equity value of unlisted company increases/(decreases) by 10%, then fair value would (decrease)/increase.
Derivative -Put option		1.000	
Tamil Nadu Industrial Corporation Limited ("TIDCO") Put option	The Company has used Monte Carlo Simulation to estimate the fair value of the options.	2020: * Volatility of share price of comparable companies (48.18% of volatility) (2019-21%)	2020: "If Equity value of unlisted company increases/(decreases) by 10%, then fair value would (decrease)/increase. " If there is an Increase / (decrease) of 10% volatility in share prices of comparable companies , then the fair value would increase/(decrease).
Put / call options			
The Indian Hotels Company Limited ("IHCL") Call option	The Company has used Monte Carlo Simulation to estimate the fair value of the options.	2020: * Volatility of share price of comparable companies (45.44% of volatility) (2019: 21%)	2020: 11 Equity value of unlisted company increases/(decreases) by 10%, then fair value would (decrease)/increase. 11 Ihere is an Increase / (decrease) of 10% volstility in share prices of comparable companies, then the fair value would increase/(decrease).
Amortised cost type items (including security deposits, loans . cash , trade receivables and payables )	Discounted cash flow approach; The valuation model considers the present value of expected payment, discounted using a risk adjusted discount rate.	Not applicable	Not applicable

#### 43 Disclosure in respect of Sale of Residential Flats a) Disclosure with respect to transition adjustme a (a) and too

a)	Disclosure with respect to transition adjustment of IndAS 115	For the year ended 31 March 2019
	the same with the same of the state of the same same	Amount
	Opening Retained Earnings (before Ind AS 115)	(17,728.54)
	Reversal of revenue	(3,174.43)
	Reversal of Cost of sale	1.870.38
	Opening Retained Earning (Atter Ind AS 115)	(19,032.59)
	Increase in Inventory	1.870.38
	Decrease in Trade Receivable (Other than related Party)	(3,174.43)

Particulars	For the year ended 31 March 2020	For the year ended 31 March 2019
Contract revenue recognised as revenue during the year	13,050.27	9,463.27
Particulars	31 March 2020	31 March 2019
Cumulative revenue recognised (2019:net of Ind AS 115 adjustment INR 3,174.42 Lakhs)	1,20,381,48	1,07,331.21
Cumulative costs recognised (2019:net of Ind AS 115 adjustment INR 1,870.38 Lakhs)	84,724.33	72,606.23
Cumulative margins accounted Advances paid	35,657,15	34,724,98
Retention money payable	40.10	83.30 227.54
inancial instruments – Fair values and risk management		
(i) Financial risk management		
The Company has exposure to the following risks arising from financial instruments		
A. Credit risk		
B. Liquidity risk		
C. Market risk		



## Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in takhs)

Risk management framework The Company's board of directors has overall responsibility for the establishment and oversight of the Company's risk management framework. The Company manages market risk through a treasury department, which evaluates and exercises independent control over the entire process of market risk management. The treasury department recommends risk management objectives and policies, which are approved by Board of Directors. The activities of this department include management of cash resources, borrowing strategies, and ensuring compliance with market risk limits and policies.

The Company's risk management policies are established to identify and analyse the risks laced by the Company, to set appropriate risk limits and controls and to monitor risks and adherence to limits. Risk management policies and systems are reviewed regularly to reflect changes in market conditions and the Company's activities. The Company, through its training and management standards and procedures, aims to maintain a disciplined and constructive control environment in which all employees understand their roles and obligations.

The audit committee oversees how management monitors compliance with the company's risk management policies and procedures, and reviews the adequacy of the risk management framework in relation to the risks faced by the Company. The audit committee is assisted in its oversight role by internal audit. Internal audit undertakes both regular and ad hoc reviews of risk management controls and procedures, the results of which are reported to the audit committee.

#### A. Credit Risk

Credit risk is the risk of financial loss to the Company if a customer or counterparty to a financial instrument fails to meet its contractual obligations, and arises principally from the Company's receivables from customers, bans and advances to related parties and investments at amountainer and the receivables from customers. Bans and advances to related parties and investments at amountainer and the receivables from customers. The Company establishing credit limits and continuously monitoring the creditworthiness of customers to which the Company grants credit terms in the normal curse of business. The Company establishes an allowance for doubtful debts and impairment that represents its estimate of incurred losses in respect of trade and other receivables. Ioans and advances and investments.

#### Trade and other receivables

The Company's exposure to credit risk is influenced mainly by the individual characteristics of each customer. The demographics of the customer, including the default risk of the industry and country in which the customer operates, also has an influence on credit risk assessment. Credit risk is managed through credit approvals, establishing credit limits and continuously monitoring the creditworthiness of customers to which the Company grants credit terms in the normal course of business.

#### Credit risk for receivables pertaining to residential business

The risk for trade receivables pertaining to residential ousiness is considered nil as the possession of the residential property is transferred only after the receipt of payment from the customer

Summary of the Company's exposure to credit risk by age of the outstanding from various customers is as follows:

Particulars	31 March 2020	31 March 2019
Past due but not impaired		
Past due 1-90 days	1,411.99	643.12
Past due 91-180 days	33.05	94.48
Past due 181-365 days	9.17	96.93
Past due more than 365 days	424.21	563.28
	1,878.42	1,397.81

#### Cash and cash equivalents

The Company held cash and bank balances with credit worthy banks and linancial institutions of INR 26,000.30 lakhs and INR 890.69 lakhs as at 31 March 2020 and 31 March 2019 respectively. The credit worthiness of such banks and linancial institutions is evaluated by the management on an ongoing basis and is considered to be good.

#### Derivatives

The derivatives are entered into with credit worthy counterparties. The credit worthiness is evaluated by the management on an ongoing basis and is considered to be good.

Security deposits given to lessors The security deposit majorly pertains to rent deposit amounting to INR NIL lakhs and INR 22.64 lakhs as at 31 March 2020 and 31 March 2019 respectively. The Company does not expect any losses from non-performance by theses counter-parties.

#### B. Liquidity risk

Liquidity risk is the risk that the Company will not be able to meet its financial obligations as they become due. The Company manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Company's reputation.

The Company has obtained fund and non-fund based working capital lines from banks, commercial papers issued to Nutual funds and through issue of debentures. The Company also constantly monitors funding options available in the debt and capital markets with a view to maintaining financial flexibility

#### Exposure to liquidity risk

The table below analyses the Company's financial liabilities into relevant maturity groupings based on their contractual maturities for

\* all non derivative financial liabilities \* net and gross settled derivative financial instruments for which the contractual maturities are essential for the understanding of the timing of the cash flows, Contractual cash flows As at 31 March, 2020 Carrying amount Total 1 year or less 1-2 years 2-5 years More than 5 years Non-derivative financial liabilities Commercial paper issued to mutual funds Inter Corporate Deposits 75.303.21 78.000.00 78.000.00 Trade and other payables 3,185.65 3.165.65 3.185.65 Bank overdraft Other financial liabilities 63,082.62 62,934.34 57,001.77 5,932.57 Term Loan from Banks 20.000.00 20.000.00 20.000.00 Non convertible debentures 1 39 419 04 1,79,500.00 10 000 00 79 500 00 67 000 00 3.00,990.52 3,43,619.99 1,98,187.42 78,432.57 67,000.00 **Derivative financial liabilities** Derivatives liabilities at fair 7,121,00 7.121.00 7.121.00 value 7,121.00 7.121.00 7.121.00

#### Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

177 . S &	AT MALE AND A COMPANY		Con	tractual cash flows	Contraction of the second s	
As at 31 March, 2019	Carrying amount	Total	1 year or less	1-2 years	2-5 years	More than 5 years
Non-derivative financial liabili	ties					
Commercial paper issued to mutual funds	76,567.29	77,500.00	77,500.00	-	-	
Inter Corporate Deposits	14,000.00	14,000.00	14,000.00	-		
Trade and other payables	3,458.03	3,458.03	3,458.03	4	-	
Bank overdraft	4,117,86	4,117.86	4,117,86	1.1.1.1.1		
Other financial liabilities	1,29,395.78	1,29,135.88	1,17,710.30	11,425.58	-	
Term Loan from Banks				-	÷	
Non convertible debentures	72,460.15	72,460,15	•	72,460.15	-	
	2,99,999.11	3,00,671.92	2,16,786.19	83,885.73		
Derivative financial liabilities						
Derivatives liabilities at fair value	7,122.00	7,122.00	7.122.00	~		
	7,122.00	7,122.00	7,122.00	240		

The inflows/(outflows) disclosed in the above table represent the contractual undiscounted cash flows relating to derivative financial liabilities held for risk management purposes and which are not usually closed out before contractual maturity. The disclosure shows net cash flow amounts for derivatives that are net cash-settled and gross cash inflow and outflow amounts for derivatives that have simultaneous gross cash settlement.

#### C. Market risk

Market risk is the risk that changes in market prices – such as foreign exchange rates, interest rates and equity prices – will affect the Company's income or the value of its holdings of linancial instruments. The Company is domiciled in India and has its revenues and other transactions in its functional currency i.e. INR. Accordingly the Company is not exposed to any currency risk.

#### (ii) Interest rate risk

(a) interest rate taks is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Company's exposure to market risk for changes in interest rates relates to fixed deposits and borrowings from financial institutions.

#### Exposure to interest rate risk

The interest rate profile of the Company's interest-bearing financial instruments is as follows:

Particulars	31 March 2020	31 March 2019
Zero-rate instruments		
Financial assets	4,55,690,82	3.63.988.20
Financial liabilities	73,389.27	1,39,975,81
Fixed-rate instruments		
Financial assets	1,17,741,19	1.01.982.92
Financial liabilities	2,34,722.25	1,67,145,30
Variable-rate instruments		
Financial assets		
Financial fiabilities		-

#### Interest rate sensitivity - fixed rate instruments

Interest rate sensitivity - fixed rate instruments
The Company does not account for any fixed-rate financial assets or financial liabilities at fair value through profit or loss, and the Company does not have any designate
derivatives (interest rate swaps) except for the following:
1. Financial liability - 0.01% Compulsorily convertible debentures subscribed by Tata Sons Private Limited which are carried at fair value through profit or loss
2. Financial asset - 0.01% Compulsorily convertible debentures invested in TRIL Infopark Limited which are carried at fair value through profit or loss

Since both the instruments are compulsorily convertible in nature, there is no redemption value. Further sensitivity pertaining to risk free rate will not have any impact on fair values due to monte carlo simulation techniques used. Refer Note 42 for valuation techniques used to determine fair value.

Therefore, a change in interest rates at the reporting date would not affect profit or loss for any of these fixed interest bearing financial instruments accounted at fair value through profit or loss

#### Interest rate sensitivity - variable rate instruments

The Company is having only fixed rate borrowings and fixed rate bank deposits which are carried at amortised cost. They are therefore not subject to interest rate risk as defined in IND AS 107, since neither the carrying amount nor the future cash flow will fluctuate because of a change in market interest rates.

45 Capital Management The Company's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. It sets the amount of capital required on the basis of annual business and long-term operating plans which include capital and other strategic investments.

The funding requirements are met through a mixture of equity, convertible debt securities, and other borrowings. The Group's policy is to use short-term and long-term borrowings to meet anticipated funding requirements.

The Company monitors capital using a ratio of "adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total liabilities, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity other than amounts accumulated in the hedging reserve.

Particulars	31 March 2020	31 March 2019
Total liabilities (comprising of interest bearing borrowings and interest accrued thereon)	2,97,656.59	2,96,281.18
Less : Cash and cash equivalent	26,000.30	890.69
Adjusted net debt	2,71,656.29	2.95,390,49
Adjusted equity	2,93,425.62	1,98,685,95
Adjusted net debt to adjusted equity ratio	0.93	1.49

#### 46 Events after the balance sheet date

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 Notes to Ind AS financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

47 World Health Organisation (WHO) declared outbreak of Coronavirus Disease (COVID-19) a global pandemic on March 11, 2020. Consequent to this, Government of India declared lockdown on March 25, 2020 and the Company suspended its operations in all ongoing projects in compliance with the lockdown instructions issued by the Central and State Governments. COVID-19 has impacted the normal business operations of the Company during the lock-down period.

The Company has made detailed assessment of its liquidity position for the next year and the recoverability and carrying value of its assets comprising property, plant and equipment, intangible assets, investments, investory, advances, trade receivables. Deterred faxes, other financial and non-financial assets etc. Based on current indicators of future economic conditions, the Company expects to recover the carrying amount of these assets. The situation is changing rapidly giving rise to inherent uncertainty around the extent and timing of the potential future impact of these OVID-19 parademic, which may be different from that estimated as at the date of approval of these financial statements.

The Central and State Governments have initiated steps to lift the lockdown and the Company will adhere to the same as it resumes its activities. Construction at sites has already restarted. Since it is only about thirteen weeks into the pandemic, the Company will continue to closely observe the evolving scenario and take into account any luture developments arising out of the same.

#### 48 Previous Year's Figures

Previous year figures have been regrouped / reclassified to conform to current half year presentation, wherever considered necessary.

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 utt of Nhubedal NO ma Banmali Agrawala F. N. Subedar Sanjay Dutt Director DIN - 00028428 Chairman DIN - 00120029 Managing Director DIN - 05251670 2 and Sanjay Sharma Sudhakar Shetty Company Secretary Membership No: A13200 **Chief Financial Officer** Mumbai Dated : 06 July 2020



de

# **TATA** REALTY AND INFRASTRUCTURE LTD.

# CONSOLIDATED FINANCIAL STATEMENT FOR THE YEAR 2018-19



Chartered Accountants Lotus Corporate Park 1" Floor, Wing A – G CTS No.185/A, Jay Coach Off Western Express Highway Goregaon (East) Mumbal – 400 063 Maharashtra, India

Tel: ÷91 22 6245 1000 Fax: +91 22 6245 1001

## **INDEPENDENT AUDITOR'S REPORT**

To The Members of Tata Realty and Infrastructure Limited

## Report on the Audit of the Consolidated Ind AS financial statements

## Opinion

We have audited the accompanying consolidated Ind AS financial statements of Tata Realty and Infrastructure Limited ("the Parent") and its subsidiaries, (the Parent and its subsidiaries together referred to as "the Group") which includes the Group's share of loss in its joint ventures, which comprise the Consolidated Balance Sheet as at 31<sup>st</sup> March 2019, and the Consolidated Statement of Profit and Loss (including Other Comprehensive Income), the Consolidated Statement of Cash Flows and the Consolidated Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information (hereinafter referred to as "the consolidated Ind AS financial statements")

In our opinion and to the best of our information and according to the explanations given to us, and based on the consideration of reports of the joint ventures referred to in the Other Matters section below, the aforesaid consolidated Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended ('Ind AS') as amended ("Accounting Standards")], and other accounting principles generally accepted in India, of the consolidated state of affairs of the Group as at 31 March 2019, and their consolidated loss, their consolidated total comprehensive income, their consolidated cash flows and their consolidated changes in equity for the year ended on that date.

## **Basis for Opinion**

We conducted our audit of the consolidated Ind AS financial statements in accordance with the Standards on Auditing specified under section 143 (10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Consolidated Ind AS financial statements section of our report. We are independent of the Group in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the consolidated Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us is sufficient and appropriate to provide a basis for our audit opinion on the consolidated Ind AS financial statements.



Accountants

## Information Other than the Financial Statements and Auditor's Report Thereon

- The Parent's Board of Directors is responsible for the other information. The other information obtained at the date of this auditor's report is included in the Directors Report, but does not include the consolidated Ind AS financial statements and our auditor's report thereon.
- Our opinion on the consolidated Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.
- In connection with our audit of the consolidated Ind AS financial statements, our responsibility is to read the other information, compare with the financial statements of the group and, joint ventures audited by the other auditors, to the extent it relates to these entities and, in doing so, place reliance on the work of the other auditors and consider whether the other information is materially inconsistent with the consolidated Ind AS financial statements or our knowledge obtained during the course of our audit or otherwise appears to be materially misstated. Other information so far as it relates to the group and joint ventures is traced from their financial statements audited by other auditors.
- If, based on the work we have performed, we conclude that there is a material
  misstatement of this other information, we are required to report that fact. We have
  nothing to report in this regards.

## Management's Responsibility for the Consolidated Ind AS financial statements

The Parent's Board of Directors is responsible for the matters stated in section 134(5) of the Act with respect to the preparation of these consolidated financial statements that give a true and fair view of the consolidated financial position, consolidated financial performance including other comprehensive income, consolidated cash flows and consolidated changes in equity of the Group including its joint ventures in accordance with the Ind AS and other accounting principles generally accepted in India. The respective Board of Directors of the companies included in the Group and of its joint ventures are responsible for maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Group and its associates and its jointly controlled entities/ joint ventures and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the financial statements that give a true and fair view and are free from material misstatement, whether due to fraud or error, which have been used for the purpose of preparation of the consolidated Ind AS financial statements by the Directors of the Parent Company, as aforesaid.

SKIN companies included in the Group and joint ventures are responsible for assessing the ability of the Group and of its joint ventures to continue as a going concern, disclosing, as

Chartered

applicable, matters related to going concern and using the going concern basis of accounting unless the management either intends to liquidate or cease operations, or has no realistic alternative but to do so.

The respective Board of Directors of the companies included in the Group and of its joint ventures are also responsible for overseeing the financial reporting process of the Group and of its joint ventures.

# Auditor's Responsibility for the Audit of the Consolidated Ind AS financial statements

Our objectives are to obtain reasonable assurance about whether the consolidated Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these consolidated Ind AS financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

- Identify and assess the risks of material misstatement of the consolidated Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.
- Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under section 143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Parent has adequate internal financial controls system in place and the operating effectiveness of such controls.
- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to events or conditions that may cast significant doubt on the ability of the Group and its joint ventures to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the consolidated Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Group and its associates and joint ventures to cease to continue as a going concern.
- Evaluate the overall presentation, structure and content of the consolidated Ind AS financial statements, including the disclosures, and whether the consolidated Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.

Obtain sufficient appropriate audit evidence regarding the financial information of the pranches, entities or business activities within the Group and its joint ventures to

express an opinion on the consolidated Ind AS financial statements. We are responsible for the direction, supervision and performance of the audit of the financial statements of such branches or entities or business activities included in the consolidated Ind AS financial statements of which we are the independent auditors. For the other branches or entities or business activities included in the consolidated Ind AS financial statements, which have been audited by the branch auditors or other auditors, such branch auditors and other auditors remain responsible for the direction, supervision and performance of the audits carried out by them. We remain solely responsible for our audit opinion.

Materiality is the magnitude of misstatements in the consolidated Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the consolidated Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the consolidated Ind AS financial statements.

We communicate with those charged with governance of the Parent and such other entities included in the consolidated Ind AS financial statements of which we are the independent auditors regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

## **Other Matters**

We did not audit the financial statements two subsidiaries, and whose financial statements reflect total assets of Rs. 1,31,571.12 lakhs as at 31st March, 2019, total revenues of Rs. 61,815.73 lakhs and net cash inflows amounting to Rs.1,694.01 lakhs for the year ended on that date, as considered in the consolidated financial statements. The consolidated financial statements also include the Group's share of net loss of Rs. 599.14 lakhs for the year ended 31<sup>st</sup> March, 2019, as considered in the consolidated financial statements have not been audited by us. These financial statements have been audited by other auditors whose reports have been furnished to us by the Management and our opinion on the consolidated financial statements, in respect of two joint ventures and our report in terms of subsection (3) of Section 143 of the Act, in so far as it relates to the aforesaid group and joint ventures and associates is based solely on the reports of the other auditors.

Our opinion on the consolidated Ind AS financial statements above and our report on Other Legal and Regulatory Requirements below, is not modified in respect of the above matters with respect to our reliance on the work done and the reports of other auditors.

# Report on Other Legal and Regulatory Requirements

1. As required by Section 143(3) of the Act, based on our audit and on the consideration Chartered of the reports of the branch auditors and other auditors on the separate financial

KINS

Chartered

statements/ financial information of the subsidiaries, joint ventures referred to in the Other Matters section above we report, to the extent applicable that:

- a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit of the aforesaid consolidated Ind AS financial statements.
- b) In our opinion, proper books of account as required by law relating to preparation of the aforesaid consolidated Ind AS financial statements have been kept so far as it appears from our examination of those books, returns and the reports of the other auditors.
- c) The reports on the accounts of the branch offices of the Companies included in the Group audited under Section 143(8) of the Act by subsidiaries and jont venture auditors have been sent to us other auditors and have been properly dealt with by us in preparing this report.
- d) The Consolidated Balance Sheet, the Consolidated Statement of Profit and Loss including Other Comprehensive Income, the Consolidated Statement of Cash Flows and the Consolidated Statement of Changes in Equity dealt with by this Report are in agreement with the relevant books of account maintained for the purpose of preparation of the consolidated Ind AS financial statements.
- e) In our opinion, the aforesaid consolidated financial statements comply with the Ind AS specified under Section 133 of the Act.
- f) On the basis of the written representations received from the directors of the Parent as on 31st March, 2019 taken on record by the Board of Directors of the Company and the reports of the statutory auditors of its, subsidiary companies, and joint venture companies incorporated in India, none of the directors of the Group companies, its associate companies and joint venture companies incorporated in India is disqualified as on 31st March, 2019 from being appointed as a director in terms of Section 164 (2) of the Act.
- g) With respect to the adequacy of the internal financial controls over financial reporting and the operating effectiveness of such controls, refer to our separate Report in "Annexure A" which is based on the auditors' reports of the Parent, subsidiary companies, joint venture companies incorporated in India. Our report expresses an unmodified opinion on the adequacy and operating effectiveness of internal financial controls over financial reporting of those companies.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of section 197(16) of the Act, as amended, In our opinion and to the best of our information and according to the explanations given to us, the remuneration paid by the Parent to its directors during the year is in accordance with the provisions of section 197 of the Act.
- i) With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended in our opinion and to the best of our information and according to the explanations given to us:

- The consolidated Ind AS financial statements disclose the impact of pending litigations on the consolidated financial position of the Group, and joint ventures.
- ii) Provision has been made in the consolidated Ind AS financial statements, as required under the applicable law or accounting standards, for material foreseeable losses, if any, on long-term contracts including derivative contracts;
- iii) There were no amounts which were required to be transferred to the Investor Education and Protection Fund by the Parent/ Holding Company, and its subsidiary companies, associate companies and jointly controlled companies/ joint venture companies incorporated in India.

## For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No.117366W/W-100018)

Auchla

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai, 08 May 2019

## ANNEXURE "A" TO THE INDEPENDENT AUDITOR'S REPORT

# (Referred to in paragraph g under 'Report on Other Legal and Regulatory Requirements' section of our report of even date)

## Report on the Internal Financial Controls Over Financial Reporting under Clause (i) of Sub-section 3 of Section 143 of the Companies Act, 2013 ("the Act")

In conjunction with our audit of the consolidated Ind AS financial statements of **Tata Realty and Infrastructure Limited** ("the Parent"/ "the Company") as of and for the year ended 31<sup>st</sup> March, 2019, we have audited the internal financial controls over financial reporting of the Parent and its subsidiary companies, which are companies incorporated in India, its jointly controlled companies, which are companies incorporated in India, as of that date.

## Management's Responsibility for Internal Financial Controls

The respective Board of Directors of the Parent, its subsidiary companies and jointly controlled companies, which are companies incorporated in India, are responsible for establishing and maintaining internal financial controls based on the internal control over financial reporting criteria established by the respective Companies considering the essential components of internal control stated in the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting issued by the Institute of Chartered Accountants of India (ICAI). These responsibilities include the design, implementation and maintenance of adequate internal financial controls that were operating effectively for ensuring the orderly and efficient conduct of its business, including adherence to the respective company's policies, the safeguarding of its assets, the prevention and detection of frauds and errors, the accuracy and completeness of the accounting records, and the timely preparation of reliable financial information, as required under the Companies Act, 2013.

## Auditor's Responsibility

4

Accountants

Our responsibility is to express an opinion on the internal financial controls over financial reporting of the Parent, its subsidiary companies and its jointly controlled companies, which are companies incorporated in India, based on our audit and based on the consideration of reports of the other auditors on separate Ind AS financial statements of the subsidiaries and joint ventures referred to below in the Other Matters paragraph. We conducted our audit in accordance with the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting (the "Guidance Note") issued by the Institute of Chartered Accountants of India and the Standards on Auditing, prescribed under Section 143(10) of the Companies Act, 2013, to the extent applicable to an audit of internal financial controls. Those Standards and the Guidance Note require that we comply with ethical requirements and plan and perform the audit to obtain reasonable assurance about whether adequate internal financial controls over financial reporting was established and maintained and if such controls operated effectively in all material respects.

Our audit involves performing procedures to obtain audit evidence about the adequacy of the internal financial controls system over financial reporting and their operating effectiveness. Our audit of internal financial controls over financial reporting included obtaining an understanding of internal financial controls over financial reporting, assessing skithe tisk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. The procedures Chartered

selected depend on the auditor's judgement, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error.

We believe that the audit evidence we have obtained and the audit evidence obtained by the other auditors of subsidiary companies and jointly controlled companies, which are companies incorporated in India, in terms of their reports referred to in the Other Matters paragraph below, is sufficient and appropriate to provide a basis for our audit opinion on the internal financial controls system over financial reporting of the Parent, its subsidiary companies and its jointly controlled companies, which are companies incorporated in India.

## Meaning of Internal Financial Controls Over Financial Reporting

A company's internal financial control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal financial control over financial reporting includes those policies and procedures that:

- 1. pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company;
- provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorisations of management and directors of the company; and
- provide reasonable assurance regarding prevention or timely detection of unauthorised acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

## Inherent Limitations of Internal Financial Controls Over Financial Reporting

Because of the inherent limitations of internal financial controls over financial reporting, including the possibility of collusion or improper management override of controls, material misstatements due to error or fraud may occur and not be detected. Also, projections of any evaluation of the internal financial controls over financial reporting to future periods are subject to the risk that the internal financial control over financial reporting may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

## Opinion

In our opinion to the best of our information and according to the explanations given to us and based on the consideration of the reports of the other auditors referred to in the Other Matters paragraph below, the Parent, its subsidiary companies and jointly controlled companies, which are companies incorporated in India, have, in all material respects, an adequate internal financial controls system over financial reporting and such internal financial controls over financial reporting were operating effectively as at 31<sup>st</sup> March, 2019, based on the criteria for internal financial control over financial reporting established by the respective companies considering the essential components of internal control stated in the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting issued by the Institute of Chartered Accountants of India.



## **Other Matters**

Our aforesaid report under Section 143(3)(i) of the Act on the adequacy and operating effectiveness of the internal financial controls over financial reporting insofar as it relates to two subsidiary companies and two jointly controlled companies, which are companies incorporated in India, is based solely on the corresponding reports of the auditors of such companies incorporated in India.

Our opinion is not modified in respect of the above matters.

## For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No.117366W/W-100018)

Keinchita

Kalpesh J. Mehta Partner Membership No. 48791

Mumbai, 08 May 2019

Consolidated Balance Sheet as at 31 March 2019 (Currency: Indian rupees in lakhs)

	ETS	Note	31 March 2019	31 March 20
I Non-	-current assets			
(a)	Property, plant and equipment	3	3,906	3,60
(b)	Capital work-in-progress	5	121	11
(c)		8	71	8
(d)	Investment property	4	362,000	367,66
(e)		6	141,468	112,81
(f)		72	21,216	21,96
(g)		7		and the second se
(h)		8	224,165	143,35
()		8	45,354	37,65
344	i) Investments in Joint Ventures			1222
	(ii) Other Investments	9	73,130	63,29
		10	23	2
VA	(iii) Other Financial assets	11	1,677	1,66
(1)		12	6,456	7,24
(k)		13	10,889	9,57
(1)	Other non-current assets	14	26,275	29,72
Non	-current assets		916,751	798,78
2 Curr	rent assets			
(a)	Inventories	15	43,718	44,43
(6)	Financial assets		-1741023.	16,45,77
200	(i) Trade and other receivables	16	2,846	5.78
	(ii) Cash and cash equivalents	17	5,921	714
	(iii) Bank balances other than (ii) above	18	1,579	34
	(iv) Other Investments	19	1,383	795
	(v) Short-term loans and advances	20	10 PS 18 57	
	(vi) Other Financial assets		3,054	5,02
14	A STATE OF THE ACCOUNT OF THE ACCOUN	21	1,981	1.32
100	Other current assets	22	5,388	7,42
Curi	rent assets	-	65,870	79.42
	TOTAL ASS	=	982,621	878,21
	JITY AND LIABILITIES			
1 Equi	The second			
(a)	Equity share capial	23	101,731	101,73
(b)	Other equity	24	(78,836)	(53,87
	Equity attributable to owners		22,895	47.85
(c)	Non-controlling interests		6,617	6,82
152		-	29,512	54,68
	-Current Liabilities			
2 Non-				
	Financial liabilities			
	Financial liabilities (i) Borrowings	25	544,488	580.33
	(i) Borrowings		544,488	580,32
(1)	<ul><li>(i) Borrowings</li><li>(ii) Other financial liabilities</li></ul>	26	103,165	89.26
(s) (b)	<ul> <li>(i) Borrowings</li> <li>(ii) Other financial liabilities</li> <li>Long-term provisions</li> </ul>	26 27	103,165 3,260	89.20 2,23
(x) (b) (c)	<ul> <li>(i) Borrowings</li> <li>(ii) Other financial liabilities</li> <li>Long-lerm provisions</li> <li>Deferred tax liabilities (Net)</li> </ul>	26 27 28	103,165 3,260 1,584	89.20 2,2 4
(s) (b)	<ul> <li>(i) Borrowings</li> <li>(ii) Other financial liabilities</li> <li>Long-lerm provisions</li> <li>Deferred tax liabilities (Net)</li> </ul>	26 27	103,165 3,260 1,584 11,440	89.20 2,2 4: 11,5-
(x) (b) (c) (d)	(i) Borrowings (ii) Other financial liabilities Long-lerm provisions Deferred tax liabilities (Net) Other non-current liabilities	26 27 28	103,165 3,260 1,584	580,33 89,20 2,23 45 11,54 683,83
(x) (b) (c) (d) 3 Curr	(i) Borrowings (ii) Other financial liabilities Long-lerm provisions Deferred tax liabilities (Net) Other non-current liabilities rent liabilities	26 27 28	103,165 3,260 1,584 11,440	89.20 2,2 45 11,54
(i) (b) (c) (d) 3 Curr	(i) Borrowings (ii) Other financial liabilities Long-lerm provisions Deferred tax liabilities (Net) Other non-current liabilities rent liabilities Financial liabilities	26 27 28 29	103,165 3,260 1,584 11,440 663,937	89.24 2,21 41 11,5- 683,81
(i) (b) (c) (d) 3 Curr	(i) Borrowings (ii) Other financial liabilities Long-term provisions Deferred tax liabilities (Net) Other non-current liabilities rent liabilities Financial liabilities () Short term borrowings	26 27 28 29 30	103,165 3,260 1,584 11,440 663,937 100,574	89.2( 2,2) 4: 11,5- 683,8) 61,2'
(i) (b) (c) (d) 3 Curr	(i) Borrowings (ii) Other financial liabilities Long-term provisions Deferred tax liabilities (Net) Other non-current liabilities rent liabilities Financial liabilities () Short term borrowings (i) Trade and other payables	26 27 28 29 30 31	103,165 3,260 1,584 11,440 663,937 100,574 9,468	89.24 2.2: 4: 11,5- 683,8: 61.2: 10,7:
(s) (b) (c) (d) 3 Curr (a)	(i) Borrowings (ii) Other financial liabilities Long-term provisions Deferred tax liabilities (Net) Other non-current liabilities Financial liabilities () Short term borrowings (i) Trade and other payables (iii) Other financial liabilities	26 27 28 29 30 31 32	103,165 3,260 1,584 11,440 663,937 100,574 9,468 169,725	89.2: 4: 11,5: 683,8: 61,2: 10,7: 59,2:
(s) (b) (c) (d) 3 Curr (a) (b)	(i) Borrowings (ii) Other financial liabilities Long-term provisions Deferred tax liabilities (Net) Other non-current liabilities Financial liabilities () Short term borrowings (i) Trade and other payables (ii) Other financial liabilities Other current liabilities	26 27 28 29 30 31 32 33	103,165 3,260 1,584 11,440 663,937 100,574 9,468 169,725 7,354	89 2( 2,2) 4( 11,5- 683,8) 61,2' 10,7' 59,2' 630
(s) (b) (c) (d) 3 Curr (a) (b) (c)	(i) Borrowings (ii) Other financial liabilities Long-term provisions Deferred tax liabilities (Net) Other non-current liabilities Financial liabilities () Short term borrowings (i) Trade and other payables (ii) Other financial liabilities Other current liabilities Short term provisions	26 27 28 29 30 31 32	103,165 3,260 1,584 11,440 663,937 100,574 9,468 169,725	89.2; 2,2; 4; 11,5; 683,8; 61,2; 10,7; 59,2; 6,3;
(s) (b) (c) (d) 3 Curr (a) (b) (c)	(i) Borrowings (ii) Other financial liabilities Long-term provisions Deferred tax liabilities (Net) Other non-current liabilities Financial liabilities () Short term borrowings (i) Trade and other payables (ii) Other financial liabilities Other current liabilities	26 27 28 29 30 31 32 33	103,165 3,260 1,584 11,440 663,937 100,574 9,468 169,725 7,354 263 1,788	89 2: 2,2 4: 11,5 <b>683,8</b> : 61,2 10,7 59,2 6,3 1 1,3
(s) (b) (c) (d) 3 Curr (a) (b) (c)	(i) Borrowings (ii) Other financial liabilities Long-term provisions Deferred tax liabilities (Net) Other non-current liabilities Financial liabilities () Short term borrowings (i) Other financial liabilities Other current liabilities Short term provisions Current tax liabilities (Net)	26 27 28 29 30 31 32 33 34 35	103,165 3,260 1,584 11,440 663,937 100,574 9,468 169,725 7,354 263	89.20 2,2 45 11,54
(s) (b) (c) (d) 3 Curr (a) (b) (c)	(i) Borrowings (ii) Other financial liabilities Long-term provisions Deferred tax liabilities (Net) Other non-current liabilities Financial liabilities () Short term borrowings (i) Trade and other payables (ii) Other financial liabilities Other current liabilities Short term provisions	26 27 28 29 30 31 32 33 34 35	103,165 3,260 1,584 11,440 663,937 100,574 9,468 169,725 7,354 263 1,788	89.24 2,23 45 683,83 61,25 683,83 61,25 10,75 59,25 6,37 11 1,39
(1) (b) (c) (d) 3 Curr (1) (4) (5) (c) (d)	(i) Borrowings (ii) Other financial liabilities Long-term provisions Deferred tax liabilities (Net) Other non-current liabilities Financial liabilities () Short term borrowings (i) Other financial liabilities Other current liabilities Short term provisions Current tax liabilities (Net)	26 27 28 29 30 31 32 33 34 35	103,165 3,260 1,584 11,440 663,937 100,574 9,468 169,725 7,354 263 1,788 289,172	89.24 2.22 43 683,83 61.24 10,72 59,22 6,33 11 1,39 139,7

The notes referred above form an integral part of these consolicated Ind AS financial statements As per our report of even date attached

## For DELOITTE HASKINS & SELLS LLP

Chartered Accountants

(Frm's Registration No. 117366W/W-100018)

la mman Agrain

Kalpesh J. Mehta Partner

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

Rajiv Sabharval Director pupeda

DIN - 00057333

Cinay Gaokar

Company Secretary Membership No: A6 20

> Mumbai May 8, 2019

Mumbai May 8, 2019 S. Santhanakrishnan Director DIN - 00032049

Banmalijagrawala

Chairman DIN - 00120029

Neera Saggi Director DIN - 00501029

UN

Sanjay Dutt

Managing Director DIN - 05251670

306

0 1 Sanjay Sharma Chief Financal Officer

F. N. Subedar

DIN - 00028428

Director

Consolidated Statement of Profit and Loss for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

		Ncie	31 March 2019	31 March 2018
1	Revenue from operations	36	167,816	127,862
11	Other income	37	_6,616	3,395
111		Total income (1+11)	174,432	131,257
1V	Expenses			
	Cost of flats sold	38	6,122	4,497
	Construction costs	39	\$3,095	57,023
	Employee benefit expenses	4)	8,259	8,229
	Finance costs	41	52,055	54,471
	Depreciation and amortization expense	3/4/8	16,858	17,600
	Other expenses	4.2	28,555	28,650
		Total expenses (EV)	194,944	170,470
v	Loss before exceptional items, share of profits (net) and tax (III- $tV$	)	(20,512)	(39,213)
	Add: Share of (loss) from associates			
	Add Share of (loss) from joint ventures		(633)	(1,023)
VI	Loss before exceptional items and tax	-	(21,145)	(40,236)
VII	Exceptional items		·	
VIII IX	Loss before tax (V2-VII) Tax Expenses		(21,145)	(40,236)
	Current tax		864	799
	Deferred tax charge / (credit)		\$,906	(1,007)
	Current tax expenses relating to prior years		-	262
х	Loss for the year (VIII-IX)		(23,915)	(40,290)
	Less Share in loss transferred to non-controlling interest		(222)	(92)
XI	Loss for the year attributable to owners		(23,693)	(40,198)
XII	Other Comprehensive Income			
	A (i) Items that will not be reclassified to profit or loss			
	Remeasurements of defined benefit hability (asset)		55	35
	Equity instruments fair valued through OCi		•	•
	focome tax relating to items that will not be reclassified to profit	t or loss	(14)	(9)
	B (ii) Items that will be reclassified to prafit or loss			
	Income tax relating to items that will be reclassified to profit or 1		-	
	Share of other comprehensive income of associates, joint ventur	res accounted for using equaty method	-	-
хщ	Total Comprehensive income for the year (XI+XII)		(23,652)	(40,172)
	(Comprising Loss and Other Comprehensive Income for the year after n	on-controlling interest)		
XIV	Earnings per equity share . (Face Value per share Rs 10 each)			
	(1) Basic	42.	(2.33)	(3.95)
	(2) Diluted		(2,33)	(3.95)
Signi	icant accounting policies	2		
	to the consolidated lpd AS financials statements	3-60		

The notes referred above form an integral part of these consolidated Ind AS financial statements As per our report of even date attached

#### For DELOITTE HASKINS & SELLS LUP

Chartered Accommants (Firm s Registration No 117366W/W-100018)

ď a

Kalpesh J. Mehta Partner

Mombai May 8, 2019

awal. hanmali Agriswala Chairman DIN - 00120029

alla S. Sauthanakrishnan Director

DIN - 00032049

For and on behalf of the Board of Directors of Tata Resity and Infrastructure Limited CIN No: U70102MH2007PLC168300

hill F. N. Subedar Director DIN - 00028428

(or)

Sanjay Sharma

Chief Financial Officer

14

1

Sanjay Dutt

Managing Director

8

DIN - 95251670

wer

Rajiv Saturnal Director DJN - 00057333

1 0 Vinate Szokas

Company Secretary Membership No A6120

Mumbai May 8, 2019

Po

DIN - 00501029

Neera Saggi

Director

## Consolidated statement of cash flows as at 31 March 2019

(Currency, Indian rupees in lakhs)

A	Cash flows from operating activities :	31 March 2019	31 March 2018
	(Loss) before tax	(21,145)	(40,236)
	Adjusted for	(()	(10,000)
	Depreciation and amortisation	16,858	17,600
	Excess provision written back	(177)	
	(Gam) on sale of current investments	(273)	(161)
	Provision for Major Maintenance	1,011	931
	Mark to Market Gain on Forward / Derivative Contracts	(3,329)	1,732
	Gain on re-valuation of Investments Provision for doubtful and bad debts	(100)	(118)
	Advance written off	322	94
	Interest income	4,050	16
	Finance costs (including MTM on forward contract)	(1,953) 52,055	(1,912)
	Share of profit (loss) from joint ventures and associates	52,055 633	54,471 1,022
	Sundry Liabilities/ Provision written off	-	1,022
	Profit on sale of fixed assets		(124)
	Impairment loss	750	5,219
	Provision for Employee Benefits	156	113
	Reversal of provision for compensated		-
	Loss on sale of propety, plant and equipment	31	41
		70,034	79,046
	Operating profit before working capital changes	48,889	38,810
	Changes to working capital		
	Decrease in Trade Receivables	6,315	1,004
	Decrease in Inventories	714	8,850
	Decrease in Loans & Advances, Other Financial Assets and Other Current Assets	7,995	2,681
	Decrease in Trade payables	(1,254)	(2,066)
	Increase / (decrease) in Other financial habilities, current and non current liabilities and provision	(2,769)	8,030
	—	6,001	18,500
	Cash flows generated from operations	54,890	57,309
	Taxes paid (net of refund received)	(2,390)	(1,808)
	Net cash flows generated from operating activities	52,500	55,501
в	Cash flows from investing activities :		
	Payment for purchase and construction of property plant and equipment	(83,018)	(64,518)
	Proceeds on sale of fixed assess Purchase of investment property / investment property under construction	103	304
	Investment in joint ventures	(38,022)	(25,688)
	Proceeds from sale of investments in mutual funds	(10,470) 152,375	(7,981) 35,185
	Investment in in mutual funds	(145,428)	(24,256)
	Investment in fixed deposits (net) under hen	(1,234)	(274)
	Acquisition of subsidiaries joint ventures	-	(24,122)
	Interest received	2,192	1,752
	Net cash flows (used in) investing activities	(123,502)	(109,598)
С	Cash flows from financing activities ;		
	Proceeds / (Repayment) from short term borrowings (net)	39,278	(132,172)
	Proceeds them long term borrowings	125,585	412,380
	(Repayment) of long term borrowings	(36,479)	(185,575)
	Proceeds from Minority Interest	13	-
	Finance costs paid Net cash generated from financing activities	(58,615) 69,782	<u>(52,354)</u> 42,279
	Net (decrease) in cash and bank balances (A+B+C)	(1,220)	41.00
	Cash and bank balances, beginning of the year	7,141	(11,818) 18,714
	Increase in cash and eash equivalents on account of acquisition of subsidiary	-	245
	Cash and bank balances, end of year (refin upte 17)	5,921	7,146
			r yarren





Consolidated statement of cash flows as at 31 March 2019 (Continued) (Currency, Indian rupees in lakhs)

Notes:	
- D	

<ol> <li>Cash and bank balances include the following :</li> </ol>			
Cash and cash equivalents		31 March 2019	31 March 2018
Cash balance		93	72
Balance with scheduled banks:			
- in current accounts		2,099	4,034
<ul> <li>in deposit accounts</li> </ul>		3,729	3,035
		5,921	7,141
New Sec. of New Sec.	-		
nificant accounting policies	2		

Signi Notes to the consolidated Ind AS financial statements

3-60

For and on behalf of the Board of Directors of

The notes referred above form an integral part of these consolidated Ind AS financial statements As per our report of even date attached

For DELOITTE HASKINS & SELLS LLP Chartered Accountants (Firm's Registration No. 117366W/W-100018)

Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 N hubedar Davis Salaharment la Q murali A M Kalpesh J. Mehta Banmali Agrawala Sanjay Dutt Chairman DIN - 00120029 Managing Director DIN - 05251670 Director DIN - 00057333 Director DIN - 00028428 ¢ avive 0 U.S Ą 0 Vinay Gaokar Neera Saggi Sanjay Sharma s Director Director Chief Financial Officer Company Secretary DIN - 00032049 DJN - 00501029 Membership No. A6120 Mumbai May 8, 2019

Munbaj May 8, 2019

v t

Partner

Consolidated statement of changes in equity as at 31 March 2019 Tata Realty and infrastructure Limited

(Currency Indian rupees in lakhs)

(a) Equity share capital Balance as at I April

Changes in equity share capital during the year Balance as at 31 March

101,731 101,731 2

101,731

101,731

31-Mar-18

31-Mar-18

Amount

Amount

Notes

	2
1	ĩ
	ĩ
,	ě
1	Š
1	÷

	Re	Reserves and surphy	lus	Other our	Other cumpreliensive		No-	
Particulars				inc	income	Total ether	controlline	Total
	Capital	Securities	Retained	EVOCI -	Defined	equity	interests	-
	reserve	Prenium	Earnings	equity	benefit plan			
				instrum Ends	adjustment			
Balance as at 1 A prd 2017	6,457	15,769	(35,158)	806	(01)	(12,136)	6.550	(5,586)
Loss for the year	•	,	(40,198)	'		(40,198)	(92)	(40,290)
Reversal of capital reserve during the year	(1:576)	(ð)	•	,		(1,576)	•	(1,576)
Remeasurements of defined benefit	1		•	•	3.5	35		35
Issues of New Shares / CCD in Subsidiaries	,		ı	,	ı	•	368	368
Other comprehensive income for the year	•		,		(6)	(6)	'	(6)
Consolidation Adjustments		'	6	•	,	6		4
FVOCI transfer to Retained Earnings	•	ı	805	(908)	•	1		,
Balance as al 31 March 2018	4,911	15,769	(74,541)	'	(1+5)	(53,875)	6,826	(660,74)
Loss for the year		'	(23,693)	•	•		(222)	(23,915)
Remeasurements of defined benefit	•	•	ı		55	55		55
Issues of New Shares / CCD in Subsidiaries	•	•	,		,	•	13	[]
Income tax relating to items that will not be reclassified to profit or loss	I	•	•	•	(14)	(14)		(14)
Consolidation Adjustments	I		(5)	•	•	(s)		(S)
Transition adjustment due to application of DVD AS 115	•	*	(1, 304)	-		(1,,104)		(1,304)
Balance as at 31 March 2019	4,911	15.769	(99,543)	4	27	(78,836)	6.617	(72,220)
Significant accounting policies	5							
Notes to the concelled and AS facencials statements	1.60							

Significant accounting policies Notes to the consolidated Ind AS financials statements

The notices referred above form an integral part of these comodificanci ind AS feasarcial statements As per our report of even date attached

For DELOUTTE HASKINS & SEILS LLP

(Firm's Registration No. 117366W/W-100018) Chartered Accountants

g Kalpesh J. Mehra Partner

Mumbai May 8, 2019

Ar

Tata Realty and Infrastructure Limited CBN No. U70102MB12007PLC168300 Rajiv Sabharwal DIN - 00057333 Company Secretary CHnay-Gantar Ser and a ~ MMMON IN Sulle Director F. N. Subedar DBN - 00028428 iñjay Sharma hief Financial Officer Sanjay Dutt Manayug Director DIN - 05251670 Director Neera Sal some hours Chickman S. Santhanakrishnan. Banmali Agewala Director DTN - 00120029 aura 2

Director

2 anno 19

For and on behalf of the Board of Directors of

30

Mumbai May 8, 2019

Membership No. A6120

DIN - 00501029

DfN - 00032049

### Notes to the consolidated Ind AS financial statements

for the year ended 31 March 2019

(Currency: Indian rupees in lakbs)

#### 1 Background and basis of preparation

#### 1A Background

Tata Realty & Infrastructure Limited ("TRIL," or 'the Group') along with its subsidiaries, joint ventures and associates, collectively referred to as the "the Group' is engaged in the business of investment advisory services, project management consultancy services and real estate and infrastructure development. The parent company is a wholly owned subsidiary of Tata Sons Private Limited.

#### 1B Basis of preparation

#### (a) Statement of compliance

These Consolidated Ind AS financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules. 2015 untified under Section 133 of the Companies Act, 2013 (the Act) and other relevant provisions of the Act,

#### (b) Going Concern

As at March 31, 2019, the Group is having .

- accumulated losses of INR 99,543 Lakhs.
- · current habilities exceeding its current assets by INR 104,502 Laklis.
- repayment of INR, 1,18,800 Lakhs pertaining to the current portion of the long term debt (NCD) due in next 12 months
- · cash losses during the current year and previous year

All of the above may indicate doubt about the Group's ability to continue as a going concern.

The Board of Directors of the parent company have assessed going concern ability considering the following factors

 Negative working capital is on account of management decision to borrow short-term funds through commercial papers considering benefits of interest arbitrage under current economic scenario

• The parent company has credit ratings of AA and A1+ from two credit rating agencies which supports raising of long-term funds or refinance short-term loans on a need basis respectively. Based on the credit rating, the parent company has a long-term borrowing limit of INR 80,000 lakhs and short-term borrowing limit of INR 1.00,000 lakhs. The parent company has a term sheet from a bank for long term funding up to INR 1,00,000 lakhs.

• The Board of Directors has approved rights issue of INR 3,60,000 laklis approximately through equity capital from the parent i.e. Tata Sons Private Limited. The Board of Directors are confident that the same will be fully subscribed and thereby the Authorized Capital is realigned to accommodate the above rights issue

• The Board of Directors of the parent company are actively considering disposal of certain projects which are not strategically important to the business and yield positive cashiflows in the next 12 months.

Hence in the opinion of the Board of Directors, the Group is able to continue as going concern. Accordingly, the financial statements have been prepared on a going concern basis.

#### (c) Functional and presentation currency

These Consolidated funncial statements are presented in Indian Rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian Rupess (INR) and all amounts have been rounded-off to the nearest lakhs with two decimals, except for share data and as otherwise stated. Due to rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.

#### (d) Basis of measurement

The Consolidated and AS financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value.

- Certain financial assets and liabilities (including derivative instruments)
- 2 Defined benefit plans plan assets measured at fair value

#### (e) Critical accounting judgements and key sources of estimation of uncertainity

In preparing these Consilidated Ind AS financial statements, management has made judgements, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates Estimates and underlying assumptions are reviewed on an ongoing basis. Revisions to accounting estimates are recognised prospectively.

#### (i) Judgements

Information about judgements made in applying accounting policies that have the most significant effects on the amounts recognised in the Ind AS financial statements is included in the following notes.





#### Notes to the consolidated Ind AS financial statements

for the year ended 31 March 2019

(Currency Indian rupees in lakhs)

#### (ii) Control

If the Group owns less than one-half of the voting power and it is able to control the company which inter-alta provides the Group with power to appoint majority of the board of directors and power over relevant activities. Consequently, the Group consolidates its investment in the company.(Refer Note No. 52(a))

If the Group owns more than one-half of the voting power but n does not have control over the Companies then the Group has joint control over the companies. In other words, decisions about the relevant activities - i.e. those that significantly affect the returns of the arrangement - require the affirmative consent of the investors. Consequently, the Group has classified its interest in these companies as jointly controlled entity. (Refer Note no. 52(b))

#### (iii) Assumptions and estimation uncertainties

Information about assumptions and estimation uncertainties that have a significant risk of resulting in a material adjustment in the year ended 31 March 2019 is included in the following notes:

Note - implaiment test of non-financial assets: Rey assumptions underlying recoverable amounts including the recoverability of expenditure on acquired intangible assets (goodwill); Usefial life of Property, plant and equipment

Note - recognition of deferred tax assets, availability of future taxable profit against which tax losses carried forward can be used;

Note - recognition and measurement of provisions and contingencies key assumptions about the likelihood and magnitude of an outflow of resources

#### Note measurement of defined benefit obligations key actuarial assumptions;

Note acquisition of subsidiary, fair value of the consideration transferred (including contingent consideration) and fair value of the assets acquired and habilities assumed, measured on a provisional basis; and

Note - determining the fair value of investments on the basis of significant unobservable mpins,

Note - impairment of fmancial assets.

#### (f) Measurement of fair values

The Group's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Group has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the CFO.

They regularly review significant unobservable uputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, methoding the level is the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorized into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

Level 1 quoted prices (unadjusted) in active markets for identical assets or liabilities.

Level 2 inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

Level 3 inputs for the asset or liability that are not based on observable market data (unobservable inputs)

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorized in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement.

#### (g) Foreign currency transactions

Foreign currency transactions are translated into the functional currency using the exchange rates at the dates of the transactions. Foreign exchange gains and losses resulting from the settlement of such transactions and from the translation of monetary assets and liabilities denominated in Joreign currencies at year end exchange rates are generally recognised in profit or loss.

Non-monetary items that are measured at fair value in a foreign currency are translated using the exchange rates at the date when the fair value was determined. Translation differences on assets and liabilities carried at fair value are reported as part of the fair value gain or loss. The gain or loss arising on translation of nonmonetary items measured at fair value is treated in line with the recognition of the gain or loss on the change in fair value of the item (i.e., translation differences on items whose fair value gain or loss is recognised in OC1 or profit or loss are also recognised in OC1 or profit or loss, respectively).

#### 2.01 Basis of consolidation

#### (i) Subsidiaries

Subsidiaries are all entities (including structured entities) over which the group has control. The group controls an entity when the group is exposed to, or has rights to, variable returns from its involvement with the entity and has the ability to affect those returns through its power to direct the relevant activities of the entity. Subsidiaries are fully consolidated from the date on which control is transferred to the group. They are deconsolidated from the date that control ceases

The group combines the financial statements of the parent and its subsidiaries line by fine idding together fike neuro of assets, liabilities, equity, income and expenses. InterGroup transactions, balances and unrealised gains on transactions between group comparises are eliminated. Chrealised losses are also eliminated unless the transaction provides evidence of an impairment of the transferred asset. Accounting policies of subsidiaries have been changed where necessary to ensure consistency with the policies adopted by the group.

Non-controlling interests in the results and equity of subsidiaries are shown separately in the consolidated statement of profit and loss, consolidated statement of changes in equity and balance sheet respectively





## Notes to the consolidated Ind AS financial statements

for the year ended 31 March 2019

#### (Currency Indian rupees in lakhs) (ii) Joint arrangements

Under Ind AS 111 Joint Arrangements, investments in joint arrangements are classified as either joint operations or joint ventures. The classification depends on the contractual rights and obligations of each investor, rather than the legal structure of the joint arrangement. The group has only joint ventures

(iii) John ventures

Joint ventures are entities over which the group has joint control along with another entity. Interests in joint ventures are accounted for using the equity method, after initially being recognized at cost in the consolidated balance sheet. The investment is initially recognized at cost, and the carrying amount is increased or decreased to recognize the investor's share of the profit or loss of the investee after the acquisition date.

#### (iv) Associates

Associates are entities over which the group has significant influence but not control. Investments in associates are accounted for using the equity method of accounting. The investment is initially recognized at cost, and the carrying amount is increased or decreased to recognize the investor's share of the profit or loss of the investee after the acquisition date.

#### 2.02 Segment reporting

Operating segments are reported in a manner consistent with the internal reporting provided to the chief operating decision maker (CODM). Chief operating decision maker's function is to allocate the resources of the entity and access the performance of the operating segment of the Group.

The Managing Director assesses the financial performance and position of the Group and makes strategic decisions and is identified as being the chief operating decision maker for the Group. Refer note 46 for segment information presented

#### 2.03 Revenue recognition

#### (1) Sale of completed property

Revenue from sale of completed property (residential and commercial) is recognised when

1 The Company has transferred to the huyer signicificant risk and rewards of ownership of the completed property.

2 The Company retains neither continuing managerial involvement to the degree usually associated with the ownership nor effective control over the completed property sold;

- 3. The amount of revenue can be measured reliably:
- 4 It is probable that the economic henefit associated with the transaction will flow to the Company, and
- 5 Cost incurred or to be incurred in respect of the transaction can be measured reliably

#### (ii) Service concession arrangements

Concession arrangements are recognized in accordance with Appendix C of Iral AS 115, Service Concession Arrangements It is applicable to concession arrangements comprising a public service obligation and satisfying all of the following criteria.

the concession granter controls or regulates the services to be provided by the operator using the asset, the infrastructure, the beneficiaries of the services and prices applied;

the granter controls the significant residual interest in the infrastructure at the end of the term of the arrangement

Pursuant to Appendix C of Ind AS 115, such infrastructures are not recognized in assets of the operator as property, plant and equipment but in financial assets ("financial asset model") and/or intangible assets ("intangible asset model") depending on the remuneration community given by the granitor

The intangible asset model applies where the operator is paid by the users or where the concession grantor has not provided a contractual guarantee in respect of the recoverable amount. The intangible asset corresponds to the right granted by the concession grantor to the operator to charge users of the public service in remuneration of concession vervices.

Intangible assets resulting from the application of Appendix C of Ind AS 115 are recorded in the financial statements as intangible assets. The Group accounts for such intangible asset (along with the present value of committed payments towards concession arrangement to the grantor at the appointed date e.g. Negative Grant, premium etc.) in accordance with the previsions of Ind AS 38 and is amortized based on projected fraffic count or revenue, as detailed in Note 2.87(m), taking into account the estimated period of commercial operation of infrastructure which generally on a straight-me basis, over the contract term

#### Under the intangable assist model, revenue includes

revenue recorded on a completion basis for assets and infrastructure under construction (in accordance with IND AS 115); charges collected from users

However, in certain concession arrangements, contracts may include a payment commitment on the part of the concession grantor covering only part of the investment, with the balance covered by amount charged to users. Where this is the case, the investment amount guaranteed by the concession grantor is recognized under the financial asset model and the residual balance is recognized under the intangible asset model. Financial assets resulting from that are recorded in the financial statements under the heading other financial assets and recognized at amounted cost.

Firstiver, where infrastructive is partly regulated and partly unregulated, the portion of infrastructure that is physically separable and capable of being operated independently and meets definition of easing generating unit as defined under find AS 36 is analysis separately if it is used wholly for unregulated purposes

#### (iii) Rendering of services

Asset management fees and Project management consultancy fees are recognized in accordance with terms of agreement with customers.

#### (iv) Rental income

Rental income from investment property is recognised as part of revenue from operations in profit or loss on a straight-line basis over the term of the lease except where the rentals are structured to increase in line with expected general inflation. Lease incentives granted are recognised as an integral part of the total rental income, over the term of the lease. Rental moone, over the term of the lease. Rental moone from sub-leasing is also recognised in a similar manner and included under other moone.





#### Notes to the consolidated Ind AS financial statements

for the year ended 31 March 2019

(Currency Indian rupees in lakhs)

(v) loterest and dividend:

A dividend is recognized as revenue when the right to receive payment has been established. For all debt instruments measured either at amortised cost or at fair value through other comprehensive income, interest income is recorded using the effective interest rate (EIR) EIR is the rate that exactly discounts the estimated future cash payments or receipts over the expected life of the financial instrument or a shorter period, where appropriate, to the gross carrying amount of the financial asset or to the amortised cost of a financial liability.

#### 2.04 Government grants

Grants from the government are initially recognised as deferred income at their fair value where there is a reasonable assurance that the grant will be received and the group will comply with all attached conditions, they are then recognised in profit or loss as other income on a systematic basis over the useful life of the asset

Grants that compensate the Group for expenses incurred are recognised in profit or loss on a systematic basis in the periods in which the expenses are recognised

### 2.05 Property, plant and equipment

#### (i) Recognition and measurement

Freehold land is carried at historical cost All other items of property, plant and equipment are measured at cost less accumulated depreciation and any accumulated impairment losses

The cost of an item of property, plant and equipment comprises.

- a) its purchase price, including import duties and non-refundable purchase taxes, after deducting trade discounts and rebates.
- b) any costs directly attributable to bringing the asset to the location and condition necessary for it to be capable of operating in the manner intended by

c) the initial estimate of the costs of dismantling and removing the item and restoring the site on which it is located, the obligation for which an entity incurs either when the item is acquired

Income and expenses related to the incidental operations, not necessary to bring the item to the location and condition necessary for it to be capable of operating in the manner intended by management, are recognized in profit or loss.

If significant parts of an item of property, plant and equipment have different useful lives, then they are accounted for as separate items (major components) of property, plant and equipment and equipment and equipment and equipment are recognised in profit or loss

#### (ii) Subsequent expenditure

Subsequent expenditure is capitalised only if it is probable that the future economic benefits associated with the expenditure will flow to the Group.

#### 2.85 Property, plant and equipment (Continued)

#### (iii) Depreciation

Depreciable amount for assets is the cost of an asset, or other amount substituted for cost, less its estimated residual value.

Depreciation is provided using the straight line method in the manner and at the rates prescribed by Part 'C' of Schedule II of the Act, Depreciation is charged on a montily pro-rate basis for assets purchased or sold during the year.

In the following cases, the useful life is less than the corresponding useful life presenbed in Part 'C' of Schedule II of the Act, based on internal technical evaluation, taking into account the nature of the assets, the estimated usage of the asset, the operating conditions of the asset, past history of replacement, anticipated technological changes etc.

Plant & Machinery	8-15 years
Furniture & Fixture	6-10 years
Office equipments	3-6 years
Computers	3-4 years
Motor Vehicles	5-7 years
Commercial building	5-32 years

Leasehold land and leasehold improvements are amortised over the lease period

Deprectation methods, useful lives and residual values are reviewed at each reporting date and adjusted if appropriate.

Gains and losses on disposals are determined by comparing proceeds with carrying amount. These are included in profit or loss within other gains! (losses)

#### 2.06 Investment property

Property that is held for long-term rental yields or for capital appreciation or both, and that is not occupied by the group, is classified as investment property investment property is measured initially at its cost, including related transaction costs and where applicable borrowing costs. Subsequent expenditure is capitalised to the asset's carrying amount only when it is probable that future occurring benefits associated with the expenditure will flow to the group and the cost of the item can be measured reliably. All other replaced maintenance costs are expensed when incurred. When part of an investment property is replaced, the carrying amount of the replaced part is derecogoused.

#### Investment property under construction

Property that is being constructed for future use as investment property is accounted for as investment property under construction until construction or development is complete. All costs which are directly attributable to construction of the investment property are capitalized,





## Notes to the consolidated Ind AS financial statements

for the year ended 31 March 2019

(Currency Indian rupces in lakhs)

#### 2.07 Intangible assets

#### Goodwill

Goodwill on acquisitions of subsidiaries is included in intangible assets. Goodwill is not amortized but it is tested for impairment annually. It is carried at cost less accumulated impairment losses. Gains and losses on the disposal of an entity include the carrying amount of goodwill relating to the entity sold.

Goodwill is allocated to cash-generating units for the purpose of impairment testing. The allocation is made to those cash-generating units or groups of cash-generating units that are expected to benefit from the business combination in which the goodwill arose. The units or groups of units are identified at the lowest level at which goodwill is monitored for internal management purposes, which in our case are the operating segments.

Intangible assets comprise application software purchased, which are not an integral part of the related hardware, and are amortized on a written down value basis over a period of ten years, which in Management's estimate represents the period during which the economic benefits will be derived from their use

Subsequent expenditure is capitalized only when it increases the future economic benefits embodied in the specific to which it relates

#### Intangible assets under development

#### i) Passenger ropeway facility

Intangible assets comprise a right to charge users for passenger ropeway facility, which is currently under development. The cost of intangible assets acquired in exchange of monetary or non-monetary assets is measured at fair value of le exchange transaction lacks commercial substance or (b) the fair value of neither the asset received nor the asset given up is reliably measurable. If acquired assets cannot be measured at fair value, its cost is measured at the carrying amount of the asset given up. The fair value of the asset is measured with reference to the fair value of construction services provided

Subsequent expenditure is capitalized only when it increases the future economic benefits embodied in the specific to which a relates

#### (ii) Toll collection right

Toll collection rights representing right to receive fees from users of facility are obtained in consideration for rendering construction, operation and maintenance services in relation to building and maintenance of the project on Build. Operate and Transfer basis. The consideration received in a service concession arrangement is measured at fair value upon initial recognition, i.e. construction cost. Subsequent to initial recognition the intangible asset is measured at cost, which includes capitalised borrowing costs, less accumulated amortisation and accumulated impairment lesses

#### Amortization of concession intangible assets

Amortization of intangible asset under SCA. The intangible rights relating to infrastructure assets, which are recognized in the form of right to charge users of the infrastructure asset are amortized by taking proportionate of actual traffic count for the period over total projected traffic count from project to cost of intangible assets, i.e. proportionate of actual traffic for the period over total projected traffic count from the infrastructure assets are amortized by taking proportionate of actual traffic count for the period over total projected traffic count from the infrastructure assets expected to be carned over the balance concession period as estimated by the management. However with respect to total projected revenue from the project over the balance concession period as estimated of traffic count. Total projected revenue / traffic count is reviewed at the end of each financial year and is adjusted to reflect any charges in the estimates which lead to the actual collection at the end of the concession period.

#### 2.08 Impairment of non-financial assets

Goodwill and intangible assets that have an indefinite useful life are not subject to amortisation and are tested annually for impairment, or more frequently if events or changes in circumstances indicate that they might be impaired. Other assets are tested for impairment whenever events or changes in circumstances indicate that the carrying amount may not be recoverable. Management periodically assesses using, external and internal sources, whether there is an indication that an asset may be impaired.

The recoverable amount is higher of the asset's net selling price or value in use, which means the present value of future cash flows expected to arise from the continuing use of the asset and its eventual disposal. An impairment loss for an asset is reversed if, and only if, the reversal can be related objectively to an event occurring after the impairment loss was recognized. The carrying amount of an asset is increased to its revised recoverable amount, provided that this amount does not exceed the carrying amount that would have been determined (net of any accumulated amortization or depreciation) had no impairment loss been recognized for the asset in prior years,

#### 2,09 Borrowing cost

General and specific borrowing costs that are directly attributable to the acquisition, construction or production of a qualifying asset are capitalised during the period of time that is required to complete and prepare the asset for its intended use or sale. Qualifying assets are assets that necessarily take a substantial period of time to get ready for their intended use or sale.

Investment income earned on the temporary investment of specific borrowings pending their expenditure on qualifying assets is deducted from the borrowing costs eligible for capitalisation

Other borrowing costs are expensed in the period in which they are incurred

#### 2.10 Leases

- (i) Determining whether an arrangement contains a lease
  - An arrangement, which is not in the legal form of a lense, should be accounted for as a lease, if
  - a) fulfilment of the arrangement is dependent on the use of a specific asset or assets (the asset), and
  - b) the attangement conveys a right to use the asset
  - At inception of an arrangement, the Group determines whether the arrangement is or contains a lease.

At inception of on reassessment of an arrangement that contains a lease, the Group separates payments and other consideration required by the arrangement into those for the lease and those for other elements





## Notes to the consolidated Ind AS financial statements

for the year ended 31 March 2019

(Currency Indian rupees in lakhs)

(ii) Finance lease

Agreements are classified as finance leases, if substantially all the risks and rewards incidental to ownership of the leased asset is transferred to the lessee.

Assets given under finance leases are recognised as a receivable at an amount equal to the net investment in the lease. Finance income is allocated over the lease term on a systematic and rational basis. This income allocation is based on a pattern reflecting a constant periodic return on the net investment in the finance lease.

Minimum lease payments, for assets taken under finance lease, are apportioned between the finance expense and the reduction of the outstanding liability. The finance expense is allocated to each period during the lease term so as to produce a constant periodic rate of interest on the remaining balance of the hability.

#### (iii) Operating lease

Agreements which are not classified as finance leases are considered as operating lease.

Payments made under operating leases are recognised in profit or loss. Lease incentives received are recognised as an integral part of the total lease expense, over the lease term. Lease payments under an operating lease are recognised as an expense on a straight line basis over the lease term unless the payments to lessor are structured to increase in line with expected general inflation to compensate for the lessor's expected inflationary cost increases.

Payments received under operating leases are recognised in profit or loss. Lease incentives received are recognised as an integral part of the total lease income, over the lease term. Lease receipts under an operating lease are recognised as income on a straight line basis over the lease term unless the receipts from the lessee are structured to increase in line with expected general inflation to compensate for the Group's expected inflationary cost increases.

#### 2.11 Income-tax

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised directly in equity or in OCI

(i) Current tax

Current tax comprises the expected tax payable or receivable on the taxable income or loss for the year and any adjustment to the tax payable or receivable in respect of previous years. It is measured using tax rates enacted or substantively enacted at the reporting date. Current tax also includes any tax arising from dividends. Management periodically evaluates positions taken in tax returns with respect to situations in which applicable tax regulation is subject to interpretation. It establishes provisions where appropriate on the basis of amounts expected to be paid to the tax authorities.

Minimum Alternate Tax ('MAT') under the provisions of Income-tax Act, 1961 is recognized as current tax in the statement of profit and loss MAT paid in accordance with the tax laws, which gives future economic benefits in the form of adjustment to future income tax liability, is considered as an asset if there is a convincing evidence that the Group will pay normal tax. Accordingly, MAT is tecognized as an asset in the balance sheet when it is probable that the future economic benefit associated with it will flow to the Group.

- Current tax assets and liabilities are offset only if, the Group
- a) has a legally enforceable right to set off the recognised amounts, and
- b) intends either to settle on a net basis, or to realise the asset and settle the hability simultaneously
- (ii) Deferred tax

Deferred tax is recognised in respect of temporary differences between the carrying amounts of assets and habilities for financial reporting purposes and the amounts used for taxation purposes. However, deferred tax liabilities are not recognised if they arise from the taitial recognition of goodwill. Deferred income tax is also not accounted for if it arises from initial recognition of an asset or liability in a transaction other than a business combination that at the time of the transaction affects neither accounting profit nor taxable profit (tax loss).

Deferred tax assets are recognised for unused tax losses, unused tax credits and deductible temporary differences to the extent that it is probable that future taxable profits will be available against which they can be used. Deferred tax assets are reviewed at each reporting date and are reduced to the extent that it is no longer probable that the related tax benefit will be realised, such reductions are reviewed when the probability of future taxable profits improves. Unrecognised deferred tax assets are reassessed at each reporting date and recognised to the extent that it has become probable that future taxable profits will be available against which they can be used.

Deferred tax is measured at the tax rates that are expected to be applied to temporary differences when they reverse, using tax rates enacted or substantively coacted at the reporting date and are expected to apply when the related deterred income tax asset is realised or the deferred income tax fieldly is settled.

Deferred tax habilities are not recognised for temporary differences between the carrying amount and tax bases of investments in subsidianes, branches and associates and interest in joint arrangements where the group is able to control the timing of the reversal of the temporary differences and it is probable that the differences will not reverse in the foreseeable future.

Deferred tax assets are not recognised for temporary differences between the carrying amount and tax bases of investments in subsidiaties, branches and associates and interest in joint arrangements where it is not probable that the differences will reverse in the foreseeable future and taxable profit will not be available against which the temporary difference can be utilised.

Deferred tax assets and liabilities are offset only if:

- a) the entity has a legally enforceable right to set off current tax assets against current tax habilities, and
- b) the deferred tax assets and the deferred tax liabilities relate to income taxes levied by the same taxation authority on the same taxable entity.





Notes to the consolidated Ind AS financial statements for the year ended 31 March 2019

#### (Currency: Indian rupees in lakhs)

#### 2.12 Business combinations

The acquisition method of accounting is used to account for all business combinations, regardless of whether equity instruments or other assets are acquired. The consideration transferred for the acquisition of a subsidiary comprises the.

- fair values of the assets transferred;
- habilities incurred to the former owners of the acquired business,
- equity interests issued by the group; and
- Fair value of any asset or liability resulting from a contingent consideration arrangement

Identifiable assets acquired and liabilities and contingent liabilities assumed in a business combination are, with limited exceptions, measured initially at their fair values at the ocquisition date. The group recognises any non-controlling interest in the acquired entity on an acquisition-by-acquisition basis either at fair value or at the non-controlling interest's proportionate share of the acquired entity's net identifiable assets. Acquisition-related costs are expensed as incurred,

#### The excess of the

## consideration transferred;

amount of any non-controlling interest in the acquired entity, and

Acquisition-date fair value of any previous equity interest in the acquired entity over the fair value of the net identifiable assets acquired is recorded as goodwill if those amounts are less than the fair value of the net identifiable assets of the business acquired, the difference is recognised in other comprehensive income and accumulated in equity as capital reserve provided there is clear evidence of the underlying reasons for classifying the business combination as a bargain purchase in other cases, the bargain purchase gain is recognised directly in equity as capital reserve.

Where settlement of any part of cash consideration is deferred, the amounts payable in the future are discounted to their present value as at the date of exchange. The discount rate used is the entity's incremental borrowing rate, being the rate at which a similar borrowing could be obtained from an independent financier under comparable terms and conditions.

Contingent consideration is classified either as equity or a financial liability. Amounts classified as a financial liability are subsequently remeasured to fait value with changes in fair value recognised in profit or loss

If the business combination is achieved in stages, the acquisition date carrying value of the acquirer's previously held equity interest in the acquiree is remeasured to fair value at the acquisition date. Any gains or losses arising from such remeasurement are recognised in profit or loss or other comprehensive income, as appropriate,

#### 2.13 Cash and cash equivalents

For the purpose of presentation in the statement of cash flows, cash and cash equivalents includes cash on hand, deposits held at call with financial institutions, other short-term, highly liquid investments with original maturities of three months or less that are readily convertible to known amounts of cash and which are subject to an insignificant risk of changes in value, and bank overdrafts. Bank overdrafts are shown within borrowings in current liabilities in the balance sheet.

#### 2.14 Inventories

Direct expenses like land cost, development rights, site labour cost, material used for project construction, cost of borrowing, project management consultancy, costs for moving the plant and machinery to the site and general expenses incurred specifically for the respective project like insurance, design and technical assistance, and construction overheads are taken as the cost of the project work in progress and cost of unsold flats

#### Material at site comprise of building material, components, stores and spares,

Inventories are valued at lower of cost or net realizable value, cost is determined on weighted average basis. Net realizable value is the estimated selling price in the ordinary course of business, less estimated costs of completion and estimated costs necessary to make the sale

#### 2.15 Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity. Financial instruments also include derivative contracts such as foreign currency foreign exchange forward contracts, interest rate swaps and currency options; and embedded derivatives in the host contract,

#### (i) Financial assets

#### Classification

The Group shall classify financial assets as subsequently measured at amortised cost, fair value through other comprehensive mcome (FVOCI) or fair value through profit or loss (FVTPL) on the basis of its business model for managing the financial assets and the contractual cash flow characteristics of the financial asset.

#### Initial recognition and measurement

All financial assets are recognised initially at fair value plus, in the case of financial assets not recorded at fair value through profit or loss, transaction costs that are attributable to the acquisition of the financial asset. Purchases or sales of favancial assets hat require delivery of assets within a time frame established by regulation or convention in the market place (regular way trades) are recognised on the trade date, i.e., the date that the Group commits to purchase or sell the asset.





#### Notes to the consolidated Ind AS financial statements for the year ended 31 March 2019

in the year ended of march 201

#### (Currency, Indian rupees in lakhs)

### Debt instruments

§ A 'debt instrument' is measured at the amortised cost if both the following conditions are met

a) The asset is held within a business model whose objective is to hold assets for collecting contractual cash flows, and

b) Contractual terms of the asset give rise on specified dates to eash flows that are solely payments of principal and interest (SPPI) on the principal amount outstanding.

§ After initial measurement, such financial assets are subsequently measured at amortised cost using the effective interest rate (EIR) method. Amortised cost is calculated by taking into account any discount or premain and fees or costs that are an integral part of the EIR. The EIR amortisation is included in finance income in the profit or loss.

§ Debt instruments included within the fair value through profit and loss (FVTPL) category are measured at fair value with all changes recognized in the statement of profit and loss.

#### Equity instruments

§ The group subsequently measures all equity investments in companies other than equity investments in, joint ventures and associates at fair value. Where the group's management has elected to present fair value gains and losses on equity investments in other comprehensive income, there is no subsequent reclassification of fair value gains and losses to profit or loss. Dividends from such investments are recognised in profit or loss as other income when the group's right to receive payments is established.

#### De-recognition

§ A financial asset (or, where applicable, a part of a financial asset or part of a Group of similar financial assets) is primarily derecognized (i.e. removed from the Group's balance sheet) when:

§ The rights to receive cash flows from the asset have expired, or

§ The Group has transferred us rights to receive eash flows from the asset or has assumed an obligation to pay the received cash flows in full without material delay to a third party under a 'pass-through' arrangement; and either (a) the Group has transferred substantially all the risks and rewards of the asset, or (b) the Group has neither transferred nor retained substantially all the risks and rewards of the asset, but has transferred control of the asset

When the Group has transferred its rights to receive cash flows from an asset or has entered into a pass-through arrangement, it evaluates if and to what extent it has retained the risks and rewards of ownership. When it has neither transferred nor retained substantially all of the risks and rewards of the asset, nor transferred control of the asset, the Group continues to recognise the transferred asset to the extent of the Group's continuing involvement. In that case, the Group also recognises an associated liability. The transferred asset and the associated liability are measured on a basis that reflects the rights and obligations that the Group has retained.

§ Continuing involvement that takes the form of a guarantee over the transferred asset is measured at the lower of the original carrying amount of the asset and the maximum amount of consideration that the Group could be required to repay.

#### Impairment of financial assets

In accordance with Ind-AS 109, the Group applies expected credit loss (ECL) model for measurement and recognition of impairment loss on the following funancial assets and credit risk exposure,

a) Financial assets that are debt instruments, and are measured at amortised cost e.g., loans, debt securities, deposits, and bank balance

- b) Lease receivables
- c) Trade receivables

The Group follows 'samplified approach' for recognition of impairment loss allowance on

§ Trade receivables which do not contain a significant financing component

§ All lease receivables resulting from transactions.

The application of simplified approach does not require the Group to track changes in credit risk. Rather, it recognises impairment loss allowance based on lifetime ECLs at each reporting date, right from its initial recognition

For recognition of impairment loss on other funancial assets and risk exposure, the Group determines that whether there has been a significant increase in the credit risk since initial recognition. If credit risk has not increased significantly, 12-month ECL is used to provide for impairment loss. However, if credit risk has increased significantly, lifetime ECL is used. If, in a subsequent period, credit quality of the instrument improves such that there is no longer a significant increase in credit risk since initial recognition, then the entity reverts to recognising impairment loss allowance based on 12-month ECL.

#### 2.15 Financial Instruments (Countined)

#### Derivative contracts

The group enters into a variety of derivative financial instruments to manage its exposure to interest rate and foreign exchange rate risk, including foreign exchange forward contracts, interest rate swaps and cross currency swaps. Derivatives are initially recognised at fair value at the date the derivative contracts are entered into and are subsequently remeasured to their fair value at the end of each reporting period. The resulting gain/loss is recognised in profit or loss immediately.





### Notes to the consolidated Ind AS financial statements

for the year ended 31 Morch 2019

(Currency Indian rapees in lakhs)

#### (li) Financial liabilities

The Group classifies all financial liabilities as subsequently measured at amorused cost, except for financial habilities at fair value through profit or loss. Such habilities, including derivatives that are liabilities, shall be subsequently measured at fair value

#### Initial recognition and measurement

Financial fiabilities are classified, at initial recognition, as financial liabilities at fair value through profit or loss, loans and borrowings, payables, or as derivatives designated as hedging instruments in an effective hedge, as appropriate

All financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly autobutable transaction costs

The Group's financial habilities include trade and other payables, loans and borrowings including bank overritails, financial guarantee contracts and derivative financial instruments

#### Financial liabilities at fair value through profit or loss

Financial liabilities at fair value through profit or loss include financial liabilities held for trading and financial liabilities designated upon initial recognition as at fair value through profit or loss. Financial liabilities are classified as held for trading if they are incurred for the purpose of repurchasing in the near term. This category also includes derivative financial instruments entered into by the Group that are not designated as hedging instruments in hedge relationships as defined by Ind-AS 109. Separated embedded derivatives are also classified as held for trading unless they are designated as effective hedging instruments.

#### Gains or losses on habilities held for trading are recognised in the profit or loss.

Pinancial liabilities designated upon initial recognition at fair value through profit or loss are designated at the initial date of recognition, and only if the criteria in Ind-AS 109 are satisfied. For liabilities designated as FVTPL, fair value gains/losses attributable to changes in own credit risk are recognized in OCI. These gains/loss are not subsequently transferred to profit and loss. However, the Group may transfer the cumulative gain or loss within equity. All other changes in fair value of such liability are recognised in the statement of profit or loss. The Group has not designated any financial liability as at fair value through profit and loss.

#### Loans and borrowings

After initial recognition, interest-bearing loans and borrowings are subsequently measured at amortised cost using the EIR method. Gains and losses are recognised in profit or loss when the liabilities are derecognized

Amortised cost is calculated by taking into account any discount or premium on acquisition and fees or costs that are an integral part of the EIR. The EIR amortisation is included as finance costs in the statement of profit and loss

#### This category generally applies to interest-bearing loans and borrowings

Preference shares, which are mandatorily redeemable on a specific date, are classified as liabilities. The dividends on these preference shares are recognised in profit or loss as finance costs.

The fait value of the liability portion of an optionally convertible bonds is determined using a market interest rate for an equivalent non-convertible bonds. This amount is recorded as a liability on an amortised cost basis until extinguished on conversion or redemption of the bonds. The remainder of the proceeds is attributable to the equity portion of the compound instrument. This is recognised and included in shareholders' equity, net of income tax effects, and not subsequently remeasured.

Where the terms of a financial liability are renegotiated and the entity issues equity instruments to a creditor to extinguish all or part of the liability (debt for equity swap), a gain or loss is recognized in profit or loss, which is measured as the difference between the carrying amount of the financial liability and the fair value of the equity instruments issued.

#### Derecognition

A financial liability is derecognised when the obligation under the hability is discharged or cancelled or expires. When an existing financial liability is replaced by another from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit or loss.

#### Embedded derivatives

If the hybrid contract contains a host that is a financial asset within the scope Ind-AS 109, the Group does not separate embodded derivatives. Rather, it applies the classification requirements contained in Ind AS 109 to the entire hybrid contract. Derivatives embedded in all other host contracts are accounted for as separate derivatives and recorded at fair value if their economic characteristics and risks are not closely related to those of the host contracts are accounted for as contracts are not held for trading or designated at fair value through profit or loss. These embedded derivatives are measured at fair value with changes in fair value contract that significantly modifies the cash flows.

#### Offsetting of financial instruments

Financial assets and liabilities are offset and the net amount is reported in the halance sheet where there is a legally enforceable right to offset the recognised amounts and there is an intermon to settle on a net basis or realise the asset and settle the hability simultaneously. The legally enforceable right must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the group or the counterparty.

#### Derivative financial instruments

The Group uses derivative financial instruments, such as foreign exchange forward contracts and interest rate swaps to manage its exposure to interest rate and foreign exchange risks. For contracts where fieldge accounting is not followed, such derivative financial instruments are initially recognised at fair value on the date on which a derivative contract is entered uno and are subsequently re-measured at fair value through profit or loss account. Derivatives are carried as financial assets when the fair value is positive and as financial liabilities when the fair value is negative





## Notes to the consolidated 1nd AS financial statements

for the year ended 31 March 2019

(Currency, Indian rupces in lakhs)

#### 2.16 Employee benefits

#### (i) Shurt-term employee benefits

Short-term employee benefits are expensed as the related service is provided. A liability is recognised for the amount expected to be paid if the Group has a present legal or constructive obligation to pay this amount as a result of past service provided by the employee and the obligation can be estimated reliably

#### (ii) Compensated absences

The liabilities for carned leave and sick leave are not expected to be settled wholly within 12 months after the end of the period in which the employees render the related service. They are therefore measured as the present value of expected future payments to be made in respect of services provided by employees up to the end of the reporting period using the projected unit credit method. The benefits are discounted using the market yields at the end of the reporting period that have terms approximating to the terms of the related obligation. Re-measurements as a result of experience adjustments and changes in actuanal assumptions are recognised in profit or loss.

#### (iii) Defined contribution plans

Obligations for contributions to defined contribution plans are expensed as the related service is provided. Prepaid contributions are recognised as an asset to the extent that a cash refund or a reduction in future payments is available.

#### (iv) Defined benefit plans

The Group's net obligation in respect of defined benefit plans is calculated separately for each plan by estimating the amount of future benefit that employees have earned in the current and prior periods, discounting that amount and deducting the fair value of any plan assets

The calculation of defined benefit obligations is performed annually by a qualified actuary using the projected unit credit method. When the calculation results in a potential asset for the Group, the recognised asset is limited to the present value of economic benefits available in the form of any future refunds from the plan or reductions in future contributions to the plan. To calculate the present value of economic benefits, consideration is given to any applicable minimum funding requirements.

Re-measurement of the net defined benefit liability, which comprise actuarial gains and losses, the return on plan assets (excluding interest) and the effect of the asset ceiling (if any, excluding interest), are recognised immediately in other comprehensive income (OCI). Net interest expense (income) on the net defined liability (assets) is computed by applying the discount rate, used to measure the net defined liability (asset), to the net defined liability (asset) is computed by applying the discount rate, used to measure the net defined liability (asset), to the net defined liability (asset) as the start of the financial year after taking into account any changes as a result of contribution and benefit payments during the year. Net interest expense and other expenses related to defined benefit plans are recognised in profit or loss.

When the benefits of a plan are changed or when a plan is curtailed, the resulting change in benefit that relates to past service or the gain or loss on curtailment is recognised immediately in profit or loss. The Group recognises gains and losses on the settlement of a defined benefit plan when the settlement occurs.

#### (v) Other long-term employee benefits

The Group's net obligation in respect of long-term employee benefits is the amount of future henefit that employees have earned in rotum for their service in the current and prior periods. That benefit is discounted to determine its present value. Re-measurement are recognised in profit or loss in the period in which they arise

#### 2.17 Provisions

Long-term provisions are determined by discounting the expected future cash flows specific to the fiability. The unwinding of the discount is recognised as finance cost.

Provisions are measured at the present value of management's best estimate of the exponditure required to settle the present obligation at the end of the reporting period. The discount rate used to determine the present value is a pre-tax rate that reflects currant enacket assessments of the time value of money and the risks specific to the highlity. The merease in the provision due to the passage of time is recognized as interest expense.

#### Onerous contracts

A provision for onerous contracts is measured at the present value of the lower of the expected cost of terministing the contract and the expected net cost of continuing with the contract. Before a provision is established, the Group recognises any impairment loss on the assets associated with that contract

#### 2.18 Earnings per share

Basic carrings per skene is computed by dividing the profit / (loss) after tax by the weighted average munitier of equity shares outstanding during the year. The weighted average number of equity shares outstanding during the year is adjusted for the events for bonus issue, bonus element in a rights issue to existing shareholders, share split and reverse share split (consolidation of shares).

Deluted earnings per share is computed by dividing the profit / (loss) after tax as adjusted for dividend, interest and other charges to expense or income (net of any attributable taxes) relating to the dilutive potential equity shares, by the weighted average number of equity shares considered for deriving basic earnings per share and the weighted average number of equity shares which could have been issued on conversion of all dilutive potential equity shares.





### Notes to the consolidated Ind AS financial statements

for the year ended 31 March 2019

#### (Currency Indian rupees in lakhs)

#### 2.19 Recent Indian Accounting Standards (Ind AS)

Ministry of Corporate Affairs ("MCA"), through Companies (Indian Accounting Standards) Amendment Rules, 2019 and Companies (Indian Accounting Standards) Second Amendment Rules, has notified the following new and amendments to Ind ASs which the Group has not applied as they are effective from April 1, 2019.

#### tnd AS - 116

Ind AS 116 will replace the existing leases standard, Ind AS 17 Leases. Ind AS 116 sets out the principles for the recognition, measurement, presentation and disclosure of leases for both lessees and lessors. It introduces a single, on-balance sheet lessee accounting model for lessees. A lessee recognises right-of-use asset representing its right to use the underlying asset and a lease liability representing its obligation to make lease payments. The standard also contains enhanced disclosure requirements for lessees. In dAS 116 substantially earnies forward the lessor accounting requirements in Ind AS 17. The Group will adopt Ind AS 16, effective annual reporting period beginning April 1, 2019.

#### Ind AS 12 Income taxes (amendments relating to income tax consequences of dividend and uncertainty over income tax treatments)

The amendment relating to income tax consequences of dividend clarify that an entity shall recognise the income tax consequences of dividends in profit or loss, other comprehensive income or equity according to where the entity originally recognised those past transactions or events. The Company does not expect any impact from this pronouncement,

#### Ind AS 109 - Prepayment Features with Negative Compensation

The amendments rolate to the existing requirements in Ind AS 109 regarding termination rights in order to allow measurement at amortised cost (or, depending on the business model, at fair value through other comprehensive income) even in the case of negative. The Compensation payments. The Group does not expect this amendment to have any impact on its financial statements.

#### Ind AS 19 - Plan Amendment, Curtailment or Settlement

The amendments clarify that if a plan amendment, curtailment or settlement occurs, it is mandatory that the current service cost and the net interest for the period after the re-measurement are determined using the assumptions used for the re-measurement. In addition, amendments have been included to clarify the effect of a plan amendment, curtailment or settlement on the requirements regarding the asset ceiling. The Group does not expect this amendment to have any significant impact on its financial statements.

#### Ind AS 23 - Borrowing Costs

The amendments clarify that if any specific borrowing remains outstanding after the related asset is ready for its intended use or sale, that borrowing becomes part of the funds that an entity borrows generally when calculating the capitalisation rate on general borrowings. The Group does not expect any impact from this amendment







Tata Realty and Infrastructure Limited	Notes to the Consolidated financial statements	for the year ended 31 March 2019	(Currency fadian rupees in lakhs)
Tata	Notes t	for the ye	(Currency

Note 3 Property, plant and equipment

				0	Tangible Assets	ssets				
Particulars	Leasehold	Freehold	Buildings	Plant &	Furniture	Electrical	Computers	Office	Mater	Total of
	improvements	land		Machinery	and fixtures	Fittings		equipments,	vehicles	Tangible
								Mobile and Other		Assels
								Enipments		
Cost										
At 31 March 2017	'	1,145	16	39	474	120	395	438	107	2,734
On account Acquisition of Subsidiaries		1,473		153	9	•	80	£	57	1.680
Additions	80	'	25	11	37		55	85	134	415
Disposais		(2)		1	(00)		(12)	(25)	(21)	(06)
At 31 March 2018	\$	2,616	31	263	487	120	446	105	257	4,739
On account Acquisition of Subsidiaries	'	,			'	,				,
Additions	467			\$0	40	ę.	30	110	5	672
Disposals	'			6	(52)	•	(2)	(20)	(44)	(35)
At 31 March 2019	47S	2,616	41	172	498	123	474	16S	227	5,316
Depreciation / авіогіїзатіов алої інпраіттелі								_		
AI 31 March 2017		ŀ	0	5	266	63	259	209	31	833
On account Acquisition of Subsidiaries	•		'	28	~	,	\$	61	16	\$9
Charge for the year	-	,	ŕ	29	74	15	74	77	38	311
Disposals	-		·		(61)		(12)	(3)	(18)	(12)
At 31 March 2015	-		2	62	326	78	329	265	63	1,(13)
On account Acquisition of Subsidiaries	'	•	'	'	,		'	,	'	•
Charge for the year	ñ		ŝ	15	58	12	69	78	30	314
Disposals	•			(0)	(8)		(1)	(6)	(12)	(35)
A131 March 2019	37		S.	¥	946	9:	202	334	88	1,410
			\$			!		ł		
Net Block as at 31 March 2018	6	2,656	38	201	191	4	127	236	190	3.608
Net Block as at 31 March 20(9	143	2.616	33	173	122	33	77	257	147	3,906



## Notes to the Consolidated financial statements

for the year ended 31 March 2019 (Currency: Indian repeas in lakhs)

Note 4

#### Investment Property

Particulars	Freehold land	Lease hold land	Buildings	Computers	Furniture and fixtures	Office equipments	Electrical fittings	Plant & Machinery	Total
Cost								<u> </u>	
At 34 March 2017	20,688	141,279	233,386	129	490	924	11,775	34,86 t	443,532
Additions	· ·		1,935		U	249	130	477	2,791
Disposals		· .	-	-			-	(674)	(674)
At 31 March 2018	20,688	141 279	235,321	1 29	490	1,173	11,905	34,664	445,649
Addition on Amalgamation / merger	· ·		-				-		-
Additions			7,628	-	35	91	\$7	1.623	9,434
Disposals		-	-		-	(3)	(0)	(669)	(672)
A1 31 March 2019	20,688	141 279	242,949	129	525	1,261	11,962	35,618	454,411
Depreciation / amortisation and impairment									
At 31 March 2017	5,724	12,319	32 005	102	266	82	2.955	10 081	63,534
Impairment loss (Refer Note below)	467	-	-	•					467
Charge for the year	-	1,434	8,405	18	36	70	1,106	3,304	14,373
Reversal of impairment loss (Refer Note below)		-			-	-		(389)	(389)
At 31 March 2015	6,191	13,753	40 410	120	302	152	4.061	12.996	17,985
Impairement loss (Refer Note below)			-	-			× .		-
Charge for the year	-	1,427	9.171	8	38	127	1,137	3,127	15,035
Disposals	-	-				(2)	(0)	(607)	(609)
At 31 March 2019	6,191	15 180	49,581	128	340	277	5,198	15,516	92,411
Net Block as at 31 March 2018	14,497	127,526	194,911	9	188	1,021	7,844	21,668	367,664
Net Block as at 31 March 2019	14,497	126,099	193,368	1	185	984	6,764	20,102	362,000

\* Depreciation capitalised to Investment Property under Construction Rs 85 lakhs (Previous year Rs. Nil)

i) Amounts recognised in profit & loss for favostment property		
Particolars	31 March 2019	31 March 2018
Rental luome	53,431	48.214
Direct operating expenses from property that generated restal income	25.228	23.067
Direct operating expenses from property that did not generate rental income	\$\$3	1.472

#### i) Leasing arrangements

The Group has leased properties under non-cancellable operating leases in the capacity of a kaser. Refer Note 47 for future minimum lease payments in respect of date properties till the expiry of lock in

#### ii) Impurment loss

During the year Rs. Nil (previous year, Rs.467 Lakies) of impairment loss was recognised during the year. Based on the performance of the project during the current year and future projections, an impairment analysis had been earned out during the year and such impairment loss was recognised during the year.

#### iii) Measurement of fair values

The fair value of investment property are determined after considering valuation by an independent valuer who holds a recognised and relevant professional qualification and experience in respect of the investment property leving valued

The fair value measurement for all of the investment property has been categorised as a Level 3 fair value based on the inputs to the valuation technique used

#### iv) Valuation technique:

(a) Valuation technique, a. Valuation feature subject properts has been done by Sales Companison Method under Market Approach. A comparison as reade for the purpose of valuation with similar properties that have recently been stull in the market and thus have a transaction price. The sales comparison approach is the preferred approach when sales data are available. Comparable properties are selected for similarity to the subject property considering attributes like age, size, shape, quality of construction, building features, condition, design, goarty, etc. They sale prices are then adjusted for their difference from the subject property. Finally a market value for the subject property is estimated from the adjusted safes price of the comparable properties.

b. The Group also has followed discounted cash flows technique for some property which considers the present value of net each flows to be generated from the property, asing risk-adjusted discount rates.

in case of a valuation of a large land parcel like the subject property, where the development potential is realised over a period of time (i.e. time value of money comes into the picture) and also where there are no or few immediate similar properties (i.e. comparable) available for comparison, DCF asothed considering relevant potential developments of the project is used.

Investment property comprises a number of commercial proputies that are leased to third parties. Each of the leases entered are normally for a period of 5 to 10 years. Subsequent renewals are negotiated with the lease or as per the terms of mitial agreement are automatically remeated. No contingent renewals are charged.

#### Reconciliation of fair value

The Company evaluates impairment for Investment property and Investment property under construction alrogether as both relate to same project. The following is the reconciliation in the fair values as of Morch 31, 2018 and Morch 31, 2019

Particulars	Amount	
Opening halance as of April 5, 2017	444,261	
Additions	2,791	
Fair vidue difference	56.364	
Clasing Indance as of March 31, 2018	503.321	
Addutions	-9,434	
Fair value difference	R 386	
Closing balance as of March 31, 2019	505.041	





Tata Realty and Infrastructure Limited Notes to the consolidated Ind AS financial statements (Continued) (Currency: Indian rupees in lakhs)

	31 March	31 March
Note 5		
Capital work-in-progress		
Land demarcation and reforestation	55	55
Project consultancy and technical charges	66	63
Total Capital Work in Progress	121	118
Note 6		
Investment property under construction		
Land	24,922	24,922
Direct expenses and other adjustments	117,576	\$8,925
	142,498	\$13,847
Less Impairment Loss *	(1,030)	(1,030)
	141,468	112,817

\* Based on the performance of the project and future projections, an impairment analysis had been carried out and such impairment loss was recognised in past years

Note 7 Intangible assets under development under SCA		
Project Development Expenses	62,308	26,300
Professional Fees	1,368	1,138
Project Management Fees	5,100	1,034
Finance Costs (including unwinding of interest)	15,937	6,465
Construction Cost to Contractor	74,996	45,159
Financial hability for premium payable recognised at fair value	62,155	61,442
Other Expense	2,301	1,815
	224,165	143,353




Notes to the Consolidated financial statements for the year ended 31 March 2019 (Currency, Indian rupces in lakhs)

## Note 8

## Other intangible assets

	Intangible Assets			
Particolars	Softwares	Service Concession Arrangements #	Total of Intangible Assets	
Cost				
At 31 March 2017	387		387	
On account Acquistuon of Subsidiaries	· · ·	41,079	41,079	
Additions	23		23	
Disposals	(J2)		(12	
At 31 March 2018	398	41,079	41,477	
On account Acquisition of Subsidiance				
Additions	10	9,271	9,281	
Disposals		• : :		
At 31 March 2019	408	\$0,350	50,758	
Depreciation / amorbisation and impairment				
At 31 March 2017	285	•	285	
On account Acquisition of Subsidiaries	-	1,271	),271	
Charge for the year"	42	2 153	2,195	
Disposals	(11)		(1	
At 31 March 2018	316	3,424	3,740	
On account Acquisition of Subsidiance	-		-	
Charge for the year	21	1 572	1,593	
Disposals	-	•	-	
A1 31 March 2019	337	4,996	5,333	
Net Block as at 31 March 2018		37,655	37,737	
Net Block as at 31 March 2019	71	45,354	45,425	

## Passenger Ropeway Facility

The Group has received the tight to charge users a fart for the passenger topeway facility which the Group will collect and return. The Group has a dispectantic decide the user fare to be charged Hence the condition prescribed for service concession Amagement under ind AS 115 Construction Contracts is not fulfible benee SCA accounting is not applied. The Group shall pay to the Authority every year (once construction is complete), a concession for, a Minimum Guaranteed Amount (MGA) of NR 100 lakks for Dharanshala & DNR 150 lakks for Manali which shall be increased by 5% and 10% respectively every year over the previous year's annual concession for the Concessionaire shall remote shall project assets including land astranged by the concessioning Authority (CA) or its designated agency/authority after the expiry/termination of Concession Period at Ni Value's evidentiary costs obligation to CA. At the time of transfer the project assets including Ropeway. Parking structure, equipment etc shall have been transferred in good condition with all specifications, drawings, write ups etc

# Acquisition of subsidiary accounting During the previous year Group acquired 100% stake in one of the subsidiary company. The excess of the

Consigning the previous your consigning room and in the original yourgany for exect of the consideration transformed. Acquisition-databook value of any previous equity interest in the acquired entity over the book value of the net identifiable assets acquired is recorded as goodwill. The goodwill of Rs. 14.256.23 lakks has been allocated towards Intangible Asset (Road Asset SCA) being the fair value increase on the date of aquisition.

## Service enneession agreement

Name of entity	Description of the arrangement Significant terms of the arrangement
International โกร์าสไขมใช้ Pvt Ltd	As per the Concession Agreement (CA) the Company is required a Period of concession 33 yrs build and operate Multis Level Car Parking and collect Parking fee: lawestment grant from concession grantor - Mil from Vehicles and User State for Public Convenience Facilitations. The lawestment grant doing atoms - Mil CA also specifies that the Company is required to Sufficient Basis upon which re-prioring or re-negotiation is determined. Inflation of the total area of commenceal infrastructure and an area of 5% b.c. 466 sq m shall be reserved for Acto Showroom/Auto Workshop.
Harop Expressways Private Limned	Design Build Finance. Operate and Transfor (DBFOT) basis. Period of concession 2016 - 2041 augmentation of the existing road from km 299 00 to km 418 750 Remuneration. Toll (approximately 120 18 km) on the Hospet-Chittadurge section of Investment guant from concession granter. Nil National Highway No 13 (New National Highway No 50) National Highway No 13 (New National Highway No 50) Basis upon which re-pricing or re-negotiation is determined. Inflation Premium pagable to granter. Yes
Dharamshala Ropeway Litnited	The Company has entered agreement with The Department of Period of concussion: 40 Years Tournern & Civil Aviation, Himachal Brudesh ('Authority') for the Retrumention Rs 1 Cr Plus 5% Increase construction of Passenger Ropeway between Backinde of Investment grant from concession granter NIL Dharamshala Bastand unto Dalah Lama Temple, Michedigan Investment grant from concession granter NIL Dharamshala Bastand unto Dalah Lama Temple, Michedigan Investment grant from concession granter NIL Dharamshala Bastand unto Dalah Lama Temple, Michedigan Investment and reneval obligations. Non Renewal under DBFOT Model (Design, Build, Funnee, Operate, Bacily)" The Build France Operate & Transfer Passenger Ropeway facility is under construction as on date. The Crvil & Lami Stre Devlopments works have been started during the year Under the terms of the agreement, the Company will operate and make the Teasenger rupeway facility available to the public once construction's complete





Notes to the Consolidated financial statements for the year ended 31 Morch 2019 (Carrency: Indian rupoes in Takhs)

Note 8 Other intangible assets

Other intangible assets	
Munali Ropeways Pravate Lanued	The Company and encered line a Concession Agreement on dated [Priod of concession 49 Years 21 October 2015 - with The Department of Tourism & Civil Remuneration 18x 3,50 Cr. Plus 10% Increase Aviation, Himachal Pradesh [Authority] for the constituction of [Investment prant from concession granter: NIL Passenger Ropeway between Palchan and Rohteng at Manak under Investment and renewal obligations: Non Renewal BOT Model [Unit], Operate and Transfer Model]. The construction and support which respectively or re-negotiation is determined in has not started as on date and the company is in process of obtaining Bailt/Operate and Transfer Various government approvals for commencing the construction of Premium payable to granter (Rs,6,80 Cr ropeway.Under the strans of the agreement, the Company will operate and make the passenger ropeway facility available to the public once construction is complete.
Uchit Expressway Private Lid.	Six Laning of Chittorgarh-Udapur Soction of NH-76 from Design Period of concession, 29 Years Channage Km 214.870 to Km 308.370 in the State of Rajastlank Remineration : NIL (Length 92.500 Km) on DBPOT (Tell) Mode under NHDP Phase V Investment grant from concession granter : NIL (Package -III) Basis upon which re-pricing or re-negotiation is determined : NA Prendum payable to granter : NIL
Darg Shinnadh Fixgnasoways Private Lumited	The company is anguged in the business of toll collections from the users and operation and maintenance of total stretch of 18.5 kms at leads on (50/5/2004) NH-6. Durg Bypess as aggreed between the Company and National 21 Remineration. Toll Fee Collection from Read Users (approx. 6% to Highways Authority of India (NHAI) on Build. Operate and 8% growth is year on year) Transfer (BOT) basis 2) investment from Concession Grantor (NHAI) – NIL 4) Basis upon which re-printing or re-negatistion is determined – Toll Tanff will be revised for April of every year considering WP1 5) Premium payable to Grantor – NiL
Matheran Ropeway Private Limited	The Company is developing a Ropeway project in Matherau. The Company las signed an Agreement with The Matheran Guisthan Matherakhira of apprex, 4500 Meters connecting Bhuivati Village Nagar Parishad, Matheran on 20th May 2003 for a period of 99 years, to Madharqi Point at Matheran, on Design, Baidd, Fmanne, Operate The Company las agreed to pay compensation equal to 5% of list net and Transfer (DBFOT) basis. The Project would have all elements profit of each year to Matheran Nagar Parishad. Matheran, of sustainability and would be noise free, electricity driven. Eco-friendly and son-polluting; It is estimated that the Project would The Ropeway will be divided into 2 sections. Length of Section 1 is approx. 2.7 Km with manceable detachable gondola system. Length of Section 2 is approx. 1.75km with tricable jigback system. Daring the FY 2017 -18, company its achieved Environmental clearance from MOEF and made significant advancement in Stage II. Forest Clearance. The company its plauning to commence the construction work in October 2018 and commercial operations by Decomber 2019.
Pruk Solapur Expressivity Pit Lid	the Company ("concessionaire') entered into a service concession Period of concession, 20 Years agreement with a NHAB (The National Highways Authority to India Remineration : NIL also the "grantor") to construct a tolk highway between Pine and Investment grant from concession grantor : NIL Solapur. The construction of the toll road started thereafter and in Investment and renewal obfigations : NIL was parsially completed on 3 February 2013 and was Tully Basis upon which re-pricing or ro-negotiation is determined ; NA completed and available for use on 31 January 2015





Verent de la company and minastracture Dimited		
Notes to the consolidated Ind AS financial statements (Continued)		
(Currency Indian rupees in Jakhs)		
N	31 March 2019	31 March 2018
Note 9		
Carrying Cust of investment in joint ventures:		
Unquoted equity shares, fully paid-up 2,386,711 (2018, 2,386,711) equity shares of Pune Solapur	2 150	1 170
Expressways Private Limited, a joint venture, of Rs.10 each fully paid	2,170	2,770
up		
740,000 (2018 740,000) equity shares of INR 10 each in TRIL IT4		
Private Limited (Formerly known as Albrecht Builder Private		
Limited) ##		
10,000 (2018 10,000) equity shares of A & T Road Construction	-	
Management and Operation Private Limited, a joint venture, of Rs 10		
each fully paid up, ##		
2,997 (2018 2,997) equity shares of INR 10 each in Industrial	21,910	21,921
Minerals and Chemicals Company Private Limited		
Unquoted equity shares, fully paid-up		
16,989,290 (2018 16,989,290) equity shares of INR 10 each in	30,436	30,459
Mikado Realtors Pv4 Ltd		
-	54,516	55,150
Investment in debentures of joint venture		
814 (2018 814) Rated Listed Redeemable Non-convertible	8,140	8,140
Debentures of INR 10 Lacs each in TRIL IT4 Private Limited		
10,473,960 (2018: NIL-) Compulsarily Convertible Debentures of	10,474	
INR 100 each in Industrial Minerals and Chemicals Company Private		
Limited		
-	18,6)4	8,140
	10,074	5,140
-	73,130	63,290
## Unrecognised share of Losses in Joint Venture		
TR1L IT4 Private Limited	(16,815)	(12,550)
A & T Road Construction Management and Operation Private		
Limked	(1)	(1)
New 10		
Note 10		
lavestment in other companies fair value through profit and loss		
Unquoted equity instruments, fully paid-up		
Others		
16,500 (2018: 16,500) equity shares of OPG Power Generation	-	2
Private Limited		
117600 (2018: 117600) equity shares of Vagara Windfarms Limited	12	12
119,187 (2018: 119,187) equity shares of Echanda Urja Private Limited	11	11
-	23	25
Note 11		
Other non-current financial assets		
Unsecured, considered good		
Fixed deposit having maturity more than 12 months under hen	500	375
Lease rental receivable	193	94
Interest accrued on bank deposits	•	\$
Recoverable from Punjab Urban Development Authority ( Refer foot		
note below)	517	
Security Deposits	467	1,195
	1,677	L,669

Foot Note:

The respective group company has won the matter under litigation with Punjab & Haryana High Court single bench for recovery of extension fee amounting to INR 372-77 lakhs alongwith interest from Punjab Urban Development Authority (PUDA) in March 2016. PUDA filed an appeal with Punjab & Haryana High Court in August, 2017 with double bench against the judgement of single bench of Punjab & Haryana High Court. The matter is pending with honourable High court for hearing and the next hearing date for the same is May 16, 2019.





Tata Realty and Infrastructure Limited Notes to the consolidated Ind AS financial statements (Continued) (Currency: Indian rupees in lakus)

(Currency: Indian rupees in lakits)	31 March 2019	31 March 2018
Note 12		
Deferred tax assets		
- investments in associates and joint ventures	2,487	2,274
- Impact of Effective Interest rate on Borrowings		(173)
- other financial assets at fair value through profit and loss	1,340	-
- Unused Tax losses carried forward		19
<ul> <li>On other provisions (including provision for employee benefits and other amounts allowable on a payment basis)</li> </ul>	239	278
- on account of PPE and Intangibles	791	2,365
- derivatives (including call put options)	1,599	2,481
	6,456	7,244
Note 13		
Non Current fax assets (Net) Advance tax	15,845	14,591
Less: Provision for tax	(4,956)	(5,020)
	10,889	9,571
=		
Note 14		
Other non-current assets Unsecured, considered good		
Capital advances	16,676	22,175
Balance with Government Authorities Prepaid expenses	112	42 14
Lease equalisation reserve	9,310	7,496
Other non-current assets	169	-
-	26,275	29,727
Note 15		
Inventories		
(valued at cost or net realisable value whichever is less)		
Construction materials	52	256
Consumables stores and spares	77	83
Work-in-progress	15,341	13,592
Finished goods	28,248	30,501 44,432
=		
Note 16 Trade and other receivables		
Secured, considered good	2,061	2,929
Unsecured, considered good	785	2,858
Unsecured, considered doubtful	275	393
Allowance for doubtful debts	(275)	(393)
_	2,846	5,787
Note 17		
Cash and cash equivalents		
Cash on hand	93	72
Balance with banks - in current accounts	2,099	4,034
- in deposit accounts	3,729	3,035
	5,921	7,141
=		
Note 18		
Bank balances other than cash and cash equivalents		
Term deposit with original maturity less than twelve months under lien	1,579	345
	1,579	345
0		





## Notes to the consolidated Ind AS financial statements (Continued)

(Currency Indian rupees in lakhs)

	31 March	31 March
Note 19		
Investment in mutual funds:		
Fair valued through Profit and Loss:		1.076
Units Held 10,576.43 (2018 - 2,54.398.32) of Face value of Rs 1.000 (NAV - Rs 2,944.44) each in TATA money market fund - Direct Plan -	311	6,966
Growth		
28,831.67 units [2018 28,831.67] of Face value of Rs 1,000 (NAV -	884	747
Rs 2,788.16) each in Franklin India Treasury Management Account -		
Super Institutional Plan - Growth		
10,19,407 79 units [2018: 10,19,407 79] of Face value of Rs 10 (NAV	265	245
- Rs 26.27) each in Franklin India Ultra Short Bond - Super	208	243
Institutional Plan - Growth		
Inancerional Field - Colouri		
	1,383	7,958
Note 20		
Short-term loans and advances		
Unsecured, considered good		
Advances given	2,000	•
Security deposits	38	•
Advances recoverable from Related Parties	683	-
Other loans and advances	333	970
	3,054	970
Secured considered good		
Other advances		4,050
	3,054	5,020
Note 21		
Other current financial assets		
Unsecured, considered good		
Interest accrued on deposits	115	353
Claims Receivable from NHAJ (utility shifting)	674	223
Recoverable from Punjab Urban Development Authority		517
Unbilled revenue	537	102
Foreign Exchange Forward Contract Receivable		114
Security deposits	263	5
Others current financial asset	392	6
	1,981	1,320
Note 22		
Other current assots		
Unsecured, considered good		- 020
Advance to vendors	681	3,070
Balances with Government and other authorities	1,463	1,222
Prepaid expenses	805	862
Lease equalisation reserve	1,360	2,236
Claim Receivable from NHAI (Refer foot note below)	1,048	•
Others	31	35
	5,388	7,425

Foot Note:

Claim Receivable from NHAI pertains to claims towards change in law os per clause 41.1 of Service Concession Agreement entered with NHAI by the respective group company.

Note 23 Share capital

	101,731	101,731
Equity share capital	101,731	101,731





Notes to the Consolidated financial statements

as at 31 March 2019 (Currency: Indian rupees in lakhs)

Note 23 Equity share capital Authorised share capital

Equity share capital Authorised share capita)	31 March 2019	31 March 2018
3,00,00,000 (2018: 2,00,00,00,000) equity shares of INR 10 each	300,000	200,000
1,00.00,00.000 5% Non cumulative convertible preference shares of Rs 10 each.	-	100,000
lssued, subscribed and paid-up		
1.01,73.07,692 (2018: 1,01,73,07,692) equity shares of INR 10 each	101,731	101,731
	101.731	101.731

a. Reconciliation of the shares outstanding at the beginning and at the end of the year

Equity shares	31 March	2019	31 March	2018
	Number of shares	Amount	Number of shares	Amount
At beginning of the year Issued during the year	1,017,307,692	101,731	1.017,307,692	101,731
Outstanding at the end of the year	1,017,307,692	101,731	1,017,307,692	101,731

## b. Terms and rights attached to the equity share

The Company has only one class of equity shares having par value of INR 10 per share. Each holder of equity shares is entitled to one vote per share. The Company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject to the approval of shareholders in the ensuing Annual General Meeting. In the event of liquidation of the Company, the holders of the equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the shareholders

## c. Shares of the company held by the Holding company

31 March 2019		31 March 2018		
Number of shares Amount		Number of shares	Amount	
1,017,307,692	101,731	1,017,307,692	101,731	

## Tata Sons Private Limited and its nominee Equity shares of INR 10 each, fully paid-up

Equity shares of INR 10 each, fully paid-up

## d. Details of shareholding more than 5% in the company

 31 March 2019
 31 March 2018

 Number of shares
 % Holding
 Number of shares
 % Holding

 1,017,307,692
 100%
 1,017,307,692
 100%



Tata Sons Private Limited



Notes to the consolidated Ind AS financial statements (Continued)

(Currency Indian rupees in taklas)

	31 March 2019	31 March 2018
Note 24 Other equity		
Reserves and surplus		
Scourities premium	15.769	15,769
Capital reserve	4,911	4,911
Retained earnings	(99,543)	(74,541)
Other comprehensive income		
Defined benefit plan adjustment	27	(14)
FVOCI - equity instruments	<u> </u>	<u> </u>
-	(78,836)	(53,875)
Securities promium reservo		
Opening balance	15,769	15,769
	15,769	15,769
Opening balance	4,911	6,487
Addition during the year		
Deletion during the year		(1,576)
	4,911	4,911
Opening balance	(74,541)	(35,158)
Net loss for the year attributable to owners	(23,693)	(40,198)
Consolidation adjustment	(5)	(+0,198)
Transition adjustment due to application of IND AS 115	(1,304)	-
(Refer Note 51 (a))	(1,504)	
Transferred from FVOCI		\$06
	(99,543)	(74,541)
Other comprehensive income		
Opening balance	(14)	(40)
Remeasurements of defined benefit hability (asset)	55	35
Income tax relating to items that will not be reclassified to profit or loss	(14)	(9)
-	27	(14)
– Equity Instruments through Other Comprehensive Income		
reserve Opening balance		806
Opening balance Transferred to retained earnings	-	
riansiented to recamed cartings	·	(806)
	•	-

## Nature and purpose of the reserve

Securities premium Securities premium reserve is used to record the premium on issue of shares.

Capital reserve

Capital reserve is used to record excess of net assets taken over pursuant to scheme of merger sanctioned by High Court

## Convertible preference shares classified as equity

The aroup has classified compulsorily convertible preference convertible into fixed number of equity shares as part of other equity

## FVOCI - equity instruments

The group has elected to recognise changes in the fair value of certain investments in equity securities in other comprehensive income. These changes are accumulated within the FVOCI equity investments reserve within equity. The group transfers amounts from this reserve to retained earnings when the relevant equity securities are derecognised.

404,786	345,606
24,910	3,281
114,792	231,439
541.488	580.326
	24,910





ata Reathy and infrastructure Lumited	lotes to the consultidated Ind AS funnechal statements (Combinned)	n the year ended 51 March 2019	Currency Indian repose in lakhs)
ata Reatly	oles to the	the year	NITTERICY

Tar. Reath, and infinitucture Lennied Notes to the cousalidated hud AS fanacial statements ( <i>Continued</i> <i>for the year outed 21 March</i> 2019 ( <i>Coursers</i> , Indian rupees in lakis)	ments <i>fCimta</i>	(neu		
Name of the Bank FI	Autonot	Terrus of Repayment	Security	Rate of Interest
Banks / Financial દostiturioze				
HBFC Ltd	15,782	F 5 5	eron Loan repay able in single installment after 40 First charge on the entire land and building pertaining to the project ionits from date of first disbursement $i \neq \frac{1}{4}$ db is 2020	The rate of interest on term loan HDFC CPLR less spread of 785 bas, moreset barry, within the range of 8.90% to 10.40% per annum pay able at monthy rests.
HDFC Ltd	20,815	E 2 01	Term Loan repayable m single installment after 54 First charge on the entire tand and building pertaring to the project moduls from date of first disbursement i.e. 13th september 2021	The rate of interest on term laan HDFC CPLR less spread of 785 bps, interest being, within the range of 8.90% to 10.4.2%, per annum pa, able at nonthyr rests.
HDFC Lid	(8)5.4	Tarm Loan reportable in single installment after 72 monds from date of first disbursement (.s. 13th March 2023	First charge on the entire land and building pertaining to the project	The rate of interest on term loan HDFC CPLR less spread of 785 bps, interests being: within the range of 8.97% to 10 40% per anuum pay able at monthy rests.
Bink of Maharashtra	3.519	Reportent of principal amount will be to quarterly installments over period of 10 years commencing from March 2019 till March 2029	Reproment of principal amount will be no quarterly (i) First charge by vary of montgage on numvouble project assets astallments over period of 10 years commencing(ii) First charge by vary of hoottage on the montable project assets from March 2019 ittl March 2029 (iii) First charge on the Project's book clebre, operating cash flows, receivables, commissions, reviewes of rate of interest is 10% per whatsoever nature and wherever aiting, present and future, intaagibles, good/nill, uncolled capital of interest is 10% per present and future (iv) First charge on therever aiting, present and future, intaagibles, good/nill, uncolled capital of interest is 10% per present and future (iv) First charge on therever aiting, present and future, intaagibles, good/nill, uncolled capital of interest is 10% per present and future (iv) First charge on Projects Ecrons Account, Current Account, and the amounts therein. (v) First charge on Projects Ecrons Account, Current Account, and the amounts therein.	Rate of interest will base rate +1.25% per annum is presend, ide rate of interest is 10% per annum (approx) Interest payment frequency will be monthly.
Inden Bank	7.250		Pay able in quarterity installments starting frond i) First charge on the entire fand and building including project assets September, 2529, 711e berrowning on the source on the entire moviable and numeriable assets Interest rate of Benchmark rate, being base rate of fil): Charge on ill the revenue/receivables accruing to the project accumits, any other bunk Source Bank of India (SBI), plus spread iv) H5 politication' mortaguge' assignment of all the rights, title, interest, benefits, claims and demands whatsoever of the companies in the Project Documents	The taile of faithrest during the 1 can was in the range of 2.5% to 9.80%.
Huon Bark of India (TL )	825-61	Repayable in quarterly rastallinents commencing (a), First charge on after immovable from June, 2022 and termutating on March, 2040 [furte, save and accept proper Assers (b) First charge on all tangither movable and machinery, machinery parces and and machinery, machinery parces and the all accounts of the Accounts (or any acceution for Departure and the Marchardent from time to time, a deposited therein, from time to time, a	Reparation in quarterly installments commencing (a) first charge on entries immortable properties of the respective group company, both present and rade of interest is 9.74% (upvou) from June, 2022 and termunating on March, 2040 [future, save and accept propert Assets. (b) First Analysis machinery, imachinery improved Assets. (c) First Analysis machinery, imachinery, imachinery interest and looks and accept Propert Assets. (Intriume, firstware, vehicles and all other more able assets, both present and future, if any save and except Propert Assets. (c) First Analysis and Interest and looks and accept Propert Assets. (c) First charge on all angulte movimes and future, if any save and except Propert Assets. (c) First charge on the assets, both present and future, if any save and except Propert Assets. Accounts for any accounts of the respective group company including the Escow Account. Sub- Accounts for any account to Able assets, both present and future, if any is accounts of the respective group company intervals and all inde- dependence, from time to finance, and reservable and Permitted Investments on other securities deposited therein, from time to time, all receivable and Permitted Investments on other securities	iale ut interest is 9.74%, a (apr.u.)







al .	mageral statements (Cuntinned)		
ata Realty and infrastructure Lamsed	otes to the consolidated Ind AS financial statements (Cuntinned)	r the year ended 31 March 2019	Currency. Indian rupees in lakhs)

Name of the Bank/ F1	Anigual	Terns of Repayment	Security Rate of Intervet
State Bark of fadua	+2894	Repairable in quarterly installinemis commencing from lune, 2022 and terminating on March, 2040	Repstate in quartery installments commercing (a) First charge on entire immovality properties of the Company, including movable plant and mactimeer, project Activities 2022 and terminating on March 2040. Direct charge carall weighle movable passes of the Company, including movable plant and mactimeer, morthmery spares and total accessories. furnitime, fixtures, vehicles and all other movable plant and mactimeer, morthmery spares and total accessories. furnitime, fixtures, vehicles and all other movable plant and mactimeer, morthmery spares and total accessories. furnitime, fixtures, vehicles and all other movable plant and mactimeer, morthmery spares and total accessories. furnitime, fixtures, vehicles and all other movable access. (SF First charge or and lacement and fixe Activity and the Activity and Procedure or any of the vehicles and all other movable access. (SF First charge or all intrapellies assets of the Company including blue not limited to goodwill, rights, underthinking there intra-strange or all intrapellies assets of the Company including blue not limited to goodwill, rights, underthinking there intra-strange or mediating the subject to the extent permissible as period and a strange or mediating the strated blue of concession for an arigination provided that all recercibers arrain directive specifies that a data and in the Arriste Direct Activity and a strateging to the Company in a data and a charge or the extent permissible as period and a strateging to the Company. That and strateging to the Company in the first charge or media and and and a strateging to the Company in a strateging to the Company. The Arriste Direct Activity and a strateging to the Company in the Arriste Direct Activity and the antigenees of the Company in the antigenees of the Company in the antigenees of the Company in the Arriste Direct Activity appendent and the Arriste Direct Activity appendent an
State Bank of India State Bank of Commerce Connerce Connerce	26,017 26,017 7 038 14,330 21,499	tepsy able in structured quarterly, installments commanding from June 30, 2020 and terminating on September 30, 2033	(i) First charge on entire immovable properties of the Company. If any, save and except project Assets are average rate of interest of PW+. (ii) First charge on entire immovable average (if) Company, both present and funct. If any, save and except Project Assets (iii) First charge on entire movable average (if) Company, both present and funct. If any, save and except Project Assets (is defined in the Concession Agreement). (iii) First charge on the Eaconv Account Thrat and Relemine Account (TRA), Depr Karvite Reserve Account Of TAA). They for the project Assets (is defined in the Concession Agreement). (iii) First charge on the Eaconv Account Trat and Relemine Account (TRA), Depr Karvite Reserve Account Of TAA). The first charge as the Eaconv Account Trat and Relemine Account (TRA). They for the project of the trate as a set on therite And any and the trate and the each of the first charge as end and the configurated for the Project Assets and other thank account? That and Reserve Agreement and Reserve Agreement. (v) First charge on all instangletes including but not limited to goodwill and insteadled capital, present and for the Concession Agreement. (v) Assignment of security Merement. (v) Assignment of security index on the Borrower's right, fulle and interest to the extent covered by and interest to the extent overed by and interest to the fact the first charge as used in any account and and account of the eacter or the Borrower's right, fulle and interest to the extent overed by and interest on the Borrower's right. After and interest to the extent overed by and interest of the first charge as the insurance proceedid exposited fund the Eacore Account (1). First charge as the insurance proceedid exposited fund the extent overed b
Conporation Bank State Bank of India	6.471 82.000		Scened by a first ranking part pass charge on the entire fired assets of the Company, including the The mitterst rate from the date of lessehold rights under the lesse deed, the projects and the movables thereto and file Company's nighted behavior 2019 under the project documents: collateral security mortgage charge over the investment properties overaed was in the range of \$20%, to by the Company's nighted out the science of \$20%, to by the Company's nighted out the science of \$20%, to and such other accounts of the Company and the receivables there to not second. the \$70% is contact account and such other accounts of the Company and there from: first ranking





			-	
Name of the Bank/ FI	Amount	Torms of Repayment	Security	Rate of Interest
State Bank of India	NG 744		Repayment in -18 instalments commencing frond. Secured by a first ranking part passu charge on the antire current assets, current accounts, rescrow The interest rate from the date of accounts, project documents and such other assets of the Company. Collateral security over the johtaining the loan to 31 March 2019 to restrict and projection or the other asset of the Company. Collateral security over the johtaining the loan to 31 March 2019 docds. Have deck and the amenuities approached to the factor of lease neural security over the johtaining the loan to 31 March 2019 docds. Have deck and the amenuities agreement and other leases entered by the Company in the investment property. As per this tarists of the company has submitted Financial Guarantee of Rs. 12,40,40,000 for year ended 31 March 2019	a interest rate from the date of aining the learn to 31 March 2019 i N 30%.
HDFC Lid	83.175		The 650cr - Repayment of 175cr per month Secured by a first ranking purt pass what go on the entire fixed assets of the Company's rights obtaining the free of from due due of from the off assets of assets of assets of the projects and the morables thereta and the Company's rights obtaining the loarn in 31 March 2019 2020, with minimum sectors and the morables thereta and the Company's rights obtaining the loarn in 31 March 2010 as principal to be repaid in 108 equated under the project documents: collected assets of charge over the morables thereta and the Company's rights obtaining the loarn in 31 March 2010 as the first ranking part pass uptact and the project document properties owned the active the part pass of the Company and the project document to the company of the Company and the project document properties owned the active the part pass of the Company and the repeat meters are also for the Company and the repeating the loarn in 31 March 2010 the base of the Company and the project document properties owned the repeat the transmitter account of the Company and the repeating the first ranking part pass of the Company and the requires there are an account the frame of 181 part pass observes on the debt service reserve account and the renounts therein. The 20nder is the ranking part of 50 months of the conversion of the company therein.	interest rate from the date of anong lite loan to 31 March 2019 IS 195%, IS 2019%,
Financial Institution				
Adır e Brita Frrance Lımitod	12.250		Pay able in quarterly installinearts starting from[1) First charge on the entire land and building including project assets September, 2020. The borrowings carry a floating(1) First charge on the entire moviable and immoviable assets interest rate of Beachmark nue, being base rate of (1) Charge on all first revensive revensive revensive revensive rate of the project, all the project accounts, any other bank interest rate of findia (SBI), plus spread in Project and the rate of and of any proceeds bying in the Escrow Accounts, any other bank is the project accounts, any other bank is the project accounts and all the revensive assignment of all the rights, tile, interest, benefits, claims and demands whatsoever of the companies in the Project Documents.	The rate of Interest during the year was in the tange of 9,25% to 9,80%.
Adış a Bıria Fransıce i td	12.788		<ul> <li>Repeable in quartery installaeets commencing (a) First charge on entire immovable properties of the Company, both present and foure, carve and except Project Assis.</li> <li>(b) First sharge on all coupling investige family including movable plant and machinen, muchinery sparses and tools and accessores. (unitative, fixtures, vehicles and all other movi able assets, both present and funct, if 'usy are and accessores. (unitative, fixtures, vehicles and all other movi able assets, both present and funct, if 'usy are and accessores. (unitative, fixtures, vehicles and all other movi able assets, both present and funct, if 'usy are and accessores. (unitative, fixtures, vehicles and all other movi able assets of the Company including the Eserow Accounts for any eccount in Substitution thereach that may be opened in accordance with this Agreement and the Menorabulative for the all reservices on any of the Ecopyany including but not limited to goodivalit. rights provided that all reservices and all constrained to other second. The Project Asset provided that all reservices on Automatic and all and a subject to the extent permissible as priod fractional, present and funct, effect.</li> <li>(d) First divage an all sitiangibles assets of the Company, including but not limited to goodivalit. rights provided that all reservices on Automatic and all and all of the Concesson Agreement. The fit of the Concesson Agreement and funct, effect.</li> <li>(d) First divage and strangibles assets of the Company, in the famile of goodivalit. rights provided that all reservices and haves of a depasted in accordance state. The fit of the state all as above, able assets of the Company, in the analyter of the Concesson Agreement and funct, effect.</li> <li>(d) First divage and state of the Econy Agreement and funct all cost variables as a set advises are all as above. Accemany.</li> <li>(d) First divage and all states of the Company.</li> <li>(d) first divage and all states of the Company.</li> <li>(d) first divage and all states o</li></ul>	of interest is 9.14% p.a. (apress)







Naur of the Baulu F? Non Convertible Debentures Non Convertible Debentures				
Non Convertible Debentures Non Convertible Debentures	Ansual	Terms of Repayment	Security	Rate of Interest
Non Convertible Debeatures				
	17,5(10	The regorment shall be as commencing from FY 2020-21 uil 2029-2030	(i) Hypotheration by way of first charge on all movtable assets, bank accounts including all receivables The debehrures shall carry, a fixed indicates both present and future. (a) Rupit's of the company under the project docurrents including Instance contracts (a) Rupit's of the company under the project docurrents including Instance contracts (a) Rupit's of the company under the project docurrents including Instance contracts (a) Rupit's of the company under the project docurrents including Instance contracts (a) Rupit's of the company under the project docurrents including Instance contracts (a) Rupit's of the company under the project docurrents including the sume contracts (a) Rupit's and the company under the project docurrents including the sume contracts (a) Rupit's and the company under the project docurrents including the sume contracts (a) Rupit's and the company under the project docurrents including the sume contracts (b) Rupit's and the sume docurrent docurrents including the react of the contract and the react docurrent docurrent is react. (a) Rupit's and the react docurrent docurrent is react. (b) Rupit's and the react docurrent docurrent is react. (b) Rupit's and the react docurrent docurrent is react. (c) Rupit's and the react docurrent docurrent is react. (c) Rupit's and the react docurrent is react. (c) Rupi	The debeltures shall carry a fixer rate of inturees of 8,75% with a put call option in March 2020
Non Convertible Debeatures	K.K20	The NCD's are for a tenure of 2 years from the date of first drawdown. Accordingly, the same is to be redeemed on 29th June 2019.	i)first and exclusive charge over the building and premuses .14 mov cable fixed asets and cash flow of the project. The security creation is in process as on March 31, 2018	Non Convertible Dabantures (NCDs) interest @ 9.2%, payable at half yearly rests monthly penal interest @1%
Non Convertible Debrakues	12.150 The mon	The Debentures shall be redeemed after 144 months from date of disbussement (20 June 2017).	<ul> <li>(a) Second charge on Company's fand allotted by the Puajab Urban Planning &amp; Dor elopmont Authority (PUDA).</li> <li>(PUDA).</li> <li>(b) First anclurative charge on the building of the project and (c) First ancharge on the building of the project and (c) First angle program dange on:</li> <li>All receivables from the project and the Escrow Account</li> <li>Abbi Serrice Reserve Account (DSRA).</li> </ul>	
Non Convertible Detreatures	1×2,500	The Debantures shall be redeemed from 234d May Nul 2019 to 204h April 2421		The rate of interest on debenures is 8.25% to 9.25% per annur, and coupen to be paid annually Atauruty
Non Converble Debumurks	13,000	The Debenures shalf be rederned on 25th March 2022	First ranking charge over 100% of issued & paid up capital of Durg Shivnath Expressions Private Limited	The rate of interest on debentures is 9-75% per annum, coupon to be paid on semi-ornually.
TOTAL	683,730			
HDFC Bank Lid	5,000	Reparable within next f2 months	Secured for a first ranking pur passu charge on the entire fixed assets of the Company	The tate of interest is \$ 75%
Union Bank of India Bank OD	784	784 Rapey able within next 12 months	vany, both present	The rate of unterest is 9.50%
Avis Bank Ltd	4,610		Unsecured	7002 6
Commercial Paper				
Frankin India Liquid Fund	12,500	Repayable on 29 April 2019	Unsecured	7.83%
Frankin India Liotaid Fund	20,000	20.000 Repayable on 9 May 2019		8.05%
Frankin India Liquid Fund	10,000	10.000 Repayable on 10 June 2019		8.06%
Frankin India Liquid Fund	20,060	20.0481 Repayable on 13 June 2019		8.06%
Atahindra Liquid Fund	5,060	Repayable on 21 May 2019		7.85%
SBI Metual Fund - SBI Savings Fund	10,000	10.000 Repayable on 20 June 2019	Unsecured	8.05%
futer Corporate Deposit (rom Related Party				
Tata Housing Development Co Ltd	14,000	14.000 Repayable within next 3 months	Unsecured	9,01%
TOTAL	101,407			

## Notes to the consolidated Ind AS financial statements (Continued)

(Currency Indian rupees in lakhs)

	31 March 2019	31 March 2018
Note 26		
Other financial liabilities		
Financial liability for premium payment at fair value	69,110	65,869
Interest- free security deposits from customers	18,210	12,073
Interest accrued but not due on borrowings	11,439	4,945
Derivatives not designated as hedges		
- Forwards	425	3,128
- Put option	1,818	1,498
Liability towards purchase of shares	649	885
Retention money payable	1,444	752
Security deposits	25	66
Project development fees pavable	45	45
	103,165	89,261
Note 27		
Long-term provisions		
Provision for employee benefits		
Gratuity	430	502
Leave encashment and compensated absences	614	626

Provision for Major Maintainence of Road # 2,216 1,109
3,260 2,237
# As per the best estimate of the management, provision is been made, towards cost of major maintainence of the roads. Such major

# As per the best estimate of the management, provision is been made, towards cost of major maintainence of the roads. Such major maintainence is to be executed in the Financial Year 2019-20 in terms of the agreement entered into with National Highways Authority of India.

Particulars	31st March, 2019	31st March, 2018
On Acquisition of Subsidiary	1,109	157
Add: Provision for the year	1,011	933
Add: Unwinding of discounts on provision	96	19
less: Utilisation / Settlement	-	
Closing balance	2,216	1,109
Note 28		
Deferred Jax liabilities		
<ul> <li>on account of PPE and Intangibles</li> </ul>	1,117	454
- other financial assets at fair value through profit and loss	467	-
	1,584	454
Note 29		
Other non-current liabilities		
Advance received	7,050	7,050
Advance rent received	4,390	4,496
		11,546
Note 30		
Short-term borrowings		
Secured		
- From Bank	5,789	1,501
Unsecured		
Commercial paper		
- From Mutual funds #	76,567	59,695
Bank Overdraft	4,118	•
Inter Corporate Deposits from Related Parties	\$4,000	•
Loans from Others *	100	100
	100,574	61,296
Description of the second second second second second		

Details of security and repayment terms:

# The said borrowing represents Commercial paper issued to mutual funds at a discount rate ranging from 7 20% - 9.10% per annum (2018) 7.05% - 7.50% per annum), and the same are repayable within one year at the agreed upon full face value.

\* Borrowings outstanding as on March 2019 and March 2018 is towards interest free, unsecured loan which shall be due and repayable after satisfaction of tranche 1 couldnons in Mathema Ropeways Pvt Ltd





2 4

Chartered

Accountants

Tata Realty and Infrastructure Limited Notes to the cansolidated lod AS financial statements (Continued) (Currency Indian rupees in lakhs)

(Currency Indian rupees in taxins)		
	31 March	31 March
Note 33		
Trade and other payables		
Due to Micro, Small and Medium Enterprises (Refer Note 48)	53	42
Due to others	9,415	10,680
	9,468	10,722
Note 32		
Current - Other financial liabilities		
Current Maturity of Long Term Borrowings		
Non-convertible dehermares	118,794	
From banks	9,465	7,071
Interest- free security deposits from customers	3,341	9,924
Capital creditors	12,408	15,758
Interest accrued but not due on borrowings	8,079	7,959
Interest accrued but due on borrowings	253	
Retention money payable	34	256
Derivative Liability on put option	15,477	16,425
Financial liability for premium payment at fair value	1,050	1,800
Other financial liabilities	824	27
-	169,725	59,220
Note 33		
Other current liabilities		
Advances from oustomers	3,984	3,058
Statutory dues	1,143	1,318
Corpus Fund collection	676	623
Advance Maintenance Charges	95	243
Compensation on delayed possession payable	)(	54
Unearned rent	L,332	997
Security Deposit from Customers	64	64
Other current habilities	-49	5
-	7,354	6,362
Note 34		
Shurt-term provisions		
Provision for employee benefits		
Gratuity	82	22
Leave encasiment and compensated absences	263	92
Note 35		
Current tax liabilities (Net)		
Provision for Tax	9,149	9,540
Advance Tax and Tax Deducted at Source	(7,361)	(7,543
SKINS	1,788	1,997



## Notes to the consolidated financial statements (Continued)

(Currency: Indian rupees in lakhs)

	31 March 2019	31 March 2018
Note 36		
Revenue from operations		
(A) Sales of products		
Sale of residential flats (refer foot note below)	9,463	7,081
(B) Sale of services		
Construction revenue	88,760	57,934
Toll revenue (includes demonatisation claim receipt of Rs. 119.07 Lakhs)	12,819	11,321
Lease rentals income	41,870	37,425
Maintenance and other receipts	5,953	5,434
Utility income	6,243	5,726
Parking fees income	1,251	836
Project management consultancy fccs	1,123	1,574
Asset Management Charges Income	148	171
Revenue from other services	186	360
	167,816	127,862

## Foot Note:

Effective 1 April 2018, the Group has transited to Ind AS 115 and has availed exemption from retrospective restatement of revenue by adopting the modified approach as permissible under the transition provisions of the standard. The effect of initially applying this standard is recognised at the date of initial application (i.e. 1 April 2018). The standard is applied retrospectively only to contracts that are not completed as at the date of initial application and the comparative information in the statement of profit and loss is not restated (also refer Note 51 (a)).

## Note 37 Other income

Interest income (including unwinding of interest)	1,953	2,036
Interest on income-tax refund	74	336
Profit on sale of investments	273	161
Gain on re-valuation of Investments	100	54
Profit on sale of Property Plant and equipment	-	124
Foreign exchange (Forward Contract Gain MTM)	-	114
Gain on fair value of financial liabilities	-	65
Other income from residential projects	162	84
Fair value gain on derivatives (net)	3,329	-
Forfeiture of security deposits	4	22
Excess provision written back	177	-
Miscellaneous income	544	399
	6,616	3,395





## Notes to the consolidated financial statements (Continued)

(Currency: Indian rupees in lakhs)

		31 March 2019	31 March 2018
Note 38			
Cost of materials consumed			
Opening balance		256	394
Add: Purchase of materials		-	
Less: Closing inventories	-	(52)	(256)
	A j	204	138
Expenditure incurred during the year:			
Opening work in progress		41,485	43,268
Construction cost		3,319	2,459
Approval and permission expenses		70	28
Professional fees and technical fees		47	40
Other expenses	-	86	49
	В	45,007	45,844
	A + B	45,211	45,982
Less: Closing Stock of Inventories		40,959	41,485
Add: Transition adjustment pertaining to IND AS 115 application		1.870	-
Cost of flats sold		6,122	4,497
	1		-,
Note 39			
Construction cost		83,095	57,023
Note 40			
Employee benefit expenses			
Salaries, wages and bonus		7,787	7,725
Contributions to : Provident and pension funds		267	268
Staff welfare		341	476
Gratuny expenses		29	27
Compensated absences		126	49
Less: capitalised to investment property under construction		(217)	(194)
Less, capitalised to intangible assets under development		(74)	(121)
Less. capitalised to Inventories		8,259	(1) 8,229
Note 41		·	· · · ·
Finance costs			
Interest costs.			
- on term loans and NCD's		50,752	48,558
- on cash credit and overdrafts		33	1,818
- on commercial paper		6,140	5,505
MTM on forward contract		605	-
Finance charges		440	553
Unwinding of interest expense		3,756	1,568
less: capitalised to investment property under construction less, capitalised to intangible assets under development		(3,901) (5,770)	(3,131) (400)
		52,055	54,471
CKINS			24,171





Notes to the consolidated financial statements (Continued)

(Currency, Indian rupees in lakhs)

( · · · · · · · · · · · · · · · · · · ·	31 March 2019	31 March 2018
Note 42		
Other expenses		
Power and fuel	7,218	6,34)
Impairment loss (Refer Note 50)	750	5,219
Repairs and maintenance		
- Building	74	159
- Plant & Machinery	89	87
- Provision for Major Maintenance (Refer Note 30)	1,011	933
- Others	1,121	357
Operating and maintenance charges for infrastructure facilities	5,973	6,714
Fair value loss on derivatives	-	1,846
Advertisement and business promotion expenses	1,229	1,473
Legal and professional fees	2,257	1,536
Rates and taxes	1,129	1,122
Rent (Refer Note 47)	858	863
Travelling and conveyance	387	453
Bank charges	314	210
Telephone and communication expenses	157	149
Business development expenses	56	135
Training and recruitment expenses	127	122
Provisions and writeoffs:		
Excess service tax credit written off		122
Provision for doubtful debts	300	49
Bad debts	22	45
Advance written off	4.050	16
Audutor's remuneration	90	82
Brokerage	101	83
Insurance charges	101	78
Fees to NHAI	100	-
Printing, courier and stationery	40	63
Directors sitting fees	48	45
Loss on sale of property, plant and equipment	31	4)
Expenditure on Corporate Social Responsibility	177	40
Tender expenses	3	26
Membership fees	16	9
Compensation paid to customers	588	7
Miscellaneous expenses	138	225
<u> </u>	28,555	28,650
VINO		





Notes to the consolidated financial statements (Continued) for the year ended 31 March 2019

(Currency Indian supces in lakhs)

## Note 43

Earnings per share			
	_	31 March 2019	31 March 2018
Basic earnings per share			
Loss after tax attributable to equity shareholders	A	(23,693)	(40,198)
Number of equity shares outstanding at the beginning of the year		1,017,307,692	1,017,307,692
Equity shares issued during the year		-	-
Number of equity shares outstanding at the end of the year		1,017,307,692	1,017,307,692
Weighted average number of equity shares outstanding during the year (based on date of issue of shares)	В	1,017,307,692	1,017,307,692
Basic carnings per share of face value of Rs 10 each	(C ≈ A/8)	(2.33)	(3 95)
Diluted earnings per share			
Diluted earnings per share of face value of INR 10 each		(2.33)	(3 95)

## Note 44

Contingencies and commitments (i) Contingent Liabilities (Refer footnote 1)

Claims against the Company not acknowledged as debts	31 March 2019	31 March 2018
<ul> <li>Income TAX demands contested by the Group</li> </ul>	2,567	3,707
<ul> <li>Indirect TAX demands contested by the Group</li> </ul>	467	467
- Claims made by contractors	1,179	1,104

Foot Notes

1 The Group does not expect any outflow of economic resources in respect of the above and therefore no provision is made in respect thereaf

## (ii) Letter of comfort and going concern

(a) The Parent Company has issued letter of comfort to banks in respect of loans availed by a few of its subsidiaries

	Name of Subsidiaries	Nature of Comfort given
	Mikado Realtors Pvt. Ltd	Shortfall undertaking to meet any shortfall during the renure of facility
	International Infrabuild Private Limited	To ensure payment to debenture holders in the event of termination of the concession "agreement.
li li	i TRIF Real Estate Development Limited	Undertake to meet cost overrun to the extent of 10% of project Cost.

(b) The Parent Company has issued financial support letter to following subsidiaries and joint ventures on the basis of which the separate financials statements have been prepared on going concern basis

- Acrea Living Solutions Private Limited
- MIA Infrastructure Private Limited
- in Welkept Facility Mangement Services Private Limited (Previously known as TRIL Hospitality Private Limited)
- Iv TRIF Gurgaon Housing Projects Private Limited
- v TRIL Amntsar Projects Limited (formerly known as TRIF Amntsar Projects Limited)
- v) TRPL Roadways Private Limited
- vii Gurgaon Constructivell Private Limited
- vill HV Farms Private Limited





Notes to the consolidated financial statements (Continued) for the year ended 31 March 2019

(Currency Indian rupees in lakhs)

## Note 45

## Capital and other commitments

Particulars	31 Mareb 2019	31 March 2018
Estimated amount of contracts remaining to be executed on capital account and not provided for (net of advances)	60,491	178.382
Others		

Note 46

## Segment reporting A. Busis for segmentation

The Group has three reportable segments, as described below, which are the Group's strategic business units. These business units offer different products and services, and are managed separately because they require different technology and marketing strategies

The following summary describes the operations in each of the Group's reportable segments

Reportable segment
 Overlappenet of Residential property for outright sale

- Real Estate

- Infrastructure

## B. Information about reportable segments

Information regarding the results of each reportable segment is included below. Performance is measured based on segment profit (before tax), as included in the internal management reports that are reviewed by the Group's CODM. Segment profit is used to measure performance as management believes that such information is the most relevant in evaluating the results of certain segments relative to other entities that operate within these industries inter-segment pricing is determined on an arm's length basis.

Refer Annexore A

## C. Information about major customers

There are no customers from whom Group recognises revolue more than 10% of total revenue of the Group





Notes to the Consolidated financial statements for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

91-	
Refer Note No. 4	Апасхиге А

	Development of residential	of residential	Real Estate	state	InfraStructure	ucture	Unallocated	cated	Total	al
Particulars	31 March 2019	31 March 2018	31 March 2019	31 March 2018	31 March 2019	31 March 2018	31 March 2019	31 March 2018	31 March 2019	31 March 2018
REVENUE Net sales										
Revenue	10,253	7.830	55,278	55.459	102,328	64,061	6,573	3,907	174,432	131,257
Fotal revenue	10,253	7,830	55,278	55,459	102,328	64,061	6,573	3,907	[74,432	131,257
Expenses	8,236	6,789	17,123	26,628	89,240	53.927	11,418	11,055	126,018	665.86
RESULT Finance costs Denecisation/Immairment		21	22,406 15,166	25,439 14,490	6,662 1,550	5,455 2,965	22,988	23,577 124	52,055 16,858	54,471 17,600
Net loss before Taxes									(20,512)	(23,213)
OTHER INFORMATION										
ASSETS Segment Assets	43,382	45370	563,247	506.576	290,662	201,377	85,329	124,894	982,621	878,217
Total Assets	43.382	45,370	563,247	\$06,576	290,662	201,377	85,329	124,894	982,621	878,217
LLABILITIES Segment Liabilities	6,691	549.5	367,457	380.981	225.726	170.120	353,234	266,738	953,109	823.534
Total Liabilities	6,691	5,695	367,457	380,981	225,726	170,120	353,234	266,738	953,109	\$23,534





Notes to the consolidated financial statements (Continued) for the year ended 31 March 2019 (Currency: Indian rupees in takins)

## Note 47

## Lease airrangements

A Operating leases - as a lessee

a) The Company has taken various premises under cancellable operating leases.

b) The Company has also taken a commercial property and certain residential premises on non-cancellable operating leases. The future minimum lease payments in respect of these properties as on 31 March 2019 is as follows.

Lease Payments	31 March 2019	31 March 2018
Not later than one yeas	424	328
Later than one year but not later than five years	791	243
Later than five years		-
Payments of lease rentats during the year (includes recorded under personnel costs (NR 29,16 lakh (2018, 47,88 lakh)	426	430

c) There are no exceptional/restrictive covenants in the lease agreements

## B As a lessor:

Cinema las

## (i) Operating Lease

The Company has leased some investment properties. As on 31 March 2019, the future minimum lease payments in respect of these properties till the expiry of lock in period is as follows.

	31 March 2019	31 March 2018
Not later than one year	33,891	31,252
Later than one year but not later than five years	140,618	74,205
Later than five years	42,081	22,954
	216,590	128,412
Lease Rental Income for the year	53,431	48,214

## (ii) Finance leases - Fit-out and interior work

mahle as falles

The Company's leasing arrangement represents the fit-out of interior work completed for the customers which have been classified under Ind AS 17 on Leases as Finance lease. The lease terms are for the periods of five to seven years where substantially all the risks and rewards of ownership are transferred to the lessee. The Company records disposal of the property concerned and recognizes the subsequent interest in the finance lease. No contingent rent is receivable in this regard.

Gross investment in lease	31 March 2019	31 March 2018
Receivable within one year	166	166
Receivable between one and five years	152	318
Receivable after five years		•
	318	484
Present value of minimum lease payments	31 March 2019	31 March 2018
Not later than one year	114	94
Later than one year but not later than 5 years	125	239
Later than 5 years		-
	239	334





## Tata Realty and Infrastructure Limited Notes to the consolidated financial statements (Continued) for the year ended 31 March 3019 (Currency: Indian rupees in Inkins)

## Note 48

## Micro. Small and Medium Euterprises

Under the Micro, Small and Medium Enterprises Development Act, 2006, (MSMED) which came into force from 2 October 2006, certain disclosures are required to be made relating to Micro Small and Medium enterprises. On the basis of the information and records available with the Management, there are no outstanding dues to the Micro. Small and Medium enterprises as defined in the MSMED as set out in following disclosure.

	31 March 2019	31 March 2018
Principal amount remaining unpaid to any supplier as at the year end (o/s, for less than 30 days)	\$3	42
Interest due thereon	-	-
The amount of interest paid by the buyer as per the Miero Small and Medium Enterprises Development Act, 2006 (MSMED Act, 2006)		•
The amounts of the payments made to nucro and small suppliers beyond the appointed day during each accounting year	•	
The amount of interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under MSMED Act, 2006		
The amount of interest accrued and remaining unpaid at the end of each accounting year	-	
The amount of further interest remaining due and payable even in the succeeding years, until such date when the interest dues as above are actually paid to the small enterprise	-	-
Amount of interest accrued and remaining unpaid at the end of the accounting year		

Note 49		
Expenditure in foreign currency (on accenal basis)	31 March 2018	31 March 2018
Professional fees	15	Z4
Training and conference expenses	12	62
Travelling expenses	4	7
Sales and Marketing expenses	· ·	

## Note 50

Impairment Loss

The group has tested the impairment of goodwill arising on account of consolidation / merger. Based on the performance of the project during the current year and future projections, an impairment analysis had been carried out during the year and impairment loss recognised during the year of Rs 750 Lakhs (2017 Rs, 5.219 Lakhs)

## Note SI

## a) Disclosure with respect to transition adjustment of IndAS 115

Particulars	Amount
Opening Retained Earnings (before Ind AS 115)	(74,541)
Reversal of revenue	(3,174)
Reversal of Cost of sale	1,870
Opening Retained Earning (After Ind AS 115)	(75,845
lucrease in Inventory	1,876
Decrease in Trade Receivable (Other than related Party)	(3,174

b) Disclosure in respect of Construction Contracts

Particulars	For the Year ended March 31, 2019	For the Year ended March 31, 2018
Contract revenue recognized as revenue during the year	88,760	57,934
Particulars	As at March 31, 2019	As at March 31, 2018
Cumulative revenue recognized	170,979	82,219
Cumulative costs incurred	163,942	80,847
Cumulative margins accounted	7,037	1,371
Advances paid	11.907	16,744
Retention money payable	-	-





Notes to the consolidated financial statements (Continued) for the year ended 31 March 1019 (Currency, Indian rupees in lakhs)

Note 52

## Interests to other entities

(a) Subsidiaries The company's subsidiaries at 31 March 2019 are set out below

	Place of	% of owner	rship interest
Name of entity	business	31 March 2019	31 March 2013
Acme Living Solutions Private Limited	India	100.00%	109,00%
Arrow Infraestate Private Lumited	India	100 00%	100,00%
Gurgaon Constructivell Private Limited	India	100.00%	100 00%
Gurgaon Realtech Limited	India	100 00%	100.00%
TRIL Roads Private Limited	India	1.00.00%	100.00%
TRIF Gurgaon Housing Projects Private Limited	India	100 00%	100,00%
TRIL Urban Transport Private Limited	India	100.00%	109.00%
TRIL Hospitality Private Limited	India	100.00%	100.00%
TRIL Constructions Limited	fischia	67.50%	67.50%
TRIL Infopulk Linuted (refer note no. (ii) & (iii)	India	100.00%	100.00%
TRIL Amotsar Projects Limited	India	100.00%	100,00%
Hampi Expressway Private Linuted	India	100.00%	100.00%
TRIF Real Estate Development Limited	India	100.00%	100,00%
HV Farms Private Limited	India	100.00%	100.00%
International Infrabuild Private Linuted (UPL)	(ndia	26.00%	26.00%
Dharamshala Ropeway Lamited	fadja	74.00%	74.00%
Manali Ropeway Private Linuted	յուրն	72.00%	72,00%
Uchat Expressways Private Limited	India	100,00%	100.00%
TRPL Roadways Private Limited	India	100.00%	100.00%
Matheran Ropeways Private Limited	India	70%	70%
Durg Shivnath Expressways Private Limited	նուան	100%	100%
MIA Infrastructure Private Limited	India	100%	100%

Significant judgement: consolidation of entities with less than 50% voting interest i) Although the Group owns less than one-half of the voting power of International Infrabuild Private Limited, it is able to control the company by virtue of an agreement with the other investors of International Infrabuild Private Limited which inter-alia provides the Group with power to appoint majority of the heard of directors of International Infrabuild Private Limited, it is able to control the company by virtue of an agreement with the other investors of International Infrabuild Private Limited which inter-alia provides the Group with power to appoint majority of the heard of directors of International Infrabuild Private Limited. consolidates its investment in the company

(1) The Parent company has entered into call option with one of the shareholder of TRLL Infopark Ltd expring on 10th July 2021 wherein the Parent Company holds the call options. Company has paid full consideration of Rs 7110 Lakhs as option deposit. Hence Parent company is consolidating shareholding in TRLL Infopark Ltd of the said shareholder as stake held by it

iii) Similary, the Parent Company has entered into put option with another shareholder of TRIL Infopark Ltd exptring on 9th March 2019. Based on the put option conditions Parent Company has provided full liability under other current financial liabilities, since the put options rights are available with the other shareholders. Hence, Parent Company is consolidating TRIL Infopark Ltd as 100% subsidiary,

(b) Non-controlling interests (NCI) Set out below is summarised financial information for each subsidiary that has non-controlling interests to the group. The amounts disclosed for each subsidiary are before inter-company eliminations

Summarised balance sheet										
	Dharaoshala R	opeway Limited	TRIL Construction	as Limited	Manali Ropew	nys Private		PL	Matherau	Rope- Way
					Limit	ed			Private	[/turited
	31 March 2019	31 March 2018	31 March 2019	JI March 2018	31 March 2019	31 March 2018	31 March 2019	31 Marsh 2018	31 March 2019	31 Mareb 2018
Non-current assets	7,168	5,482	21,691	21,487	5,144,64	5,749	13,645	12,937	1,587	1,587
Current assets	108	278	2,648	2,657	4 70	1	605	287	951	954
Non-current liabilities	(5.641)	(4,077)	(8,615)	(7,050)	(4,934.02)	(5,510)	(6.474)	(10,849)		-
Current habilities	(1,132)	(682)	(289)	(1,148)	(9.18)	(28)	(9,701)	(2,389)	(101)	(129)
Net assets	504	1.001	15,434	15,946	206	212	(1,925)	(13)	2,436	2,411
Net assets attributable to NCI *	104	260	6,304	6,470		(27)		(79)		201

	Dharamshala R	apeway Limited	TRIL Constructio	ns Limited	Manali Ropew Lindi	-	11	ւ		Rope- Way Limited
	31 March 2019	31 March 2018	31 March 2019	31 March 2018	31 March 2019	31 March 2018	31 March 2019	31 March 2018	31 March 2019	31 March 2018
Summarised statement of profit and loss					_					
Revenue	2.11	116	-	-		-	136	5,831	55	37
(Loss)/Profit	(649)	77	(5)2)	(28)	(55)	(119)	(1.912)	(95)	25	(0)
OCI		-		-		1.1				-
Total comprehensive income	(649)	22	(512)	(28)	(55)	(119)	(1,912)	(95)	25	(0)
Loss allocated to NCI	(169)	20	(166)	(9)	(15)	(33)	(1.415)	(70)	7,38	(0.06)
OCI allocated to NCI				-		-		- 1		
Total comprehensive income allocated to NCI	(169)	20	(166)	(2)	(15)	(33)	(1,415)	(70)	7	(0.06)
Loss to NCI Restricted	(169)	20	(166)	(9)	27	(33)	79	(70)	7	(0)
Summarised cash flows										
Cash flows from operating activities	(171)	(231)	(723)	(238)	(10)	(23)	(147)	(378)	(60)	(38)
Cash flows from investing activities	(1,421)	(2.385)	(204)	(\$43)	(34)	(158)	(1,420)	(6,860)	54	(742)
Cash flows from financing activities	1,634	2,613	895	628	48	179	1,990	7,255		1,715
Net increase/ (decrease) in cash and cash equivalents	42.39	(2.55)	(32.65)	47.79	4.08	(1.35)	423.37	17.55	(6.08)	935.31





Notes to the consolidated financial statenents (Continued)

for the year orded 31 March 2019

(Currency Indian rupees in takhs) Note 52 (continued)

## Interests in other entities (continued)

(c) Transactions with Non-controlling interests

# These are no transactions with non-controlling interests in 2018- 2019

(d) Interests in joint vontores Set out below are the joint remutes of the group as at 31 March 2019 which in the opinion of the descore are material to the group. The country of morporation or regaration is also their principal pass of besiness.

	Place of	Place of %. of ownership		
Name of entity	business	interest		
Pune Solapur Expressmans Private Littated	ndia	50%	Joint venture	Joint venture Equity method
A & T Road Construction Management and Operation Physics		Ervel.		
Limited	India	12/20	Joint venture	Jemt venture Equity method
TRUU JF4 Private Lunuted	lindiz	74%	Joint ventuse	Joint venture Equity method
Mikado Realtors Pv1 Ltd	elpul	21%	Jourt venture	Jount venture Equity method
ladustical Minerals and Chamicals Compary, Private Lurghed	terdia	74%	Joint venture	Joint venture Equity method

(1) All the above joint ventues are engaged in the biomess of real estate and infrastructure development. It is a strategic in extreme which utilises the group's knowledge and expertise in the development of real strate and infrastructure development.

(ii) Significants padgment: Although the Group own: more than one-half of the voting power of TRU. IT4 Private United, Mittado Realtors Pvt. Ltd. and Industrial Minurats and Chemeals Company. Private Limited, it does not have control over these Company: The Graup has joint control over these companies by viftue of an agreement with the other nivestors. In other words decisions about the relativities - i.e. those that significantly alfort the returns of the arrangement - require the affirmative concerts of the investors Consequently, the Group has described its mercet in these companies as jourity controlled entrity.

Summarised fittometal information for associates and joint ventures

Latical         Magnetic and operation         (Ader Name)         (Ader Name) <th></th> <th>Punc Solapur Ex</th> <th>Punc Solapur Expressways Private</th> <th>A &amp; T Road</th> <th>A &amp; T Road Construction</th> <th>WI</th> <th>INCC</th> <th>TRIL I'' Pri</th> <th>vate Limited</th> <th>TRUL ITA Private Limited Miloudo Realtors Pyt Utd</th> <th>ors Pyt Lid</th> <th>Tolal</th> <th>-</th>		Punc Solapur Ex	Punc Solapur Expressways Private	A & T Road	A & T Road Construction	WI	INCC	TRIL I'' Pri	vate Limited	TRUL ITA Private Limited Miloudo Realtors Pyt Utd	ors Pyt Lid	Tolal	-
		Cla	lited	Managenicul Private	and Operation Limited			Keler	(01c 5)				
erroring entering interim         SM         SM <th< th=""><th></th><th>31 March 2019</th><th>31 Marsh 2018</th><th>31 March 2019</th><th>31 March 2018</th><th>31 March 2019</th><th>31 March 2015</th><th>31 March 2019</th><th>31 Мансіі 2018</th><th>31 March 2019</th><th>31 March 2018</th><th></th><th>31 March 2018</th></th<>		31 March 2019	31 Marsh 2018	31 March 2019	31 March 2018	31 March 2019	31 March 2015	31 March 2019	31 Мансіі 2018	31 March 2019	31 March 2018		31 March 2018
	Percealage องรถะรางไม่ๆ intertst	20%	%05	\$0.5	20%	ĺ	74%	%†L	24%	24%	74%		
	Current assets	2417	3,654		0	58	16	3,224	2.822	\$	32	9,262	6,523
	Non-cutteri assets	127.66	532.79			16.460	15,492	29.232	35.(1)7	29,353	7,535	168 266	155,407
	Current babulities	9,520	2.620	¢1	_	25	23	3,069	3.025	2,966	816	- 15,5KI	- 6,485
Ite steel         114         5 44         (1)         (6,93)         135         (1,2,53)         1432         435         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3318         236         3345         236         3345         236         3345         236         3345         236	Non-current habilities	\$55,333	92.776	ı			[3,]3]	52 109	51,852	21,940	2,266	159 382	160 026
Terration         2.173         2.770         (1)         (1)         (1,1,1,1)	Min	17 E 1	2.6.1	(2) (2)		16.407	1.25.1	1542 661	116 050		A 3965	1354	145.14
Andress         All Reception         2,173         2,730         (1)         (0)         2,0,10         2,1,21         2,1,41<	Crosses of man avents	1 F C		96	e e	12,206	112	(16.835)	(12,550)		3 3 18	856	(022 P)
Arrying amound of Interest in joint venture         2,72         2,70         (1) <td>Coolwil Recombed</td> <td></td> <td>,</td> <td>5.</td> <td></td> <td>9.705</td> <td>20.179</td> <td></td> <td></td> <td>~</td> <td>27.141</td> <td>36.546</td> <td>47.520</td>	Coolwil Recombed		,	5.		9.705	20.179			~	27.141	36.546	47.520
Pure Subput Expressways Private     A & T Road Construction     INCC     TRUL IT 4 Private Limited     Misculo Realtons Pri Lid     Total       Limited     Lanisted     Management and Operation     INCC     TRUL IT 4 Private Limited     Misculo Realtons Pri Lid     Total       numaristed Statement of profit and loss     31 Alarch, 2018     31 March, 2018     31 March, 2018     31 March, 32 March, 31 March,	Carrying amount of interest in joint venture	2/1/2	2 770	Ξ	(0)	016'1Z	21,921		'	30,436	30,459	54.516	55,150
		rune soupur ext mil	ited	Manuerona A	Construction and Overation			Befer	date 9)		<b>הויד וא ז כו</b> מ		=
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		CI03	1940	Private	ana operanan Limited			(nates	1016 21				
Answer         Style         Style         Style         Table         <	Summan should stationant all months and been	36 March, 2019	51 March, 2018	31 March, 2019	35 March, 2018	31 March. 7619	31 March, 2018	31 March, 2019	31 March, 2018	31 March. 2019	3J Ма <del>к</del> b, 2018		31 March, 2018
Contention         15.52         14.058         -         -         1         7,864         7.457         -         -         22,3416           Operation ind anonustion $467$ $3.717$ -         -         -         25,3416         -         -         25,3416         -         -         25,3416         -         -         25,3416         -         -         -         0,000         -         -         -         0,000         -         -         -         -         0,000         -         -         -         -         -         -         -         -         -         -         0,000         - <t< td=""><td></td><td>50%</td><td>%05</td><td>20%</td><td>%05</td><td>24%</td><td>74%</td><td>74%</td><td>74%</td><td>14%</td><td>24%</td><td></td><td></td></t<>		50%	%05	20%	%05	24%	74%	74%	74%	14%	24%		
Operation $-1687$ $3.717$ $\cdot$	Revenue	15 552	14,085			,	-	7,864	7467	,		13.416	21.557
interst express $8,651$ $8,841$ ·         · $5,592$ $5,461$ ·         · $14,247$ nonze tax express         (1,98)         (1,91)         (2,01)         (1)         (1)         (2,02) $5,461$ ·         ·         [1,2,17]           nonze tax express         (1,198)         (2,010)         (0)         (1)         (15)         (2,91)         (5,73)         (5,73)         (7,2)         (7,00)           non preference         (1,197)         (2,010)         (0)         (1)         (15)         (3)         (16)         (7,00)           non preference nonce         (1,197)         (2,010)         (0)         (1)         (1)         (6)         (7,01)         (6)         (7,01)           non preference nonce         (1,197)         (2,010)         (0)         (1)         (6)         (7,01)         (7,01)         (6)         (7,01)         (6)         (7,01)         (6)         (7,01)         (6)         (7,01)         (6)         (7,01)         (6)         (7,01)         (6)         (7,01)         (6)         (7,01)         (6)         (7,01)         (6)         (7,01)         (6)         (7,01)         (6)	Depreciation and amorusation	180	717.E	٠		•		6.841	7,055	•	,	190.928	10,773
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Interest expense	8,654	8,841	•		•	,	5,592	5,461		•	14,247	14,302
Alse comprehensive incomes         (1,198)         (1,2010 23)         (0)         (1)         (1,2)         (2)         (5,734)         (32)         (16)         (7.008)           Mare comprehensive incomes         1         1         -         -         -         -         1         -	Income tay expense		`				,		(362)	•			(202)
More completensive income     1     1     1       Ordel completensive income $(1,197)$ $(2,010)$ $(0)$ $(1)$ $(13)$ $(3,714)$ $(32)$ $(16)$ $(7007)$ Ordel comprehensive income $(1,197)$ $(2,010)$ $(0)$ $(1)$ $(13)$ $(3,714)$ $(32)$ $(16)$ $(7007)$ Promp 5 share of loss $(599)$ $(1,003)$ $ (0)$ $(11)$ $(6)$ $   -$	Loss	(861'1)	(2.010.23)	9	e	(51)	6	(5.763)	(5,974)	(32)	(je)	(7,008)	(8,010)
Ordl comprehensive measure     (1,197)     (2,010)     (0)     (1)     (15)     (9)     (5,764)     (3,274)     (32)     (16)     (7,007)       Promp's share of loss     (1005)     -     (0)     (11)     (6)     -     (12)     (6.3)       Promp's share of loss     (2,010)     -     (0)     (11)     (6)     -     -     0       Promp's share of OCI     0     1     -     -     -     (11)     (5)     -     -     0       Promp's share of OCI     0     1     -     -     -     -     -     -     0       Promp's share of Iolal comprehensive inceme     (19,04)     -     -     -     -     -     -     0	Other comprehensive income	-			'		,	•		,	,	-	
Shower's share of loss     (599)     (1.005)     -     (0)     (11)     (6)     -     (24)     (12)     (633)       Prover's share of OCI     0     1     -     -     (1)     (6)     -     -     0       Prover's share of Iopal comprehensive incease     (1)     1     -     -     -     0     0       Prover's share of Iopal comprehensive incease     (598)     (1,004)     -     (0)     (11)     (6)     -     1     (12)     (633)	Total comprehensive income	(1,197)	(2.010)	(0)	0	(51)	(6)	(S.763)	(5.974)	(32)	(91)	(7.007)	(600)\$)
Arrent's share of OC1	Group's share of loss	(565)	(\$007)		(0)	(II)	(9)		•	(54)	(12)	(633)	(1,023)
Arver's share of lotal comprehensive increase (598) (1.004) - (0) (11) (6) (2J) (12) (633) -	Group's share of OCI	0	-		'	•	,		•	ı		¢	-
	SKINS & Where 's share of total comprehensive incame	(\$65)	(1.004)	'	(0)	(II)	(9)		I	(rz)	(13)	(633)	(1,023)



up fast subscribed to the memorandum of Pund IT City Metric Rail Lumied w E. March 13, 2019 (or 74% stake to share capital of the antity, having equal participative rights with the juint venture partner

Eit Page S

Chartered

31110

Accountants

## Tata Reality and Infrastructure Limited Notes to the Generalized financial sustemants for the scar world \$1 klassis 2019 (Curreans, indum rupets in faktur

Note 53 Financial instrumenta – čnar naliset pusi risk unstagen out

A. Accounting classification and fair values

	Carryland				Fair value			
31 March 19	PYTPL	PVTOCI	Amurtised Cost	Tom	Level 1 - Quatrif price in active markets	Level 2 - Significant abservable inputs	Level J - Significant unotservable Inputs	Tatal
Pluaneta) assets								
Investments				2.9				
<ul> <li>Equity instruments in others</li> </ul>	32			22			22	21
Bonds and debentures			18,614	18.6-14			18,614	18,614
Corrent investments in mutual funds	1 383			1.383	1,78,1			1,383
Inade recessables			2.846	2.846			2,846	2,840
Cash and eash equivalents	1		5.921	\$.921			5.931	5,92
Logise real receivable			193	193			193	193
Unbilled revenue			\$37	532			\$37	537
Deposit other than raluated in crish and crish equivalent			2,079	2.079			2.079	2,075
Security deposits			730	730			7,30	73(
Other Jouns and advances			3.054	3.054			3.054	3,05-
Otier financeal assets			1.696	1,696			1,696	1,696
	1,405	-	35,670	37,075	دهدا	-	35,692	37,075
Financial liabilities	1							
Non convertible dehentures			233,586	233 584			233,586	233.5%
Financial liability fee premium payment			69 110	69.110			69 110	69,110
Project development fees payable			45	45			-15	
Secured loan and working capital facility from Bank			444 949	441,939			444,149	444,949
Commercial paper associd to mutual fond			76 567	76,567			74,567	76,561
Commercial puper issued to banks			- E.					
inter Corporate Deposite from Related Parties			14,000	14,000			14,000	14,006
Derivative pistromenta	l l						- parter	14,001
Pat option.	17,295			17,295			17 295	17,294
Forwards	125			425			425	429
interest- tree secondly deposits them existences			21 551	21.551			25 55	21.55
Other financial labelities			16,305	36.305			36. 205	36,30
Trade pavables.			9,468	9,468			9.463	9,468
Barry State State (1970)	17,720		909.699	927,419			927,419	927,419

	Carrying				Fuir value	- 2.5		
3) March 18	FVTPL	FVTOCI	Autoritized Cost	Total	Lovof I - Quasted price In active markets	Lovel 2 - Significant observable Inputs	Level 3 - Significant unobservable inputs	Total
Fittancial magts								
Investments								
- Equity instruments in others	24			34			24	2
- Benda and debentures			8,140	8,140			8,140	8,14
Current investments in motual fitteds	א <u></u> כע ד	-		7,958	7,958			7,955
Trade receivables	· -	-	5,787	5.787			5 787	5,78
Cosh and cash equivalents	· .	-	7,141	7,141			7,145	7,14
Lease tent eccevable	· .	-	94	94			9.6	5
Unbilled revenue	· ·		102	102		-	[u2	160
Deposit other thiss toluded in main and easily equivalent	· ·		720	720			720	72
Scenty deposits	· ·	-	1,202	1.299		-	1,2300	6,20
Other boons and advances	-	-	5,020	5.020		-	\$,020	5,02
Other financial assets (Forward Deravate Contract Rs. 114 Laths)	•		1.278	1,218	•	-	1.218	1,211
	7,982		29,423	37,405	7,959	-	29,447	37,40
Financial fieldities								
Non convertible debensures	· .		231,439	231,439			231.439	231,43
Financial liability for premium payment	i .		65,869	45.869			45,869	65,865
Project development fees payable			45	45			45	4
Secured loan and working onpital facility from Bank			357,459	357.459			157 459	357.45
Constitutional paper usued to matical fland			59,695	\$3.695			39 495	59,69
- Put netion	17.923			17,923			17.923	17.93
- Forwards	3 128			3.128			3.128	3,12
Interest- free security deposits from customers			21.997	21,997			21.997	21,99
Other financial leabilities			22 44%	32.348			32.548	32.54
Trade payables			(0.722	10,722			10,722	10,72
	21,051		779,775	800,825			8041,825	800,82





Notes to the Consulidated financial state for the year ended 31 (Jarch 2019 (Currency Indian inpres (a lakha)

## B. Monsterement of fair values

Volumino techniques and significant anoheers able logates The followorp table shows the volumines techniques used in incoming Level 2 and Level 3 for ratives for financial instruments measured at the value in the sutement of financial position as well as the significant molecurable inputs used

210	Voluation technique	Significant unobservable inputs	Inter-relationship between significant undeservable inputs and fair value measurement
Innucial assets			
en estiments Equate instruments in others (GPG Provid)	Discounted cash flow technique: DCF methodinarysis is a finnesi modeling technique based on expirite assumptions regarding the prospective income areing out of the development to be carried out on the subject land parcel in case of a valuation of a lange and parcel like the subject property, where the development, potential is realized over a period of true (i.e. time value of morely comes) available for comparison, DCF method considering technism developments of the problet as used.		Not avajuile
Equity instruments in others (1)(1)	Discounted cash Dow technique: DCK methodiamiyas is a financial modeling technique based on explicit assurptions regarding the pispective income arising out of the development to be carred out on the subject lend parcel is case of a velucition of alrupa tand parcel like the subject picpetry, where the development potential is replated over a previoo of the tak line value of money comes into the potucy and also where there are not dress members sumparable for comparation (are comparable) available for comparison, DCF mainted considering relevant potential developments of the project a used.		Plot nominhte
Perkalise instruments Put option - PGBJ,	The Contrany has used Monte Carlo simulation to		2019
	estimate the fair value of the options. Monte Callo Simulation: The variantion resolution annulates the equaty value per slate of the indefeoting securities considering (i) a risk, free rose rule and B() velatility of took prices of aleatical securities in matches. The expected pay-off is determined ensakering the constructed article prices of the explor and the simulated equity value per shume at each node. The fair value of the eption as relationships the rule for similar equition is to whatever node! works on a task netural framework.	*Traderbring county value of the commony (Re.3.6.2 failer) 2018 * Updoriving optic value of the communy (Re.4.609 Tables)	*If or advance enterprise value of the company increases (decrease) by 10%, then fair value would (decrease rimorease by INR (10*a)/10*a 2018 *11 underlying enterprise value of fait company increases (decreases) by 10%, then fair value would (decrease) increase by INR (10*a) 10*a
Forwards	The Company has used Monte Carlo simulation to estimate the fait value of the options. Monte Carlo Simulation: This values on the option involutes the equity value per above of the underlying sectoristic sconaderum (1) a risk free role rate and (1) valuelity of tack prices of identical accurates in markets. The expected proved fi's determined considering the entranced strike price of the option with the inimized end option in estimated by disconning the entranced strike price of the option with the inimized end option in estimated by disconning the entranced strike price of the option with the inimized end option with per- dure at each node. The fait value of the option ro- dition prodet works on a risk neoreal transventk.	* Underlying Equity value of the company (Rs. 590 laklas)	2019 HT materising enterprise value of the estimate interanse (decreases) (is 10%, then fair other would (decrease)) increase (interaction) 2018 HE understring enterprise value of the company increases (decreases) (interaction) (decrease)) increase (interaction) (decrease)) increase (interaction) (decrease)) increase (interaction) (interac
Fullenting have been recorded at ancortised cast: 1. Investments in Non-convertible debantures of equity accounted investors 2. Long rent receivables 3. Security deposit given 4. Non convertible debantures (berrowings) 5. Piranett labelity for prensum parquent 6. Security deposits recorded from constrance 7. Janual Nadu Industrial Corporation Linuted ("TIDCO") Put option	Discounted each flow approach: The saturation model considers life persent value of especied payment, discounted using a resk adjusted discound rate.	Not applicable	Not applicable

Financial risk management The Company has expressed to the following risks arrange from timescal distributions A. Credit risk B. Liquidity not, and C. Market risk

Risk management framework The Company's based of directory has overall tesponsibility for the satablishment and overagin of the Company's risk numagement framework. The Company namages market risk through a treasury department, which evaluates and ever(test satables and ever the rule over the rule overaging of the company's risk numagement framework. The Company namages market risk through a treasury department, which evaluates and ever(test satables and policies, which are approved by Board of Directors. The set while a of this department include management of cash resources, horizoning strategies, and containing compliance with market risk limits and policies.

The Company a trik numerator policy and addresses are established to identify and analyze the trike incode to the Company, to set appropriate risk invites and constrols and to monitor trake and adherence to finitis. Risk management policies and systems are reveved repolarly to tellest damper in market conduction and the Company and reactions and sustained and constrols and adherence to finitis. Risk management policies are reveved repolarly to tellest damper in market conduction and the Company are reveved repolarly to tellest damper in market conduction and the Company are revealed in the control externation of the company and management standards and proceedures are to maximize a dasaptered and control externation of the company and management standards and proceedures are to maximize a dasaptered and controls externation of the company.

The audit committee overaces how management monitors compliance with the company's rask management policies and provolutes, and reviews the adspace of the risk management framework in relation to the risks faced by the Company. The audit committee a assisted in its oversight role by internal and i internal and is a series of the risk management reactions in the results of which are reported to the and i committee





## Tata Realty and Infrastructure Limited Notes to the Consolidated financial statements for the year ended 31 March 2019

(Currency Indian rupees in lakhs)

## Note 53 (continued)

## Liquidity risk

Lequidity risk is the risk that the Group will not be able to meet its financial obligations as they become due. The Group manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring inacceptable losses or risk to the Group's reputation

The Group has obtained fund and non-fund based working capital lines from various banks. Furthermore, the Group has access to funds from compulsory convertivitie and non-convertible debentures from holding Group and other financial institution. The Group also constantly monitors funding options available in the debt and capital markets with a view to maintaining financial flexibility

## Exposure to liquidity risk

The table below analyses the group's fractical inhibitings into relevant maturity groupings based on their contractual maturities for \* all non-derivative financial liabilities.

\* net and gross settled derivative foranoid instruments for which the contractual maturities are essential for the understanding of the timing of the cash flows

31 March 2019 Non-derivative financial liabilities		Co	ntractual cush flows		
	Carrying amount	I year or less	1-2 years	2-5 years	More than 5 years
Non convertible debentures	233 586	118.820	40,000	45,500	24,650
Inter Corporate Deposits from Related Parties	L1 (HH)	14,000			
Finalitial liability for premium payment	69,110	1,316	17 1-30	13-485	322 010
Project development fees payable	.45	45			
Bank Overdraft	4,115	4,118			
Interest- free security deposits from customers	21,551	3,341	5,059	13,773	4,726
Commercial paper issued to mutual fixed	76 567	77,500			
Trade and other payables	9 468	9,465	-	-	-
Secured loan and working capital facility from Banks and Financial Institutions	444 040	13 365	41,0KU	122 3:05	286,0**
Other Gnancial habilities*	36,305	34.865	•	11.439	•
	909,699	266.838	83,854	206,591	642,441
<b>Derivative financial habilities</b>					
- Put option	17 295	15.477	-	1 \$18	
- Forwards	425			425	
	17.720	15,477		2,243	-

31 March 2018					
Non-derivative financial Kabilities			Contra	actual cash Nows	
	Carrying anoman	l year or less	I-2 years	2-5 years	More than 5 years
Non convertible debentures	231,439	6,044	131,240	108,515	18,135
Financial liability for premium payment.	152,8694	1.855	4,786	30,500	259,024
Project des elepment fees payable	-15	45			
Interest- free securary deposits from customers	24,007	7,409	1 795	13.084	4,676
Commercial paper assued to mutual fund	sa dats	143,5FR)			
Trade and other payables	10,722	48,722			
Secured loan and working capital facility from					
Rank	137,454	13,331	17,876	133,754	152,497
Other financial liabilities	12,548	27,617		4,931	-
	779,775	147,224	175,698	291,780	434369
Derivative financial III dubilities					
- Put option	17 923	10.425		-	1,498
- Ferwards	5 128			3.128	5 S
	21,051	16.425		3,128	1,498

\* Forward contract entered with the bunks with respect to the first commutation of supply of assets services estaid with the foreign contractor to be constructed and installed on the project site. Notional amount of foreign currency Euro 59.40 taking and notional INR 114.55 laking





Notes to the Consolidated financial statements for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

Note 53 (continued) Credit Risk

Credit risk is the risk of financial loss to the Group if a customer or counterparty to a financial instrument fails to meet its contractual obligations, and arises principally from the Group's receivables from customers. Credit risk is managed through credit approvals, establishing credit limits and continuously monitoring the creditworthiness of customers to which the Group grants credit terms in the normal course of business. The Group establishes an allowance for doubtful debts and impairment that represents its estimate of incurred losses in respect of trade and other receivables, loans and advances and investments.

## Trade and other receivables

The Group's exposure to credit risk is influenced mainly by the individual characteristics of each customer. The demographics of the customer, including the default risk of the industry and country in which the customer operates, also has an influence on credit risk assessment. Credit risk is managed through credit approvals, establishing credit limits and continuously monitoring the creditworthiness of customers to which the Group grants credit terms in the normal course of business

## Credit risk for receivables pertaining to residential business

The risk for trade receivables pertaining to residential business is considered nil as the possession of the residential property is transferred only after the receipt of payment from the customer

Summary of the Group's exposure to credit risk by age of the outstanding from various customers is as follows:

	31 March 2019	31 March 2018
Neither past due nor impaired	1,571	1,341
Past due but not impaired		
Past due 1-90 days	800	531
Past due 91-180 days	293	1,340
Past due 181-270 days	155	1,459
Past due 271-365 days	-	805
Past due more than 365 days	26	312
	2,846	5,787





## Notes to the Consolidated financial statements

for the year ended 31 March 2019

(Currency: Indian rupees in lakhs)

## Note 53 (continued)

(ii) Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Group's exposure to market risk for changes in interest rates relates borrowings from financial institutions.

## Exposure to interest rate risk

The interest rate profile of the Group's interest-bearing financial instruments is as follows:

Particulars	31 March 2019	31 March 2018
Fixed-rate instruments Financial assets	4,229	3,411
Financial liabilities	280,277	369,310
Variable-rate instruments Financial liabílitics	404,304	354,178

## Interest rate sensitivity - fixed rate instruments

The Group does not account for any fixed-rate financial assets or financial habilities at fair value through profit or loss, and the Group does not have any designate financial habilities.

## Interest rate sensitivity - variable rate instruments

A reasonably possible change of 100 basis points in interest rates at the reporting date would have increased / decreased equity and profit or loss by amounts shown below. This analyses assumes that all other variables, in particular, foreign currency exchange rates, remain constant. This calculation also assumes that the change occurs at the balance sheet date and has been calculated based on risk exposures outstanding as at that date. The period end balances are not necessarily representative of the average debt outstanding during the period.

	100 bp increase	100 bp decrease
As at '31 March 2019		
Variable-rate instruments	4,043	(4,043)
Cash flow sensitivity (net)	4,043	(4,043)
	100 bp increase	100 bp decrease
As at '31 March 2018		

3,542

3,542

(3,542)

(3,542)

(Note: The impact is indicated on the profit/loss and equity before tax basis)



Variable-rate instruments

Cash flow sensitivity (net)



Notes to the Consolidated financial statements for the year ended 31 March 2019 (Currency, Indian rupees in lakhs)

Note 54 Capital Management

The Group's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business it sets the amount of capital required on the basis of annual business and long-term operating plans which include capital and other strategic investments.

The funding requirement of the Group are met from fund and non-fund based working capital lines from various banks. Furthermore, the Group has access to funds from compulsory convertible and non-convertible debentures from holding Group and other financial institution. The Group's policy is to use short-term and long-term borrowings to meet anticipated funding requirements.

The Group monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total liabilities, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity other than amounts accumulated in the hedging reserve.

	As at 3) March 2019	As at 31 March	
Total borrowings	773,320	648,693	
Less . Cash and cash equivalent	5,921	7,141	
Adjusted net debt	767,400	641,552	
Adjusted equity	29,512	54,682	
Adjusted net debt to adjusted equity ratio	26	12	





Tata Realty and infrastructure Limited Notes to the consolidated financial statements (Continued) for the year anded 31 March 2019 (Currency Indian supers in lakhs)

Note 55 Tax expense (a) Amounts recognised in profit and loss

31 March 2019 31 March 2018

Current income tax	864	799
Current Tax Expense		262
Original and reversal of temperory diffeence		
linciease in tax rate	1,906	(1.007)
Recognition of previously unrecognized tax losses		-
Change in recognised deductible temporary differences		
Total deferred tax expense/(benefit)	1,906	(1,007)
Tax expense for the year	2,769	54

(h) Amounts recognised in other comprehensive income

	For the year Before tax INR	l ax (expense) benefit iNK	Net of tax	For the year Before tax INR	Tax (expense) benefit INR	Net of lax
ttems that will not be reclassified to profit or loss						
Changes in revaluation surplus				-		
Remeasurements of the defined benefit plans	55	(14)	41	35	(9)	26
Equity Instruments through Other Comprehensive Income						
items that will be reclassified to profit or loss			-			-
Share of Other Comprehensive Income in Associates and Joint						
Ventures, to the extent to be classified into profit or loss						
	55	(14)	41	35	(9)	26

## (c) Reconciliation of effective tax rate

	33 March 2019	31 March 2018
Profit before tax	(21,145)	(40,236)
Tax using the Company's domestic tax rate	(4,787)	(5,419)
(Current year 34 944% or 26%(Previous year 30 90% or 25 75%))		
Tax effect of:		
Differences in tax rates		-
frems not taxable/ considered seperately	99	455
Impact on account of Ind As adjustments	(688)	(1,647)
Current-year losses for which no deferred tax asset is recognised	7,363	7,404
Effect of deduction claimed under chapter IV/ VIA of income tax act. 1961	(1.041)	(1.7)2)
Unutilised MAT Credit	855	779
Others	968	
Total	2,769	54





## Notes to the consolidated financial statements (Continued)

for the year ended 31 March 2019 (Currency, Indian rupees in lakhs)

## Note 56

Disclosure pursuant to Ind AS 19 on "Employee Benefits"

(i) The Group has adopted Ind AS 19 on "Employee Benefits" as prescribed by the Companies (Accounting Standards) Rules, 2006 issued by the Central Government, which is deemed to be applicable under the provisions of Section 133 of the Companies Act, 2013

	31 March 2019	31 March 2018
Contribution to provident fund recognised as an expense under "Employee benefits".	136	120

## (iii) Defined Benefit Plans

## Gratuity

The Group has a defined benefit gratuity plan. Every employee who has completed five years or more of service gets a gratuity on death or resignation or retirement calculated as per the Payment of Gratuity Act, 1972 with no ceiling.

	31 March 2019	31 March 2018
I Change in the defined benefit obligation		
Liability at the beginning of the year	454	393
Liability on acquistion of subsidiaries during the year		-
Interest Cost	35	29
Current Service Cost	86	87
Past Service Cost (Vested Benefit)		-
Benefit Paid	(94)	(1)
Actuarial Loss on obligations	(60)	(34)
Liability acquired on acquisition / (settled on Divestiture)	-	-
Liability at the end of the year	421	454
11 Amount Recognised in the Balance Sheet		
Liability at the end of the year	425	454
Fair Value of Plan Assets at the end of the year		-1
Difference	421	451
Amount recognised in the Balance Sheet	421	451
III Expenses Recognised in the statement of profit and loss		
Current Service Cost	86	87
Interest Cost	35	29
Expected Return on Plan Assets		(21
Net Actuarial Loss to Be Recognised	(60)	-
Addition on acquisition	-	-
Expense Recognised in statement of profit and loss	61	95
IV Balance Sheet Reconciliation		
Opening net hability	454	393
Liability on acquisition of subsidiary	-	-
Expense as above	61	82
Employers contribution (paid)	(94)	(21
Pre acquisition liability of subsidiary	-	-
Amount recognised in Balance Sheet	421	454





Notes to the consolidated financial statements (Continued)

for the year ended 31 March 2019 (Currency Indian rupees in takhs)

## (iv) Defined Benefit Plans (Continued)

Gratuity (Continued)	31 March 2019	31 March 2018
V Actuarial Assumptions :		
Discount Rate	7.05%	7.85%
Salary escalation	7.00%	7.00%

Attrition Rate: Directors - Nil, Age 21-30 years - 5%, Age 31-40 years - 3%, Age 41-59 years -2%

Estimates of future salary increases, considered in actuarial valuation, take account of inflation, seniority, promotion and other relevant factors, such as supply and demand in the employment market

The Group's liability on account of gratuity is not funded and hence, the disclosures relating to the planned assets are not applicable.

		2019	2018
vı	Experience Adjustments:		
	Defined benefit obligation	368	408
	Plan assets	-	-
	(Deficit)	(368)	(408)
	Experience adjustment on plan liabilities		-
	Experience adjustment on plan assets	-	-

## Sensitivity analysis

Reasonably possible changes at the reporting date to one of the relevant actuarial assumptions, holding other assumptions constant, would have affected the defined benefit obligation by the amounts shown below.

	31 March 2019		31 March	31 March 2018	
	Increase	Decrease	Increase	Decrease	
Discount rate ( % movement)	(2.60)	2.70	0.45	-	
Salary escalation (% movement)	2.70	(2.60)	-	1.00	

## (v) Other long term employment benefits

## Compensated absences

The liability towards compensated absences for the year ended 31 March 2019 recognised in the Balance Sheet based on actuarial valuation using the projected unit credit method amounted to INR 795 /- (2018 INR 542/-) and the charge to the Statement of profit and loss amounted to INR 126/- (2018; INR 25/-)





Notes to the consolidated financial statements (Continued) for the year ended 31 March 2019 (Currency Indian rupees in takhs)

Note 57 Related Party Disclosures

Holding company: Tata Sons Private Limited

Joint Venture TRIL IT4 Private Limited MIA Infrastructure Private Limited Mikado Realtors Private Limited Industrial Mineral and Chemicals Company Limited

Joint Venture of a Subsidiary Pune Solapur Expressways Private Limited (a JV of TRJL Roads Private Limited) A & T Road Construction Manajement and Operation Private Limited Pune IT City Metro Rail Limited (wef March 13, 2019)

Other related parties with whom transactions have taken place during the year:

## Fellow Subsidiaries:

Ewart Investments Limited Fata Consultancy Services Limited Tata AIG General Insurance Company Limited Tata AIA Life Insurance Company Limited TC Travel and Services Limited Tata Capital Financial Services Limited Infinity Retail Ltd Tata Asset Management Limited Tata Housing Development Company Limited One Colombo Project (Private) Limited Tata Teleservices Limited Tata Teleservices Limited Tata Consulting Engineers Ltd

## Key Management Personnel

Sanjay Dutt Sanjay Ubale Sanjay Sharma Arvind Chokhany Vinay Gaokar Managing Director & CEO - w c.f. 1st April 2018 Managing Director - upto 31st March 2018 Chief Financial Officer - w.e.f. 10th September 2018 Chief Financial Officer - upto 28th February 2019 Company Secretary





Tata Realty and Infrastructure Limited Notes to the consolidated financial statements (Continued) for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

## Related party disclosures (Continued)

## Summary of related party transactions

Transactions	Holding company	Fc)low subsidiaries	Joint Venture	Key Management Personnel	Total
Unsecured Ioan taken					
31 March 2019		14.000	•	•	14,00
31 March 2018	-		-		-
Tata Housing Development Company Limited					
31 March 2019	-	14,000	•		14,0
31 March 2018	-	•	-	-	
Interest expenses on unsecured loans					
31 March 2019		21			
31 March 2018	-		•	•	
Tata Housing Development Company Lunited					
31 March 2019	•	21	•	•	
31 March 2018	•	•		• -	
Purchase of fixed assets					
31 March 2019	-				
31 March 2018		7	-		
Infinity Retail Limited					
31 March 2019	-	•			
31 March 2018	•	7			
Purchase of Investments					
31 March 2019	-	-	105		1
31 March 2018		<u> </u>	1,744	<u> </u>	ι.
Industrial Minerals and Chemicals Private Limited					
31 March 2019	-	•	105	-	
31 March 2018		-	1,744		1,1
Asset Management Consultancy fees					
31 March 2019	-		148		
31 March 2018	-	-	143		
TRIL IT4 Private Lumited					
31 March 2019		-	148	•	
31 March 2018		-	143	•	
Project Management Consultancy fees					
31 March 2019	-		1,118		1,
31 March 2018		251	1,314		1,
Mikado Realtors Private Limited					
31 March 2019	-		1,118	•	١,
31 March 2018	-		1,314		1,
Tata Consultancy Services Limited					
31 March 2019	-		-		
31 March 2018	•	251	-	-	
Interest locome			•		
31 March 2019	•	•	1,465	-	1.
31 March 2018	-		1,465	-	١,
TRIL IT4 Private Limited			1.440		
31 March 2019 31 March 2018	-	-	1,465 1,465		L, 1.
ST COLUMN TO	-				
Reut Income					2
3) March 2019	-	5,315	-	•	<u>5</u> ,
31 March 2018 Tata Consultancy Services Limited	-	5,092		*	5.
31 March 2019	-	4,629		-	4
31 March 2018	_	4,481			4
Tata AJG General Insurance Co Ltd.					
31 March 2019	-	686			
31 March 2018	-	611			
Managerial comparation					
Managerial remuneration	_			1,008	1
31 March 2019		-		1,000	



Tata Realty and Infrastructure Limited Notes to the consolidated financial statements (Continued) for the year ended 31 March 2019

(Currency: Indian rupees in lakhs)

## Related party disclosures (Continued)

## Summary of related party transactions

Transactions	Holding company	Fellow subsidiaries	Joint Veoture	Key Managemens Personnel	Total
31 March 2018	-		-	755	
Reimbursement of exponses (receivable)					
31 March 2019	8	605			6
31 March 2018	\$\$	-	-	-	
Others					
31 March 2019	8	605	-		
31 March 2018	55		-		
Services Received					
31 March 2019	5	712			:
31 March 2018	21	503	-	-	
Tata AIG General Insurance Limited					
31 March 2019	-	275	-	•	
31 March 2018	-	249	-	-	
Tata AIA Life Insurance Limited					
31 March 2019		8	-	-	
31 March 2018	-	9			
Tata Sons Private Lanuted					
31 March 2019	5	-			
31 March 2018	21	-	•	-	
Ewart Investment Limited					
31 March 2019	-	208	•	•	
31 March 2018	-	238	-	•	
Tata Consulting Engineers Ltd					
31 March 2019	-	158		-	
31 March 2018	-	•	-	-	
Tata Elxsi Ltd.					
31 March 2019	-	8	•		
31 March 2018		-	-	-	
Tata Teleservices Ltd	-				
31 March 2019	-	20	-	•	
31 March 2018	-	8	-	-	
Tata Housing Development Company Limited	-			1	
31 March 2019	-	35			
3) March 2018	•		-	-	
Travelling					
31 March 2019	•		-	-	
31 March 2018 TC Travel and Services Limited	•	166			
31 March 2019	-	_			
31 March 2019		-	-	-	
Sala OF Final Anata					
Sale Of Fixed Assets 31 March 2019	20				
31 March 2019	28	2	•	-	
Tata Sons Private Limited		-		·	
31 March 2019	28	-	_	-	
31 March 2018	2 <b>0</b> -	-	-	-	
Ewart Investments Limited					
Ewart Investments Limited 31 March 2019		2			
31 March 2019	-	2	-		
Deposit Refund Received 31 March 2019	150	450		_	
31 March 2018	130		•	-	





Tata Realty and Infrastructure Limited Notes to the consolidated financial statements (Continued) for the year ended 31 March 2019 (Currency, Indian rupees in lakhs)

## Related party disclosures (Continued)

## Summary of related party transactions

Traosactions	Holding company	Fellow subsidiaries	Joint Venture	Key Management Personnel	Total
Tata Sons Private Lunited					
31 March 2019	150	•		-	15
31 March 2018			-		-
Ewart Investments Limited					
31 March 2019		450			44
31 March 2018			-		-
Employee Benefit Transfer					
Tata Sons Private Limited					
31 March 2019					
31 March 2018	56	-	•	-	2
Balances receivable at the year-end towards interest and debit notes and advances					
31 March 2019	_	688	79		-
31 March 2018	•			•	76
Pune Solapur	•		865	•	86
31 March 2019	_		10		
31 March 2018	•	•	10	-	
51 Watch 2018	•	•	10		
TRUL IT4 Private Limited					
31 March 2019	•		69	-	6
31 March 2018	-	-	66		6
Mikado Realtors Private Lunited					
31 March 2019			273		27
31 March 2018	-	•	789		78
Tata Consunting Engineers Ltd					
31 March 2019		35			3
31 March 2018					-
Tata Housing Development Company Limited		-	-		•
31 March 2019		653			65
31 March 2018		•	-		-
Deposits placed					
31 March 2019		240			
31 March 2019	150	300	-	•	24
Ewart Investments Limited	130	500	·	<u> </u>	4:
31 March 2019		240			24
31 March 2018		300	-		30
31 March 2018	-	300	-		30
Tata Sons Pvt Limited					
31 March 2019 31 March 2018	150	-	-		-
Trade receivables					
31 March 2019	•	793	15		80
31 March 2018	6	743	15		70
Tata Consultancy Services Limited					-
31 March 2019	•	752	•	•	7:
31 March 2018	-	712	-	-	7
TRIL IT4 Private Limited					
31 March 2019	-	-	15	-	




Tata Realty and Infrastructure Limited Notes to the consolidated financial statements (Continued) for the year ended 31 March 2019 (Currency: Indian rupces in lakhs)

#### Related party disclosures (Continued)

#### Summary of related party transactions

Transactions	Holding	Fellow subsidíaries	Joint Venture	Key Management Personnel	Total
Tata Sons Private Limited	company	300310121103		i ersönner	
31 March 2019					
31 March 2019	- 6	•		•	
31 March 2018	0	-	•	•	
Tata AIG General Insurance Company Ltd					
31 March 2019		41	-		
31 March 2018	-	31	-	-	
Balances payable at the year-end					
towards advances from customers					
31 March 2019		2,922	-	-	2,5
31 March 2018	-	2,876			2,4
Tata Consultancy Services Limited					
31 March 2019		2,583		-	2,
3) March 2018	-	2,583			2,
Tata AIG General Insurance Company Ltd.					
31 March 2019	-	339		-	
31 March 2018		293	-	-	
towards expenses					
31 March 2019		25	-	•	
31 March 2018	-	7			_
Tata Consultancy Services Limited					
31 March 2019	-	7			
31 March 2018		7			
Tata Housing Development Company Limited					
31 March 2019		19			
31 March 2018		-			





Notes to the consolidated financial statements for the year ended 31 March 2019 (Currency, Indian rupees in lakhs)

#### Note 58

Additional information pursuant to paragraph 2 of Division II of Schedule III to the Companies Act 2013- 'General instructions for the preparation of consolidated financial statements' of Division II of Schedule III

# Part A March 2019

March 2019	Net A	usets	Share in prof	it or lost	Share in other comp	rehensive	Share in total compri	ehensive Income
	(Total Assets - i	otal Liabilities)			income			
Name of the entity	As a % of consolidated net assets	Απουπτ	As a % of consolidated profit or loss	Amount	As a % of consolidated other comprehensive	Amount	As 8 % of total comprehensive income	Amount
Parent					iorome		<u> </u>	
Tata Realty and infrastructure Limited	54.67%	198,686	62.72%	[18,627]	68.53%	28,871	60.97%	10,244
Subsidiairies Indian								
Acme Living Solutions Private Limited	C%	(8)	0%	(1)	0%		-0.01%	(1)
Arrow Infra Estates Private Limited	0.22%	806	0.02%	(7)		-	-0.05%	(7)
Gurgaon Constructivell Private Limited	0.76%	2,760	D.53%	(158)		-	-1.75%	(158)
Gurgaon Realtech Limited	0.40%	1,437	0.32%	(96)		-	-0 76%	(96)
TRIL Roads Private Limited	7 69%	27,942	-2.27%	675	25.35%	10,680	89.75%	11,355
TRIF Gungson Housing Projects Private Limited	0.00%	(3)	0.00%	(1)	0%	-	-0.01%	0
TRIL Urban Transport Private Limited	1 37%	4,970	3.19%	(948)		270	-5.36%	(678)
Wellkept Facility Management Services Private Limited	0.00%	(1)	0.00%	(1)		-	-D.01%	(1
TRIL Constructions Limited	4.25%	15,434	1 72%	(512)			-4.05%	(512)
TRIL Infoperk Limited	10.43%	37,911	4,74%	1,408)	0%	(2)	-15 14%	(1,410)
TRIL Amrilsar Projects Limited	0 7 9%	2,889	5 12%	(1,520)		1		(1,519)
Hampi Expressway Private Limited	5.37%	19,524	-1.54%	457	0%		3.61%	457
TRIF Real Estate Development Limitea	1.49%	5,433	0.50%	(147)	0%	1	-1.15%	(146
HV Farms Private Limited	0.32%	1,148	0.00%	0			-0.01%	(1
International Infrabuild Private Limited	-0.53%	(1,925)	5.86%	1,741		-	-13 76%	(1,741
Dharamshala Ropeway Umited	0 14%	504	2.21%	(655)		-	-5.18%	[655
Manali Ropeway Private Limited	0.06%	206	0.24%	(71)		-	-0.56%	(71
Uchit Expressways Private Limited	3.11%	11,286	-10.03%	2,978	0%	(D)		2,978
TRPL roadways	3.66%	13,313	4,31%	(1,280)		2,305		1,025
Durg Shivnath Expressways Pvt Ltd	2.62%	9,507	1.99%	590	0%	(0		590
Matheran Ropeway Pvt Ltd	0 57%	2,436	-0.11%	33	0%	-	0.26%	33
MIA Infrastructure Private Limited	0%	(31)	0.09%	(26)	0%	-	-0.20%	152
Non-controlling interests in all subsidiaries	1.82%	6,517	0 75%	(222)	0%	-	0%	-
Joint Venture (investment as per equity method) Indian								
Pune Splagur Expressways Private Limited	1.20%	4,344	4.03%	[1,198]	D%	2	-9.45%	(1,196
A & T Road Construction Management and Operation Private Limited	0.00%	(2)	0%	(0)	0%	-	0.00%	(0
TRIL IT4 Private Limited	-6.25%	[22,722]	19.41%	[5,763]			-45 55%	(5,763
Mikado Realtors Pvt Ltd	1 23%	4,452	0.11%	(32)		-	-0.25%	[32
Industrial Minerals and Chemicals Company Private Limited	4.54%	15,493	0.05%	[15]		-	-0 12%	[15
Total	100%	363,406	100%	(29,699)	100%	42, <u>129</u>	100%	12,652
Consolidation Adjustment		(333,894)		5,006		(42,087	<b>b</b>	(36,304
Total Net Assets		29,512		(23,693)		41		(23,652





# 1 atta Keatty and Intrastructure Limited Notes to the consolidated fin ancial statements (Contraned) for the year coded 31 March 2019 (Currency Indum tupes in lakits)

B Pard

Name of the entity	Reparting Currency	Evchange Rats	Share Capitat	Reserve and Surplus	Tutal Assets	Totat Labilities	Investment	Turnuver / Toral Revenue	Tatal Income	Frufti Befure Provision for Proposed Tax Taxation Divergend	Pruvisiun for Taxatlori	Profit After Taxation	Proposed Divadend	rofit After Prupowed % of Taxation Divodend Shareholthag
Рассти Тара Ясану, али Indiasencture Emined	NR.	00 1	101 231	96.955	526 847	(13X 161)	406,347	£6†'†I	247.02	(17,328)	(32)	(18.627)		
Subsidiatries										_				
Indian	914	001			2					117		40		200 001
Active Living Solutions Filvare cumues	a a	100		Ser.	101 A04	(865.7.1)				6		6		100 00%
Cuttanon Constructivell Private Lanued	INR	1 60	7	2,754	11 566	(3,846)			¥	(158)	1321	(336)		100.00%
Gurizaon Realtech Limited	INR	1 80	7	1.450	25.009	(24,232)			•	(96)	'	(96)		100 00%
TRIL Rouds Private Limited	INR	1 80	775	27 168	K2 723	(184,781)	74,682	3.737	125.4	675	202	£2]		100 00%
TRIF Gurgaon Housing Projects Private Lamaed	INR	1 40	5	[8]	e	(3)		1		(1)	•	(1)		100 00%
TRU Urban Transport Private 1 anated	INR	1.00	2,684	2,286	N.068	(SHA) ET	1.571	63	3	(NPGJ	r <sup>(1)</sup>	(178)	•	t 00 00%
Wellkept Facility Management Services Private Limited	INR	00 1	14	(11)	0	69				(1)	1	(1)		160 00%
TRIL Constantions Limited	RNR	00.1	3 615	11.820	628712	(8.905)		•		(512)	•	(512)		67 50%
TRIL Infopark Limited	RNR	00/1	105 001	(67 089)	306.518	(328.607)	12	52.171	52 600	11,408)	'	(301.1)	•	%00.001
TR11 Anuritsar Projects Lamited	NNR	1:00	32.431	129,5421	35 644	(32,755)	ŗ	2.48%	2.541	(1.520)	'	(1.520)	×.	%00 001
Hainpi Expression Private Limited	BVR	1.00	5,313	14 211	132,746	(113.272)	•	40, 269	10 272	151	399	23	¢	100.00%
TRIF Real Estate Development Loruted	INR	100	XXE	5,045	30,698	(23,266)	•			(147)		(147)	•	100 00%
HV Farms Private Lumited	ING	1 00		1.147	1,156	121	9		,	()	,	(1)	¢	100 00%
International Infrabuild Pirvate Lumod	NNR.	1 40	2	(510.1)	14 250	(16 1 35)	•	5	901	(11,241)	121	(216-1)	è	26 00%
Dharamshala Ropeway Lumited	NR.	1.00	(IDE	PT.	37276	(6. 772)		•	2	(635)	(9)	(614))		74 00%
Manuel Ropoway Private Limited	NR.	100	30	170	5149	(576 71	1	1		(12)	(91)	(\$3)		72 00%
Untrue Expressions Private Lutrated	INR	1.00	112012	4 216	102.028	1217.061	9	55,944	56,035	14Z4 Z	185,1	1.597		100 00%
TRPL loadways	<b>ENR</b>	1.00	5	405,61	680'21	(13,775)	1	,	,	101927 13		(1.280)	4	100 00%
Durg Shivnath Expressions Pvz Ltd	INR	1.00	2.362	7 145	24.725	120.216}	1.072	5,24 <b>7</b>	5 781	045	122	468		100.00%
Matheran Ropeway Pvi Lid	INK	1 60	750	1 686	2.537	(161)	•		8. 2	56	•	2	3	70.00%
NITA Infrastructure Private Linsted	INR	00.1	222	(305)	¢	40	ï		'	(26)		(26)	k	100 00%
Total			262,879	2HC,14	1,452,494	(1,098,270)	486,694	174,536	189,303	(21,169)	3,962	(151,22)		

# Associates and Joint Ventures

Name of the Associates / Junit Ventures	tratest audited Balance Shiret date	rocanares of the Asswellate / Joint Ventures beld by the Company	Account of Investment in Associates / Julin Venture	Laken u Hotáng (%)	description of back Incred to Applicant influence	Associate / Joint Vebture is not cursuitated	reconnects attributable to Shareholding as per latest audited Balance Sheel		
								Considered in Consolidation	Not Considerci In Cutsolidation
<del>նոնդ Ventures</del> Ստու Նանգրա (-Նլեսեռնեւթ. Վ.Ի. ։ ։։ է՝ նշունքն	6 02 60 18	2,386,711	239	3	There is sentitizent affuence due to sharcheideng and The entry to pour central over the resonants activenes inducation of the way drough turbadany company.	The entity is paracilitated by way of equity accounting	217	(\$55)	
તે & 1 K.હન է ખાતવાહરાલા મિત્રમહુલ્લાસા હાથે (ગુસ્તાહાલા મિલગાલ વિભારન	9102 ED 1E	norioi		8	There is significant influence due to shareholding, and The entity is joint cortrol over the economic activities indirectlyconsolidated by way through adisationery company.	The entroy is consolidated by way of equity accounting	(1)		
العالم معددا سابعا	31 03 2019	740,000	74 00	12	These is significant influence due to shareholding and The entry is joint control over the constraint starvide indirectly personidated by way through subordinary company and an equily seconduce	The entity is consolidated by way of equity accounting	((16,815)		(4,265)
tado Realson Vervard I misel	31 13 2019	1019,782,4001	666'1	74	These is significant influence due to shareholding and The exity is purit carried over the economic setivities indirectlypercondulated by way through subsidiary company.	The entry is consolidated by way of equity accounting	295,6	(124)	•
serial McOyd and Chameste Company Privise Limited	GIOZ EN IE	3256	~	14	There is superformer influence due to shareholding and the outly to joint control over the occoromic activities indexed/activities through ubsuitary company	The entry is consolidated by way of equity accounting	12,205	(1)	



### Notes to the consolidated Ind AS financial statements (Continued)

for the year ended 31 March 2019 (Currency: Indian rupees in lakhs)

#### Note 59

#### Events after the balance sheet date

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

#### Note 60

#### **Previous Year's Figures**

Previous year figures have been regrouped / reclassified to conform to current year presentation, wherever considered necessary,

SKINS 4 Chartered Accountants è 0

m Banmali Agravala Chairman QIN - 00120029

S. Santh Director DIN - 00032049

DIN - 05251670 Neera Saggi Director DIN - 00501029

Sanjay Dutt

Managing Director

upeda F. N. Subedar Director

DIN - 00028428 Qua anjay Sharma

nancial Officer

Chief

1 Vinas Cankar Company Secretary Membership No: A6120

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

Ba

Mumbai May 8, 2019

d Rajiv Sabbarwal

DIN - 00057333

Salabasures

Director

Certified True Copy Tata Relaty and Infrastructure Limited

**Company Secretary** 

Chartered Accountants Indiabulls Finance Centre Tower 3, 27<sup>th</sup> – 32<sup>nd</sup> Floor Senapati Bapat Marg Elphinstone Road (West) Mumbai – 400 013 Maharashtra, India Tel: +91 22 6185 4000 Fax: +91 22 6185 4001

#### INDEPENDENT AUDITOR'S REPORT

To The Members of TATA Realty and Infrastructure Limited

#### Report on the Audit of the Consolidated Ind AS Financial Statements

#### Opinion

We have audited the accompanying consolidated Ind AS financial statements of TATA Realty and Infrastructure Limited ("the Parent") and its subsidiaries, (the Parent and its subsidiaries together referred to as "the Group") and the Group's share of loss in its joint ventures, which comprise the Consolidated Balance Sheet as at 31<sup>st</sup> March 2020, and the Consolidated Statement of Profit and Loss (including Other Comprehensive Income), the Consolidated Statement of Cash Flows and the Consolidated Statement of Changes in Equity for the year then ended, and a summary of significant accounting policies and other explanatory information.

In our opinion and to the best of our information and according to the explanations given to us, and based on the consideration of reports of the other auditors on separate financial statements of the subsidiaries and joint ventures referred to in the Other Matters section below, the aforesaid consolidated Ind AS financial statements give the information required by the Companies Act, 2013 ("the Act") in the manner so required and give a true and fair view in conformity with the Indian Accounting Standards prescribed under section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, as amended ('Ind AS'), and other accounting principles generally accepted in India, of the consolidated total comprehensive loss, their consolidated cash flows and their consolidated changes in equity for the year ended on that date.

#### **Basis for Opinion**

We conducted our audit of the consolidated Ind AS financial statements in accordance with the Standards on Auditing specified under section 143 (10) of the Act (SAs). Our responsibilities under those Standards are further described in the Auditor's Responsibility for the Audit of the Consolidated Ind AS financial statements section of our report. We are independent of the Group and its joint ventures in accordance with the Code of Ethics issued by the Institute of Chartered Accountants of India (ICAI) together with the ethical requirements that are relevant to our audit of the consolidated Ind AS financial statements under the provisions of the Act and the Rules made thereunder, and we have fulfilled our other ethical responsibilities in accordance with these requirements and the ICAI's Code of Ethics. We believe that the audit evidence obtained by us and the audit evidence obtained by the other auditors in terms of their reports referred to in the sub-paragraph (a) of the Other Matters section below, is sufficient and appropriate to provide a basis for our audit opinion on the consolidated Ind AS financial statements.

RKU

Regd. Office: Indiabulls Finance Centre, Tower 3, 27<sup>th</sup>-32<sup>nd</sup> Floor, Senapati Bapat Marg, Elphinstone Road (West), Mumbai – 400013, Maharashtra, India. (LLP Identification No. AAB-8737)

### Information Other than the Financial Statements and Auditor's Report Thereon

- The Parent's Board of Directors is responsible for the other information. The other information comprises the information included in the Board of Directors report, but does not include the consolidated Ind AS financial statements and our auditor's report thereon.
- Our opinion on the consolidated Ind AS financial statements does not cover the other information and we do not express any form of assurance conclusion thereon.
- In connection with our audit of the consolidated Ind AS financial statements, our responsibility is to
  read the other information, compare with the financial statements of the subsidiaries and joint
  ventures audited by the other auditors, to the extent it relates to these entities and, in doing so,
  place reliance on the work of the other auditors and consider whether the other information is
  materially inconsistent with the consolidated Ind AS financial statements or our knowledge obtained
  during the course of our audit or otherwise appears to be materially misstated. Other information
  so far as it relates to the subsidiaries and joint ventures is traced from their financial statements
  audited by other auditors.
- If, based on the work we have performed, we conclude that there is a material misstatement of this other information, we are required to report that fact. We have nothing to report in this regard.

### Management's Responsibility for the Consolidated Ind AS financial statements

The Parent's Board of Directors is responsible for the matters stated in section 134(5) of the Act with respect to the preparation of these consolidated financial statements that give a true and fair view of the consolidated financial position, consolidated financial performance including other comprehensive income, consolidated cash flows and consolidated changes in equity of the Group including its joint ventures in accordance with the Ind AS and other accounting principles generally accepted in India. The respective Board of Directors of the companies included in the Group and of its joint ventures are responsible for maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding the assets of the Group and its joint ventures and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial controls, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the financial statements that give a true and fair view and are free from material misstatement, whether due to fraud or error, which have been used for the purpose of preparation of the consolidated Ind AS financial statements by the Directors of the Parent Company, as aforesaid.

In preparing the consolidated financial statements, the respective Board of Directors of the companies included in the Group and joint ventures are responsible for assessing the ability of the respective entities to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless the respective Board of Directors either intends to liquidate their respective entities or to cease operations, or has no realistic alternative but to do so. The respective Board of Directors of the companies included in the Group and of its joint ventures are also responsible for overseeing the financial reporting process of the Group and of its joint ventures.

#### Auditor's Responsibility for the Audit of the Consolidated Ind AS financial statements

Our objectives are to obtain reasonable assurance about whether the consolidated Ind AS financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance,

R RKO

but is not a guarantee that an audit conducted in accordance with SAs will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these consolidated Ind AS financial statements.

As part of an audit in accordance with SAs, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

- Identify and assess the risks of material misstatement of the consolidated Ind AS financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.
- Obtain an understanding of internal financial control relevant to the audit in order to design audit procedures that are appropriate in the circumstances. Under section 143(3)(i) of the Act, we are also responsible for expressing our opinion on whether the Parent has adequate internal financial controls system in place and the operating effectiveness of such controls.
- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to events or conditions that may cast significant doubt on the ability of the Group and its joint ventures to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the consolidated Ind AS financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Group and its joint ventures to cease to continue as a going concern.
- Evaluate the overall presentation, structure and content of the consolidated Ind AS financial statements, including the disclosures, and whether the consolidated Ind AS financial statements represent the underlying transactions and events in a manner that achieves fair presentation.
- Obtain sufficient appropriate audit evidence regarding the financial information of the entities or business activities within the Group and its joint ventures to express an opinion on the consolidated Ind AS financial statements. We are responsible for the direction, supervision and performance of the audit of the financial statements of such entities or business activities included in the consolidated Ind AS financial statements of which we are the independent auditors. For the other entities or business activities included in the consolidated Ind AS financial statements, which have been audited by the other auditors, such other auditors remain responsible for the direction, supervision and performance of the audits carried out by them. We remain solely responsible for our audit opinion.

Materiality is the magnitude of misstatements in the consolidated Ind AS financial statements that, individually or in aggregate, makes it probable that the economic decisions of a reasonably knowledgeable user of the consolidated Ind AS financial statements may be influenced. We consider quantitative materiality and qualitative factors in (i) planning the scope of our audit work and in evaluating the results of our work; and (ii) to evaluate the effect of any identified misstatements in the consolidated Ind AS financial statements.

& PKH

We communicate with those charged with governance of the Parent and such other entities included in the consolidated Ind AS financial statements of which we are the independent auditors regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

#### **Other Matters**

- (a) We did not audit the financial statements of three subsidiaries, whose financial statements reflect total assets of Rs. 3295,03.88 lakhs as at 31 March, 2020, and total revenue of Rs. 816,11.20 lakhs and net cash inflows amounting to Rs. 23,10.52 lakhs for the year ended on that date, as considered in the consolidated financial statements. The consolidated financial statements also include the Group's share of net loss of Rs. 1,38.59 lakhs for the year ended 31<sup>st</sup> March, 2020, as considered in the consolidated financial statements, in respect of three joint ventures, whose financial statements have not been audited by us. These financial statements have been audited by other auditors whose reports have been furnished to us by the Management and our opinion on the consolidated financial statements, in so far as it relates to the amounts and disclosures included in respect of these subsidiaries and joint ventures and our report in terms of subsection (3) of Section 143 of the Act, in so far as it relates to the aforesaid subsidiaries and joint ventures is based solely on the reports of the other auditors.
- (b) We did not audit the financial statements of two entities who were subsidiaries up to 9 December 2019 whose financial statements reflect total revenues of Rs.25,54.37 lakhs and net cash inflows amounting to Rs. 4,66.39 lakhs, as considered in the consolidated financial statements for the year ended 31 March 2020. The group divested their shareholdings in these entities on 9 December 2019. These financial statements are unaudited and have been furnished to us by the Management and our opinion on the consolidated financial statements, in so far as it relates to the amounts and disclosures included in respect of these subsidiaries, is based solely on such unaudited financial statements. In our opinion and according to the information and explanations given to us by the Management, these financial statements are not material to the Group

Our opinion on the consolidated Ind AS financial statements above and our report on Other Legal and Regulatory Requirements below, is not modified in respect of the above matters with respect to our reliance on the work done and the reports of the other auditors and the financial statements certified by the Management.

#### Report on Other Legal and Regulatory Requirements

- As required by Section 143(3) of the Act, based on our audit and on the consideration of the reports of the other auditors on the separate financial statements of the subsidiaries and joint ventures referred to in the Other Matters section above we report, to the extent applicable that:
  - a) We have sought and obtained all the information and explanations which to the best of our knowledge and belief were necessary for the purposes of our audit of the aforesaid consolidated Ind AS financial statements.

& PKU

- b) In our opinion, proper books of account as required by law relating to preparation of the aforesaid consolidated Ind AS financial statements have been kept so far as it appears from our examination of those books and the reports of the other auditors.
- c) The Consolidated Balance Sheet, the Consolidated Statement of Profit and Loss including Other Comprehensive Income, the Consolidated Statement of Cash Flows and the Consolidated Statement of Changes in Equity dealt with by this Report are in agreement with the relevant books of account maintained for the purpose of preparation of the consolidated Ind AS financial statements.
- d) In our opinion, the aforesaid consolidated financial statements comply with the Ind AS specified under Section 133 of the Act.
- e) On the basis of the written representations received from the directors of the Parent as on 31st March, 2020 taken on record by the Board of Directors of the Company and the reports of the statutory auditors of the subsidiary companies and the joint venture companies incorporated in India, none of the directors of the Group companies, and joint venture companies incorporated in India, is disqualified as on 31 March, 2020 from being appointed as a director in terms of Section 164 (2) of the Act.
- f) With respect to the adequacy of the internal financial controls over financial reporting and the operating effectiveness of such controls, refer to our separate Report in the Annexure which is based on the auditors' reports of the Parent, subsidiary companies and joint venture companies incorporated in India. Our report expresses an unmodified opinion on the adequacy and operating effectiveness of internal financial controls over financial reporting of those companies.
- g) With respect to the other matters to be included in the Auditor's Report in accordance with the requirements of section 197(16) of the Act, as amended, in our opinion and to the best of our information and according to the explanations given to us, the remuneration paid by the Parent to its directors during the year is in accordance with the provisions of section 197 of the Act.
- h) With respect to the other matters to be included in the Auditor's Report in accordance with Rule 11 of the Companies (Audit and Auditors) Rules, 2014, as amended, in our opinion and to the best of our information and according to the explanations given to us:
  - The consolidated Ind AS financial statements disclose the impact of pending litigations on the consolidated financial position of the Group, and joint ventures.
  - ii) Provision has been made in the consolidated Ind AS financial statements, as required under the applicable law or accounting standards, for material foreseeable losses, if any, on long-term contracts including derivative contracts.

R PKb

iii) There were no amounts which were required to be transferred to the Investor Education and Protection Fund by the Parent Company, and its subsidiary companies, and joint venture companies incorporated in India.

For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No.117366W/W-100018)

ekuihas

Rajesh K. Hiranandani Partner Membership No. 32690 UDIN: 20036920AAAACC1788

Mumbai, 06 July 2020

& pro

#### ANNEXURE TO THE INDEPENDENT AUDITOR'S REPORT

# (Referred to in paragraph f under 'Report on Other Legal and Regulatory Requirements' section of our report of even date)

#### Report on the Internal Financial Controls Over Financial Reporting under Clause (i) of Subsection 3 of Section 143 of the Companies Act, 2013 ("the Act")

In conjunction with our audit of the consolidated Ind AS financial statements of TATA Realty and Infrastructure Limited ("the Parent"/ "the Company") as of and for the year ended 31 March, 2020, we have audited the internal financial controls over financial reporting of the Parent and its subsidiary companies and joint ventures, which are companies incorporated in India, as of that date.

#### Management's Responsibility for Internal Financial Controls

The respective Board of Directors of the Parent, its subsidiary companies and joint ventures, which are companies incorporated in India, are responsible for establishing and maintaining internal financial controls based on the internal control over financial reporting criteria established by the respective Companies considering the essential components of internal control stated in the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting issued by the Institute of Chartered Accountants of India (ICAI). These responsibilities include the design, implementation and maintenance of adequate internal financial controls that were operating effectively for ensuring the orderly and efficient conduct of its business, including adherence to the respective company's policies, the safeguarding of its assets, the prevention and detection of frauds and errors, the accuracy and completeness of the accounting records, and the timely preparation of reliable financial information, as required under the Companies Act, 2013.

#### Auditor's Responsibility

Our responsibility is to express an opinion on the internal financial controls over financial reporting of the Parent, its subsidiary companies and its joint ventures, which are companies incorporated in India, based on our audit and based on the consideration of reports of the other auditors on separate Ind AS financial statements of the subsidiaries and joint ventures referred to below in the Other Matters paragraph. We conducted our audit in accordance with the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting (the "Guidance Note") issued by the Institute of Chartered Accountants of India and the Standards on Audit of internal financial controls. Those Standards and the Guidance Note require that we comply with ethical requirements and plan and perform the audit to obtain reasonable assurance about whether adequate internal financial controls over financial reporting was established and maintained and if such controls operated effectively in all material respects.

Our audit involves performing procedures to obtain audit evidence about the adequacy of the internal financial controls system over financial reporting and their operating effectiveness. Our audit of internal financial controls over financial reporting included obtaining an understanding of internal financial controls over financial reporting, assessing the risk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. The procedures selected depend on the auditor's judgement, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error.

We believe that the audit evidence we have obtained and the audit evidence obtained by the other auditors of the subsidiary companies and joint ventures, which are companies incorporated in India, in terms of their reports referred to in the Other Matters paragraph below, is sufficient and appropriate to provide a basis for our audit opinion on the internal financial controls system over financial reporting of the Parent, its subsidiary companies and its joint ventures, which are companies incorporated in India.

B PEG

#### Meaning of Internal Financial Controls Over Financial Reporting

A company's internal financial control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal financial control over financial reporting includes those policies and procedures that:

- 1. pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company;
- provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorisations of management and directors of the company; and
- provide reasonable assurance regarding prevention or timely detection of unauthorised acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

4

#### Inherent Limitations of Internal Financial Controls Over Financial Reporting

Because of the inherent limitations of internal financial controls over financial reporting, including the possibility of collusion or improper management override of controls, material misstatements due to error or fraud may occur and not be detected. Also, projections of any evaluation of the internal financial controls over financial reporting to future periods are subject to the risk that the internal financial control over financial reporting may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

#### Opinion

In our opinion to the best of our information and according to the explanations given to us and based on the consideration of the reports of the other auditors referred to in the Other Matters paragraph below, the Parent, its subsidiary companies and joint ventures, which are companies incorporated in India, have, in all material respects, an adequate internal financial controls system over financial reporting and such internal financial controls over financial reporting were operating effectively as at 31 March, 2020, based on the criteria for internal financial control over financial reporting established by the respective companies considering the essential components of internal control stated in the Guidance Note on Audit of Internal Financial Controls Over Financial Reporting issued by the Institute of Chartered Accountants of India

PRU

372

#### **Other Matters**

Our aforesaid report under Section 143(3)(i) of the Act on the adequacy and operating effectiveness of the internal financial controls over financial reporting insofar as it relates to three subsidiary companies and three joint ventures, which are companies incorporated in India, is based solely on the corresponding reports of the auditors of such companies incorporated in India.

Our opinion is not modified in respect of the above matters.

For DELOITTE HASKINS & SELLS LLP Chartered Accountants (Firm's Registration No.117366W/W-100018)

R

Rajesh K. Hiranandani Partner Membership No. 36920 UDIN: 20036920AAAACC1788

Mumbai, 06 July 2020

& pro

Tata Realty and Infrastructure Limited Consolidated Balance Sheet as at 31 March 2020 (Currency: Indian rupees in lakhs)

(Currenc	y, mutan rupees in takits)			
I ASS	SETS	Note	31 March 2020	31 March 201
	-current assets			
(a)		3(a)	3,682	3,900
(b)		5	118	
(c)		8	90	12
(d)				
(e)		4 (a)	2,41,824	3,62,000
		4 (b)	1,24,833	
(f)		6	80,673	1,41,468
(g		4 (c)	19,053	21,216
(h)	Intangible assets under development under Service Concession Arrangements	.7	1,48,518	2,24,165
(i)		8	1,77,244	45,354
(j)				
	(i) Investment accounted using Equity Method	9	78,687	73,13
	(ii) Other Investments	10	212	2
	(iii) Other Financial assets	11	7,204	1,67
(k)	a contract and and a contract of the second s	12	9,592	6,450
(1)	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	13	12,359 -	10,889
(m)	Other non-current assets	14	21,311	26,275
Non	-current assets		9,25,400	9,16,751
	rent assets			
(a)	Inventories	15	33,587	43,718
(b)	Financial assets			
	(i) Trade and other receivables	16	2,865	2,846
	(ii) Cash and cash equivalents	17	29,940	5,921
	(iii) Bank balances other than (ii) above	18	219	1,579
	(iv) Other Investments	19	71,180	1,38
	(v) Short-term loans and advances	20	1,389	
	(vi) Other Financial assets			3,054
1.5		21	5,503	1,981
(c)	Other current assets	22	10,161	5,388
Curi	rent assets	_	1,54,844	65,870
	TOTAL ASSETS		10,80,244	9,82,621
	ITY AND LIABILITIES			
1 Equi	ty Equity share capital	23	1,01,731	1.01.771
	Other equity	23		1,01,731
(0)(1)	Equity attributable to owners	24	4,910	(78,836
			1,06,641	22,895
(D)(II	) Non-controlling interests	24	6,384	6,617
2 Non-	Current Liabilities		1,13,025	29,512
(a)	Financial liabilities			
	(i) Borrowings	25	5,96,749	5,44,488
	(ii) Other financial liabilities	26	98,814	1,03,165
(b)	Long-term provisions	27	2,368	
				3,260
(c)	Deferred tax liabilities (Net)	28	5,021	1,584
(d)	Other non-current liabilities	29	9,775	11,440
3 Curr	ent liabilities		7,12,727	6,63,937
(a)	Financial liabilities			
	(i) Short term borrowings	30	1,05,414	1,00,574
	(ii) Trade and other payables			a decide .
	from Micro and Small Enterprises	31	6	53
	from other than Micro and Small Enterprises	31	5,873	9,415
	(iii) Other financial liabilities	32	1,29,765	
(b)	Other current liabilities			1,69,725
		33	7,901	7,354
(c)	Short term provisions	34	3,777	263
(d)	Current tax liabilities (Net)	35	1,756	1,788
			2,54,492	2,89,172
	TOTAL EQUITY AND LIABILITIES	-	10,80,244	9,82,621
	accounting policies	2		
otes to th	e consolidated Ind AS financials statements	3-61		

# Significant accounting policies Notes to the consolidated Ind AS financials statements

The notes referred above form an integral part of these consolidated financial statements As per our report of even date attached

#### For DELOITTE HASKINS & SELLS LLP

Chartered Accountants

(Firm's Registration No 117366W/W-100018)

Rajesh K. Hiranandani Partner

JANA AT

Banmali grawala DIN - 00120029

3 - 61

ha

ann

Sanjay Dutt Managing Director DIN - 05251670 nor

Sanjay Sharma

Chief Financial Officer

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

utt

Director DIN - 00028428 the

F. N. Subedar

J. N. hubedar

Sudhakar Shetty Company Secretary Membership No: A13200

Mumbai 06 July, 2020

Mumbai 06 July, 2020

#### Consolidated Statement of Profit and Loss for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

		Note	31 March 2020	31 March 2019
1	Revenue from operations	36	1,57,879	1,67,816
П	Other income	37	5,998	6,616
ш	Total income (I +II)	- 12 LE	1,63,877	1,74,432
IV	Expenses			
	Cost of flats sold	38	12,119	6,122
	Construction costs	39	64,023	83,095
	Employee benefit expenses	40	7,985	8,259
	Finance costs	41	60,186	52,055
	Depreciation and amortisation expense	3(b)	17,251	16,858
	Other expenses	42	37,951	28,555
	Total expenses (IV)		1,99,515	1,94,944
v	Loss before share of loss (net) from Joint ventures and tax (III- IV)		(35,638)	(20,512)
VI	Add: Share of (loss) from joint ventures		(198)	(633)
VII	Loss before tax (V-VI)		(35,836)	(21,145)
VIII	Tax Expenses:			10.0
	Current tax		353	864
	Deferred tax charge	_	300	1,906
IX	Loss for the year (VII-VIII)		(36,489)	(23,915)
X	Less: Share in loss transferred to non-controlling interest		(233)	(222)
XI	Loss for the year attributable to owners (IX - X)		(36,256)	(23,693)
XII	Other Comprehensive Income			
	A Items that will not be reclassified to profit or loss			
	Remeasurements of defined benefit liability		3	55
	Income tax relating to items that will not be reclassified to profit or loss		(1)	(14)
	B Items that will be reclassified to profit or loss			•
XIII	Total Comprehensive Income for the year (XI + XII)		(36,254)	(23,652)
	(Comprising Loss and Other Comprehensive Income for the year after non-controlling interest	est)		
XIV	Earnings per equity share : (Face Value per share Rs. 10 each)		and	
	(1) Basic	43	(3.56)	(2.33)
	(2) Diluted *	-	(3.56)	(2.33)
	* Diluted EPS for the year ended 31 March 2020 is anti-dilutive hence not consider	ed		
Signi	ficant accounting policies	2		
Notes	s to the consolidated Ind AS financials statements	3 - 61		

Note

31 March 2020

31 March 2010

The notes referred above form an integral part of these consolidated financial statements. As per our report of even date attached.

For DELOITTE HASKINS & SELLS LLP

Chartered Accountants (Firm's Registration No. 117366W/W-100018)

Rajesh K. Hiranandani Partner

Mumbai 06 July, 2020

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

Bruns ŀ¥

Luchidas

Banmali Agrawala Chairman DIN - 00120029

Sanjay Dutt Managing Director DIN - 05251670

F. N. Subedar Director DIN - 00028428

onn Sanjay Sharma

Chief Financial Officer

Sudhakar Shetty Company Secretary Membership No: A13200

Mumbai 06 July, 2020

Consolidated statement of cash flows for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

	Cash flows from operating activities :	31 March 2020	31 March 2
	(Loss) before tax <u>Adjusted for:</u>	(35,836)	(21,1
	Depreciation and amortisation	17.351	1.0
	Excess provision written back	17,251	16.8
	Profit on sale of current investments in Mutual Funds	(10)	(1
	Provision for doubtful debts - credit impaired written back	(892)	(2
	Provision for Major Maintenance of Road	(172)	1.0
	Mark to Market loss / (gain) on Forward / Derivative Contracts	1,219	1.0
	Mark to Market gain on current investments in Mutual funds	1,103	(3,3
	Provision for credit impaired Trade Receivables	(435)	(1
	Provision for credit impaired Inter corporate deposits	306	3
	Advances written off	517 207	
	Interest income	(3,867)	4,0
	Finance costs		(1,9
	Share of loss from joint ventures	60,186 198	52,0
	Profit on sale of property, plant and equipment		6
	Impairment loss on Goodwill	(1)	
	Impairment loss on assets	11.11.1	7
	Loss on Sale of Non-Current Investments	11,314	
	Provision for Employee Benefits	314	
	Loss on sale of property, plant and equipment	1	I
		87,240	70.0
	Operating profit before working capital changes	51,404	48,8
	Changes in working capital	21,101	40,0
	(Increase) / Decrease in Trade Receivables	1153	1.3
	Decrease in Inventories	(153)	1.3
	(increase) / Decrease in Loans & Advances, Other Financial Assets and Other Current Assets	10,131	7
	Decrease in Trade payables	(1,130)	7.9
	Increase / (Decrease) in Other financial liabilities, current and non current liabilities and provisions	(3,584) 13,132	(1.2)
	Cash flows generated from operations	18,396	6,00
	Taxes paid (net of refund received)	69,800 (1,855)	54.8
	Net cash flows generated from operating activities	67,945	(2,3)
	Cash flows from investing activities :	41,945	52,0
	Payment for purchase and construction of property, plant and equipment	1100	
	Proceeds on sale of property, plant and equipment	(109)	(6
	Purchase of intangible assets / intangible assets under development	15	1
	Purchase of investment property / investment property under construction	(68,281)	(82,3-
	Investment in joint ventures	(23,651)	(38,0)
	Investment in equity shares of other companies	(5,756)	(10,4)
	Proceeds from sale of investments in mutual funds	(189)	
	Investments in mutual funds	3,78,060	1.52,37
	Redemption / (Investment) in fixed deposits under lien (net)	(4,46,530)	(1,45,42
	Proceeds on account of divestment of investments in subsidiaries	1,360	(1,23
	Interest received	9,177	
	Net cash flows (used in) investing activities	3,783	2,19
		(1,52,121)	(1.23.50
	Cash flows from financing activities : Proceeds / (Repayment) from short term borrowings (net)	1.010	
	Proceeds from long term borrowings	4,840	39,27
	(Repayment) of long term borrowings	99,216	1,25,58
	Rights Issue Application Money Received	(47,330)	(36,47
	Proceeds from Minority Interest	1,20,000	
	Finance costs paid		- 1
	Net cash generated from financing activities	(68,060)	(58,61
		1,08,666	69,78
	Net (decrease) in cash and bank balances (A+B+C) Cash and bank balances, beginning of the year	24,490	(1,22
		5,921	7.14
10			18.
	Cash and bank balances, end of year (refer note 17)	(471) 29,940	5.92

#### Tata Realty and Infrastructure Limited Consolidated statement of cash flows as at 31 March 2019 (Continued)

(Currency: Indian rupees in lakhs)

Notes: 1)	Cash and bank balances include the following :			
	Cash and cash equivalents	-4	31 March 2020	31 March 2019
	Cash balance		30	93
	Balance with scheduled banks:			
	- in current accounts		3,300	2,099
	- in deposit accounts		26,610	3,729
			29,940	5,921
Significar	accounting policies	2		

#### Significant accounting policies Notes to the consolidated Ind AS financial statements

3 - 61

The notes referred above form an integral part of these consolidated financial statements As per our report of even date attached

For DELOITTE HASKINS & SELLS LLP Chartered Accountants (Firm's Registration No. 117366W/W-100018)

Rajesh K. Hiranandani

Partner

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

hubedair samuel tt 6 170 Banmali Agrawala Sanjay Dutt F. N. Subedar Managing Director DIN - 05251670

Chairman DIN - 00120029

Director DIN - 00028428

anna Sanjay Sharma Chief Financial Officer

Sudhakar Shetty

Company Secretary Membership No: A13200

Mumbai

July 06, 2020

Mumbai July 06, 2020

Balance as at 31 March 2019 Stare application pending allotment received during the year Loss for the year Remeasurements of defined benefit liability Income tax relating to items that will not be reclassified to profit or loss Balance as at 31 March 2020

11

Significant accounting policies Notes to the consolidated Ind AS financials statements The notes referred above form an integral part of these consolidated financial statements As per our report of even date attached

Consolidated statement of changes in equity for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)								
(a) Equity share capital						Notes	31 March 2020 Amount	Amount
Balance as at 1 April Changes in equity share capital during the year						23	1,01,731	1,01,731
Balance as at 31 March						-	1,01,731	1.01.731
(b) Other equity		-						
		A	tributable to th	e owners of the	parent		Non-	Total
Particulars	Share application money	Re	erves and surpl	us	Other comprehensive income	Total other equity	controlling interests	
	pending allotment	Capital reserve	Securities premium reserve	Retained earnings	Defined benefit plan adjustment			
Balance as at 1 April 2018		4,911	15,769	(74,541)	(14)	(53,875)	6,826	(47,049)
	-	4,911	15,769	(74,541) (23,693)	(14)	(53,875) (23,693)	6,826 (222)	(47,049) (23,915)
Loss for the year	10.00	4,911						(23,915
Loss for the year Remeasurements of defined benefit	-	-	** 20	(23,693)		(23,693)		
Loss for the year Remeasurements of defined benefit Issue of new shares / CCD in subsidiaries		2	21 21 21	(23,693)	55	(23,693) 55	(222)	(23,915 55
Balance as at 1 April 2018 Loss for the year Remeasurements of defined benefit Issue of new shares / CCD in subsidiaries Income tax relating to items that will not be reclassified to profit or loss Consolidation adjustments	-	2	** 20	(23,693)	55	(23,693) 55	(222)	(23,915 55 13

1,20,000

1,20,000

2

4,911

4,911

15,769

15,769

Mumbai 06 July, 2020

Rajesh K. Hiranandani Partner

Balance as at 31 March 2019

For DELOITTE HASKINS & SELLS LLP Chartered Accountants (Firm's Registration No 117366W/W-100018)

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300 F. N. Subedar Director DIN-00028428 Ranna tt Sanjay Dutt Managing Director DIN - 05251670 Banmali Agrawala Chairman DIN - Ou 20029 we. sig Sanjay Sharma Sudhakar Shetty Chief Financial Officer Company Secretary Membership No: A13200

27

29

(78,836) 1,20,000 (36,256)

3

4,910

(72,220) 1,20,000 (36,489)

(1) 11,294

Mumbai 06 July, 2020

6,617

(233)

6,384

(99,543)

(36,256)

(1.35,799)

378

#### Notes to the consolidated financial statements

for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

#### 1 Background and basis of preparation

#### 1A Background

Tata Realty & Infrastructure Limited ("TRIL" or 'the Group') along with its subsidiaries, joint ventures and associates, collectively referred to as the "the Group" is engaged in the business of investment advisory services, project management consultancy services and real estate and infrastructure development. The parent company is a wholly owned subsidiary of Tata Sons Private Limited.

#### 1B Basis of preparation

#### (a) Statement of compliance

These Consolidated Ind AS financial statements have been prepared in accordance with Indian Accounting Standards (Ind AS) as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 ('the Act') and other relevant provisions of the Act.

#### (b) Going Concern

The Group has incurred losses amounting to Rs. 36,256 Lakhs in the current year (previous year Rs. 23,693 Lakhs). As at 31 March 2020 the Company has a net current liability position of Rs. 99,648 Lakhs where the current liabilities at Rs. 2,54,492 Lakhs exceed the current assets at Rs. 1,54,844 Lakhs. Based on scheduled repayment Rs 1,98,492 Lakhs is due for repayment within 12 months from the from the reporting date.

Assessment: The Board of Directors have assessed the above operational conditions and indicators and have come to the conclusion that no material uncertainty exists that may cast significant doubt on the Group's ability to continue as a going concern based on cash flow forecasts and the plan management has put in place.

Mitigating factors: The Group is in the business of developing commercial & residential Real Estate and Infrastructure Development to generate stable cash flows over the life of the assets. The Group is developing asset through investments in Subsidiaries and Joint Ventures in various project SPVs. During the year, two projects become operational and started generating lease rental and one infrastructure project has started toll collection in addition to ongoing projects.

During FY 2020-21 based on projections, the SPV's are expected to generate operational net cash flow of more than Rs 450,00 Lakhs which will increase the value of investments in subsidiaries and joint venture SPVs. The Group has robust divestment strategy which is demonstrated by divestment of Retail business SPV during FY 2020. The management is evaluating possibility of divestment of selected assets and change in capital structure in its project SPVs' which is expected to generate more than Rs 1,50,000 Lakhs as equity value. Also, the free cash flow from sale of Ready to Move in (RTMI) residential inventory will support its operations during coming financial years.

Negative working capital is on account of management decision to borrow short-term funds through commercial papers to take advantage of interest arbitrage. However, management has modified the strategy to replace, to the extent possible, short term funding with long term funding arrangement going forward.

The equity capital from the parent i.e. Tata Sons Private Limited, of an amount of Rs. 1,20,000 Lakhs received during the year has improved the company's networth allowing the company further ability for additional borrowing in future. This plan is also supported by ratings from ICRA and CARE, for short term and long term borrowings of A1+ and AA with stable outlook respectively.

Conclusion: The Board of Directors based on cash flow forecasts and management plans have concluded on ability of the Company to continue as going concern and the financial statements have been prepared on that basis.

#### (c) Functional and presentation currency

These Consolidated financial statements are presented in Indian Rupees (INR), which is also the Company's functional currency. All the financial information have been presented in Indian Rupees (INR) and all amounts have been rounded-off to the nearest lakhs with two decimals, except for share data and as otherwise stated. Due to rounding off, the numbers presented throughout the document may not add up precisely to the totals and percentage may not precisely reflect the absolute figures.

#### (d) Basis of measurement

The Consolidated Ind AS financial statements have been prepared on the historical cost basis except for the following assets and liabilities which have been measured at fair value:

Certain financial assets and liabilities (including derivative instruments)

2 Defined benefit plans - plan assets measured at fair value

#### Notes to the consolidated financial statements

for the year ended 31 March 2020

#### (Currency: Indian rupees in lakhs)

#### (e) Critical accounting judgements and key sources of estimation of uncertainty

In preparing these Consolidated Ind AS financial statements, management has made judgements, estimates and assumptions that affect the application of accounting policies and the reported amounts of assets, liabilities, income and expenses. Actual results may differ from these estimates. Estimates and underlying assumptions are reviewed on an ongoing basis. Revisions to accounting estimates are recognised prospectively.

#### (i) Judgements

Information about judgements made in applying accounting policies that have the most significant effects on the amounts recognised in the Ind AS financial statements is included in the subsequent notes.

#### (ii) Control

If the Group owns less than one-half of the voting power and it is able to control the company which inter-alia provides the Group with power to appoint majority of the board of directors and power over relevant activities. Consequently, the Group consolidates its investment in the company. (Refer Note No. 52(a))

If the Group owns more than one-half of the voting power but it does not have control over the Companies then the Group has joint control over the companies. In other words, decisions about the relevant activities - i.e. those that significantly affect the returns of the arrangement - require the affirmative consent of the investors. Consequently, the Group has classified its interest in these companies as jointly controlled entity, (Refer Note no. 52(b))

#### (iii) Assumptions and estimation uncertainties

Information about assumptions and estimation uncertainties that have a significant risk of resulting in a material adjustment in the year ended 31 March 2020 is included in the following notes:

Note - impairment test of non-financial assets: key assumptions underlying recoverable amounts including the recoverability of expenditure on acquired intangible assets (goodwill); Useful life of Property, plant and equipment;

Note - recognition of deferred tax assets availability of future taxable profit against which tax losses carried forward can be used;

Note - recognition and measurement of provisions and contingencies; key assumptions about the likelihood and magnitude of an outflow of resources;

Note - measurement of defined benefit obligations: key actuarial assumptions,

Note - acquisition of subsidiary: fair value of the consideration transferred (including contingent consideration) and fair value of the assets acquired and liabilities assumed, measured on a provisional basis; and

Note - determining the fair value of investments on the basis of significant unobservable inputs;

Note - impairment of financial assets.

#### (f) Measurement of fair values

The Group's accounting policies and disclosures require the measurement of fair values, for both financial and non-financial assets and liabilities. The Group has an established control framework with respect to the measurement of fair values. The finance team has overall responsibility for overseeing all significant fair value measurements, including Level 3 fair values, and reports directly to the CFO.

They regularly review significant unobservable inputs and valuation adjustments. If third party information is used to measure fair values then the finance team assesses the evidence obtained from the third parties to support the conclusion that such valuations meet the requirements of Ind AS, including the level in the fair value hierarchy in which such valuations should be classified.

When measuring the fair value of an asset or a liability, the Company uses observable market data as far as possible. Fair values are categorized into different levels in a fair value hierarchy based on the inputs used in the valuation techniques as follows:

Level 1: quoted prices (unadjusted) in active markets for identical assets or liabilities.

Level 2: inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly (i.e. as prices) or indirectly (i.e. derived from prices).

Level 3: inputs for the asset or liability that are not based on observable market data (unobservable inputs).

If the inputs used to measure the fair value of an asset or a liability fall into different levels of the fair value hierarchy, then the fair value measurement is categorized in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement.

#### (g) Foreign currency transactions

Foreign currency transactions are translated into the functional currency using the exchange rates at the dates of the transactions. Foreign exchange gains and losses resulting from the settlement of such transactions and from the translation of monetary assets and liabilities denominated in foreign currencies at year end exchange rates are generally recognised in profit or loss.

Non-monetary items that are measured at fair value in a foreign currency are translated using the exchange rates at the date when the fair value was determined. Translation differences on assets and liabilities carried at fair value are reported as part of the fair value gain or loss. The gain or loss arising on translation of nonmonetary items measured at fair value is treated in line with the recognition of the gain or loss on the change in fair value of the item (i.e., translation differences on items whose fair value gain or loss is recognised in OCI or profit or loss are also recognised in OCI or profit or loss, respectively).

#### Notes to the consolidated financial statements

#### for the year ended 31 March 2020

#### (Currency: Indian rupees in lakhs)

#### 2.01 Basis of consolidation

(i) Subsidiaries

Subsidiaries are all entities (including structured entities) over which the group has control. The group controls an entity when the group is exposed to, or has rights to, variable returns from its involvement with the entity and has the ability to affect those returns through its power to direct the relevant activities of the entity. Subsidiaries are fully consolidated from the date on which control is transferred to the group. They are deconsolidated from the date that control ceases.

The group combines the financial statements of the parent and its subsidiaries line by line adding together like items of assets, liabilities, equity, income and expenses. Intergroup transactions, balances and unrealised gains on transactions between group companies are eliminated. Unrealised losses are also eliminated unless the transaction provides evidence of an impairment of the transferred asset. Accounting policies of subsidiaries have been changed where necessary to ensure consistency with the policies adopted by the group.

Non-controlling interests in the results and equity of subsidiaries are shown separately in the consolidated statement of profit and loss, consolidated statement of changes in equity and balance sheet respectively.

#### (ii) Joint arrangements

Under Ind AS 111 Joint Arrangements, investments in joint arrangements are classified as either joint operations or joint ventures. The classification depends on the contractual rights and obligations of each investor, rather than the legal structure of the joint arrangement. The group has only joint ventures.

#### (iii) Joint ventures

Joint ventures are entities over which the group has joint control along with another entity. Interests in joint ventures are accounted for using the equity method, after initially being recognized at cost in the consolidated balance sheet. The investment is initially recognized at cost, and the carrying amount is increased or decreased to recognize the investor's share of the profit or loss of the investee after the acquisition date.

(iv) Associates

Associates are entities over which the group has significant influence but not control. Investments in associates are accounted for using the equity method of accounting. The investment is initially recognized at cost, and the carrying amount is increased or decreased to recognize the investor's share of the profit or loss of the investee after the acquisition date.

#### 2.02 Segment reporting

Operating segments are reported in a manner consistent with the internal reporting provided to the chief operating decision maker (CODM). Chief operating decision maker's function is to allocate the resources of the entity and access the performance of the operating segment of the Group.

The Managing Director assesses the financial performance and position of the Group and makes strategic decisions and is identified as being the chief operating decision maker for the Group. Refer note 46 for segment information presented.

#### 2.03 Revenue recognition

#### (i) Sale of completed property

Revenue from sale of completed property (residential and commercial) is recognised when:

1. The Group has transferred to the buyer significant risk and rewards of ownership of the completed property;

2. The Group retains neither continuing managerial involvement to the degree usually associated with the ownership nor effective control over the completed property sold;

- 3. The amount of revenue can be measured reliably;
- 4. It is probable that the economic benefit associated with the transaction will flow to the Group; and
- 5. Cost incurred or to be incurred in respect of the transaction can be measured reliably.

#### (ii) Service concession arrangements

Concession arrangements are recognized in accordance with Appendix C of Ind AS 115, Service Concession Arrangements. It is applicable to concession arrangements comprising a public service obligation and satisfying all of the following criteria:

• the concession grantor controls or regulates the services to be provided by the operator using the asset, the infrastructure, the beneficiaries of the services and prices applied;

· the grantor controls the significant residual interest in the infrastructure at the end of the term of the arrangement.

Pursuant to Appendix C of Ind AS 115, such infrastructures are not recognized in assets of the operator as property, plant and equipment but in financial assets ("financial asset model") and/or intangible assets ("intangible asset model") depending on the remuneration commitments given by the grantor.

The intangible asset model applies where the operator is paid by the users or where the concession grantor has not provided a contractual guarantee in respect of the recoverable amount. The intangible asset corresponds to the right granted by the concession grantor to the operator to charge users of the public service in remuneration of concession services.

Intangible assets resulting from the application of Appendix C of Ind AS 115 are recorded in the financial statements as intangible assets. The Group accounts for such intangible asset (along with the present value of committed payments towards concession arrangement to the grantor at the appointed date e.g.

Negative Grant, premium etc.) in accordance with the provisions of Ind AS 38 and is amortized based on projected traffic count or revenue, as detailed in Note 2.07(ii), taking into account the estimated period of commercial operation of infrastructure which generally coincides with the concession period. And are amortized, generally on a straight-line basis, over the contract term.

#### Under the intangible asset model, revenue includes:

- revenue recorded on a completion basis for assets and infrastructure under construction (in accordance with IND AS 115);
- charges collected from users





#### Notes to the consolidated financial statements

for the year ended 31 March 2020

#### (Currency: Indian rupees in lakhs)

However, in certain concession arrangements, contracts may include a payment commitment on the part of the concession grantor covering only part of the investment, with the balance covered by amount charged to users. Where this is the case, the investment amount guaranteed by the concession grantor is recognized under the financial asset model and the residual balance is recognized under the intangible asset model. Financial assets resulting from that are recorded in the financial statements under the heading other financial assets and recognized at amortized cost.

Further, where infrastructure is partly regulated and partly unregulated, the portion of infrastructure that is physically separable and capable of being operated independently and meets definition of cash generating unit as defined under Ind AS 36 is analysed separately if it is used wholly for unregulated purposes.

#### (iii) Rendering of services

Asset management fees and Project management consultancy fees are recognized in accordance with terms of agreement with customers.

#### (iv) Rental income

Rental income from investment property is recognised as part of revenue from operations in profit or loss on a straight-line basis over the term of the lease. Lease incentives granted are recognised as an integral part of the total rental income, over the term of the lease. Rental income from sub-leasing is also recognised in a similar manner and included under other income.

#### (v) Interest and dividend:

A dividend is recognized as revenue when the right to receive payment has been established. For all debt instruments measured either at amortised cost or at fair value through other comprehensive income, interest income is recorded using the effective interest rate (EIR). EIR is the rate that exactly discounts the estimated future cash payments or receipts over the expected life of the financial instrument or a shorter period, where appropriate, to the gross carrying amount of the financial asset or to the amortised cost of a financial liability.

#### 2.04 Government grants

Grants from the government are initially recognised as deferred income at their fair value where there is a reasonable assurance that the grant will be received and the group will comply with all attached conditions, they are then recognised in profit or loss as other income on a systematic basis over the useful life of the asset.

Grants that compensate the Group for expenses incurred are recognised in profit or loss on a systematic basis in the periods in which the expenses are recognised.

#### 2.05 Property, plant and equipment

#### (i) Recognition and measurement

Freehold land is carried at historical cost. All other items of property, plant and equipment are measured at cost less accumulated depreciation and any accumulated impairment losses.

The cost of an item of property, plant and equipment comprises:

a) its purchase price, including import duties and non-refundable purchase taxes, after deducting trade discounts and rebates.

any costs directly attributable to bringing the asset to the location and condition necessary for it to be capable of operating in the manner intended by
 the initial estimate of the costs of dismantling and removing the item and restoring the site on which it is located, the obligation for which an entity incurs either when the item is acquired

Income and expenses related to the incidental operations, not necessary to bring the item to the location and condition necessary for it to be capable of operating in the manner intended by management, are recognised in profit or loss.

If significant parts of an item of property, plant and equipment have different useful lives, then they are accounted for as separate items (major components) of property, plant and equipment. Any gain or loss on disposal of an item of property, plant and equipment is recognised in profit or loss.

#### (ii) Subsequent expenditure

Subsequent expenditure is capitalised only if it is probable that the future economic benefits associated with the expenditure will flow to the Group.

#### 2.05 Property, plant and equipment (Continued)

#### (iii) Depreciation

Depreciable amount for assets is the cost of an asset, or other amount substituted for cost, less its estimated residual value.

Depreciation is provided using the straight line method in the manner and at the rates prescribed by Part 'C' of Schedule 11 of the Act. Depreciation is charged on a monthly pro-rata basis for assets purchased or sold during the year.

In the following cases, the useful life is less than the corresponding useful life prescribed in Part 'C' of Schedule II of the Act, based on internal technical evaluation, taking into account the nature of the assets, the estimated usage of the asset, the operating conditions of the asset, past history of replacement, anticipated technological changes etc.:

Plant & Machinery	8-15 years
Furniture & Fixture	6-10 years
Office equipments	3-6 years
Computers	3-4 years
Motor Vehicles	5-7 years
Commercial building	5-32 years
Leasehold improvements are an	nortised over the lease ne

Leasehold improvements are amortised over the lease period.

Depreciation methods, useful lives and residual values are reviewed at each reporting date and adjusted if appropriate.

Gains and losses on disposals are determined by comparing proceeds with carrying amount. These are included in profit or loss within other gains/ (losses).

#### Notes to the consolidated financial statements

for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

#### 2.06 Investment property

Property that is held for long-term rental yields or for capital appreciation or both, and that is not occupied by the group, is classified as investment property. Investment property is measured initially at its cost, including related transaction costs and where applicable borrowing costs. Subsequent expenditure is capitalised to the asset's carrying amount only when it is probable that future economic benefits associated with the expenditure will flow to the group and the cost of the item can be measured reliably. All other repairs and maintenance costs are expensed when incurred. When part of an investment property is replaced, the carrying amount of the replaced part is derecognised.

#### Investment property under construction

Property that is being constructed for future use as investment property is accounted for as investment property under construction until construction or development is complete. All costs which are directly attributable to construction of the investment property are capitalized.

#### 2.07 Intangible assets

#### Goodwill

Goodwill on acquisitions of subsidiaries is included in intangible assets. Goodwill is not amortised but it is tested for impairment annually. It is carried at cost less accumulated impairment losses. Gains and losses on the disposal of an entity include the carrying amount of goodwill relating to the entity sold.

Goodwill is allocated to cash-generating units for the purpose of impairment testing. The allocation is made to those cash-generating units or groups of cashgenerating units that are expected to benefit from the business combination in which the goodwill arose. The units or groups of units are identified at the lowest level at which goodwill is monitored for internal management purposes, which in our case are the operating segments.

Intangible assets comprise application software purchased, which are not an integral part of the related hardware, and are amortized on a written down value basis over a period of ten years, which in Management's estimate represents the period during which the economic benefits will be derived from their use.

Subsequent expenditure is capitalized only when it increases the future economic benefits embodied in the specific to which it relates

#### Intangible assets under development

#### (i) Passenger ropeway facility

Intangible assets comprise a right to charge users for passenger ropeway facility, which is currently under development. The cost of intangible assets acquired in exchange of monetary or non-monetary assets is measured at fair value unless (a) the exchange transaction lacks commercial substance or (b) the fair value of neither the asset received nor the asset given up is reliably measurable. If acquired asset cannot be measured at fair value, its cost is measured at the carrying amount of the asset given up. The fair value of the asset is measured with reference to the fair value of construction services provided.

Subsequent expenditure is capitalized only when it increases the future economic benefits embodied in the specific to which it relates.

#### (ii) Toll collection right

Toll collection rights representing right to receive fees from users of facility are obtained in consideration for rendering construction, operation and maintenance services in relation to building and maintenance of the project on Build, Operate and Transfer basis. The consideration received in a service concession arrangement is measured at fair value upon initial recognition. i.e. construction cost. Subsequent to initial recognition the intangible asset is measured at cost, which includes capitalised borrowing costs, less accumulated amortisation and accumulated impairment losses.

#### Amortization of concession intangible assets

Amortization of intangible asset under SCA. The intangible rights relating to infrastructure assets, which are recognized in the form of right to charge users of the infrastructure asset are amortized by taking proportionate of actual traffic count for the period over total projected traffic count from project to cost of intangible assets; i.e. proportionate of actual traffic for the period over total projected traffic count from the intangible assets expected to be earned over the balance concession period as estimated by the management. However with respect to total projected revenue and in operation as at March 31, 2016, the amortization of such intangible assets, intend of traffic count. Total projected revenue earned to total projected revenue from the project over the balance concession period to cost intangible assets, instead of traffic count. Total projected revenue / traffic count is reviewed at the end of each financial year and is adjusted to reflect any changes in the estimates which lead to the actual collection at the end of the concession period.

#### 2.08 Impairment of non-financial assets

Goodwill and intangible assets that have an indefinite useful life are not subject to amortisation and are tested annually for impairment, or more frequently if events or changes in circumstances indicate that they might be impaired. Other assets are tested for impairment whenever events or changes in circumstances indicate that the carrying amount may not be recoverable. Management periodically assesses using, external and internal sources, whether there is an indication that an asset may be impaired.

The recoverable amount is higher of the asset's net selling price or value in use, which means the present value of future cash flows expected to arise from the continuing use of the asset and its eventual disposal. An impairment loss for an asset is reversed if, and only if, the reversal can be related objectively to an event occurring after the impairment loss was recognized. The carrying amount of an asset is increased to its revised recoverable amount, provided that this amount does not exceed the carrying amount that would have been determined (net of any accumulated amortization or depreciation) had no impairment loss been recognized for the asset in prior years.

#### Notes to the consolidated financial statements

for the year ended 31 March 2020

#### (Currency: Indian rupees in lakhs)

#### 2.09 Borrowing cost

General and specific borrowing costs that are directly attributable to the acquisition, construction or production of a qualifying asset are capitalised during the period of time that is required to complete and prepare the asset for its intended use or sale. Qualifying assets are assets that necessarily take a substantial period of time to get ready for their intended use or sale.

Investment income earned on the temporary investment of specific borrowings pending their expenditure on qualifying assets is deducted from the borrowing costs eligible for capitalisation.

Other borrowing costs are expensed in the period in which they are incurred.

#### 2.10 Income-tax

Income tax expense comprises current and deferred tax. It is recognised in profit or loss except to the extent that it relates items recognised directly in equity or in OCI

#### (i) Current tax

Current tax comprises the expected tax payable or receivable on the taxable income or loss for the year and any adjustment to the tax payable or receivable in respect of previous years. It is measured using tax rates enacted or substantively enacted at the reporting date. Current tax also includes any tax arising from dividends. Management periodically evaluates positions taken in tax returns with respect to situations in which applicable tax regulation is subject to interpretation. It establishes provisions where appropriate on the basis of amounts expected to be paid to the tax authorities.

Minimum Alternate Tax ('MAT') under the provisions of Income-tax Act, 1961 is recognised as current tax in the statement of profit and loss. MAT paid in accordance with the tax laws, which gives future economic benefits in the form of adjustment to future income tax liability, is considered as an asset if there is a convincing evidence that the Group will pay normal tax. Accordingly, MAT is recognized as an asset in the balance sheet when it is probable that the future economic benefit associated with it will flow to the Group.

Current tax assets and liabilities are offset only if, the Group:

- a) has a legally enforceable right to set off the recognised amounts; and
- b) intends either to settle on a net basis, or to realise the asset and settle the liability simultaneously.

#### (ii) Deferred tax

Deferred tax is recognised in respect of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for taxation purposes. However, deferred tax liabilities are not recognised if they arise from the initial recognition of goodwill. Deferred income tax is also not accounted for if it arises from initial recognition of an asset or liability in a transaction other than a business combination that at the time of the transaction affects neither accounting profit nor taxable profit (tax loss).

Deferred tax assets are recognised for unused tax losses, unused tax credits and deductible temporary differences to the extent that it is probable that future taxable profits will be available against which they can be used. Deferred tax assets are reviewed at each reporting date and are reduced to the extent that it is no longer probable that the related tax benefit will be realised; such reductions are reversed when the probability of future taxable profits improves. Unrecognised deferred tax assets are reassessed at each reporting date and recognised to the extent that it has become probable that future taxable profits will be available against which they can be used.

Deferred tax is measured at the tax rates that are expected to be applied to temporary differences when they reverse, using tax rates enacted or substantively enacted at the reporting date and are expected to apply when the related deferred income tax asset is realised or the deferred income tax liability is settled.

Deferred tax liabilities are not recognised for temporary differences between the carrying amount and tax bases of investments in subsidiaries, branches and associates and interest in joint arrangements where the group is able to control the timing of the reversal of the temporary differences and it is probable that the differences will not reverse in the foreseeable future.

Deferred tax assets are not recognised for temporary differences between the carrying amount and tax bases of investments in subsidiaries, branches and associates and interest in joint arrangements where it is not probable that the differences will reverse in the foreseeable future and taxable profit will not be available against which the temporary difference can be utilised.

Deferred tax assets and liabilities are offset only if:

- a) the entity has a legally enforceable right to set off current tax assets against current tax liabilities; and
- b) the deferred tax assets and the deferred tax liabilities relate to income taxes levied by the same taxation authority on the same taxable entity.

#### 2.11 Business combinations

The acquisition method of accounting is used to account for all business combinations, regardless of whether equity instruments or other assets are acquired. The consideration transferred for the acquisition of a subsidiary comprises the:

- fair values of the assets transferred;
- liabilities incurred to the former owners of the acquired business;
- equity interests issued by the group; and
- · Fair value of any asset or liability resulting from a contingent consideration arrangement.

Identifiable assets acquired and liabilities and contingent liabilities assumed in a business combination are, with limited exceptions, measured initially at their fair values at the acquisition date. The group recognises any non-controlling interest in the acquired entity on an acquisition-by-acquisition basis either at fair value or at the non-controlling interest's proportionate share of the acquired entity's net identifiable assets. Acquisition-related costs are expensed as incurred.

#### The excess of the

· consideration transferred;

- amount of any non-controlling interest in the acquired entity, and



#### Notes to the consolidated financial statements

for the year ended 31 March 2020

#### (Currency: Indian rupees in lakhs)

- Acquisition-date fair value of any previous equity interest in the acquired entity over the fair value of the net identifiable assets acquired is recorded as goodwill. If those amounts are less than the fair value of the net identifiable assets of the business acquired, the difference is recognised in other comprehensive income and accumulated in equity as capital reserve provided there is clear evidence of the underlying reasons for classifying the business combination as a bargain purchase. In other cases, the bargain purchase gain is recognised directly in equity as capital reserve.

• Where settlement of any part of cash consideration is deferred, the amounts payable in the future are discounted to their present value as at the date of exchange. The discount rate used is the entity's incremental borrowing rate, being the rate at which a similar borrowing could be obtained from an independent financier under comparable terms and conditions.

· Contingent consideration is classified either as equity or a financial liability. Amounts classified as a financial liability are subsequently remeasured to fair value with changes in fair value recognised in profit or loss.

 If the business combination is achieved in stages, the acquisition date carrying value of the acquirer's previously held equity interest in the acquire is remeasured to fair value at the acquisition date. Any gains or losses arising from such remeasurement are recognised in profit or loss or other comprehensive income, as appropriate.

#### 2.12 Cash and cash equivalents

For the purpose of presentation in the statement of cash flows, cash and cash equivalents includes cash on hand, deposits held at call with financial institutions, other short-term, highly liquid investments with original maturities of three months or less that are readily convertible to known amounts of cash and which are subject to an insignificant risk of changes in value, and bank overdrafts. Bank overdrafts are shown within borrowings in current liabilities in the balance sheet.

#### 2.13 Inventories

Direct expenses like land cost, development rights, site labour cost, material used for project construction, cost of borrowing, project management consultancy, costs for moving the plant and machinery to the site and general expenses incurred specifically for the respective project like insurance, design and technical assistance, and construction overheads are taken as the cost of the project work in progress and cost of unsold flats.

#### Material at site comprise of building material, components, stores and spares.

Inventories are valued at lower of cost or net realizable value, cost is determined on weighted average basis. Net realizable value is the estimated selling price in the ordinary course of business, less estimated costs of completion and estimated costs necessary to make the sale.

#### 2.14 Financial instruments

A financial instrument is any contract that gives rise to a financial asset of one entity and a financial liability or equity instrument of another entity. Financial instruments also include derivative contracts such as foreign currency foreign exchange forward contracts, interest rate swaps and currency options; and embedded derivatives in the host contract.

#### (i) Financial assets

#### Classification

The Group shall classify financial assets as subsequently measured at amortised cost, fair value through other comprehensive income (FVOCI) or fair value through profit or loss (FVTPL) on the basis of its business model for managing the financial assets and the contractual cash flow characteristics of the financial asset.

#### Initial recognition and measurement

All financial assets are recognised initially at fair value plus, in the case of financial assets not recorded at fair value through profit or loss, transaction costs that are attributable to the acquisition of the financial asset. Purchases or sales of financial assets that require delivery of assets within a time frame established by regulation or convention in the market place (regular way trades) are recognised on the trade date, i.e., the date that the Group commits to purchase or sell the asset.

#### Debt instruments

§ A 'debt instrument' is measured at the amortised cost if both the following conditions are met:

a) The asset is held within a business model whose objective is to hold assets for collecting contractual cash flows, and

b) Contractual terms of the asset give rise on specified dates to cash flows that are solely payments of principal and interest (SPPI) on the principal amount outstanding.

§ After initial measurement, such financial assets are subsequently measured at amortised cost using the effective interest rate (EIR) method. Amortised cost is calculated by taking into account any discount or premium and fees or costs that are an integral part of the EIR. The EIR amortisation is included in finance income in the profit or loss.

§ Debt instruments included within the fair value through profit and loss (FVTPL) category are measured at fair value with all changes recognized in the statement of profit and loss.

#### Equity instruments

§ The group subsequently measures all equity investments in companies other than equity investments in, joint ventures and associates at fair value. Where the group's management has elected to present fair value gains and losses on equity investments in other comprehensive income, there is no subsequent reclassification of fair value gains and losses to profit or loss. Dividends from such investments are recognised in profit or loss as other income when the group's right to receive payments is established.

#### Notes to the consolidated financial statements

#### for the year ended 31 March 2020

#### (Currency: Indian rupees in lakhs)

#### **De-recognition**

§ A financial asset (or, where applicable, a part of a financial asset or part of a Group of similar financial assets) is primarily derecognised (i.e. removed from the Group's balance sheet) when:

§ The rights to receive cash flows from the asset have expired, or

§ The Group has transferred its rights to receive cash flows from the asset or has assumed an obligation to pay the received cash flows in full without material delay to a third party under a 'pass-through' arrangement; and either (a) the Group has transferred substantially all the risks and rewards of the asset, or (b) the Group has neither transferred nor retained substantially all the risks and rewards of the asset, but has transferred control of the asset.

§ When the Group has transferred its rights to receive cash flows from an asset or has entered into a pass-through arrangement, it evaluates if and to what extent it has retained the risks and rewards of ownership. When it has neither transferred nor retained substantially all of the risks and rewards of the asset, nor transferred control of the asset, the Group continues to recognise the transferred asset to the extent of the Group's continuing involvement. In that case, the Group also recognises an associated liability. The transferred asset and the associated liability are measured on a basis that reflects the rights and obligations that the Group as retained.

§ Continuing involvement that takes the form of a guarantee over the transferred asset is measured at the lower of the original carrying amount of the asset and the maximum amount of consideration that the Group could be required to repay.

#### Impairment of financial assets

In accordance with Ind-AS 109, the Group applies expected credit loss (ECL) model for measurement and recognition of impairment loss on the following financial assets and credit risk exposure:

a) Financial assets that are debt instruments, and are measured at amortised cost e.g., loans, debt securities, deposits, and bank balance

- b) Lease receivables
- c) Trade receivables

The Group follows 'simplified approach' for recognition of impairment loss allowance on:

§ Trade receivables which do not contain a significant financing component.

§ All lease receivables resulting from transactions.

The application of simplified approach does not require the Group to track changes in credit risk. Rather, it recognises impairment loss allowance based on lifetime ECLs at each reporting date, right from its initial recognition.

For recognition of impairment loss on other financial assets and risk exposure, the Group determines that whether there has been a significant increase in the credit risk since initial recognition. If credit risk has not increased significantly, 12-month ECL is used to provide for impairment loss. However, if credit risk has increased significantly, lifetime ECL is used. If, in a subsequent period, credit quality of the instrument improves such that there is no longer a significant increase in credit risk since initial recognition, then the entity reverts to recognising impairment loss allowance based on 12-month ECL.

#### 2.14 Financial instruments (Continued)

#### **Derivative contracts**

The group enters into a variety of derivative financial instruments to manage its exposure to interest rate and foreign exchange rate risk, including foreign exchange forward contracts, interest rate swaps and cross currency swaps. Derivatives are initially recognised at fair value at the date the derivative contracts are entered into and are subsequently remeasured to their fair value at the end of each reporting period. The resulting gain/loss is recognised in profit or loss immediately.

#### (ii) Financial liabilities

The Group classifies all financial liabilities as subsequently measured at amortised cost, except for financial liabilities at fair value through profit or loss. Such liabilities, including derivatives that are liabilities, shall be subsequently measured at fair value.

#### Initial recognition and measurement

Financial liabilities are classified, at initial recognition, as financial liabilities at fair value through profit or loss, loans and borrowings, payables, or as derivatives designated as hedging instruments in an effective hedge, as appropriate.

All financial liabilities are recognised initially at fair value and, in the case of loans and borrowings and payables, net of directly attributable transaction costs.

The Group's financial liabilities include trade and other payables, loans and borrowings including bank overdrafts, financial guarantee contracts and derivative financial instruments.

#### Financial liabilities at fair value through profit or loss

Financial liabilities at fair value through profit or loss include financial liabilities held for trading and financial liabilities designated upon initial recognition as at fair value through profit or loss. Financial liabilities are classified as held for trading if they are incurred for the purpose of repurchasing in the near term. This category also includes derivative financial instruments entered into by the Group that are not designated as hedging instruments in hedge relationships as defined by Ind-AS 109. Separated embedded derivatives are also classified as held for trading unless they are designated as effective hedging instruments.

#### Gains or losses on liabilities held for trading are recognised in the profit or loss.

Financial liabilities designated upon initial recognition at fair value through profit or loss are designated at the initial date of recognition, and only if the criteria in Ind-AS 109 are satisfied. For liabilities designated as FVTPL, fair value gains/ losses attributable to changes in own credit risk are recognized in OCI. These gains/loss are not subsequently transferred to profit and loss. However, the Group may transfer the cumulative gain or loss within equity. All other changes in fair value of such liability are recognised in the statement of profit or loss. The Group has not designated any financial liability as at fair value through profit and loss.



#### Notes to the consolidated financial statements

for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

#### Loans and borrowings

After initial recognition, interest-bearing loans and borrowings are subsequently measured at amortised cost using the EIR method. Gains and losses are recognised in profit or loss when the liabilities are derecognized.

Amortised cost is calculated by taking into account any discount or premium on acquisition and fees or costs that are an integral part of the EIR. The EIR amortisation is included as finance costs in the statement of profit and loss.

This category generally applies to interest-bearing loans and borrowings.

Preference shares, which are mandatorily redeemable on a specific date, are classified as liabilities. The dividends on these preference shares are recognised in profit or loss as finance costs.

The fair value of the liability portion of an optionally convertible bonds is determined using a market interest rate for an equivalent non-convertible bonds. This amount is recorded as a liability on an amortised cost basis until extinguished on conversion or redemption of the bonds. The remainder of the proceeds is attributable to the equity portion of the compound instrument. This is recognised and included in shareholders' equity, net of income tax effects, and not subsequently remeasured.

Where the terms of a financial liability are renegotiated and the entity issues equity instruments to a creditor to extinguish all or part of the liability (debt for equity swap), a gain or loss is recognised in profit or loss, which is measured as the difference between the carrying amount of the financial liability and the fair value of the equity instruments issued.

#### Derecognition

A financial liability is derecognised when the obligation under the liability is discharged or cancelled or expires. When an existing financial liability is replaced by another from the same lender on substantially different terms, or the terms of an existing liability are substantially modified, such an exchange or modification is treated as the derecognition of the original liability and the recognition of a new liability. The difference in the respective carrying amounts is recognised in the statement of profit or loss.

#### Embedded derivatives

If the hybrid contract contains a host that is a financial asset within the scope Ind-AS 109, the Group does not separate embedded derivatives. Rather, it applies the classification requirements contained in Ind AS 109 to the entire hybrid contract. Derivatives embedded in all other host contracts are accounted for as separate derivatives and recorded at fair value if their economic characteristics and risks are not closely related to those of the host contracts and the host contracts are not held for trading or designated at fair value through profit or loss. These embedded derivatives are measured at fair value with changes in fair value recognised in profit or loss, unless designated as effective hedging instruments. Reassessment only occurs if there is either a change in the terms of the contract that significantly modifies the cash flows.

#### Offsetting of financial instruments

Financial assets and liabilities are offset and the net amount is reported in the balance sheet where there is a legally enforceable right to offset the recognised amounts and there is an intention to settle on a net basis or realise the asset and settle the liability simultaneously. The legally enforceable right must not be contingent on future events and must be enforceable in the normal course of business and in the event of default, insolvency or bankruptcy of the group or the counterparty.

#### Derivative financial instruments

The Group uses derivative financial instruments, such as foreign exchange forward contracts and interest rate swaps to manage its exposure to interest rate and foreign exchange risks. For contracts where hedge accounting is not followed, such derivative financial instruments are initially recognised at fair value on the date on which a derivative contract is entered into and are subsequently re-measured at fair value through profit or loss account. Derivatives are carried as financial assets when the fair value is positive and as financial liabilities when the fair value is negative.

#### 2.15 Employee benefits

#### (i) Short-term employee benefits

Short-term employee benefits are expensed as the related service is provided. A liability is recognised for the amount expected to be paid if the Group has a present legal or constructive obligation to pay this amount as a result of past service provided by the employee and the obligation can be estimated reliably.

#### (ii) Compensated absences

The liabilities for earned leave and sick leave are not expected to be settled wholly within 12 months after the end of the period in which the employees render the related service. They are therefore measured as the present value of expected future payments to be made in respect of services provided by employees up to the end of the reporting period using the projected unit credit method. The benefits are discounted using the market yields at the end of the reporting period that have terms approximating to the terms of the related obligation. Re-measurements as a result of experience adjustments and changes in actuarial assumptions are recognised in profit or loss.

#### (iii) Defined contribution plans

Obligations for contributions to defined contribution plans are expensed as the related service is provided. Prepaid contributions are recognised as an asset to the extent that a cash refund or a reduction in future payments is available.

#### Notes to the consolidated financial statements

for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

#### (iv) Defined benefit plans

The Group's net obligation in respect of defined benefit plans is calculated separately for each plan by estimating the amount of future benefit that employees have earned in the current and prior periods, discounting that amount and deducting the fair value of any plan assets.

The calculation of defined benefit obligations is performed annually by a qualified actuary using the projected unit credit method. When the calculation results in a potential asset for the Group, the recognised asset is limited to the present value of economic benefits available in the form of any future refunds from the plan or reductions in future contributions to the plan. To calculate the present value of economic benefits, consideration is given to any applicable minimum funding requirements.

Re-measurement of the net defined benefit liability, which comprise actuarial gains and losses, the return on plan assets (excluding interest) and the effect of the asset ceiling (if any, excluding interest), are recognised immediately in other comprehensive income (OCI). Net interest expense (income) on the net defined liability (assets) is computed by applying the discount rate, used to measure the net defined liability (asset), to the net defined liability (asset) at the start of the financial year after taking into account any changes as a result of contribution and benefit payments during the year. Net interest expense and other expenses related to defined benefit plans are recognised in profit or loss.

When the benefits of a plan are changed or when a plan is curtailed, the resulting change in benefit that relates to past service or the gain or loss on curtailment is recognised immediately in profit or loss. The Group recognises gains and losses on the settlement of a defined benefit plan when the settlement occurs.

#### (v) Other long-term employee benefits

The Group's net obligation in respect of long-term employee benefits is the amount of future benefit that employees have earned in return for their service in the current and prior periods. That benefit is discounted to determine its present value. Re-measurement are recognised in profit or loss in the period in which they arise.

#### 2.16 Provisions

Long-term provisions are determined by discounting the expected future cash flows specific to the liability. The unwinding of the discount is recognised as finance cost.

Provisions are measured at the present value of management's best estimate of the expenditure required to settle the present obligation at the end of the reporting period. The discount rate used to determine the present value is a pre-tax rate that reflects current market assessments of the time value of money and the risks specific to the liability. The increase in the provision due to the passage of time is recognised as interest expense.

#### **Onerous contracts**

A provision for onerous contracts is measured at the present value of the lower of the expected cost of terminating the contract and the expected net cost of continuing with the contract. Before a provision is established, the Group recognises any impairment loss on the assets associated with that contract.

#### 2.17 Earnings per share

Basic earnings per share is computed by dividing the profit / (loss) after tax by the weighted average number of equity shares outstanding during the year. The weighted average number of equity shares outstanding during the year is adjusted for the events for bonus issue, bonus element in a rights issue to existing shareholders, share split and reverse share split (consolidation of shares).

Diluted earnings per share is computed by dividing the profit / (loss) after tax as adjusted for dividend, interest and other charges to expense or income (net of any attributable taxes) relating to the dilutive potential equity shares, by the weighted average number of equity shares considered for deriving basic earnings per share and the weighted average number of equity shares which could have been issued on conversion of all dilutive potential equity shares

#### 2.18 Leases

Ministry of Corporate Affairs ("MCA") through Companies (Indian Accounting Standards) Amendment Rules, 2019 and Companies (Indian Accounting Standards) Second Amendment Rules, has notified Ind AS 116 Leases which replaces the existing lease standard, Ind AS 17 leases, and other interpretations. Ind AS 116 sets out the principles for the recognition, measurement, presentation and disclosure of leases for both lessees and lessors. It introduces a single, onbalance sheet lease accounting model for lessees.

All of the Group leases at 1 April 2019 were either cancellable or short term or had a remaining period of less than one year from that date. Accordingly, the transition to Ind AS 116 did not have any impact on the financial statements of the Company as at that date.



N	
-W	
K	4
	/

# Tata Realty and Infrastructure Limited Notes to the Consolidated financial statements as at 31 March 2020 (Currency: Indian rupees in lakhs)

Note 3(a) Property, plant and equipment

Particulars	Leasehold improvements	Freehold land	Buildings	Plant & Machinery	Furniture and fixtures	Electrical fittings	Computers	Office and Other Euipments	Motor vehicles	Total
Cost / Deemed Cost			1			1			· · · · · · · · · ·	
At 31 March 2018	8	2,616	41	263	487	120	446	501	257	4,739
Additions	467	1.1.1		8	40	3	30	110	14	672
Disposals	· · · · · ·		2.1	(0)	(29)		(2)	(20)	(44)	(95)
At 31 March 2019	475	2,616	41	271	498	123	474	591	227	5,316
Additions			9	86	2	1.1	34	43	25	199
On account of disposal of investments in subsidiaries			÷	(8)	(88)	(7)	(21)	(88)		(212)
Disposals / adjustments			- 21	2	(109)		(22)	(46)	(30)	(205)
At 31 March 2020	475	2,616	50	351	303	116	465	500	222	5,098
Depreciation / amortisation										
At 31 March 2018	ī	5 m	3	62	326	78	329	265	67	1,131
Charge for the year	31	-	5	31	58	12	69	78	30	314
Disposals / adjustments				(0)	(8)		(1)	(9)	(17)	(35)
At 31 March 2019	32	~	8	93	376	90	397	334	80	1,410
Charge for the year	109		5	30	22	12	50	63	30	321
On account of disposal of investments in subsidiaries	-		-	(2)	(33)	(1)	(14)	(75)	4.4	(125)
Disposals	10 million (1996)	-		0	(108)		(20)	(45)	(17)	(190)
At 31 March 2020	141		13	121	257	101	413	277	93	1,416
Net Block as at 31 March 2019	443	2,616	33	178	122	33	77	257	147	3,906
Net Block as at 31 March 2020	334	2,616	37	230	46	15	52	223	129	3,682

# Note 3(b) Depreciation and amortisation expense

Particulars	31 Mar 2020	31 Mar 2019
Property, plant and equipment	321	314
Investment Property	13,737	15,035
Right to use an asset	1,417	L. C.
Intangible assets	1,776	1,593
Sub-total	17,251	16,942
Less: Capitalised to Investment property under construction		(85)
	17,251	16,858

R

389

Notes to the Consolidated financial statements

as at 31 March 2020 (Currency: Indian rupees in lakhs)

Note 4 (a)

#### Investment Property

Particulars	Freehold land	Leasehold land	Buildings	Computers	Furniture and fixtures	Office equipments	Electrical fittings	Plant & Machinery	Total
Cost / Deemed Cost								-	
At 31 March 2018	20,688	1,41,279	2,35,321	129	490	1,173	11,905	34,664	4,45,649
Additions			7,628		35	91	57	1,623	9,434
Disposals	10 10 10 W		1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		1	(3)	(0)	(669)	(672
At 31 March 2019	20,688	1,41,279	2,42,949	129	525	1,261	11,962	35,618	4,54,411
Less reclassified to Right to use an assets	1	(1,41,279)		31	-				(1,41,279
(Refer note 4 (b))						1.1.1.1.1.1.1			for series
Additions	1,029		40,255		107	- 11-	5,864	7.417	54,683
On account of disposal of investments in subsidiaries	(19,065)		(25,776)	(128)	(506)	(114)	(1,513)	(8,249)	(55,351
Disposals	1					- 1			ter per s
At 31 March 2020	2,652		2,57,428	1	126	1,158	16,313	34,786	3,12,464
Depreciation / amortisation and impairment	1.1.1.5				1.1	11.1	1.11	100.00	
At 31 March 2018	6,191	13,753	40,410	120	302	152	4,061	12,996	77,985
Charge for the year*	1.1	1.427	9,171	8	38	127	1,137	3,127	15,035
Disposal	and the second second second					(2)	(0)	(607)	(609)
At 31 March 2019	6,191	15,180	49,581	128	340.	277	5,198	15,516	92,411
Less reclassified to Right to use an assets		(15,180)	(Arre)				Shees	1000	(15,180)
Refer note 4 (b))			1.00						
Impairment loss (Refer Note 50 (b))	1.		1,878	÷		15.14	1.4	1.00	1,878
Charge for the year *			9,658	~	5	811	1,157	2,799	13,737
On account of disposal of investments in subsidiaries	(5,724)		(10,091)	(128)	(340)	(47)	(1,033)	(4,843)	(22,206)
Disposal	÷						1		
At 31 March 2020	467		51,026	×.	5	348	5,322	13,472	70,640
Net Block as at 31 March 2019	14,497	1,26,099	1,93,368	1	185	984	6,764	20,102	3,62,000
Net Block as at 31 March 2020	2,185		2,06,402	t	121	810	10,991	21,314	2,41,824

\* Depreciation capitalised to Investment Property under Construction Rs. Nil (Previous year Rs, 85 lakhs).

i) Amounts recognised in profit & loss for Investment property		
Particulars	31 March	31 March
	2020	2019

	2020	2019	
Rental Income	46,332	44,168	
Direct operating expenses from property that generated rental income	23,475	25,228	
Direct operating expenses from property that did not generate rental income	975	883	

#### ii) Leasing arrangements

The Group has leased properties under non-cancellable operating leases in the capacity of a lessor. Refer Note 47 for future minimum lease payments in respect of these properties till the expiry of lock in period.

#### iii) Measurement of fair values

The fair value of investment property are determined after considering valuation by an independent valuer who holds a recognised and relevant professional qualification and experience in respect of the investment property being valued.

The fair value measurement for all of the investment property has been categorised as a Level 3 fair value based on the inputs to the valuation technique used.

#### iv) Valuation technique:

a. Valuation of some subject property has been done by Sales Comparison Method under Market Approach. A comparison is made for the purpose of valuation with similar properties that have recently been sold in the market and thus have a transaction price. The sales comparison approach is the preferred approach when sales data are available. Comparable properties are selected for similarity to the subject property considering attributes like age, size, share, quality of Comparable property, similarity a market value for the subject property is estimated from the adjusted sales price of the comparable properties.

b. The Group also has followed discounted cash flows technique for some property which considers the present value of net cash flows to be generated from the property, using risk-adjusted discount rates.

c. In case of a valuation of a large land parcel, where the development potential is realised over a period of time (i.e. time value of money comes into the picture) and also where there are no or few immediate similar properties (i.e. comparable) available for comparison, DCF method considering relevant potential developments of the project is used.

Investment property comprises a number of commercial properties that are leased to third parties. Each of the leases entered are normally for a period of 5 to 10 years. Subsequent renewals are negotiated with the lessee or as per the terms of initial agreement are automatically renewed. No contingent rents are charged.

#### Reconciliation of fair value

The following is the reconciliation in the fair values as of March 31, 2019 and March 31, 2020:

Particulars	Amount
Opening balance as of April 1, 2018	5,03,321
Additions	9,434
Fair value difference	2,286
Closing balance as of March 31, 2019	5,15,041
Additions	54,683
Deletion on account of disposal of investments in subsidiaries	(33,145)
Reclassified to Right to use an assets (Refer note 4 (b))	(1,26,099)
Fair value difference	360
Closing balance as of March 31, 2020	4,10,840

# Tata Realty and Infrastructure Limited Notes to the Consolidated financial statements as at 31 March 2020 (Currency: Indian rupees in lakts)

#### Note 4 (b) Right to use an asset

Particulars	Leasehold land	Office Premises	Total
Cost		1 million 1 million 1 million	· · · · · ·
At 31 March 2018	7	-	~
Additions		8	5
Disposals	1000000000	2	
At 31 March 2019			2
Add : reclassified from Investment property	1,41,279	~	1,41,279
Additions		151	151
Disposals		1	- Q
At 31 March 2020	1,41,279	151	1,41,43(
Amortisation and impairment			
At 31 March 2018			1.0
Charge for the year			-
At 31 March 2019	To be	×	1.1.101
Add : reclassified from Investment property	15,180	-	15,180
Charge for the year	1,413	-4	1,417
Disposals			
At 31 March 2020	16,593	4	16,597
Net Block as at 31 March 2019			1.1.1.1
Net Block as at 31 March 2020	1,24,686	147	1,24,833

Note: As at April 01, 2019 the Group has reclassified gross amount of Right to use asset of Rs. 1.41,280 lakhs with accumulated amortisation of Rs. 15,180 lakhs and the amount amortised for the year ended March 31, 2020 is Rs. 1,417 lakhs which has been charged to Statement of Profit & Loss Account (Refer Note No. 2.18).

# Note 4 (c) Goodwill on consolidation

Goodwill acquired on business combination is allocated, at acquisition to the cash generating units (CGUs) that are expected to benefit from that business combination. The carrying amount of Goodwill has been allocated as follows:

Particulars	31 March 2020	31 March 2019
Goodwill in respect of:		
TRIL Infopark Limited	19,053	19,053
TRIL Amritsar Projects Limited		1,506
(Net of impairment) (Refer Note No. 50 (a) and (b))		1.
TRIF Real Estate Development Limited		657
(Net of impairment) (Refer Note No. 50 (a) and (b))	1	
	19,053	21,216

#### Tata Realty and Infrastructure Limited Notes to the consolidated financial statements (Continued)

Notes to the consolidated financial statements (Continued)		
(Currency: Indian rupees in lakhs)	As at	As at
	31 March 2020	31 March 2019
Note 5	SI Waten 2020	51 March 2015
Capital work-in-progress		
and demarcation and reforestation	55	55
Project consultancy and technical charges	63	66
	118	121
Note 6		
nvestment property under construction Land	20,384	24,922
Direct expenses	60,289	1,16,546
	80,673	1,41,468
Note 7		
Intangible assets under development under Service Concession Arrangements		
Project Development Expenses	95,172	62,308
rofessional Fees	420	1,368
roject Management Fees		5,100
inance Costs (including unwinding of interest)	13,140	15,937
onstruction Cost to Contractor	7,485	74,996
inancial liability for premium payable recognised at fair value	30,278	62,155
Other Expenses	2,023	2,301
	1,48,518	2,24,165

Notes to the Consolidated financial statements as at 31 March 2020 (Currency: Indian rupees in lakhs)

# Note 8 Intangible assets

Particulars	Softwares	Service Concession Arrangements (refer foot note below)	Total
Cost / Deemed Cost			
At 31 March 2018	398	41,079	41,477
Additions	10	9,271	9,281
Disposals			-
At 31 March 2019	408	50,350	50,758
Additions	47	1,43,079	1,43,126
On account of disposal of investments in subsidiaries	(17)		(17
Disposals	(33)		(33
At 31 March 2020	405	1,93,429	1,93,834
Amortisation and impairment			
At 31 March 2018	316	3,424	3,740
Charge for the year	21	1,572	1,593
Disposals			-
At 31 March 2019	337	4,996	5,333
Impairment loss (Refer Note 50 (b))	× 1	9,436	9,436
Charge for the year	23	1,753	1,776
On account of disposal of investments in subsidiaries	(12)		(12
Disposals	(33)		(33)
At 31 March 2020	315	16,185	16,500
Net Block as at 31 March 2019	71	45,354	45,425
Net Block as at 31 March 2020	90	1,77,244	1,77,334

#### Service concession agreements

Name of entity	Description of the arrangement	Significant terms of the arrangement
International Infrabuild Pvr. Ltd.	As per the Concession Agreement (CA) the Company is required to build and operate Multi Level Cur Parking and collect Parking fees from Vehicles and User fees for Public Convenience Pacificus. The CA also specifies that the Company is required to built and operate of the total area of commercial infrastructure and an area of 5% i.e. 466 sq.m.shall be reserved for Auto Shownion/Auto Workshop.	Period of concession: 33 years (2017-2050) Investment grant from concession gramter: Nill Investment and renewal obligations : Nil Basis upon which re-pricing or re-negotiation is iderrained : Inflation Premium payable to granter :Rs. 2.66 urs plus escalation increase
Hampi Expressways Private Limited	Design, Build, Finance, Operate and Transfer (DBFOT) basis, augmentation of the existing road from km 299.00 to km 418.750 (approximately 12.018 km) on the Hospet-Chitradurga section of National Highway No.13 (New National Highway No.50)	Period of concession: 26 years (2016 - 2041) Remuneration Toll Fee Collection from Road Users Investment grant from concession grantor - Nil Investment and renewal obligations - Nil Basis upon which re-pricing or re-negotiation is determined - Inflation Premium paylet to grantor Rs. 18.15 crs tplus escalation increase
Dharanshala Ropeway Linuted	Agreement entered with The Department of Tourism & Civil Aviation, Fitmachal Pradesh ("Authority) 1 for the construction of Passenger Ropeway between Backide of Dharanstatal Bustand tipto Dalhi Lama Temple, Mcleodgany under DBFOT Model (Design, Build, Finance, Operate and Transfer Model), referred to as "Passenger Ropeway Eacility", The Passenger Ropeway facility is under construction as on date. The Civil & Land Site Developments works have been started during the year.	Period of concession: 40 Years from the appointment date (2020-2060) Remuneration: Collection on Ropeway tickets Investment paint from concession grantor - NIL Investment and renewal obligations : Nil Basis upon which re-pricing or re-negotiation is determined. AS per concession agreement Premium payable to grantor:Rs. 1 cr plus escalation Increase
Manali Ropeways Private Limited	Concession Agreement entered on dated 21 October 2015, with The Department of Toursan & C(n) Avatton, Humachal Pradesh ("Authority") for the construction of Passenger Ropeway between Palehan and Robutang at Manali under BOT Model (Built, Operate and Transfer Model). The construction has not started as on date and the company is in process of obtaining various government approvals for commercing the construction of ropeway.	Period of concession: 40 Years from the appointment date Remaneration: Collection on Ropeway tickets Investment grant from concession granuor (NIL Investment and renewal obligations: Nil Basis upon which re-pricing or re-negotiation is determined. As per concessio agreement Premium payable to grantor :Rs.1.50 Cr plus escalation increase
Uchit Expressway Private Lid.	Six Laning of Chintorgarh-Udaipur Section of NH-76 from Design Chainage Km 214.870 to Km 308.870 in the State of Rajasthan (Length 93.500 Km) on DBFOT (Toil) Mode under NHDP Phase V (Package-III)	Period of concession: 29 Years (2017-2046) Remuneration: Toll Fee Collection from Road Users Investinger grant from concession grantor: NIL Investment and renewal obligations: NIL Basis upon which re-pricing or re-negotiation is determined. NA Premium payable to grantor: Rs. 25 crs plus escalation increase
Durg Slävnath Expressways Private Limited	Toll collections from the users and operation and maintenance of total stretch of 18.5 kms at NH-6, Durg Bypass as agreed between the Company and National Highways Authority of India (NHAI) on Build. Operate and Transfer [BOT] basis	<ol> <li>Period of Concession: 32 Years and 6 months (2007-2030)</li> <li>Remaneration: Toll Fee Collection from Road Users</li> <li>Investment Grant from Concession Granter – NIL.</li> <li>H Basis upon which te-preting or te-negotiation is determined – Toll Tariff. wi be revised (st April of every year considering WPI 5) Premaum payable to Granter – NIL.</li> </ol>
Matheran Ropeway Private Lumited	Development of a Ropeway project in Matheran, Maharashtra of approx. 4500 Meters connecting Blattavali Village to Madhavji Point at Matheran, on Design, Build, Finance, Operate and Transfer (DBFOT) basis. The Project would have all elements of sustainability and would be noise free, electricity driven, Eco-friendly and non-polluting. It is estimated that the Project would save ~ 1000 tome of carbon dioxide emissions.	The Company has signed an Agreement with The Matheran Giristhan Nagar Parishad, Matheran on 20th May 2003 for a period of 99 years. The Company has agreed to pay compensation equal to 5% of its net profits of each year to Matheran Giristhan Nagar Parishad, Matheran.
nne Solapur Expressway Pvt Lid	Highways Authority to India also the graniter') to construct a toll fughway between Pune and Solapur. The construction of the toll road started thereafter and it was partially completed on 3 February 2013 and was fully completed and available for use on 31 January 2015.	Period of concession: 21 Years (2009-2030) Remmeration - Toll Fee Collection from Road Users Investment grant from concession grantor - NIL Investment and renewal obligations - NIL Basis upon which res-pricing or re-negotiation is determined - NA Premium payable to grantor - NIL

RI

Notes to the consolidated financial statements (Continued)

(Currency: Indian rupees in lakhs)		
	As at	As at
	31 March 2020	31 March 2019
Note 9		
Investment accounted using Equity Method (at carrying cost):		
Unquoted equity shares, fully paid-up		
2,386,711 (2019: 2,386,711) equity shares of Pune Solapur	2,248	2,170
Expressways Private Limited, of Rs.10 each fully paid up.		
740,000 (2019: 740,000) equity shares of INR 10 each in TRIL IT4	1.4	·
Private Limited (Formerly known as Albrecht Builder Private Limited)		
##		
30,000 (2019: 10,000) equity shares of A & T Road Construction	1	15
Management and Operation Private Limited, of Rs.10 each fully paid		
up.##	22.002	6.6.
3,256 (2019:3,256) equity shares of INR 100 each in Industrial	21,950	21,910
Minerals and Chemicals Company Private Limited.	20.224	20.425
19,987,400 (2019: 19,987,400) equity shares of INR 10 each in Mikado Realtors Pvt, Ltd.	30,335	30,436
44,407,400 (2019: Nil) equity shares of INR 10 each in Pune IT City	4,226	
Metro Rail Limited	4,220	
(A)	58,760	54,516
Investment in debentures of joint ventures 814 (2019: 814) Quoted Redeemable Non-convertible Debentures of	0.140	0.140
INR 10 Lacs each in TRIL IT4 Private Limited	8,140	8,140
11,787,460 (2019: 10,473,960) Unquoted Compulsorily Convertible	11,787	10,474
Debentures of INR 100 each In Industrial Minerals and Chemicals Company Private Limited		
Company Private Eminted		
(B)	19,927	18,614
(A + B) -	78,687	73,130
## Unrecognised share of Losses in Joint Venture	- decare	ALL STATES
TRIL IT4 Private Limited	(20,919)	(16,815)
A & T Road Construction Management and Operation Private Limited		(1)
Note 10		(1)
Other Investments (non-current)		
Investment in Unquoted, Fully paid-up, Equity instruments of Other Companies (Fair Valued Through Profit and Loss) (Refer Foot note)		
117600 (2019: 117600) equity shares of Vagarai Windfarms Limited	12	12
18,90,000 (2019: Nil) equity Shares o of Perinyx Neep Private Limited	189	
119,187 (2019: 119,187) equity shares of Echanda Urja Private	11	11
Limited		
	212	23

Note:

The above investments represents investment in equity shares of the above mentioned entities pursuant to the requirement under the Electricity Act, in connection with the power purchase arrangement that the group has in place with these parties. As per the terms of these investments, the group is not entitled to any other returns or benefits and will be entitled to receive the amount invested equivalent to the face value of the equity shares upon expiry of such agreements.

#### Note 11

Other financial assets (non-current)		
Unsecured, considered good		
Fixed deposit having maturity more than 12 months under lien	5,977	500
Lease rental receivable	243	193
Recoverable from Punjab Urban Development Authority		517
Security Deposits	984	467
	7,204	1,677

A

The set of		
	As at	As at
and the	31 March 2020	31 March 2019
Note 12 Deferred tax assets (Net)		
- investments in associates and joint ventures	2,274	2,487
- other financial assets at fair value through profit and loss	757	1,340
<ul> <li>On other provisions (including provision for employee benefits and other amounts allowable on a payment basis)</li> </ul>	34	239
- on account of PPE and Intangibles	4,397	791
- derivatives (including call put options)	2,130	1,599
and an	9,592	6,456
Note 13		
Non Current tax assets (Net)		
Advance tax	18,787	15,845
Less: Provision for tax	(6,428)	(4,956
	12,359	10,889
Note 14 Other non-current assets		
Other non-current assets Unsecured, considered good		
Capital advances Balance with Government Authorities	8,634	16,676
Security Deposit - Others	184 117	112
Prepaid expenses	620	54
Lease equalisation reserve	11,599	9,310
Unbilled Revenue	157	7,510
Other non-current assets		115
	21,311	26,275
Note 15		
Inventories		
(valued at cost or net realisable value whichever is less)		
Bought out construction materials	52	52
Consumables stores and spares		77
Work-in-progress Finished goods (Refer foot note)	2,630	15,341
inisited goods (Refer tool note)		
	30,905	28,248
	33,587	28,248 43,718
Foot note:	33,587	
	33,587	
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16	33,587	
Foot note: Represent value of residential units. Net off NRV provision on inventories	33,587	
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16 Irade and other receivables	33,587	
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16 Irade and other receivables Dutstanding for a period exceeding six months:	33,587	43,718
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16 Frade and other receivables Dutstanding for a period exceeding six months: From others	33,587	
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16 Frade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months:	33,587 INR 1,420.78 Lakhs (2019: NIL) 24	<u>43,718</u> 43
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16 Frade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003	43,718 43 2,120
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From related parties	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144	43,718 43 2,120 958
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16 Frade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306)	43,718 43 2,120 958 (275)
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From others From related parties e.ess : Provision for credit impaired Trade Receivables	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144	43,718 43 2,120 958
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From related parties	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306) 2,865	43,718 43 2,120 958 (275) 2,846
Foot note: Represent value of residential units. Net off NRV provision on inventories Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From others From related parties ess : Provision for credit impaired Trade Receivables Break-up for security details:	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306)	43,718 43 2,120 958 (275) 2,846 2,061
Foot note: Represent value of residential units. Net off NRV provision on inventories i Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From related parties ess : Provision for credit impaired Trade Receivables Break-up for security details: ecured, considered good	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306) 2,865 1,675	43,718 43 2,120 958 (275) 2,846
Foot note: Represent value of residential units. Net off NRV provision on inventories i Note 16 Frade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From related parties .ess : Provision for credit impaired Trade Receivables Greak-up for security details: ecured, considered good Insecured, considered good	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306) 2,865 1,675 1,190	43,718 43 2,120 958 (275) 2,846 2,061 785
Foot note: Represent value of residential units. Net off NRV provision on inventories i Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From related parties Less : Provision for credit impaired Trade Receivables <b>Break-up for security details:</b> Recured, considered good Dissecured, considered good Dissecured, considered doubtful Ulowance for doubtful debts	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306) 2,865 1,675 1,190 306	43,718 43 2,120 958 (275) 2,846 2,061 785 275
Foot note: Represent value of residential units. Net off NRV provision on inventories i Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From related parties ess : Provision for credit impaired Trade Receivables Break-up for security details: ecured, considered good Insecured, considered good Insecured, considered doubtful Ulowance for doubtful debts Foote 17	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306) 2,865 1,675 1,190 306 (306)	43,718 43 2,120 958 (275) 2,846 2,061 785 2,75 (275)
Foot note: Represent value of residential units. Net off NRV provision on inventories i Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Justanding for a period less than six months: From others From related parties ess : Provision for credit impaired Trade Receivables Break-up for security details: ecured, considered good Insecured, considered good Insecured, considered good Insecured, considered doubtful Llowance for doubtful debts Fote 17 Eash and cash equivalents	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306) 2,865 1,675 1,190 306 (306) 2,865	43,718 43 2,120 958 (275) 2,846 2,061 785 2,75 (275)
Foot note: Represent value of residential units. Net off NRV provision on inventories i Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From related parties .ess : Provision for credit impaired Trade Receivables Break-up for security details: ecured, considered good Insecured, considered good Insecured, considered good Insecured, considered doubtful Illowance for doubtful debts Fote 17 Eash and cash equivalents ash on hand	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306) 2,865 1,675 1,190 306 (306)	43,718 43 2,120 958 (275) 2,846 2,061 785 2,75 (275)
Foot note: Represent value of residential units. Net off NRV provision on inventories I Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From related parties ess : Provision for credit impaired Trade Receivables Break-up for security details: ecured, considered good Insecured, considered good Insecured, considered doubtful Allowance for doubtful debts Note 17 ash and cash equivalents ash on hand alance with banks	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306) 2,865 1,675 1,190 306 (306) 2,865 30	43,718 43 2,120 958 (275) 2,846 2,061 785 2,75 (275) 2,846 93
Foot note: Represent value of residential units. Net off NRV provision on inventories i Note 16 Trade and other receivables Dutstanding for a period exceeding six months: From others Dutstanding for a period less than six months: From others From related parties .ess : Provision for credit impaired Trade Receivables Break-up for security details: ecured, considered good Insecured, considered good Insecured, considered good Insecured, considered doubtful Illowance for doubtful debts Fote 17 Eash and cash equivalents ash on hand	33,587 INR 1,420.78 Lakhs (2019: NIL) 24 2,003 1,144 (306) 2,865 1,675 1,190 306 (306) 2,865	43,718 43 2,120 958 (275) 2,846 2,061 785 275 (275) 2,846

Note 18 Bank balances other than cash and cash equivalents

Term deposit with original maturity less than twelve months under lien

A

219

219

1,579

1,579

Notes to the consolidated financial statements (Continued) (Currency: Indian runees in lakhs)

(Currency: Indian rupees in lakhs)		
	As at	As at
	31 March 2020	31 March 2019
Note 19		
Other Investments (current)		
Investment in mutual funds, fair valued through Profit and Loss		
Units Held 491,528.95 (2019 : 10,576.43) of Face value of Rs 1,000	15,395	311
(NAV - Rs 3,131.99 (2019 : 2,944.44)) each in TATA money market		
fund -Growth Option		
Units Held 3,289,874.36 [2019: Nil] of Face value of Rs 10 (NAV -	10,513	
Rs 319.56) each in ABSL Liquid Fund - Growth		
Units Held 385,284.66 [2019: Nil] of Face value of Rs 10 (NAV - Rs	15,052	1 m
3,906.61) each in HDFC Liquid Fund - Growth Option		
Units Held 10,286,570.04 [2019: Nil] of Face value of Rs 10 (NAV -	30,220	
Rs 293.78) each in ICICI Prudential Liquid Fund - Growth Option	50,220	
Units Held Nil [2019: 28,831.67] of Face value of Rs 1,000 each		804
(NAV - Rs. 2,788.16) in Franklin India Treasury Management Account		001
Super Institutional Plan - Growth Option		
Units Held Nil [2019: 10,19,407.79] of Face value of Rs 10 each		268
(NAV - Rs. 26.27) in Franklin India Ultra Short Bond - Super	-	200
Institutional Plan - Growth Option		
	71,180	1,383
	/1,100	1,505
Note 20		
Short-term loans and advances		
Unsecured, considered good		
Advances given	1983	2,000
Security deposits	3	38
Advances recoverable from Related Parties	579	683
Other loans and advances	807	333
	1,389	3,054
Note 21		
Other financial assets (current)		
Unsecured, considered good		
Interest accrued on deposits	198	115
Claims Receivable from NHAI (utility shifting)	1,232	674
Inter-corporate Deposits given to Related Parties	2,510	
Unbilled revenue	438	537
Foreign Exchange Forward Contract Receivable	103	
Security deposits	775	263
Others current financial asset	247	392
	5,503	1,981
Note 22		
Other current assets		
Unsecured, considered good		
Advance to vendors	184	681
Balances with Government and other authorities	3,107	1.463
Prepaid expenses	789	805
Lease equalisation reserve	1,079	1,360
Claim Receivable from NHAI (Refer foot note below)	4,647	1,048
Others	355	31
	10,161	5,388
	10,101	5,000

Foot Note:

Claim Receivable from NHAI pertains to claims towards change in law as per clause 41.1 of Service Concession Agreement entered with NHAI. These are back to back claims made by EPC Contractor as per clause 24.2 of EPC agreement and the same are payable to the EPC Contractor only to the extent such claims are recognised by NHAI as change in law and payments are released. To the extent such claims are not accepted by NHAI the same shall accordingly not be payable to EPC contractor and accordingly corresponding liability shall reduce.
### Notes to the Consolidated financial statements

### as at 31 March 2020

(Currency: Indian rupees in lakhs)

### Note 23

Equity share capital Authorised share capital	31 March 2020	31 March 2019
8,00,00,000 (2019: 3,00,00,00,000) equity shares of INR 10 each	8,00,000	3,00,000
Issued, subscribed and paid-up 1,01,73,07,692 (2019: 1,01,73,07,692) equity shares of INR 10 each	1.01.731	1,01,731
	1,01,731	1,01,731

### a. Reconciliation of the shares outstanding at the beginning and at the end of the year

Equity shares	31 March 2	31 March 2020		
	Number of shares	Amount	Number of shares	Amount
At beginning of the year	1,01,73,07,692	1,01,731	1,01,73,07,692	1,01,731
Issued during the year	1 Provide 1 1			1.
Outstanding at the end of the year	1,01,73,07,692	1,01,731	1,01,73,07,692	1,01,731

### b. Terms and rights attached to the equity share

The Company has only one class of equity shares having par value of INR 10 per share. Each holder of equity shares is entitled to one vote per share. The Company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject to the approval of shareholders in the ensuing Annual General Meeting. In the event of liquidation of the Company, the holders of the equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in proportion to the number of equity shares held by the shareholders.

### c. Shares of the company held by the Holding company

Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited

31 March 2	2020	31 March 2	2019
Number of shares	Amount	Number of shares	Amount
1,01,73,07,692	1,01,731	1,01,73,07,692	1,01,731

### d. Details of shareholding more than 5% in the company

Equity shares of INR 10 each, fully paid-up Tata Sons Private Limited

31 March	2020	31 March	2019
Number of shares	% Holding	Number of shares	% Holding
1,01,73,07,692	100%	1,01,73,07,692	100%

	(Currency: Indian rupees in lakhs)		
		As at	As at
		31 March 2020	31 March 2019
1	Note 24		
	Other equity		
1	Share application money pending allotment	1,20,000	
1	Reserves and surplus		
1	Securities premium reserve	15,769	15,769
(	Capital reserve	4,911	4,911
J	Retained earnings	(1,35,799)	(99,543)
	Other comprehensive income		
1	Defined benefit plan adjustment	29	27
		4,910	(78,836)
5	Share application money pending allotment		
	Dpening balance	-	
	Received during the year	1,20,000	
		1,20,000	
I	Retained earnings		
	Dpening balance	(99,543)	(74,541)
	ransition adjustment due to application of IND AS 115		(1,304)
	Refer Note 51 (a))		
N	let loss for the year attributable to owners	(36,256)	(23,693)
C	onsolidation adjustment		(5)
		(1,35,799)	(99,543)
(	Other comprehensive income - Defined benefit plan adjustment		
C	Dening balance	27	(14)
F	emeasurements of defined benefit liability	3	55
I	ncome tax relating to items that will not be reclassified to profit or loss	(1)	(14)
		29	27
N	on-controlling interests		
	pening balance	6,617	6,826
	oss for the year	(233)	(222)
	ssue of new shares / CCD in subsidiaries		13
		6,384	6,617

Notes to the consolidated financial statements (Continued)

Nature and purpose of the reserves: Securities premium reserve

0.01% Compulsorily convertible debentures were compulsorily convertible into equity shares by 25 August 2016 or before at the option of investor. During the Financial year 2016-17 (on 24 August 2016), these debentures were converted into 192,307,692 Equity shares of INR 10 each at a premium of INR 3 each.

### Capital reserve

Capital reserve was created to record excess of net assets taken over pursuant to scheme of merger sanctioned by the Bombay High Court in the year 2015-16 between Tata Realty and Infrastructure Limited, Mara Builder Private Limited and TRIF Real Estate and Development Limited.

### Note 25

Borrowings (Long-term) (refer footnote below)

From Banks and Financial Institutions	12,929	
Non convertible debentures	1,39,419	1,14,792
Unsecured		
From Others	19,045	24,910
From Banks and Financial Institutions	4,25,356	4,04,786
Secured		

Note:

Name of the Banks / Financial Institutions	Amount	Terms of Repayment	Security	Rate of Interest
NDFC I.d	20,291	Term Loan from HDPC Ldd is repayable in a single instalment i.e 14th July, 2020.	(i) Land including constructions thereon, including present and future. (ii) All the right, title and interest of the borrower/mortgagor in the secured receivables (including future receivables for multisum insthement/premises that have been sold/acased/licensed/etc.) from 'in respect of the Secured Property. (iii) All the right, title and interest of the borrower/mortgagor in the designated (No. Lien) aAccount secured Property. (iv) Sponsor intervestion of the borrower/mortgagor in the designated (No. Lien) aAccount secures account whether established or to be established with HDFC Bank Limited, any other bank accounts of the borrower / mortgagor. (iv) Sponsor intervoably and unconditionally undertakes to continue to legally and beneficially hold and retain 100% Stares (one hundred percent) of the Shares of the Borrower throughout the tenor of the Facility if the Final Settlement Date. Provided However the Sponsor may dispose of Shares capavalent to 49% of the Shares with prior infimitation to the Lender Until the Final Settlement Date, there shall be no change in the management control of the Borrower The Sponsor hereby agrees that in the event the each flow in the escrow account is not sufficient to service the facility, it shall find the escrow account.	spread of 110 bps, presently being 10% p.a. psyable at monthly rests.
HDFC Lat	26,451	months from date of first disbursement i.e 13th September, 2021.	(i) Land including constructions thereon, including present and future. (ii) Land including that interests of the borrower/mortgager in the secured receivables (including future receivables from flatsunits/hememet/premises that have been sold/leased/licensed/etc.) from / in respect of the secured property. (iii) All the right, title and interest of the borrower/mortgager in the designated (No. Lien) account / secret vaccount whether established or to be established with IDPC Bank Linuited, any other bank accounts of the borrower / mortgagor. (iv) Sponsor support agreement by Tata Realty & Infrastructure Lid. (Holding Company) :*- • The Sponsor inrevocably and unconditionally undertakes to continue to legally and beneficially hold and realin 100% Shares (one bundle precent) of the Shares of the Borrower throughout the tenor of the facility till the final settlement date. • Provided however the sponsor may dispose of Shares equivalent to 49% of the Shares with prior initination to the leader. • Unit the final settlement date in the event the cash flow in the eastrow account is not sufficient to service the facility, its alull fund the eserve account.	
IDFC LM	4,487	months from date of first disbursement i.e 13th March, 2023	(i) Land including constructions thereon, including present and future. (ii) All the right, title and interest of the borrower/mortgagor in the secured receivables (including future receivables from flas/mit/stemener/premises that have been sold/faesed/icensed etc.) from / in respect of the secured property. (iii) All the right, title and interest of the borrower/mortgagor in the designated (No. Lien) account / secrow account whether established or to be established with HDFC Bank Limited, any other bank accounts of the borrower /mortgagor. (iv) Sponsor support agreement by Tata Realty & Infrastructure Ld. (Holding Company) '- The sponsor inrevocably and unconditionally undertakes to continue to legally and beneficially hold and tettini 109% Shares (one hundred percent) of the Shares of the borrower throughout the tenor of the facility till the final settlement date. Provided Hhowever the sponsor may dispose of shares equivalent to 49% of the shares with prior minimation to the lender. Unlift be Final settlement date, there shall be no change in the management control of the borrower. The Sponsor hereby agrees that in the event the cash flow in the serow account is not sufficient to service the facility, its hall find the escrew account.	



(currency:	mutan rupees in takits)
Note.	

Name of the Banks / Financial Institutions	Amount	Terms of Repayment	Security	Rate of Interest
Bank of Maharashira	3,636		(i) First charge by way of mortgage on immovable project assets. (ii) First charge by way of hypothecation of all the movable assets of the company present and future. (iii) First charge on the Project's book doels, operating cash (hows, receivables, commissions, revenues of whatsoever nature and wherever arising, present and future, intangibles, goodwill, uncalled capital present and future. (iv) First charge on projects escrow account, current account and the amounts therein.	Rate of interest will be MCLR +1.25% p.a. i.e. presently the rate of interest is 10% p.a. (approx). Interest payment frequency will be monthly
Union Bank of India (TL)	33,473	Repayable in quarterly instalments commencing from June, 2022 and terminating on March, 2040.	(a)First charge on entire immovable properties of the respective group company, both present and future, sive and except project assets. (b) First charge on all tangible movable assets of the respective group company including movable plant and machinery, machinery sparse and tools and accessories, furniture, fixtners, vehicles and all other movable assets, both present and future, if any, save and except project assets. (c) First charge over all accounts of the respective group company including the escrow account, sub-accounts (or any account is nubstitution thereof) that may be opened in accordance with this agreement and the memorandum of opening procedure, or any of the other project documents and all funds deposited therein, from time to time, all receivable and permitted investments or other securities.	(approx.)
State Bank of India		from June, 2022 and terminating on March, 2040.	(a)First charge on entire immovable properties of the Company, both present and future, save and except project assets. (b) First charge on all tangible movable assets of the Company including movable plant and machinery, machinery sparses and tools and accept Project Assets. (c) First charge over all accounts of the Company including the escrow account, sub-accounts (or any account in substitution thereof) that may be opened in accordance with this agreement and therein, form lime to time, all receivable and permitted investments or other securities (additional discretion) and therein form lime to time, all receivable and permitted investments or other securities (d) First charge on all itangibles assets of the Company including but not limited to goodwill, rights, undertakings of the Company including, trues, except the project assets provided that all receivables arising therefrom shall be deposited into escrow account and a charge on the same adaltable subject to the exet permissible as per the priority specified in the Aricle 31 of the Concession Agreement. Further a charge on ancilled capital as set in above, shall subject however, to the provisions of Articles 5.3, 7.1(k) and 31 of the concession Agreement. (c) an assignment by way of security in: (i) all the rights, title, interest, benefits, claims and demands whatsoever of the hortower in, to or under the project documents; (ii) all the rights, title, interest, benefits, claims and demands whatsoever, of the portower not perioder parameters, each perioder parameters, and inducation parameters, and inquidated damages and performance bond provided by any party to the project documents;	Rate of interest is 9,50% p.a. (approx.)



Tata Realty and infrastructure Limited
Notes to the consolidated financial statements (Continued)

as at 31 March 2020 (Currency- Indian rupees in lakhs)

Name of the Banks / Financial Institutions	Amount	Terms of Repayment	Security	Rate of Interest
State Bank of India	31,271	quarterly instalments commencing from 26th	(i) First charge on entire inumovable properties of the Company, if any, save and except project Assets (as	Average rate of interest of 9.15% -
Oriental Bank of Commerce	8,446		defined in the Concession Agreement).	9.25% p.a.
Canara Bank	17,225		(ii) First charge on entire movable assets of the Company, both present and future, if any, save and except	
Union Bank	25,841	2035	project assets (as defined in the Concession Agreement).	
Corporation Baok	8.617		(iii) First charge on the servov account/trust and retention account (TRA), Dept. Service Reserve Account (OBRA) and any other reserves and other boak accounts of the Company, wherever miniatined, provided further that the first charge as set out herein shall anse only after the proceeds or realization thereof. if any, have been received in the escrow account designated for the Project and thereafter shall only be to the extent of waterfall mechanism (c. priorities for the withdrawal) of payments) as provided in the Concession Agreement and Escrow Agreement. (iv) First charge on all intangibles including but not limited to goodwill and uncalled capital, present and fluture, encluding the project assets and a charge on the uncalled capital subject however, to the provisions of Concession Agreement. (v) Assignment of security mitrest on the Borrowey's rights, title and interest to the extent overed by and in accounting with the substitution Agreement. (vi) First charge on the insurance proceeds deposited into the escrow account. (vii) The above security will runk pari-passa with other lenders participating in the facility.	
State Bank of India	79,000	This Term Loan from State Bank of India is repayable in several instalment till 30th September, 2029 as per the repayment schedale agreed	The loan is secured by a first ranking pari passa charge on the entire current assets, current accounts, escriw accounts, project documents and such other assets of the Company. Collateral security over the investment properties owned by the Company; assignment or hypothecation of lease rent as per the Lease deeds, lease deeds and the amenities agreement and other leases entered by the Company in the investment property.	state bank of India is lyear SB MCLR rate plus 0.15%, during the
State Bank of India	84.212	This Term Loan from State Bank Of India is repayable in several instalment till 30th September, 2029 as per the repayment schedule agreed	The loan is secured by a first ranking part passu charge on the entire current assets, current accounts, escrow accounts, project documents and such other assets of the Company. Collateral security over the investment properties owned by the Company, assignment or hypothecation of lease tent as per the Lease deeds, lease deeds and the amenities agreement and other leases entered by the Company in the investment property.	state bank of India is Iyear SB MCLR rate plus 0.15%, during the
HDPC LM	59,864		Assignment/hypothecation of lease rent (present & future) as per leave and license agreement and amenities agreement entered into by Company and the various lessees from the area mortgaged hereinafter referred to as the said receivables Mortgage of approximately 23.7.4 Acres of SEZ property along with leasehold rights of project "Ramanujan IT City" The above securities shall rank pari pasu with State Bank of India and Standard Chartered Bank Debenture holders as on 31-03-2018 in the ratio of the loan outstanding ie Rs 850 Cr and Rs. 900 Cr	HDFC Ltd is HDFC CPLR less spread of 930 bps, during the year i was changed twice once to 9.35% p.a. in May 2019 and then it November 2019 to 9.10% p.a.
HDPC Lid	9,330	Ltd is repayable in single instalment on 30th October, 2020.	Exclusive Mortgage of approximately 1.38 Acres of leasehold land with construction thereon. The above securities shall rank pari pasa with State Bank of India and Standard Chartered Bank Debenture bolders as on 31-03-2018. Assignment of all receivables from the project being developed including receivables from service apartments / convention centre / facilities.	was changed once to 10.25% p.a. in May 2019 and is presently still being
HDFC Ltd	3,450	This Working Capital Loan from HDFC Bank is repayable in 11 months ie in February, 2021	This is a working capital loan and is Unsecured	The rate of interest on this working capital loan is 8.00% p.a.



Name of the Banks / Financial Institutions	Amount	Torres of Barran and	P	Bata of Laborat
File production and the state of the state of the		Terms of Repayment	Security	Rate of Interest
RBL Bank	2,981	monthly instalment after 7 months from date of firs	It Charge on the project building & premises financed along with all the movable fixed asset and other eurrent assets. 2. Charge on the scheduled receivables and all the insurance proceeds.	The rate of interest on term loan from RBL Bank Ltd is 3M RBL MCLF presently being 9.25% p.a. payable a monthly rests
Financial Institution				
Adiiya Birla Finance Ltd	18,739	Repayable in quarterly instalments commencing from June, 2022 and terminating on March, 2040	(a).First charge on entire immovable properties of the Company, both present and future, save and except project Assets. (b) First charge on all tangible movable assets of the Company including movable plant and machinery, machinery sparse and tools and accessories, furniture, futures, vehicles and all other movable assets, both present and future, if any, save and except Project Assets. (c) First charge over all accounts of the Company including the eserow account, sub-accounts (or any account in Sathitution theroo) that may be opened in accountace with this agreement and this agreement, from time to time, all receivable and permitted anvestments or other securities. (a) First charge on all itangibles assets of the Company including the result is agreement and the memorandum of openating procedure, or any of the other Project Documents and all funds deposited therein, from time to time, all receivable and permitted answtaments or other securities. (a) First charge on all itangibles assets of the Company including but not limited to goodwill, rights, undertakings of the Company and uncalled capital, present and future, except the project assets provided that all receivables arising therefrom shall be deposited in the Article 31 of the Concession Agreement. (c) all the rights, title, anterest, benefits, claims and demands whatsoever of the borrower in, to or under the project documents. (c) all the rights, title, antierest, benefits, claims and demands whatsoever, of the Company in, to or under the project documents. (ii) all the rights, title, interest, benefits, claims and demands whatsoever, of the Company in, to or under the right, title, interest, benefits, claims and demands whatsoever, of the Company in, to or under all insurance contracts.	(approx.)
	12,929	Repayment is fastalment up to 30 September, 2022	Corpornte guarannee provided by TRIL Roads Private Limited, holding company. Also, non-disposal undertaking given by TRIL Roads Private Limited, with respect to 51% share of Durg Shivnath Expressways Private Limited.	
xis Bank Lid		The repayment shall be as per repayment schedule, commencing from FY 2020-21 to 2027-28	revenues both present and future;	The term loan shall carry a fixed rat of interest of 8.45% p.a. wit monthly interest payment

R

Note:

Name of the Banks / Financial Institutions	Amount	Terms of Repayment	Connector	
			funnac	Kate of Interest
Non Convertible Debentures				
Nan Convertible Debentures	1,79,500	1,79,500 The Debentures shall be redeemed From 20th Nil April, 2020 to 18th November, 2022	au Nu	The rate of interest on debentures is 8.25% to 9.50% p.a., and coupon to be paid annually/Maturity.
A-LATAL	6,73,211			
HDFC Bank Ltd	4,400	4.400 This Working Capital Loan from HDFC Bank repayable in 11 months ie in August, 2020	Capital Loan from HDFC Bank is This is a working capital loan and is unsecured months ac numer, 2020	The rate of interest on this working
HDFC Bank Ltd	5,000	5,000 This Working Capital Loan from HDFC Bank repayable in 11 months ie in February. 2021.	Capital Loan from HDFC Bank is This is a working capital loan and is unsecured months ie in February. 2021.	The rate of interest on this working
Union Bank of India	611	611 Over Draft	Unsecured	Rate of interest is 9.50% p.a.
Short Term Loan from Bank				(approx.)
Deutsche Bank	20,000		Unsecured	8 60% D a
Commercial Paper				·····
Franklin India Liquid Fund	17,500		Unsecured	8.50% p.a.
SBI Mutual Fund - SBI Savings Fund	15,000		Unsecured	8.55% 0.3
Franklin India Low Duration Fund	4,000		Unsecured	8.70% p.a.
Franklin India Ultra Short Bond Fund	3,500		Unsecured	8 70% n a
Mahindra Liquid Fund	1,000		Unsecured	8 70% n.a
Mahindra Ultra Short Term Yojana	500	500 Repayment on 15th September, 2020	Unsecured	8 70% n a
Franklin India Low Duration Fund	21,500		Unsecured	8 60% n.a
SBI Magnum Low Duration Fund	5,000	5,000 Repayment on 28th September, 2020	Unsecured	8 25% n.a
Reltance Capital Trustee Co.Lid a/c Nippon India Ultra Short Duration Fund	10,000	10,000 Repayment on 28th September, 2020	Unsecured	8.25% p.a.
Loans from Directors	100			
TOTAL-B	1108,111			
Less - EIR on Borrowings (IND AS adjustments)- (C)	(3,370)			
TOTAL - (A + B + C)	136.77.7			

 Particulars
 Annount

 Long term borrowings
 5.9749
 Refer Note No. 25

 Current maturity of long term borrowings
 75,788
 Refer Note No. 32

 Short term borrowings
 1,05,414
 Refer Note No. 30

 TOTAL
 7,77,951
 30

Tata Realty and Infrastructure Limited Notes to the consolidated financial statements (Continued) (Currency: Indian rupees in lakhs)

	As at	As at
	31 March 2020	31 March 2019
Note 26		
Other financial liabilities (non-current)		
Financial liability for premium payment at fair value	69,016	69,110
Security deposits from customers	20,787	18,210
Interest accrued but not due on borrowings	5,933	11,439
Derivatives not designated as hedges	277 D	
- Forwards		425
- Put option	1,391	1.818
Liability towards purchase of shares	601	649
Retention money payable	947	1,44
Security deposits	-	25
Project development fees payable		45
Lease Liabilities (Refer note 2.18)	139	
	98,814	1,03,165
Note 27		
Long-term provisions		
Provision for employee benefits:		
- Gratuity	464	430
- Compensated absences	502	61-
Provision for Construction Cost	1,281	
Provision for Major Maintenance of Road ≠	121	2,216
	2,368	3,260

# As per the best estimate of the management, provision is been made, towards cost of major maintenance of the roads. Such major maintenance is to be executed in the Financial Year 2020-21 in terms of the agreement entered into with National Highways Authority of India.

Particulars	31st March, 2020	31st March, 2019
Opening balance	2,216	1.109
Add: Provision for the year	1,219	1.011
Add: Unwinding of discounts on provision	186	+ 96
Closing balance:	1. Sec.	
- Long-term Provision	121	2,216
- Short-term Provision	3,500	ar ar
Note 28		
Deferred tax liabilities (Net)		
- on account of PPE and Intangibles	4,864	1,117
- other financial assets at fair value through profit and loss		467
<ul> <li>On other provisions (including provision for employee benefits and other amounts allowable on a payment basis)</li> </ul>	157	-
	5,021	1,584
Note 29		
Other non-current liabilities		
Advance received from customers	7.050	7,050
Advance rent received	2,725	4,390
	9,775	11,440
Note 30		
Short-term borrowings (refer footnote to Note No. 25)		
Secured		
- From Banks	10,011	5,789
Unsecured		
- From Bank	20,000	
Commercial paper		
- From Mutual funds ≠	75,303	76,567
Bank Overdraft		4,118
Inter Corporate Deposits from Related Parties	-	14,000
Loans from Others *	100	100
	1,05,414	1,00,574

### Note:

# The said borrowing represents Commercial paper issued to mutual funds at a discount rate ranging from 6.10% - 9.00% per annum (2019: 7.20% - 9.10% per annum), and the same are repayable within one year at the agreed upon full face value.

\* Borrowings outstanding as on 31 March 2020 and 31 March 2019 is towards interest free, unsecured loan which shall be due and repayable after satisfaction of tranche 1 conditions in Matheran Ropeways Pvt. Ltd.

	Notes to the consolidated financial statements (Continued)		
	(Currency: Indian rupees in lakhs)		
	(currency, mount rupees in mans)	As at	As at
		31 March 2020	31 March 2019
	Note 31		
	Trade and other payables		
	Due to Micro and Small Enterprises (Refer Note 48)	6	53
	Due to others	5,873	9,415
		5,879	9,468
	Note 32		
	Current - Other financial liabilities		
	Current Maturity of Long Term Borrowings (refer footnote to Note No. 25)		
	Non-convertible debentures	39,994	1,18,794
	From banks	35,794	9.465
	Security deposits from customers	1,619	3,341
	Capital creditors	10,867	12,442
	Interest accrued but not due on borrowings	17,290	8,332
	Derivative Liability on put option	17,432	15,477
	Financial liability for premium payment at fair value	1,452	1,050
	Payable - Claim to EPC (Refer foot note to Note 22)	4,647	1,030
	Other financial liabilities	4,047	824
		1,29,765	1,69,725
			190711
	Note 33		
	Other current liabilities		
	Advances from customers	1,976	3,984
1	Statutory dues	2,577	1,143
	Corpus Fund collection	781	676
	Advance Maintenance Charges		95
ġ.	Compensation on delayed possession payable	11	11
1	Unearned rent	2,385	1,332
12	Security Deposit from Customers	43	64
1	Other current liabilities	128	49
		7,901	7,354
	Note 34		
	Short-term provisions		
	Provision for employee benefits:		
	- Gratuity	85	82
	- Compensated absences	152	181
1	Provision for Major Maintenance of Road (refer footnote to Note No.	3,500	101
		5,500	
	Other Provisions	40	
		3,777	263
		S,rit	
Ĩ	Note 35		
(	Current tax liabilities (Net)		
F	rovision for Tax	8,975	9,149
ł	Advance Tax and Tax Deducted at Source	(7,219)	(7,361)
		1,756	1,788

d

# Notes to the consolidated financial statements (Continued)

(Currency: Indian rupees in lakhs)

	31 March 2020	31 March 2019
Note 36		
Revenue from operations		
(A) Sales of products		
Sale of residential flats	13,050	9,463
(B) Sale of services		
Construction revenue	68,417	88,760
Toll revenue (includes receipt of demonetisation claim Rs. Nil Lakhs (2019; Rs. 119.07 lakhs))	12,753	12,819
Lease rentals	48,077	41,870
Maintenance and other receipts	6,911	5,953
Utility income	5,614	6,243
Parking fees income	1,296	1,251
Project management consultancy fees	1,625	1,123
Asset Management Charges Income	136	148
Revenue from other services	7	186
	1,57,879	1,67,816
Note 37		
Other income		
nterest income	3,867	1,953
nterest on income-tax refund	206	74
Profit on sale of current investments in Mutual Funds	892	273
Mark to Market gain on current investment in Mutual funds	435	100
Profit on sale of Property Plant and equipment	1	
Other income from residential projects	50	162
air value gain on derivatives (net)	÷	3,329
		4
orfeiture of security deposits		10.04
orfeiture of security deposits excess provision written back	- 10	177
orfeiture of security deposits excess provision written back rovision for doubtful debts - credit impaired written back	10 172	177
orfeiture of security deposits excess provision written back Provision for doubtful debts - credit impaired written back Aiscellaneous income		

### Notes to the consolidated financial statements (Continued)

(Currency: Indian rupees in lakhs)

	31 March 2020	31 March 2019
Note 38		
Cost of materials consumed		
Opening balance - Bought out construction materials	52	256
Less: Closing inventories	(52)	(5)
A	(52)	204
Expenditure incurred during the year:		
Opening stock of Inventories - Finished Goods and Work in Progress	43,589	44,113
Add: Transition adjustment pertaining to IND AS 115 application	43,309	1,87
(Refer Note 51 (a))		1,0 //
Addition during the year		
Construction cost *	1,950	3,319
Approval and permission expenses	26	3,315
Professional fees and technical fees	69	4
Other expenses	20	86
B	45,654	49,50
A + B	45,654	49,712
Less: Closing Stock of Inventories - Finished goods and Work in progress	33,535	43,589
20st of flats sold =	12,119	6,122
<sup>e</sup> Includes NRV provision on inventories of INR 1,420.78 Lakhs (2019: NIL).		
Note 39		
Construction costs =	64,023	83,095
Note 40		
Employee benefit expenses		
Salaries, wages and bonus	7,608	7,787
Contributions to : Provident and pension funds	223	267
Staff welfare	255	341
Gratuity expenses	37	29
Compensated absences	(35)	126
ess: capitalised to investment property under construction	(26)	(217
ess: capitalised to intangible assets under development	(77)	(74
	7,985	8,259
lote 41		
inance costs		
nterest costs: on term loans and NCD's from Banks and Financial Institutions	12.022	1000
	47,160	50,752
on cash credit and overdrafts from Banks on commercial paper from Mutual Funds	142	33
ITM on forward contract	11,250	6,140
	33	605
	1,594	440
inance charges		3,756
inance charges Inwinding of interest expense	5,666	12 0.01
inance charges Inwinding of interest expense ess: capitalised to investment property under construction	(5,306)	
inance charges Inwinding of interest expense	(5,306) (353)	(5,770
inance charges Inwinding of interest expense ess: capitalised to investment property under construction	(5,306)	(3,901) (5,770) <b>52,055</b>

Notes to the consolidated financial statements (Continued)

(Currency: Indian rupees in lakhs)

	31 March 2020	31 March 2019
Note 42		
Other expenses		
Power and fuel	5,297	7,218
Impairment in value of Goodwill (Refer Note 50 (a))	5,= 77	750
Impairment in value of Intangible asset under SCA and Building	11,314	150
(Refer Note 50(b))	11,514	
Repairs and maintenance		
- Building	70	74
- Plant & Machinery	55	89
- Provision for Major Maintenance of Road (Refer Note 27)	1,219	1.011
- Others	482	317
Operating and maintenance charges for infrastructure facilities	7,159	6,777
Fair value loss on derivatives (net)	1,103	0,777
Advertisement and business promotion expenses	1,098	1,229
egal and professional fees	3,188	
Rates and taxes		2,257
Rent (Refer Note 47)	1,162	1,129
ravelling and conveyance		858
Bank charges	303	387
elephone and communication expenses	124	314
business development expenses	55	157
raining and recruitment expenses	855	56
rovisions and write-offs:	15	127
Provision for credit impaired Trade Receivables		12.63
Bad debts	306	300
Advance written off	1	22
	207	4,050
Provision for credit impaired Inter corporate deposits	517	•
oss on account of disposal of investments in subsidiaries (Refer ote 50(c))	314	1
uditor's remuneration (Refer foot note below)	107	90
rokerage	768	101
rbitration Award paid (Refer Note No. 44(i))	1,120	-
surance charges	140	101
ees to NHAI	86	100
rinting, courier and stationery	13	40
irectors sitting fees	45	48
oss on sale of property, plant and equipment	1	31
xpenditure on Corporate Social Responsibility	85	177
ompensation paid to customers		588
ompensation paid to vendors	232	200
iscellaneous expenses	190	157
	37,951	28,555
		20,333
ot Note:		
uditor's remuneration include payments towards:		
or statutory audit	84	65
or taxation matter	15	9
or other services	9	16
	107	90

- for taxati - for other

### Notes to the consolidated financial statements (Continued)

for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### Note 43

		31 March 2020	31 March 2019
Basic earnings per share			
Loss after tax attributable to equity shareholders	Ă	(36,256)	(23,693
Number of equity shares outstanding at the beginning of the year		1,01,73,07,692	1.01.73.07.692
Equity shares issued during the year		1011/2/07/072	1.91.12.07.022
Number of equity shares outstanding at the end of the year		1.01.73.07.692	1.01.73.07.692
Weighted average number of equity shares outstanding during the year (based on date of ssue of shares)	в	1,01,73,07,692	1.01,73,07,692
Rights Share application money received during the year - pending allotment		60,00,00,000	
Weighted average number of equity shares outstanding during the year	C	1,61,73,07,692	1,01,73,07,692
Basic earnings per share of face value of Rs 10 each	[D = A/B]	(3.56)	(2.33)
Diluted earnings per share			
Diluted earnings per share of face value of INR 10 each *	[E = A/C]	(3.56)	(2.33)
Anti-dilutive - hence not considered	A STA	(cited)	(add)
iote 44			
Contingencies and commitments			
i) Contingent Liabilities (Refer footnote 1)			
Claims against the Company not acknowledged as debts		31 March 2020	31 March 2019
- Income tax demands contested by the Company		1,460	2,567
- Indirect tax demands contested by the Company		467	467
<ul> <li>Claims made by contractors (Refer footnote 2)</li> <li>Other Legal Claims</li> </ul>			1,179
		602	

The Group does not expect any outflow of economic resources in respect of the above and therefore no provision is made in respect thereof.
 Claim made by contractors has been paid during the current year, and it is debited to the Statement of Profit and Loss (Refer Note 42).

(ii) Commitments

(a) Indemnity for representations and warranties for disinvestment in retail business - INR 1,350 lakhs. (2019: Nil)

(b) Bank guarantee issued on behalf of the Company and its Subsidiaries and Joint Ventures out of the overall non fund based limits of the Company INR 7.222 Lakhs (2019: INR 5,946 lakhs).

(c) The Parent Company has issued latter of comfort to banks in respect of loans availed by a few of its subsidiaries and joint ventures

1	Name of subsidiaries / joint ventures	Nature of Comfort given
ł	Mikado Realtors Pvt. Ltd.	Shortfall undertaking to meet any shortfall during the tenure of facility
ű	Arrow Infra Estates Private Limited	Shortfall undertaking to meet any shortfall during the tenura of facility
ili	Gurgaon Constructwell Private Limited	Shortfall undertaking to meet any shortfall during the tenure of facility
IV.	Gurgaon Realtech Limited	Shortfall undertaking to meet any shortfall during the tenure of facility
v	International Infrabuild Private Limited	To ensure payment to debenture holders in the event of termination of the concession 'agreement.

# (d) The Parent Company has issued financial support letter to following subsidiaries and joint ventures on the basis of which the separate financials statements have been prepared on going concern basis

Acme Living Solutions Private Limited i

MIA Infrastructure Private Limited
 Wellkept Facility Management Services Private Limited (Previously known as TRIL Hospitality Private Limited)
 TRIF Gurgaon Housing Projects Private Limited

v TRIL Urban Transport Private Limited

vi TRIL Roads Private Limited

vii Gurgaon Constructwell Private Limited

viii HV Farms Private Limited

Notes to the consolidated financial statements (Continued) for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### Note 45

Capital and other commitments

Particulars	31 March 2020	31 March 2019
Estimated amount of contracts remaining to be executed on capital account and not provided for (net of advances)	33,024	60,491

Note 46 Segment reporting

### A. Basis for segmentation

The Group has three reportable segments, as described below, which are the Group's strategic business units. These business units offer different products and services, and are managed separately because they require different technology and marketing strategies.

The following summary describes the operations in each of the Group's reportable segments:

- Reportable segment
- Development of Residential property for outright sale.
   Real Estate
- Infrastructure

### B. Information about reportable segments

Information regarding the results of each reportable segment is included below. Performance is measured based on segment profit (before tax), as included in the internal management reports that are reviewed by the Group's CODM. Segment profit is used to measure performance as management believes that such information is the most relevant in evaluating the results of certain segments relative to other entities that operate within these industries. Inter-segment pricing is determined on an arm's length basis.

### Refer Annexure A

### C. Information about major customers

There are no customers from whom Group recognises revenue more than 10% of total revenue of the Group.

Tata Realty and Infrastructure Limited Notes to the Consolidated financial statements for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

Refer Note No. 46 Annexure A

Particulars	Development of residential property for outright sale	of residential utright sale	Real Estate	state	InfraSt	InfraStructure	Unallocated	cated	Total	al
	31 March 2020 31 March 2019	31 March 2019	31 March 2020	31 March 2019		31 March 2020 31 March 2010	Arne James 11	OT ALL COLOR		
REVENUE						CIDE INIBIA IC	31 March 2020 31 March 2019	31 March 2019	31 March 2020	31 March 2019
Revenue	13,609	10,253	62,224	55,278	81,563	1,02,328	6,481	6.573	1,63,877	1,74,432
Total revenue	13,609	10,253	62,224	55,278	81,563	1,02,328	6,481	6,573	1,63,877	1,74,432
Expenses	14,062	8,236	25,103	17,123	71,517	89,240	11,397	11,432	1,22,079	1,26,031
RESULT Segment Result	(453)	2,017	37,121	38,154	10,046	13,088	(4,915)	(4,858)	41,799	48,401
Finance costs Depreciation/Impairment		1.4	24,725 15,410	22,406 15,166	6,964 1,649	6,662 1,550	28,497 184	22,987 140	60,186 17,251	52,055 16,858
Net loss before taxes and share of joint ventures									(35,638)	(20,512)
OTHER INFORMATION										
ASSETS Segment Assets	34,803	43,382	4,94,663	5,63,247	3,67,669	2,90,662	1,83,109	85,330	10,80,244	9,82,621
Total Assets	34,803	43,382	4,94,663	5,63,247	3,67,669	2,90,662	1,83,109	85,330	10,80,244	9,82,621
LIABILITIES Segment Liabilities	484,4	169'9	3,46,266	3,67,457	2,94,277	2,25,726	3,22,192	3,53,234	9,67,219	9,53,109
Total Liabilities	4,484	169'9	3,46,266	3,67,457	2,94,277	2,25,726	3.22.192	3.53.234	9 67 219	9 53 109

An

Notes to the consolidated financial statements (Continued) for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### Note 47

## A Operating leases - as a lessee

- a) The Group has taken various premises under cancellable operating leases.
- b) The Group has also taken a commercial property and certain residential premises on non-cancellable operating leases. The future minimum lease payments in respect of these properties as on 31 March 2020 is as follows:

ease Payments	31 March 2020	31 March 2019
Not later than one year	294	424
Later than one year but not later than five years	133	791
Later than five years		-
Payments of lease rentals during the year	322	858

c) There are no exceptional/restrictive covenants in the lease agreements.

### B As a lessor:

### (i) Operating Lease

The Group has leased some investment properties. As on 31 March 2020, the future minimum lease payments in respect of these properties till the expiry of lock in period is as follows:

	31 March 2020	31 March 2019
Not later than one year	45,024	33,891
Later than one year but not later than five years	1,49,442	1,40,618
Later than five years	54,384	42,081
	2,48,850	2,16,590
Lease Rental Income for the year	46,332	44,168

### (ii) Finance leases - Fit-out and interior work

The Group's leasing arrangement represents the fit-out or interior work completed for the customers which have been classified under Ind AS 116 as Finance lease. The lease terms are for the periods of five to seven years where substantially all the risks and rewards of ownership are transferred to the lessee. The Group records disposal of the property concerned and recognizes the subsequent interest in the finance lease. No contingent rent is receivable in this regard.

Gross investment in lease	31 March 2020	31 March 2019
Receivable within one year	201	166
Receivable between one and five years	244	152
ceivable between one and five years ceivable after five years esent value of minimum lease payments t later than one year	128	
	573	318
ivable within one year ivable between one and five years ivable after five years and value of minimum lease payments ater than one year than one year but not later than 5 years	31 March 2020	31 March 2019
Not later than one year	144	114
Later than one year but not later than 5 years	134	125
Later than 5 years	109	
	387	239

### Note 48

### Micro, Small and Medium Enterprises

Under the Micro, Small and Medium Enterprises Development Act, 2006, (MSMED) which came into force from 2 October 2006, certain disclosures are required to be made relating to Micro, Small and Medium enterprises. On the basis of the information and records available with the Management, there are no outstanding dues to the Micro, Small and Medium enterprises as defined in the MSMED as set out in following disclosure:

	31 March 2020	31 March 2019
Principal amount remaining unpaid to any supplier as at the year end (o/s. for less than 30 days)	б	53
Interest due thereon		-
The amount of interest paid by the buyer as per the Micro Small and Medium Enterprises Development Act, 2006 (MSMED Act, 2006)	4	4
The amounts of the payments made to micro and small suppliers beyond the appointed lay during each accounting year	-A.	-
The amount of interest due and payable for the period of delay in making payment (which have been paid but beyond the appointed day during the year) but without adding the interest specified under MSMED Act, 2006.		÷.
The amount of interest accrued and remaining unpaid at the end of each accounting year.		1
The amount of further interest remaining due and payable even in the succeeding years, intil such date when the interest dues as above are actually paid to the small enterprise		- A
Amount of interest accrued and remaining unpaid at the end of the accounting year	~	

Notes to the consolidated financial statements (Continued) for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

Expenditure in foreign currency (on accrual basis)	31 March 2020	31 March 2019
Professional fees	26	15
Training and conference expenses		12
Travelling expenses	19	4
Membership & Subscription Expenses	14	

### Note 50

(a) Impairment in value of Goodwill

The group has tested the impairment of goodwill arising on account of consolidation / merger. Based on the performance of the project during the current year and future projections, an impairment analysis had been carried out during the year and impairment loss recognised during the year of Rs. Nil (2019: Rs. 750 Lakhs)

### (b) Impairment in value of Intangible asset under SCA and Building

As on 31st March 2020, the Enterprise Value (as per the DCF technique) of the group's Investment Property and Intangible Assets (Service Concession Arrangements (SCA)) in one of its subsidiary i.e. International Infrabuild Private Limited is Rs. 5,463 lakhs which is less than the book value of Rs. 16,777 lakhs. Therefore, an impairment loss of Rs. 11,314 lakhs has been recognized in the books of accounts. The low enterprise value primarily pertains to Multi Level Car Park and hence the intangible assets under SCA pertaining to that asset is entirely written down of Rs. 9,436 lakhs. The impairment over and above the Intangible Assets under SCA pertains to the Retail space given on rent which is represented by Building under Investment Property, hence the balance amount of Rs. 1,379 lakhs has been impaired therefrom.

### (c) Loss on account of disposal of investments in Subsidiaries

During the current year the Group has disposed off investments in two subsidiaries i.e. TRIL Amritsar Projects Limited and TRIF Real Estate Development Limited as on December 09, 2019 for a sale consideration of Rs. 9,177 lakhs. The amount of net assets disposed off is Rs.9,491 Lakhs (including goodwill of Rs. 2,163 lakhs). The net loss on disposal of investments of Rs. 314 lakhs has been charged to Statement of Profit & Loss Account.

### Note 51

### a) Disclosure with respect to transition adjustment of IndAS 115

Particulars	For the Year ended March 31, 2019
Opening Retained Earnings (before Ind AS 115)	(74.541)
Reversal of revenue	(3,174)
Reversal of Cost of sale	1.870
Opening Retained Earning (After Ind AS 115)	(75,845)
Increase in Inventory	1,870
Decrease in Trade Receivable (Other than related Party)	(3,174)

### b) Disclosure in respect of Construction Contracts

Particulars	For the Year ended March 31, 2020	For the Year ended March 31, 2019
Contract revenue recognized as revenue during the year	68,417	88,760
Particulars	For the Year ended March 31, 2020	For the Year ended March 31, 2019
Cumulative revenue recognized	2.39.396	1.70.979
Cumulative costs incurred	2,27,966	1.63,942
Cumulative margins accounted	11,430	7.037
Advances paid	5,870	11,907
Retention money payable	112	

Notes to the consolidated financial statements (Continued) for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

### Note 52

Interests in other entities

(a) Subsidiaries

The company's subsidiaries at 31 March 2020 are set out below.

	Place of	% of owner	rship interest
Name of entity	business	31 March 2020	31 March 2019
Acme Living Solutions Private Limited	India	100.00%	100.00%
Arrow Infraestate Private Limited	India	100.00%	100.00%
Gurgaon Constructwell Private Limited	India	100.00%	100.00%
Gurgaon Realtech Limited	India	100.00%	100.00%
TRIL Roads Private Limited	India	100.00%	100.00%
TRIF Gurgaon Housing Projects Private Limited	India	100.00%	100.00%
TRIL Urban Transport Private Limited	India	100.00%	100.00%
Wellkept Facility Management Services Private Limited	India	100.00%	100.00%
TRIL Constructions Limited	India	67.50%	67.50%
TRIL Infopark Limited (refer note no. (ii) & (iii)	India	100.00%	100.00%
TRIL Amritsar Projects Limited (upto December 09, 2019 - 100.00%) (Refer note 50 (c))	India	1	100.00%
Hampi Expressway Private Limited	India	100.00%	100.00%
TRIF Real Estate Development Limited (upto December 09, 2019 - 100.00%) (Refer note 50 (c))	India		100.00%
HV Farms Private Limited	India	100.00%	100.00%
International Infrabuild Private Limited (IIPL)	India	26.00%	26.00%
Dharamshala Ropeway Limited	India	74.00%	74.00%
Manali Ropeway Private Limited	India	72.00%	72.00%
Uchit Expressways Private Limited	India	100.00%	100.00%
TRPL Roadways Private Limited	India	100.00%	100.00%
Matheran Ropeways Private Limited	India	70.00%	70.00%
Durg Shivnath Expressways Private Limited	India	100.00%	100.00%
MIA Infrastructure Private Limited	India	100.00%	100.00%

### Significant judgement: consolidation of entities with less than 50% voting interest

i) Although the Group owns less than one-half of the voting power of International Infrabuild Private Limited, it is able to control the company by virtue of an agreement with the other investors of International Infrabuild Private Limited which inter-alia provides the Group with power to appoint majority of the board of directors of International Infrabuild Private Limited and power over relevant activities. Consequently, the Group consolidates its investment in the company.

ii) The Parent company has entered into call option with one of the shareholder of TRIL Infopark Ltd expiring on 10th July 2021 wherein the Parent Company holds the call options. Company has paid full consideration of Rs 7,110 Lakhs as option deposit. Hence Parent company is consolidating shareholding in TRIL Infopark Ltd of the said shareholder as stake held by it.

iii) Similarly, the Parent Company has entered into put option with another shareholder of TRIL Infopark Ltd expiring on 31st March. 2021. Based on the put option conditions Parent Company has provided full liability under other current financial liabilities, since the put options rights are available with the other shareholders. Hence, Parent Company is consolidating TRIL Infopark Ltd as 100% subsidiary.

### (b) Non-controlling interests (NCI)

Set out below is summarised financial information for each subsidiary that has non-controlling interests to the group. The amounts disclosed for each subsidiary are before inter-company eliminations.

Summarised balance sheet	and the second se										
	Dharamshala Ropeway		Dharamshala Ropeway Limited TRIL Constructions Limited Man			Manali Ropew Limit	A CONTRACTOR OF	ш	PL.	Matheran Ropeways Private Limited	
	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019	
Non-current assets	8.105	7,168	21,484	21,691	5,773	5,145	5,652	13,645	1,599	1,587	
Current assets	359	108	2,645	2,648	4	5	317	605	975	951	
Non-current liabilities	(7,106)	(5,641)	(8,868)	(8,615)	(5.397)	(4,934)	(7,079)	(6,474)	-	-	
Current liabilities	(768)	(1.132)	(460)	(289)	(241)	(9)	(794)	(9,701)	(112)	(101)	
Net assets	590	503	14,801	15,435	139	207	(1,904)	(1,925)	2,462	2,437	
Net assets attributable to NCI	70	104	6,098	6,304		•			216	209	

	Dharamshala R	lopeway Limited	TRIL Constructio	ns Limited	Manali Ropew Limit		m	PL	Matheran Ropeways Private Limited	
	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019
Summarised statement of profit and loss							_			-
Revenue	2.25	2.11	-		.,		488	136	58	55
(Loss)/Profit	(133)	(649)	(634)	(512)	(68)	(55)	(9,559)	(1,912)	26	25
OCI								- <u>Ali</u> n		
Total comprehensive income	(133)	(649)	(634)	(512)	(68)	(55)	(9,559)	(1,912)	26	25
Loss allocated to NCI	(35)	(169)	(206)	(166)	(19)	(15)	(7.074)	(1,415)	8	7
OCI allocated to NCI			1		in the second	1.10	1.101.04	Cards.		
Total comprehensive income allocated to NCI	(35)	(169)	(206)	(166)	(19)	(15)	(7.074)	(1,415)	8	7
Loss to NCI Restricted	(35)	(169)	(206)	(166)		27		79	8	7
Summarised cash flows										
Cash flows from operating activities	(85)	(171)	(256)	(723)	(4)	(10)	(154)	(147)	(24)	(60)
Cash flows from investing activities	(936)	(1,421)		(204)	(230)	(34)	(1.032)	(1.420)	(877)	54
Cash flows from financing activities	1,167	1,634	253	895	234	48	780	1,990	+	-
Net increase/ (decrease) in cash and cash equivalents	146	42	(3)	(32)	1993	4	(406)	423	(901)	(6)

Notes to the consolidated financial statements (Continued)

for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

Note 52 (continued)

Interests in other entities (continued)

(c) Transactions with Non-controlling interests There are no transactions with non-controlling interests in 2019-2020

(d) Interests in joint ventures Set out below are the joint ventures of the group as at 31. March 2020 which, in the opinion of the directors, are material to the group. The country of incorporation or registration is also their principal place of business.

	Place of	% of ownership		
Name of entity	business	interest		
Pune Solapur Expressways Private Limited	India	50.00%	Joint venture	Equity method
A & T Road Construction Management and Operation Private	India	50.00%	Joint venture	Equity method
Limited				
TRIL IT4 Private Limited	India	74.00%	Joint venture	Equity method
Mikado Realtors Pvt. Ltd.	India	74.00%	Joint venture	Equity method
Industrial Minerals and Chemicals Company Private Limited	India	74.00%	Joint venture	Equity method
Pune IT City Metro Rail Limited	India	74,00%	Joint venture	Equity method

Duna Salanus Ex

Dalais

I C.T.D

(i) All the above joint ventures are engaged in the business of real estate and infrastructure development. It is a strategic investment which utilises the group's knowledge and expertise in the development of real estate and infrastructure development.

(ii) Significant judgement: Although the Group owns more than one-half of the voting power of TRIL IT4 Private Limited, Mikado Realtors Pvt. Ltd., Industrial Minerals and Chemicals Company Private Limited and Pune IT City Metro Rail Limited, it does not have control over these Companies. The Group has joint control over these companies by virtue of an agreement with the other investors. In other words, decisions about the relevant activities - i.e. those that significantly affect the returns of the arrangement - require the affirmative consent of the investors. Consequently, the Group has classified its interest in these companies as jointly controlled entity.

Summarised financial information for associates and joint ventures

	Pune Solapur Expressways Private Limited		A & T Road Construction Management and Operation Private Limited		Industrial Minerals and Chemicals Company Private Limited		TRIL IT4 Private Limited (Refer Note 9)		Mikado Realtors Pvt Ltd		I Pune II City Metro Rail Limited		Total	
	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019	31 March 2020	31 March 2019
Percentage ownership interest	50%	50%	50%	50%	74%	74%	74%	74%	74%	74%	74%	74%		
Current assets	10,824	5,977	1.1	-	1,815	58	2,374	3,224	589	5	1.912		17,514	9,262
Non-current assets	88,718	93,221	3	-	16,548	16,460	26,079	29,232	55,461	29,353	4,792	-	1,91,600	1,68,266
Current liabilities	11,820	9,520	1.1	2	41	25	43,126	3,069	5,517	2,966	665		61,170	15,581
Non-current liabilities	83,223	85,333	-		2	-	13,596	52,109	46,216	21,940	327	1.1	1,43,362	1,59,382
Net assets	4,500	4,344	2	(2)	18,322	16,493	(28,269)	(22,722)	4,316	4,452	5,711		4,582	2.565
Group's share of net assets	2,248	2,170	1	(1)	13,558	12.205	(20,919)	(16,815)	3,194	3,295	4.226		2,308	854
Goodwill Recognised			2	1.4	8,392	9,705		(10,010)	27,141	27.141	TableO		35.533	36.846
Carrying amount of interest in joint venture	2,248	2,170	Т	(1)	21,950	21,910	•	×	30,335	30,436	4,226		58,760	54,516

	Limited		A & I Road Construction Management and Operation Private Limited		Chemicals Company Private Limited		(Refer Note 9)		Mikado Realtors Pvt Ltd		Pune IT City Metro Rail Limited		То	Total	
Summarised statement of profit and loss	31 March, 2020	31 March, 2019	31 March, 2020	31 March, 2019	31 March, 2020	31 March, 2019	31 March, 2020	31 March, 2019	31 March, 2020	31 March, 2019	31 March, 2020	31 March, 2019	31 March, 2020	31 March, 2019	
	50%	50%	50%	50%	74%	74%	74%	74%	74%	74%	74%	74%			
Revenue	16,568	15,552				1	8,695	7,864	0		4.395		29,658	23,416	
Depreciation and amortisation	4,364	4,087		1.12			6.580	6,841	0		0		10,945	10,928	
Interest expense	8.085	8,654			4	~	5,659	5,592	87		47		13,882	14,247	
Income tax expense					-	-		2127-					13,064	14,247	
Loss	153	(1.198)	(D)	(0)	54	(15)	(5,547)	(5,763)	(136)	(32)	(290)	- C -	(5,766)	(7,008)	
Other comprehensive income	3	I		3.4		3,000	0	(articles)	(120)	()	(=>0)		(51/00)	(1.000)	
Total comprehensive income	156	(1,197)	(1)	(0)	54	(15)	(5,546)	(5,763)	(136)	(32)	(290)		(5,763)	17 007	
Group's share of loss	77	(599)	(1)	1.4	40	(1)	(0,0,40)	(Second)	(101)	(24)					
Group's share of OCI	2	0	(1)		40	0.0		-	(Ior)	(24)	(215)		(200)	(633)	
Group's share of total comprehensive income	78	(598)	(1)		40	(11)	- N	-	(101)	(24)	(215)		(198)	(633)	

Gro.

# Note 53 Financial instruments – Fair values and risk management

A. Accounting classification and fair values

	Carrying				Fair value			
31 March 2020	FVTPL	FVTOCI	Amortised Cost	Total	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Total
Financial assets								
Investments	1.							
- Equity instruments in others.	222	-	-	222	-	-	222	222
- Bonds and debentures		-	19,927	19,927			-	
Current investments in mutual funds	71,180	~	1 ( A C	71,180	71,180	-	-	71,180
Inter-corporate Deposits given to Related Parties	1.1.1.2.	-	2,510	2,510		-	~	
Trade receivables		-	2,865	2,865			-	
Cash and cash equivalents	÷		29,940	29,940			100	
Lease rent receivable		-	243	243		-	-	
Unbilled revenue		*	438	438		-		
Deposit other than included in cash and cash equivalent	÷		6,195	6,195				
Security deposits	-		1,759	1.759		-	-	
Other loans and advances	÷.		1,389	1.389		-		
Other financial assets	-		1,781	1,781	- · · · · · · · · · · · · · · · · · · ·	-		
	71,402		67,047	1,38,439	71,180		222	71,402
Financial liabilities								
Non convertible debentures	1.00		1.79.413	1,79,413	-			
Financial liability for premium payment			69,016	69,016				
Project development fees payable		-	0.00					
Loan and working capital facility from Bank	S		5,23,135	5.23.135				
Commercial paper issued to mutual fund			75.303	75,303				
Bank Overdraft		-		100	-	-		-
Inter Corporate Deposits from Related Parties	-	-	1.4		2	2	-	-
Derivative instruments	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1							
Put option	18,823		-	18.823			18.823	18.823
Forwards	-			-				
interest- free security deposits from customers		-	22,406	22,406				
Other financial liabilities		-	42,645	42,645	-			20
Trade payables			5.879	5,879	· · · · · · · · · · · · · · · · · · ·	14.1		
	18,823		9,17,797	9,36,620			18,823	18,823

	Carrying				Fair value			
31 March 2019	FVTPL	FVTOCI	Amortised Cost	Total	Level 1 - Quoted price in active markets	Level 2 - Significant observable inputs	Level 3 - Significant unobservable inputs	Total
Financial assets								
Investments								
- Equity instruments in others	23			23		-	23	23
- Bonds and debentures			18,614	18,614			-	
Current investments in mutual funds	1,383	~		1,383	1,383		-	1,383
Trade receivables	-	-	2,846	2,846	-	-	-	1
Cash and cash equivalents		5.0	5,921	5,921				-
Lease rent receivable		÷	193	193	1 1 E E		81	-
Unbilled revenue	-	-	537	537	9	-		-
Deposit other than included in cash and cash equivalent	÷	2	2,079	2,079	2			1
Security deposits	2	-	730	730				
Other loans and advances	2.0	2.0	3,054	3,054	1.	1.1	21	
Other financial assets (Forward Derivate Contract Rs. 114 Lakhs)	-	~	1,696	1,696		~		
	1,406	4	35,670	37,076	1,383	2	23	1,406
Financial liabilities								
Non convertible debentures		-	2,33,586	1.33,586		~	-	
Financial liability for premium payment			69,110	69,110				
Project development fees payable	-	-	45	45			-	
oan and working capital facility from Bank			4.44,949	4,44,949	1.1			
Commercial paper issued to mutual fund			76,567	76,567				1.12
Bank Overdraft		1	4,118	4,118	1.1	-	2	
Inter Corporate Deposits from Related Parties			14,000	14,000			1	
Derivative instruments				1.0 000			10.000	1.0.0
Put option	17,295			17,295		-	17,295	17,295
Forwards	425		4	425		-	425	425
interest- free security deposits from customers	~	-	21,551	21,551		-	7	-
Other financial liabilities	-	-	36,305	36,305		-		-
Trade payables	-	-	9,468	9,468	-		-	
	17,720		9,09,699	9,27,419			17,720	17,720

4 R

Notes to the Consolidated financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### B. Measurement of fair values

Valuation techniques and significant unobservable inputs The following table shows the valuation techniques used in measuring Level 2 and Level 3 fair values for tinancial instruments measured at fair value in the statement of financial position as well as the significant unobservable inputs used.

Туре	Valuation technique	Significant unobservable inputs	Inter-relationship between significant unobservable inputs and fair value measurement
Financial assets			
Investments	the second se		
<ol> <li>Equity instruments in others (Echanda Urja Private Limited)</li> <li>Equity instruments in others (Vagarai Windfarms Limited)</li> <li>Equity instruments in others (Perinyt Neep Private Limited)</li> </ol>	Measured at cost	Not available	Not available
Derivative instruments			
- Put option - TGBL	The Courpany has used Monte Carlo simulation to estimate the fair value of the options. Monite Carlo Simulation: The valuation model simulates the equity value per share of the underlying securities considering (i) a risk free raite rate and (ii) volatility of stock prices of i dentical securities in markets. The expected pay-off is determined considering the courtrated strike price of the option and the simulated equity value per share at each node. The fair value of the option is estimated by discounting the estimated pay-off using a risk free rate for similar maturity as the valuation model works on a risk netural framework.	* Underlying equity value of the company (Rs. 4,280 lakhs)	2020 "If underlying enterprise value of the company- increases (decrease) by 10%, then fair value would (decrease) increase by BNR (10%)/10%. 2019 "If underlying enterprise value of the company increases (decreases) by 10%, then fair value would (decrease) increase by INR (10%)/10%.
- Forwards - IAI	The Company has used Monte Carlo simulation to estimate the fair value of the options. Monte Carlo Simulation: The valuation model simulates the equity value per share of the underlying securities considering (1) a risk free rate rate and (ii) volatility of stock prices of identical securities in markets. The expected pay-off is determined considering the contracted strike price of the option and the simulated equity value per share at each node. The far value of the option is estimated by discounting the estimated pay-off using a risk free rate for similar maturity as the valuation model works on a risk netural finamework.	* Underlying Equity value of the company (Rs, Nil lakhs) '2019:	2020 *If underlying enterprise value of the company increases (decrease) by 10%, then fair value would (decrease) increase by INR (10%)/10%. 2019 *If underlying enterprise value of the company meanses (decrease) by 10%, then fair value would (decrease) increase by INR (10%)/10%.
Following have been recorded at amortised cost: 1. Investments in Non-convertible debentures of equity accounted nvestees 2. Lease rent receivables 3. Security deposit given 4. Non convertible debentures (borrowings) 5. Security deposits received from customers 5. Tamil Nado Industrial Corporation Limited ("TIDCO") Put ption	Discounted eash flow approach: The valuation model considers the present value of expected payment, discounted using a risk adjusted discount rate.	Not applicable	Not applicable

Financial risk management

The Company has exposure to the following risks arising from financial instruments: A. Credit risk : B. Liquidity risk ; and C. Market risk

Risk management framework The Company's bard of directors has overall responsibility for the establishment and oversight of the Company's risk management framework. The Company manages market risk through a treasury department, which evaluates and exercises independent control over the entire process of market risk management. The treasury department recommends risk management objectives and policies, which are approved by Board of Directors. The activities of this department include management of cash resources, borrowing strategies, and ensuring compliance with market risk limits and policies.

The Company's risk management policies are established to identify and analyse the risks faced by the Company, to set appropriate risk limits and controls and to monitor risks and atherence to limits. Risk management policies and systems are reviewed regularly to reflect changes in market conditions and the Company's activities. The Company, through its training and management standards and procedures, arms to maintain a disciplined and constructive control environment in which all employees understand their roles and obligations.

The audit committee oversees how management monitors compliance with the company's risk management policies and procedures, and reviews the udequacy of the risk management framework in relation to the risks faced by the Company. The audit committee is assisted in its oversight role by internal audit. Internal audit undertakes both regular and ad hoc reviews of risk management controls and procedures, the results of which are reported to the audit committee.



Notes to the Consolidated financial statements for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

### Note 53 (continued)

### Liquidity risk

Liquidity risk is the risk that the Group will not be able to meet its financial obligations as they become due. The Group manages its liquidity risk by ensuring, as far as possible, that it will always have sufficient liquidity to meet its liabilities when due, under both normal and stressed conditions, without incurring unacceptable losses or risk to the Group's reputation.

The Group has obtained fund and non-fund based working capital lines from various banks. Furthermore, the Group has access to funds from compulsory convertible and nonconvertible debentures from holding Group and other financial institution. The Group also constantly monitors funding options available in the debt and capital markets with a view to maintaining financial flexibility.

### Exposure to liquidity risk

The table below analyses the group's financial liabilities into relevant maturity groupings based on their contractual maturities for:

\* all non derivative financial liabilities

\* net and gross settled derivative financial instruments for which the contractual maturities are essential for the understanding of the timing of the cash flows.

31 March 2020					
Non-derivative financial liabilities		Co	ntractual cash flows	6	
	Carrying amount	1 year or less	1-2 years	2-5 years	More than 5 years
Non convertible debentures	1,79,413	40,000	72,500	67,000	
Financial liability for premium payment	69,016	2,108	7,078	13,570	2,93,673
Interest- free security deposits from customers	22,406	1,619	1.1		40
Commercial paper issued to mutual fund	75,303	77,500	12	1.14	
Trade and other payables	5,879	5,879			4
Loan and working capital facility from Banks and Financial Institutions	5,23,135	66,979	46,094	62,113	3,60,544
Other financial liabilities	42,645	36,713	12	5,933	-
	9,17,797	2,30,798	1,25,672	1,48,616	6,54,217
Derivative financial liabilities					
- Put option	18,823	17,432	-	1.391	
	18,823	17,432		1,391	

### 31 March 2019 Non-derivative financial liabilities

Non-derivative infancial habilities			Contra	actual cash flows	and the second sec
	<b>Carrying amount</b>	1 year or less	1-2 years	2-5 years	More than 5 years
Non convertible debentures	2,33,586	1,18,820	40,000	45,500	29,650
Inter Corporate Deposits from Related Parties	14,000	14,000	1.	1.	
Financial liability for premium payment	69,110	1,316	6,846	13,485	3,22,010
Project development fees payable	45	45	AC AC		
Bank Overdraft	4,118	4,118		$\sim$	-
Interest- free security deposits from customers	21,551	3,341	5,059	13,773	4,726
Commercial paper issued to mutual fund	76,567	77,500	1		
Trade and other payables	9,468	9,468			
Secured loan and working capital facility from Bank	4,44,949	13,365	31.950	1,22,395	2,86,055
Other financial liabilities*	36,305	24,865	1.	11,439	
	9,09,699	2,66,838	83,855	2,06,592	6,42,441
Derivative financial liabilities					
- Put option	17,295	15,477	é.	1,818	
- Forwards	425		~	425	-
	17,720	15,477		2,243	

two strict such flowers

\* Forward contract entered with the banks with respect to the firm commitment of supply of assets / services entered with the foreign contractor to be constructed and installed on the project site. Notional amount of foreign currency Euro 59.40 lakhs and notional INR 114.35 lakhs.

Notes to the Consolidated financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

Note 53 (continued) Credit Risk

Credit risk is the risk of financial loss to the Group if a customer or counterparty to a financial instrument fails to meet its contractual obligations, and arises principally from the Group's receivables from customers. Credit risk is managed through credit approvals, establishing credit limits and continuously monitoring the creditworthiness of customers to which the Group grants credit terms in the normal course of business. The Group establishes an allowance for doubtful debts and impairment that represents its estimate of incurred losses in respect of trade and other receivables, loans and advances and investments.

### Trade and other receivables

The Group's exposure to credit risk is influenced mainly by the individual characteristics of each customer. The demographics of the customer, including the default risk of the industry and country in which the customer operates, also has an influence on credit risk assessment. Credit risk is managed through credit approvals, establishing credit limits and continuously monitoring the creditworthiness of customers to which the Group grants credit terms in the normal course of business.

### Credit risk for receivables pertaining to residential business

The risk for trade receivables pertaining to residential business is considered nil as the possession of the residential property is transferred only after the receipt of payment from the customer.

Summary of the Group's exposure to credit risk by age of the outstanding from various customers is as follows:

	31 March 2020	31 March 2019
Neither past due nor impaired	987	1,572
Past due but not impaired		
Past due 1-90 days	1,412	800
Past due 91-180 days	33	293
Past due 181-270 days	9	155
Past due 271-365 days	424	-
Past due more than 365 days		26
	2,865	2,846

### Notes to the Consolidated financial statements

for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### Note 53 (continued)

### (ii) Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The Group's exposure to market risk for changes in interest rates relates borrowings from financial institutions.

### Exposure to interest rate risk

The interest rate profile of the Group's interest-bearing financial instruments is as follows:

Particulars	31 March 2020	31 March 2019	
Fixed-rate instruments			
Financial assets	32,805	5,808	
Financial liabilities	3,47,345	3,99,071	
Variable-rate instruments			
Financial liabilities	4,84,090	4,20,040	

### Interest rate sensitivity - fixed rate instruments

The Group does not account for any fixed-rate financial assets or financial liabilities at fair value through profit or loss, and the Group does not have any designate financial liabilities.

### Interest rate sensitivity - variable rate instruments

A reasonably possible change of 100 basis points in interest rates at the reporting date would have increased / decreased equity and profit or loss by amounts shown below. This analyses assumes that all other variables, in particular, foreign currency exchange rates, remain constant. This calculation also assumes that the change occurs at the balance sheet date and has been calculated based on risk exposures outstanding as at that date. The period end balances are not necessarily representative of the average debt outstanding during the period.

	100 bp increase	100 bp decrease
As at '31 March 2020		
Variable-rate instruments	4,841	(4,841)
Cash flow sensitivity (net)	4,841	(4,841)
	100 bp increase	100 bp decrease
As at '31 March 2019		-0
Variable-rate instruments	4,200	(4,200)
Cash flow sensitivity (net)	4,200	(4,200)

(Note: The impact is indicated on the profit/loss and equity before tax basis)

Notes to the Consolidated financial statements

for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### Note 54 Capital Management

The Group's policy is to maintain a strong capital base so as to maintain investor, creditor and market confidence and to sustain future development of the business. It sets the amount of capital required on the basis of annual business and long-term operating plans which include capital and other strategic investments.

The funding requirement of the Group are met from fund and non-fund based working capital lines from various banks. Furthermore, the Group has access to funds from compulsory convertible and non-convertible debentures from holding Group and other financial institution. The Group's policy is to use short-term and long-term borrowings to meet anticipated funding requirements.

The Group monitors capital using a ratio of 'adjusted net debt' to 'adjusted equity'. For this purpose, adjusted net debt is defined as total liabilities, comprising interest-bearing loans and borrowings and obligations under finance leases, less cash and cash equivalents. Adjusted equity comprises all components of equity other than amounts accumulated in the hedging reserve.

	As at 31 March 2020	As at 31 March 2019
Total borrowings	7,77,951	7,73,320
Less : Cash and cash equivalent	29,940	5,921
Adjusted net debt	7,48,011	7,67,399
Adjusted equity	1,13,025	29,512
Adjusted net debt to adjusted equity ratio	7	26

### Note 55

# Tax expense (a) Amounts recognised in profit and loss

	31 March 2020	31 March 2019
Current tax expense	353	864
Current tax expense relating to prior years		
Deferred Tax Expense		
Original and reversal of temporary difference	300	1,906
Increase in tax rate		
Recognition of previously unrecognised tax losses	-	
Change in recognised deductible temporary differences		
Total deferred tax expense/(benefit)	300	1,906
Tax expense for the year	653	2,770

(b) Amounts recognised in other comprehensive income

	For the year Before tax	Tax (expense)	Net of tax	For the year Before tax	Tax (expense)	Net of tax
	INR	INR	INR	INR	INR	INR
Items that will not be reclassified to profit or loss				1. 2. 2.		
Remeasurements of the defined benefit plans	3	(1)	2	55	(14)	41
Equity Instruments through Other Comprehensive Income	A			11		-
	3	(1)	2	55	(14)	41

(c) Reconciliation of effective tax rate

31 March 2020 31 March 2019

rofit before tax	(35,836)	(21,145)
ax using the Company's domestic tax rate	(9,353)	(4,787)
farch 2020: 26% or 29.12% (March 2019: 34.944% or 26%))		
ax effect of:		
ifferences in tax rates	138	
ems not taxable/ considered separately	92	99
apact on account of Ind As adjustments	(591)	(688)
irrent-year losses for which no deferred tax asset is recognised	10,775	7,363
fect of deduction claimed under chapter IV/ VIA of income tax act, 1961	(954)	(1,041)
nutilised MAT Credit	206	855
hers	340	969
otal	653	2,770

a

### Notes to the consolidated financial statements (Continued)

for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### Note 56

### Disclosure pursuant to Ind AS 19 on "Employee Benefits"

(i) The Group has adopted Ind AS 19 on "Employee Benefits" as per the Companies (Indian Accounting Standards) Rules, 2015 notified under Section 133 of the Companies Act, 2013 ('the Act') and other relevant provisions of the Act.

(ii)	Contribution to Provident fund		
		31 March 2020	31 March 2019
	Contribution to provident fund recognised as an expense under "Employee benefits".	223	267

### (iii) Defined Benefit Plans

....

-....

### Gratuity

The Group has a defined benefit gratuity plan. Every employee who has completed five years or more of service gets a gratuity on death or resignation or retirement calculated as per the Payment of Gratuity Act, 1972 with no ceiling.

		31 March 2020	31 March 2019
I	Change in the defined benefit obligation		
	Liability at the beginning of the year	421	454
	Interest Cost	34	35
	Current Service Cost	75	86
	Benefit Paid	(43)	(94)
	Actuarial Loss on obligations	17	(60)
	Liability at the end of the year	504	421
п	Amount Recognised in the Balance Sheet		
	Liability at the end of the year	504	421
	Fair Value of Plan Assets at the end of the year	-	-
	Difference	504	421
	Amount recognised in the Balance Sheet	504	421
ш	Expenses Recognised in the statement of profit and loss		
	Current Service Cost	68	86
	Interest Cost	29	35
	Net Actuarial Loss to Be Recognised	17	(60)
	Expense Recognised in statement of profit and loss	114	61
IV	Balance Sheet Reconciliation		
	Opening net liability	421	454
	Expense as above	114	61
	Employers contribution (paid)	(31)	(94)
	Amount recognised in Balance Sheet	504	421

### Notes to the consolidated financial statements (Continued)

for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### (iv) Defined Benefit Plans (Continued)

ratuity (Continued)	31 March 2020	31 March 2019
V Actuarial Assumptions :		
Discount Rate	5.55%	7.05%
Salary escalation	7.00%	7.00%

### Attrition Rate: Directors - Nil, Age 21-30 years - 5%, Age 31-40 years - 3%, Age 41-59 years -2%

Estimates of future salary increases, considered in actuarial valuation, take account of inflation, seniority, promotion and other relevant factors, such as supply and demand in the employment market.

The Group's liability on account of gratuity is not funded and hence, the disclosures relating to the planned assets are not applicable.

		31 March 2020	31 March 2019
VI	Experience Adjustments:		
	Defined benefit obligation	381	368
	Plan assets		
	(Deficit)	(381)	(368)
	Experience adjustment on plan liabilities		
	Experience adjustment on plan assets	4	

### Sensitivity analysis

Reasonably possible changes at the reporting date to one of the relevant actuarial assumptions, holding other assumptions constant, would have affected the defined benefit obligation by the amounts shown below.

	31 March	2020	31 March 2019		
	Increase	Decrease	Increase	Decrease	
Discount rate ( % movement)	(2.10)	2.20	(2.60)	2.70	
Salary escalation (% movement)	2.20	(2.10)	2.70	(2.60)	

### (v) Other long term employment benefits

### **Compensated** absences

The liability towards compensated absences for the year ended 31 March 2020 recognised in the Balance Sheet based on actuarial valuation using the projected unit credit method amounted to INR 654/- (2019: INR 795/-) and the charge to the Statement of profit and loss amounted to INR (35)/- (2019: INR 126/-).

### Notes to the consolidated financial statements (Continued) for the year ended 31 March 2020

(Currency: Indian rupees in lakhs)

### Note 57 Related Party Disclosures

Holding company: Tata Sons Private Limited

### Joint Venture

TRIL IT4 Private Limited Mikado Realtors Private Limited Industrial Mineral and Chemicals Company Limited

### Joint Venture of a Subsidiary

Pune Solapur Expressways Private Limited (a JV of TRIL Roads Private Limited) A & T Road Construction Management and Operation Private Limited Pune IT City Metro Rail Limited

Other related parties with whom transactions have taken place during the year:

### Fellow Subsidiaries:

Ewart Investments Limited Tata Consultancy Services Limited Tata AIG General Insurance Company Limited Tata AIA Life Insurance Company Limited Tata Capital Financial Services Limited Infinity Retail Ltd. Tata Asset Management Limited Tata Housing Development Company Limited Tata Elxsi Ltd. Tata Teleservices Limited Tata Consulting Engineers Ltd

### Key Management Personnel

Sanjay Dutt Sanjay Sharma Arvind Chokhany Sudhakar Shetty Vinay Gaokar Managing Director & CEO Chief Financial Officer - w.e.f. 10th September 2018 Chief Financial Officer - upto 28th February 2019 Company Secretary - w.e.f. 1st December 2019 Company Secretary - upto 30th November 2019

### Related party disclosures (Continued)

T	ransactions	Holding company	Fellow Subsidiaries	Joint Ventures	Key Management Personnel	Total
1 5	hare Application Pending Allotment					
	1 March 2020	1,20,000	1.1	0.00	0.40	1,20.00
3	1 March 2019	-		2		
	ata Sons Private Limited					
	1 March 2020	1,20,000		-		1,20,00
	1 March 2019	-			-	-
	nsecured loan taken		AT 144			1.00
1.7	1 March 2020	1 <del>2</del> 1	43,590	2	-	43,59
_	1 March 2019	1.2	14,000	÷		14,00
	ata Housing Development Company Limited					
	1 March 2020		43,590	1.00		43,59
3	I March 2019	-	14,000		-	14,00
1	nsecured loan repaid					
	March 2020		57,590			57.59
	I March 2019		11,590	-	7	27,25
	ata Housing Development Company Limited					,
	March 2020		57 500			
			57,590		7	57,59
5.	March 2019		-	-	1	-
i In	terest expenses on unsecured loans					
	March 2020		846			84
	March 2019		21		2	2
	ata Housing Development Company Limited					
	March 2020	2.11	846		6	84
	March 2019	-	21		2.	2
	urchase of property, plant & equipments					
	March 2020	-	1	1	-	
	March 2019	-		-		-
In	finity Retail Limited					
31	March 2020		1	(-)		
31	March 2019		-			-
	irchase of Investments			Late.		
	March 2020	-	-	5,756	1.0	5,75
	March 2019			105		10
	dustrial Minerals and Chemicals Private Limited					
	March 2020			1,314		1,31
	March 2019			105		10
	ine It City Metro Rail Limited					
	March 2020		-	4,441	-	4,44
31	March 2019			-		-
A	& T Road Construction					
31	March 2020	1.4		2	2	
31	March 2019		e)	(+) <sup>*</sup>		
	Management Completion Com					
	set Management Consultancy fees			0.5		- 13
	March 2020	1.0	24	136	1.2	13
	March 2019			148		14
	RIL IT4 Private Limited. March 2020			136		13
	March 2020 March 2019	20	50 B			
	WIGIUN 2017			148		14

R

### Related party disclosures (Continued)

### Summary of related party transactions

	Transactions	Holding company	Fellow Subsidiaries	Joint Ventures	Key Management Personnel	Total
8	Project Management Consultancy fees					
	31 March 2020			1,625		1,6
	31 March 2019			1,118		1,1
	Mikado Realtors Private Limited					
	31 March 2020	-		1,485	-	1,4
	31 March 2019	-		1,118		1,1
	TRIL IT4 Private Limited					.,.
	31 March 2020			140		1
	31 March 2019	-	2	-	-	
~						
9	Interest Income			0.752		
	31 March 2020	7	-	1,465		1,4
	31 March 2019	×		1,465		1,4
	TRIL IT4 Private Limited					
	31 March 2020	T	~	1,465		1,4
	31 March 2019		5	1,465	· · ·	1,4
0	Rent Income					
	31 March 2020	-	4,747		-	4.7
	31 March 2019		5,315	-		5,3
	Tata Consultancy Services Limited					
	31 March 2020		4,722		-	4.7
	31 March 2019		4,629			4,6
	Tata AIG General Insurance Co. Ltd.		de sel			
	31 March 2020					
	31 March 2019		686			6
	Tata Housing Development Company Ltd		000			
	31 March 2020		25			
	31 March 2019	2	- 2.5		1	
	Managerial remuneration				- 1000	1.5
	31 March 2020	-	14	-	1,133	1,1
	31 March 2019	-	1	10	1,008	1,0
2	Reimbursement of expenses (receivable)					
	31 March 2020	27	306	12		3
	31 March 2019	8	605			6
1	Tata Sons Private Limited					
- 5	31 March 2020	27	1.1		÷	
- 8	31 March 2019	8		4		
	Tata Housing Development Company Ltd					
5	31 March 2020	÷	301	1 (ex)	1.2	3
1	31 March 2019	-	605			6
1	Ewart Investment Limited					
1	31 March 2020		1	1. A.		
	31 March 2019	-			14	
	Tata Services Limited					
	31 March 2020		4			

### Related party disclosures (Continued)

	Transactions	Holding company	Fellow Subsidiaries	Joint Ventures	Key Management Personnel	Total
13	Services Received					_
	31 March 2020	10	939			9
	31 March 2019	5	712			7
	Tata AIG General Insurance Limited					,
	31 March 2020		238			2
	31 March 2019	- C	275			2
	Tata AIA Life Insurance Limited		215		5	2
	31 March 2020					
	31 March 2019		8			
	Tata Sons Private Limited	-	0		~	
	31 March 2020	10				
	31 March 2019	10	-	-		
		5	-	2	-	
	Ewart Investment Limited					
	31 March 2020		182	-	-	1
	31 March 2019	-	208	÷	-	2
	Tata Consulting Engineers Ltd					
	31 March 2020	-	500		1	5
	31 March 2019	-	158		÷	1
	Tata Elxsi Ltd.					
	31 March 2020		1.1			14
	31 March 2019		8		2.1	
	Tata Teleservices Ltd					
	31 March 2020	-	2		2	
	31 March 2019	-	20			
	Tata Housing Development Company Limited	-				
	31 March 2020					
	31 March 2019		35		3	
	Tata Consultancy Services Limited		22	-		
	31 March 2020		10			
	31 March 2019		16	-	-	
	31 March 2019					•
	Sale of porperty, plant & equipments					
ş	31 March 2020		~	1.0	-	
	31 March 2019	28	2	-		
1	Tata Sons Private Limited					
	31 March 2020	-	-		~	
ł	31 March 2019	28		147		
	Ewart Investments Limited	20				
	31 March 2020					
	31 March 2019	-	2			
ĺ	or march 2019	7	2	-		
	Deposit Refund Received					
100	31 March 2020	-	-	2		-
j,	31 March 2019	150	450	-		60
	Tata Sons Private Limited					
100	31 March 2020	4	- C	147	G	
1000	31 March 2019	150	4.1			15
	Ewart Investments Limited					
	31 March 2020	1		-	8	
	31 March 2019		450			45
Ĩ		-	100			43
	Employee Benefit Transfer					
	Tata Sons Private Limited					
	31 March 2020	1.0		161	12	Q.
3	31 March 2019	56	241		-	5
/	D					

Related party disclosures (Continued)

	Transactions	Holding company	Fellow Subsidiaries	Joint Ventures	Key Management Personnel	Total
17	Balances receivable at the year-end					
i)	Towards interest and debit notes and advances					
	31 March 2020		366	106		47
	31 March 2019		688	79		76
	Pune Solapur					
	31 March 2020	4		18		1
	31 March 2019	100		10		1
	TRIL IT4 Private Limited			10		
	31 March 2020	1.00		87	1	8
	31 March 2019			69		é
	Mikado Realtors Private Limited			09	-	L.
	31 March 2020			775		72
	31 March 2019			735		73
	Tata Consulting Engineers Ltd	-		273	-	27
	31 March 2020		14			
		÷	15		~	1
	31 March 2019	÷	35	-	~	3
	Tata Housing Development Company Limited		44.			
	31 March 2020	÷	319	1.1		31
	31 March 2019	1. A.	653	1.4		65
	Pune IT City Metro Rail Limited					
	31 March 2020		32	1.1	-	3
	31 March 2019	÷		Г÷	100	
-						
ii)	Deposits placed					
	31 March 2020	7	240	1.2		24
	31 March 2019	~	240			24
	Ewart Investments Limited					
	31 March 2020	~	240	1.41		24
	31 March 2019		240	1.00	- P	24
in.	Trade receivables					
u)	31 March 2020		740			
		23	740		14	76.
	31 March 2019		793	15		80
	Tata Consultancy Services Limited					
	31 March 2020	-	740	~	~	74
	31 March 2019	-	752	(*)	~	75
	TRIL IT4 Private Limited					
	31 March 2020		-	÷		1.1
	31 March 2019	-		15	~	1
	Tata Sons Private Limited					
	31 March 2020	23		-		2
	31 March 2019	1.2		-		
	Tata AIG General Insurance Company Ltd.					
	31 March 2020	-				
	31 March 2019	1.2	41	-		4
	Balances payable at the year-end					
	Towards advances from customers					
	31 March 2020		2,583	1		2,583
	31 March 2019		2,922	1	S	2,922
	Tata Consultancy Services Limited					
	31 March 2020	1.0	2,583	1	G.)	2,583
	31 March 2019	190	2,583	1		2,583
	Tata AIG General Insurance Company Ltd.					
						2.0
	31 March 2020	-				

### Related party disclosures (Continued)

### Summary of related party transactions

	Transactions	Holding company	Fellow Subsidiaries	Joint Ventures	Key Management Personnel	Total
ii)	Towards expenses					
	31 March 2020	9	166		-	174
	31 March 2019		44			44
	Tata Sons Pvt. Limited					
	31 March 2020	9				9
	31 March 2019				-	
	Tata Consultancy Services Limited					
	31 March 2020	÷	5			5
	31 March 2019	2	7	1 A A	-	7
	Tata Teleservices Limited					
	31 March 2020		- i			1
	31 March 2019					
	Tata Consulting Engineers Limited					
	31 March 2020	2	160		-	160
	31 March 2019		-			-
	Tata Housing Development Company Limited					
	31 March 2020	2.1				
	31 March 2019		19			19

R

Notes to the consolidated financial statements for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

Note 58 Additional information pursuant to paragraph 2 of Division II of Schedule III to the Companies Act 2013- 'General instructions for the preparation of consolidated financial statements' of Division II of Schedule III

## Part A March 2020

and a second	Net As (Total Assets - To		Share in profi	t or loss	Share in other com income	prehensive	Share in total comprehensive income		
Name of the entity	As a % of consolidated net assets	Amount	As a % of consolidated profit or loss	Amount	As a % of consolidated other comprehensive income	Amount	As a % of total comprehensive income	Amount	
Parent Tata Realty and Infrastructure Limited	56.02%	202.427	CD 1001	100 5051	the state of the state of the			in the	
Tata Realty and Infrastructure cifficed	56.02%	2,93,427	68.19%	(22,585)	60.92%	(2,675)	67.76%	(25,259	
Subsidiaries Indian				1.21					
Acme Living Solutions Private Limited	0.00%	(9)	0.00%	(1)	0.00%		0.00%	(1)	
Arrow Infra Estates Private Limited	0.07%	361	1.34%	(445)	0.00%	1.	1.19%	(445)	
Gurgaon Constructwell Private Limited	0.42%	2,189	2.71%	(898)	0.00%		2.41%	(898)	
Gurgaon Realtech Limited	0.26%	1,384	0.15%	(53)	0.00%		0.14%	(53)	
TRIL Roads Private Limited	4.99%	26,116	7.34%	(2,430)	2,99%	(131)	6.87%	(2,561)	
TRIF Gurgaon Housing Projects Private Limited	0.00%	(4)	0.00%	(1)	0.00%	-	0.00%	(1)	
TRIL Urban Transport Private Limited	1.35%	7,048	7.32%	(2,425)	4.90%	(215)	7.08%	(2,640)	
Wellkept Facility Management Services Private Limited	0.00%	(1)	0.00%	(0)	0.00%	10000	0.00%	(0)	
TRIL Constructions Limited	2.83%	14,801	1.91%	(634)	0.00%		1.70%	(634)	
TRIL Infopark Limited	8.37%	43,825	-17.83%	5,905	-0.20%	9	-15.86%	5,914	
TRIL Amritsar Projects Limited	0.00%		0.00%	1.1	0.00%	1	0.00%	1913	
Hampi Expressway Private Limited	5.07%	26,544	0.47%	(156)	0.00%	+	0.42%	(156)	
TRIF Real Estate Development Limited	0.00%		0.00%	10 m	0.00%		0.00%	1	
HV Farms Private Limited	0.22%	1,147	0.00%	(1)	0.00%	-	0.00%	(1)	
International Infrabuild Private Limited	-0.36%	(1,905)	28.58%	(9,465)	0.00%		25.39%	(9,465)	
Dharamshala Ropeway Limited	0.11%	590	0.49%	(163)	0.00%		0.44%	(163)	
Manali Ropeway Private Limited	0.03%	138	0.27%	(90)	0.00%		0.24%	(90)	
Uchit Expressways Private Limited	3.53%	18,492	-3.54%	1,173	0.10%	(4)	-3.14%	1,169	
TRPL Roadways Private Limited	2.30%	12,052	3.87%	(1,281)	31.33%	(1,375)	7.13%	(2,656)	
Durg Shivnath Expressways Pvt Ltd	1.91%	10,024	-2.50%	828	-0.01%	1	-2.22%	828	
Matheran Ropeway Pvt Ltd	0.47%	2,462	-0.10%	35	0.00%		-0.09%	35	
MIA Infrastructure Private Limited	0%	(31)	0,00%	(1)	0.00%		0.00%	(1)	
Non-controlling interests in all subsidiaries	1.22%	6,384	0.70%	(233)	0.00%	-	0.00%	-	
Joint Venture (investment as per equity method) Indian				1.1					
Pune Solapur Expressways Private Limited	0.43%	2,248	-0.24%	78	0.00%		-0.21%	78	
A & T Road Construction Management and Operation Private Limited	0.00%	1	0%	(1)	0.00%		0.00%	(1)	
FRIL IT4 Private Limited	0.00%	2	0.00%	14/	0.00%		0.00%	(1)	
Mikado Realtors Pvt. Ltd.	5,79%	30,335	0.30%	(101)	0.00%		0.27%	(101)	
ndustrial Minerals and Chemicals Company Private Limited	4.19%	21,950	-0.12%	40	0.00%	-	-0.11%	40	
Pune IT City Metro Rail Limited	0.81%	4,226	0.65%	(215)	0.00%		0.58%	(215)	
Fotal	100.00%	5,23,794	100.00%	(33,120)	100.00%	(4,390)	100.00%	(37,277)	
Consolidation Adjustment		(4,10,769)		(3,136)		4,392		1,023	
fotal		1,13,025		(36,256)		2		(36,254)	



### Part B

Name of the entity	Reporting Currency	Exchange Rate	Share Capital	Reserve and Surplus	Total Assets	Total Liabilities	Investment	Turnover / Total Revenue	Total Income	Profit Before Tax	Provision for Taxation			% of Shareholding
Parent												1 1		
Tata Realty and Infrastructure Limited	INR	1.00	1,01,731	1,91,696	6,26,354	(3,32,928)	5,01,951	17.585	32,551	(20,167)	(2,418)	(22,585)		
Subsidiaries					C							1.00		
Indian														1
Acme Living Solutions Private Limited	INR	1.00	5	(14)	0	(9)			1	a		(I)		100.00%
Arrow Infra Estates Private Limited	INR	1.00	10	350	23,526	(23.165)		448	456	(445)		(445)		100.00%
Gurgaon Constructwell Private Limited	INR	1.00	7	2,182	11.691	(9.502)				(898)	(102)	(796)		100.00%
Gurgaon Realtech Limited	INR	1.00	7	1.378	32,898	(31,514)		1,095		(53)	(102)	(53)		100.00%
TRIL Roads Private Limited	INR	1.00	775	25,341	95.367	(69,252)	85,422	3,040	3.545	(2,430)	675	(3,105)		100.00%
TRIF Gurgaon Housing Projects Private Limited	INR	1.00	5	(9)	0	(4)				(1)	015	(1)		100.00%
TRIL Urban Transport Private Limited	INR	1.00	7,124	(75)	12,093	(5,044)	7.844		97	(2,425)	(158)	(2,267)	1.81	100.00%
Wellkept Facility Management Services Private Limited	ENR	1.00	40	(41)	6	(7)				(0)	1,000/	(0)	1.2.1	100.00%
TRIL Constructions Limited	INR	1.00	3.615	11.186	24,128	(9,328)	1 1 2 1	1 21		(634)	10.000	(634)	11.8.1	67.50%
TRIL Infopark Limited	INR	1.00	1.05.000	(61,175)	3.64,993	(3,21,168)	211	56,821	57.574	5.905		5,905	1.8.1	100.00%
Hampi Expressway Private Limited	INR	1.00	5,313	21,231	1.57,146	(1.30,602)		27,401	27,402	(156)	(276)	120	181	100.00%
HV Farms Private Limited	INR	1.00	1	1,146	1,156	(9)		and a second		(1)	12707	(1)	1.84	100.00%
International Infrabuild Private Limited	INR	1.00	10	(1.915)	5,969	(7,374)		476	488	(9,465)	-94	(9.559)	11.21	26.00%
Dharamshala Ropeway Limited	INR	1.00	360	230	5,464	(7.574)	1		7	(163)	(30)	(133)	11-57	74.00%
Manali Ropeway Private Limited	INR	1.00	36	103	5,777	(5,638)				(90)	(22)	(68)	1.1	72.00%
Uchit Expressways Private Limited	INR	1.00	13,312	5,180	1,40,552	(1.22,061)		48.041	48.043	1.173	205	968		100.00%
TRPL Roadways Private Limited	INR	1.00	5	12,047	25,889	(13,836)	1.1.1		0	(1,281)		(1.281)		100.00%
Durg Shivnath Expressways Pvt Ltd	INB	1.00	2,362	7,662	31,806	(21,782)	1.1	5,728	6.167	828	311	517		100.00%
Matheran Ropeway Pvi Ltd	INR	1.00	750	1,712	2.574	(112)		-	58	35	9	26	1.21	70.00%
MIA Infrastructure Private Limited	INR	1.00	277	(308)	6	(38)	8	100		(1)	0.00	(1)		100.00%
Fotal			2,40,745	2,17,907	15,70,395	(11,11,747)	5,95,428	1,60,635	1,76,383	(30,270)	(1,712)	(33,394)	10.11	

### Associates and Joint Ventures

Name of the Associates / Joint Ventures	Latest audited Balance Sheet date	No.Shares of the Associate / Joint Ventures held by the Company	Amount of Investment in Associates / Joint Venture	Extent of Holding (%)	Description of how there is significant influence	Reason why the Associate / Joint Venture is not consolidated	Networth attributable to Sharebolding as per latest audited Balance Sheet	Profit / Loss for the year		
	777			1				Considered in Comolidation	Not Considered in Consolidation	
Joint Ventures										
Pane Solapur Expressways Private Limited	31.03.2020	23,86,711	001,01	50	There is significant influence due to shareholding and joint control over the communic activities indirectly through subsidiary company.		2,250	78		
A & T Road Construction Management and Operation Private Limited	31.03.2020	30,000	3	50		The entity is consolidated by way of equity accounting		-10		
TRIL IT4 Private Limited	11.03.2020	7.40,000	74	74		The entity is consolidated by way of equity accounting	(20,919)		(4,104)	
Mikado Realton Private Limited	31.03.2020	1,99,87,400	1,999	74			3,194	(101)		
Industrial Minerals and Chemicals Company Private Limited	31.03.2020	3,256	3	74			13,558	40		
Puse IT City Metro Rail Tamild	31.03.2020	4,44,07,400	4,441	74			4,226	(215)		

### Notes to the consolidated financial statements (Continued)

for the year ended 31 March 2020 (Currency: Indian rupees in lakhs)

### Note 59

### Events after the balance sheet date

There are no significant subsequent events that would require adjustments or disclosures in the financial statements as on the balance sheet date.

### Note 60

World Health Organisation (WHO) declared outbreak of Coronavirus Disease (COVID-19) a global pandemic on March 11, 2020. Consequent to this, Government of India declared lockdown on March 25, 2020 and the Group suspended its operations in all ongoing projects in compliance with the lockdown instructions issued by the Central and State Governments. COVID-19 has impacted the normal business operations of the Group during the lock-down period.

The Group has made detailed assessment of its liquidity position for the next year and the recoverability and carrying value of its assets comprising property, plant and equipment, intangible assets, investments, inventory, advances, trade receivables, Deferred taxes, other financial and non-financial assets etc. Based on current indicators of future economic conditions, the Group expects to recover the carrying amount of these assets. The situation is changing rapidly giving rise to inherent uncertainty around the extent and timing of the potential future impact of the COVID-19 pandemic, which may be different from that estimated as at the date of approval of these financial statements.

The Central and State Governments have initiated steps to lift the lockdown and the Group will adhere to the same as it resumes its activities. Construction at sites has already restarted. Since it is only about thirteen weeks into the pandemic, the Group will continue to closely observe the evolving scenario and take into account any future developments arising out of the same.

### Note 61

### **Previous Year's Figures**

Previous year figures have been regrouped / reclassified to conform to current year presentation, wherever considered necessary.

For and on behalf of the Board of Directors of Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

1 hubedan F. N. Subedar

Banmali Agrawala Chairman DIN - 00120029

Sanjay Dutt Managing Director DIN - 05251670

ann

Sanjay Sharma Chief Financial Officer

DIN - 00028428

Director

Sudhakar Shetty Company Secretary Membership No: A13200

Mumbai 06 July, 2020

# Statement pursuant to first proviso to sub-section (3) of section 129 of the Companies Act, 2013, read with rule 5 of Companies (Accounts) Rules, 2014 in the prescribed Form AOC-1 Statement containing salient features of the financial statement of Subsidiaries / Associate companies / Joint Ventures Part A: Subsidiaries

-	Name of the subsidiary	1.0.1.0.0	1	12												(₹ crores
Sr. No.		The date since when subsidiary was acquired	Reporting period for the subsidiary concerned, if different from the holding company's reporting period	Reporting currency	Exchange rate as on the last date of the relevant financial year in case of foreign subsidiaries	Share Capital	Reserves & Surplus	Total Assets	Total Liabilities	Investments	Turnover	Profit before taxation	Provision for taxation	Profit after taxation	Proposed dividend	% of shareholding
A	Tata Realty and Infrastructure Limited	NA	31-Mar-2020	Indian Rupees	1.00	1.017.31	1.916.96	6.263 54	3,329,28	5.019.51	175.85	-201.67	-24.18	-225.85	-	-
100	Subsidiaries:						1				erester.		L TILL	223.03	-	-
1	Acme Living Solutions Private Limited	27-Jan-2009	31-Mar-2020	Indian Rupees	1.00	0.05	-0.14	0.00	0.09			(0.01)		(0.01)	-	100.00%
2	Arrow Infra Estates Private Limited	5-Jan-2009	31-Mar-2020	Indian Rupees	1.00	0.10	3.50	235.26	231.65		4.48	(4.45)		(4.45)		100.00%
3	Gurgaon Constructwell Private Limited	5-Jan-2009	31-Mar-2020	Indian Rupees	1.00	0.07	21.82	116.91	95.02			(8.98)	(1.02)	(7.96)		100.00%
4	Gurgaon Realtech Limited	5-Jan-2009	31-Mar-2020	Indian Rupees	1.00	0.07	13.78	328.98	315.14		10.95	(0 53)	12:021	(0.53)		100.00%
5	TRIL Roads Private Limited	18-Mar-2008	31-Mar-2020	Indian Rupees	1.00	7.75	253.41	953.67	692.52	854.22	30.40	(24.30)	6.75	(31.05)		100.00%
6	TRIF Gurgaon Housing Projects Private Limited	24-Jun-2009	31-Mar-2020	Indian Rupees	1.00	0.05	(0.09)	0.00	0.04		20.10	(0.01)		(0.01)		100.00%
7	TRIL Urban Transport Private Limited	25-May-2007	31-Mar-2020	Indian Rupees	1.00	71.24	(0.75)	120.93	50.44	78.44	~ ~ ~	(24.25)	(1.58)	(22.67)		100.00%
В	Wellkept Facility Management Services Pvt. Ltd.	17-Jan-2011	31-Mar-2020	Indian Rupees	1.00	0.40	(0.41)	0.05	0.07	10.11	-	(0.00)	(1.50)	(0.00)	-	100.00%
9	TRIL Constructions Limited	25-Jun-2007	31-Mar-2020	Indian Rupees	1.00	36,15	111.86	241.28	93.28			(6.34)		(6.34)	11 2121	67,50%
10	TRIL Infoperk Limited	22-Mar-2015	31-Mar-2020	Indian Rupees	1.00	1.050.00	(611.75)	3,649.93	3.211.68	2.11	568.21	59.05		59.05		83.85%
11	Hampi Expressway Private Limited	23-Apr-2015	31-Mar-2020	Indian Rupees	1.00	53.13	212.31	1.571.46	1,306.02		274.01	(1.56)	(2.76)	1.20	-	100.00%
12	HV Farms Private Limited	25-Apr-2012	31-Mar-2020	Indian Rupees	1.00	0.01	11.46	11 56	0.09			(0.01)	(All of	(0.01)	-	100.00%
13	International Infrabuild Private Limited	1-Apr-2016	31-Mar-2020	Indian Rupees	1.00	0.10	(19.15)	59,69	78.74		4.75	(94.65)	0.94	(95.59)		25.00%
14	Dharamshala Ropeway Limited	8-May-2015	31-Mar-2020	Indian Rupees	1.00	3.60	2.30	84.64	78.74			(1.63)	(0.30)	(1.33)		74.00%
15	Manali Ropeway Private Limited	19-Oct-2015	31-Mar-2020	Indian Rupees	1.00	0.36	1.03	57.77	56.38			(0.90)	(0.22)	(0.68)		72.00%
16	Uchit Expressways Private Limited	10-Oct-2016	31-Mar-2020	Indian Rupees	1.00	133.12	51.80	1,405.52	1,220.61		480.41	11 73	2.05	9.68	-	100.00%
17	TRPL Roadways Private Limited	3-Nov-2016	31-Mar-2020	Indian Rupees	1.00	0.05	120.47	258.89	138.36			(12.81)	2.05	(12.81)	-	100.00%
18	Durg Shivnath Expressways Pvt. Ltd.	25-Apr-2017	31-Mar-2020	Indian Rupees	1.00	23.62	76.62	318.06	217.82		57.28	8.28	3.11	5.17		100.00%
19	Matheran Ropeway Pvt. Ltd.	19-Jun-2017	31-Mar-2020	Indian Rupees	1.00	7.50	17.12	25.74	1.12	-		0.35	0.09	0.26		70.00%
20	MIA Infrastructure Private Limited	17-Oct-2017	31-Mar-2020	Indian Rupees	1.00	2.77	(3.08)	0.06	0.38			(0.01)	0.03	(0.01)		100.00%

\* % of share holding of the Company and its subsidiaries

### Additional details:

Banmali Agri Chairman DIN - 001200 Date: 06 July, Place: Mumb

nuona eesaan: Name of subsidiaries which are yet to commence operation TRIF Gurgaon Housing Projects Pvt Limited Acme Living Solutions Pvt Limited Wellkept Facility Management Services Private Limited MIA infrastructure Private Limited

3 4

Name of subsidiaries which have been liquidated or sold during the year
 TRIL Amritsar Projects Limited (upto December 09, 2019)
 TRIF Real Estate and Development Limited (upto December 09, 2019)

For Tata Realty and Infrastructure Limited CIN No: U70102MH2007PLC168300

DUNTA

utt Sanjay Dutt Managing Director DIN - 05251670

J. 1 hubedar N F, N. Subedar Director DIN - 00028428

Sanjay Sharma Chief Financial Officer

adhakar Shetty

Company Secretary Membership No: A13200

### Part B: Associates and Joint Ventures Statement pursuant to Section 129(3) of the Co es Act, 2013 rel ted to Associate Companies and Joint Ventures

Name of the entity	Latest audited balance sheet date	Date on which the Associate or Joint Venture was associated or acquired	Reporting	Shares of the Associate / Joint Ventures held by the Company as on March 31, 2020			Description of how there is significant influence	Reason why the Associate / Joint Venture	Networth attributable to	Profit / Loss for the year		
				No of shares held by the company in associate / joint venture *	Amount of investment in associate / joint venture* ( <pre>crores</pre>	Extent of holding (in percentage)*		is not consolidated	shareholding as per latest balance sheet (₹ crores)		Not Considered In Consolidation (₹ crores)	
Joint Ventures												
1 Pune Solapur Expressways Private Limited	31-Mar-2020	20-Mar-09	Indian Rupees	23,86,711	103.00	50.00%	There is significant influence due to shareholding and joint control over the economic activities indirectly through subsidiary company	The entity is consolidated by way of equity accounting	22.50	0.78		
2 A & T Road Construction Management and Operation Private Limited	31-Mar-2020	02:May-13	Indian Bupees	30,000	0.03	50.00%	There is significant influence due to shareholding and joint control over the economic activities indirectly through subsidiary company.	The entity is consolidated by way of equity accounting	0.01	(0.01)	1	
3 TRIL IT4 Private Limited	31-Mar-2020.	17-Dec-14	Indian Rupees	7,40,000	0.74	74.00%	There is significant influence due to shareholding and joint control over the economic activities indirectly through subsidiary company.	The entity is consolidated by way of equity accounting.	(209.19)		(41.04)	
4 Mikado Realtors Private Limited	31-Mar-2020	07-Sep-16	Indian Rupees	1,99,87,400	19.99	74.00%	There is significant influence due to shareholding and joint control over the economic activities indirectly through subsidiary company.	The entity is consolidated by way of equity accounting	31.94	(1.01)		
5 Industrial Minerals and Chemicals Company Private Limited	31-Mar-2020	11-Mar-17	Indian Rupees	3,256	0.03	74.00%	There is significant influence due to shareholding and joint control over the economic activities indirectly through subsidiary company	The entity is consolidated by way of equity accounting	135.58	0.40	ત્	
6 Pune IT City Motro Rail Limited	31-Mar-2020	28-May-19	Indian Rupees	4,44,07,400	44.41	74.00%	There is significant influence due to shareholding, and joint control over the economic activities indirectly through subsidiary company.	The entity is consolidated by way of equity accounting	42.26	(2.15)	-	

\* No direct holding by TRIL of the shares of Pune Solapur Expressways Private Limited, A & T Road Construction Management and Operation PVI. Limited and Pune IT City Metro Rail Limited.

Shares of Pune Solapur Expressways Private Limited is held a	Number	Amount of investment	Extent of Holding (%)	Shares of A & T Road Construction Management and Operation Private Ltd is held as under;	Number		Amount of investment in Joint	Extent of Holding
		00013 (3.3.4	0.4				Venture	(961
TRIL Roads Private Limited (WOS of TRIL)	23,86,711	103.00	SOW	TRUL Roads Private Limited (WOS of TRU)		30,000	0.03	50%
Atlantia SpA	23,86,711	103.00	50%	Autostrade Indian infrastructure Development Private Limited		30,000	0.03	50%
Total	47,73,422	296.00	100%	Total	- 6	60,000	0.06	100%
Shares of Pune IT City Metro Rail Limited is held as under.								
	Number	Amount of Investment In Joint Venture	Extent of Holding (%)					
TRIL Urban Transport Pvt Limited (WOS of TRIL)	4,44,07,400	44.41	74%					
Seimens Project Venture GMBH	1,56,02,600	15.60	2.5%					
Total	6,00,10,000	60.01	100%	1 Juda C				
Chairman Mar	Jay Dutt naging Director - 05251670		J. F. N. Subedar Director DIN - 00028428	Saujay Sharma Chief Financial Office Statistics Chief Financial Office				